

# **Biomek Software**

Version 3.2 User's Manual

Beckman Coulter PN 987835 Revision AC September 2004

Beckman Coulter, Inc 4300 N. Harbor Boulevard, Fullerton, CA 92834-3100 Copyright © 2004 Beckman Coulter, Inc. Printed in U.S.A. Except as provided in writing signed by an officer to Beckman Coulter, Inc., this system and any related documentation are provided "as is" without warranty of any kind, expressed or implied, including that the system is "error free." This information is presented in good faith, but Beckman Coulter, Inc. does not warrant, guarantee, or make any representations regarding the use or the results of the use of this system and related documentation in terms of correctness, accuracy, reliability, currentness, omissions, or otherwise. The entire risk as to the use, results, and performance of this system and related documentation is assumed by the user.

Except as expressly provided herein, Beckman Coulter, Inc. makes no other warranty, whether oral or written, expressed or implied, as to any matter whatsoever, including but not limited to those concerning merchantability and fitness for a particular purpose, nor is freedom from any patent owned by Beckman Coulter, Inc. or by others to be inferred.

#### LIMITATIONS OF LIABILITY

Beckman Coulter, Inc. shall not be liable, to any extent whatsoever, for any damages resulting from or arising out of the use or performance of this system and related documentation or the procedures specified in this manual, regardless of foreseeability or the form of action, whether in contract, tort (including negligence), breach of warranty, strict liability or otherwise, and including but not limited to damages resulting from loss of data, loss of anticipated profits, or any special, indirect, incidental or consequential damages. In no event shall Beckman Coulter, Inc.'s liability to the user exceed the amount paid by the user to Beckman Coulter, Inc. hereunder. The user assumes full responsibility for the results obtained from the use of this system and related documentation and for application of such results.

Beckman Coulter, SAGIAN Operations Indianapolis, Indiana www.beckmancoulter.com

Biomek, SAMI, Orca, and the Beckman Coulter logo are registered trademarks and Beckman Coulter and SAGIAN are trademarks of Beckman Coulter, Inc.

Microsoft and MS-DOS are registered trademarks and Windows is a trademark of Microsoft Corporation. All other trademarks appearing in this manual are owned by their respective companies.

Copyright 2004 by Beckman Coulter, Inc. All rights reserved. No part of this publication may be reproduced, distributed, or transmitted in any form or by any means, electronic, mechanical, photocopying, recording, or otherwise, or stored in a database or retrieval system, without the prior written permission of Beckman Coulter, Inc.

#### Warranty and Returned Goods Requirements

All standard Beckman Coulter, Inc. policies governing returned goods apply to this product. Subject to the exceptions and upon the conditions stated below, the Company warrants that the products sold under this sales agreement shall be free from defects in workmanship and materials for one year after delivery of the products to the original Purchaser by the Company, and if any such product should prove to be defective within such one year period, the Company agrees, at its option, either (1) to correct by repair or at the Company's election by replacement, any such defective product provided that investigation and factory inspection discloses that such defect developed under normal and proper use, or (2) to refund the purchase price. The exceptions and conditions mentioned above are as follows:

- a. Components or accessories manufactured by the Company which by their nature are not intended to and will not function for one year are warranted only to reasonable service for a reasonable time. What constitutes a reasonable time and a reasonable service shall be determined solely by the Company. A complete list of such components and accessories is maintained at the factory.
- b. The Company makes no warranty with respect to components or accessories not manufactured by it. In the event of defect in any such component or accessory, the Company will give reasonable assistance to Purchaser in obtaining from the manufacturer's own warranty.
- c. Any product claimed to be defective must, if required by the Company, be returned to the factory, transportation charges prepaid, and will be returned to Purchaser with transportation charges collect unless the product is found to be defective, in which case the product must be properly decontaminated of any chemical, biological, or radioactive hazardous material.
- d. The Company shall be released from all obligations under all warranties, either expressed or implied, if any product covered hereby is repaired or modified by persons other than its own authorized service personnel, unless such repair by others is made with the written consent of the Company.
- e. If the product is a reagent or the like, it is warranted only to conform to the quantity and content and for the period (but not in excess of one year) stated on the label at the time of delivery.

It is expressly agreed that the above warranty shall be in lieu of all warranties of fitness and of the warranty of merchantability, and that the company shall have no liability for special or consequential damages of any kind or from any cause whatsoever arising out of the manufacture, use, sale, handling, repair, maintenance, or replacement of any of the products sold under the sales agreement.

Representatives and warranties made by any person, including dealers and representatives of the Company, which are consistent or in conflict with the terms of this warranty, shall not be binding upon the Company unless reduced in writing and approved by an expressly authorized officer of the Company.

Parts replaced during the warranty period are warranted to the end of the instrument warranty.

**Note:** Performance characteristics and specifications are only warranted when Beckman Coulter replacement parts are used.

## **Safety Information**

All Warnings and Cautions in this document include an exclamation point, a lightning bolt, or a light burst symbol framed within a triangle. Please pay special attention to the specific safety information associated with these symbols.



WARNING: If the equipment is used in a manner not specified by Beckman Coulter, Inc., the protection provided by the equipment may be impaired.

### Warning and Caution Definitions



The exclamation point symbol is an international symbol which serves as a reminder that all safety instructions should be read and understood before installation, use, maintenance, and servicing is attempted.

When this symbol is displayed in this manual, pay special attention to the specific safety information associated with the symbol.

#### WARNING

A WARNING calls attention to a condition or possible situation that could cause injury to the operator.

#### CAUTION

A CAUTION calls attention to a condition or possible situation that could damage or destroy the product or the operator's work.

#### **Electrical Safety**

To prevent electrically related injuries and property damage, properly inspect all electrical equipment prior to use and immediately report any electrical deficiencies. Contact a Beckman Coulter Service Representative for any servicing of equipment requiring the removal of covers or panels.

#### **High Voltage**



This symbol indicates the potential of an electrical shock hazard existing from a high voltage source and that all safety instructions should be read and understood before proceeding with the installation, maintenance, and servicing of all modules.

Do not remove system covers. To avoid electrical shock, use supplied power cords only and connect to properly grounded (three-holed) wall outlets. Do not use multiplug power strips.

#### Laser Light



This symbol indicates that a potential hazard to personal safety exists from a laser source. When this symbol is displayed in this manual, pay special attention to the specific safety information associated with the symbol.

#### **Laser Specifications**

Laser Type:	Class II Laser Diode
Maximum Output:	1mW
Wavelength:	670 nm

#### **Chemical and Biological Safety**

Normal operation of laboratory equipment may involve the use of materials that are toxic, flammable, or otherwise biologically harmful. When using such materials, observe the following precautions:

- Handle infectious samples according to good laboratory procedures and methods to prevent the spread of disease.
- Observe all cautionary information printed on the original solutions containers prior to their use.
- Dispose of all waste solutions according to your facility's waste disposal procedures.
- Operate the Biomek instrument in accordance with the instructions outlined in this manual, and take all the necessary precautions when using pathological, toxic, or radioactive materials.
- Objects dropped onto plates, accidental tool release, or other accidental collisions may result in splashing of liquids; therefore, take appropriate safety precautions, such as using safety glasses and wearing protective clothing, when working with potentially hazardous liquids.
- Use an appropriately contained environment when using hazardous materials.
- Observe the appropriate cautionary procedures as defined by your safety officer when using flammable solvents in or near a powered-up instrument.
- Observe the appropriate cautionary procedures as defined by your safety officer when using toxic, pathological, or radioactive materials.

**Note:** Observe all warnings and cautions listed for the Biomek instrument and any external devices attached or used during operation of the instrument. Refer to the applicable instrument and external device user's manuals for operating procedures of that device.

#### **Moving Parts**

To avoid injury due to moving parts, observe the following:

- Never attempt to exchange labware, reagents, or tools while the instrument is operating.
- Never attempt to physically restrict any of the moving components of the Biomek instrument.
- Keep the Biomek instrument work area clear to prevent obstruction of the movement.

### Cleaning

Observe the cleaning procedures outlined in the user's manual for the Biomek instrument, automated labware positioner (ALP), or external device. Prior to cleaning equipment that has been exposed to hazardous material:

- Appropriate Chemical and Biological Safety personnel should be contacted.
- The Chemical and Biological Safety information contained in this user's manual should be reviewed.

### Maintenance

Perform only the maintenance described in this manual or the user's manual for the Biomek instrument. Maintenance other than that specified in these manuals should be performed only by Beckman Coulter Service Representatives.



It is your responsibility to decontaminate the Biomek instrument or any of its accessories before requesting service by a Beckman Coulter Service Representative or returning parts to Beckman Coulter for repair. Beckman Coulter will NOT accept any items which have not been decontaminated where it is appropriate to do so. If any parts are returned, they must be enclosed in a sealed plastic bag stating that the contents are safe to handle and are not contaminated.

### Warnings and Cautions Found in this Manual

Please read and observe all cautions and instructions. Remember, the most important key to safety is to operate the Biomek instrument and accessories with care.

The WARNINGs and CAUTIONs found within this document are listed below.

**Note:** Also refer to the warnings and cautions in the applicable Biomek instrument user's manual for additional safety information.



WARNING: If the equipment is used in a manner not specified by Beckman Coulter, Inc., the protection provided by the equipment may be impaired.



CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height and failure to choose the correct ALP in the Deck Editor may result in collisions between pod(s) and ALPs during operation.



CAUTION: After new tips are added, the correct properties must be defined in the Tip Type Editor according to the manufacturer's specifications. Inaccurate specifications may lead to hardware crashes.



CAUTION: The correct well properties must be defined in Labware Types according to the manufacturer's specifications. Inaccurate specifications may lead to inaccurate pipetting.



CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height and failure to choose each ALP correctly in the software may result in collisions between pod(s) and ALPs during operation.



CAUTION: No changes to the Biomek state are permitted while a method is paused. Changes can be made to the labware contents, but not the deck or the devices.



WARNING: The light curtain is a safety device. Use it to stop a method only in an emergency.



CAUTION: An inaccurate Instrument Setup may result in pod and labware collisions, or in inappropriate pipetting.



CAUTION: To make sure the Biomek pod avoids all obstacles during its travel, always specify the labware, ALPs, and devices that are on the deck, and the deck position each occupies.



CAUTION: Do not place labware other than a tip box on a tip loader position (TL#).



CAUTION: Do not purge the system without mandrels installed and tubing attached to disposable or fixed tips. Purging the system without the mandrels installed and the tubing attached to tips may cause corrosion in the tip interface.



CAUTION: Do not attempt to access a 96-Channel or 384-Channel Tip Wash ALP with a Multichannel Pod equipped with an HDR Tool Body. The gripper may crash and damage the pod, HDR Tool Body, or Tip Wash ALP.



CAUTION: Do not access labware positioned on a 1 x 5 Passive ALP with the HDR Tool Body. The gripper may crash into the ALP.



CAUTION: Do not access labware on a Stirring ALP with the HDR Tool Body. The magnetic stirrer may bend the pins or interfere with the liquid transfer performance of the pins.



CAUTION: When setting the height for pipetting operations on the Stirring ALP, the presence of the stir bar must be taken into consideration. Pipetting operations that do not account for the height of the stir bar could damage the tips.



WARNING: The Ignore error recovery option is potentially dangerous since almost every action depends upon the successful completion of previous actions. Choose Ignore at your own risk.



CAUTION: Resuming a method assumes that the Biomek instrument is in the same state as when the error occurred. The pod may be moved to deal with a problem, but no changes can be made to the Biomek instrument deck.



CAUTION: The partially completed steps assume the pods are in the same position they were in when the method was halted. Removing the Recovery step without also removing these partially completed steps could result in a crash.



WARNING: To eliminate the possibility of replacing a Continuation method before corrections are made, it is recommended that a Continuation method be edited and run as soon as it is created.



CAUTION: A Continuation method assumes that the instrument is in the same state as when the Continuation was snapped. If this is not true, the Continuation method must be edited to account for the changes.

## **Table of Contents**

	Warranty	and Returned Goods Requirements iii
Saf	ety Info	rmationv
	Warning	and Caution Definitions
	Electrical	Safetyvi
	Hig	h Voltage vi
	Laser Lig	ht vi
	Las	er Specifications vi
	Chemical	and Biological Safetyvii
	Moving F	'artsvii
	Cleaning	viii
	Maintena	nce viii
	Warnings	and Cautions Found in this Manual
Tab	ole of Co	ontents
Tab List	ole of Co t of Figu	ontents
Tab List List	ble of Co t of Figu t of Tabl	ontentsxi ıresxiii lesxxxv
Tab List List	ole of Co t of Figu t of Tabl Biomel	ontentsxi resxiii esxxiii s Software Introduction
Tab List List	ble of Co t of Figu t of Tabl Biomel	ontents xi   ires xxiii   es xxxv   x Software Introduction   bout Biomek Software 1-1
Tab List List	ole of Co t of Figu t of Tabl Biomel 1.1 A 1.	ontents       xi         ores       xxiii         es       xxxv         x Software Introduction       1-1         bout Biomek Software       1-1         1.1       Instrument Files       1-1
Tab List List	ble of Co t of Figu t of Tabl Biomel 1.1 A 1. 1.	with the second state of the second
Tab List List	ble of Co t of Figu t of Tab Biomel 1.1 A 1. 1. 1.	with the second seco
Tab List List	ble of Co t of Figu t of Tabl Biomel 1.1 A 1. 1. 1.2 A	ires       xxiii         es       xxxv         x Software Introduction       xxxv         bout Biomek Software       1-1         1.1       Instrument Files       1-1         1.2       Project Files       1-2         1.3       Methods       1-2         bout this Manual       1-3

#### 2 Using Accounts & Permissions

2.1	Overvie	2W	2-1
2.2	Installir	ng and Setting the Level of Support For Accounts & Permissions	2-2
	2.2.1	Installing Accounts & Permissions	2-2
	2.2.2	Setting the Level of Support For Accounts & Permissions	2-3
2.3	Admini	stering User Accounts and Permissions.	2-5
	2.3.1	Assigning Biomek Software Permissions	2-6
2.4	Restori	ng the Administrator Password	2-7
2.5	Viewing	g the Audit Log	2-8
2.6	Using the	he Biomek Software With Accounts & Permissions Enabled	2-9
	2.6.1	Logging Onto the Biomek Software	2-9
	2.6.2	Logging Off the Biomek Software	. 2-11

#### 3 Using Instrument Files and Settings

3.1	Overvi	iew	. 3-1
	3.1.1	Instrument Display in Method Editor	. 3-2
	3.1.2	Contents of an Instrument File	. 3-2
3.2	Saving	and Opening Instrument Files	. 3-3
	3.2.1	Saving an Instrument File	. 3-3
	3.2.2	Creating a New Instrument File	. 3-3
	3.2.3	Opening an Existing Instrument File	. 3-4
3.3	Restori	ing Instrument Settings from Backup Files	. 3-5
3.4	Sharing	g Instrument Settings Using the Import/Export Utility	. 3-6
	3.4.1	Exporting Settings to an Import File.	. 3-7
	3.4.2	Importing Settings from an Import File	. 3-9
	3.4.3	Restoring Instrument Settings from Backup Files	3-10

#### 4 Configuring Hardware Setup

4.1	Overview	<i>w</i>	-1
4.2	Accessir	ng Hardware Setup 4-	-2
4.3	Understa	anding the Options in Hardware Setup 4-	-3
4.4	Configu	ring the Biomek Simulator 4-	-4
	4.4.1	Configuring Simulator Settings 4-	-4
	4.4.2	Configuring Camera Controls 4-	-9
	4.4.3	Configuring Materials 4-1	1

#### 5 Preparing and Managing the Deck

5.1	Overvi	ew
	5.1.1	Framing the Deck
5.2	Unders	tanding the Deck Editor
	5.2.1	Viewing the ALP Types List 5-6
	5.2.2	Biomek FX Deck
	5.2.3	Biomek 3000 Deck
	5.2.4	Biomek NX-MC Deck 5-8
	5.2.5	Biomek NX-S8 Deck 5-9

	5.3	Opening, Selecting, Creating, Deleting, or Renaming a Deck	5-10
		5.3.1 Opening and Selecting a Deck	5-10
		5.3.2 Creating a New Deck	5-11
		5.3.3 Deleting a Deck	5-12
		5.3.4 Renaming a Deck	5-13
	5.4	Modifying a Deck	5-14
		5.4.1 Deleting ALPs and Deck Positions from a Deck	5-14
		5.4.2 Adding ALPs and Deck Positions	5-15
		5.4.3 Setting ALP Properties and Deck Positions	5-16
6	Und	erstanding and Using Project Files	
	6.1	Overview	6-1
		6.1.1 Contents of a Project File	6-3
	6.2	Creating a New Project File	6-4
	6.3	Opening a Project File	6-5
	6.4	Deleting and Restoring Project Files	6-6
		6.4.1 Deleting Projects.	6-6
		6.4.2 Restoring Projects	6-7
	6.5	Renaming a Project File.	6-8
		6.5.1 Renaming the Active Project File	6-8
		6.5.2 Renaming a Saved Project File	6-9
	6.6	Creating a Copy of a Project File.	6-10
	6.7	Checking In a Project File	6-11
	6.8	Viewing Project Contents	6-13
		6.8.1 Restoring Deleted Project Items	6-14
	6.9	Reverting Projects	6-15
		6.9.1 Reverting Project Contents to a Certain Date and Time	6-15
		6.9.2 Discarding All Changes Since Last Check In.	6-17
		6.9.3 Reverting Individual Project Items to a Specific Revision	6-19
	6.10	Importing and Exporting Project Files.	6-20
		6.10.1 Exporting a Project File to a Biomek Import File	6-20
		6.10.2 Importing a Project File from a Biomek Import File	6-22
7	Cros	ating and Modifying Tip and Labware Types	
1	7 1		7_1
	7.1	Defining Ting	/-1
	1.2	7.2.1 Accessing the Tip Type Editor	· · · 7-3
		7.2.1 Accessing the Tip Type Editor	/-4
		7.2.2 Adding New Tips	7-5
		7.2.5 Changing Tip Properties	/-0
	7 2	7.2.4 Removing Tips	/-0
	1.3	7.2.1 Accessing the Labyers Type Editor	··· /-9
		7.3.1 Accessing the Laborate Type Editor	/-9
		7.3.2 Adding a New Labware Type	/-10
		7.2.4 Deleting a Labware Type	/-11
		7.5.4 Deterning a Labware Type	/-12
		7.2.6 Editing Labora Tring Properties	/-13
		7.5.0 Editing Laoware Type Properties	/-14

8	Und	erstand	ling and Creating Liquid Types	
	8.1	Overvie	2W	8-1
		8.1.1	Understanding Liquid Types	8-3
	8.2	Creating	g New Liquid Types.	8-4
		8.2.1	Changing the Aspirate and Dispense Speeds in the Liquid Type Edito	r8-7
		8.2.2	Changing the Cutoff Velocity for Span-8 Pods in Liquid Type Edito	r
			(FX, NX-S8 only)	8-8
	8.3	Removi	ng Liquid Types	. 8-10
	8.4	Copying	g and Pasting Liquid Types Within a Project File	. 8-10
	8.5	Specify	ing the Default Liquid Type	. 8-11
9	Und	erstand	ling and Creating Techniques	
	9.1	Overvie	www	9-1
		9.1.1	How Techniques Work	9-1
		9.1.2	Why Techniques are Valuable	9-2
	9.2	Accessi	ng the Technique Browser.	9-3
		9.2.1	Identifying Techniques	9-4
	9.3	Creating	g New Techniques	9-5
	9.4	Setting	Technique Values	9-9
		9.4.1	Viewing Technique Description	. 9-10
		9.4.2	Setting Aspirate Values	. 9-12
		9.4.3	Setting Dispense Values	. 9-14
		9.4.4	Setting Mix Values	. 9-17
		9.4.5	Setting Liquid Level Sensing Values (FX, 3000, and NX-S8 only)	. 9-19
		9.4.6	Setting Clot Detection Values (NX-S8 only)	. 9-25
		9.4.7	Setting Piercing Values (NX-S8 only)	. 9-29
		9.4.8	Overriding Liquid Type Values	. 9-30
	9.5	Calibrat	ting Techniques	. 9-34
	9.6	Condition	oning Techniques for Multiple Dispense (FX, NX-S8 only)	. 9-36
	9.7	Creating	g Technique Groups	. 9-39
		9.7.1	Renaming Groups	. 9-40
		9.7.2	Adding and Removing Techniques from Groups	. 9-41
		9.7.3	Removing Groups	. 9-42
	9.8	Removi	ng Techniques	. 9-43
	9.9	Modify	ing Saved Techniques	. 9-44
	9.10	Selectin	g and Modifying Techniques Manually in a Method	. 9-45
		9.10.1	Modifying a Technique Through a Method Step	. 9-46
		9.10.2	Saving Custom Techniques	. 9-48
10	Usir	ng the F	Pipetting Template Editor	
	10.1	Overvie	W	. 10-1
	10.2	Creating	g and Modifying Pipetting Templates	. 10-2
		10.2.1	Viewing Existing Pipetting Templates	. 10-3
		10.2.2	Creating a New Pipetting Template	. 10-4

	10.3	Understanding Context Variables	10-8
		10.3.1 Inserting Variables and Expressions.	10-8
		10.3.2 Defining Context Variables	10-10
	10.4	Using Volume Conditioning in Pipetting Templates (FX, NX-S8 only)	10-15
		10.4.1 About ListMath	10-15
	10.5	Configuring Pipetting Template Steps.	10-17
		10.5.1 If, Comment, and Loop	10-17
		10.5.2 Clot Detection (NX-S8 only)	10-18
		10.5.3 Aspirate Air Gap	10-20
		10.5.4 Aspirate	10-21
		10.5.5 Moving Aspirate	10-22
		10.5.6 Dispense Air Gap	10-24
		10.5.7 Dispense	10-25
		10.5.8 Moving Dispense	10-26
		10.5.9 Mix	10-28
		10.5.10 Prewet	10-30
		10.5.11 Tip Touch	10-31
		10.5.12 Axes Move	10-33
		10.5.13 Pause	10-34
		10.5.14 Bulk Dispense (FX and NX-S8 only)	10-35
	0		
11	Crea	ating well Patterns	
	11.1	Overview	11-1
	110		
	11.2	Creating a New Well Pattern	11-3
	11.2 11.3	Creating a New Well Pattern	11-3
	11.2 11.3 11.4	Creating a New Well Pattern	11-3 11-5 11-6
	11.2 11.3 11.4 11.5	Creating a New Well Pattern	11-3 11-5 11-6 11-7
12	11.2 11.3 11.4 11.5	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern	11-3 11-5 11-6 11-7
12	11.2 11.3 11.4 11.5 <b>Crea</b>	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern ating and Using Methods	11-3 11-5 11-6 11-7
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern ating and Using Methods Overview Creating a New Method	11-3 11-5 11-6 11-7 12-1
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1 12.2	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Star	11-3 11-5 11-6 11-7 12-1 12-3
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1 12.2	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step	11-3 11-5 11-6 11-7 12-1 12-3 12-4
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1 12.2	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method	11-3 11-5 11-6 11-7 12-1 12-3 12-4 12-7 12-13
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1 12.2 12.3	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques	11-3 11-5 11-6 11-7 12-1 12-3 12-4 12-7 12-13 12-14
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1 12.2 12.3	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>Ating and Using Methods</b> Overview Creating a New Method. 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method. 12.3.1 Understanding Techniques Using Additional Step Palettes	11-3 11-5 11-6 11-7 12-1 12-3 12-4 12-7 12-13 12-14 12-15
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1 12.2 12.3 12.4 12.5	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying Cutting and Pasting Steps in a Method	11-3 11-5 11-6 11-7 12-1 12-3 12-4 12-7 12-13 12-14 12-15 12-16
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1 12.2 12.3 12.4 12.5 12.6	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>Ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying, Cutting, and Pasting Steps in a Method Deleting Steps in a Method	11-3 11-5 11-6 11-7 12-1 12-3 12-4 12-7 12-13 12-14 12-15 12-16 12-17
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1 12.2 12.3 12.4 12.5 12.6 12.7	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying, Cutting, and Pasting Steps in a Method Deleting Steps in a Method Lising Lindo and Redo in Method Building	11-3 11-5 11-6 11-7 12-1 12-3 12-4 12-7 12-13 12-14 12-15 12-16 12-17 12-18
12	<ul> <li>11.2</li> <li>11.3</li> <li>11.4</li> <li>11.5</li> <li><b>Crea</b></li> <li>12.1</li> <li>12.2</li> <li>12.3</li> <li>12.4</li> <li>12.5</li> <li>12.6</li> <li>12.7</li> </ul>	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>Ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying, Cutting, and Pasting Steps in a Method Deleting Steps in a Method Using Undo and Redo in Method Building 12.7.1 Using Undo	11-3 11-5 11-6 11-7 12-1 12-3 12-4 12-7 12-13 12-14 12-15 12-16 12-17 12-18 12-18
12	11.2 11.3 11.4 11.5 <b>Crea</b> 12.1 12.2 12.3 12.4 12.5 12.6 12.7	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying, Cutting, and Pasting Steps in a Method Deleting Steps in a Method Using Undo and Redo in Method Building 12.7.1 Using Undo 12.7.2 Using Redo	11-3 11-5 11-6 11-7 12-1 12-3 12-4 12-7 12-13 12-14 12-15 12-16 12-17 12-18 12-18 12-18
12	<ul> <li>11.2</li> <li>11.3</li> <li>11.4</li> <li>11.5</li> <li><b>Crea</b></li> <li>12.1</li> <li>12.2</li> <li>12.3</li> <li>12.4</li> <li>12.5</li> <li>12.6</li> <li>12.7</li> <li>12.8</li> </ul>	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying, Cutting, and Pasting Steps in a Method Deleting Steps in a Method Using Undo and Redo in Method Building 12.7.2 Using Redo Entering and Viewing Method Properties	11-3          11-5          11-7          12-1          12-3          12-4          12-7          12-13          12-16          12-17          12-18          12-18          12-18
12	<ul> <li>11.2</li> <li>11.3</li> <li>11.4</li> <li>11.5</li> <li><b>Crea</b></li> <li>12.1</li> <li>12.2</li> <li>12.3</li> <li>12.4</li> <li>12.5</li> <li>12.6</li> <li>12.7</li> <li>12.8</li> <li>12.9</li> </ul>	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying, Cutting, and Pasting Steps in a Method Deleting Steps in a Method 12.7.1 Using Undo 12.7.2 Using Redo Entering and Viewing Method Properties Saving a Method	11-3          11-5          11-7          12-1          12-3          12-4          12-7          12-13          12-14          12-15          12-16          12-18          12-18          12-18          12-19         12-20       12-20
12	<ul> <li>11.2</li> <li>11.3</li> <li>11.4</li> <li>11.5</li> <li><b>Crea</b></li> <li>12.1</li> <li>12.2</li> <li>12.3</li> <li>12.4</li> <li>12.5</li> <li>12.6</li> <li>12.7</li> <li>12.8</li> <li>12.9</li> <li>12.10</li> </ul>	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method. 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying, Cutting, and Pasting Steps in a Method Deleting Steps in a Method Using Undo and Redo in Method Building. 12.7.1 Using Undo. 12.7.2 Using Redo. Entering and Viewing Method Properties. Saving a Method	11-3          11-5          11-7          12-1          12-3          12-4          12-7          12-15          12-16          12-18          12-18          12-19          12-20
12	<ul> <li>11.2</li> <li>11.3</li> <li>11.4</li> <li>11.5</li> <li><b>Crea</b></li> <li>12.1</li> <li>12.2</li> <li>12.3</li> <li>12.4</li> <li>12.5</li> <li>12.6</li> <li>12.7</li> <li>12.8</li> <li>12.9</li> <li>12.10</li> <li>12.11</li> </ul>	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying, Cutting, and Pasting Steps in a Method Deleting Steps in a Method Using Undo and Redo in Method Building 12.7.1 Using Undo 12.7.2 Using Redo Entering and Viewing Method Properties Saving a Method Validating a Method	11-3          11-5          11-7          12-1          12-3          12-4          12-7          12-13          12-14          12-15          12-16          12-17          12-18          12-18          12-19          12-20          12-22
12	<ul> <li>11.2</li> <li>11.3</li> <li>11.4</li> <li>11.5</li> <li><b>Crea</b></li> <li>12.1</li> <li>12.2</li> <li>12.3</li> <li>12.4</li> <li>12.5</li> <li>12.6</li> <li>12.7</li> <li>12.8</li> <li>12.9</li> <li>12.10</li> <li>12.11</li> <li>12.12</li> </ul>	Creating a New Well Pattern Copying a Well Pattern Deleting a Well Pattern Renaming a Well Pattern <b>ating and Using Methods</b> Overview Creating a New Method. 12.2.1 Configuring the Start Step 12.2.2 Configuring the Finish Step Inserting Steps in a Method 12.3.1 Understanding Techniques Using Additional Step Palettes Copying, Cutting, and Pasting Steps in a Method. Deleting Steps in a Method Using Undo and Redo in Method Building. 12.7.1 Using Undo. 12.7.2 Using Redo. Entering and Viewing Method Properties Saving a Method. Validating a Method Validating a Method	11-3          11-5          11-7          12-1          12-3          12-4          12-7          12-7          12-13          12-14          12-15          12-16          12-18          12-18          12-18          12-20          12-22          12-24

	12.13	Viewing Method History	12-27
		12.13.1 Viewing Revision History	12-27
		12.13.2 Viewing Validated Run History	12-28
	12.14	Opening a Saved Method.	12-30
	12.15	Checking Out a Method.	12-32
	12.16	Importing and Exporting Methods.	12-34
		12.16.1 Exporting a Method	12-35
		12.16.2 Exporting All Methods Associated With a Project File	12-36
		12.16.3 Importing Methods	12-37
	12.17	Deleting and Restoring Methods	12-40
		12.17.1 Deleting Methods	12-40
		12.17.2 Restoring Methods	12-41
	12.18	Running, Pausing, and Stopping a Method	12-44
		12.18.1 Running a Method	12-44
		12.18.2 Pausing a Method in Progress.	12-51
		12.18.3 Stopping a Method in Progress	12-52
		12.18.4 Snapping a Continuation (FX, NX only)	12-52
		12.18.5 Disabling Steps Within a Method	12-53
		12.18.6 Purging Air From Span-8 Pod Before a Method Run	
		(FX and NX-S8 Only)	12-54
	12.19	Printing a Method	12-55
	12.20	Improving a Method	12-56
		12.20.1 Running the Method at Slow Speed	12-56
		12.20.2 Snapping a Continuation (FX and NX Only)	12-57
		12.20.3 Using Logging	12-58
13	Usin	g Variables and Expressions in a Method	
	13.1	Overview	13-1
	13.2	Using Variables	13-2
		13.2.1 Creating Variables	13-3
		13.2.2 Using Variables in a Method	13-4
	13.3	Using Expressions	13-6
14	Usin	g Sample Tracking and Data Sets in a Method	
	14 1	Overview	14-1
		14.1.1 Sample Tracking	14-1
		14.1.2 Data Set Definition	
		14 1 3 Data Set Steps in the Biomek Software	14-2
		14.1.4 Example of Sample Tracking Using Data Sets	14-2
	14 2	Understanding How Data Sets Are Created	14-3
	- ·· <b>-</b>	14.2.1 Understanding the Volume Data Set	14-3
	143	Using Data Sets in a Transfer or Combine Step	14-4
	1 1.2	14.3.1 Defining Well Patterns Using Data Sets	14-5

14.3.1Defining wen Patterns Using Data Sets.14-314.3.2Specifying Volumes Using Data Sets.14-8

#### 15 Using the Basic Step Palette

	15.1	Overview	15-1
	15.2	Instrument Setup Step	15-3
		15.2.1 Selecting a Deck Layout	15-5
		15.2.2 Verifying Deck Layout	15-6
		15.2.3 Configuring Pod Setup Options	15-7
		15.2.4 Populating the Deck with Labware	15-9
		15.2.5 Configuring Tip and Labware Properties	15-15
		15.2.6 Creating the SampleID Data Set Using the Instrument Setup Step	15-23
		15.2.7 Creating Custom Labware	15-26
	15.3	Configuring Transfer and Combine Steps	15-29
		15.3.1 Configuring Tip Handling	15-30
		15.3.2 Configuring Source Labware	15-35
		15.3.3 Configuring Destination Labware	15-44
		15.3.4 Configuring Transfer Details	15-53
	15.4	Move Labware Step	15-56
		15.4.1 Inserting a Move Labware Step	15-56
		15.4.2 Configuring the Move Labware Step	15-57
	15.5	Pause Step	15-58
		15.5.1 Configuring a Pause for a Specific Deck Position or Pod	15-60
		15.5.2 Pausing the Whole System and Displaying a Message	15-61
	15.6	Comment Step	15-62
16	Usin	g the Intermediate Step Palette	
	16.1	Overview	16-1
	16.2	Displaying the Intermediate Step Palette.	16-3
	16.3	Aspirate Step	16-4
		16.3.1 Configuring the Aspirate Step	16-6
	16.4	Dispense Step.	16-9
		16.4.1 Configuring the Dispense Step	16-10
	16.5	Mix Step.	16-14
		16.5.1 Configuring the Mix Step	16-15
	16.6	Wash Step	16-17
		16.6.1 Configuring the Wash Step.	16-18
	16.7	New Tips Step	16-19
	16.8	Unload Tips Step	16-21
	16.9	Loop Step	16-23
		16.9.1 Configuring a Loop Step and Creating a New Variable	16-24
		16.9.2 Configuring a Loop Step Using an Existing Variable	16-27
	16.10	Cleanup Step	16-30
	16.11	Move Pod Step	16-33
	16.12	Group Step	16-35

#### 17 Using the Advanced Step Palette

17.1	Overview	17-1
17.2	Displaying the Advanced Step Palette.	17-3
17.3	Run Method Step	17-4
	17.3.1 Configuring the Run Method Step	17-5
17.4	Worklist Step	17-6
	17.4.1 Configuring the Worklist Step	17-8
17.5	Just in Time Step	17-9
17.6	Let Step	17-11
	17.6.1 Configuring the Let Step.	17-12
17.7	If Step	17-13
	17.7.1 Configuring the If Step	17-14
17.8	Script Step	17-15
	17.8.1 Configuring the Script Step	17-17
17.9	Scripted Let Step	17-18
	17.9.1 Configuring the Scripted Let Step	17-20

#### 18 Using the Span-8 Step Palette

18.1	Overvie	W	. 18-1
18.2	Displayi	ing the Span-8 Step Palette	. 18-3
18.3	Serial D	ilution Step (including 3000)	. 18-4
	18.3.1	Configuring the Serial Dilution Step	. 18-6
	18.3.2	Configuring Diluent Properties.	18-12
18.4	Span-8	Aspirate Step	18-16
	18.4.1	Configuring the Span-8 Aspirate Step	18-18
18.5	Span-8 l	Dispense Step	18-23
	18.5.1	Configuring the Span-8 Dispense Step	18-25
18.6	Span-8 l	New Tips Step	18-31
	18.6.1	Configuring the Span-8 New Tips Step	18-32
18.7	Span-8	Fip Discard Step	18-33
	18.7.1	Configuring the Span-8 Tip Discard Step	18-34
18.8	Span-8	Wash Tips Step	18-35
	18.8.1	Configuring the Span-8 Wash Tips Step for a Passive Wash	18-36
	18.8.2	Configuring the Span-8 Wash Tips Step for an Active Wash	18-39
18.9	Transfer	From File Step (including 3000)	18-41
	18.9.1	Configuring Tip Handling	18-43
	18.9.2	Configuring File Properties	18-47
	18.9.3	Configuring Source Labware	18-51
	18.9.4	Configuring Destination Labware	18-55
	18.9.5	Configuring Transfer Details	18-62
18.10	Define I	Pattern Step (including 3000)	18-64
	18.10.1	Configuring the Define Pattern Step by Specifying the	
		Wells Manually	18-65
	18.10.2	Configuring the Define Pattern Step Reading from a File	18-66
	18.10.3	Using a Defined Pattern in a Transfer or Combine Step	18-69

19	Usir	ng the HDR Step Palette (FX, 3000 only)
	19.1	Overview
	19.2	Displaying the HDR Step Palette
	19.3	Using the HDR Transfer and Combine Steps 19-3
		19.3.1 Washing Pins Between Transfers
		19.3.2 Configuring Source Labware 19-7
		19.3.3 Configuring Destination Labware
	19.4	Using the HDR Tool Cleaning Step 19-20
		19.4.1Configuring Cleaning Operations19-21
	19.5	Using the HDR Move Labware Step (FX only) 19-30
		19.5.1Configuring the HDR Move Labware Step19-31
20	Usir	ng the Biomek 3000 Step Palette
	20.1	Overview
	20.2	Displaying the Biomek 3000 Step Palette
	20.3	Change Tool Step
	20.4	Load Tool Step
	20.5	Unload Tool Step
	20.6	Remove to Waste Step 20-8
	20.7	Bulk Dispense Step 20-10
	20.8	Purge Wash Tool Step    20-13
	20.9	Wash Plate Step    20-15
21	Usir	ng the Specialty Step Palette
	21.1	Overview
	21.2	Displaying the Speciality Step Palette
	21.3	Create Group Step
		21.3.1 Configuring the Create Group Step
	21.4	Next Labware Step
		21.4.1 Configuring the Next Labware Step 21-7
		21.4.2 Using Group Variables in a Method
	21.5	Run Program Step         21-10
		21.5.1 Configuring the Run Program Step 21-10
	21.6	Next Item Step         21-13
		21.6.1Configuring the Next Item Step21-14
	21.7	Set Global Step 21-16
		21.7.1Configuring the Set Global Step.21-16
	21.8	Defining and Running Procedures
		21.8.1Define Procedure Step21-18
		21.8.2         Run Procedure Step         21-20
	21.9	Break Step
		21.9.1 Configuring the Break Step 21-21

### 22 Using the Devices Step Palette

	22.1	Overview	22-1
	22.2	Displaying the Devices Step Palette	22-3
	22.3	SPE Step	22-4
		22.3.1 Configuring the SPE Step.	22-5
	22.4	Using External Devices with Biomek Laboratory Automation Workstations	22-7
		22.4.1 Configuring Devices Using the Device Editor	22-7
	22.5	Device Setup Step	22-9
		22.5.1 Configuring the Device Setup Step	. 22-10
	22.6	Device Action Step	. 22-16
		22.6.1 Configuring the Device Action Step.	. 22-17
	22.7	Tip Loader Step (FX, NX-MC only)	. 22-37
		22.7.1 Configuring the Tip Loader Step	. 22-38
	22.8	SILAS Module Step	. 22-39
		22.8.1 Configuring the SILAS Module	. 22-39
		22.8.2 Configuring Runtime Data	. 22-41
		22.8.3 Creating Data Sets from SILAS Messages	. 22-42
23	Usin	g the Stacker Carousel Step Palette	
	23.1	Overview	23-1
	23.2	Displaying the Stacker Carousel Step Palette	23-2
	23.3	Using the Stacker Setup Step	23-3
		23.3.1 Configuring the Placement of Labware in the Hotels	23-3
		23.3.2 Configuring the Properties of the Labware in the Hotels	23-6
	23.4	Using the Present Step	. 23-12
		23.4.1 Configuring from which Hotel Labware is Presented	. 23-12
		23.4.2 Configuring Bar Code Reading Options.	. 23-13
	23.5	Using the Store Step	. 23-14
24	Usin	g the Data Sets Step Palette	
	24.1	Overview	24-1
	24.2	Displaying the Data Sets Step Palette	. 24-2
	24.3	Configuring the Create Data Set Step	. 24-3
		24.3.1 Configuring File Options	24-4
		24.3.2 Configuring Input Options	24-9
		24.3.3 Configuring Output Options	. 24-13
	24.4	Configuring the Data Set Management Step	. 24-16
		24.4.1 Copying a Data Set	. 24-16
		24.4.2 Renaming a Data Set.	. 24-17
		24.4.3 Removing a Data Set	. 24-19
		24.4.4 Changing the Properties of a Data Set	. 24-20
	24.5	Configuring the Data Set Processing Step.	. 24-22
		24.5.1 Understanding Expressions.	. 24-24
	24.6	Configuring the Reporting Step.	. 24-25
		24.6.1 Report Styles.	. 24-26
		24.6.2 Configuring a SQL Server Report Style	. 24-29

#### 25 Handling and Preventing Errors

	25.1	Overvie	ew	
		25.1.1	Error Message Overview	
		25.1.2	Error Recovery Overview	
		25.1.3	Error Prevention Overview	
	25.2	Configu	uration Errors	
		25.2.1	Configuration Error Messages	
		25.2.2	Recovering from Configuration Errors	
	25.3	Validat	ion Errors	
		25.3.1	Validation Error Messages	
		25.3.2	Validating a Method Before Run	
		25.3.3	Recovering from Validation Errors	25-10
	25.4	Manual	Control Errors	
		25.4.1	Manual Control Error Messages and Recovery	
	25.5	Run-Ti	me Errors	25-12
	20.0	25.5.1	Run-Time Error Messages	25-12
		25.5.2	Recovering from Run-Time Errors.	25-13
	25.6	Light C	Curtain Errors	25-15
	20.0	25.6.1	Light Curtain Error Messages	25-15
		25.6.2	Recovery Options for Light Curtain Errors	25-15
		25.6.2	Using the Stop Button on the Biomek 3000	25-17
	257	Liquid	Level Sensing Frrors	25-18
	20.1	25 7 1	Liquid Level Sensing on the FX and NX-S8	25-18
		25.7.2	Liquid Level Sensing on the Biomek 3000	25-18
		25.7.2	Recovering from Liquid Level Sensing Errors	25-18
	25.8	Clot De	etection Error Message (NX-S8 only)	25-21
	23.0	25.8.1	Retrying Has Not Resolved the Clot	25-21
	25.9	Multin	le Frror Messages	25-22
	20.7	25 9 1	Recovering from Multiple Errors	25-22
	25 10	Observe	ed Frrors	25-23
	23.10	25 10 1	Recovering from Observed Errors	25-23
	25 11	Using (	Continuations (not available on a 3000)	25-25
	23.11	25 11 1	Characteristics of Continuations	25-25
		25.11.1	Snapping a Continuation	25-25
		25.11.2	Loading and Running a Continuation	25_20
	25 12	23.11.3 Dorform	ning Single Operations Within Stops	
	23.12	renom	ning Single Operations within Steps	
26	Gene	erating	and Using Log Data	
	26.1	Overvie	ew	
	26.2	Types of	of Logs	
		26.2.1	Details Log	
		26.2.2	Errors Log.	
		26.2.3	Pipetting Log.	
		26.2.4	Span8Pipetting Log (FX, NX-S8 only)	26-4
		26.2.5	Span8Transfer Log (FX, NX-S8 only)	26-5
		26.2.6	UnifiedPinetting Log	26-5
		26.2.7	UnifiedTransfer Log	26-5
		20.2.1		

	26.3	Log Co	ontents	26-6
		26.3.1	Details Log Contents	26-6
		26.3.2	Errors Log Contents	26-6
		26.3.3	Pipetting Log Contents	26-7
		26.3.4	Span-8Pipetting Log Contents (FX, NX-S8 only)	26-7
		26.3.5	Span-8Transfer Log Contents (FX, NX-S8 only)	26-8
		26.3.6	UnifiedPipetting Log Contents	26-9
		26.3.7	UnifiedTransfer Log Contents	. 26-11
	26.4	Selectin	ng Log Types	. 26-13
	26.5	Viewing	g Log Files	. 26-14
27	Usin	ig Man	ual Control	
	27.1	Overvie	ew	27-1
	27.2	Accessi	ing Manual Control	27-2
	27.3	Underst	tanding the Options in Manual Control	27-3
••	<u> </u>			
28	Scri	pting		
	28.1	Overvie	ew	28-1
	28.2	Using the second	he Script Steps	28-2
	28.3	Scriptin	ng Data Sets.	28-2
		28.3.1	Procedure to Script Data Sets	28-3
29	Cha	nging V	Window Appearance	
	29.1	Overvie	ew	29-1
	29.2	Changii	ng Display Preferences	29-4
		29.2.1	Configuring General Options	29-5
		29.2.2	Configuring View Options	29-7
		29.2.3	Configuring Errors Options	29-9
	29.3	Togglin	ng Toolbars and Step Palettes	. 29-11
	29.4	Custom	nizing Step Palettes	. 29-14
		29.4.1	Displaying Large Step Palette and Toolbar Views	. 29-14
		29.4.2	Changing Step Properties	. 29-16
		29.4.3	Modifying a Step Caption in Method View	. 29-17
	29.5	Using the	he Step Palette Builder	. 29-18
		29.5.1	Accessing the Step Palette Builder	. 29-19
		29.5.2	Displaying and Hiding Step Palettes	. 29-20
		29.5.3	Customizing Step Palettes	. 29-20
		29.5.4	Creating a New Step Palette	. 29-23
		29.5.5	Modifying Step Palette Properties	. 29-23
		29.5.6	Deleting a Step Palette	. 29-24
Ind	ex			xxxix

# **List of Figures**

Figure 2-1. Support Options	
Figure 2-2. Account Management	
Figure 2-3. Password Restore	
Figure 2-4. Accounts & Permissions Audit Log	
Figure 2-5. Log on/off button — no user logged on	
Figure 2-6. Logon	
Figure 2-7. Administrator Notification	
Figure 2-8. Log on/off button — user logged on	
Figure 2-9. Current Permissions	
Figure 2-10. Log on/off button — user logged on	
Figure 3-1. Biomek method editor	
Figure 3-2. Save Instrument	
Figure 3-3. Open Instrument	
Figure 3-4. Import/Export Utility	
Figure 3-5. Exporting deck layout Deck1 to an import file	
Figure 3-6. Save Import File	
Figure 3-7. Confirmation to address to overwrite an imported setting	
Figure 3-8. Open Instrument file	
Figure 3-9. Open Instrument showing backup files	
Figure 4-1. Hardware Setup for a Biomek FX instrument	
with a Multichannel Pod and Span-8 Pod	
Figure 4-2. Hardware Setup — configuring the simulator settings	
Figure 4-3. Selecting a background color	
Figure 4-4. Simulator Preview with camera navigation buttons enabled	
Figure 4-5. Camera Navigation buttons	
Figure 4-6. Hardware Setup — configuring camera controls	
Figure 4-7. Hardware Setup — configuring materials	
Figure 5-1. Deck relationship in Biomek 3000	
Figure 5-2. Deck Editor showing Deck1 for a Biomek FX	
Figure 5-3. Deck Editor for a Biomek FX	
Figure 5-4. Deck Editor for a Biomek 3000	

Figure 5-5. Deck Editor for a Biomek NX-MC	5-8
Figure 5-6. Deck Editor for a Biomek NX-S8	5-9
Figure 5-7. Previews deck selected	5-10
Figure 5-8. Enter a new name for the deck	5-11
Figure 5-9. Confirmation to cancel changes	5-11
Figure 5-10. Delete Deck confirmation	5-12
Figure 5-11. Change deck name	5-13
Figure 5-12. Error in renaming	5-13
Figure 5-13. Confirmation to delete ALP	5-14
Figure 5-14. Possible OneByThree deck positions highlighted on a Biomek FX deck	5-15
Figure 5-15. Warning appears because an ALP is about to be placed	
where it will overlap another ALP	5-15
Figure 5-16. Position and Labware Offset locations on the OnebyOne ALP	5-17
Figure 5-17. ALP Properties of a OneByThree ALP	5-18
Figure 5-18. Position Properties for Deck Positions	5-19
Figure 5-19. Per-Labware Offsets	5-21
Figure 6-1. Biomek method editor	6-2
Figure 6-2. Create new project	6-4
Figure 6-3. Opening a project file	6-5
Figure 6-4. Open Project	6-6
Figure 6-5. Confirm deletion of selected project	6-6
Figure 6-6. Recycled projects	6-7
Figure 6-7. Renaming the open project	6-8
Figure 6-8. Open Project	6-9
Figure 6-9. Saving project as to create a copy	6-10
Figure 6-10. Checking in project items as official records	6-11
Figure 6-11. Check-In (Accounts & Permissions enabled)	6-12
Figure 6-12. Viewing project contents (Accounts & Permissions enabled)	6-13
Figure 6-13. History of a project record	6-14
Figure 6-14. Viewing project contents	6-15
Figure 6-15. Reverting a project file to a specific date and time	6-16
Figure 6-16. Viewing project contents	6-17
Figure 6-17. Discarding changes to a project file after a specific date and time	6-18
Figure 6-18. Viewing project contents	6-19
Figure 6-19. History of a project item	6-19
Figure 6-20. Selecting project contents to export	6-20
Figure 6-21. Save Import File	6-21
Figure 6-22. Open Import File	6-22
Figure 6-23. Checking in project items as official records	6-23
Figure 6-24. Information that appears when no project items are imported	6-23
Figure 6-25. Check-In (Accounts & Permissions enabled)	6-24
Figure 7-1. Relationships for labware and tips in Biomek Software	
Figure 7-2. Tip Types	
Figure 7-3. Adding a new tip type	
Figure 7-4. Warning to confirm cancellation of changes	
Figure 7-7. Remove tips warning	

Figure 7-8. Labware Types	7-9
Figure 7-9. New Labware	7-10
Figure 7-10. Copy Labware Type	7-11
Figure 7-11. Information	7-12
Figure 7-12. Rename Labware Type	7-13
Figure 7-13. Basic Information chosen	7-16
Figure 7-14. Labware Colors	7-17
Figure 7-15. Color	7-17
Figure 7-16. Warning to confirm cancellation of changes made	7-18
Figure 7-17. Sections chosen	7-19
Figure 7-18. Error reminder that sections are deleted	7-19
Figure 7-19. Edit Section Profile	7-20
Figure 7-20. Edit Section Top	7-20
Figure 7-21. Error for an improperly configured reservoir	7-21
Figure 7-22. Stacking chosen	7-22
Figure 7-23. Stackable Items	7-23
Figure 7-24. Secure Stacking	7-25
Figure 7-25. Wells 1 chosen	7-26
Figure 7-26. Well Configuration Dialog	7-27
Figure 7-27. Well Configuration Dialog with Bottom Section checked	7-28
Figure 8-1. Liquid Type Editor	8-2
Figure 8-2. Liquid Type Editor with a new liquid	8-4
Figure 8-3. Liquid Type Editor	8-9
Figure 9-1. Technique Browser	9-3
Figure 9-2. Technique Browser where technique parameters are displayed	9-4
Figure 9-3. Setting Technique Properties for a Biomek FX technique	9-7
Figure 9-4. Technique Editor — General tab	9-10
Figure 9-5. Technique Editor — Aspirate tab	9-12
Figure 9-6. Technique Editor — Dispense tab	
Figure 9-7. Technique Editor — Mix tab	
Figure 9-8. Technique Editor — General Settings on Liquid Level Sensing tab	
Figure 9-9. Technique Editor — Repetition Settings on Liquid Level Sensing tab	
Figure 9-10. Technique Editor — Sensitivity Settings on Liquid Level Sensing tab	
Figure 9-11, Technique Editor — Error Handling Settings on Liquid Level Sensing tab	
Figure 9-12 Liquid Level Sensing Error Handling prompt for Biomek 3000	9-24
Figure 9-13 Technique Editor — General Settings on Clot Detection tab	9-26
Figure 9-14 Technique Editor - Sensitivity Settings on Clot Detection tab	9-27
Figure 9-15 Technique Editor - Error Handling Settings on Clot Detection tab	9-28
Figure 9-16 Technique Editor - Piercing tab	9-29
Figure 9-17 Technique Editor — Liquid Type tab	9-31
Figure 9-18 Technique Editor — Volume Calibration on Calibration Tab	9-35
Figure 9-20 Technique Editor — Volume Conditioning on Calibration Tab	9-38
Figure 9-21 Creating a new group in the Technique Browser	9_39
Figure 9-22 New group created in the Technique Browser	9-40
Figure 9-23 Creating a new group in the Technique Browser	9 <u>4</u> 1
Figure 9-24 Remove technique confirmation prompt	<u>9_4</u> 3
i igue / 27. Remove teeninque commution prompt	) <del>-</del> +J

#### xxvi List of Figures

Figure 9-25. Customize techniques or manually select techniques from a list in source or	
destination configurations of a pipetting step	9-45
Figure 9-26. Technique selection within a method	9-46
Figure 9-27. Dispense Tab appears when Customize is selected from destination	9-47
Figure 9-28. Save a custom technique within a step	9-48
Figure 10-1. Pipetting Template Editor	. 10-2
Figure 10-2. New pipetting template	. 10-4
Figure 10-3. Rename pipetting template	. 10-5
Figure 10-4. Technique Browser	. 10-5
Figure 10-5. Technique Editor	. 10-6
Figure 10-6. Selecting variable from selected group	. 10-9
Figure 10-7. Identifying the location of Technique Editor context variables	10-10
Figure 10-8. Identifying the location of Liquid Type Editor context variables	10-11
Figure 10-9. Clot Detection step configuration	10-18
Figure 10-10. Aspirate Air Gap step configuration	10-20
Figure 10-11. Aspirate step configuration	10-21
Figure 10-12. Moving Aspirate step configuration	10-22
Figure 10-13. Dispense Air Gap step configuration	10-24
Figure 10-14. Dispense step configuration	10-25
Figure 10-15. Moving Dispense step configuration	10-26
Figure 10-16. Mix step configuration	10-28
Figure 10-17. Prewet step configuration	10-30
Figure 10-18. Tip Touch step configuration	10-31
Figure 10-19. Axes Move step configuration	10-33
Figure 10-20. Pause step configuration	10-34
Figure 10-21. Bulk Dispense step configuration	10-35
Figure 11-1. Well Patterns	. 11-2
Figure 11-2. New Pattern	. 11-3
Figure 11-3. Warning when Cancel is chosen	. 11-4
Figure 11-4. Copy Pattern	11-5
Figure 11-5. Warning confirming deletion of a pattern	. 11-6
Figure 11-6. Rename Pattern	. 11-7
Figure 12-1. Biomek Software main editor for method creation	. 12-2
Figure 12-2. Biomek Software main editor for dual-pod Biomek FX instrument	. 12-3
Figure 12-3. Start step and configuration	. 12-5
Figure 12-4. Start step with variables configured	. 12-6
Figure 12-5. Prompt to enter value for variable at runtime	. 12-7
Figure 12-6. The Finish step configuration for a Biomek FX instrument	. 12-8
Figure 12-7. Finish step options for a Biomek 3000 instrument	12-10
Figure 12-8. Reporting enabled on Finish step	12-11
Figure 12-9. Prompt when method is aborted.	12-12
Figure 12-10. Inserting a step into a method	12-13
Figure 12-11. Inserting a step by drag and drop	12-13
Figure 12-12. Step palette menu	12-15
Figure 12-13. Confirmation to delete steps in a method	12-17
Figure 12-14. Method Properties	12-19

Figure 12-15. Save method	12-20
Figure 12-16. Biomek Software main editor after saving method ExampleMethod	12-21
Figure 12-17. Check In	12-22
Figure 12-18. Check-In	12-23
Figure 12-19. Signature	12-25
Figure 12-20. Signature	12-26
Figure 12-21. Method history	12-27
Figure 12-22. Validated Run History displayed in validated mode	12-28
Figure 12-23. Print Preview.	12-29
Figure 12-24. Open saved methods	12-30
Figure 12-25. Check Out Method (Accounts & Permissions enabled)	12-32
Figure 12-26. Export Method	12-35
Figure 12-27. Browse For Folder	12-36
Figure 12-28. Import Method	12-37
Figure 12-29. Confirming the method file(s) to import	12-37
Figure 12-30. Choosing to import project items corresponding to the method	12-38
Figure 12-31. Check In Project.	12-38
Figure 12-32. Check-In	12-39
Figure 12-33. Open saved methods	12-40
Figure 12-34. Confirm deletion of selected method	12-41
Figure 12-35. Open saved methods	12-42
Figure 12-36. Recycled methods	12-43
Figure 12-37. Deck layout confirmation prompt for a Biomek 3000 method	12-45
Figure 12-38. The Biomek Simulator showing a method run on an NX-MC	12-46
Figure 12-39. Camera navigation controls	12-47
Figure 12-40. Biomek Simulator with all simulator controls enabled	12-48
Figure 12-41. Step disabled in Method View	12-53
Figure 12-42. Printed method	12-55
Figure 12-43. Changing the method speed in Hardware Setup	12-56
Figure 13-1. Transfer step using the AspValue variable	13-2
Figure 13-2. Variables configured in the Start step	13-3
Figure 13-3. Transfer step using the AspValue and AspValue2	
variables defined in the Start step	13-5
Figure 14-1. Transfer step in Method View after data set is created	14-5
Figure 14-2. Transfer step — zoomed-in destination labware	14-6
Figure 14-3. Transfer step in Method View after data set is created	14-8
Figure 15-1. Instrument Setup step and configuration	15-4
Figure 15-2. Pause to Confirm Setup? Deck Layout confirmation prompt	15-6
Figure 15-3. Pod Setup prompt for dual-pod Biomek FX instrument	15-8
Figure 15-4. Populating the Deck Layout of a Biomek 3000 instrument	15-9
Figure 15-5. A second Instrument Setup step with deck positions specified As Is	15-11
Figure 15-6. Toggle applied to the Deck Layout	15-13
Figure 15-7. Clear Deck Layout designation applied to deck position TL1	15-14
Figure 15-8. Confirmation prompt to Clear All labware from the Deck Layout	15-15
Figure 15-9. Labware Properties for tips	15-16
Figure 15-10. Labware Properties for tips with Available Tips shown	15-17

Figure 15-11. Labware Properties for microplates, reservoirs, and tube racks	15-18
Figure 15-12. Labware Properties for a 96-well microplate with Labware Volumes shown	15-20
Figure 15-13. Labware Properties for a Biomek 3000 Tool Rack	15-21
Figure 15-14. Labware Properties for a Biomek 3000 Tool Rack with tools loaded	15-22
Figure 15-15. Enabling bar code input in the Instrument Setup step	15-23
Figure 15-16. Bar code input	15-24
Figure 15-17. Entering sample IDs for labware in Bar code input	15-25
Figure 15-18. Custom labware category displayed	15-27
Figure 15-19. Custom labware created	15-28
Figure 15-20. Transfer step configuration for a Multichannel Pod	15-30
Figure 15-21. Combine step — Adding Source Labware	15-34
Figure 15-22. Combine step — Source Labware configuration	15-36
Figure 15-23. Transfer Step — Zoomed-in Source Labware configuration	15-37
Figure 15-26. Combine step — Source Labware configuration	15-41
Figure 15-27. Custom Height prompt	15-42
Figure 15-28. Technique Editor — Dispense options	15-43
Figure 15-29. Transfer step — adding destination labware	15-45
Figure 15-30. Transfer step — destination labware configuration	15-46
Figure 15-33. Transfer step — destination labware technique configuration	15-50
Figure 15-34. Technique Editor — Dispense options	15-52
Figure 15-35. Transfer step — additional Transfer Details	15-53
Figure 15-36. Advanced button — JIT Block	15-55
Figure 15-37. Move Labware step and configuration	15-56
Figure 15-38. Inserting a Pause step in a method	15-59
Figure 15-39. Pausing a deck position for a specified length of time	15-60
Figure 15-40. Pause step that provides instruction	15-61
Figure 15-41. Pause message prompt	15-61
Figure 15-42. Comment step after text is inserted	15-62
Figure 16-1. Biomek Software main editor with Intermediate Step Palette displayed	16-3
Figure 16-2. Aspirate step and configuration	16-5
Figure 16-3. Aspirate step — Adjusting the aspirate height	16-8
Figure 16-4. Custom Height prompt	16-8
Figure 16-5. Dispense step and configuration	16-10
Figure 16-6. Dispense step — Adjusting the dispense height	16-13
Figure 16-7. Custom Height prompt	16-13
Figure 16-8. Mix step and configuration	16-15
Figure 16-9. Wash step and configuration	16-17
Figure 16-10. New Tips step and configuration	16-20
Figure 16-11. Unload Tips step and configuration	16-22
Figure 16-12. Loop step and configuration	16-23
Figure 16-13. Loop step — variable created in Variable	16-24
Figure 16-14. Loop step variable 'Count' used in an Aspirate substep	16-26
Figure 16-15. Loop step — using a previously created variable as the Increment	16-27
Figure 16-16. Loop step — the 'AspValue' variable is used as the Volume for the substeps	16-29
Figure 16-17. Cleanup step and configuration.	16-31
Figure 16-18. Move Pod step and configuration	16-33
- · ·	

Figure 16-19. Group step and configuration	16-35
Figure 17-1. Biomek Software main editor with Advanced Step Palette displayed	17-3
Figure 17-2. Run Method step and configuration	17-5
Figure 17-3. Worklist example	17-6
Figure 17-4. Worklist step and configuration	17-7
Figure 17-5. Worklist — text file selection	
Figure 17-6. Just In Time step	17-9
Figure 17-7. Let step and configuration	17-11
Figure 17-8. If step and configuration	17-13
Figure 17-9. Script step and configuration	17-16
Figure 17-10. Scripted Let step and configuration	17-19
Figure 18-1. Biomek main editor with Span-8 Step Palette displayed	
Figure 18-2. Serial Dilution step and configuration	
Figure 18-3. Serial Dilution step configuration with labware configured	
Figure 18-4. Serial Dilution from left to right	
Figure 18-5. Serial Dilution from top to bottom	
Figure 18-6. Serial Dilution — Diluent Properties configuration	18-12
Figure 18-7. Serial Dilution step configuration with diluent labware	18-13
Figure 18-8. Span-8 Aspirate step and configuration	18-17
Figure 18-9. Span-8 Aspirate step configuration with labware configured	18-18
Figure 18-10. Example of how probe and spacing selection affects well selection	18-19
Figure 18-11. Individual Volumes	18-20
Figure 18-12. Span-8 Aspirate step — Adjusting the aspirate height	18-21
Figure 18-13. Custom Height prompt	18-22
Figure 18-14. Span-8 Dispense step and configuration	18-24
Figure 18-15. Span-8 Dispense step configuration with labware configured	18-25
Figure 18-16. Example of how probe and spacing selection affects well selection	18-26
Figure 18-17. Individual Volumes	18-27
Figure 18-18. Span-8 Dispense step — Adjusting the dispense height	18-28
Figure 18-19. Custom Height prompt	18-29
Figure 18-20. Span-8 New Tips step and configuration	18-31
Figure 18-21. Span-8 Tip Discard step and configuration	18-33
Figure 18-22. Span-8 Wash Tips step — Passive Wash configuration	18-36
Figure 18-23. Span-8 Wash Tips step — Active Wash configuration	18-40
Figure 18-27. Transfer From File step and configuration	18-42
Figure 18-28. Transfer From File step — Tip Handling collapsed	18-46
Figure 18-29. Transfer From File — File Properties	18-47
Figure 18-30. Open dialog	18-48
Figure 18-31. Transfer From File — Example data displayed	18-48
Figure 18-32. Transfer From File — Example data with header row	18-49
Figure 18-33. Transfer From File step — Source Labware selection	18-52
Figure 18-34. Transfer From File step — Source Labware configuration	18-53
Figure 18-35. Custom Height prompt	18-54
Figure 18-36. Transfer From File step — Destination Labware selection	18-56
Figure 18-37. Transfer From File step — Destination Labware configuration	18-57
Figure 18-38. Custom Height prompt	18-58

Figure 18-39. Wells accessed down first, then left to right	18-60
Figure 18-40. Wells accessed to the right first, then top to bottom	18-60
Figure 18-41. Transfer From File step — configuring additional Transfer Details	18-62
Figure 18-42. Advanced button — JIT Block	18-63
Figure 18-43. Define Pattern step and configuration	18-65
Figure 18-44. Define Pattern step — Read from a file configuration	18-66
Figure 18-45. Open dialog	18-67
Figure 18-46. Define Pattern step — Example data displayed	18-67
Figure 18-47. Define Pattern step — Example data with header row	18-68
Figure 18-48. Transfer step using Sample pattern	18-70
Figure 19-1. Biomek main editor with HDR Step Palette displayed	19-2
Figure 19-2. Order of transfer to array positions in destination wells with a 3x3 array	19-3
Figure 19-3. HDR Transfer step configuration	19-4
Figure 19-4. Define Procedure step and configuration	19-6
Figure 19-5. HDR Transfer step — Source Labware configuration	19-8
Figure 19-6. HDR Transfer step — Source Labware selected	19-9
Figure 19-7. Dipping Configuration	19-10
Figure 19-8. Custom Height prompt	19-12
Figure 19-9. HDR Transfer step — Adding Destination Labware	19-14
Figure 19-10. HDR Combine step — Destination Labware configuration	19-15
Figure 19-11. Dipping Configuration	19-16
Figure 19-12. Array Configuration	19-18
Figure 19-13. Custom Height prompt	19-19
Figure 19-14. HDR Tool Cleaning step configuration	19-20
Figure 19-15. HDR Tool Cleaning step — new cleaning operation configuration	19-22
Figure 19-16. HDR Tool Cleaning step — configuration for a wash operation	19-23
Figure 19-17. Dipping Configuration	19-25
Figure 19-18. Custom Height prompt	19-26
Figure 19-19, HDR Tool Cleaning step — new cleaning operation configuration	19-27
Figure 19-20. HDR Tool Cleaning step — configuration for a drving operation	19-28
Figure 19-21. Custom Height prompt	19-29
Figure 19-22. HDR Move Labware step and configuration	19-30
Figure 20-1, Biomek Software main editor with Biomek 3000 Step Palette displayed	20-3
Figure 20-2. Changing tools during a method	20-4
Figure 20-3. Loading tool during a method	20-5
Figure 20-4. Load Tool step configuration with wash tool selected	20-6
Figure 20-5. Unloading a tool during a method	20-7
Figure 20-6. Remove to Waste step configuration	
Figure 20-7. Bulk Dispense step configuration	20-11
Figure 20-8. Wash Tool Purge step configuration.	20-14
Figure 20-9. Wash Plate step configuration.	20-16
Figure 21-1. Biomek Software main editor with Specialty Step Palette displayed	
Figure 21-2. Create Group step and configuration	
Figure 21-3. Create Group step configuration with labware added	
Figure 21-4. Next Labware step and configuration	21-7
Figure 21-5. Transfer step using the Sources group defined in a Create Group step	21-9

Figure 21-6. Run Program step and configuration	21-10
Figure 21-7. Next Item step and configuration	21-14
Figure 21-8. Set Global step and configuration	21-16
Figure 21-9. Define Procedure step and configuration	21-19
Figure 21-10. Run Procedure step and configuration	21-20
Figure 21-11. Break step and configuration	21-21
Figure 22-1. Main editor for a Biomek FX with Devices Step Palette displayed	22-3
Figure 22-2. SPE step and configuration	22-5
Figure 22-3. Device Editor	22-7
Figure 22-4. Light Curtain Access configuration	22-8
Figure 22-5. Device Setup step and configuration	22-9
Figure 22-6. Labware Properties for tips	22-11
Figure 22-7. Labware Properties for tips with available tips shown	22-12
Figure 22-8. Labware Properties for microplates, reservoirs, and tube racks	22-13
Figure 22-9. Labware Properties for 96-well microplate Labware Volumes shown	22-15
Figure 22-10. Device Action step and configuration	22-16
Figure 22-11. Device Action step for the 6-port valve	22-18
Figure 22-12. Device Action step for the 6-port valve with Device Setup chosen	22-19
Figure 22-13. Device Action step for the autolatch	22-20
Figure 22-14. Device Action step for the Drainable/Refillable Reservoir ALP	22-21
Figure 22-15. Device Action step for the MicroMix with Shake chosen	22-23
Figure 22-16. Device Action step for the MicroMix with Custom Shake chosen	22-24
Figure 22-17. Device Action step for the Microplate Shaking ALP	22-26
Figure 22-18. Device Action step for a Multichannel Tip Wash ALP	22-27
Figure 22-19. Device Action Step for the Positive Position ALP	22-28
Figure 22-20. Device Action for a Source/Waste Sensor	22-29
Figure 22-21. Device Action step for the Speed Pump	22-30
Figure 22-22. Device Action step for the Stirring ALP	22-31
Figure 22-23. Device Action step for the vacuum	22-32
Figure 22-24. Device Action for the vacuum with Timed Vacuum chosen	22-33
Figure 22-25. Device Action for the Wash System	22-34
Figure 22-26. Device Action for the Wash System with Latch chosen	22-35
Figure 22-27. Device Action for the Wash System with Aspirate chosen	22-36
Figure 22-28. Tip Loader step and configuration	22-37
Figure 22-29. A Barcode Reader SILAS Module Step inserted in a method	22-39
Figure 22-30. Runtime Data Defaults	22-41
Figure 22-31. SILAS step with data sets chosen	22-42
Figure 22-32. Runtime Data Defaults	22-43
Figure 23-1. Biomek main editor with Stacker Carousel Step Palette displayed	
Figure 23-2. Stacker Setup step configuration	
Figure 23-3. Dialog to drop labware into hotels	
Figure 23-4. Menu accessing Labware Properties	
Figure 23-5. Labware Properties for a tip box	
Figure 23-6. Labware Properties for tips with Show Available Tips activated	
Figure 23-7. Labware Properties for microplates, reservoirs, and tube racks	

Figure 23-8. Labware Properties for microplates, reservoirs, or tube racks	
with Show Labware Volumes activated	23-11
Figure 23-9. Present Step configuration	23-12
Figure 23-10. Store Step configuration	23-14
Figure 24-1. Biomek Software main editor with Advanced Step Palette displayed	
Figure 24-2. Create Data Set step configuration	24-3
Figure 24-3. Create Data Set step — File Options displayed	
Figure 24-4. Create Data Set step — File Options displayed	24-5
Figure 24-5. Reading from a database	
Figure 24-6. SQL Server Connection	24-7
Figure 24-7. Connected to SQL Server	
Figure 24-8. Create Data Set step — expanding Input Options	
Figure 24-9. Create Data Set step — Input Options displayed	24-10
Figure 24-10. Create Data Set step — Input Options displayed	24-12
Figure 24-11. Create Data Set step — expanding Output Options	24-13
Figure 24-12. Create Data Set step — Output Options	24-14
Figure 24-13. Changing data set properties	24-15
Figure 24-14. Choosing Copy in Data Set Management step	24-16
Figure 24-15. Choosing Rename in Data Set Management step	24-18
Figure 24-16. Choosing Remove in Data Set Management step	24-19
Figure 24-17. Change chosen on Data Set Management step configuration	
Figure 24-18. Data Set Processing step configuration	24-22
Figure 24-19. Reporting step configuration	24-25
Figure 24-20. Report saved as Text File	
Figure 24-21. Report saved as Per-Plate HTML Files	
Figure 24-22. Report saved as SOL Server	
Figure 24-23. SOL Server report style chosen	
Figure 24-24. SOL Server Connection	
Figure 24-25. List of databases in SOL Server Connection	
Figure 24-26. Server and database chosen	
Figure 25-1. Example of a configuration error generated due to a	
Liquid Type Editor configuration problem	
Figure 25-2. Example of a configuration error generated from the Deck Editor	
Figure 25-3 Example of an error generated during method validation	
because a step is not configured correctly	25-4
Figure 25-4 Example of an expanded error message	25-5
Figure 25-5 Validate the current method before running it option in Preferences	25-7
Figure 25-6 Example of a validation error message generated during	
method development as a result of incomplete step configuration	25-7
Figure 25-7 Example of a validation error message detected before method run	25-9
Figure 25-8 Example of a validation error message detected during method run	25-9
Figure 25-9 Example of a Manual Control error because the	<b>_</b>
Biomek FX tried to move the pod too far on the Z axis	25-11
Figure 25-10 An example of a Run-Time error due to undetermined gripper state	25-12
Figure 25-11 Choosing STOP on the front rail of the Biomek 3000	25-17
Figure 25-12. Inconsistent liquid level value detected error message	
	····· — — — — /

Figure 25-13. Clot detection error	
Figure 25-14. Error message dialog with multiple errors	
Figure 25-15. An example of multiple error messages	
Figure 25-16. An example of multiple error message generated during Run-Time	
Figure 25-17. A Continuation method snapped during a Loop step	
Figure 25-18. Snap Continuation Information	
Figure 25-19. Confirm saving changes to method	
Figure 25-20. Single Step prompt	
Figure 25-21. First move of method using Single Step	
Figure 26-1. Well numbering in log files	
Figure 26-2. UnifiedPipetting log	
Figure 26-3. UnifiedTransfer log	
Figure 26-4. Select the log type(s) generated here	
Figure 26-5. Span8Pipetting Log viewed in Notepad	
Figure 26-6. Span8Transfer Log viewed in Microsoft Excel	
Figure 27-1. Confirms Manual Control is connecting	
Figure 27-2. Manual Control	
Figure 27-3. Information displaying firmware version	
Figure 29-1. Main editor for a Biomek FX	
Figure 29-2. Step Palette Builder	
Figure 29-3. Preferences	
Figure 29-4. Confirm to delete steps from a method	
Figure 29-5. Preferences - View	
Figure 29-6. Preferences - Errors	
Figure 29-7. Parameter Information	
Figure 29-8. Options for step palettes and toolbars	
Figure 29-9. Step palette menu for a Biomek FX	
Figure 29-10. Main editor with default toolbar options displayed	
Figure 29-11. Step palette views with and without labels	
Figure 29-12. Preferences — main editor setting options	
Figure 29-13. Step Properties	
Figure 29-14. Change the step caption in Edit Step Caption	
Figure 29-15. Modified step caption in Method View	
Figure 29-16. Step Palette Builder	
Figure 29-17. Step Palette — Steps tab	
Figure 29-18. Confirmation for adding a step to a step palette	
Figure 29-19. New Step Palette	
Figure 29-20. Palette Properties	

## **List of Tables**

Table 2-1. Biomek Software Permissions	
Table 4-1. Hardware Setup Options	
Table 4-2. Materials	
Table 5-1. Deck Editor Options	5-5
Table 5-2. ALP and Position Properties.	5-16
Table 7-1. Tip Properties	
Table 7-2. Labware Type Properties	
Table 8-1. Minimum and Maximum Pipetting Speeds in D-Axis	
Table 8-2. Span-8 Cutoff Velocities for Syringes for Span-8 Pods	
Table 9-1. Technique Icons	
Table 9-2. Span-8 Cutoff Velocities for Syringes for Span-8 Pods	
Table 10-1. Context Variables	
Table 12-1. Simulator Lighting Controls	
Table 23-1. Example of a Comma-Separated Value File	
Table 24-1. Some common symbols used in expressions	
Table 24-2. Some expressions and the results	
Table 27-1. Manual Control Options	
Table 29-1. General options	
Table 29-2. View Options	
Table 29-3. Toolbar Options	
*	
# Biomek Software Introduction

## 1.1 About Biomek Software

Biomek Software is used to control the Beckman Coulter Biomek line of laboratory automation workstations. Biomek Software consists of a method editor and other tools and editors for configuring the system to perform desired operations.

Configuring the Biomek instrument is necessary to achieve the desired results from automated methods. The software must have information on the deck layout, labware and tips utilized in the method, liquid types to pipette, and how to perform pipetting operations.

Biomek Software comes with a default set of parameters suitable for many applications; many opportunities are presented for customization to achieve a wide variety of desired results.

Information about the liquid-handling system is stored in two separate files. The instrument file includes information specific to an instrument, including additional components installed, deck layout, and framing information. A project file stores information about labware, tip, and liquid types, and pipetting techniques and templates.

Effectively using Biomek Software includes using the method editor for method building and the various tools and editors to appropriately configure the instrument and project files for the desired task or application.

#### 1.1.1 Instrument Files

An instrument file contains all pertinent information relating to the physical hardware. This includes:

- instrument type and configuration.
- devices installed on the instrument deck.
- external devices integrated with the instrument.
- deck layout and framing information.

The instrument configuration must accurately represent the Biomek instrument hardware to prevent collisions between the instrument and any on-deck components. All instrument configuration is done using Hardware Setup and the Deck Editor.

Hardware Setup is used to configure the heads, devices, and accessories available to the instrument. The Deck Editor is used to configure the precise location of all labware positions on the deck and associate any devices or accessories to those positions. The pod must then be precisely aligned to each deck position by framing the deck.

Refer to Chapter 3, *Using Instrument Files and Settings*, for more information on instrument files.

#### 1.1.2 Project Files

A project file stores custom information about liquid types, labware and tip types, and pipetting techniques and templates that are used to configure the actions of the instrument in a database. Project files store a history of all changes, additions, and deletions from the project file.

Project items are configured using the following editors:

- Labware Type Editor
- Tip Type Editor
- Liquid Type Editor
- Technique Editor
- Pipetting Template Editor
- Well Patterns Editor

Project items may be checked in, which creates a revision of the project item. Checked in revisions can always be recovered and reused, ensuring that checked in or validated methods are reproducible even if project items are subsequently modified or deleted. Refer to Chapter 6, *Understanding and Using Project Files*, for more information on project files.

#### 1.1.3 Methods

Methods contain the precise information for performing a specific sequence of actions to complete a task, and utilize information from the project and instrument files to configure and customize those actions.

The Biomek method editor is used to create methods that control the Biomek liquidhandling system. Methods comprise a series of steps that together perform various operations, such as liquid transfers or labware moves using the gripper. Additional operations, such as microplate shaking, can be performed in a Biomek method through active ALPs or integrated devices that are controlled through the Biomek instrument via serial communications. Refer to Chapter 12, <u>Creating and Using</u> <u>Methods</u>, for more information on building and working with methods.

**Note:** ALPs (Automated Labware Positioners) are removable and interchangeable platform structures installed on the Biomek deck to allow automated assays to be performed.

# 1.2 About this Manual

The information in this manual on Biomek Software generally refers to all Biomek instruments; however, the following conventions present specific conditions or information that apply to one or more, but not all, Biomek instruments:

- **FX** Biomek FX Laboratory Automation Workstation
- > **3000** Biomek 3000 Laboratory Automation Workstation
- > **NX-MC** Biomek NX Multichannel Laboratory Automation Workstation
- > **NX-S8** Biomek NX Span-8 Laboratory Automation Workstation

This manual should be used in conjunction with the appropriate hardware manual for the specific instrument.

An index and glossary are also provided. The glossary contains terms and definitions relating to the Biomek instruments, ALPs, and Biomek Software. It is a separate .pdf file that may be accessed from this manual or any of the related manuals.

#### 1.2.1 Related Manuals

The following related user's manuals for accessing more in-depth information on the Biomek instruments and ALPs are included as specific cross-referenced links to this manual:

- Biomek® FX Laboratory Automation Workstation User's Manual
- Biomek® 3000 Laboratory Automation Workstation User's Manual
- <u>Biomek® NX Multichannel Laboratory Automation Workstation User's</u> <u>Manual</u>
- Biomek® NX Span-8 Laboratory Automation Workstation User's Manual
- ALPs User's Manual

# **2** Using Accounts & Permissions

# 2.1 Overview

Beckman Coulter Accounts & Permissions is an integrated set of features built into Beckman Coulter software that assists users in complying with electronic signature requirements (such as 21 CFR Part 11) for closed systems. With Biomek Software, support is extended only for the Biomek instrument; devices integrated with the Biomek instrument are not supported unless specified in separate documentation.

Accounts & Permissions only provides support for closed systems; networked systems are not supported. In a location where several Beckman Coulter systems are present, Accounts & Permissions must be installed and enabled separately for each system where compliance is desired. Users require a separate account for each system they need to access.

For each Beckman Coulter system, a single administrator sets up the level of support provided by Accounts & Permissions, creates, manages, and sets permissions for user accounts, and configures system parameters relating to Accounts & Permissions.

This chapter covers:

- <u>Installing and Setting the Level of Support For Accounts & Permissions</u> (Section 2.2).
- <u>Administering User Accounts and Permissions</u> (Section 2.3).
- <u>Restoring the Administrator Password</u> (Section 2.4).
- <u>Viewing the Audit Log</u> (Section 2.5).
- <u>Using the Biomek Software With Accounts & Permissions Enabled</u> (Section 2.6).

# 2.2 Installing and Setting the Level of Support For Accounts & Permissions

Some Accounts & Permissions functionality is built into Biomek Software; however, additional options that allow Accounts & Permissions to be enabled and configured are installed separately. This section covers:

- Installing Accounts & Permissions (Section 2.2.1).
- <u>Setting the Level of Support For Accounts & Permissions</u> (Section 2.2.2).

#### 2.2.1 Installing Accounts & Permissions

Beckman Coulter Accounts & Permissions support may be installed before or after Biomek Software is installed on the system.

**Note:** Site processes should be in place to ensure that access to the Accounts & Permissions installation CD is controlled.

To install support for Accounts & Permissions:

- 1. Close all open applications on the computer.
- 2. Place the Beckman Coulter Accounts & Permissions Installer CD in the drive, and browse to the contents of the CD.
- 3. Double-click **Beckman Coulter Accounts & Permissions.exe**. Beckman Coulter Accounts & Permissions Setup appears.

**Note:** If Accounts & Permissions is already installed, options to repair, modify, and remove the application appear.

- 4. Follow the steps in the setup wizard.
- 5. Restart the computer.
- After the computer is restarted, set the level of Accounts & Permissions support desired (refer to Section 2.2.2, <u>Setting the Level of Support For Accounts &</u> <u>Permissions</u>).

#### 2.2.2 Setting the Level of Support For Accounts & Permissions

The administrator may change the level of support for Accounts & Permissions at any time.

**Note:** The site must have processes in place to govern all system administration activities, as specified in the 21 CFR Part 11 regulation.

To set the level of support for Accounts & Permissions:

- 1. Log off and close Biomek Software and any other Beckman Coulter software applications using Accounts & Permissions, if necessary.
- 2. Place the Beckman Coulter Accounts & Permissions Installer CD in the drive, and browse to the contents of the CD.
- 3. Double-click Beckman Coulter Accounts & Permissions Support Options.exe.
- When prompted, enter the Administration Password and choose OK. Support Options appears (Figure 2-1). A series of tabs displays all installed software applications compatible with Beckman Coulter Accounts & Permissions.

**Note:** If the Administration Password is lost, forgotten, or not known, follow the steps in Section 2.4, *<u>Restoring the Administrator Password</u>*.

Beckman Coulter Accounts & Permissions - Support Options					
SAMI® Workstation EX Software	Biomek 9	Software			
O No support					
Accounts and Permissions					
C Accounts and Permissions, with password checks for signing and check-in					
OK	(	Can	cel		

5. Choose the **Biomek Software** tab, if multiple Beckman Coulter software applications supporting Accounts & Permissions are installed on the system.

Figure 2-1. Support Options

- 6. Select the level of support. When several Beckman Coulter software applications are installed on the same system, the level of support selected must be the same for each application.
  - No support User accounts are not required to access Biomek
     Software. Users have access to all software operations and functionality.

**Note:** System activity, such as creating, editing, or running methods, is still logged in the audit trail and can be viewed in the Audit Log (refer to Section 2.5, *Viewing the Audit Log*).

• Accounts and Permissions — Enables the use of user accounts and permissions for Biomek Software. Users must log in to use the software and may access only features and operations for which they have permission. Operations such as checking in, validating, or signing methods do not require password confirmation.

**Note:** Accounts and Permissions without password checks may not provide adequate support for 21 CFR Part 11 compliance. Each site must evaluate the level of support required for a given system.

• Accounts and Permissions, with password checks for signing and check-in — Enables the use of user accounts and permissions with electronic signatures for Biomek Software. Users must log in to use the software and may access only features and operations for which they have permission. Support for 21 CFR Part 11 is provided by requiring password checks for operations such as checking in, validating, and signing methods.

**Note:** Compliance with 21 CFR Part 11 requires implementing site processes beyond the control of the software.

7. Choose **OK** to activate the level of support chosen and close Support Options.

OR

Choose **Cancel** to close Support Options without changing the level of support.

# 2.3 Administering User Accounts and Permissions

System administration tasks for Beckman Coulter Accounts & Permissions are performed in Account Management, a separate application from Biomek Software. The system administrator sets up and configures user accounts, passwords, and permissions, and configures system settings such as automatic password expiration and system logout time.

A single system administrator password is used on a system. System administration tasks may be performed only on the computer where Account Management is installed; access to Account Management over a network is not supported.

**Note:** The administrator is not automatically given a user account on the system. If the administrator requires an account, one must be created in Account Management.

**Note:** The 21 CFR Part 11 regulation contains additional requirements for account management beyond the control of the software.

This section covers opening Account Management and provides an overview of permissions available for Biomek Software users (refer to Section 2.3.1, <u>Assigning</u> <u>Biomek Software Permissions</u>). Refer to the Account Management online help for detailed information about administering user accounts and permissions.

To open Account Management:

- 1. Log off and close Biomek Software and any other Beckman Coulter software applications using Accounts & Permissions.
- 2. In the Windows<sup>®</sup> Start menu, choose **Settings>Control Panel**. Control Panel appears.
- In Control Panel, double-click Administrative Tools. Administrative Tools appears.
- In Administrative Tools, double-click Beckman Coulter Accounts & Permissions Administrator. Beckman Coulter Accounts & Permissions appears.
- In Beckman Coulter Accounts & Permissions, enter the administrator password and choose OK. Account Management appears (Figure 2-2).

Beckman Coulter Account	s & Permissions - Account	Management X
🚨 Accounts 🔐 Settings 🚺	🗊 Repositories 🛛 😨 Audit 🗍	
Berkeley BrianD cathyg Danni DGJUNES FPD Lambeau LMVelour MarkFu PDQ1 REJones Roget	Full Name: Berkeley Rattan Description: Set Description Permissions: Biomek® Software - Develo Biomek® Software - Rotkor Biomek® Software - Rotkor Biomek® Software - Ruhka Biomek® Software - Setup I Biomek® Software - Setup I Biomek® Software - Validate	Choose ? or press F1 to access online help.
Create New Account	Add Permissions	Remove Permissions

Figure 2-2. Account Management

#### 2.3.1 Assigning Biomek Software Permissions

Permissions provide the ability to control user access to program operations. The administrator assigns permissions for each user in Account Management. Table 2-1 describes the permissions available for Biomek Software users.

**Note:** Refer to Account Management online help for detailed information about assigning permissions.

Permission	Description		
Develop Methods	Allows the user to open and run methods that have not been validated, develop new methods and edit existing methods (refer to Chapter 12, <u>Creating and Using</u> <u>Methods</u> ).		
Develop Projects	Allows the user to create and edit project files (refer to Chapter 6, <u>Understanding and Using Project Files</u> ).		
Editor Preferences	Allows the user to modify <b>Preferences</b> and Log Configurations (refer to Section 29.2, <u>Changing</u> <u>Display Preferences</u> and Chapter 26, <u>Generating and</u> <u>Using Log Data</u> ).		
Run Validated Methods	Allows the user to open and run validated methods. A validated method is a revision of a method that cannot be modified or associated with a different instrument (refer to Section 12.15, <u>Checking Out a Method</u> ).		
Setup Instrument	Allows the user to configure instrument settings in Hardware Setup (refer to Chapter 4, <u>Configuring</u> <u>Hardware Setup</u> ), the Deck Editor (refer to Chapter 5, <u>Preparing and Managing the Deck</u> ), the Device Editor (refer to Section 22.4.1, <u>Configuring Devices</u> <u>Using the Device Editor</u> ), and the Import/Export Utility (refer to Section 3.4, <u>Sharing Instrument</u> <u>Settings Using the Import/Export Utility</u> ).		
Use Manual Control	Allows the user to control instrument actions outside a method using Manual Control (refer to Chapter 27, Using Manual Control).		
Validate Methods	Allows the user to validate a revision of a method. Once validated, that revision of the method cannot be modified or associated with a different instrument (refer to Section 12.15, <u>Checking Out a Method</u> ).		

Table 2-1. Biomek Software Permissions

# 2.4 Restoring the Administrator Password

Only one administrator account exists on a system with Accounts & Permissions installed. If the administrator password is lost or forgotten, Beckman Coulter Customer Technical Support must be contacted to restore access to the Account Management application.

To restore the administrator password:

- 1. Place the Beckman Coulter Accounts & Permissions Installer CD in the drive, and browse to the contents of the CD.
- 2. Double-click Beckman Coulter Accounts & Permissions Admin Password Restore.exe. Password Restore appears (Figure 2-3).

Beckman Coulter Accounts & Permissions - Password Restore		
Give this to technical support (do not close this app until you get the code back)		
B928-7D2E-410B-4030-9BA5-0D8D-XXXX-XXXX		
Type the code technical support gives you here::		
OK Cancel		

Figure 2-3. Password Restore

3. Contact Beckman Coulter Customer Technical Support and provide the code displayed in the upper field of Password Restore.

**Note:** Leave Password Restore open until Beckman Coulter Customer Technical Support supplies a new code. The new code is based on the code displayed in the upper field, which changes each time Password Restore is opened.

- 4. In the lower field of **Password Restore**, enter the new code provided by Beckman Coulter Customer Technical Support.
- 5. Choose **OK** to close **Password Restore** and accept the new code.

OR

Choose Cancel to close Password Restore without accepting the new code.

6. Follow any additional instructions provided by Beckman Coulter Customer Technical Support.

# 2.5 Viewing the Audit Log

Audit Log displays the audit trail for all user activity in software applications that support Beckman Coulter Accounts & Permissions. System activity is logged, even when Accounts & Permissions is set to No Support (refer to Section 2.2.2, <u>Setting</u> the Level of Support For Accounts & Permissions).

**Note:** Administrator activity other than changing the level of support for Accounts & Permissions is not available in the Audit Log. However, all administrator activity is logged and may be viewed in Account Management.

To view the Audit Log:

- 1. In the Windows<sup>®</sup> Start menu, choose **Settings>Control Panel**. Control Panel appears.
- In Control Panel, double-click Administrative Tools. Administrative Tools appears.
- 3. In Administrative Tools, double-click **Beckman Coulter Accounts & Permissions Audit Log**. The Audit Log appears (Figure 2-4).

🕮 Beckman Coulter Accounts & Permissions Audit Log	_ IOI XI
8/6/2004 9:56:11 AM: Berkelev (Berkelev Rattan) - Account Berkelev (Berkelev Rattan) logged in.	
8/6/2004 9:55:18 AM: Berkeley (Berkeley Rattan) - Account Berkeley (Berkeley Rattan) logged off.	
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 1 of record BiomekFX\Biomek\Tip Classes\AP96_P1000_LL5_Barrier was checke	ed in b
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - The working revision of record BiomekFX\Biomek\Tip Classes\AP96_P1000_LLS_Barrier v	vas ch
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 1 of record BiomekFX\Biomek\Tip Classes\AP96_P1000_LLS was checked in by I	Berkel
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - The working revision of record BiomekFX\Biomek\Tip Classes\AP96_P1000_LLS was check	:ked in
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 2 of record Blomekt-XtBlomekt) echniques(standard was checked in by Berkeley 0//2004 0.54:40 AM: Berkeley (Berkeley Rattan) - Revision 2 of record Blomekt-XtBlomekt) echniques(standard was checked in by Berkeley	/ (Berk
8)6/2004 9:54:49 APR: Berkeley (Berkeley Rattan) - The Working revision or record Biomek-X (Biomek) (echniques)standard was checked in D 9/c/2004 0:54:40 APR existence (Berkeley Datas). Devision of record Biomek-X (Biomek) (echniques)standard was checked in D	iy Berk
o (a) 2004 9134149 AMI Berkeley (Berkeley Rattan) - Revision 1 or record biometry (biometry (bio	by bei becke
9/0/2004 9:54:49 AM: Barkeley (Barkeley Rattan) - The working revision on record biomek Applications, reciming estimates to induce species was a 8/6/2004 9:54:49 AM: Barkeley (Barkeley Rattan) - Revision 1 of record Biomek EV/Biomek/Techniques/P1000 Defailt was checked in by Bar	riecke kelev
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - The working revision of record Biomek, teaming and the second biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was revision of record Biomek (Biomek) techniques) P1000 Default was rev	d in h
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 2 of record BiomekEXBiomekTechniquesLow-Volume was checked in by Berke	elev (B
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - The working revision of record BiomekFX\Biomek\Techniques\Low-Volume was checked in	n by B
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 1 of record BiomekFX\Biomek\Techniques\Low-vol CirculatingReservoir was che	ecked i
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - The working revision of record BiomekFX\Biomek\Techniques\Low-vol CirculatingReservo	bir was
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 2 of record BiomekFX\Biomek\Techniques\Low Volume 384 Plate was checked in	n by Bi
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - The working revision of record BiomekFX\Biomek\Techniques\Low Volume 384 Plate was	check
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 1 of record BiomekFX[Biomek] Techniques[CirculatingReservoir Span-8 Low 80 v 0// 2004 0.51:40 AM: Berkeley (Berkeley Rattan) - Revision 1 of record BiomekFX[Biomek] Techniques[CirculatingReservoir Span-8 Low 80 v	Nas ch
8)6/2004 9:54:49 Am; Berkeley (Berkeley Rattan) - The Working revision or record Biomek-X(Biomek) (arculatingkeservoir Span- 9)6/2004 9:54:49 Am; Berkeley (Berkeley Rattan) - The Working revision or record Biomek-X(Biomek) (arculatingkeservoir Span- 9)6/2004 9:54:49 Am; Berkeley (Berkeley Rattan) - The Working revision or record Biomek-X(Biomek) - Charley Berkeley (Berkeley Battan) - Span-	8 LOW
o (b) 2004 9134149 AMI: Berkeley (berkeley Rattail) - Revision 1 record biomer A (biomer) (biometric) and (receiver our spano was cried	Reu III Rimori -
0/02/00 7 3-3 47 Denkely (Denkely Ration) - The Working Testion on Four Distribution Autometer (Four Indues) (Circulating reservoir Spanic 8/6/2004 9-54-49 AM: Barkalav (Barkalav Dattan) - Devicing 1 of record Bioma/EY/Bioma/Pachoruse) (Circulating Reservoir Spanic	hy Ber
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - The working revision of record Biomek/Silliomek/Sechniques/CirculatingReservoir was checked in the second biomek/Silliomek/Sechniques/CirculatingReservoir was checked in the second biomek/Silliomek/Sechniques/CirculatingReservoir was checked in the second biomek/Sechniques/CirculatingReservoir was checked in the second biomek/Sechniques/Se	becker
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 1 of record Biomek/FX\Biomek\Pipetting Templates\Span-8 P1000 was checked i	in by E
8/6/2004 9:54:49 AM; Berkeley (Berkeley Rattan) - The working revision of record BiomekFX\Biomek\Pipetting Templates\Span-8 P1000 was	s checl
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 1 of record BiomekFX\Biomek\Pipetting Templates\CirculatingReservoir Span-8	was cl
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - The working revision of record BiomekFX\Biomek\Pipetting Templates\CirculatingReservo	pir Spa
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 1 of record BiomekFX\Biomek\Pipetting Templates\CirculatingReservoir was che	ecked i
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - The working revision of record BiomekTX\Biomek\Pipetting Templates\CirculatingReservo	pir was
8/6/2004 9:54:49 AM: Berkeley (Berkeley Rattan) - Revision 2 of record Biometk/Biometk/Liquid Types/Water was checked in by Berkeley ( 0// 2004 0:54:40 AM: Berkeley (Berkeley Rattan) - Revision 2 of record Biometk/Liquid Types/Water was checked in by Berkeley (	Berkel
opjourd 9:54:49 Am: berkeley (Berkeley Rattan) - The working revision of record biomekr-x(biomek)Liquid Types(Water was checked in by	berke
	►
Defends   Emeral	
Kerresn Export	

Figure 2-4. Accounts & Permissions Audit Log

4. Choose **Export** to export the entire audit log to a text file, if desired. The exported file may be opened, read, and printed in any application that supports text files.

**Note:** When the Audit Log is open while Biomek Software is running, system activity is logged, but not refreshed automatically in the Audit Log. Choose **Refresh** to to view system activity logged since the Audit Log was opened or the last time displayed data was refreshed.

# 2.6 Using the Biomek Software With Accounts & Permissions Enabled

When Beckman Coulter Accounts & Permissions is enabled, all users must log on to use Biomek Software. Only one user may be logged on at a time. Users are permitted to perform only the operations for which the administrator has assigned them permission.

**Note:** This section covers logging on and off Biomek Software. Refer to Chapter 12, <u>Creating and Using Methods</u> for information about checking in, validating, and signing methods when Accounts & Permissions is enabled. Refer to Section 6.4, <u>Deleting and Restoring Project Files</u> for information about checking in projects.

#### 2.6.1 Logging Onto the Biomek Software

To log onto Biomek Software:

1. On the toolbar, choose the log on/off button (Figure 2-5). Logon appears (Figure 2-6).

Log on/off button
No user is currently logged on.

Logon	
Ausermus	t be logged in to use the Biomek software.
User Name:	
Password:	
	OK Cancel

Figure 2-6. Logon

2. Enter the account User Name and Password.

**Note:** The first time a user logs on using a new account, or after having the password changed by the administrator, Biomek Software automatically prompts for the password to be changed.

The new password must be different from the original and may include alphanumeric characters and spaces and be up to 250 characters in length. Accounts & Permissions passwords are case insensitive.

3. Choose **OK** to log onto Biomek Software. The name of the user logged on appears to the left of the log on/off button (Figure 2-8).

#### 2.6.1.1 Handling Disabled Accounts

User accounts may be disabled by the administrator, or automatically after a number of logon attempts for the account fail. Allowing accounts to be automatically disabled after a number of failed logon attempts is optional, and configured by the administrator.

When an account is automatically disabled, Administrator Notification appears (Figure 2-7). The administrator password must be entered before Biomek Software can be accessed by users.



Figure 2-7. Administrator Notification

#### 2.6.1.2 Viewing the Logged On User's Permissions

To view the permissions of the user currently logged onto Biomek Software:

1. On the toolbar, choose the user name displayed next to the log on/off button (Figure 2-8). Current Permissions appears (Figure 2-9).



Figure 2-8. Log on/off button — user logged on

Current Permissions			
Current user: Berkeley Rattan ID: Berkeley			
Permission	Description		
✓ Develop Projects	Create, modify, or delete projects		
✓ Develop Methods	Create, modify, and run Biomek® methods		
✓ Validate Methods	Validate Biomek® methods		
✓ Run Validated Methods	Run validated Biomek® methods		
✓ Editor Preferences	Change the Biomek® Software preferences		
✓ Setup Instrument	Change the Biomek® instrument setup		
🗸 Use Manual Control	Control the Biomek® instrument via manual control		
	ОК		

Figure 2-9. Current Permissions

2. Choose OK to close Current Permissions.

#### 2.6.2 Logging Off the Biomek Software

To log off Biomek Software:

On the toolbar, choose the log on/off button (Figure 2-10). The user is logged off the system.

**Note:** Closing Biomek Software will not automatically log a user off. The current user will remain logged on until the software is reopened and the account is logged off, or until the login times out automatically.



Figure 2-10. Log on/off button — user logged on

# Using Instrument Files and Settings

### 3.1 Overview

An instrument file stores information about the hardware configuration and deck layout of the Biomek instrument. Instrument files can represent different Biomek instruments, or different hardware configurations for the same instrument.

The instrument configuration must accurately represent the Biomek instrument hardware to prevent collisions between pods and any on-deck components. All instrument configuration is done using Hardware Setup (refer to Chapter 4, *Configuring Hardware Setup*) and the Deck Editor (refer to Chapter 5, *Preparing and Managing the Deck*).

The **Instrument** menu contains options to open and save instrument files, as well as provide access to the editors to configure the instrument file.

The sections in this chapter include:

- <u>Saving and Opening Instrument Files</u> (Section 3.2) Save the current configuration of an instrument or open an existing instrument file.
- <u>Restoring Instrument Settings from Backup Files</u> (Section 3.3) If settings are deleted or lost, workspace settings or an entire workspace can be restored from one of the automatic daily, weekly, or monthly backup files created by Biomek Software.
- <u>Sharing Instrument Settings Using the Import/Export Utility</u> (Section 3.4) Settings may be shared with another host computer or sent to Beckman Coulter Technical Support for analysis using the Import/Export Utility. This utility allows instrument settings (decks, framing tools, pod settings) to be imported and exported as import files (.imp).

#### 3.1.1 Instrument Display in Method Editor

The active instrument file and project file are displayed in the status bar of the Biomek method editor (Figure 3-1). Any instrument file can be used with any project file and method. Validated methods, however, are validated for a specific instrument and are no longer validated when used with a different instrument file, even if the instrument file is equivalent to the one on which the method is validated (refer to Section 12.11, *Validating a Method*).

**Note:** A validated method for a specific instrument is not the same as validating a method before a run to internally test the method for errors before it is actually run. To validate a method to test for errors before it is run, check **Validate the current method before running it** in Preferences (refer to Section 29.2, <u>Changing</u> <u>Display Preferences</u>).



Figure 3-1. Biomek method editor

#### 3.1.2 Contents of an Instrument File

An instrument file contains all:

- Hardware configurations information about the instrument and integrated ALPs or devices as configured in Hardware Setup.
- Pod settings axis limit values and other settings for the pod(s).
- Deck layouts and framing information layout of ALPs and devices on the instrument deck as configured in the Deck Editor.

**Note:** When an instrument file is loaded or imported, the deck framing is reset to **Unframed** and the deck must be framed again.

# 3.2 Saving and Opening Instrument Files

Instrument files are saved as \*.bif files that may be opened on any system in Biomek Software.

Instrument files can represent different Biomek instruments, or different hardware configurations for the same instrument. These instrument files can be used to:

- quickly change the instrument configuration between two or more commonly used configurations for an instrument.
- change instruments when developing methods in simulation mode for multiple instruments on one computer.
- share the instrument configuration with other users or Beckman Coulter Technical Support to ensure a method is being run using an equivalent instrument.

#### 3.2.1 Saving an Instrument File

An instrument file may be saved at any time. The instrument file is automatically saved when another instrument file is opened or Biomek Software is closed; however, to save the current instrument file without closing it, use Save Instrument.

To save changes made to an instrument file:

From the Instrument menu, choose **Save Instrument**. The active instrument file is saved.

#### 3.2.2 Creating a New Instrument File

A new instrument file is created by saving the current active instrument file with a new file name. The new instrument file has the same settings as the active instrument when it was saved; however, any changes made to the original instrument file are saved as the new instrument and not with the original instrument file.

To create a new instrument file:

1. From the Instrument menu, choose **Save Instrument As**. Save Instrument appears (Figure 3-2).

Save Instrument	<u>?</u> ×
Save in: 🖻 Biomek 🔹 🖛 🖽 🖝	
Backup	
Logs	
Biomek3000-2.bif	
Biomek3000.bif	
File name: Biomek3000.bif Sav	е
Save as type: Instrument Files (*.bif)	el

Figure 3-2. Save Instrument

- 2. In **Save Instrument**, browse to the desired drive and directory in which to save the instrument file.
- 3. In File Name, enter the desired file name for the instrument file.
- 4. Choose **Save**. The instrument file is saved in the selected drive and directory with the specified file name and opened as the active instrument file.

#### 3.2.3 Opening an Existing Instrument File

Any existing instrument file may be opened at any time to develop methods using a specific instrument configuration.

To open an existing instrument file:

1. From the Instrument menu, choose Open. Open Instrument appears.

Open Instrun	ment	N N
Look in: 📔	)Biomek 💌 🖛 🛍 📸 💷 -	
🗎 Backup		
Logs		- 11
Biomek300	00-2.bif	- 11
Biomek300	00.bif	
File name:	Biomek3000.bif Open	
Files of type:	Instrument Files (*.bif)	

Figure 3-3. Open Instrument

2. In Open Instrument, browse to and select the desired instrument file (\*.bif) to open.

**Note:** To open one of the daily, weekly, or monthly backup files, enter **\*.\*backup** in File name and press **Enter**. All the backup files are displayed in Open Instrument.

3. Choose **Open** to open the selected instrument file. The selected instrument file is opened.

# 3.3 Restoring Instrument Settings from Backup Files

Backup files are created daily, weekly, and monthly by Biomek Software whenever an instrument file is saved, typically when **File>Save Instrument** is selected or when Biomek Software is closed.

**Note:** By default, backup files for instruments are saved in the C:\Documents and Settings\All Users\Shared Documents\Biomek\ directory with the following naming convention:

<Instrument File Name>.bif.DailyBackup <Instrument File Name>.bif.WeeklyBackup <Instrument File Name>.bif.MonthlyBackup

To restore an instrument file from a backup file when settings are deleted or lost:

- 1. Close the Biomek main editor.
- 2. Open Windows Explorer and locate the instrument file (\*.bif) in the shared application data directory. Typically, the path to this directory is: C:\Documents and Settings\All Users\Shared Documents\Biomek\.
- 3. If the instrument file needs to be replaced rather than recovered, rename the latest instrument file to <Instrument File Name>.old.

Note: Rename the original only if it is available.

OR

If the instrument file needs to be recovered, rename the backup file to the instrument file name, such as <Instrument File Name>.bif.DailyBackup to <Instrument File Name>.bif.

4. Open the Biomek main editor. The instrument file has been restored.

# 3.4 Sharing Instrument Settings Using the Import/ Export Utility

Specific segments of a current instrument file, called instrument file settings, can be restored using the Import/Export Utility (Figure 3-4). Exporting settings allows settings to be shared with another system, while importing settings allows settings developed on another system to be used. Settings are imported from and exported to import files (.imp).

To share a method between two computers, any instrument settings different from the default installed with the software must be exported from the original computer and imported to the new computer using the Import/Export Utility.

Figure 3-4 shows how the Import/Export Utility works. The Instrument side of the utility displays the list of settings in the current instrument file. The Import File side of the utility displays the list of settings in the currently opened import file.

To restore settings of a current instrument file, settings from Instrument are dragged (exported) to the Import File. From Import File, these import settings can be dragged (imported) to the current Instrument.

**Note:** Settings may also be imported or exported by choosing **Import** or **Export** on the toolbar.

Terrent /Freesh Likiliter	
New 🚔 Open 🔚 Save   → Export ← Imp	ort E Close
Instrument	Import File
Instrument	Import File Displays list of settings in the current import file.
Import settings from the import	t file to the current instrument.
Drag and drop workspace elements to import or export.	

Figure 3-4. Import/Export Utility

#### 3.4.1 Exporting Settings to an Import File

To export settings to an import file:

1. From the Instrument menu, choose Import/Export Utility. Import/Export Utility appears (Figure 3-5).

Import/Export Utility			
│ 🗋 New 😂 Open 🔚 Save 🛛 → Export ← Imp	ort Close		
Instrument	Import File		
<ul> <li>Instrument</li> <li>Deck Layouts</li> <li>Deck1</li> <li>Standard</li> <li>Framing Tools</li> <li>Pod Settings</li> </ul>	Untitled Deck Layouts Deck1		

Figure 3-5. Exporting deck layout Deck1 to an import file

2. Drag and drop the desired settings from Instrument to Import File.

OR

Highlight the desired settings from Instrument and choose **Export**. The folder containing the desired setting and the setting itself appear in the Import File (Figure 3-5).

**Note:** Dragging a folder copies all of the settings in that folder

3. Continue to export settings using step 2 until all desired settings are exported.

4. Choose Save to save the Biomek Import File. Save As (Figure 3-6) appears.

Save Import File	<u>?</u> ×
Save in: 🕒 My Documents 💽 🖛 🗈 💣 💷 ד	
🛅 Biomek	
BiomekFX	
My eBooks	
📸 My Music	
My Pictures	
My Videos	
File name: My First Biomek Import Sav	e
Save as type: Biomek® Import Files	el

Figure 3-6. Save Import File

- 5. Browse to the desired file location to save the import file.
- 6. In File name, enter a name for the import file.
- 7. Choose Save. The import file is created and Save As closes.

**Note:** Note that the name of the new import file is now displayed on the top of the Import File window.

8. Choose Close. Import/Export Utility closes.

**Note:** When exporting a Deck with a stationary labware type, such as a Wash Station, the Wash Station must be present in the destination workspace or exported to the Import File.

#### 3.4.2 Importing Settings from an Import File

To import settings from an import file:

- 1. From the Instrument menu, choose Import/Export Utility. Import/Export Utility appears (Figure 3-5).
- 2. Choose **Open** and open the .imp file.
- 3. Open the desired folder by clicking the + containing the desired settings from the import file.

**Note:** If a setting with the same name is used in the import file and in the current instrument, it can be overwritten.

4. Drag and drop the setting from Import File to Instrument to overwrite the setting in the current workspace with the one from the import file.

OR

Rename the setting in the Import side, then drag and drop the setting from Import File to Instrument to not overwrite the setting.

**Note:** Rename a setting by highlighting the setting, single-clicking, pausing until the name becomes editable, typing a new name, and pressing return.

Note: Dragging an entire folder copies all of the settings in that folder.

**Note:** If a setting being imported exists in the instrument file and, therefore, can be overwritten, a **Confirm** similar to Figure 3-7 appears. Answer **Yes** to overwrite the setting in the instrument file. Answer **No** to cancel the import of that setting. **Yes** to All and **No** to All allow multiple imported settings to overwrite the contents of the instrument file or be cancelled at the same time.

Confirm	×
Ŷ	The deck layout "Deck1" already exists in your workspace. Overwrite it?
	Yes to All Yes to All

Figure 3-7. Confirmation to address to overwrite an imported setting

- 5. Continue to import settings using step 4 until all desired settings are imported.
- 6. Choose Close to exit the Import/Export Utility.
- 7. Select **Instrument>Save Instrument** to save any changes made to the workspace.

**Note:** When Biomek Software is updated, a backup of the instrument file is made to save previous settings. The backup instrument file is located in C:\Documents and Settings\All Users\Shared Documents\Biomek\Backup\.

#### 3.4.3 Restoring Instrument Settings from Backup Files

To restore instrument file settings from a backup file, such as when settings are deleted or incorrectly changed:

1. From Instrument, choose **Open** to open the backup file for the instrument file. Open Instrument appears (Figure 3-8).

Open Instrum	nent	<u>?</u> ×
Look in: 🛅	Biomek 💌 🔄 🖆 🔠 🕇	
Backup		
Biomek300	0-2.bif	
Biomek300	0.bif	
File name:	Biomek3000.bif Ope	n
Files of type:	Instrument Files (*.bif)	el /
		111

Figure 3-8. Open Instrument file

2. In File name, enter **\*.\*backup** and press **Enter** to show all daily, weekly, and monthly backup files in Open Instrument (Figure 3-9).

Open Instrument					<u>?</u> ×
Look in: 🛅 Biomek	:	•	← 🗈 🖻	* 📰 🕶	
Backup Logs Biomek3000-2.bif. Biomek3000-2.bif. Biomek3000-2.bif. Biomek3000.bif.Da	DailyBackup MonthlyBackup WeeklyBackup iilyBackup	Biomek3000.b	if . MonthlyBa if . WeeklyBac	ckup :kup	
File name: <sup>*</sup> .*bac	kup			Oper	
Files of type: Instru	ment Files (*.bif)		•	Cance	el //

Figure 3-9. Open Instrument showing backup files

- 3. Select the desired backup file.
- 4. Choose **Open**. The selected backup file is opened as the current instrument.

- 5. Use the Import/Export Utility to create an import file with the desired settings from the backup instrument file (refer to Section 3.4.1, *Exporting Settings to an Import File*).
- 6. From the Instrument menu, choose Open Instrument.
- 7. Select the original instrument file, named <Instrument File>.bif.
- 8. Choose Open.
- 9. Use the Import/Export Utility to import settings from the import file created in step 5 (refer to Section 3.4.2, *Importing Settings from an Import File*).

**Note:** When Biomek Software is updated, a backup of the instrument file is made to save previous settings. The backup instrument file is located in C:\Documents and Settings\All Users\Shared Documents\Biomek\Backup\.

# 4 Configuring Hardware Setup

# 4.1 Overview

Hardware Setup is used to configure Biomek Software with the appropriate Biomek instrument information, including the instrument type and which heads and devices are available for use. The Biomek Simulator, which shows a 3-D animation of the instrument performing methods, is also configured in Hardware Setup.

After a Beckman Coulter Service Engineer has physically installed a device, the device is detected on the Biomek instrument and must be properly installed and configured in Hardware Setup. While the Service Engineer normally installs and configures new devices, it may be necessary to install, configure, and remove other devices using Hardware Setup.

For specific information on configuring Biomek instruments in Hardware Setup, refer to the appropriate hardware user's manual:

- FX <u>Biomek® FX Laboratory Automation Workstation User's Manual</u>.
- 3000 <u>Biomek® 3000 Laboratory Automation Workstation User's</u> <u>Manual</u>.
- NX-MC <u>Biomek® NX Multichannel Laboratory Automation Workstation</u> <u>User's Manual</u>.
- NX-S8 <u>Biomek® NX Span-8 Laboratory Automation Workstation User's</u> <u>Manual</u>.

Refer to <u>*ALPs User's Manual*</u> or the appropriate integration manual for information on configuring specific ALPs and devices.

The sections in this chapter include:

- <u>Accessing Hardware Setup</u> (Section 4.2).
- <u>Understanding the Options in Hardware Setup</u> (Section 4.3).
- <u>Configuring the Biomek Simulator</u> (Section 4.4).

# 4.2 Accessing Hardware Setup

Hardware Setup is accessed from within Biomek Software.

To access Hardware Setup:

From the Instrument menu, choose **Hardware Setup**. Hardware Setup appears (Figure 4-1).

Biomek® Hardware Setup		
👔 Reconnect 🔺 Home All Axes 🛛	🖶 Add Device 👞 Remove Device 🖌 Accept 🔀 Cancel	
Biomek@ FX (SN: None) Characterization Biomek@ FX (SN: None) Characterization Biomek@ FX (SN: None) Characterization Biomek@ FX (SN: None) Characterization Devices ShakerALP0 ShakerALP0 ShakerALP1 SpeedPump0 Simulator Digital Devices WashPump1 Digital Devices Camera Camera Stacker Carousels	Serial Number: Port: Simulate This is a dual-armed system Left Pod Type: Left Multichannel Pod Right Pod Type: Right Span-8 Pod T	
BiomekFX		

Figure 4-1. Hardware Setup for a Biomek FX instrument with a Multichannel Pod and Span-8 Pod

**Note:** The devices displayed in **Hardware Setup** are dependent on instrument type and configuration.

# 4.3 Understanding the Options in Hardware Setup

An understanding of the options on the toolbar in Hardware Setup is necessary to properly install, configure, and remove devices.

Table 4-1 lists and describes the toolbar options in Hardware Setup:

#### Table 4-1. Hardware Setup Options

Option	Description
Reconnect	Allows Hardware Setup to reexamine the devices present. Choose this option to determine what devices are present rather than closing and reopening Hardware Setup.
Home All Axes	Gives the instrument a point of reference from which to make subsequent moves. For a single- pod system, home position is left, back. For a dual- pod system, home position for the first (left) pod is left, back and for the second (right) pod is right, back. <b>Note:</b> Pods should be homed each time the instrument is powered on. Depending on the type of pods on the system, a Warning appears. After confirming that the actions have been addressed properly, choose <b>OK</b> .
Add Device	Installs a device.
Remove Device	Removes a device.
Accept	Saves all changes to the instrument and closes Hardware Setup. Choose this option after the device has been installed and configured.
Cancel	Closes Hardware Setup without saving the modifications to the instrument.

### 4.4 Configuring the Biomek Simulator

When a method is run in Simulate mode, the Biomek Simulator appears, showing an animated 3-D model of the instrument performing the method. Simulator settings can be configured in Hardware Setup.

**Note:** Modifying simulator settings is optional. The default settings are appropriate for most computers.

Configuring the simulator includes:

- <u>Configuring Simulator Settings</u> (Section 4.4.1).
- <u>Configuring Camera Controls</u> (Section 4.4.2).
- <u>Configuring Materials</u> (Section 4.4.3).

#### 4.4.1 Configuring Simulator Settings

Simulator configures view settings, including animation speed, simulator display size, and image quality. Many of the view settings increase or decrease the image quality of the 3-D model shown in simulations. In general, increasing image quality, such as enabling blending, requires a faster computer to run the animated simulation smoothly. Conversely, decreasing image quality requires less computing power, and will run simulations more smoothly on slower computers.

The default settings in Simulator provide a good balance between image quality and smooth animation playback on most computers. Change the settings only if higher image quality or smoother playback is desired.

To configure simulator settings:

1. From the left pane of Hardware Setup, choose **Simulator**. The Biomek Simulator settings appear (Figure 4.2).



Figure 4-2. Hardware Setup — configuring the simulator settings

- 2. In Warp Factor, change the playback speed for simulated methods. Warp Factor 1 runs simulated methods in real time.
- 3. Enter the **Max Height** in pixels for the simulator display.
- 4. Enter the **Max Width** in pixels for the simulator display.
- 5. In Max Poly Size, enter the maximum size of the polygons that make up the model of the instrument shown in the simulator. Decreasing the polygon size creates a more detailed simulator view, but requires more computing power. Increasing the polygon size creates a less detailed simulator view, but requires less computing power.
- 6. Choose the method of Perspective Correction:
  - **Nicest** performs more accurate perspective correction; better for faster computers.
  - **Fastest** performs less accurate correction; better for slower computers.
  - **Don't Care** Biomek Software automatically chooses the most appropriate correction method.
- 7. Choose **Use Textures** to include textures, such as microplate wells, on the instrument and labware surfaces displayed in the simulator.
- 8. If **Use Textures** is chosen, choose a **Labware Texture Size**, if desired. Dragging the slider to the right increases texture detail; dragging to the left reduces texture detail.
- 9. Change Tesselation Type to **Triangle Fan** or **Quad Strip** if a slight enhancement in image quality is desired. The default setting, **Triangle Strip**, is adequate for most computers.
- 10. Choose the colored box to the right of Background Color to change the background color in the simulator display. Color appears (Figure 4-3).



Figure 4-3. Selecting a background color

11. Select a new color, or define a custom color for the simulator background.

12. Choose **OK** to set the new background color.

OR

Choose **Cancel** to retain the original background color.

13. Select **Blending** to render semi-transparent objects, such as tips, correctly.

**Note:** When Blending is disabled, semi-transparent objects appear opaque; for example, liquid is not visible inside tips. Leaving Blending disabled is recommended for slower computers.

- 14. Select **Round Edges** to smooth out and round the edges of the instrument in the simulator display.
- 15. Select **Show camera navigation buttons** to display camera navigation controls by default during simulated method runs and in **Simulator Preview**.

**Note:** The camera is the position from where the 3-D model is being viewed. The camera navigation controls are used to move or rotate the camera, which changes the viewing angle of the model in the simulator.

**Note:** Camera navigation controls may also be toggled on and off during simulated method runs by pressing the tilde ( $\sim$ ) key to enable the simulator controls (refer to Section 12.18.1.1.1, <u>Using the Simulator Controls</u>).

- 16. Select **Use hardware OpenGL acceleration if available** to provide smoother rendering of the model and animation on supported computers.
- 17. Hide the bridge or canopy of the instrument to provide an unobstructed view of the pod(s) performing the steps in the method.
  - **FX**, 3000 Select Hide front section of pod bridge.
  - > NX-MC, NX-S8 Select Hide Canopy.



18. Choose **Preview** to access Simulator Preview at any time during simulator configuration (refer to Section 4.4.1.1, *Viewing the Simulator Preview*).

#### 4.4.1.1 Viewing the Simulator Preview

Simulator Preview shows the 3-D model of the instrument, including current configuration changes, as it appears in animated simulations. Simulator Preview is accessible from Simulator, Camera, or Materials at any time.

To view the Simulator Preview:



1. Choose **Preview** to view changes made to the settings, if desired. Simulator Preview appears (Figure 4-4).

Simulator Preview	
↑ ↑     ♪     ♪     Restore camera	position 1 2 3 4 5 6 7 8 9 0
Store camera po	sition 1 2 3 4 5 6 7 8 9 0
	🗸 ОК

Figure 4-4. Simulator Preview with camera navigation buttons enabled

2. Change the view by placing the cursor in the simulator display, then clicking and dragging the mouse in the desired direction of rotation.

#### OR

Change the view by clicking the cursor in the simulator display and using the keyboard control keys defined in **Camera** (refer to Section 4.4.2, <u>Configuring</u> <u>Camera Controls</u>).

OR

Use the camera navigation buttons to change, save, and reset the view (Figure 4-5). The camera navigation buttons are displayed only when enabled in Simulator (refer to Section 4.4.1, *Configuring Simulator Settings*).

**Note:** The camera is the position from where the 3-D model is being viewed. The camera navigation buttons move or rotate the camera, which changes the angle at which the model is being viewed in the simulator.



Figure 4-5. Camera Navigation buttons

3. Choose **OK** to close Simulator Preview and return to the simulator configuration in Hardware Setup.
#### 4.4.2 Configuring Camera Controls

The camera is the position from where the 3-D model of the instrument is being viewed. To change the view of the model, the camera can be moved using the keyboard, mouse, or optional camera navigation buttons. In **Camera**, movements are called **Actions**. Keyboard and mouse controls are called **Bindings**. Keyboard and mouse controls can be created, edited, or deleted in **Camera**.

**Note:** Camera navigation buttons are enabled and disabled in Simulator (refer to Section 4.4.1, *<u>Configuring Simulator Settings</u>*).

To add or modify a camera control:

1. From the left pane of Hardware Setup, expand the **Simulator** tree, if necessary, and choose **Camera**. Camera appears (Figure 4-6).



Figure 4-6. Hardware Setup — configuring camera controls

- 2. In the Actions pane, select the camera control to modify. The current keyboard and mouse control settings appear in Bindings.
- 3. In Bindings, select the binding to delete or modify.
- 4. Choose **Delete Bind** to delete the selected binding.

#### OR

壷

Modify the binding. Modifications include:

- <u>Creating or Editing a Mouse Binding</u> (Section 4.4.2.1).
- <u>Recording or Editing a Key Binding</u> (Section 4.4.2.2).

#### 4.4.2.1 Creating or Editing a Mouse Binding

Mouse bindings control camera actions using mouse controls. They may be configured to use a single mouse control, a combination of mouse controls, or a combination of mouse controls and modifier keys on the keyboard.

**Note:** The mouse buttons, wheel, and directional movement are all considered mouse controls.

To create or edit a mouse binding:

<u> </u>
----------

1. To create a new mouse binding, choose **Add Mouse Bind**.

OR

To edit an existing mouse binding, select the desired binding.

- 2. In Mouse Command, choose the desired command.
- 3. In Modifiers, select any desired keyboard modifiers and/or mouse commands.

**Note:** The Ctrl, Alt, and Shift keys are keyboard modifiers. Mouse command modifiers are Right, Left, and Middle, which represent mouse buttons and Double, which represents double-clicking.



- 4. If creating a new mouse binding, choose **Add Mouse Bind** to add the new binding to the selected **Action**.
- To test the mouse binding, choose **Preview** to access Simulator Preview at any time during simulator configuration (refer to Section 4.4.1.1, <u>Viewing the</u> <u>Simulator Preview</u>).

#### 4.4.2.2 Recording or Editing a Key Binding

Key bindings control camera actions with keys on the keyboard. They may be configured to require a single key, such as "T", or a combination of a key and modifier.

To record or edit a key binding:



1. To create a new key binding, choose **Record Key Bind**.

OR

To edit an existing key binding, select the desired binding.

- 2. If creating a new binding, press the desired key.
- 3. In Modifiers, select a modifier key, if desired.



 To test the key binding, choose **Preview** to access Simulator Preview at any time during simulator configuration (refer to Section 4.4.1.1, <u>Viewing the</u> <u>Simulator Preview</u>).

#### 4.4.3 Configuring Materials

Each component of the 3-D instrument model is made of a material which has visual properties, such as color and reflectivity, specified. Properties can be edited in **Materials**. However, configuring the properties of materials is not necessary or recommended.



Materials can also be created and deleted using the buttons in Materials. Materials *should never* be deleted, because the deleted material will no longer be visible in the simulator. Likewise, materials created by users will not appear in the simulator either.

**Note:** Always preview changes in Simulator Preview before saving the new configuration. Refer to Section 4.4.1.1, *Viewing the Simulator Preview*, for more information.

To configure Materials:

1. From the left pane of Hardware Setup, expand the **Simulator** tree, if necessary, and choose **Materials**. Materials appears (Figure 4.4).



Figure 4-7. Hardware Setup — configuring materials

**Note:** In Materials, select the material to edit. The properties of the material appear. Table 4-2 provides descriptions of each material

2. In Properties, *do not* change the Name. Changing the name of a material causes it to no longer be visible in the simulator.

3. Choose the colored box to the right of a color parameter to change the color of the material. Color appears.

**Note:** Ambient and Diffuse define the color of the material when light is present. Specular defines the color of specular highlights (glare). Emission defines the color of the material in the absence of light.

- 4. In Color, select a new color, or define a custom color for the color parameter.
- 5. Enter a new value for Shininess, if desired.

**Note:** Shininess defines the reflectivity of the material on a scale of 0 - 100, with 100 being the most reflective.

6. To view changes made to materials, choose **Preview**. to access **Simulator Preview** at any time during simulator configuration (refer to Section 4.4.1.1, <u>Viewing the Simulator Preview</u>).

Material	Description
ALPBase	Base or stand of an ALP.
AmberLightCurtainDisplay	Color of the front panel indicator light during a method run.
BarcodeBeam	Laser beam emitted by a barcode reader.
BlackPlastic	Biomek 3000 tool bodies.
DAxisIndicator	Color band that encircles the head to show upward and downward movement.
Deck	Instrument deck.
Default	Default color of a material when deleted from the simulator.
GreenLightCurtainDisplay	The color of the front panel indicator light when the instrument is not running a method.
Gripper	Gripper.
LightCurtain	Color of light curtain during a method run.
Pod	Instrument pod.
StackerCarouselSupport	Brackets that support a Stacker Carousel.
Tips	Tips.
TLRods	Tip Loader locking rods.
ViolatedLightCurtain	Color of light curtain after a violation has occurred.

#### Table 4-2. Materials

# Preparing and Managing the Deck

# 5.1 Overview

The **Deck Editor** is used to define and change the deck configurations stored in the current instrument file. A deck is a software representation of the Biomek instrument deck (Figure 5-1) and can be stored and used for multiple methods; however, the software deck must always match the physical deck of the instrument used in the method.

A deck is created and modified in the **Deck Editor**, configured for the method in the **Instrument Setup** step, and used extensively by the software. A deck also stores framing information and device associations for each position on the deck.

The Deck Editor can be used to:

- Create new decks.
- Delete unused decks.
- Define the types and locations of ALPs, labware positioners, and devices positioned on a Biomek instrument deck.
- Control device associations.
- Frame (teach) the physical positions of ALPs, labware positioners, and devices.

The sections in this chapter include:

- <u>Understanding the Deck Editor</u> (Section 5.2)
- <u>Opening, Selecting, Creating, Deleting, or Renaming a Deck</u> (Section 5.3)
- Modifying a Deck (Section 5.4)

The Deck Editor displays decks to be used during the method-building process. A deck is set as the default deck in the Deck Editor. The default deck appears when the Deck Editor is opened and is selected automatically when the Instrument Setup step is inserted into a method. The Instrument Setup step allows selection of any previously defined deck (Figure 5-1).



Figure 5-1. Deck relationship in Biomek 3000

#### 5.1.1 Framing the Deck

Framing the deck allows the Biomek Software to remember each deck position for each Biomek pod, and the Deck Editor is used to set the properties of each ALP and position. Several properties, such as Position Span and Labware Offset, are preprogrammed into Biomek Software; however, other properties, such as Position, must be set by teaching.

The position offsets are taught using the framing tools with the Biomek instrument. However, different framing tools are used according to the type of instrument, type of pod, or the specific deck position. Refer to the appropriate hardware user's manual for information on framing the specific Biomek instrument.

# 5.2 Understanding the Deck Editor

The Deck Editor allows for creation and modification of decks. While the instructions for opening, creating, renaming, and deleting decks are the same, decks for the specific Biomek instrument have different positions since decks in the Deck Editor are based on the physical decks of the instruments.

To access the Deck Editor:

Select Instrument>Deck Editor. The Deck Editor appears (Figure 5-2).



Figure 5-2. Deck Editor showing Deck1 for a Biomek FX

The Deck Editor has several options on the toolbar used to modify decks (Table 5-1).

Options	Description
New Deck	Creates a new deck based upon the current deck.
Delete Deck	Permanently removes the current deck.
Rename Deck	Changes the current deck name.
Open Deck	Opens an existing deck and sets it as the current deck.
Clear Deck	Removes all ALPs and devices from the deck.
Renumber	Renames all of the deck items based on position starting in the upper left corner.
Delete ALP	Removes an ALP or device from the deck.
Properties	Opens the properties for the selected deck position or ALP.
Save	Saves the deck and closes the Deck Editor.
Cancel	Cancels the changes made to the opened deck and closes the Deck Editor.

#### Table 5-1. Deck Editor Options

#### 5.2.1 Viewing the ALP Types List

The ALP Types List (Figure 5-2) shows all standard ALPs, labware positioners, and devices available to place on the deck. Only the ALPs, labware positioners, and devices that can be placed on the deck for the current instrument type are displayed. Use the filter to list only specific types of ALPs.

CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height and failure to choose the correct ALP in the Deck Editor may result in collisions between pod(s) and ALPs during operation.



Figure 5-3. Deck Editor for a Biomek FX

**Note:** Refer to Section 5.4.2, <u>Adding ALPs and Deck Positions</u>, for more information on adding ALPs to a deck. Refer to the <u>ALPs User's Manual</u> for more information about specific ALPs.

#### 5.2.2 Biomek FX Deck

Each initial Biomek FX deck (Figure 5-2) contains 1 Tip Loader ALP (TipLoader) and 19 deck positions (areas where ALPs may be placed). The 19 deck positions are made of five 1 x 3 Passive ALPs and four 1 x 1 Passive ALPs.

- Deck1 initially has 1 Tip Loader ALP position and 19 deck positions; no tip loader device is associated with the tip loader position; can be modified to include other ALPs and other external hardware devices (Figure 5-2).
- Standard has 1 Tip Loader ALP position and 19 deck positions; no tip loader device is associated with the tip loader position; cannot be modified; can be copied to create new decks.

#### 5.2.3 Biomek 3000 Deck

Each initial Biomek 3000 deck (Figure 5-4) contains 1 tool rack position (ToolRack) and 7 deck positions (areas where tool racks and devices may be placed).

- Deck1 initially has 1 tool rack position and 7 deck positions; can be modified to include other tool racks and devices (Figure 5-4).
- Standard has 1 tool rack position and 7 deck positions; cannot be modified; can be copied to create new decks



Figure 5-4. Deck Editor for a Biomek 3000

#### 5.2.4 Biomek NX-MC Deck

Each initial Biomek NX-MC deck (Figure 5-5) contains one Tip Loader ALP (TipLoader) position and 12 labware positions. One 4 by 3 High Density ALP position provides the 12 labware positions. Two more ALPs may be placed in the back row to allow a total of 15 positions.

- Deck1 initially has one Tip Loader ALP position and 12 labware positions; no tip loader device is associated with the tip loader position; can be modified to include other ALPs and other external hardware devices (Figure 5-5).
- Standard one Tip Loader ALP position and 12 labware positions; no tip loader device is associated with the tip loader position; cannot be modified; can be copied to create new decks.



Figure 5-5. Deck Editor for a Biomek NX-MC

#### 5.2.5 Biomek NX-S8 Deck

Each initial Biomek NX-S8 deck (Figure 5-6) contains one Span-8 Disposal ALP (Span8TipTrash) position, one Span-8 Tip Wash position (Span8WashLeft), and 12 labware positions. One 4 by 3 High Density ALP position provides the 12 labware positions. One more ALP may be placed in the back row to allow a total of 15 positions.

- Deck1 initially has one Span-8 Disposal ALP position, one Span-8 Tip Wash position, and 12 labware positions; can be modified to include other ALPs and other external hardware devices (Figure 5-6).
- Standard has one Span-8 Disposal ALP position, one Span-8 Tip Wash position, and 12 labware positions; cannot be modified; can be copied to create new decks.



Figure 5-6. Deck Editor for a Biomek NX-S8

#### 5.3 **Opening, Selecting, Creating, Deleting, or Renaming a Deck**

The Deck Editor is used to open, select, create, delete, or rename a deck in the Biomek Software.

#### 5.3.1 **Opening and Selecting a Deck**

A deck is selected in **Deck Editor** as the default deck. Each time an **Instrument** Setup step is inserted into a method, the default deck appears.

To select a deck:



1. In the Deck Editor, choose **Open Deck**. Select a Deck appears (Figure 5-7).





Figure 5-7. Previews deck selected

- 2. Select the desired deck from the Deck List.
- 3. Choose **OK**. The deck appears as the current deck.
- 4. Choose Save to save the deck and exit Deck Editor.

#### 5.3.2 Creating a New Deck

When a new deck is created, it is initially a copy of the currently open deck. A new deck may be created for many reasons, including:

- To maintain the previous deck configuration when switching between different deck layouts.
- When similar deck configurations exist, but with differences between one or two positions.
- When devices replace deck positions.

To create a new deck:

1. In the Deck Editor, open the deck to copy as a basis for the new deck (refer to Section 5.3.1, *Opening and Selecting a Deck*).

2. Select **New Deck**. Choose a name for this deck: appears (Figure 5-8).

D <u>N</u>ew Deck

Choose a name for this deck:	×
Deck2	
OK Cancel	

Figure 5-8. Enter a new name for the deck

- 3. Enter a name for the new deck.
- 4. Choose **OK** to save the deck name and return to the **Deck Editor**. A copy of the selected deck appears, using the new name and is ready for modification.
- 5. If desired, make modifications to the new deck.
- 6. Choose Save to save the new deck and exit Deck Editor.

**Note:** Choose **Cancel** to cancel the modifications made to the deck. A **Confirm** appears to confirm cancellation of the changes (Figure 5-9).

Confirm	×
?	This will discard all of your changes. Are you sure that you want to do this?
	<u>Yes</u> <u>N</u> o

Figure 5-9. Confirmation to cancel changes

#### 5.3.3 Deleting a Deck

If a deck is no longer needed, it may be deleted.

**Note:** Once a deck is deleted in the Deck Editor, it cannot be retrieved via the Deck Editor. It can be retrieved, however, using the Import/Export Utility (refer to Chapter 3, <u>Using Instrument Files and Settings</u>).

To delete a deck:

1. Open the deck to delete (refer to Section 5.3.1, *Opening and Selecting a Deck*).



2. Select **Delete Deck**. A Warning appears (Figure 5-10).

₩arning			×
$\triangle$	If you delete a decł Are you sure about	< you can't get it this?	back again.
	Yes	<u>N</u> o	

Figure 5-10. Delete Deck confirmation

- 3. Choose **Yes** to delete the deck. The deck is deleted.
- 4. Choose Save to exit the Deck Editor.

Note: Choose Cancel to cancel the deletion of the deck.

#### 5.3.4 Renaming a Deck

Rename a deck to easily identify it.

To rename a deck:

1. Open the deck to rename (refer to Section 5.3.1, *Opening and Selecting a Deck*).



Choose a name for t	his deck:			×
Deck2				
0	ĸ	Cancel	]	

Figure 5-11. Change deck name

3. Enter the desired deck name.

**Note:** Names for deck positions must be alphanumeric with no spaces; the only non-alphanumeric character allowed is "\_" (underscore). The first character must be a letter.

4. Choose **OK** to save the deck under the new name.

**Note:** An Error similar to Figure 5-12 appears if the new name already exists. Choose another name.

Error	×
8	A deck named Deck2 already exists.
	ОК

Figure 5-12. Error in renaming

5. Choose Save to exit the Deck Editor.

Note: Choose Cancel to cancel the renaming of the deck.

# 5.4 Modifying a Deck

3000 — Labware positioners and devices on a Biomek 3000 are referred to as ALPs in Deck Editor.

Modifying a deck includes:

- Opening the deck to modify or create a new deck (refer to Section 5.3.2, *Creating a New Deck*).
- Deleting ALPs and deck positions from the open deck as necessary.
- Adding ALPs and deck positions as necessary.
- Setting ALP and deck position properties.

#### 5.4.1 Deleting ALPs and Deck Positions from a Deck

When a deck is copied or a new deck is created, it is necessary to remove ALPs no longer needed. Remove these ALPs so that other ALPs can be added.

To remove ALPs:

1. Click on the ALP to delete in the Deck View (Figure 5-2).

**Note:** Notice that the ALP is highlighted with a pink border, and deck positions are highlighted with a yellow border. If a single position within a multiposition ALP is selected, it is highlighted pink.

2. Select **Delete ALP** from the toolbar.

OR

Press the **Delete** key on the keyboard.

OR

**Right-click** on the deck position and select **Delete** from the menu that appears. A Warning prompt appears (Figure 5-13).

₩arning	×
	Are you sure you want to delete this?
ĺ.	Yes <u>N</u> o

Figure 5-13. Confirmation to delete ALP

3. Choose **Yes** to delete the ALP. The modified deck appears in Deck View (Figure 5-2).

#### 5.4.2 Adding ALPs and Deck Positions

New ALPs can be added to the deck in the appropriate locations. Deck positions are part of ALPs and are named automatically when added to a deck.

To add ALPs to the deck:

1. Click and hold the mouse button on the desired ALP in the ALP Types List (Figure 5-14). Notice that the locations capable of supporting the ALP are indicated by dashed boxes.



Figure 5-14. Possible OneByThree deck positions highlighted on a Biomek FX deck

2. **Drag and drop the ALP** from the ALP Types list to the desired location. The ALP appears and deck positions are named automatically.

**Note:** The positions may be renamed (refer to Section 5.4.3, <u>Setting ALP</u> <u>Properties and Deck Positions</u>).

**Note:** If the desired ALP is about to be placed where another ALP is currently placed on the deck, the following warning appears (Figure 5-15). Delete the currently-placed ALP before placing the desired ALP on the deck (refer to Section 5.4.1, *Deleting ALPs and Deck Positions from a Deck*).



Figure 5-15. Warning appears because an ALP is about to be placed where it will overlap another ALP

#### 5.4.3 Setting ALP Properties and Deck Positions

After a new ALP is placed on the deck, set the properties of the ALP and related deck positions (Table 5-2).

The following properties may be configured for ALPs or deck positions:

Table 5-2. ALP and Position Properties.

Property	Description	Applies To
Name	The name of the deck position as shown in the Instrument Setup step. ALPs have related names that do not appear in the Instrument Setup.	Position and ALP
Pod1 and Pod2 Coordinates	The coordinate location of the back left corner of the position on the instrument platform for each pod (Figure 5-16).	Position and ALP
Device	Instrument, such as a Stacker Carousel or Tip Loader, associated with a deck position (refer to Section 5.4.3.1, <u>Setting ALP Properties</u> ). Displayed only when More>> is selected.	Position
Device Index	Location of labware in the device associated with a deck position (refer to Section 5.4.3.1, <u>Setting ALP Properties</u> ). Displayed only when <b>More&gt;&gt;</b> is selected.	Position
Sensor Device	Associates the position with the Source/Waste Sensor which is used with an off deck source or waste container.	Position
Labware Offset	The coordinate difference from the Position to the back left corner of the labware (Figure 5-16). Refer to Section 5.4.3.1, <u>Setting ALP</u> <u>Properties</u> . Displayed only when More>> is selected.	Position
Position Span	Area of the Biomek platform the ALP or position covers. (The span for an ALP may be different from the span of a position.) Displayed only when <b>More&gt;&gt;</b> is selected.	Position and ALP
Min Safe Height	Reserved height above position needed to avoid collisions. Displayed Position only when More>> is selected.	
Per-Labware Offsets	Coordinate difference between the set Labware Offset to where specific types of labware sit on the deck position (Figure 5-16). Useful for tip loading and when using the SPE Manifold (refer to Section 5.4.3.1, <u>Setting ALP Properties</u> ). Displayed only when More>> is selected.Position	

**Note:** Not all properties listed in Table 5-2 may be present. Properties vary according to the instrument and configuration.



#### 5.4.3.1 Setting ALP Properties

To set properties of the ALP:

1. From the Deck View, select the ALP to modify.

**Note:** The ALP is selected when the outer edges of the ALP are highlighted pink, and the associated deck positions are highlighted in yellow within the Deck View. Deck positions with green highlights are associated with devices.

2. Choose Properties.

OR

Right-click on the ALP in the Deck Editor or any of its positions and select ALP **Properties**.

OR

Double-click on the ALP on the deck layout. ALP Properties appears (Figure 5-17).

ALP Propertie	es						
<u>N</u> ame OneB	yThree1			ALP Type:	OneBy1	Three	
		X (cm)	Y (cm)	Z (cm)	Pre	ecision	
Pod <u>1</u> Co	ordinates	-4.085	7.072	-15.7	N/A		
Pod <u>2</u> Co	ordinates	-4.085	7.072	-15.7	N/A		
Pod © Pod1 © Pod2	<u>A</u> dvance Manua <u>[</u>	ed MC Teach	<u>I</u> each A <u>u</u> to Tea	ch		More > <u>&gt;</u>	
			OK				

Figure 5-17. ALP Properties of a OneByThree ALP

**Note:** Not all properties listed on Figure 5-17 may be present. Properties vary according to the instrument and configuration.

- 3. Change the Name of the ALP, if desired.
- 4. Choose **OK** to save the ALP properties. Current Deck appears.
- 5. Choose Save to close the Deck Editor

#### 5.4.3.2 Setting Deck Position Properties

Deck position properties are similar to ALP properties; however, deck positions include Labware Offsets that place the labware in the appropriate place for pipetting operations. Because some deck positions are associated with devices, such as a Stacker Carousel, the Device property must be set for these deck positions.

To set deck position properties:

1. Select the position to modify.



#### 2. Select Properties.

OR

Double-click on the deck position.

OR

Right-click on the desired position and select **Properties** from the menu that appears. Position Properties appears (Figure 5-18).

Position Properties
Name P1 ALP Type: FourByThreeHD
X (cm)         Y (cm)         Z (cm)         Precision           Pod1 Coordinates         6.645         13.436         -15.935         Not Framed
Advanced MC Ieach Sectors
Device #none# Device Index Device Control
Sensor Device #none#
X (cm)     Y (cm)     Z (cm)       Labware Offset     0     0     Per-labware Offsets       Position Span     12.812     8.58     Min Safe Height     1
OK Cancel

Figure 5-18. Position Properties for Deck Positions

**Note:** Not all properties listed on Figure 5-18 may be present. Properties vary according to the instrument and configuration.

Note: The deck position is highlighted with a pink line inside the ALP.

3. Rename the deck position, if desired.

**Note:** Names for deck positions must be alphanumeric with no spaces; the only non-alphanumeric character allowed is "\_" (underscore). The first character must be a letter.

- 4. Enter the **X**, **Y**, and **Z Coordinates** of the appropriate pod by framing the ALP:
  - FX refer to the Biomek® FX Laboratory Automation Workstation User's Manual, Chapter 5, <u>Framing the Biomek® FX</u>.
  - 3000 refer to the Biomek® 3000 Laboratory Automation Workstation User's Manual, Chapter 13, <u>Framing the Biomek® 3000</u>.
  - NX-MC refer to the Biomek® NX Multichannel Laboratory Automation Workstation User's Manual, Chapter 4, <u>Framing Instructions</u>.
  - NX-S8 refer to the Biomek Software User's Manual, Chapter 5, <u>Preparing and Managing the Deck</u>.
- 5. Select **More>>**, if necessary.
- 6. Choose the **Device** name to associate with the position, if required.

**Note:** For devices controlled through SILAS, the SILAS module for the device must be installed prior to associating the device with a deck position.

7. If appropriate, select the **Device Index**.

**Note:** The Device Index works with software modules operating instruments that have more than one position, such as the MicroMix 5 Shaker.

- 8. If appropriate, choose the desired **Sensor Device** to associate the Waste/Sensor with the position.
- Choose Per-labware Offsets to customize the offsets for specific types of labware (refer to Section 5.4.3.3, <u>Changing Per-Labware Offsets</u>).
- 10. Set the Min Safe Height.

**Note:** The Labware Offsets and Position Span are predefined in the software. Do not modify these properties.

- 11. Choose **OK** to save the deck position properties. Current Deck appears.
- 12. Choose **Save** to close the Deck Editor.

#### 5.4.3.3 Changing Per-Labware Offsets

Modify the per-labware offsets when specialized needs arise. The X, Y, and Z coordinates for the deck position may be modified based upon the labware type. For example, the Z offsets can be changed for a plasmid filter when performing a filtration-based DNA plasmid purification.

To modify the per-labware offsets:

- 1. Select Instrument>Deck Editor.
- 2. Open the desired deck (refer to Section 5.3.1, *Opening and Selecting a Deck*).
- 3. Select the position to modify.
- 4. Select **Properties**

OR

Double-click on the deck position

OR

Right-click on the desired position and select **Properties**. Position Properties appears (Figure 5-18).

- 5. Choose More>>, if necessary.
- 6. Choose Per-labware Offsets.
- 7. In Per-Labware Offsets, select the labware to modify (Figure 5-19).

Per-Labware Offsets	
Per-Labware Offsets AP384 30uL AP96_200uL AP96_200uL_LLS AP96_20uL_LLS AP96_20uL_LLS BCDeep96Round BCDeep96Round BCDeep96Square BCFlat96 BCSPECollar BCT uberack_10mm BCT uberack_12mm BCT uberack_13mm BCUpsideDownTipBoxLid CostarCone96Round CostarDeep96Square	X 0 cm Y 0 cm Z 0 cm
CostarCone96Round CostarDeep96Square CostarFlat384Square FilterHolder GreinerFlat384Square GreinerShallow384Round LJLShallow384Round NuncFlat384Square PlasmidFilter	
Reservoir	

Figure 5-19. Per-Labware Offsets

**Note:** Per-Labware Offsets includes labware that is specific to the instrument.

8. Enter the desired **X**, **Y**, and **Z** offsets.

**Note:** While the offset values may be calculated using calipers, a more precise measurement should be obtained from the manufacturer's drawings.

- 9. Choose OK to save the offset values. Position Properties appears (Figure 5-18).
- 10. Choose **OK**.
- 11. Choose **Save** to close **Deck Editor**.

# **6** Understanding and Using Project Files

# 6.1 Overview

A project file stores information about liquid types; labware and tip types; well patterns; and pipetting templates and techniques as revisions that are used by a method file to configure the actions of the instrument. Project files store a history of all changes, additions, and deletions of items from the project file.

When a project file is checked in, all items become revisions that are permanently saved. If an item is deleted, that item is no longer available in method building; however, the revision remains in the project file and is used for older validated methods that utilize that revision. If an item is modified, the new definition is stored as a separate revision. Older revisions remain as part of the project file and are used by older validated methods that utilize that record. However, only the newest revision, or working revision, is available in method building. In this way, a validated method can always be run with the exact same revisions, even if those revisions have since been modified or deleted (refer to Section 12.11, *Validating a Method*).

**Note:** If Beckman Coulter Accounts & Permissions is used with Biomek Software, project files also maintain a history of who makes changes to any revisions in the project file.

The active project file and instrument file are displayed in the status bar of the Biomek method editor (Figure 6-1). Any project file can be used with any instrument file. Validated methods, however, are validated for a specific instrument and are no longer validated when used with a different instrument file, even if the instrument file is equivalent to the one on which the method is validated.

**Note:** Validating a method for a specific instrument is not the same as validating a method before a run to internally test the method for errors before it is run. To validate a method to test for errors before it is run, check **Validate the current method before running it** in Preferences (refer to Section 29.2, <u>*Changing Display Preferences*</u>).



Figure 6-1. Biomek method editor

The **Project** menu contains options to access, create, view contents, and check in projects, as well as provide access to the editors to create, modify, and delete project items.

The sections in this chapter include:

- <u>Creating a New Project File</u> (Section 6.2).
- *Opening a Project File* (Section 6.3).
- <u>Deleting and Restoring Project Files</u> (Section 6.4).
- <u>Renaming a Project File</u> (Section 6.5).
- <u>Creating a Copy of a Project File</u> (Section 6.6).
- <u>Checking In a Project File</u> (Section 6.7).
- <u>Viewing Project Contents</u> (Section 6.8).
- <u>Reverting Projects</u> (Section 6.9).
- Importing and Exporting Project Files (Section 6.10).

#### 6.1.1 Contents of a Project File

A project file contains all:

- Labware Classes definitions for microplates, reservoirs, tip boxes, and other labware used in a method, as specified in the Labware Type Editor (refer to Section 7.3, <u>Defining Labware Types</u>).
- Labware Patterns definitions of patterns of wells to aspirate or dispense from, as specified in the Well Pattern Editor (refer to Chapter 11, <u>Creating</u> <u>Well Patterns</u>).
- Liquid Types definitions for reagents used in a method, as specified in the Liquid Type Editor (refer to Chapter 8, <u>Understanding and Creating</u> <u>Liquid Types</u>).
- Pipetting Templates templates of how pipetting operations are performed, as specified in the Pipetting Template Editor (refer to Chapter 10, <u>Using the Pipetting Template Editor</u>).
- Techniques configuration of properties and parameters for all pipetting operations, as specified in the Technique Browser and Technique Editor (refer to Chapter 9, <u>Understanding and Creating Techniques</u>).
- Tip Classes definitions for tips used in a method, as specified in the Tip Type Editor (refer to Section 7.2, <u>Defining Tips</u>).

In addition, methods are associated with a project file. To open a method, the project file that was used to create the method must be the active project.

**Note:** If a method is open when a project file is closed to open a new project, a prompt appears to check in, save, or discard the current method.

# 6.2 Creating a New Project File

Any number of project files may exist on any system. This allows separate project files to be created for different applications or different users. Project files customize the instrument performance to the needs of users. For example, users could create their own project files. In this way, users can guarantee that the definitions used are not changed without their knowledge, and keep labware and liquid type catalogs limited to only the ones they use.

**Create Project** is used to create a new, blank project file. A new project file has no existing revisions for labware types, liquid types, tip types, pipetting templates or techniques, or well patterns.

**Note:** A default project file is automatically created during installation of Biomek Software. The contents of the default project file depend on the options selected during the installation wizard.

To create a new blank project file:

1. From the Project menu, choose **New Project**. Create Project appears (Figure 6-2).

Create Project	X
Enter new project name:	
New Project	
UK Lancel	

Figure 6-2. Create new project

2. Enter a name for the new project file.

Note: The project file name must be different from all other project files.

3. Choose **OK** to create a new project with the specified name. Create Project closes and the new project becomes the active project file.

OR

Choose **Cancel** to close **Create Project** without creating a new project file.

# 6.3 Opening a Project File

A project file can be opened at any time. Opening a project file closes the current active project and any methods that may be open, and opens the selected project.

To open a project file:

1. From the Project menu, choose **Open Project**. Open Project appears (Figure 6-3).

Open Project	
My Projects Recycled	Select a project to open: Biomek/FX Biomek/NX Biomek/NX MC NX Project04 MultiChan Project05
	Project Name: Project05 OK
	Cancel

Figure 6-3. Opening a project file

- 2. Select the desired project file to open.
- 3. Choose **OK** to open the selected (highlighted) project file. Open Project closes and the selected project file becomes the active project.

OR

Choose **Cancel** to close **Open Project** without opening a new project file. The current project remains active.

# 6.4 Deleting and Restoring Project Files

Projects may be deleted, but are never permanently removed from the system, and may be restored at any time.

**Note:** When multiple instances of the Biomek Software are running simultaneously, such as when controlling multiple Biomek instruments, project files are locked. Locked project files may be opened, but not deleted, renamed, or saved.

#### 6.4.1 Deleting Projects

To delete a project:

1. From the Project menu, choose **Open Project**. Open Project appears (Figure 6-4).

Open Project						
My Projects	Select a proj BiomekFX Project05	ect to open: BiomekNX	BiomekNX MC	NX MultiChan	Project04	
	Project Name:	Project05				ОК
						Cancel

Figure 6-4. Open Project

2. Right-click the desired project to delete and choose **Delete**. A confirmation appears (Figure 6-5).



Figure 6-5. Confirm deletion of selected project

 Choose Yes to confirm deletion of the selected project. The project is removed from My Projects and placed in Recycled.

OR

Choose **No** to keep the selected project.

#### 6.4.2 Restoring Projects

To restore a deleted project:

- 1. From the Project menu, choose **Open Project**. Open Project appears (Figure 6-4).
- 2. In the left panel of Open Method, choose **Recycled** to view all the projects that have previously been deleted.

Open Project		
<b>7A</b> -	Select a project to open:	
My Projects	Project05 Project04	
ŝ	······	
Recycled		
	I	
	Project Name: Project05	OK
		Cancel

Figure 6-6. Recycled projects

3. Right-click the desired project to restore and choose **Restore**. The project is moved from **Recycled** to **My Projects**.

**Note:** A project may be opened directly from the **Recycled** folder by selecting the project and choosing **OK**. A confirmation appears stating that the project will be restored prior to opening the project. Choose **OK** to restore and open the project.

# 6.5 Renaming a Project File

A project file may be renamed at any time, regardless of whether it is open or closed. Renaming a project file does not create a copy of the file with the new name.

**Note:** To save a copy of a project file with a new name, use the Save Project As command (refer to Section 6.6, <u>Creating a Copy of a Project File</u>).

**Note:** When multiple instances of the Biomek Software are running simultaneously, such as when controlling multiple Biomek instruments, project files are locked. Locked project files may be opened, but not deleted, renamed, or saved.

#### 6.5.1 Renaming the Active Project File

To rename the project file currently open in the Biomek Software:

1. From the Project menu, choose **Rename Project**. Rename Project appears (Figure 6-7).

Rename Project		X
Enter project name:		
Project05		
ОК	Cancel	

Figure 6-7. Renaming the open project

- 2. Enter a new name for the project file.
- 3. Choose **OK** to rename the project file.

OR

Choose **Cancel** to retain the original name.

#### 6.5.2 Renaming a Saved Project File

To rename a saved project file:

1. From the Project menu, choose **Open Project**. Open Project appears (Figure 6.8).

Open Project						
	Select a project to open:					
My Projects						
	BiomekFX	BiomekNX	BiomekNX MC	NX MultiChan	Project04	
Recycled						
	Project05					
	Project Name:	roject05				ОК
						Cancel

Figure 6-8. Open Project

- 2. Right-click the desired project to rename and choose **Rename**. The project file name is highlighted.
- 3. Enter a new name, and press Enter.
- 4. Choose **OK** to close **Open Project**, save the project file with the new name, and open the new project file.

OR

Choose **Cancel** to close **Open Project** and save the project with the new name only.

# 6.6 Creating a Copy of a Project File

A project file can be copied to create a new project file based on the original using the Save Project As command. A new project created using Save Project As maintains all the records from the original project in a new project file.

**Note:** Using Save Project As copies only the working revision of each project item and does not include prior revisions of project items or any method files in the project. Methods must be imported separately using the import and export options in the File menu. Refer to Section 12.16, *Importing and Exporting Methods*, Section 12.16.1, *Exporting a Method*, and Section 12.16.2, *Exporting All Methods Associated With a Project File*, for more information.

To create a copy of a project file:

- If the project file to copy is not the active file, open the desired project file (refer to Section 6.3, <u>Opening a Project File</u>).
- 2. From the Project menu, choose **Save Project As**. Save Project As appears (Figure 6-9).

Save Project As	×
Enter new project name:	
New Project	
OK Cancel	

Figure 6-9. Saving project as to create a copy

3. Enter a name for the new project file.

**Note:** The project file name must be different from all other project files.

4. Choose **OK** to create a new project with the specified name. Save Project As closes and the new project becomes the active project file.

OR

Choose Cancel to close Save Project As without creating a new project file.
# 6.7 Checking In a Project File

Changes made to any item in a project file are saved but do not become revisions until checked in. Once checked in, a new revision is created for the project items. Revisions can never be lost. If a project item is later modified or deleted, any revision of the project item may be restored.

**Note:** If Beckman Coulter Accounts & Permissions is enabled, methods may be validated. A validated method is a revision of a method that is checked in, approved with a reason or electronic signature, and protected from further modification (refer to Section 12.15, <u>Checking Out a Method</u>). Before a method can be validated, all items used in the method must be checked in as revisions in the project file.

To check in a project file to create new revisions:

 From the Project menu, choose Check In Project. Check In Project displays a list of all project items that differ from the working revision in the project (Figure 6-10).

Check In Project						
Select the project items to	Select the project items to check in:					
Project Item	Change	Last Check In	Check In Time			
🖃 🗹 🤌 Labware Classes						
- 🗹 AB384WellReaction	🖸 Modified	Berkeley Rattan	2/3/2004 1:29:06 PM			
- 🗹 BCDeep96Square	🖸 Modified	Berkeley Rattan	2/4/2004 1:49:16 PM			
- 🗹 CopyOf_AB384Wel	🛨 New		2/4/2004 1:49:17 PM			
– 🗹 Greiner384Lid	🖸 Modified		1/21/2004 5:40:14 PM			
🖵 🔽 Greiner384	🛨 New		2/4/2004 1:04:46 PM			
	ОК	Cancel				

Figure 6-10. Checking in project items as official records

2. Select all project items to check in as revisions in the project file by selecting the checkbox to the left of each project item. By default, all project items are selected.

**Note:** Project items are listed in a tree structure that lists items by type. Branches may be expanded or collapsed using the + and - buttons to the left.

**Note:** Right-click on a type of project item and choose **Select All** from the menu to select all project items of that type, or choose **Select None** to deselect all project items of that type. Right-click on whitespace and choose **Select All** or **Select None** to select or deselect all project items for all types.

3. Choose OK.

**Note:** If Beckman Coulter Accounts & Permissions is not in use, check in is complete. All selected project items are checked in as revisions in the project file.

OR

Choose **Cancel** to close **Check In Project** without checking any project items in as revisions.

4. If Accounts & Permissions is enabled, Check-In appears (Figure 6-11). Enter a **Reason** for checking in the project.

Check	-In					
Userl	Name:	BerkeleyR				
Checl	king In:	Project (Biomek FX Study 14)				
Reas	on:	Added new labware types.				
		Password				
	Only appears when Accounts & Permissions is configured to require password checks (refer to Section 2.2, <u>Installing and</u> <u>Setting the Level of Support For Accounts &amp; Permissions</u> ).					
Password:						
		OK Cancel				

Figure 6-11. Check-In (Accounts & Permissions enabled)

- 5. If Accounts & Permissions is configured to require password checks for signing and check-in, enter the user **Password**.
- 6. Choose **OK** to check in all selected project items as revisions in the project file and close **Check In**.

OR

Choose Cancel to close Check-In without checking in the project.

# 6.8 Viewing Project Contents

The contents of a project file and status of each project item can be viewed at any time.

To view the contents of a project file:

- 1. From the Project menu, choose **Project Contents**. Project Contents displays a list of all project items and their status (Figure 6-12). There are six possible states that may be listed for the status:
  - Current the project item matches the working revision.
  - New there are no existing revisions of the project item that have been checked in.
  - Removed the project item has been deleted and is not available for method editing.

**Note:** To view Removed items, right-click in Project Contents and select **Show Deleted Items**. All project items that have been deleted are listed in Project Contents with a status of Removed.

- Modified the project item has changed since it was last checked in.
- Renamed the project item has been renamed since it was last checked in as a revision, but is otherwise unchanged.
- Reverted the project item has been reverted to a previous revision.

Project Contents						
The following project items have changed:						
Project Item	Change	Last Check In	Check In Time	$\square$		
🕞 🤔 Labware Classes						
AB384WellReactionPlate	Removed	Berkeley Rattan	10/13/2003 1:32:54 PM			
- AP384_30uL	Current	Berkeley Rattan	10/13/2003 1:32:54 PM			
- AP96_200uL	Current	Berkeley Rattan	10/13/2003 1:32:55 PM			
AP96_200uL_Barrier	Removed	Berkeley Rattan	10/13/2003 1:32:55 PM			
AP96_200uL_LLS	Current	Berkeley Rattan	10/13/2003 1:32:56 PM			
- AP96_20uL	Current	Berkeley Rattan	10/13/2003 1:32:56 PM			
AP96_20uL_Barrier	Removed	Berkeley Rattan	10/13/2003 1:32:56 PM			
AP96_20uL_LLS	Current	Berkeley Rattan	10/13/2003 1:32:56 PM			
- BCDeep96Round	N Renamed	Berkeley Rattan	10/13/2003 1:32:57 PM			
BCDeep96Square	N Renamed	Berkeley Rattan	10/13/2003 1:32:57 PM			
BCFlat96	Current	Berkeley Rattan	10/13/2003 1:32:57 PM			
- BCFullReservoir	Modified	Berkeley Rattan	10/13/2003 1:32:57 PM			
- BCSPECollar	Current	Berkeley Rattan	10/13/2003 1:32:57 PM			
BCTuberack_10mm	Current	Berkeley Rattan	10/13/2003 1:32:58 PM			
BCTuberack_12mm	Modified	Berkeley Rattan	10/13/2003 1:32:58 PM			
BCTuberack_13mm	Current	Berkeley Rattan	10/13/2003 1:32:58 PM			
BCTuberack_B3K_10mm	Removed	Berkeley Rattan	10/13/2003 1:32:58 PM			
BCTuberack_B3K_12mm	Removed	Berkeley Rattan	10/13/2003 1:32:59 PM			
BCTuberack_B3K_13mm	🗙 Removed	Berkeley Rattan	10/13/2003 1:32:59 PM	-		
	Revert	ок				

Figure 6-12. Viewing project contents (Accounts & Permissions enabled)

**Note:** When Accounts & Permissions is not enabled, Last Check In does not appear and Check In Time is named Modified Time.

2. Right-click on any project item and choose **History** from the menu to view the history for that record (Figure 6-13). **History** displays the history of changes, who made the changes, the time changes were made, and any comments that may have been entered.

**Note:** Changed By and Comment are blank unless Beckman Coulter Accounts & Permissions is enabled.

History	History					
History c	History of BCFlat96:					
#	Change	Changed By	Time of Change	Comment		
▶ 5	🕑 Checked In		9/15/2003 4:09:35 PM			
4	🕑 Checked In		9/15/2003 4:08:41 PM			
3	🕑 Checked In		9/15/2003 4:08:22 PM			
2	🕑 Checked In		9/15/2003 3:29:58 PM			
1	🕑 Checked In		9/15/2003 2:55:05 PM			
	()					

Figure 6-13. History of a project record

- 3. When finished viewing the history for the record, choose **OK** to close History.
- 4. Repeat steps 2 and 3 to view the history for additional records, if desired.
- 5. Choose **OK** to close Project Contents.

#### 6.8.1 Restoring Deleted Project Items

A deleted project item may be restored at any time and made available again for method building. Restoring a project item makes available the last revision of the project item that was checked in before it was deleted.

To restore a deleted project item:

- 1. From the Project menu, choose **Project Contents**. Project Contents displays a list of all project items and their status.
- 2. Right-click in Project Contents and select **Show Deleted Items**. All project items that have been deleted are listed in Project Contents with a status of Removed.
- 3. Right-click on the desired deleted item to restore and select **Restore** from the menu. The status of the project item changes to **Current** and the project item may now be used in method building.

# 6.9 Reverting Projects

A project file can be reverted back to a previous revision state, if desired.

Note: A revision is created each time a project is checked in.

Options for reverting project files are:

- <u>Reverting Project Contents to a Certain Date and Time</u> (Section 6.9.1).
- <u>Discarding All Changes Since Last Check In</u> (Section 6.9.2).
- <u>Reverting Individual Project Items to a Specific Revision</u> (Section 6.9.3).

# 6.9.1 Reverting Project Contents to a Certain Date and Time

All items in a project file may be reverted to the working revision at a previous date and time. When reverting project contents to a date and time, the latest revision checked in prior to the specified date and time is restored as the active working revision. Revisions checked in after the specified date and time remain in the project file, but are not active.

To revert project contents to a specific date and time:

1. From the Project menu, choose **Project Contents**. Project Contents displays a list of all project items and their status (Figure 6-14).

Project Contents	Project Contents					
The following project items	have changed:					
Project Item	Change	Modified Time				
🕞 🤔 Labware Classes		<u> </u>				
<ul> <li>AB384WellReactionPlate</li> </ul>	🛛 Removed	9/15/2003 1:12:01 PM				
- AP384_30uL	Current	9/15/2003 1:12:01 PM				
- AP96_200uL	Current	9/15/2003 1:12:01 PM				
<ul> <li>AP96_200uL_Barrier</li> </ul>	🛛 Removed	9/15/2003 1:28:36 PM				
<ul> <li>AP96_200uL_LLS</li> </ul>	Current	9/15/2003 1:12:01 PM				
- AP96_20uL	Current	9/15/2003 1:12:01 PM				
<ul> <li>AP96_20uL_Barrier</li> </ul>	🗵 Removed	9/15/2003 1:28:36 PM				
<ul> <li>AP96_20uL_LLS</li> </ul>	Current	9/15/2003 1:12:01 PM				
<ul> <li>Barrier_Tips_200uL</li> </ul>	N Renamed	9/15/2003 1:28:36 PM				
<ul> <li>Barrier_Tips_20uL</li> </ul>	N Renamed	9/15/2003 1:28:36 PM				
<ul> <li>BCDeep96Round</li> </ul>	Current	9/15/2003 1:12:01 PM				
<ul> <li>BCDeep96Square</li> </ul>	Modified	9/15/2003 1:28:35 PM				
<ul> <li>BCFlat96</li> </ul>	Current	9/15/2003 1:12:01 PM				
<ul> <li>BCFullReservoir</li> </ul>	Current	9/15/2003 1:12:01 PM				
<ul> <li>BCNewPlate</li> </ul>	Modified	9/15/2003 1:28:36 PM				
<ul> <li>BCSPECollar</li> </ul>	Current	9/15/2003 1:12:01 PM				
<ul> <li>BCTuberack_10mm</li> </ul>	🗵 Removed	9/15/2003 1:12:01 PM				
<ul> <li>BCTuberack_12mm</li> </ul>	🗵 Removed	9/15/2003 1:12:01 PM				
<ul> <li>BCTuberack_13mm</li> </ul>	🖹 Removed	9/15/2003 1:12:01 PM				
	<u>R</u> evert	ок				

Figure 6-14. Viewing project contents

2. Choose **Revert**. Revert Project appears (Figure 6-15).



Figure 6-15. Reverting a project file to a specific date and time

- 3. Select Revert the project contents back to this date and time.
- 4. On the calendar, browse to and select the desired date.
- 5. Beneath the calendar, enter the desired time.
- 6. Choose **OK**. Revert Project closes and the project contents are reverted to the latest revision checked in prior to the specified date and time.

OR

Choose **Cancel** to close **Revert Project** without reverting the project contents.

#### 6.9.2 Discarding All Changes Since Last Check In

All the changes that have been made to any project contents since the project was last checked in may be discarded. Any changes made to a project item since the last time that project item was checked in are discarded.

To discard changes since last check in:

1. From the Project menu, choose **Project Contents**. Project Contents displays a list of all project items and their status (Figure 6-16).

Project Contents			
The following project items	have changed	:	
Project Item	Change	Modified Time	
🕞 🤔 Labware Classes			-
AB384WellReactionPlate	Removed	9/15/2003 1:12:01 PM	
- AP384_30uL	Current	9/15/2003 1:12:01 PM	
- AP96_200uL	Current	9/15/2003 1:12:01 PM	
AP96_200uL_Barrier	Removed	9/15/2003 1:28:36 PM	
AP96_200uL_LLS	Current	9/15/2003 1:12:01 PM	
- AP96_20uL	Current	9/15/2003 1:12:01 PM	
AP96_20uL_Barrier	Removed	9/15/2003 1:28:36 PM	
AP96_20uL_LLS	Current	9/15/2003 1:12:01 PM	
Barrier_Tips_200uL	N Renamed	9/15/2003 1:28:36 PM	
Barrier_Tips_20uL	N Renamed	9/15/2003 1:28:36 PM	
- BCDeep96Round	Current	9/15/2003 1:12:01 PM	
- BCDeep96Square	Modified	9/15/2003 1:28:35 PM	
- BCFlat96	Current	9/15/2003 1:12:01 PM	
- BCFullReservoir	Current	9/15/2003 1:12:01 PM	
- BCNewPlate	Modified	9/15/2003 1:28:36 PM	
- BCSPECollar	Current	9/15/2003 1:12:01 PM	
BCTuberack_10mm	Removed	9/15/2003 1:12:01 PM	
BCTuberack_12mm	Removed	9/15/2003 1:12:01 PM	
BCTuberack_13mm	🗙 Removed	9/15/2003 1:12:01 PM	-
	<u>R</u> evert	ОК	

Figure 6-16. Viewing project contents

2. Choose **Revert**. Revert Project appears (Figure 6-17).





- 3. Select Discard all changes since the project was last checked in.
- 4. Choose **OK**. Revert Project closes and the all changes made to each project item since that item was last checked in are discarded.

OR

Choose **Cancel** to close Revert Project without discarding changes.

# 6.9.3 Reverting Individual Project Items to a Specific Revision

An individual project item may also be reverted back to a specific revision, if desired.

To restore an individual project item to a specific revision:

1. From the Project menu, choose **Project Contents**. Project Contents displays a list of all project items and their status (Figure 6-18).

P	Project Contents						
1	The following project items have changed:						
Γ	Project Item	Change	Modified Time				
Γ	🕞 🤔 Labware Classes			<b></b>			
L	<ul> <li>AB384WellReactionPlate</li> </ul>	Removed	9/15/2003 1:12:01 PM				
L	- AP384_30uL	Current	9/15/2003 1:12:01 PM				
L	- AP96_200uL	Current	9/15/2003 1:12:01 PM				
L	<ul> <li>AP96_200uL_Barrier</li> </ul>	🛛 Removed	9/15/2003 1:28:36 PM				
L	- AP96_200uL_LLS	Current	9/15/2003 1:12:01 PM				
L	- AP96_20uL	Current	9/15/2003 1:12:01 PM				
L	<ul> <li>AP96_20uL_Barrier</li> </ul>	🛛 Removed	9/15/2003 1:28:36 PM				
L	<ul> <li>AP96_20uL_LLS</li> </ul>	Current	9/15/2003 1:12:01 PM				
L	<ul> <li>Barrier_Tips_200uL</li> </ul>	N Renamed	9/15/2003 1:28:36 PM				
L	<ul> <li>Barrier_Tips_20uL</li> </ul>	N Renamed	9/15/2003 1:28:36 PM				
L	<ul> <li>BCDeep96Round</li> </ul>	Current	9/15/2003 1:12:01 PM				
L	<ul> <li>BCDeep96Square</li> </ul>	Modified	9/15/2003 1:28:35 PM				
L	<ul> <li>BCFlat96</li> </ul>	Current	9/15/2003 1:12:01 PM				
L	<ul> <li>BCFullReservoir</li> </ul>	Current	9/15/2003 1:12:01 PM				
L	<ul> <li>BCNewPlate</li> </ul>	Modified	9/15/2003 1:28:36 PM				
L	<ul> <li>BCSPECollar</li> </ul>	Current	9/15/2003 1:12:01 PM				
L	<ul> <li>BCTuberack_10mm</li> </ul>	🖈 Removed	9/15/2003 1:12:01 PM				
L	<ul> <li>BCTuberack_12mm</li> </ul>	🛛 Removed	9/15/2003 1:12:01 PM				
L	<ul> <li>BCTuberack_13mm</li> </ul>	🛛 Removed	9/15/2003 1:12:01 PM	-			
		<u>R</u> evert	ОК				

Figure 6-18. Viewing project contents

2. Right-click the desired project item to revert. **History** appears and displays a list of all the revisions of the project item that have been checked in. A blue arrow points to the active working revision for the project item (Figure 6-19).

History	History						
History (	of BCFlat96:						
#	Change	Changed By	Time of Change	Comment			
▶ 5	🖌 Checked In		9/15/2003 4:09:35 PM				
4	🕑 Checked In		9/15/2003 4:08:41 PM				
3	🗳 Checked In		9/15/2003 4:08:22 PM				
2	🗳 Checked In		9/15/2003 3:29:58 PM				
1	🗳 Checked In		9/15/2003 2:55:05 PM				
		OK					

Figure 6-19. History of a project item

3. Double-click on the desired revision number of the project item to revert.

OR

Right-click the desired revision number and choose **Revert Changes** from the menu. The blue arrow moves to the revision to use as the working revision.

4. Choose **OK**. History closes and the selected revision is set as the working revision for the project item.

# 6.10 Importing and Exporting Project Files

Project files and individual records can be freely shared among projects or transferred from one computer to another using Import Project and Export Project.

**Note:** When Beckman Coulter Accounts & Permissions is enabled, users with the Run Validated Methods or Develop Methods permission may export methods, but only those with the Develop Methods permission may import methods into a project (refer to Chapter 2, <u>Using Accounts & Permissions</u>).

**Note:** Methods may also be imported and exported. Refer to Section 12.16, *Importing and Exporting Methods*, for more information.

This section covers:

- Exporting a Project File to a Biomek Import File (Section 6.10.1).
- *<u>Importing a Project File from a Biomek Import File</u> (Section 6.10.2).*

#### 6.10.1 Exporting a Project File to a Biomek Import File

A project file or individual records from a project file can be exported to a Biomek import file, which may be imported into a different project file. Exporting a project file always exports the working revisions from the current active project file. The specific items to export from the project file are then selected from all available project items.

To export a project file:

1. From the Project menu, choose **Export Project**. Export Project appears, displaying a list of all records in the project file (Figure 6-20).

Export Project					
Select the project items to export:					
Project Item	Last Check In	Check In Time			
🕞 🗹 🤔 Labware Classes					
AB384WellReactionPla	te	2/3/2004 1:29:06 PM			
- 🗹 AP384_30uL		2/3/2004 1:29:07 PM			
- 🗹 AP96_200uL		2/3/2004 1:29:07 PM			
- 🗹 AP96_200uL_Barrier		2/3/2004 1:29:07 PM			
- 🗹 AP96_20uL		2/3/2004 1:29:07 PM			
- 🖌 AP96_20uL_Barrier		2/3/2004 1:29:07 PM			
🛛 🗕 🗹 BCDeep96Round		1/20/2004 10:55:16 AM			
🛛 🕂 🗹 BCDeep96Square		1/20/2004 10:55:16 AM			
- 🗹 BCFlat96		1/20/2004 10:55:16 AM			
- 🗹 BCFullReservoir		2/3/2004 1:29:07 PM			
- 🗹 BCSPECollar		2/3/2004 1:29:07 PM			
🛛 🕂 🗹 BCUpsideDownTipBoxL	id	1/20/2004 10:55:16 AM			
— 🗹 Collar_36mm		2/3/2004 1:29:07 PM			
— 🗹 CollarSpacer_23mm		2/3/2004 1:29:07 PM			
- 🗹 CostarCone96Round		1/20/2004 10:55:18 AM			
- 🗹 CostarDeep96Square		2/3/2004 1:29:07 PM			
— 🗹 CostarFlat384Square		2/3/2004 1:29:07 PM			
🛛 🕂 🗹 CostarFlat384SquareLi	d	1/20/2004 10:55:18 AM			
🛛 🕂 🗹 DrainableRefillableRese	ervoir	1/20/2004 10:55:18 AM	-		
	ОК	Cancel			

Figure 6-20. Selecting project contents to export

2. Select all project items to include in the Biomek import file by selecting the checkbox to the left of each project item. By default, all project items are selected.

**Note:** Project items are listed in a tree structure that lists items by type. Branches may be expanded or collapsed using the + and - buttons to the left of the project item name.

To select or deselect all project items of a specific type, select the checkbox to the left of folder representing the desired type of project item. Alternately, right-click on a type of project item and choose **Select All** from the menu to select all project items of that type, or choose **Select None** to deselect all project items of that type. Right-click on whitespace and choose **Select All** or **Select None** to select or deselect all project items for all types.

3. Choose **OK** to include all selected project items in the Biomek import file and close **Export Project**. Save As appears (Figure 6-21).

OR

Choose **Cancel** to close **Export Project** without exporting the project to a Biomek import file.

Save As			<u>?</u> ×
Save in: 🗀	Biomek 💌 🖛 🛍 (	•	
Logs			
File name:	BiomekFX.imp	Save	•
Save as type:	Biomek® Software Import Files (*.imp)	Canc	el
			111

Figure 6-21. Save Import File

- 4. In Save As, browse to and select the location to save the Biomek import file.
- 5. In File Name, enter a name for the import file.
- 6. Choose **Save** to create the new import file.

# 6.10.2 Importing a Project File from a Biomek Import File

A Biomek import file previously exported from a project file can be imported into the current active project file. The records to import are selected from all available records in the import file. Biomek Software compares the selected project items in the import file to the current project and only imports those items that are different.

To import a project file from a Biomek import file:

- 1. If necessary, open the project file in which to import the Biomek import file.
- 2. From the Project menu, choose **Import Project**. Open Import File appears (Figure 6-22).

Open Import	File					<u>?</u> ×
Look in: 🔯	Biomek Software		•	۱	r 🗐 🕈	
3.0 Backup Bitmaps Logs Biomek300	OProject.imp roject.imp					
File name: Files of type:	Biomek® Import	Files		•	Oper Canc	

Figure 6-22. Open Import File

3. In Open Import File, browse to and select the Biomek import file to import.

**Note:** Workspace settings from Biomek FX software versions 2.5 or later may also be imported into a project file. To import a Biomek FX 2.5 software workspace file, from Files of type, select **Biomek® FX Workspace Files** to show all workspace files (\*.wld).

4. Choose **Open**. Check In Project appears, displaying a list of all project items in the project file to import (Figure 6-23). Only the project items that do not match any of the revisions in the current project file are listed.

**Note:** Biomek Software compares the project items in the import file to the current project. If the item in the import file matches any revision in the current project, that item is not displayed in Check In Project; if all items from the import file match the current project, Information (Figure 6-24) appears instead of Check In Project.

Check In Project						
Select the project items to	Select the project items to check in:					
Project Item	Change	Last Check In	Check In Time			
🖃 🗹 🤔 Labware Classes						
- 🗹 AB384WellReaction	🖸 Modified	Berkeley Rattan	2/3/2004 1:29:06 PM			
— 🗹 BCDeep96Square	🖸 Modified	Berkeley Rattan	2/4/2004 1:49:16 PM			
– 🗹 CopyOf_AB384Wel	🛨 New		2/4/2004 1:49:17 PM			
— 🗹 Greiner384Lid	🖸 Modified		1/21/2004 5:40:14 PM			
🗌 🖵 🗹 Greiner384	🛨 New		2/4/2004 1:04:46 PM			
	ОК	Cancel				



Informa	tion	×
(j)	The BiomekNX MC project is up-to-date.	
	ОК	

Figure 6-24. Information that appears when no project items are imported

5. Select all project items from the Biomek import file to import by selecting the checkbox to the left of each project item. By default, all project items are selected.

**Note:** Project items are listed in a tree structure that lists items by type. Branches may be expanded or collapsed using the + and - buttons to the left of the project item name.

Right-click on a type of project item and choose **Select All** from the menu to select all project items of that type, or choose **Select None** to deselect all project items of that type.

6. Choose **OK** to import all selected project items from the Biomek import file into the project file and close Check In Project.

**Note:** If Beckman Coulter Accounts & Permissions is not in use, check in is complete.

OR

Choose **Cancel** to close Check In Project without including any project items in the Biomek import file.

7. If Accounts & Permissions is enabled, Check-In appears (Figure 6-25). Enter a **Reason** for checking in the project.

Check-In	
User Name:	BerkeleyR
Checking In:	Project (Biomek FX Study 14)
Reason:	Added new labware types.
Only a require <u>Setting</u>	Password ppears when Accounts & Permissions is configured to password checks (refer to Section 2.2, <u>Installing and</u> the Level of Support For Accounts & Permissions).
Password:	OK Cancel

Figure 6-25. Check-In (Accounts & Permissions enabled)

- 8. If Accounts & Permissions is configured to require password checks for signing and check-in, enter the user **Password**.
- 9. Choose **OK** to close **Check-In**. The project items from the import file are imported to the project file.

# Creating and Modifying Tip and Labware Types

# 7.1 Overview

Biomek Software must know the properties of the tips and labware used with the instrument to correctly perform liquid handling and gripper movement operations. A set of predefined tip and labware types commonly used in automated laboratories is preloaded with the software.

However, there may be instances when a tip or labware type used in the laboratory is not predefined in Biomek Software, or specifications of a predefined tip or labware type change. When this occurs, tip modifications are made in the Tip Type Editor and labware type modifications are made in the Labware Type Editor.

**Note:** Tips must be associated with a labware type in order to be used in a method.

All labware types have a graphic associated with them in Biomek Software. Any new labware type created in the Labware Type Editor will have a corresponding graphic generated, based upon the labware type properties. The graphic then appears in the Labware Category Graphical Display of the Instrument Setup step for use in methods (Figure 7-1).

Tip and labware types, along with information about liquid types; well patterns; and pipetting templates and techniques, are stored as part of a project file. Project files store a history of all changes, additions, and deletions of items from the project file. Refer to Chapter 6, <u>Understanding and Using Project Files</u>, for more information on project files.

**Note:** When Beckman Coulter Accounts & Permissions is enabled, only users with the **Develop Projects** permission assigned may open, edit, create, and delete tip and labware types (refer to Chapter 2, <u>Using Accounts & Permissions</u>).

The sections in this chapter include:

- <u>Defining Tips</u> (Section 7.2)
- <u>Defining Labware Types</u> (Section 7.3)





# 7.2 Defining Tips

All Beckman Coulter tip types may be used with Biomek instruments with some exceptions:

- FX Fixed tips can only be used on the Span-8 Pod. LLS tips should only be used on the Span-8 Pod since a Multichannel Pod cannot sense the liquid level of wells.
- FX Fixed60 and Fixed100 tips predefined in the Tip Type Editor for a Biomek FX project have a maximum capacity that is different from those in a Biomek NX project. Use caution when importing and exporting methods to and from instruments.

**Note:** The Span-8 Pod is capable of accessing all types of labware supported by the Biomek FX instrument; however, specific types of tips are recommended to access specific types of labware (refer to the *Biomek*® *FX Laboratory Automation Workstation User's Manual*, Section 3.3.2.1, *Labware and Tip Compatibility*, with the Span-8 Pod).

3000 — WashTips are only used internally with Biomek Software on the wash tool.

**Note:** The Biomek 3000 head is capable of accessing all types of labware supported by the Biomek 3000 instrument; however, some tools cannot access specific types of labware. For example, multi-channel pipette tools cannot access test tubes (refer to the *Biomek*® 3000 Laboratory Automation Workstation User's Manual, Section 2.3.1, Labware and Tip Compatibility with the Pipette Tools).

- NX-S8 LLS tips are conductive and also used for clot detection which is available only on this instrument.
- NX-S8 SeptaFluted tips are conductive and used for clot detection and for piercing septa which is available only on this instrument.
- FX, NX-S8 Fixed60 and Fixed100 tips predefined in the Tip Type Editor for a Biomek NX project have a maximum capacity that is different from those in a Biomek FX project. Use caution when importing and exporting methods to and from instruments.

**Note:** The pod is capable of accessing all types of labware supported by the Biomek NX instrument; however, specific types of tips are recommended to access specific types of labware (refer to the *Biomek*® *NX Span-8 Laboratory Automation Workstation User's Manual*, Section 2.3.3.1, *Labware and Tip Compatibility with the Span-8 Pod*).

**Note:** Refer to the specific hardware manual for a Biomek instrument for more information on tips and tip properties.

In order to use tips that are not predefined, use the Tip Type Editor to:

- Add new tips.
- Change tip properties, such as capacity and size.
- Remove tips.

## 7.2.1 Accessing the Tip Type Editor

To access the Tip Type Editor, choose **Project>Tip Type Editor**. Tip Types appears (Figure 7-2).

	Tip Types					
	P20 Fixed100	<u>N</u> ame	P20			
	Fixed60 P20 Barrier	<u>C</u> apacity	100	μL		
	P20_LLS	Air Capacit <u>y</u>	120	μL		
	P200_Barrier	<u>H</u> eight	3.825	cm		
defined tin types	P30	Seating Depth	0.75	cm		Tip Properties
The tips that appear	Span8_P20 Span8_P20_Barrier	Conic <u>L</u> ength	2.047	cm		Displays properties
here predefined are	Span8_P200 Span8_P200_Barrier	$Ma\underline{\times}$ Tip Radius	0.152	cm	•	tips.
dependent on the	Pins Span8_P20_LLS_Barrier	<u>M</u> in Tip Radius	0.04	cm		
Biomek instrument	Span8_P200_LLS_Barrie Span8_P1000_LLS	R <u>u</u> nout Radius	0.025	cm		
configuration.	Span8_P1000_LLS_Bar	LLS Tips				
		Fixed Tips	an Table -			
		Piercipa Tips	em Lubing			
		, nording rips				
		Add	Remove			
		ОК	Cancel			
	I					

Figure 7-2. Tip Types

#### 7.2.2 Adding New Tips

Add tips that are not previously defined to ensure that liquid transfers are performed accurately.

To add tips:

1. In the Tip Type Editor, highlight the tip type that most closely resembles the new tip type.

**Note:** Adding tips copies the selected tip type, including its properties.

2. Choose **Add**. The tip type appears in the Tip List as "Tip" followed by a number. For example, the first tips created are named "Tip1".



Figure 7-3. Adding a new tip type



CAUTION: After new tips are added, the correct properties must be defined in the Tip Type Editor according to the manufacturer's specifications. Inaccurate specifications may lead to hardware crashes.

- 3. Modify the tip properties according to the manufacturer's specifications (refer to Section 7.2.3, *Changing Tip Properties*).
- 4. Choose OK.

**Note:** Choose **Cancel** to cancel the addition of the tip type. A Warning appears to confirm the cancellation of changes (Figure 7-4).

Warning	×
	You will lose all the changes that you made in this dialog. Are you sure you want to do this?
	<u>Yes</u> <u>N</u> o



# 7.2.3 Changing Tip Properties

Change tip properties in the **Tip Type Editor** when defining new tips or changing the existing tip properties.

Tip properties include the following (Table 7-1):

Table 7-1. Tip Properties

	Description
Name	Names the tips for easy identification.
	<b>Note:</b> Tip names must begin with a letter, and only letters, numbers, and underscores (_)can be used.
Capacity	The maximum liquid volume the tip holds.
Air Capacity	The total amount of leading air gap, plus liquid that can be aspirated. Trailing air gaps are counted against liquid capacity since they displace liquid further into the tips (Figure 7-5).
Height	The total height of the tip (Figure 7-6).
Seating Depth	The depth at which the tip seats properly on the mandrels (Figure 7-6).
Conic Length	The lowest triangular section of a tip (Figure 7-6).
Max Tip Radius	Radius of the tip at the top of the conic length (Figure 7-6).
Min Tip Radius	Radius of the tip at its narrowest point (Figure 7-6).
Runout Radius	Defined area around a tip to account for manufacturing tolerances and tip bending.
LLS Tips	Indicates the tips are conductive and capable of Liquid Level Sensing and clot detection.
	FX — Liquid Level Sensing is available on a Span-8 Pod only. Clot detection is not available.
	NX-S8 — Liquid Level Sensing and clot detection are available.
Fixed Tips	Indicates the tips are not disposable.
	> <b>FX, NX-S8</b> — Used on Span-8 Pods only.
Use System Tubing	Indicates the sample aspiration is allowed to enter the system tubing.
	> <b>FX, NX-S8</b> — Used on Span-8 Pods only.
	> <b>3000</b> — Used for WashTips.
Piercing Tips	Indicates the tips are capable of piercing septa.
	NX-S8 — Septa piercing is available on this instrument only.



Figure 7-5. Tip capacity and air capacity



Figure 7-6. Tip dimensions

To change tip properties:

- 1. In the Tip Type Editor, select the desired tips from the Tip List (Figure 7-2). The current properties of the tips display in Tip Properties.
- 2. Change the desired properties (Table 7-1) according to the manufacturer's specifications.
- 3. Choose OK.

**Note:** Choose **Cancel** to cancel the changes made to the properties. A Warning appears to confirm the cancellation of changes (Figure 7-4).

### 7.2.4 Removing Tips

Remove obsolete tips from the Tip List.

**Note:** The Tip Type Editor does not allow tips to be deleted that are used by any labware class or that are on the pod. To delete a tip type, delete the labware type in the Labware Type Editor first (refer to Section 7.3.6, *Editing Labware Type Properties*).

To remove tips:

- 1. In the Tip Type Editor, select the desired tips from the Tip List.
- 2. Choose **Remove**. A Warning appears (Figure 7-7).

Warning	X
	This will delete the Tip1 tip type. Are you sure you want to do this?
	Yes <u>N</u> o

Figure 7-7. Remove tips warning

- 3. Choose **Yes** to confirm the removal. The tips are removed from the Tip List.
- 4. Choose OK.

**Note:** Choose **Cancel** to cancel the removal of the tip type. A **Warning** appears to confirm the cancellation of changes (Figure 7-4).

# 7.3 Defining Labware Types

Labware types are divided into categories, such as titerplates, reservoirs, tube racks, lids, and tip boxes. In Biomek Software, each labware category contains specific predefined labware, such as BCDeep96Square, BCUpsideDownTip, BoxLid, and BCTuberack\_10mm.

Edit labware type properties. Use the Labware Type Editor for:

- <u>Adding a New Labware Type</u> (Section 7.3.2).
- <u>Copying a Labware Type</u> (Section 7.3.3)
- <u>Deleting a Labware Type</u> (Section 7.3.4)
- <u>Renaming a Labware Type</u> (Section 7.3.5)
- *<u>Editing Labware Type Properties</u>* (Section 7.3.6)

#### 7.3.1 Accessing the Labware Type Editor

To access the Labware Type Editor, choose **Project>Labware Type Editor**. Labware Types appears (Figure 7-8).



Figure 7-8. Labware Types

**Note:** The labware that is displayed in the Labware Type Editor is dependent on the current project file.

## 7.3.2 Adding a New Labware Type

Add a new labware type when the desired type is not predefined in Biomek Software. When a new labware type is added, default properties are inserted into the appropriate fields. The default properties may need to be changed to meet the manufacturer's specifications of the new labware type.

To add a new labware type:

1. From Labware Types, select **New** from the toolbar at the top. New Labware appears (Figure 7-9).

New Labware	×
Type: Titerplate	
Name: Labware1	
OK Cancel	

Figure 7-9. New Labware

- 2. Choose the labware type from the pull-down list.
- 3. Enter a name for the labware or use the default name.

**Note:** The new labware is assigned the default name of Labware plus a number representing the order in which the labware was created. For example, the first labware created is given the default name Labware1 and the second is given the default name Labware2.

 Choose OK. The new labware appears graphically in Labware Types. (Figure 7-8).

**Note:** Use Filter or scroll down Labware Types to locate the new labware type (Figure 7-8).

#### 7.3.3 Copying a Labware Type

Copying a labware type creates an exact copy of it, including current properties. The properties of the copied labware type can be modified to create a new labware type.

To copy a labware type:

1. From Labware Types, highlight the graphic representing the labware and select **Copy** from the toolbar at the top.

OR

Right-click the graphic for the labware type and choose **Copy**. Copy Labware Type appears (Figure 7-10).

Copy Labware Type		×
CopyOf_BCFlat96		
OK	Cancel	

Figure 7-10. Copy Labware Type

2. Enter a name for the labware or use the default name.

**Note:** Labware names must begin with a letter, and can contain only letters, numbers, and underscores (\_) or an error message appears. The default name assigned to a labware type consists of the name of the labware type preceded by CopyOf\_. For example, a copy of BCFlat96 is CopyOf\_BCFlat96.

3. Choose **OK**. The copied labware appears graphically in Labware Types. (Figure 7-8).

#### 7.3.4 Deleting a Labware Type

Delete any labware type from Labware Types that is no longer used.

To delete a labware type:

1. From Labware Types, highlight the graphic representing the labware and select **Delete** from the toolbar at the top.

OR

Right-click the graphic for the labware type and choose **Delete**. Information appears (Figure 7-11).



Figure 7-11. Information

2. Choose **Yes** to confirm the removal. The labware is deleted from Labware Types (Figure 7-8).

#### 7.3.5 Renaming a Labware Type

Renaming a labware type provides the opportunity to assign a descriptive name to a labware type for easy identification.

**Note:** If labware types are renamed, any methods with labware under the old name will generate an error when the method is validated. The name must be changed to reflect the new name in the **Instrument Setup** step and any step that refers to the old name, such as in the **Transfer** step.

**Note:** The labware type names appear on the **Instrument Setup** Configuration. Refer to Chapter 15.2, *Instrument Setup Step*, for more information about **Instrument** Setup.

To rename a labware type:

1. From Labware Types, highlight the graphic representing the labware and select **Rename** from the toolbar at the top

OR

Right-click the graphic for the labware type and choose **Rename**. Rename Labware Type appears (Figure 7-12).

	×
Cancel	
	Cancel

Figure 7-12. Rename Labware Type

2. Enter a new name.

**Note:** Labware type names must begin with a letter, and only letters, numbers, and underscores (\_)can be used.

3. Choose **OK** to save the new labware name. The new labware name appears in Labware Types (Figure 7-8).

## 7.3.6 Editing Labware Type Properties

The applicable properties for a labware type are arranged in the Labware Type Editor according to the following categories (Table 7-2):

Category	Properties
Basic Information	span, height, and description of the labware; bitmap graphic; and an option to edit the color of the labware (refer to Section 7.3.6.1, <u>Changing</u> <u>Labware Colors</u> )
Magbead	allows properties to be set for labware used on the Magnetic Bead ALP, such as magnet type; height; and options to clamp during pipetting or engaging magnet, move arms to maximum height when clamping; and sense labware before pipetting or gripping (refer to the <i>ALPs User's</i> <i>Manual</i> , Section 10.4.3, <u>Configuring</u> <u>LabwareTypes for Use on the</u> <u>Magnetic Bead ALP</u> ).
Miscellaneous	name of the default lid, sensitivity, and clamp force, an option to hide the labware so it is not visible in the Biomek editor, and an option to require piercing tips when the labware is accessed
	> <b>NX-S8</b> — Septum piercing is available on this instrument only.
Movement Information	gripper offset, gripper squeeze, gripper unsqueeze, speed limit, and an option to use the gripper sensor when available
Orbital Shaker	allows the maximum speed to be set for labware used on the Orbital Shaker ALP (refer to the <i>ALPs User's</i> <i>Manual</i> , Section 14.4.2, <u>Configuring</u> <u>a New Labware Type for Use on the</u> <u>Orbital Shaker ALP</u> ).
Ordering Information	manufacturer and part number
Sections	an option to configure sections used in a reservoir (refer to Section 7.3.6.2, <u>Configuring a Reservoir with</u> <u>Sections</u> )

Table 7-2. Labware Type Properties

Category	Properties
Stacking	stack offsets and options to edit the stack offsets (refer to Section 7.3.6.3, <u>Specifying Stackable Items</u> ) and to edit on which labware types or ALP types the labware can be securely stacked on top (refer to Section 7.3.6.4, <u>Using Secure Stacking</u> )
Reservation Information	"Caption" field that identifies the label that appears on the reservation and a "Reserve For" field that identifies the reason the space is reserved
Wells_1	well offset, count, spacing, maximum volume, format, column 2 offset, default volume, known default volume, and an option to configure well dimensions (refer to Section 7.3.6.5, <u>Configuring Well Properties</u> )
Tips	tip offset, count, spacing, tip load Z offset, tip unload Z offset, and tip type

Table 7-2. Labware Type Properties

To edit labware properties:

1. From Labware Types, highlight the graphic representing the labware and choose **Edit** from the toolbar.

OR

Right-click the graphic and choose **Edit**.

OR

Double-click the graphic.

2. Choose the appropriate labware category and make the desired changes to the labware properties that appear.

Note: All distances are in centimeters unless otherwise specified.

3. Choose Save.

**Note:** Three sections of hints are offered at the bottom of the edit option. The first section displays a graphic of the labware, the second section explains the selected field, and the third section explains any errors resulting from editing the labware properties. Place the cursor over the desired labware properties field and click to access these hints.

#### 7.3.6.1 Changing Labware Colors

Make labware stand out in the Instrument Setup step by changing the colors of the labware as desired. For example, each titerplate could be a different color. Changing labware colors also helps to make labware distinguishable in the Biomek main editor.

**Note:** The color fuchsia (RGB 255, 0, 255) is the system transparency color and causes odd visual behavior.

To change labware colors:

1. From Labware Types, highlight the graphic representing the labware and choose **Edit** from the toolbar at the top.

OR

Right-click the graphic and choose Edit.

OR

Double-click the graphic.

2. Choose Basic Information (Figure 7-13).

GreinerFlat3845quare	
<u>S</u> ave <u>C</u> ancel	
Basic Information	X Y
Magpead Miscellaneous	Span 12.78 8.56 cm
Orbital Shaker	Height 1.44 cm
Ordering Information Stacking	Colors Edit
Wells_1	Bitmap
	Description Greiner microplate with 384 square, flat-bottomed wells
Preview	Hint Errors
	The shape and general description of the There are no errors in this labware type definition

Figure 7-13. Basic Information chosen

3. Choose Edit next to Colors. Labware Colors appears (Figure 7-14).

Labware Colors	X
Wide Edge	(Set )
Narrow Edge	Set
Тор	Set
Well	Set
	*** *** *** ***
ОК	Cancel

Figure 7-14. Labware Colors

- 4. Choose **Set** to configure the color for the following sections of the labware:
  - Wide Edge
  - Narrow Edge
  - Top
  - Well

Color appears (Figure 7-15).

Color ? X
Basic colors:
<u>C</u> ustom colors:
Define Custom Colors >>
OK Cancel

Figure 7-15. Color

- 5. Select the desired color or define a custom color for the labware section.
- 6. Choose OK.

Note: Choose Cancel to cancel the color selections.

- 7. Repeat the process for each section of the labware, if desired.
- 8. Choose **OK** in Labware Colors when complete.

Note: Choose Cancel to cancel the color selections.

9. Choose **Save**. Labware Types appears (Figure 7-8).

**Note:** Choose **Cancel** to cancel the changes made to the labware type. A Warning appears to confirm the cancellation of unsaved changes to the labware (Figure 7-16).

₩arning	×
$\triangle$	You will lose all unsaved changes to this labware type. Continue?
	Yes <u>N</u> o

Figure 7-16. Warning to confirm cancellation of changes made

10. Choose Exit.

#### 7.3.6.2 Configuring a Reservoir with Sections

Some methods require the use of a reservoir with sections. Use **Sections** to configure a reservoir with sections. This allows the pod to accurately aspirate, dispense, and transfer liquid using a reservoir with sections.

To configure a reservoir with sections:

- 1. Open Labware Types (refer to Section 7.3.1, <u>Accessing the Labware Type</u> <u>Editor</u>).
- 2. Copy an existing reservoir (refer to Section 7.3.3, *Copying a Labware Type*) or create a new one (refer to Section 7.3.2, *Adding a New Labware Type*). The new labware appears in Labware Types.
- 3. Highlight the graphic representing the new reservoir and choose **Edit** at the top of Labware Types.

OR

Right-click the graphic and choose **Edit**.

OR

Double-click the graphic.

4.	Choose	Sections	(Figure	7-17	)
----	--------	----------	---------	------	---

Reservoir	
Basic Information Miscellaneous Movement Information Orbital Shaker Ordering Information Sections Stacking	Section shapes         Drag the desired section         shapes to this areas after         step 6 below.         Delete         Edit Profile         Edit Top View
	Describes the regions where liquid may be placed in the reservoir. There are no errors in this labware type definition.

Figure 7-17. Sections chosen

- 5. Click in the reservoir. A green rectangle appears.
- 6. Choose **Delete**. Error appears as a reminder that sections are deleted and new sections must be added (Figure 7-18).



Figure 7-18. Error reminder that sections are deleted

- 7. Choose OK.
- 8. Drag the shape of the section(s) to the reservoir.
- 9. Click the area around a specific section. A green rectangle appears.

10. Choose Edit Profile. Edit Section Profile appears (Figure 7-19).

Edit Section Pr	ofile			
Height	1.453	cm		
Max Volume	110000	μL		
Absolute Max	117214.323	μL		
Default Volume	0	μL		
🗖 Bottom Secti	on			
Bottom Section H	Height 0	cm		
	OK		Cancel	

Figure 7-19. Edit Section Profile

- 11. If necessary, change the values in **Height**, **Max Volume**, and **Default Volume** according to the manufacturer's specifications.
- 12. Check **Bottom Section** if the section includes a triangular bottom.

**Note:** If Bottom Section is checked, enter the value in **Bottom Section Height**.

- 13. Choose OK.
- 14. Choose Edit Top View. Edit Section Top appears (Figure 7-20).

Edit Se	ection Top				
Left	0.826	cm	M	110000	
Right	11.93	cm	Max Volume	195650 209	μL
Back	0.635	cm	Absolute Max	0	μL
Front	7.9	cm	Derauit Volume	lo lo	μι
		OK	Cancel		

Figure 7-20. Edit Section Top

- 15. If necessary, change the values in Left, Right, Back, Front, Max Volume, and Default Volume according to the manufacturer's specifications.
- 16. Choose OK.

- 17. Repeat steps 8 through 16 for each section in the reservoir.
- 18. When all sections have been configured, choose **Save** (Figure 7-17). A graphic representing the reservoir with the sections appears in Labware Types.

**Note:** An Error (Figure 7-21) appears if the sections are not configured properly. Choose **OK** and reconfigure the sections.

Error	×
$\otimes$	This reservoir has sections that overlap each other
	<u> </u>

Figure 7-21. Error for an improperly configured reservoir

**Note:** Choose **Cancel** to cancel the changes made to the labware type. A Warning appears to confirm the cancellation of unsaved changes to the labware (Figure 7-16).

#### 7.3.6.3 Specifying Stackable Items

Many common microplates are stackable up to six centimeters in height; however, items such as reservoirs, tube racks, and tip boxes are not.

Specifying stackable items is necessary to allow the software to effectively move labware stacks during liquid-handling operations. For items that are stackable, use **Stackable Items** to specify the distance from one piece of labware to the labware above it (stack offset). A stack offset is the distance between stacked plates, measured from the bottom, back-left corner on a microplate to the bottom back-left corner on the microplate above it. The distance is measured using X, Y, and Z coordinates.

**Note:** If it is desired to place more than two pieces of labware in a stack, they must stack securely to each other (refer to Section 7.3.6.4, *Using Secure Stacking*).

Stackable Items:

- specifies with a checkmark which labware types can be stacked on the labware being edited.
- may be configured to allows checkmarks to be entered when it is necessary to stack a specific type of labware on labware being edited for a specific method.

The information entered in Stackable Items is used by the software when moving labware, for example, in the Move Labware step.

To specify stacking offset:

1. From Labware Types, highlight the graphic representing the labware and choose **Edit** from the toolbar at the top.

OR

Right-click the graphic and choose Edit.

OR

Double-click the graphic.

#### 2. Choose Stacking (Figure 7-22). GreinerFlat3845quare 5 <u>S</u>ave <u>C</u>ancel **Basic Information** Х Y Ζ Magbead Miscellaneous 1.25 0 0 Stack Offset cm Movement Information Stacking Speed Limit 100 Orbital Shaker % Ordering Information Stack Offsets E dit... Stacking Wells\_1 E dit... Secure Stacking ☑ Allow Self-stacking Preview Hint Errors Describes how other labware types can There are no errors in this labware type be stacked on this one. definition.

Figure 7-22. Stacking chosen
Choose Edit next to Stack Offsets. Stackable Items appears (Figure 7-23).



Figure 7-23. Stackable Items

4. Choose the desired labware types that can be stacked on top of the labware type being edited.

**Note:** If an item does not have a checkmark next to it, that type of labware cannot be stacked on the labware type being edited (selected in Labware Types); however, if Allow Self-stacking is checked in Stacking, it is not necessary to check it in Stackable Items.

5. If applicable, type the desired value for **Stack Offset X**, **Stack Offset Y**, and **Stack Offset Z**.

**Note:** While the offset values may be calculated using calipers, a more precise measurement should be obtained from the manufacturer's specifications.

- 6. Choose **OK** to save the stackable items settings.
- 7. Choose Save. Labware Types appears.

**Note:** Choose **Cancel** to cancel the changes made to the labware type. A Warning appears to confirm the cancellation of unsaved changes to the labware (Figure 7-16).

8. Choose Exit.

### 7.3.6.4 Using Secure Stacking

Use Secure Stacking when it is necessary to securely stack more than two plates. The notations made in Secure Stacking are used by the software to allow the creation of stacks for a Solid Phase Extraction (SPE) assembly or tall stacks. It is necessary to use Secure Stacking with Stackable Items to ensure that specific labware can be stacked on top of other specific labware.

Like Stackable Items, Secure Stacking allows checkmarks to be entered when it is necessary to stack a specific type of labware on another specific type of labware for certain methods. The information entered in Secure Stacking will be used by the software when moving labware, for example, in the Move Labware step.

To effectively use Secure Stacking, an understanding of the Biomek stacking rules is necessary. The Biomek stacking rules apply to stacks on all Biomek instrument decks.

#### 7.3.6.4.1 Biomek Stacking Rules

When stacking or unstacking labware, Biomek Software works from the bottom up. If there is a stack of four microplates on deck, the top three are moved by the gripper, then the top two, and finally the top piece of labware.

The maximum height of stacks of microplates allowed on the Biomek instrument decks is as follows:

- > **FX, NX-MC** 5.5 cm (2.17 in.)
- ➤ 3000 6.0 cm (2.36 in.)
- **NX-S8** 6.5 cm (2.56 in.)

Note: A stack of four standard 96-well plates is approximately 5.2 cm (2.05 in.) tall.

A plate (or stack of plates) can be removed from the source position if one of the following conditions is true:

- The stack at the source position has fewer than three plates.
- The entire stack is being moved.
- The selected plate to be moved is the plate immediately above the bottom plate.
- All of the plates below the selected plate to be moved securely stack to each other.
- The selected plate to be moved securely stacks to the source position.
- The plate immediately below the selected plate to be moved securely stacks to the source position.

A plate (or stack of plates) can be moved to a destination position if one of the following conditions is true:

- The destination position has fewer than two plates (one or none).
- All of the plates that are currently stacked at the destination position securely stack to each other.
- The top plate of the destination stack securely stacks to the destination position.
- The selected plate to be moved securely stacks to the destination position.

### 7.3.6.4.2 Configuring Secure Stacking

To use Secure Stacking:

1. From Labware Type Editor, highlight the graphic representing the labware and choose **Edit**.

OR

Right-click the graphic and choose Edit.

OR

Double-click on the graphic.

- 2. Choose Stacking (Figure 7-22).
- 3. Choose **Edit** next to Secure Stacking. Secure Stacking appears (Figure 7-24).

Labware Types:     ALP types:       AP384_30uL     BarcodeRe       AP96_200uL     BarcodeRe       AP96_200uL     FourByFour       AP96_20uL     OneByFive       AP96_20uL     OneByFive	
BCDeep96Round       OneByThree         BCDeep96Square       PositivePos         BCFlat96       ShakerALF         BCFulReservoir       Span8TipT         BCSPECollar       Span8Tras         BCTuberack_10mm       Span8Tras         BCTuberack_12mm       Span8Was         BCTuberack_13mm       Span8Was         BCUpsideDownTipBoxLid       SPE         Collar_36mm       SPE2x1Lef         CollarSpacer_23mm       SPE2x1Rig         CostarCone96Round       SPEHolder         CostarFlat384Square       StirrerALP         CostarFlat384Square       TipLoader         FilterHolder       TipTrash	aderLeft aderRight r sitionALP rash hLeft hRight theft that tousel

Figure 7-24. Secure Stacking

- 4. Check the labware types and/or ALP types which lock securely to the labware type being edited.
- 5. Choose **OK** to save the settings.
- 6. Choose **Save**. Labware Types appears.

**Note:** Choose **Cancel** to cancel the changes made to the labware type. A Warning appears to confirm the cancellation of unsaved changes to the labware (Figure 7-16).

7. Choose Exit.

### 7.3.6.5 Configuring Well Properties

Well properties must be configured when new labware is added to the deck.



CAUTION: The correct well properties must be defined in Labware Types according to the manufacturer's specifications. Inaccurate specifications may lead to inaccurate pipetting.

To configure well properties:

1. From Labware Types, highlight the graphic representing the labware and choose **Edit**.

OR

Right-click the graphic and choose **Edit**.

OR

Double-click the graphic.

2. Choose Wells\_1 (Figure 7-25).

GreinerFlat384Square	
Save Cancel	
Basic Information Magbead Miscellaneous Movement Information Orbital Shaker Ordering Information Stacking Wells 1	X     Y       Well Offset     1.215     0.905     cm       Well Count     24     16       Well Spacing     0.45     0.45     cm       Maximum Volume     140.874     µL       Well Configuration     Edit       Format     Offset     I       Column 2 Offset     1.12268     cm       Default Volume     0     µL
Preview	Hint     Errors       Describes how a set of wells are arranged and configured.     There are no errors in this labware type definition.

Figure 7-25. Wells\_1 chosen

3. Choose **Edit** next to Well Configuration. Well Configuration Dialog appears (Figure 7-26).

Well Configuration Dialog
Shape Round
Upper Radius 0.343 cm
Lower Radius 0.3175 cm
Height 1.0668 cm
E Bottom Section
Shape Flat
Maximum possible volume 365.7078
OK Cancel

Figure 7-26. Well Configuration Dialog

4. Choose the Shape from the drop down menu.

5. Configure the specifications according to the manufacturer's specifications.

**Note:** Placing the cursor in the specification fields identifies the measurement area on the graphic shape. Placing a check in the Bottom Section field identifies the measurement area on the graphic shape that represents the bottom section of the well (Figure 7-27).

Well Configuration Dialog
Shape Round
Upper Radius 0.185 cm
Lower Radius 0.185 cm
Height 1.15 cm
☑ Bottom Section
Shape Hemisphere
Radius 0.33 cm
Maximum possible volume 198.9154
OK Cancel

Figure 7-27. Well Configuration Dialog with Bottom Section checked

- 6. Choose **OK** in Well Configuration Dialog when complete.
- 7. Choose Save. Labware Types appears.

**Note:** Choose **Cancel** to cancel the changes made to the labware type. A Warning appears to confirm the cancellation of unsaved changes to the labware (Figure 7-16).

8. Choose Exit.

# Understanding and Creating Liquid Types

## 8.1 Overview

The Liquid Type Editor (Figure 8-1) is used to create new liquid types or to modify existing liquid types for methods. Each time a new liquid type is created, it becomes available for all new and existing methods created in a Biomek Software project file. Any liquid type may be set as the default liquid type. In a method, the default liquid type is used when a piece of labware is not configured with a specific liquid.

**Note:** When Beckman Coulter Accounts & Permissions is enabled, only users with Develop Projects permission may open, edit, create, and delete liquid types (refer to Chapter 2, <u>Using Accounts & Permissions</u>).

The following is configured in the Liquid Type Editor:

 Aspirate — configures the delay, speed, and trailing air gap applicable to the liquid.

**Note:** A trailing air gap draws air into the tips after aspirate to help prevent drips during a method run.

- Blowout configures the amount of air used in the blowout, and the delay after the air has been dispensed.
- Prewet configures the amount of additional liquid to aspirate, and the amount of time to pause before dispensing.
- Wash configures the amount of liquid used to clean a tip and the default number of aspirate and dispense cycles used during the wash.
  - > **3000** Wash is *not* applicable for liquid types.
- Dispense configures the delay and speed of the dispense operation during a method run and the cutoff velocity.
  - FX, NX-S8 Cutoff velocity is applicable when the liquid is handled by a Span-8 Pod *only*.
- Tip Touch configures the position and amount of time to pause while the tip is in contact with the side of the well and the speed at which the pod should move during the operation.

- Liquid Level Sensing Sensitivity configures the sensitivity of the liquid level sensing for the liquid type.
  - FX, NX-S8 Liquid Level Sensing Sensitivity is applicable when the liquid is handled by a Span-8 Pod *only*.
- Clot Detection Sensitivity configures the sensitivity of the clot detection for the liquid type.
  - > **NX-S8** Clot Detection is applicable on this instrument *only*.

**Note:** When Override Liquid Type Settings is selected in the technique, the liquid type settings created in Liquid Type Editor are not used (overridden). Refer to Chapter 9, <u>Understanding and Creating Techniques</u> for more information about creating and editing techniques.

Liquid types, along with information about tip and labware types; well patterns; and pipetting templates and techniques, are stored as part of a project file. Project files store a history of all changes, additions, and deletions of items from the project file. Refer to Chapter 6, <u>Understanding and Using Project Files</u>, for more information on project files.

Liquid Type Editor New Remove Copy Liquid Type DMSO Ethanol Serum Water Default Liquid Type A checkmark appears next to the default liquid type.	Paste Paste Aspirate Trailing Air ( Delay: 0 Speed: 50 Blowout Volume: 10 Delay: 0 Prewet Overage: 0 Delay: 0	Defau Sap: 2	ult μL μL/s μL/s μL ms	Dispense Delay: 0 Speed: 50 Cutoff Velocity: 150 Tip Touch Height: -1 from Top V Angle: 90 Speed: 100	ms μL/s μL/s mm		
Liquid Type View Shows all defined liquid types.	Overage: 0 Delay: 0 Wash Default Cycl Default Volu	les: 4 me: 10	μL ms	Speed: 100 Delay: 0 Sensitivity Liquid Level Sensing 2500 3500 Clot Detection 2500 3500	% ms 4000	Color	
Liquid Type Configuration Shows configuration of selected type, and allows for modificated existing liquid types or the add new liquid types.	n View ed liquid ion of lition of	][	Graphical 1 of tip) and Colors corr Current De Gap is alw	OK Cancel Tip Displa representation of Tra Blowout (at top of t respond with the Tip ck Display of the ma rays green and Blow	ailing tip), i Disp tin ed vout i	Apply Air Gap f applicate play in the itor. Trail s always f	(at end ole. e ing Air red.

Figure 8-1. Liquid Type Editor

## 8.1.1 Understanding Liquid Types

Liquid type information is used by techniques to help control the aspirate and dispense operations of a method. Due to liquid characteristics such as viscosity or volatility, each type of liquid may need to be pipetted in a different way. By defining liquid types in the Liquid Type Editor, the liquids will pipette consistently among methods. Liquid types containing properties such as Prewet Delay, Tip Touch, or Blowout Volume may be configured within the Liquid Type Editor. Any configured liquid type may be set as the default and used when labware is not configured for any particular liquid type.

# 8.2 Creating New Liquid Types

New liquid types are created by setting information in the Liquid Type Editor. Liquid type settings are used by Biomek Software in the steps of a method to determine how pipetting operations occur.

To create a new liquid type:

1. Choose **Project>Liquid Type Editor**. Liquid Type Editor appears (Figure 8-1).



2. Choose **New**. The new liquid type appears in the Liquid Type with the name "NewLiquid" (Figure 8-2).

Liquid Type Editor					
	<b>b</b> (12)	<b>N</b>			
New Remove	Copy Paste	Default		1	
Liquid Type	Aspirat	e		Dispense	
DMSO Ethapol	Trailing	Air Gap: 0	μL	Delay: 0 ms	
NewLiquid	Delay:	0	ms	Speed: 100 µL/	's
Serum	Speed:	100	μL/s	Cutoff Velocity: 150 µL/	's
	Blowou	t		Tip Touch	
	Volume	10	μL	Height: -1	mm
	Delay:	0	ms	from Top 🗾 🧷	
	Prewet	_		Angle: 90	
	Overag	e: 0	μL	Speed: 100	%
	Delay:	0	ms	Delay: 0	ms
	Wash			Sensitivity	
	Default	Cycles: 1		Liquid Level Sensing	
	Default	Volume: 100%	μL		
				2500 3500 40	000
				Clot Detection	
				2500 3500 40	
				OK Cancel	Apply

Figure 8-2. Liquid Type Editor with a new liquid

- 3. Click on **NewLiquid**, pause briefly, and click on **NewLiquid** again to select the liquid name as text.
- 4. Enter the name of the new liquid and press **Enter** to save it.

**Note:** Liquid type names may contain only alphanumeric characters and the underscore (\_).

5. In Aspirate, enter the volume of the **Trailing Air Gap** in microliters (μL). After aspirating the liquid, a **Trailing Air Gap** is aspirated to prevent dripping.

- 6. In Aspirate, enter the desired **Delay** in milliseconds (ms). Delay specifies the amount of time after an aspirate operation before continuing.
- 7. In Aspirate, enter the desired **Speed** in microliters per second ( $\mu$ L/s). This indicates the speed along the D-axis during aspirate.
- 8. In Blowout, enter the desired **Volume** of the leading air gap to blow out after dispensing the liquid.
- 9. In Blowout, enter the desired **Delay** in milliseconds (ms). Delay specifies the amount of time after a blowout before continuing.
- 10. In **Prewet**, enter the desired **Overage** in microliters. **Overage** is the amount of liquid to be aspirated over the amount specified in the step configuration.
- 11. In **Prewet**, enter the desired **Delay** in milliseconds (ms). **Delay** specifies the amount of time from when the **Prewet** volume is completely aspirated until it begins to dispense.
- 12. In Wash, enter the **Default Cycles**. Default Cycles is the number of aspirate and dispense cycles to use when washing.
- 13. In Wash, enter the **Default Volume** as a percentage of the maximum used volume followed by the percent sign (%) or as an absolute amount in microliters. The **Default Volume** is the amount of liquid to aspirate and dispense in each cycle.
- 14. In Dispense, enter the desired **Delay** in milliseconds (ms). Delay specifies the amount of time after a dispense operation before continuing.
- 15. In Dispense, enter the desired **Speed** in microliters per second ( $\mu$ L/s). This indicates the speed along the D-axis during dispense.
- 16. In Dispense, enter the desired **Cutoff Velocity** in microliters per second ( $\mu$ L/s).
  - FX, NX-S8 Cutoff velocity is applicable when the liquid is handled by a Span-8 Pod *only*.

**Note:** Cutoff Velocity is the dispense speed of the syringe before it abruptly stops, allowing droplets to break off from the tip of a Span-8 Pod. This is particularly useful for low-volume dispensing.

**Note:** The minimum and maximum cutoff velocities vary with the syringe type. The value entered must fall within the acceptable range for the syringe type. Refer to Table 8-2 for the minimum, maximum, and recommended cutoff velocities for each syringe type.

17. In Tip Touch, enter the desired **Height**.

**Note:** Height, along with from and Angle, specify where along the well the Tip Touch occurs.

18. In Tip Touch, select the desired **from**: Top, Bottom, or Liquid. This indicates the location the Height is measured from.

**Note:** Top measures from the top of the well, Bottom from the bottom of the well, and Liquid from the top of the liquid.

19. In Tip Touch, enter the desired **Angle**.

OR

Select the position along the edge on the graphic by positioning the smaller circle at the desired location. The Angle adjusts to the current location.

**Note:** The Angle specifies the location along the well wall where the Tip Touch occurs at the specified Height and from location.

- 20. In Tip Touch, enter the desired Delay.
- 21. In Tip Touch, enter the desired **Speed** as a percentage of the total speed.
- 22. In Liquid Level Sensing, select the desired **Sensitivity**. Sensitivity is the magnitude of change required for LLS to detect the liquid. A larger Sensitivity detects smaller changes.
  - FX, NX-S8 Liquid Level Sensing Sensitivity is applicable when the liquid is handled by a Span-8 Pod *only*.

**Note:** The Sensitivity value is only used if the selected technique has Inherit From Liquid Type selected for the technique sensitivity. Refer to Section 9.4.5, *Setting Liquid Level Sensing Values (FX, 3000, and NX-S8 only)*.

23. In Clot Detection, select the desired **Sensitivity**. Sensitivity is the magnitude of change required for clots to be detected during an aspiration. A larger Sensitivity detects smaller changes.

> **NX-S8** — Clot Detection is applicable on this instrument *only*.

**Note:** The Sensitivity value is only used if the selected technique has Inherit From Liquid Type selected for the technique sensitivity.

24. Choose **Color...** to designate a new display color for the liquid, if desired.

**Note:** The color affects only the display of aspirated liquid. The **Trailing Air** Gap is always green and Blowout is always red.

25. Choose **OK** to enter the new liquid type and close the Liquid Type Editor.

OR

Choose **Apply** to save the new liquid type without closing Liquid Type Editor.

OR

Choose **Cancel** to close the Liquid Type Editor without saving changes.

# 8.2.1 Changing the Aspirate and Dispense Speeds in the Liquid Type Editor

The aspirate and dispense speed is expressed in units of microliters per second. This represents the speed of the plunger in the D-axis on the Biomek instrument during pipetting operations. Each head or tool has a minimum and maximum allowable speed, as shown in Table 8-1.

**Note:** Earlier versions in Biomek Software expressed this speed as a percentage of the maximum speed.

Pod and H	ead or Tool	Minimum Speed (µL/s)	Maximum Speed (µL/s)
Biomek FX — Multichannel Pod	AP96 96-channel pod: P20 Head	0.3715	37.15
	P200 Head	1.486	148.6
	AP384 384-channel pod P30 Head	0.3715	37.15
Biomek FX — Span-8 Pod	250 μL syringe	2.083	241.67
	500 μL syringe	4.167	483.33
	1000 µL syringe	8.333	966.67
Biomek 3000	P20 Single-Tip Pipette Tool	0.039	39.4
	MP20 Multi-Tip Pipette Tool	0.039	39.4
	P200 Single-Tip Pipette Tool	0.304	304
	MP200 Multi-Tip Pipette Tool	0.304	304
	P1000 Single-Tip Pipette Tool	1.779	1780
Biomek NX Multichannel	AP96 96-channel pod: P20 Head	0.3715	37.15
	P200 Head	1.486	148.6
	AP384 384-channel pod P30 Head	0.3715	37.15
Biomek NX Span-8	100 μL syringe	0.833	96.667
	250 μL syringe	2083	241.667
	500 μL syringe	4.167	483.33
	1000 µL syringe	8.333	966.667
	2.5 mL	20.833	2416.667
	5 mL	41.667	4833.333

### Table 8-1. Minimum and Maximum Pipetting Speeds in D-Axis

### 8.2.2 Changing the Cutoff Velocity for Span-8 Pods in Liquid Type Editor (FX, NX-S8 only)

The minimum and maximum Cutoff Velocity and the instruments on which they are supported are listed in Table 8-2. When changes to Cutoff Velocity are required, they must be made to the technique governing the pipetting operation in the Technique Editor, and to the parameters of the fluid used during the pipetting operation in the Liquid Type Editor.

**Note:** The minimum Cutoff Velocity for each syringe is established using the equation:  $50 \times \text{Syringe}$  size in  $\mu L > /6000$ . The maximum Cutoff Velocity for each syringe is established using the equation:  $2700 \times \text{Syringe}$  size in  $\mu L > /6000$ .

The cutoff velocity should be adjusted in all Span-8 techniques for better pipetting accuracy every time syringes are changed from one type to another. The Cutoff Velocities displayed in Table 8-2 are intended as a starting point. The values provided should be experimented with to determine the most accurate Cutoff Velocity for a specific liquid handling operation.

Syringe Size	Minimum Cutoff Velocity	Maximum Cutoff Velocity	Instruments						
100 µL	0.833 µL/second	$45.0 \ \mu L/second$	NX-S8						
250 μL *	2.08 µL/second	112.5 µL/second	FX, NX-S8						
500 µL**	4.17 µL/ second	225 µL/second	FX, NX-S8						
1 mL ***	8.33 $\mu$ L/ second	450 µL/second	FX, NX-S8						
2.5 mL	20.8 µL/second	1125 µL/second	NX-S8						
5 mL	41.7 µL/second	2250 µL/second	NX-S8						
* Recommended cutoff velocity is 100 μL/second. ** Recommended cutoff velocity is 150 μL/second.									

 Table 8-2.
 Span-8 Cutoff Velocities for Syringes for Span-8 Pods

To change the parameters of a fluid to conform to the Cutoff Velocity supported by 250  $\mu$ L and 500  $\mu$ L syringes:

1. With Biomek Software open, choose **Project>Liquid Type Editor**. The Liquid Type Editor appears (Figure 8-3).

Liquid Type	Editor							
D	×	Ē	C2	ľ				
New	Remove	Сору	Paste	Default		1		
Liquid Type			Aspirate			Dispense		
DMSO Ethanol			Trailing Air	Gap: 2	μL	Delay: 0	ms	ЦШШЦЦ
Serum			Delay: 0		ms	Speed: 50	μL/s	
Water			Speed: 50		μL/s	Cutoff Velocity: 150	μL/s	
			Blowout			Tip Touch		
			Volume: 10		μL	Height: -1	mm	
			Delay: 0		ms	from Top		
			Prewet			Angle: 90		
			Overage: 0	)	μL	Speed: 100	%	
			Delay: 0		ms	Delay: 0	ms	
			Wash			Sensitivity		
			Default Cyc	les: 4		Liquid Level Sensing		
			Default Volu	ıme: 100%	μL			
						2500 3500	4000	
							<u> </u>	1
						2500 3500	4000	Color
,						ОК Са	ancel	Apply

Figure 8-3. Liquid Type Editor

- 2. Select the liquid used during the pipetting operation.
- 3. In Dispense, enter the desired **Cutoff Velocity**, in microliters per second  $(\mu L/s)$  (refer to Table 8-2 for the correct Cutoff Velocity).
- 4. Choose **OK** to save the liquid type and exit the Liquid Type Editor.

## 8.3 Removing Liquid Types

If a liquid type is no longer needed, it can be removed from the project.

**Note:** When a liquid type is removed, previous revisions of the liquid type will remain available for validated methods, but will not be available for future method editing tasks.

To remove a liquid type:

1. In the Liquid Type Editor, select the liquid type to remove.



- 2. Select Remove.
- 3. Choose **OK**. The liquid type is removed.

## 8.4 Copying and Pasting Liquid Types Within a Project File

To use the same liquid type with different parameters within a project, copy and paste the liquid type within the Liquid Type Editor.

**Note:** Liquid types can be exported to and imported from other projects. Refer to Section 6.10, *Importing and Exporting Project Files* for information about importing and exporting liquid types.

To copy and paste liquid types:

1. In the Liquid Type Editor, select the liquid type to copy.



2. Choose **Copy**.



Paste

- 3. Choose **Paste**. The new liquid appears with the name "Copy\_of\_(Liquid Type)".
- 4. Rename the liquid type as desired.

**Note:** Liquid type names may contain only alphanumeric characters and the underscore (\_).

- 5. Make any desired changes to the liquid type (refer to Section 8.2, <u>*Creating New</u></u><u>Liquid Types</u>).</u>*
- 6. Choose OK.

# 8.5 Specifying the Default Liquid Type

Any defined liquid type can be the default. Any labware on the deck not configured with a specific liquid type uses the default liquid type for pipetting operations.

To set the default liquid type:

1. In the Liquid Type Editor, select the liquid type to set as default.



#### 2. Choose Default.

**Note:** The default liquid type appears with a check mark in the Liquid Type list. When the default liquid type is highlighted in the Liquid Type list, the Default button is toggled on.

# 9 Understanding and Creating Techniques

## 9.1 Overview

A technique stores a set of values and properties that instruct the Biomek instrument in performing pipetting operations, such as aspirate, dispense, mix, pod height, pod speed, and tip touch. The Biomek Software also stores a set of properties related to each technique, such as labware type and liquid type. Based upon these values and properties, the appropriate technique is selected automatically for the pipetting operation.

Techniques, along with information about tip and labware types; liquid types; well patterns; and pipetting templates, are stored as part of a project file. Project files store a history of all changes, additions, and deletions of items from the project file. Refer to Chapter 6, <u>Understanding and Using Project Files</u>, for more information on project files.

**Note:** When Beckman Coulter Accounts & Permissions is enabled, only users with the **Develop Projects** permission assigned may open, edit, create, and delete techniques (refer to Chapter 2, *Using Accounts & Permissions*).

### 9.1.1 How Techniques Work

Biomek Software is preprogrammed with a variety of techniques. The properties input during method building are matched with technique properties to select the most appropriate technique for use in the pipetting operation. Techniques are selected automatically based upon the number of properties that match the method. For example, a technique that matches five properties in a method is selected over a technique with four matching properties.

In most instances, one technique will have the closest match with the properties of the current pipetting operation; however, if more than one technique matches the same number of properties, Biomek Software selects the technique with the highest rank.

Because Biomek Software selects techniques automatically, any change within a method may result in a new technique selection. When a single value or property is changed in the method, Biomek Software makes sure the technique is still the most appropriate technique to use.

## 9.1.2 Why Techniques are Valuable

Techniques are a flexible way to make the pipetting process easier. Techniques allow the pipetting settings to be saved globally and used in multiple methods. Each time a method is created, the technique most appropriate for the pipetting operation is selected with no additional configuration required. When multiple sources and liquids are used in a method, each pipetting operation may use a different technique. This allows for appropriate pipetting throughout the entire method.

Customized techniques may also be created. Once additional techniques are created, they appear and function like predefined techniques.

# 9.2 Accessing the Technique Browser

To access the Technique Browser:

From the Project menu, choose Technique Browser.

The Technique Browser appears (Figure 9-1). The Technique Browser contains two main views:

- **Groups** displays all user-created groups that contain a subset of techniques in the project; refer to Section 9.7, <u>*Creating Technique Groups*</u>, for more information.
- Techniques View lists all techniques defined in the selected group with their technique properties.

Technique Browser									
🔊 New Group 🍵 Remove Group 🚿 Edit Group 😭 Rem	name Grou	o 🖪 🖼 Clo	se						
R New Beneve Eth Conv Blacto Still P	Bropertie								
S Tenne de Coby De Caste Z fort E	rigperde:	°		1			1 .		
Groups	Tips	Head	Pod	Labw	liquidt	maxi	rank	minim	Syrin
🕲 (All) 🖉 Washing	Fixed	*	Pod9	Wash	*	1000	40	*	*
🖉 1536 Plate	P30	ATTIL	Pod96	Grein	*	15	45	*	*
🖉 Low Volume 384 Plate	P30	ATTIL	Pod96	AB38	*	15	50	*	*
🖉 Low Volume 384 Wet Plate	P30	*	*	BCUp	*	2	50	0.5	*
🔊 Low-vol Reservoir	*	*	Pod96	BCDe	*	25	58	*	*
🗐 Span-8 Viscous	Fixed	*	Pod8	*	Serum	*	59	*	*
🗐 Span-8 Low 80	Fixed	*	Pod8	*	*	80	59	*	*
🔊 Low-Volume	*	*	Pod96	AB38	*	25	59	*	*
🔊 Span-8	Fixed	*	Pod8	*	*	*	59	80	*
🔊 Reservoir	*	*	Pod96	BCDe	*	*	60	15	*
🔊 Standard	*	*	Pod96	AB38	*	*	60	15	*
🔊 Reservoir (ETOH)	*	*	Pod96	BCDe	Ethanol	*	61	*	*
😰 Span-8 MultiDispense	Fixed	*	Pod8	*	*	*	60	*	*
🔊 Default	*	*	PodB	*	*	*	99	*	*
						1			
				-					
Gro	ups				ecnni	ques	view		
Selects group t	o displ	lay in		Show	s defin	ed tecl	hnique	s	
Techniques Vi	ew			that m	natch se	electio	ns fror	n	
	• • • •			Conto	ovt Inf	ormat	ion		
				COIL		onnat			
14 Techniques									

Figure 9-1. Technique Browser

## 9.2.1 Identifying Techniques

Techniques are identified by name in the **Technique Browser** (Figure 9-2). The name of the technique is a control name (identifier) which is related to and identifies the values used during a method run. When a technique is selected, the name appears in the **Technique** field (Figure 9-25).

🖉 <u>N</u> ew 📗 <u>R</u> emo	ve 🖻 Copy 💼 Paste 🔊 Edit	🚰 Propertie	s							
Groups	Name	Tips	Head	Pod	Labw	liquidt	maxi	rank	minim	Syrin.
🕑 (All)	🖉 Washing	Fixed	*	Pod9	Wash	*	1000	40	*	*
	🔊 1536 Plate	P30	ATTIL	Pod96	Grein	*	15	45	*	*
	🖉 Low Volume 384 Plate	P30	ATTIL	Pod96	AB38	*	15	50	*	*
	🔊 Low Volume 384 Wet Pl	late P30	*	*	BCUp	*	2	50	0.5	*
	🖉 Low-vol Reservoir	*	*	Pod96	BCDe	*	25	58	*	*
	Span-8 Viscous	Fixed	*	Pod8	*	Serum	*	59	*	*
	Span-8 Low 80	Fixed	*	Pod8	*	*	80	59	*	*
	Low-Volume	*	*	Pod96	AB38	*	25	59	*	*
	Span-8	Fixed	*	Pod8	*	*	*	59	80	*
	Reservoir	*	*	Pod96	BCDe	*	*	60	15	*
	Standard	*	*	Pod96	AB38	*	*	60	15	*
	Reservoir (ETOH)	*	*	Pod96	BCDe	Ethanol	*	61	*	*
	Span-8 MultiDispense	Fixed	*	Pod8	*	т •	т •	60	*	*
	Derault		*	POdB	*			99		
						<b>Techniques</b> Available techniques are selected automatically for different pipetting operations in a method.				
4 Techniques	<b>▲</b> )									
<b>Gr</b> Filters all te	oups echniques to	Te Disp	echnic lays all	ues \ techni	<b>/iew</b>	r				

Figure 9-2. Technique Browser where technique parameters are displayed

# 9.3 Creating New Techniques

The default techniques are sufficient for some pipetting operations and are meant as a starting point; there are instances when additional techniques may be required. For example, a method may call for a technique for a 384-well titerplate that transfers a volume between 5  $\mu$ L and 10  $\mu$ L of DMSO. When creating a new technique, the properties of the technique must be configured. For best results, all techniques should be evaluated and fine-tuned for the specific application through experimentation.

Techniques are automatically selected using properties. Properties identify certain aspects of the pipetting operation which might affect how the pipetting operation is performed. For a technique to be available for selection for a specific pipetting operation, all properties for the operation must match the properties in the technique.

The following properties are used to determine the optimal technique to use:

- Head identifies the tool types for which the technique is applicable; for example, a technique could be created which is only to be used with a specific tool or tools, such as single-channel tools, eight-channel tools, or wash tools. The technique is only used when the tool used for the pipetting operation matches this selection.
- Labware identifies labware types for which the technique is applicable; for example, a technique could be created which is only to be used when pipetting from a certain type of labware, such as reservoirs, deepwell microplates, or test tube racks. The technique is only used when the labware type used in the pipetting operation matches this selection.
- Liquid type identifies the liquid types for which the technique is applicable; for example, a technique could be created which is only to be used when pipetting a certain liquid type, such as DMSO or water. This can be useful to create special techniques when aspirating or dispensing viscous liquids. The technique is only used when the liquid type used in the pipetting operation matches this selection.
- Pod identifies the pod performing the pipetting operation; for example, separate techniques could be created for use with each pod type. The technique is only used when the pod type used in the pipetting operation matches this selection.
- Syringe Type identifies the syringe sizes for probes on a Span-8 Pod for which the technique is applicable. The technique is only used when the syringe type for the probes used in the pipetting operation matches this selection.

**FX**, **NX-S8** — Applicable for Span-8 Pods only.

Tips — identifies the tip types for which the technique is applicable; for example, a technique could be created which is only to be used with a certain type of tips, such as barrier tips or fixed tips. The technique is only used when the tip type used in the pipetting operation matches this selection.

- Volume identifies the volume range for which a technique is applicable; for example, a technique could be created that is only to be used when pipetting low volumes, such as 0-10  $\mu$ L. The technique is only used when the volume entered in the step configuration falls within the specified range.
- Group if groups have been created (refer to Section 9.7, <u>Creating</u> <u>Technique Groups</u>), an additional property is listed to allow the new group to be added to an existing group, if desired. Group is for organizing techniques only and is not used for selecting the technique to use.

Create additional techniques using the Technique Browser (Figure 9-1).

**Note:** Automatic selection of techniques may be turned off and new techniques created within a step configuration (refer to Section 9.10, <u>Selecting and Modifying</u> <u>Techniques Manually in a Method</u>).

To create a new technique and set its properties:

 From the Project menu, choose Technique Browser. Technique Browser appears (Figure 9-1).



2. In the Technique Browser, choose New. Technique Properties appears (Figure 9-3).

OR

Right-click in the Techniques View and choose New from the menu.





3. In Technique Name, enter a name to identify the technique.

4. In Rank, enter a value to set the relative preference of the technique to other techniques with similar properties.

**Note:** Rank allows Biomek Software to give techniques preference over others. For example, if two techniques use a 384-well titerplate and P20 tips, with the difference between techniques being the Mix Before Aspirate values, one technique may have a higher priority (Rank of 1) over the other (Rank of 2) for automatic selection, depending upon properties input during method building.

5. In Context Information, select the desired **Head**, **Labware**, **Liquid Type**, **Pod**, **Syringe Type**, and **Tips** to use for the technique.

**Note:** If no selection is made for a category, the technique is applicable for all items within that category. For example, if no labware types are selected under Labware, the technique is available for use with all labware types.

**Note:** To remove all current selections from the Technique Properties, rightclick in Context Information and select **Clear Selections**. Clear Selections removes all selections from all categories.

6. In Volume Range, enter the **Minimum Volume** and **Maximum Volume** for the technique to aspirate or dispense.

OR

Change the minimum and maximum volumes graphically using the handles on the graphical Volume Range (Figure 9-3).

**Note:** The left gauge is for Minimum Volume and the right gauge is for Maximum Volume.

7. Select **Do not Auto-Select** to exclude the new technique from the possible selected techniques for a step, such as **Transfer** or **Combine**, when the **Auto-Select** has been checked for the step configuration. The technique will not be automatically selected by Biomek Software for any pipetting operations, but is still available when manually selecting a technique if the properties match (refer to Section 9.10, *Selecting and Modifying Techniques Manually in a Method*).

**Note:** A technique that has Do not Auto-Select selected is displayed in the Technique Browser with a red **x** in the icon next to a technique.

8. Choose **OK**. The technique is created and added to the list in Technique Browser.

To access the properties of an existing technique to view or modify them:

Choose the **Properties** button on the Technique Browser toolbar.

OR

Properties

Right-click on the technique entry in the browser and select **Properties**.

**Note:** Check **Do not Auto-Select** to exclude the existing technique from the possible selected techniques for a step, such as **Transfer** or **Combine**, when the **Auto-Select** has been checked for the step configuration.

# 9.4 Setting Technique Values

After creating a technique and setting the related properties, set the technique values, such as Tip Touch or Mix After Dispense, in Technique Editor before performing a method.

**Note:** The **Technique Browser** lists all available techniques. The **Technique** Editor allows for configuration of the technique values for a specific technique.

In the Techniques Editor, a drop-down list is used to select the Pipetting Template (refer to Chapter 10, <u>Using the Pipetting Template Editor</u>). When the Pipetting Template is selected, the following nine tabs appear depending on instrument type and configuration.

The nine tabs are:

- General <u>Viewing Technique Description</u> (Section 9.4.1).
- Aspirate <u>Setting Aspirate Values</u> (Section 9.4.2).
- Dispense <u>Setting Dispense Values</u> (Section 9.4.3).
- Mix <u>Setting Mix Values</u> (Section 9.4.4).
- Calibration <u>Calibrating Techniques</u> (Section 9.5).
- Liquid Level Sensing <u>Setting Liquid Level Sensing Values (FX, 3000,</u> <u>and NX-S8 only)</u> (Section 9.4.5).
  - > **FX**, **NX-S8** Available with a Span-8 Pod.
  - > **3000** Available with P200L or P1000L Single-Tip Pipette Tool
- Clot Detection <u>Setting Clot Detection Values (NX-S8 only)</u> (Section 9.4.6).

> **NX-S8** — Available on this instrument only.

- Piercing <u>Setting Piercing Values (NX-S8 only)</u> (Section 9.4.7).
  - > **NX-S8** Available on this instrument only.
- Liquid Type <u>Overriding Liquid Type Values</u> (Section 9.4.8).

## 9.4.1 Viewing Technique Description

The General tab (Figure 9-4) summarizes the technique configuration on the Aspirate, Dispense, and Mix tabs in sentence form. The icons below each heading identify some of the operations the technique performs (Table 9-1).

Technique Editor - Default
Pipetting Template: Default Template
Calibration     Liquid Level Sensing     Liquid Type       General     Aspirate     Dispense     Mix
Aspirate Move within the well at 100% speed. Aspirate at 0 mm from the Bottom.
ä
Dispense Move within the well at 100% speed. Dispense at 0 mm from the Bottom.
u u u u u u u u u u u u u u u u u u u
Mix Move within the well at 100% speed. Aspirate and dispense at 0 mm from the Bottom.
đ
OK Cancel

Figure 9-4. Technique Editor — General tab

The icons are:

ruere y 1. roomique rooms	
lcon	Associated Operation
₹Œ	Tip Touch — touches the tip on the side of the well.
₩.	Blowout (used in Dispense operations only) — aspirates an air gap before the liquid aspirates, and dispenses the air gap after the liquid has been dispensed.
*	Follow Liquid — tip follows liquid level during pipetting operations.
<b>1</b>	Prewet (Aspirate operations only) — quickly aspirates and dispenses liquid before the main aspirate operation occurs.
Ğ	Mix — mixes liquid in wells before aspirate or after dispense.
<b>i</b>	Trailing Air Gap (Aspirate operations only) — aspirates air after liquid is aspirated to help prevent drips.

Table 9-1. Technique Icons

To view technique values in the General tab:

- 1. From the Project menu, choose **Technique Browser**. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.

3. Choose Edit.

OR

💁 <u>E</u>dit

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

**Note:** Access the Technique Editor (Figure 9-4) through the step configuration by choosing **Customize...** in the step configuration. This overrides the current technique in the pipetting operation only.

### 9.4.2 Setting Aspirate Values

Setting aspirate values allows for the technique to perform specific aspirate operations by changing related values, such as Aspirate Height, Pod Speed, and Mix Before Aspirate.

**Note:** When setting aspirate values from the **Customize** button in a step configuration, the **Technique Editor** is specific to the operation (Figure 9-5).

To configure Aspirate options:

- From the Project menu, choose Technique Browser. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.



3. Choose **Edit**.

OR

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

4. In Technique Editor, select the **Aspirate** tab (Figure 9-5).

Technique Editor - Default		
Pipetting Template: Default Template		
Calibration Liquid Level Sensing Liquid Type		
General Aspirace Dispense Mix		
Move within the well at 100 % speed.		
Aspirate at 0 mm from the Bottom		
Eollow liquid level when aspirating or dispensing liquid		
☑ Iouch tips on the sides of the wells		
Prewet the tips		
Aspirate a leading air gap for blowout		
Mix prior to aspirating liquid		
Mig 10 µL 1 time,		
Aspirate at 0 mm from the Bottom $\checkmark$ at 100 $\mu$ L/s.		
Dispense at 0 mm from the Bottom $\checkmark$ at 100 $\mu$ L/s.		
Aspirate a trailing air gap after leaving the liquid		
OK Cancel		

Figure 9-5. Technique Editor — Aspirate tab

5. In Move within the well at, enter the speed as a percentage of the maximum speed of the pod. This value must be between 0 and 100.

**Note:** This controls the speed of the X-, Y-, and Z-axes. This will affect the speed at which the pod moves to the aspirate and dispense heights. The actual aspirate and dispense speeds and the speed for tip touch are controlled by settings in the Liquid Type Editor (refer to Section 8.2, <u>*Creating New Liquid Types*</u>).

- 6. In Aspirate at, enter the distance in millimeters (mm) to measure from the selected position for the aspirate.
- 7. Select from where to measure the aspirate height: **Top, Bottom,** or **Liquid**.

**Note:** Top measures the aspirate height from the top of the well, Bottom from the bottom of the well, and Liquid from the top of the liquid. The only time a liquid level is required is when Aspirate is set to Liquid. It is not necessary to define the liquid level to use Follow Liquid.

When a specific height is entered in Aspirate at, the aspirate operation occurs at that distance relative to the aspirate position.

**Note:** A positive value is measured above the selected location; a negative value is measured below the selected location. For example, to aspirate a liquid 1.0 mm below the liquid level, a value of **-1.0** is entered in Aspirate at and Liquid is chosen as the aspirate position.

- 8. Choose **Follow liquid level when aspirating or dispensing liquid** to allow the tip to follow the liquid level during the aspirate operation.
- Choose Touch tips on the sides of the wells to allow tips to rest against the sides of the wells for a specific amount of time after aspirating (refer to Section 8.2, <u>Creating New Liquid Types</u>).
- Choose Prewet the tips to perform an aspirate and dispense cycle before performing the actual aspirate operations (refer to Section 8.2, <u>Creating New</u> <u>Liquid Types</u>).
- 11. Choose **Aspirate a leading air gap for blowout** to dispense an air gap on the Dispense operation.

**Note:** Blowout all leading air gaps must also be chosen on the Dispense tab to perform a complete blowout; just choosing Aspirate a leading air gap for blowout on the Aspirate tab will not perform a complete blowout.

 Choose Mix prior to aspirating liquid to mix liquid before the aspirate occurs, if desired. Refer to Section 9.4.2.1, <u>Setting Mix Before Aspirate Values</u>, for instructions on configuring mix properties.

**Note:** Mix aspirates and dispenses a specified number of times within a piece of labware.

 Choose Aspirate a trailing air gap after leaving liquid to use the defined Trailing Air Gap from the Liquid Type Editor or from the Liquid Type tab (refer to Section 8.2, <u>Creating New Liquid Types</u>).

### 9.4.2.1 Setting Mix Before Aspirate Values

Set the Mix prior to aspirating liquid values when a mix occurs before the liquid aspirates.

To set the Mix prior to aspirating liquid values:

- 1. In Mix, enter the amount of liquid to mix, and the number of times to mix.
- 2. Enter the **Aspirate at** height in millimeters (mm) at which to perform the Aspirate.
- 3. Choose the desired aspirate position: **Top, Bottom,** or **Liquid**.
- 4. Enter the speed at which to aspirate in  $\mu$ L/s.
- 5. Enter the **Dispense at** height in millimeters (mm) at which to perform the Dispense.
- 6. Choose the Dispense location: Top, Bottom, or Liquid.
- 7. Enter the speed at which to dispense in  $\mu$ L/s.

### 9.4.3 Setting Dispense Values

Setting the dispense values allows for specific dispense operations to occur by changing related values, such as Dispense Height, Pod Speed, and Mix After Dispense.

**Note:** When setting dispense values from the Customize button in a step configuration, the Technique Editor is specific to the operation (Figure 9-6).

To set Dispense values, complete the following:

- 1. From the Project menu, choose **Technique Browser**. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.

💁 <u>E</u>dit I

OR

3. Choose Edit.

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

4. In Technique Editor, select the **Dispense** tab (Figure 9-6).

Technique Editor - Default
Pipetting Template: Default Template
<u>C</u> alibration <u>L</u> iquid Level Sensing Liquid <u>Type</u> <u>G</u> eneral <u>A</u> spirate <u>D</u> ispense <u>Mix</u>
Move within the well at 100 % speed.
Dispense at 0 mm from the Bottom
Eollow liquid level when aspirating or dispensing liquid
✓ Iouch tips on the sides of the wells
✓ Blowout all leading air gaps
Mix after dispensing liquid
Mix 10 µL 1 time.
Aspirate at 0 mm from the Bottom $\checkmark$ at 100 $\mu$ L/s.
Dispense at 0 mm from the Bottom $\checkmark$ at 100 $\mu$ L/s.
DK Cancel

Figure 9-6. Technique Editor — Dispense tab

5. In Move within the well at, enter the speed as a percentage of the maximum speed of the pod. This value must be between 0 and 100.

**Note:** This controls the speed of the X-, Y-, and Z-axes. This will affect the speed at which the pod moves to the aspirate and dispense heights. The actual aspirate and dispense speeds and the speed for tip touch are controlled by settings in the Liquid Type Editor (refer to Section 8.2, <u>*Creating New Liquid Types*</u>).

6. In **Dispense at**, enter the distance in millimeters (mm) to measure from the selected position for the dispense.

7. Select from where to measure the dispense height: **Top, Bottom,** or **Liquid**.

**Note:** Top measures the dispense height from the top of the well, Bottom from the bottom of the well, and Liquid from the top of the liquid. The only time a liquid level is required is when Dispense is set to Liquid. It is not necessary to define the liquid level to use Follow Liquid.

When a specific height is entered in **Dispense at**, the dispense operation occurs at that distance relative to the dispense position.

**Note:** A positive value is measured above the selected location; a negative value is measured below the selected location. For example, to dispense liquid 1.0 mm below the liquid level, a value of **-1.0** is entered in **Dispense at** and **Liquid** is chosen as the dispense position.

- 8. Choose **Follow liquid level when aspirating or dispensing liquid** to allow the tip to follow the liquid level during the dispense cycle.
- 9. Choose **Touch tips on the sides of the wells** to allow tips to rest against the sides of the wells for a specific amount of time after dispensing.
- Choose Blowout all leading air gaps to dispense an air gap, if necessary (refer to Section 8.2, <u>Creating New Liquid Types</u>).

**Note:** Aspirate a leading air gap for blowout must also be chosen on the Aspirate tab to perform a complete blowout; just choosing Blowout all leading air gaps on the Dispense tab will not perform a complete blowout.

 Choose Mix after dispensing liquid to mix liquid after the dispense occurs. Refer to Section 9.4.3.1, <u>Setting Mix After Dispense Values</u>, for instructions on configuring mix properties.

### 9.4.3.1 Setting Mix After Dispense Values

Set the Mix after dispensing liquid values when a mix occurs after the liquid dispenses.

To set the Mix after dispensing liquid values:

- 1. In Mix, enter the amount of liquid to mix, and the number of times to mix.
- 2. Enter the **Aspirate** height in millimeters (mm) at which to perform the Aspirate.
- 3. Choose the desired aspirate position: **Top, Bottom,** or **Liquid**.
- 4. Enter the speed at which to aspirate liquid in  $\mu$ L/s.
- 5. Enter the **Dispense** height in millimeters (mm) at which to perform the Dispense.
- 6. Choose the Dispense location: **Top**, **Bottom**, or **Liquid**.
- 7. Enter the speed at which to dispense liquid in  $\mu$ L/s.

### 9.4.4 Setting Mix Values

Set Mix values to create a custom mix. Mix step aspirates and dispenses liquid a specified number of times during a method.

Note: The number of cycles is configured in the Mix step configuration.

To configure Mix:

- 1. From the Project menu, choose **Technique Browser**. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.



3. Choose Edit.

OR

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

4. In Technique Editor, select the **Mix** tab (Figure 9-7).

Technique Editor - Default		
Pipetting Template: Default Template		
Calibration     Liquid Level Sensing     Liquid Type       General     Aspirate     Dispense     Mix       Move within the well at     100     % speed.     %		
A <u>s</u> pirate at 0 mm from the Bottom ▼ at 100 μL/s. Dispense at 0 mm from the Bottom ▼ at 100 μL/s.		
Eollow liquid level when aspirating or dispensing liquid		
<ul> <li>Aspirate a leading air gap prior to mix and blowout after mix is complete</li> </ul>		
OK Cancel		

Figure 9-7. Technique Editor — Mix tab

5. In Move within the well at, enter the speed as a percentage of the maximum speed of the pod. This value must be between 0 and 100.

**Note:** This controls the speed of the X, Y, and Z axes. This will affect the speed at which the pod moves to the aspirate and dispense heights. The actual aspirate and dispense speeds and the speed for tip touch are controlled by settings in the Liquid Type Editor (refer to Section 8.2, <u>Creating New Liquid Types</u>).

- 6. Enter the **Aspirate at** height in millimeters (mm) at which to perform the Aspirate.
- 7. Choose the desired aspirate position: **Top, Bottom,** or **Liquid**.
- 8. Enter the speed at which to aspirate liquid in  $\mu$ L/s.
- 9. Enter the **Dispense at** height in millimeters (mm) at which to perform the Dispense.
- 10. Choose the Dispense location: **Top, Bottom,** or **Liquid**.
- 11. Enter the speed at which to dispense liquid in  $\mu$ L/s.
- 12. Choose **Follow liquid level when aspirating or dispensing liquid** to allow the tip to follow the liquid level during the aspirate and dispense cycles.
- 13. Choose **Touch tips on the sides of the wells** to allow tips to rest against the sides of the wells for a specific amount of time after mixing.
- Choose Aspirate a leading air gap prior to mix and blowout after mix is complete to dispense an air gap, if necessary (refer to Chapter 8, <u>Understanding and Creating Liquid Types</u>).
# 9.4.5 Setting Liquid Level Sensing Values (FX, 3000, and NX-S8 only)

- **FX** Available with a Span-8 Pod only.
- 3000 Available on P200L or P1000L Single-Tip Pipette Tool only. Some options are not available with this instrument and do not appear.

Set Liquid Level Sensing values to utilize liquid level sensing to determine the liquid level. Liquid level detection is necessary only if the Aspirate, Dispense, or Mix height is measured from the liquid and the liquid level is unknown. The pod, head, and tips must all be compatible with liquid level sensing to detect the liquid level.

The Liquid Level Sensing tab consists of four collapsible sections to configure various aspects of using LLS. Each section is expanded by clicking the down arrow or sentence summary for that section and may then be configured. Only one section may be expanded at a time. When collapsed, each sections displays a sentence summary describing the configuration for that section.

The four sections are:

- General Settings specify the start height and sense speed.
- Repetition Settings specify the number of times to perform LLS
  - FX, NX-S8 Available on these instruments only.
- Sensitivity Settings select to inherit sensitivity from labware type, liquid type, or define a custom setting
  - **FX, NX-S8** Available on these instruments only.
- Error Handling Settings specify the action to take if LLS fails

To configure Liquid Level Sensing:

- From the Project menu, choose Technique Browser. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.



3. Choose Edit.

OR

•

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

Technique Editor - Span-8 Pipetting Template:				
<u>G</u> eneral <u>A</u> spirate <u>D</u> is <u>C</u> alibration <u>L</u> iquid Level Sensing	spense <u>M</u> ix 9 Liquid <u>T</u> ype			
➤ General Settings Move to 5 mm from the Top ▼ Sense at 20 % speed.	before sensing.			
<ul> <li>✓ Sense 1 time.</li> <li>✓ Inherit sensitivity from labware.</li> <li>✓ On error, retry 0 times. Then, prompt the user.</li> <li>General Settings specify the starting height within the well and pod speed in</li> </ul>				
Arrow to collapse/expand General Settings				
	OK Cancel			

4. In Technique Editor, select the Liquid Level Sensing tab (Figure 9-8).

Figure 9-8. Technique Editor - General Settings on Liquid Level Sensing tab

#### 9.4.5.1 Configuring General Settings

General Settings is used to configure the starting height of the probe within the well and the pod speed in the Z axis during liquid level sense operations.

- 1. If General Settings is not displayed, click on the topmost sentence summary to expand the General Settings configuration (Figure 9-8).
- 2. Enter the height in millimeters (mm) to **Move to** and start sensing from.
- 3. Select the from the start location: Top or Bottom. The probe moves to the specified height measured from the top or bottom of the well before sensing.
- 4. Enter the **Sense at** speed as a percentage of the maximum speed of the pod.

**Note:** The speed affects the Z-axis speed only. The default sense speed is 20 percent.

#### 9.4.5.2 Configuring Repetition Settings (FX, NX-S8 only)

Repetition Settings is used to configure the number of times to sense and conditions to accept results early or fail.

1. If **Repetition Settings** is not displayed, click on the sentence summary second from the top to expand the **Repetition Settings** configuration (Figure 9-9).

Technique Editor - Span-8				
Pipetting Template: Span-8				
General Aspirate Dispa Calibration Liquid Level Sensing	ense <u>M</u> ix   Liquid <u>T</u> ype			
➡▼ Start at 5 mm from the Top. Sense at 20% speed	i.			
A Repetition Settings				
Sense 1 time.				
Accept LLS result early if difference drops below	μL.			
Fail if LLS result difference never drops below 0 μL.				
✓ Inherit sensitivity from labware.				
<sup> </sup>				
Repetition Settings         Repetition Settings           configures options         for multiple senses.				
	OK Cancel			

Figure 9-9. Technique Editor - Repetition Settings on Liquid Level Sensing tab

2. In **Sense**, enter the number of times to sense. Specifying a **Sense** of zero (0) disables liquid level sensing for the technique.

**Note:** Sensing multiple times helps to pop air bubbles in the wells and increases the accuracy in the liquid level detection.

- 3. Select **Accept LLS result early** to halt liquid level sensing if consecutive readings are within a specified tolerance, if desired.
- 4. In **drops below**, enter the desired tolerance for Accept LLS result early. When a reading falls within the entered tolerance of the previous reading, the result is accepted as the liquid level.
- 5. Select **Fail if LLS result** to fail liquid level sensing if consecutive readings never fall within a specified tolerance, if desired.
- 6. In never drops below, enter the desired tolerance for Fail if LLS result. If after completing all the desired Sense readings consecutive values never fall within the entered tolerance, the liquid level sensing fails and an error displays.

#### 9.4.5.3 Configuring Sensitivity Settings (FX, NX-S8 only)

Liquid level is determined using specially designed liquid level sensing (LLS) tips that detect a shift in the capacitance of the probe. The tip moves to a specified height within the well and then slowly moves down into the well. When the tip contacts liquid, there is a large change in capacitance detected. The liquid level is sensed by determining the height at which this change in capacitance occurs.

Sensitivity Settings is used to configure how sensitive the probes are to changes in capacitance. A larger sensitivity value detects smaller capacitance changes more easily.

1. If Sensitivity Settings is not displayed, click on the sentence summary second from the bottom to expand the Sensitivity Settings configuration (Figure 9-10).

Technique Editor - Sp	an-8				
Pipetting Template:	an-8				
<u>G</u> eneral Calibration	Aspirate Liquid Level Sens	Dispense   sing   Li	<u>M</u> ix   iquid <u>T</u> ype		
♥ Start at 5 mm from ♥ Sense 1 time.	<ul> <li>∇ Start at 5 mm from the Top. Sense at 20% speed.</li> <li>∇ Sense 1 time.</li> </ul>				
<ul> <li>Sensitivity Settings</li> <li>Inherit from Liquid Type</li> </ul>					
Inherit from Labware     Use this Setting: 2500					
2500 4000					
♥ On error, retry 0 times. Then, prompt the user. Sensitivity Settings					
Arrow to collapse/expand       Sensitivity Settings         Sensitivity Settings       affects the required         capacitance change to sense liquid.       sense liquid.					
		ОК	Cancel		

Figure 9-10. Technique Editor - Sensitivity Settings on Liquid Level Sensing tab

- 2. Select the **Sensitivity** of the liquid detection. **Sensitivity** is the magnitude of change required for LLS to detect the liquid. A larger sensitivity value detects smaller capacitance changes more easily. Options include:
  - Inherit from Liquid Type uses sensitivity specified for the liquid in Liquid Type Editor; refer to Chapter 8, <u>Understanding and Creating</u> <u>Liquid Types</u>.
  - Inherit from Labware uses sensitivity specified for the labware in Labware Type Editor; refer to Section 7.3, <u>Defining Labware Types</u>.
  - Use this Setting uses the setting specified in the Technique Editor; specify the **Sensitivity** by entering the value or dragging the bar to the desired values.

Note: The default sensitivity setting is Inherit from Labware.

#### 9.4.5.4 Configuring Error Handling Settings

Error Handling Settings specify the action to take if liquid level sensing fails.

1. If Error Handling Settings is not displayed, click on the last sentence summary to expand the Error Handling Settings configuration (Figure 9-11).

Technique Editor - Default				
Pipetting Template:	efault Template			
<u>G</u> eneral <u>C</u> alibration	Aspirate <u>D</u> ispense <u>M</u> ix Liquid Level Sensing Liquid <u>T</u> ype			
♥ Start at 0 mm from	the Top. Sense at 20% speed.			
<b>Error Handling Settings</b> Error Handling Settings specifies an action to perform if liquid level sensing fails.				
▲ Error Handling Set	tings			
When a sensing erro Retry 0	r occurs: times.			
If the sensing error ha	as not been resolved by retrying:			
Pipette from the second sec	he bottom of the well instead.			
O Pipette from a	bove the well instead.			
<ul> <li>Prompt the user for further input.</li> </ul>				
Arrow to collapse/expand Error Handling Settings				
	OK Cancel			

Figure 9-11. Technique Editor - Error Handling Settings on Liquid Level Sensing tab

2. In **Retry**, enter the number of times to try to detect the liquid level when a sensing error occurs.

> FX, NX-S8 — Retry only applies to these instruments.

- 3. In If the sensing error has not been resolved by retrying, select the action to perform:
  - Pipette from the bottom of the well instead measures the pipetting height from the bottom of the well instead of from liquid level

**Note:** When pipetting from the bottom of the well, the minimum pipetting height as defined on the **Calibration** tab for the technique (refer to Section 9.5, *Calibrating Techniques*) is used.

• Pipette from the top of the well instead — measures the pipetting height from the top of the well instead of from liquid level

**Note:** When pipetting from the top of the well, the operation is performed at a height 1.0 millimeters (mm) above the top of the well.

• Prompt the user for further input — prompts the user for further instruction (Figure 9-12) and provides options to pipette from the top of the well (Pipette Air), pipette from the bottom of the well (Pipette from Bottom), retry (Seek Again), snap a continuation (Snap), or stop the method (Abort).

Biomek® Software	
There is insufficient volume in well 1 for this operation. Check mandril #1 before	continuina
	containing.
Pipette Air Pipette from Bottom Seek Again At	port
	11/3/2003 2:39:15 PM
	11/3/2003 2:39:15 PM

Figure 9-12. Liquid Level Sensing Error Handling prompt for Biomek 3000

#### 9.4.6 Setting Clot Detection Values (NX-S8 only)

The settings for clot detection allow the technique to perform clot detection during an aspirate. Refer to the *Biomek*® *NX Span-8 Laboratory Automation Workstation User's Manual,* Chapter 2.3.2, <u>Clot Detection</u>, for information on how clot detection works with the Biomek NX Span-8 instrument and Biomek Software.

The Clot Detection tab consists of three collapsible sections to configure various aspects of using clot detection. Each section is expanded by clicking the down arrow or sentence summary for that section and may then be configured. Only one section may be expanded at a time. When collapsed, each sections displays a sentence summary describing the configuration for that section.

The three sections are:

- General Settings specify the detection height and speed.
- Sensitivity Settings select to inherit sensitivity from labware type, liquid type, or define a custom setting
- Error Handling Settings specify the action to take if a clot is detected.

To configure Clot Detection:

- 1. From the Project menu, choose **Technique Browser**. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.

#### 3. Choose Edit.

OR

💁 Edit .

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

Technique Editor - Span-8		
Pipetting Template: Span-8		
Count   Anista   Discours		Caliburation
Liquid Level Sensing Clot Detection	Piercing	
A General Settings		(
	r.	
Move to 0 mm from the Liquid	before dete	ting.
Sense at 30 % speed.		
▽ Inherit sensitivity from labware.		
<ul> <li>On error, retry 1 time. Then, prompt the use</li> </ul>	r.	
	ок	Cancel
L		

4. In Technique Editor, select the **Clot Detection** tab (Figure 9-13)

Figure 9-13. Technique Editor — General Settings on Clot Detection tab

#### 9.4.6.1 Configuring General Settings

General Settings is used to configure the height of the probe within the well and the pod speed in the Z axis during clot detection operations.

- 1. If General Settings is not displayed, click on the topmost sentence summary to expand the General Settings configuration (Figure 9-13).
- 2. Enter the height in millimeters (mm) to Move to and detect from.

**Note:** The suggested starting height for **Move to** should be between 3.175 and 4.7625 mm. The height may be adjusted, but starting with these heights is suggested.

- 3. Select the from the start location: Liquid, Top, or Bottom. The probe moves to the specified height measured from the liquid, top, or bottom of the well before detecting.
- 4. Enter the **Sense at** speed as a percentage of the maximum speed of the pod.

Note: The speed affects the Z-axis speed only.

#### 9.4.6.2 Configuring Sensitivity Settings

Sensitivity Settings is used to configure how sensitive the probes are to changes in capacitance. A clot is detected by determining the change in capacitance as the tip moves up after aspiration. Clot detection is determined using conductive tips (identified as LLS tips in the Tip Type Editor), fixed tips, or SeptaFluted tips that detect a shift in the capacitance of the probe.

The tip moves to a specified height within the well and then slowly moves down into the well. When the tip aspirates and rises from the liquid and if there is a large change in capacitance, no clot is detected. If there is no change or a small change in capacitance, a clot is detected.

To configure Sensitivity Settings:

1. If Sensitivity Settings is not displayed, click on the sentence summary second from the bottom to expand the Sensitivity Settings configuration (Figure 9-14).

Technique Editor - Span-8		
Pipetting Template: Span-8		•
General Aspirate Dispense Liquid Level Sensing Clot Detection	<u>M</u> ix Piercing	<u>C</u> alibration Liquid <u>T</u> ype
$\nabla$ Start at 0 mm from the Liquid. Sense at 30%	speed.	
A Sensitivity Settings		
C Inherit from Liquid Type		
Inherit from Labware		
🔿 Use this Setting: 3000		
more sensitive		less sensitive
1000		3500
$\nabla$ On error, retry 1 time. Then, prompt the use	er.	
	ОК	Cancel



- 2. Select the **Sensitivity** of the clot detection. Sensitivity is the magnitude of change required for the conductive tips to detect a clot. A smaller sensitivity value detects smaller capacitance changes more easily. Options include:
  - Inherit from Liquid Type (default sensitivity setting) uses sensitivity specified for the liquid in Liquid Type Editor; refer to Chapter 8, <u>Understanding and Creating Liquid Types</u>.
  - Inherit from Labware uses sensitivity specified for the labware in Labware Type Editor; refer to Section 7.3, <u>Defining Labware Types</u>.
  - Use this Setting uses the setting specified in the Technique Editor; specify the **Sensitivity** by entering the value or dragging the bar to the desired values.

#### 9.4.6.3 Error Handling Settings

Error Handling Settings specify the action to take if a clot has been detected.

 If Error Handling Settings is not displayed, click on the last sentence summary to expand the Error Handling Settings configuration (Figure 9-11).

Technique Editor - Span-8					
Pipetting Template: Span-8		-			
	1				
<u>General</u> <u>A</u> spirate <u>Dispense</u>	<u>M</u> ix Diavaia a	<u>Calibration</u>			
Eiguid Level Sensing	Piercing	Liquia <u>T</u> ype			
abla Start at 0 mm from the Liquid. Sense at 309	6 speed.				
$\nabla$ Inherit sensitivity from labware.					
A Error Handling Settings					
When a potential clot is detected:					
Retry detection 1 times.					
If the clot detection has not been resolved by retrying:					
$\bigcirc$ Dispense the sample, aspirate the samp	ple again, then	retry detection.			
$\bigcirc$ Dispense the sample, then pipette air a	ind continue th	e method.			
Prompt the user for further input.					
Ignore the error and continue the meth	nod.				
	ОК	Cancel			

Figure 9-15. Technique Editor - Error Handling Settings on Clot Detection tab

- 2. In Retry detection, enter the number of times to retry the detection of a clot.
- 3. In If the clot detection has not been resolved by retrying, select the action to perform:
  - Dispense the sample, aspirate the same again, then retry detection.
  - Dispense the sample, then pipette air and continue the method.
  - Prompt the user for further input. Requests further instruction and provides the options Dispense, Aspirate Again, then Retry Detection; Ignore and Continue; Snap [a Continuation]; or Abort.
  - Ignore the error and continue the method. Ignores the error and continues the method.

**Note:** Ignoring the error and continuing the method when a clot exists may contaminate the deck.

#### 9.4.7 Setting Piercing Values (NX-S8 only)

Setting the piercing values allows the technique to pierce septum within specified parameters during an aspirate.

To configure Piercing:

- 1. From the Project menu, choose **Technique Browser**. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.

3. Choose Edit.

OR

<u>E</u>dit

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

4. In Technique Editor, select the **Piercing** tab (Figure 9-13)

Technique Editor - Span-8				
Pipetting Template: Span-8				
General     Aspirate     Dispense     Mix     Calibration       Liquid Level Sensing     Clot Detection     Piercing     Liquid Type				
When access to a well requires a piercing move:				
first move to 2 mm from the top of the well,				
then move to -5 mm from the top of the well				
at 20 % speed,				
then execute the specified pipetting template.				
Limit all Z moves inside a pierced well to 20 % speed.				
When leaving a well that required a piercing move:				
move to 2 mm from the top of the well.				
at 20 % speed.				
OK Cancel				

Figure 9-16. Technique Editor - Piercing tab

- 5. In When access to a well requires a piercing move: first move to, enter the distance in millimeters (mm) from the top of the well to move to before piercing.
- 6. In then move to, enter the distance in millimeters (mm) to then move to from the top of the well to pierce.
- 7. In **at**, enter the speed as a percentage of the maximum speed of the pod while accessing the well. The value must be between 0 and 100.

- 8. In Limit all Z moves inside a pierced well to, enter the speed of all Z-axis moves as a percentage of the maximum speed of the pod.
- 9. In When leaving a well that required a piercing move: move to, enter the distance in millimeters (mm) to move away from the top of the well that was pierced.
- 10. In at, enter the speed as a percentage of the maximum speed of the pod while leaving the well. The value must be between 0 and 100.

#### 9.4.8 Overriding Liquid Type Values

The settings configured on the Liquid Type tab override values set in the Liquid Type Editor (refer to Section 8.2, <u>Creating New Liquid Types</u>). Configure the liquid type override values so specific pipetting characteristics will result, such as the amount of air to use as a blowout, or speeds for aspirate and dispense.

Configuring the Liquid Type tab is optional. When overriding liquid type values, the settings entered in the Liquid Type tab are used instead of the settings from the particular liquid type. To use the settings from the liquid type, do not select Override Liquid Type Settings.

**Note:** Use the Liquid Type tab only when overriding the liquid type. Create additional liquid types when using common liquid type settings.

To set liquid type override values for this technique:

- 1. From the Project menu, choose **Technique Browser**. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.



3. Choose Edit.

OR

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

T	Clb		
Technique Edicor - De	arault	_	
Pipetting Template: De	sfault Template		•
General Calibration	Aspirate	Dispense   el Sensing	Mix   Liquid Type
🔽 Override Liquid Ty	pe Settings		
Aspirate		Dispense	
Trailing Air Gap:	1 μL	Delay: 0	ms
Delay: 0	ms	Speed: 60	μL/s
Speed: 60	μL/s	Cutoff Velocity: 15	0 µL/s
Blowout		Tip Touch	
Volume: 10	μL	Height: -1	mm
Delay: 0	ms	from Top	
Prewet		Angle: 90	
Overage: 0	μL	Speed: 50	%
Delay: 0	ms	Delay: 0	ms
		OK	Cancel

4. In Technique Editor, choose the Liquid Type tab (Figure 9-17).

Figure 9-17. Technique Editor — Liquid Type tab

5. Select **Override Liquid Type Setting** to activate the Liquid Type tab configuration.

**Note:** When overriding liquid type values, the settings entered in the Liquid Type tab are used instead of the settings from the particular liquid type. To use the settings from the liquid type, do not select Override Liquid Type Settings.

- 6. In Aspirate, enter the volume of the **Trailing Air Gap** in microliters (μL). After aspirating the liquid, a **Trailing Air Gap** is aspirated to prevent dripping.
- 7. In Aspirate, enter the desired **Delay** in milliseconds (ms). Delay specifies the amount of time after an aspirate operation before continuing.
- 8. In Aspirate, enter the desired **Speed** in microliters per second ( $\mu$ L/s). This indicates the speed along the D axis during aspirate.
- 9. In Blowout, enter the desired **Volume** of the leading air gap to blow out after dispensing the liquid.
- 10. In Blowout, enter the desired **Delay** in milliseconds (ms). Delay specifies the amount of time after a blowout before continuing.

- 11. In **Prewet**, enter the desired **Overage** in microliters. **Overage** is the amount of liquid to be aspirated over the amount specified in the step configuration.
- 12. In **Prewet**, enter the desired **Delay** in milliseconds (ms). **Delay** specifies the amount of time from when the **Prewet** volume is completely aspirated until it begins to dispense.
- 13. In **Dispense**, enter the desired **Delay** in milliseconds (ms). **Delay** specifies the amount of time after a dispense operation before continuing.
- 14. In **Dispense**, enter the desired **Speed** in microliters per second ( $\mu$ L/s). This indicates the speed along the D axis during dispense.
- 15. In Dispense, enter the desired **Cutoff Velocity**, in microliters per second ( $\mu L/s$ ).
  - FX, NX-S8 Cutoff Velocity applies to a Span-8 Pod only.

**Note:** The minimum and maximum cutoff velocity varies with the syringe type. The value entered must fall within the acceptable range for the syringe type. Refer to Section 9.4.8.1, *Syringe Cutoff Velocities (FX, NX-S8 only)*, for more information on the cutoff velocity.

16. In Tip Touch, enter the desired **Height**.

**Note:** Height, along with from and Angle, specify where along the well the Tip Touch occurs.

- 17. In Tip Touch, select the desired **from**: Top, Bottom, or Liquid. This indicates the location the Height is measured from.
- 18. In Tip Touch, enter the desired Angle

OR

Select the position along the edge on the graphic by positioning the smaller circle at the desired location. The Angle adjusts to the current location.

**Note:** The Angle specifies the location along the well wall where the Tip Touch occurs at the specified Height and from location.

- 19. In Tip Touch, enter the desired Delay.
- 20. In **Tip Touch**, enter the desired **Speed** as a percentage of the total pod speed. This indicates the pod speed along the X-, Y-, and Z-axes as the pod moves to the tip touch position.

#### 9.4.8.1 Syringe Cutoff Velocities (FX, NX-S8 only)

The speed per second at which fluid is dispensed prior to the dispense action stopping abruptly is called the Cutoff Velocity. If the Cutoff Velocity in the technique or liquid type governing the pipetting operation is too slow, droplets can be left at the end of tips after a dispense pipetting operation. Accurately setting the Cutoff Velocity causes the liquid to be ejected at a speed sufficient to prevent droplets from forming on the end of a tip.

The minimum and maximum Cutoff Velocity and the instruments on which they are supported are listed in Table 9-2. When changes to Cutoff Velocity are required, they must be made to the technique governing the pipetting operation in the Technique Editor, and to the parameters of the fluid used during the pipetting operation in the Liquid Type Editor.

The cutoff velocity should be adjusted in all Span-8 techniques for better pipetting accuracy every time syringes are changed from one type to another. The Cutoff Velocities displayed in Table 9-2 are intended as a starting point. The values provided should be experimented with to determine the most accurate Cutoff Velocity for a specific liquid handling operation.

Syringe Size	Minimum Cutoff Velocity	Maximum Cutoff Velocity	Instruments
100 µL	0.833 µL/second	45.0 µL/second	NX-S8
250 µL *	2.08 µL/second	112.5 µL/second	FX, NX-S8
500 µL**	4.17 $\mu$ L/ second	225 µL/second	FX, NX-S8
1 mL ***	8.33 $\mu$ L/ second	450 μL/second	FX, NX-S8
2.5 mL	20.8 µL/second	1125 µL/second	NX-S8
5 mL	41.7 µL/second	2250 µL/second	NX-S8
* D			

 Table 9-2.
 Span-8 Cutoff Velocities for Syringes for Span-8 Pods

\* Recommended cutoff velocity is 100 µL/second.

\*\* Recommended cutoff velocity is 150 µL/second.

\*\*\* Recommended cutoff velocity is 150 µL/second.

**Note:** The minimum Cutoff Velocity for each syringe is established using the equation:  $50 \times \text{Syringe size in } \mu \text{L} > /6000$ . The maximum Cutoff Velocity for each syringe is established using the equation:  $2700 \times \text{Syringe size in } \mu \text{L} > /6000$ .

To change techniques to conform to the **Cutoff Velocity** supported by the syringe size:

- From the Project menu, choose Technique Browser. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.

🔰 <u>E</u>dit

3. Choose **Edit**.

OR

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

- 4. In Technique Editor, choose the Liquid Type tab (Figure 9-17).
- 5. In Dispense, enter the desired **Cutoff Velocity**, in microliters per second (μL/s).
- 6. Choose **OK** to save the technique and exit the **Technique Editor**.
- 7. Choose Close to exit the Technique Browser.

**Note:** If the size of syringes installed in the Biomek instrument changes frequently, or if two different sizes of syringes are installed at the same time, create a duplicate set of the Span-8 techniques for each syringe size (refer to Section 9.9, *Modifying Saved Techniques*) and edit the Cutoff Velocity on the Liquid Type tab to support the syringes.

# 9.5 Calibrating Techniques

Calibrating techniques improves the accuracy of pipetting operations. All method steps involving aspirating, dispensing, and mixing volumes of liquids use calibration to make sure an accurate volume is aspirated and dispensed. Calibration is also necessary to adjust for liquid properties, such as viscosity.

During a method run, the Biomek instrument aspirates and dispenses liquid based on the results of the following equation:

Displaced Volume = Desired Volume \* Scaling Factor + Offset Volume

The properties in the equation are:

- Displaced Volume the amount physically aspirated and dispensed.
- Desired Volume the amount that is expected to be aspirated and dispensed.
- Scaling Factor the amount to multiply a given volume by to achieve the desired calibration.
- Offset Volume the amount of overage by a specific amount, such as  $2 \mu L$ .

The results of the equation are displayed in sentence form in the middle section of the Calibration tab (Figure 9-18). The displaced volume changes as adjustments are made to the Scaling Factor and Offset to indicate the result of the equation.

To calibrate techniques:

- From the Project menu, choose Technique Browser. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.

3. Choose **Edit**.

OR

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

4. In Technique Editor, select the Calibration tab.

5. If the Volume Calibration configuration is not displayed, click on Volume Calibration to expand the Volume Calibration configuration.

Technique Editor - Default				
Pipetting Template: Default Template				
<u>G</u> eneral	<u>A</u> spirate	<u>D</u> ispense	<u>M</u> ix	
<u>C</u> alibration	Liquid Leve	el Sensing	Liquid <u>T</u> ype	
A Volume Calibration It may be necessary to adjust the displacement volume to account for varying properties of liquids such as viscosity to accurately deliver a desired volume. Using the equation:				
Displaced Volume	e = (Desired Volume)	* (Scaling Factor) + (	Offset Volume)	
adjust the offset (a fixed volume) and the scaling factor (multiply, amplify, etc. the desired volume) to deliver the desired volume.				
Scaling Factor: 1 Offset: 0				
Using these parameters, the instrument will displace $10 \mu$ L to deliver the desired volume of $10 \mu$ L.				
♥ Volume Conditioning				
Minimum pipetting <u>h</u> eight: 0.2 mm above the well bottom.				
		OK	Cancel	

Figure 9-18. Technique Editor — Volume Calibration on Calibration Tab

6. Enter the **Scaling Factor** at which to calibrate. The sentence display adjusts accordingly.

**Note:** Scaling Factor should be greater than 0. A Scaling Factor of 1 and an Offset of 0 does not alter the volume manipulated during a pipetting operation.

- 7. Enter the desired **Offset**. The sentence display adjusts accordingly.
- 8. Verify the calibration equation by reading the sentence display. Adjust the Scaling Factor and Offset as necessary.
- 9. In Minimum pipetting height, enter the number of millimeters from the bottom of the labware that the tip is not allowed to go below.
- 10. Choose OK.
- 11. Choose Close to close Technique Browser.

**Note:** It is recommended that techniques be tested after calibrating to verify the desired results are achieved.

# 9.6 Conditioning Techniques for Multiple Dispense (FX, NX-S8 only)

Conditioning is a specified volume of liquid to dispense when performing multiple dispense operations from a single aspirate operation, and is used to enhance precision when switching from aspirate mode to dispense mode. If a multiple dispense technique is not properly conditioned, the first dispense may be greater than the volume specified in the step.

When using a technique that utilizes volume conditioning, the Biomek instrument aspirates and dispenses liquid according to Figure 9-19:

**Note:** Volume conditioning is used by a technique only if the associated pipetting template is designed to use volume conditioning. Refer to Chapter 10, <u>Using the</u> <u>Pipetting Template Editor</u>, for more information on pipetting templates and the Pipetting Template Editor. The Span-8 MultiDispense pipetting template is the only standard template designed to use volume conditioning.



Figure 9-19. Volume Conditioning identification

The volumes displayed in Figure 9-19 are:

- System Trailing Air Gap air gap between the system fluid and volume aspirated. The system trailing air gap is configured in Hardware Setup.
  - FX refer to the Biomek® FX Laboratory Automation Workstation User's Manual, Chapter 4.4.6, <u>Setting Span-8 Pod Properties</u> or
  - NX-S8 refer to the Biomek® NX Span-8 Laboratory Automation Workstation User's Manual, Chapter 3.3.6, <u>Setting Span-8 Pod</u> <u>Properties</u>

**Note:** The System Trailing Air Gap is initially configured by a Beckman Coulter Service Engineer and should not be modified without specific instructions from a Beckman Coulter Service Engineer or Technical Support.

• Leading Conditioning Excess — volume of source aspirated in excess of the sample Volume to Dispense that serves as a buffer between the system fluid and the liquid dispensed to protect from sample dilution. The Leading Conditioning Excess is purged to waste after completing all dispense operations.

**Note:** The Span-8 MultiDispense pipetting template automatically includes an excess of 15 percent of the volume aspirated.

- Volume to Dispense the total volume of source aspirated for all dispense operations. For example, if dispensing 10 μL across 12 wells of a microplate, the volume aspirated is 120 μL.
- Trailing Conditioning Volume an extra volume of source is aspirated a specified Number of Trailing Volume Sections times following the Volume Aspirated. Each trailing volume section is dispensed back into the source well separately prior to leaving the well to condition the dispense operation for more precise pipetting. For example, in Figure 9-19, each of the three bottom sections is dispensed back into the source well with three separate dispense actions.

To condition techniques:

- From the Project menu, choose Technique Browser. Technique Browser appears.
- 2. In Technique Browser, select the technique to modify.



OR

3. Choose Edit.

Double-click on the technique.

OR

Right-click in the Techniques View and choose **Edit** from the menu. The **General** tab is displayed by default (Figure 9-4).

4. In Technique Editor, select the Calibration tab.

 If the Volume Conditioning configuration is not displayed, click on Volume Conditioning to expand the Volume Conditioning configuration (Figure 9-18).

Technique Editor - Default
Pipetting Template: Default Template
<u>G</u> eneral <u>A</u> spirate <u>D</u> ispense <u>M</u> ix
Calibration Liquid Level Sensing Liquid Type
⊽ Volume C <u>a</u> libration
A Volume Conditioning
It is often necessary to use conditioning sections to provide improved pipetting precision. For example, conditioning should be used when using a single aspirate for multiple dispenses.
Leading Conditioning <u>E</u> xcess: 0 µL
Irailing Conditioning Volume: 0 μL
Number of Trailing Volume Sections: 1
The Pipetting Template used by this technique must be designed to use the conditioning volumes. Consult the User's Manual for more information on working with Pipetting Templates.
Minimum pipetting height: 0.2 mm above the well bottom.
OK Cancel

Figure 9-20. Technique Editor — Volume Conditioning on Calibration Tab

6. Enter the volume of **Leading Conditioning Excess** to aspirate as an additional buffer in excess of that aspirated automatically as part of the pipetting template between the system fluid and the source liquid.

**Note:** The **Span-8** MultiDispense pipetting template automatically includes an excess of 15 percent of the volume aspirated. The **Leading Conditioning Excess** is the volume to aspirate in addition to the 15 percent specified in the pipetting template for the buffer between the system fluid and source.

- 7. Enter the **Trailing Conditioning Volume** to aspirate for each trailing volume section.
- 8. Enter the Number of Trailing Volume Sections to aspirate.

**Note:** After aspirating the volume of source to dispense, an additional volume of source equal to the Number of Trailing Volume Sections times the Trailing Conditioning Volume is aspirated. Prior to leaving the well, each trailing volume section of Trailing Conditioning Volume is dispensed back into the source well in separate dispense actions equal to the specified Number of Trailing Volume Sections.

# 9.7 Creating Technique Groups

A Group is a user-created subset of techniques and can be used to organize techniques by their properties via the **Technique Editor** using Groups. For examples, all techniques for use with a Span-8 Pod on a Biomek FX instrument could be placed in a group called Span-8.

Techniques may be in more than one group, if desired. For example, another group could have all techniques for use with the Multichannel Pod on a Biomek FX instrument. Any techniques that can use both the Span-8 Pod and the Multichannel Pod would appear in both groups.

To create a new group:

1. From the Project menu, choose **Technique Browser**. Technique Browser appears.

🔊 New <u>G</u>roup

2. Choose New Group.

OR

Right-click under Groups and select **New Group** from the menu that appears. A new group is added under Groups and a checkbox appears to the left of each technique (Figure 9-21).

Image: Second	Parte     Edit       Name       Image: Span-8 Low 80       Image: Span-8 MultiDispense       Image: Span-8 Viscous       Image: Span-8 Viscous	Properties Tips P30 Fixed Fixed Fixed Fixed	* * * *	Pod * Pod8Span Pod8Span	Labw BCUp *	liquidt * *	maxi 2 80	rank 50 59		J J
Groups N (All) New Group	Name 고 말 Low Volume 384 Wet 과 말 Span-8 Low 80 고 말 Span-8 MultiDispense 고 말 Span-8 Viscous 기 말 Span-8 고 말 Washing	Tips P30 Fixed Fixed Fixed Fixed	Head * * *	Pod * Pod8Span Pod8Span	Labw BCUp *	liquidt * *	maxi 2 80	rank 50 59		ļ
(All)	모델 Low Volume 384 Wet 모델 Span-8 Low 80 모델 Span-8 MultiDispense 모델 Span-8 Viscous 모델 Span-8 모델 Washing	P30 Fixed Fixed Fixed Fixed	* * *	* Pod8Span Pod8Span	BCUp * *	*	2 80	50 59	0.5 *	,
New Group	9 (화 Span-8 Low 80 호텔 Span-8 MultiDispense 호텔 Span-8 Viscous 호텔 Span-8 호텔 Washing	Fixed Fixed Fixed Fixed	* *	Pod8Span Pod8Span	*	*	80	59	*	
2 2 2	모텔 Span-8 MultiDispense 고릴 Span-8 Viscous 고밀 Span-8 고릴 Washing	Fixed Fixed Fixed	*	Pod8Span	*	*				,
	모를 Span-8 Viscous 고를 Span-8 고를 Washing	Fixed Fixed	*	n log			*	60	*	,
	모 🕼 Span-8 코 🕼 Washing	Fixed		Pod85pan	*	Serum	*	59	*	>
	🗹 🕼 Washing		*	Pod8Span	*	*	*	59	80	>
		Fixed	*	Pod96, Pod8Span	Wash	*	1000	40	*	,
	🔄 🕼 1536 Plate	P30	ATTIL	Pod96	Grein	*	15	45	*	>
	🗖 🗐 Low Volume 384 Plate	P30	ATTIL	Pod96	AB38	*	15	50	*	>
	🗖 🗐 Low-vol Reservoir	*	*	Pod96	BCDe	*	25	58	*	>
	Low-Volume	*	*	Pod96	AB38	*	25	59	*	>
	🔲 🕼 Reservoir (ETOH)	*	*	Pod96	BCDe	Ethanol	*	61	*	,
	Reservoir	*	*	Pod96	BCDe	*	*	60	15	>
	Standard Standard	*	*	Pod96	AB38	*	*	60	15	,
	🗖 🕼 Default	*	*	PodB3K, Pod96	*	*	*	99	*	,
Select techni by selecting	iques to include in checkbox.	n group	]							Þ

Figure 9-21. Creating a new group in the Technique Browser

3. Enter a name for the new group.

4. In the list of techniques, select the checkbox to the left of each technique to include that technique in the new group.

**Note:** Techniques can be sorted by any of the properties by clicking on the column headings. Selecting a heading once sorts the column in ascending order; clicking on the same heading a second time sorts that column in descending order.

💊 Edit Group

5. Choose **Edit** to exit group editing mode. The checkboxes disappear and only the methods that were selected remain in the group.

**Note:** The status bar in the **Technique Browser** displays how many techniques are in the selected group and how many techniques there are in the project.

Technique Browser									
🖉 New <u>G</u> roup 🖉 Re <u>m</u> ove G	Group 🛛 🔊 E <u>d</u> it Group 😭 Rei	n <u>a</u> me Group	Close						
🛛 🖉 New 💐 Remove 🛛 🛱 🤇	Copy 🛍 Paste   🔊 Edit 🖆	<sup>I</sup> Pr <u>o</u> perties							
Groups	Name	Tips	Head Poo	ł	Labw	liquidt	maxi	rank	minim S
(AII) 御 <mark>5pan8</mark>	② Low Volume 384 Wet Plate ③ Span-8 Low 80 ③ Span-8 MultiDispense ③ Span-8 Viscous ③ Span-8 ③ Washing	P30 Fixed Fixed Fixed Fixed Fixed	* * Poo * Poo * Poo * Poo * Poo	485pan 485pan 485pan 485pan 496, Pod85pan	BCUp * * * Wash	* * Serum *	2 80 * * 1000	50 59 60 59 59 40	0.5 * * * * * 80 * * *
Status bar displ techniques in g of techniques in	lays number of roup and number n project.								
6 Techniques in Span® (12 Tetal)									
o recimiques in spano (15 rocal)	,								

Figure 9-22. New group created in the Technique Browser

#### 9.7.1 Renaming Groups

Groups may be renamed at any time, if desired.

To rename a group:

1. From the Project menu, choose **Technique Browser**. Technique Browser appears.

😭 Ren<u>a</u>me Group

#### 2. Choose Rename Group.

OR

Right-click under Groups and select **Rename Group** from the menu that appears.

3. Enter the desired new name for the group and press **Enter**. The group is renamed.

#### 9.7.2 Adding and Removing Techniques from Groups

Techniques may be added to or removed from a group at any time using the Edit Group command.

**Note:** Techniques may also be added to a group when they are created by selecting the groups to which the technique is to be added when configuring technique properties (refer to Section 9.3, *Creating New Techniques*).

To edit a group:

1. From the Project menu, choose **Technique Browser**. Technique Browser appears.



2. Choose Edit Group.

OR

Right-click under **Groups** and select **Edit Group** from the menu that appears. A checkbox appears to the left of each technique (Figure 9-21).

Technique Browser	Sorts techniques by properties by clicking on column headers.								
j <b>&amp;/New </b> ∰ <u>R</u> emove   ⊟≙ :	Copy 📴 Paste   🧏 Edit 🗃	* Properties	5	-		7			
Groups	Name	Tips	Head	Pod	Labw	liquidt	maxi	rank	minim
(All)	Low Volume 384 Wet	P30	*	*	BCUp	*	2	50	0.5
B New Group	🗹 🕼 Span-8 Low 80	Fixed	*	Pod8Span	*	*	80	59	*
	🗹 🗐 Span-8 MultiDispense	Fixed	*	Pod8Span	*	*	*	60	*
	🗹 🕼 Span-8 Viscous	Fixed	*	Pod8Span	*	Serum	*	59	*
	🗹 🕼 Span-8	Fixed	*	Pod8Span	*	*	*	59	80
	🗹 🕼 Washing	Fixed	*	Pod96, Pod8	Span Wash	*	1000	40	*
	1536 Plate	P30	ATTIL	Pod96	Grein	*	15	45	*
	Low Volume 384 Plate	P30	ATTIL	Pod96	AB38	*	15	50	*
	Low-vol Reservoir	*	*	Pod96	BCDe	*	25	58	*
	Low-Volume	*	*	Pod96	AB38	*	25	59	*
	Reservoir (ETOH)	*	*	Pod96	BCDe	Ethanol	*	61	*
	Reservoir	*	*	Pod96	BCDe	*	*	60	15
	Standard	*	*	Pod96	AB38	*	*	60	15
	Default	*	*	PodB3K, Pod	196 *	*	*	99	*
			1						
Select tech	nniques to include in	n group							
by selectir	ng checkhox	- 1							
by selectif	ig eneekoox.		]						
	<pre>I</pre>								
14 Techniques in New Group (14	Total)								

Figure 9-23. Creating a new group in the Technique Browser

3. In the list of techniques, select the checkbox to the left of each technique to include in the new group, or deselect the checkbox to the left of each technique to remove that technique from the group.

**Note:** Techniques can be sorted by any of the properties by clicking on the column headings. Selecting a heading once sorts the column in ascending order; clicking on the same heading a second time sorts that column in descending order.

**Note:** Right-clicking on the technique in the techniques list and selecting **Remove** removes the technique from the project, not the group.



4. Choose **Edit Group** to exit group editing mode. The checkboxes disappear and only the methods that were selected remain in the group.

**Note:** The status bar in the Technique Browser displays how many techniques are in the selected group and how many techniques there are in the project.

#### 9.7.3 Removing Groups

When a group is no longer needed, it can be removed from the Technique Browser.

To remove a group:

- 1. From the Project menu, choose **Technique Browser**. Technique Browser appears.
- 2. Under Groups, select the desired group to remove.

🔊 Remove Group

#### 3. Choose **Remove Group**.

OR

Right-click under Groups and select **Remove Group** from the menu that appears. The group is removed from the list under Groups.

# 9.8 Removing Techniques

To prevent Biomek Software from choosing techniques during a method run that may have become obsolete, remove the techniques from the **Technique Editor**.

When a technique is removed, it is no longer available in method building. However, the latest checked-in revision of the technique remains in the project file and may be restored at any time (refer to Section 6.8.1, *<u>Restoring Deleted Project Items</u>*).

To remove techniques:

1. In Technique Browser, select the technique.



2. Choose Remove.

OR

Right-click and choose **Remove** from the menu. A Confirm prompt appears (Figure 9-24).

Confirm	×
2	Are you sure you wish to remove the technique "Technique 2" from your workspace?
	<u>Y</u> es <u>N</u> o

Figure 9-24. Remove technique confirmation prompt

3. Choose **Yes** to remove the technique.

# 9.9 Modifying Saved Techniques

When a method uses pipetting requirements that are slightly different from the requirements of previously created techniques, modify the technique that most closely resembles the new pipetting requirements by copying and pasting.

To copy and paste techniques:

1. In Technique Browser, select the technique to copy.



2. Choose **Copy**.

OR

Right-click and choose **Copy** from the menu.



3. Choose **Paste**. The copy appears with the name "Copy of (Technique)".

OR

Right-click and choose Paste from the menu.

4. Select the copied technique.

5. Choose **Properties**.

Properties

#### OR

Right-click and choose **Properties** from the menu.

- 6. Enter a new name for the technique, if desired.
- 7. Make any changes to the technique (refer to Section 9.4, <u>Setting Technique</u> <u>Values</u>).
- 8. Choose OK.

## 9.10 Selecting and Modifying Techniques Manually in a Method

Automatic selection of techniques is turned on by default; however, there may be instances when manual selection of techniques is desired. When Auto-Select is unchecked, manual selection of techniques is possible in the source and destination configuration of a pipetting step, such as Transfer or Combine. Biomek Software shows all techniques that match the properties of the step configuration (pod, tips, labware type, liquid type, volume) in a drop-down list in the Technique field (Figure 9-25).



Figure 9-25. Customize techniques or manually select techniques from a list in source or destination configurations of a pipetting step

Techniques used in methods are normally auto-selected by Biomek Software; however, circumstances may require modifications to techniques due to the current setup or liquid type. Many pipetting steps provide access to the **Technique Editor** during method development. Customize..

# 9.10.1 Modifying a Technique Through a Method Step

Custom techniques created within a method are saved within the current method only and are accessible only in the pipetting operation for which the technique was created. The technique can be saved for global use after configuration, if desired.

When modifying techniques or creating new techniques within a method, only the parameters for the specific operation, along with Liquid Type, Liquid Level Detection, and Calibration settings, may be modified.

To modify a technique within a step or method:

- 1. Select the desired step in the method.
- 2. Select the desired source or destination.

3. Choose **Customize** (Figure 9-26). Technique Editor - [Custom] appears (Figure 9-27).



Figure 9-26. Technique selection within a method

4. Customize the technique as needed (refer to Section 9.4, <u>Setting Technique</u> <u>Values</u>).

**Note:** Technique Properties, such as Labware Type, Liquid Type, and Tips, cannot be configured for the custom technique when creating the technique within a method. The technique uses the known properties of the specific operation being customized. Technique Properties are configured upon saving the technique for global use (refer to Section 9.10.2, *Saving Custom Techniques*).

5. Choose **OK**. [Custom] replaces the technique name.

Technique Editor - [Custom]
Pipetting Template: Default Template
General Dispense Calibration Liquid Level Sensing Liquid Type
Move within the well at 100 % speed.
Dispense at 0 mm from the Bottom
Follow liquid level when aspirating or dispensing liquid
Touch tips on the sides of the wells
Blowout all leading air gaps
Mix after dispensing liquid
Mig 10 μL 1 time.
Aspirate at 0 mm from the Bottom $\checkmark$ at 100 $\mu$ L/s.
Dispense at 0 mm from the Bottom $\checkmark$ at 100 $\mu$ L/s.
OK Cancel

Figure 9-27. Dispense Tab appears when Customize is selected from destination

Save As...

#### 9.10.2 Saving Custom Techniques

A custom technique is saved within the method in which it was created; however, any custom technique can be saved for global use.

To save a custom technique for global use:

1. Choose **Save As** in the step configuration (Figure 9-28). Technique Properties appears.



Figure 9-28. Save a custom technique within a step

- 2. Enter the **Technique Name**, then select the properties desired for the technique (refer to Section 9.3, *Creating New Techniques*).
- 3. Choose **OK**. The new technique name appears in **Technique**.

# Using the Pipetting Template Editor

### 10.1 Overview

Pipetting operations within steps of a method are configured using techniques. Techniques are associated with a pipetting template that controls how the instrument performs pipetting operations for aspirating, dispensing, and mixing liquids. While using the Pipetting Template Editor is not necessary to successfully operate the instrument, it does present the opportunity for finer control over pipetting operations in order to optimize performance for specific applications.

A pipetting template is not used unless it is associated with a technique. Likewise, a technique is useless without the pipetting template that controls it. Biomek Software contains several default pipetting templates by which all predefined techniques are controlled.

Pipetting templates can be modified, or new templates created, using the Pipetting Template Editor. Templates are created by configuring a series of steps that dictate the actions the instrument makes, similar to building a method in the Main Editor. Template steps are configured using context variables; however, only the template steps that are applicable to the instrument and configuration appear. A context variable retrieves values from the technique or liquid type used in the pipetting operation.

Pipetting templates, along with information about liquid types; labware and tip types; well patterns; and pipetting techniques, are stored as part of a project file. Project files store a history of all changes, additions, and deletions of items from the project file. Refer to Chapter 6, <u>Understanding and Using Project Files</u>, for more information on project files.

**Note:** When Beckman Coulter Accounts & Permissions is enabled, only users with the **Develop Projects** permission assigned may open, edit, create, and delete pipetting templates (refer to Chapter 2, *Using Accounts & Permissions*).

# **10.2 Creating and Modifying Pipetting Templates**

Creating and modifying pipetting templates is achieved with the Pipetting Template Editor.

To open the Pipetting Template Editor:

Choose **Project>Pipetting Template Editor**. The Pipetting Template Editor appears (Figure 10-1).

- **FX**, **NX-S8** Bulk Dispense appears *only* with these instruments.
- > **NX-S8** Clot Detection appears *only* with this instrument.

Pipetting templates are created in one of the following ways:

- Create a blank template and insert template steps
- Copy and paste an existing template and modify as needed

**Note:** The most expedient way to create pipetting templates is to copy and paste the template that resembles the desired operations most closely (refer to Section 10.2.3, *Modifying Existing Pipetting Templates*).

Dis akkis <del>- T</del>	analata Cd	itee		_		_	_	
Pipetting T	emplate Edi	itor Ba	rêj.	r©1	<b>N</b> -	_	_	
New	Remove	Сору	Paste	u≡ Rename	<b>بدر</b> Variables			
Comment If Comment If Clot Detection Clot Detection Aspirate Air Aspirate	Homore Dispense → Moving Dispense → Mix → Prewet ↓ Tip Touch Axes Move Pause	Pipetting Pipetting Default Aspire200 M Disper No Aspire200 M Disper D	I Template: Template: Template ite spirate =C ix ove to 0 cm, ewet spirate =C p Touch ove to 0 cm, spirate =C ix p Touch ove to 0 cm, spense Air G ove to 0 cm, spense Air G d spirate =C ix p Touch ove to 0 cm, spense Air G d	SlowoutVolut $0^{\circ}$ , =C_He Volume µL al $0^{\circ}$ , -1 mm ap at =C_S $0^{\circ}$ , -2 He Volume µL a $0^{\circ}$ , -1 mm ap at =C_S BlowoutVolut $0^{\circ}$ , -1 mm ap at =C_S	me μL for Air Ga eight mm : =CAspirate: pVolume μL for ipeed eight mm it =CDispense ipeed me μL for Air Ga	p at =C 5peed Air Gap at =Speed	Speed =C_Speed	If =C_Blowout is True, aspirate an air gap with volume =C_BlowoutVolume µL at speed =C_Speed µL/s then delay =C_BlowoutDelay ms Template Steps Configuration Context variables are inserted in fields provided for template step configuration.
Dispense Air Gap	Bulk Dispense	• •				1		
Pipett Femplat or modizemplate	T ing Ten te steps u fying pip es.	nplate used in petting	e Steps	g	Pipett Me Display the Asp	ing Te thod ' vs operate, I function	emplate View ations in Dispense,	OK Cancel Apply

Figure 10-1. Pipetting Template Editor

#### 10.2.1 Viewing Existing Pipetting Templates

Several default pipetting templates are included with Biomek Software. These templates have specific operations which they perform, and are associated with each technique to control liquid-handling operations. In order to understand the operations the pipetting template performs, view a pipetting template.

**Note:** Refer to Section 10.5, <u>*Configuring Pipetting Template Steps*</u>, for more information regarding each template step within the technique.

To view the pipetting template and related operations:

- 1. Choose **Project>Pipetting Template Editor**. The Pipetting Template Editor opens (Figure 10-1).
- 2. Select the desired **Pipetting Template**.
- 3. Click on the appropriate step to view the settings for the step.

#### **10.2.2 Creating a New Pipetting Template**

New pipetting templates are created using template steps that make up a pipetting procedure (refer to Section 10.5, *<u>Configuring Pipetting Template Steps</u>*). The template steps control the order of operations in aspirate, dispense, and mix operations.

**Note:** Before creating or customizing templates, be sure to understand where each of the context variables finds and retrieves the values used in the configured pipetting operations (refer to Section 10.3, *Understanding Context Variables*).

To create a new pipetting template:

1. Choose **Project>Pipetting Template Editor**. The Pipetting Template Editor opens (Figure 10-1).



2. Choose **New**. A blank template with Aspirate, Dispense and Mix steps appears (Figure 10-2).



Figure 10-2. New pipetting template

3. Insert and configure the desired template steps (refer to Section 10.5, <u>Configuring Pipetting Template Steps</u>).



4. Choose **Rename** to change the template name. Rename Pipetting Template appears (Figure 10-3).

Rename Pipetting Template						
Enter the new name for this template:						
Default Template	_					
OK Cancel						

Figure 10-3. Rename pipetting template

- 5. Enter the name of the new pipetting template and choose **OK**.
- 6. Choose **OK**. The template is saved and the Pipetting Template Editor closes.

OR

Choose **Apply**. The template is saved and the **Pipetting Template Editor** remains open.

OR

Choose **Cancel**. The template is not saved and the **Pipetting Template Editor** closes.

# 10.2.2.1 Associating a Pipetting Template with a Technique

A pipetting template is not used by the instrument unless it has been associated with a technique. Every time a pipetting operation occurs in a method that uses a technique with the template assigned, the operations are executed as configured in the pipetting template.

To associate a pipetting template with a technique:

1. Select **Project>Technique Browser**. The Technique Browser appears (Figure 10-4).

Technique Browser									
🔪 New Group 🏾 💓 Remove Group 🛛 🔪 Edit Gro	up 🛛 🚰 Rename Group	Clo	ise						
🕼 New 🔊 Remove 🗈 Copy 💼 Paste 💐	🛓 Edit 🛛 😭 Properties								
Groups Name	Labw	Tips	Pod	Head	Liquid	Maxi	Rank	Minim	Syrin
🖉 (All) 🖉 Washing	Wash	Fixed	Pod9	*	*	1000	40	*	*
🖉 Selected 🖉 1536 Plate	Grein	P30	Pod96	ATTIL	*	15	45	*	*
Low Volume 384	Plate AB38	P30	Pod96	ATTIL	*	15	50	*	*
Low-vol Reserve	oir BCDe	*	Pod96	*	*	25	58	*	*
🗐 Span-8 Viscous	*	Fixed	Pod8	*	Serum	*	59	*	*
🔊 Span-8 Low 80	*	Fixed	Pod8	*	*	80	59	*	*
Low-Volume	AB38	*	Pod96	*	*	25	59	*	*
🖓 Span-8	*	Fixed	Pod8	*	*	*	59	80	*
Reservoir	BCDe	*	Pod96	*	*	*	60	15	*
🖉 Standard	AB38	*	Pod96	*	*	*	60	15	*
Reservoir (ETO	H) BCDe	*	Pod96	*	Ethanol	*	61	*	*
🔊 Span-8 MultiDis	pense *	Fixed	Pod8	*	*	*	60	*	*
🖉 Default	*	*	Pod96	*	*	*	99	*	*
Copy of Low Vo	lume 38 AB38	P30	Pod96	ATTIL	*	15	50	*	*
14 Techniques									

Figure 10-4. Technique Browser



2. Select the desired technique and choose **Edit** from the toolbar.

OR

Double-click the desired technique on the right side of the Technique Browser.

OR

Right-click the desired technique and select **Edit** from the menu. The **Technique** Editor appears (Figure 10-5).

Technique Editor - Default
Pipetting Template: Default Template
Calibration     Pipetting Template associated with the technique is selected here.     uid Type       General     Mix
Aspirate Move within the well at 100% speed. Aspirate at 0 mm from the Bottom.
ä
<b>Dispense</b> Move within the well at <b>100%</b> speed. Dispense at <b>0 mm</b> from the <b>Bottom</b> .
Mix Move within the well at 100% speed. Aspirate and dispense at 0 mm from the Bottom.
đ
OK Cancel

Figure 10-5. Technique Editor

- 3. Select the appropriate **Pipetting Template** from the list.
- 4. Choose **OK**. The changes to the technique are saved and the **Technique Editor** closes.
### **10.2.3 Modifying Existing Pipetting Templates**

Copy and paste pipetting templates when making modifications to an existing template. Once the template is copied and pasted as a new template, any changes made do not affect the original.

**Note:** Pipetting templates can be exported to and imported from other projects. Refer to Section 6.10, *Importing and Exporting Project Files* for information about importing and exporting pipetting templates.

To copy and paste a pipetting template:

- 1. Choose **Project>Pipetting Template Editor** to open the Pipetting Template Editor.
- 2. In Pipetting Template, select the template to copy.
- 3. Choose Copy.
- Choose Paste to insert the copied template as a new template. The new template appears with the name "Copy of (Original Template's Name)".
- 5. Choose **Rename** to change the template name. Rename Pipetting Template appears.
- 6. Enter the name of the new pipetting template and choose **OK**.
- 7. Make changes to the pipetting template by inserting and configuring steps (refer to Section 10.5, *Configuring Pipetting Template Steps*).
- 8. Choose **OK**. The template is saved and the **Pipetting Template Editor** closes.

OR

Choose **Apply**. The template is saved and the **Pipetting Template Editor** remains open.

OR

Choose **Cancel**. The template is not saved and the **Pipetting Template Editor** closes.

## **10.3 Understanding Context Variables**

Pipetting templates make use of context variables to configure pipetting operations. Context variables are inserted into the template steps configuration and are used to retrieve values from the appropriate technique or liquid type in the Technique Editor or Liquid Type Editor. For example, C\_\_DispenseDelay retrieves the value from Dispense Delay of the appropriate liquid type in the Liquid Type Editor.

Using context variables gives pipetting templates greater versatility, as the values for context variables are determined by the liquid type or technique used for the specific pipetting operation.

**Note:** Variables configured within **Start**, Let, and **Worklist** can be placed into a pipetting template, but doing so may limit the usefulness of the pipetting template. When variables defined in **Start**, Let, and **Worklist** are used in a pipetting template, the template may only be used in methods with the same variable names defined. Be sure that these variables are defined in the method, or an error may occur during method execution.

#### **10.3.1** Inserting Variables and Expressions

Insert variables the appropriate fields of a pipetting template step, or by using the Variables button on the Pipetting Template Editor.

Expressions using a combination of variables, numerical constants, and mathematical operators, may also be entered in the fields of the appropriate template steps. For example, an expression could be entered for adding a volume or delay to a variable.

To insert variables in a template step:

1. Insert a template step from the Pipetting Template Step Palette into the desired operation: Aspirate, Dispense, or Mix.

OR

Select a template step already configured in the current template.

- 2. Choose the field in which to insert a variable.
- 3. Enter the variable name (Table 10-1).

**Note:** When entering context variables into a field, always use a double-underscore (\_\_\_) between C and the variable name.

OR



Choose Variables. Preprogrammed variables arranged by groups appear.

4. Choose the desired variable group. The group variables appear.

5. Choose the desired variable (Figure 10-6). The variables listed depend on the variable group selected.



Figure 10-6. Selecting variable from selected group

6. Repeat steps 2 to 5 until the template step is configured.

**Note:** Refer to Section 10.5, <u>Configuring Pipetting Template Steps</u>, for more information about inserting and configuring variables in template steps.

### 10.3.2 Defining Context Variables

To retrieve values, context variables reference fields in the **Technique Editor** and Liquid Type Editor. Table 10-1 describes context variables that can be used when creating or modifying a pipetting template. Figure 10-7 and Figure 10-8 identify the fields in the **Technique Editor** and Liquid Type Editor from which the context variables defined in Table 10-1 reference and retrieve values.

**Note:** Template steps default to the most appropriate variable; however, any variable may be used in any field.



Figure 10-7. Identifying the location of Technique Editor context variables



Figure 10-8. Identifying the location of Liquid Type Editor context variables

Variable Group	Variable	Description	Associates with:
General	CBlowout	Tells Biomek instrument to perform a blowout.	Blowout in technique
	C_BlowoutDelay	The delay in milliseconds after a blowout.	Blowout Delay in liquid type
	C_BlowoutVolume	The volume of the air gap to blow out.	Blowout Volume in liquid type or technique
	C_CalibrationOffset	Additional volume aspirated or dispensed to accurately deliver the desired volume.	Offset in technique calibration
	C_CalibrationSlope	Scaling factor used to accurately deliver the desired volume	Scaling Factor in technique calibration
	CConditioningExcess	Excess volume aspirated as a buffer between the system fluid and source.	Leading Conditioning Excess in technique calibration
	C_ConditioningSectionCount	Number of conditioning sections aspirated following the source.	Number of Trailing Volume Sections in technique calibration
	C_ConditioningSectionVolume	Volume of each conditioning section following the source	Trailing Conditioning Volume in technique calibration
	CPrewet	Tells Biomek instrument to perform a prewet cycle.	Prewet in technique
	C_PrewetDelay	The delay in milliseconds from the time the Prewet volume is completely aspirated to the time it begins to dispense.	Prewet Delay in liquid type or technique
	C_PrewetOverage	The volume to add for a prewet cycle.	Prewet Overage in liquid type
	C_Speed	The speed of the pod movement (X, Y, Z axes).	Pod Speed % in technique
	CTipTouch	Tells Biomek instrument to perform a tip touch operation.	Tip Touch in technique
	CTipTouchDelay	The delay in milliseconds at the side of wells.	Tip Touch Delay in liquid type or technique
	CTipTouchHeight	The height of the tip touch.	Tip Touch Height in liquid type or technique
	CTipTouchHeightFrom	Level — top, bottom, or liquid — from which CTipTouchHeight is measured.	Tip Touch Height, from in liquid type or technique

### Table 10-1. Context Variables

Variable Group	Variable	/ariable Description				
<b>General</b> (Continued)	CTipTouchTheta	Position along well edge tip touch occurs	Tip Touch Angle in liquid type or technique			
	CTipTouchSpeed	The speed of the pod movement.	Tip Touch Speed in technique			
	CVolume	The volume given in the pipetting operation (such as <b>Transfer</b> or <b>Combine</b> ).	Volume in step configuration			
	C_CDHeight	The height of the clot detection.	Clot Detection in technique.			
	CCDHeightFrom	Level—liquid, bottom, or top— from which CCDHeight is measured	Clot Detection in technique.			
	C_CDSpeed	The speed of the pod movement.	Clot Detection in technique.			
Aspirate	C_AspirateDelay	The delay in milliseconds after an aspirate cycle.	Aspirate Delay in liquid type			
	C_AspirateSpeed	The speed in $\mu$ /sec along the D axis for an aspirate cycle.	Aspirate Speed in liquid type			
	CTrailingAirGapVolume	The volume of the post-aspirate air gap.	Aspirate Trailing Air Gap in liquid type or technique			
	CTrailingAirGap	Tells Biomek instrument to aspirate a trailing air gap.	Trailing Air Gap in technique			
Dispense	C_DispenseDelay	The delay in milliseconds after a dispense operation.	Dispense Delay in liquid type or technique			
	CDispenseSpeed	The speed in $\mu$ l/sec of the D axis for a dispense operation.	Dispense Speed in liquid type or technique			
Aspirate/ Dispense	C_Height	The height of the aspirate or dispense.	Aspirate at x mm in technique			
	CHeightFrom	Level—top, bottom, or liquid— from which CHeight is measured.	Aspirate x mm from (Top, Bottom, Liquid) in technique			
Mix	CMix	Tells Biomek instrument to perform a mix operation in labware.	Mix Before Aspirate or Mix after Dispense in technique			
	CMixAspirateHeight	The height of the mix operation from the relative well location.	Aspirate at x mm in technique			
	CMixAspirateFrom	The location for the mix aspirate to occur.	In Mix Before Aspirate or Mix After Dispense and Mix, from (Top, Bottom, Liquid) in technique			
	CMixAspirateSpeed	The speed at which to aspirate for the mix cycle.	Aspirateat x% in technique			

## Table 10-1. Context Variables (Continued)

Variable Group	Variable	Description	Associates with:
Mix (Continued)	CMixCount	The number of cycles in a mix.	Number of times to Mix Before Aspirate or Mix After Dispense in technique
	C_MixDispenseHeight	The height of the mix dispense.	In Mix Before Aspirate or Mix After Dispense and Mix, from (Top, Bottom, Liquid) in technique
	CMixDispenseFrom	The height of the mix operation from the relative well location.	In Mix, from (Top, Bottom, Liquid) in technique
	CMixDispenseSpeed	The speed at which to dispense in a mix cycle.	Dispenseat x% in technique
	CMixVolume	The volume of the mix.	In Mix before Aspirate and Mix After Dispense, cycles to mix at specific volume in technique

Table 10-1.	Context	Variables	(Continued)
-------------	---------	-----------	-------------

# 10.4 Using Volume Conditioning in Pipetting Templates (FX, NX-S8 only)

Some techniques, such as the Span-8 MultiDispense technique, use volume conditioning to achieve more precise pipetting for multiple dispenses from a single aspirate. To use volume conditioning, the pipetting template associated with that technique must be designed to use volume conditioning.

**FX**— Volume conditioning is available only for a Span-8 pod.

**Note:** The Span-8 MultiDispense pipetting template is the only standard template that is designed to use volume conditioning. To associate the Span-8 MultiDispense pipetting template to a technique, refer to Section 10.2.2.1, <u>Associating a Pipetting</u> <u>Template with a Technique</u>.

There are three context variables that are important when using conditioning volumes:

- C\_\_ConditioningExcess the Leading Conditioning Excess from the Calibration tab of the Technique Editor
- C\_\_ConditioningSectionVolume the Trailing Conditioning Volume from the Calibration tab of the Technique Editor
- C\_\_ConditioningSectionCount the Number of Trailing Volume Sections from the Calibration tab of the Technique Editor

**Note:** Refer to the *Biomek Software User's Manual*, Section 9.6, <u>Conditioning</u> <u>Techniques for Multiple Dispense (FX, NX-S8 only)</u> for more information on conditioning volumes.

### 10.4.1 About ListMath

Since a Span-8 Pod is capable of aspirating and dispensing up to eight separate volumes, the pipetting template must accommodate this ability. ListMath is a special set of commands that can be entered in a pipetting template step configuration used to perform mathematical operations uniformly through a list.

ListMath takes two parameters and returns a list of values after performing the specified operation. A parameter may be a variable, a constant, or an array of constants or variables.

In the Span-8 MultiDispense pipetting template, ListMath is used to calculate the volume to aspirate by adding:

- The total volume needed to perform all the desired dispense operations, as entered in the step configuration.
- The Leading Conditioning Excess volume, plus 15 percent of the total volume needed to perform all the desired dispense operations.
- The Trailing Conditioning Volume times the Number of Trailing Volume Sections.

The result of the above operations is a volume for each probe.

Use ListMath like a variable when configuring a template step. Always start with an equal sign (=).

- To add two parameters, enter =ListMath.Add(Parameter1, Parameter2). Parameter2 is added to Parameter1.
- To subtract two parameters, enter =ListMath.Sub(Parameter1, Parameter2). Parameter2 is subtracted from Parameter1.
- To multiply two parameters, enter =ListMath.Mult(Parameter1, Parameter2). Parameter1 is multiplied by Parameter2.
- To divide two parameters, enter =ListMath.Div(Parameter1, Parameter2). Parameter1 is divided by Parameter2.

# **10.5 Configuring Pipetting Template Steps**

A pipetting template contains template steps that create the pipetting procedure. Steps are configured using context variables (Table 10-1) that retrieve information from the Technique Editor or Liquid Type Editor, or variables configured in a Start, Let, or Worklist configuration. The following conventions apply:

- All template steps, excluding Prewet, may be placed in any part of the pipetting process. Prewet is applicable only to Aspirate operations and is limited to the Aspirate process.
- Any variable defined in the system or method, including those defined in the Start, Let, or Worklist configurations, may be used to configure template steps.
- When using variables defined in Start, Let, and Worklist in a pipetting template, the template is useful only to those methods with the same variable names defined.
- The template steps are available for use in the Pipetting Template Editor only.
- When a step is inserted into the pipetting template, each field is configured with the most appropriate context variable by default. These default values may be changed to any desired value, variable, or expression.
- Hovering the mouse pointer over a template step button displays a tool tip with the name of the template step. Some steps also display information about the default configuration for that step.

#### 10.5.1 If, Comment, and Loop

The lf, Comment, and Loop steps are available for use in both methods and pipetting templates. When creating a pipetting template using these steps, configure the steps in the same manner as when configuring the step in the Method View.

**Note:** For more information, refer to Section 17.7, <u>If Step</u>; refer to Section 15.6, <u>Comment Step</u>; and refer to Section 16.9.1, <u>Configuring a Loop Step and Creating a</u> <u>New Variable</u>.

### 10.5.2 Clot Detection (NX-S8 only)

Clot Detection allows heights for clot detection to be configured. The template step sets the Move to height after aspiration; height measured from the liquid, bottom, or top of the well; and speed for clot detection. The heights are used to determine the difference in capacitance which indicates a clot exists.

Note: Clot Detection is usually used immediately after an Aspirate steplet.



To configure Clot Detection:

1. Insert **Clot Detection** in the desired pipetting operation of the template. The template step configuration appears (Figure 10-9)



Figure 10-9. Clot Detection step configuration

 In Move to, enter the variable or value to specify the desired height, in millimeters (mm), after aspiration. The default value for this field is the context variable C\_CDHeight. (Refer to Table 10-1 for a definition of this context variable.) 3. In from, select from where to measure the height in the well: Liquid, Bottom, or **Top**. The default value for this field is the context variable

**C\_CDHeightFrom**. (Refer to Table 10-1 for a definition of this variable.)

**Note:** Liquid measures from the top of the liquid, Bottom measures from the bottom of the well and Top measures from the top of the well.

**Note:** If Liquid is selected in C\_CDHeightFrom, the liquid level of the well must be known by the time clot detection is executed, either marked as known in Labware Properties or successfully liquid level sensed.

4. In **at speed**, enter the variable or value to specify the speed of the pod during clot detection as a percentage of the maximum speed of the pod. The default value for this field is the context variable **C\_CDSpeed**. (Refer to Table 10-1 for a definition of this variable.)

### 10.5.3 Aspirate Air Gap

Aspirate Air Gap aspirates an air gap at a specific volume and speed before liquid is aspirated. If a delay between the air gap aspirate and subsequent steps is desired, configure a delay into the template step.

Aspirate Air Gap can be used to aspirate a leading air gap before aspirating liquid, or a trailing air gap after aspirating liquid, depending on where it is inserted in the template.

To configure Aspirate Air Gap:



1. Insert **Aspirate Air Gap** in the desired pipetting operation of the template. The template step configuration appears (Figure 10-10).



Figure 10-10. Aspirate Air Gap step configuration

- Enter the variable to configure the lf condition. When the variable entered is True, an air gap is aspirated. The default value, =True, always aspirates an air gap.
- 3. In **volume**, enter the variable or value to specify the amount of air to aspirate for the air gap in microliters ( $\mu$ L). The default value is **0**  $\mu$ L.

- In speed, enter the variable or value to specify the speed of the pod in the D-axis in microliters per second (μl/sec) during the aspirate operation. The default value is the context variable C\_\_Speed. (Refer to Table 10-1 for a definition of this variable.)
- 5. In **delay**, enter the variable or value to specify the length of the delay in milliseconds (ms) after aspirating an air gap. The pod pauses for the specified duration before continuing with the next operation. The default value for this field is the context variable **C\_\_AspirateDelay**. (Refer to Table 10-1 for a definition of this variable.)

#### 10.5.4 Aspirate

Aspirate aspirates a specific volume at a speed specified in microliters per second ( $\mu$ l/sec). To pause for a specific amount of time after aspirating, add a delay to the Aspirate template step.

To configure Aspirate:



1. Insert **Aspirate** at the desired position. The template step configuration appears (Figure 10-11).



Figure 10-11. Aspirate step configuration

 In volume, enter the variable or value to specify the volume of liquid to aspirate in microliters (μL). The default value for this field is the context variable C\_\_Volume. (Refer to Table 10-1 for a definition of this variable.)

- In speed, enter the variable or value to specify the speed of the pod in the D-axis in microliters per second (μl/sec) during the aspirate operation. The default value for this field is the context variable C\_\_Speed. (Refer to Table 10-1 for a definition of this variable.)
- 4. In **delay**, enter the variable or value to specify the length of the delay in milliseconds (ms) after the aspirate operation. The pod pauses for the specified duration before continuing with the next operation. The default value for this field is the context variable C\_\_AspirateDelay. (Refer to Table 10-1 for a definition of this variable.)

#### 10.5.5 Moving Aspirate

Moving Aspirate aspirates while tips are moving within a well. The template step sets the initial position, final position, and a specific volume for the aspiration.

To configure Moving Aspirate:



1. Insert **Moving Aspirate** into the pipetting method. The template step configuration appears (Figure 10-12).

Pipetting T	emplate Ed	itor					
	$\times$	te (fil	r	<b>N</b> -			
New	Remove	Copy Paste	Rename	Variables			
Comment	Moving	Pipetting Template:			1	•	Set Initial Position     Initial
IF	Dispense Mix	<ul> <li>Aspirate</li> <li></li></ul>	for Air Gap at _Volume µL at ate	=CSpeed =CSpeed			Move to 0 & 🗶
Loop	Prewet	<ul> <li>End</li> <li>Dispense</li> <li>End</li> <li>Mix</li> </ul>	Dials of m	s for adjus	ting the angl from the	e	
Aspirate Air Gap	Tip Touch	End	in de	er of the vergrees ch rdingly.)	anges		from the well Ceapter, at 0 degrees
Aspirate	Axes Move	Movi	ng Asp	irate	]		and 0 mm
Moving Aspirate	Pause	Performs operation	s an aspin n while r	rate noving			Volume: =C_Volume µL
Dispense Aix Gap	Bulk						Post Delay: =C_AspirateDelay ms
Dispense	Слуренуе	1	MOY Co Moving A	ving Asp onfigura Aspirate s	tion tion	_	]
		C	configure	ed here.			OK Cancel Apply



2. Check Set Initial Position to set the start position for aspirate, if desired.

**Note:** Set Initial Position is used if an aspirate must begin at a specific well location on the labware. If this is not set, the pod moves from its current location. Set Initial Position is not selected by default.

If Set Initial Position is checked, complete the following steps to configure Initial position parameters:

- 1. In **Move to**, enter the variable or value to specify where within the well the aspirate operation begins, either as a percentage of the distance from the center of the well to the well edge or in centimeters (cm). The default for this field is **0**.
- 2. Select the units for the variable or value entered in **Move to**: % (percentage) or cm (centimeters). The default value for this field is %.
- 3. In **from the well center, at**, enter the variable or value to specify the angle at which the Move to measurement is performed.

OR

In the graphic representing the well, position the dial at the desired angle at which the Move to measurement is performed. The value in from the well center, at displays the current angle, in degrees. The default value for the angle is **0** degrees.

- 4. In **and**, enter the variable or value to specify the height of the tip, in millimeters (mm). The default value for this field is the context variable **C\_\_\_Height**. (Refer to Table 10-1 for a definition of this variable.)
- In from, enter the variable or value to specify from where the height in and is measured. The default value for this field is the context variable
   C\_HeightFrom. (Refer to Table 10-1 for a definition of this variable.)

To configure the Final position parameters:

- 1. In **Move to**, enter the variable or value to specify where within the well the aspirate operation moves, either as a percentage of the distance from the center of the well to the well edge or in centimeters (cm). The default for this field is **0**.
- 2. Select the units for the variable or value entered in **Move to**: % (percentage) or **cm** (centimeters). The default value for this field is %.
- 3. In **from the well center, at**, enter the variable or value to specify the angle at which the Move to measurement is performed. The default value for the angle is **0** degrees.

OR

In the graphic representing the well, position the dial at the desired angle at which the Move to measurement is performed. The value in from the well center, at displays the current angle, in degrees.

- 4. In **and**, enter the variable or value to specify the height of the tip, in millimeters (mm). The default for this field is **0** mm.
- 5. In **from**, enter the variable or value to specify from where the height in and is measured. The default for this field is **Bottom**.

To complete the Moving Aspirate configuration:

- In Volume, enter the variable or value to specify the volume of liquid to aspirate, in microliters (μL). The default value for this field is the context variable
   C\_\_Volume. (Refer to Table 10-1 for a definition of this variable.)
- In Speed, enter the variable or value to specify the speed of the pod in the D-axis in the D-axis in microliters per second (μl/sec) during the moving aspirate operation. The default value for this field is the context variable C\_\_\_Speed. (Refer to Table 10-1 for a definition of this variable.)
- In Post Delay, enter the variable or value to specify the length of the delay in milliseconds (ms) after the aspirate operation. The pod pauses for the specified duration before continuing with the next operation. The default value for this field is the context variable C\_\_AspirateDelay. (Refer to Table 10-1 for a definition of this variable.)

### 10.5.6 Dispense Air Gap

Dispense Air Gap dispenses an air gap at a specified speed and allows a delay before further operations occur. Dispensing an air gap is sometimes called a *blowout* because it blows out the leading air gap.

Dispense Air Gap can be used to dispense a trailing or leading air gap, depending on where within the pipetting template it is inserted.

To configure Dispense Air Gap:



1. Insert **Dispense Air Gap** into the pipetting method. The template step configuration appears (Figure 10-13).





- In speed, enter the variable or value to specify the speed of the pod in the D-axis in microliters per second (μl/sec) during the blowout. The default value for this field is the context variable C\_\_Speed. (Refer to Table 10-1 for a definition of this variable.)
- 3. In delay, enter the variable or value to specify the length of the delay in milliseconds (ms) after the blowout. The pod pauses for the specified duration before continuing with the next operation. The default value for this field is the context variable C\_\_\_DispenseDelay. (Refer to Table 10-1 for a definition of this variable.)

#### 10.5.7 Dispense

Dispense dispenses a specific volume at a speed specified in microliters per second ( $\mu$ l/sec). To pause for a specific amount of time after dispensing, add a delay to the Dispense template step.

To configure Dispense:



1. Insert **Dispense** into the pipetting method. The template steps configuration appears (Figure 10-14).



Figure 10-14. Dispense step configuration

 In volume, enter the variable or value to specify the volume of liquid to dispense in microliters (μL). The default value for this field is the context variable C\_\_Volume. (Refer to Table 10-1 for a definition of this variable.)

- 3. In **speed**, enter the variable or value to specify the speed of the pod during the dispense operation as a percentage of the maximum speed of the pod. The default value for this field is the context variable **C\_\_\_Speed**. (Refer to Table 10-1 for a definition of this variable.)
- In delay, enter the variable or value to specify the length of the delay in milliseconds (ms) after the dispense operation. The pod pauses for the specified duration before continuing. The default value for this field is the context variable C\_\_\_DispenseDelay. (Refer to Table 10-1 for a definition of this variable.)

#### 10.5.8 Moving Dispense

Moving Dispense dispenses a specific volume of liquid while moving between positions relative to the center of the well.

To configure Moving Dispense:



1. Insert **Moving Dispense** into the pipetting method. The template step configuration appears (Figure 10-15).

Pipetting Te	emplate Edi	itor
	×	
New	Remove	Copy Paste Rename Variables
Comment	Moving	Pipetting Template:           New Template         Imitial Position
If		Aspirate Aspirate 0 µL for Air Gap at =CSpeed Aspirate =CVolume µL at =CSpeed from the well center, at 0 degrees
Loop	Prewet	End     Dispense     We Dispense Air Gap at =C_Speed     From =C_HeightFrom     From =C_HeightFrom
Aspirate Air	Tip Touch	Moving Dispense     End     Dials for adjusting the     angle of movement from     from the well center, at 0 degrees
Aspirate	Axes Move	the center of the well. (Number in degrees
Moving Aspirate	Pause	Volume: =C_Volume µL
Dispense	Bulk	Moving Dispense Moving Dispense elay =C_DispenseDelay ms
Air Gap	Dispense	Dispenses a specified amount while moving Configure Moving
Dispense		the tips inside wells. Dispense settings

Figure 10-15. Moving Dispense step configuration

2. Check Set Initial Position to set the start position for dispense, if desired.

**Note:** Set Initial Position is used if a dispense must begin at a specific well location on the labware. If this is not set, the pod moves from its current location. Set Initial Position is not selected by default.

If Set Initial Position is checked, complete the following steps to configure Initial position parameters:

- 1. In **Move to**, enter the variable or value to specify where within the well the dispense operation begins, either as a percentage of the distance from the center of the well to the well edge or in centimeters (cm). The default for this field is **0**.
- 2. Select the units for the variable or value entered in **Move to**: % (percentage) or cm (centimeters). The default value for this field is %.
- 3. In **from the well center, at**, enter the variable or value to specify the angle at which the Move to measurement is performed. The default value for the angle is **0** degrees.

OR

In the graphic representing the well, position the dial at the desired angle at which the Move to measurement is performed. The value in from the well center, at displays the current angle, in degrees.

- 4. In **and**, enter the variable or value to specify the height of the tip, in millimeters (mm). The default value for this field is the context variable **C\_\_\_Height**. (Refer to Table 10-1 for a definition of this variable.)
- In from, enter the variable or value to specify from where the height in and is measured. The default value for this field is the context variable
   C\_HeightFrom. (Refer to Table 10-1 for a definition of this variable.)

To configure the Final position parameters:

- 1. In **Move to**, enter the variable or value to specify where within the well the dispense operation moves, either as a percentage of the distance from the center of the well to the well edge or in centimeters (cm). The default for this field is **0**.
- 2. Select the units for the variable or value entered in **Move to**: % (percentage) or **cm** (centimeters). The default value for this field is %.
- 3. In **from the well center, at**, enter the variable or value to specify the angle at which the Move to measurement is performed. The default value for the angle is **0** degrees.

OR

In the graphic representing the well, position the dial at the desired angle at which the Move to measurement is performed. The value in from the well center, at displays the current angle, in degrees.

- 4. In **and**, enter the variable or value to specify the height of the tip, in millimeters (mm). The default for this field is **0** mm.
- 5. In **from**, enter the variable or value to specify from where the height in and is measured. The default for this field is **Bottom**.

To complete the Moving Dispense configuration:

- In Volume, enter the variable or value to specify the volume of liquid to dispense, in microliters (μL). The default value for this field is the context variable C\_\_Volume. (Refer to Table 10-1 for a definition of this variable.)
- In Speed, enter the variable or value to specify the speed of the pod in the D-axis in microliters per second (μl/sec) during the moving dispense operation. The default value for this field is the context variable C\_\_Speed. (Refer to Table 10-1 for a definition of this variable.)
- In Post Delay, enter the variable or value to specify the length of the delay in milliseconds (ms) after the dispense operation. The pod pauses for the specified duration before continuing. The default value for this field is the context variable C\_\_\_DispenseDelay. (Refer to Table 10-1 for a definition of this variable.)

#### 10.5.9 Mix

Mix aspirates and dispenses liquid in wells of labware as a mixing operation.

**Note:** This template step performs operations only when Mix is selected in the Technique Editor for the technique being used during method run.

To configure Mix:



1. Insert **Mix** into the pipetting method. The template steps configuration appears (Figure 10-16).

Pipetting T	emplate Ed	litor	
D	$\times$	h (2) 🖆 🍡 -	
New	Remove	Copy Paste Rename Variables	
	lait	Pipetting Template:	If =CMix is True,
~~	₩ ₹ Moving	New Template	
Comment	Dispense	Aspirate	
	l∎l±	Aspirate 0 µL for Air Gap at =CSpeed	=CMixCount times.
TE	₩ ♥ Mix	W Aspirate = Cvolume µL at = Cspeed	Aspirate atMixAspirateSpeedul /s
		End	
			then delay 0 ms
Loop	Prewet	(M) Dispense Air Gap at =⊂Speed MI Dispense =⊂ Volume µL at =⊂ Speed	Set Aspirate Height
ini.		Moving Dispense	Gio to _=CMixAspirateHeight mm
⊽t	<b>±</b>	End	from =C_MixAspirateFrom
Aspirate Air Gap	Tip Touch	Ten Mix	Dispense at =CMixDispenseSpeeduL/s
	44	End	then delay.
<b>₩</b> I	24		Cost Discourse Height
Aspirate	Axes move		Go to -C - MixDispenseHeight mm
. ♥‡	8		
Moving	Pause		from =C_MixDispenseFrom
Aspirate	5		<b>▲</b>
♥		Mix	
Dispense Air Gan	Bulk	Performs an aspirate	Mix Configuration
		and dimense to min	Mix settings are configured
		and dispense to mix	have
Dispense		liquids.	nere.
			OK Cancel Apply

Figure 10-16. Mix step configuration

- In Mix, enter the variable or value to specify the volume of liquid to mix, in microliters (μL). The default value for this field is the context variable C\_MixVolume. (Refer to Table 10-1 for a definition of this variable.)
- In times, enter the variable or value to specify the number of times to mix the liquid. The default value for this field is the context variable C\_\_\_MixCount. (Refer to Table 10-1 for a definition of this variable.)
- 4. In Aspirate at, enter the variable or value to specify the speed of the pod in the D-axis in microliters per second (μl/sec) for the aspirate operation. The default value for this field is the context variable C\_MixAspirateSpeed. (Refer to Table 10-1 for a definition of this variable.)
- 5. In **then delay**, enter the variable or value to specify the length of the delay in milliseconds (ms) after the mix operation. The pod pauses for the specified duration before continuing with the next operation. The default value for this field is **0** ms.
- 6. Check **Set Aspirate Height** to enter the variable or expression for the height of the tips during aspiration.

**Note:** If Set Aspirate Height is not selected, the mix operation occurs at the current height and the tips do not move vertically before aspirating.

- In Go to, enter the variable or value to specify the height of the tip for the aspirate operations, in millimeters (mm). The default value for this field is the context variable C\_\_\_MixAspirateHeight. (Refer to Table 10-1 for a definition of this variable.)
- In from, enter the variable or value to specify from where the height in Go to is measured. The default value for this field is the context variable
   C MixAspirateFrom. (Refer to Table 10-1 for a definition of this variable.)
- In **Dispense at**, enter the variable or value to specify the speed of the pod in the D-axis in microliters per second (μl/sec) for the dispense operation. The default value for this field is the context variable C\_\_MixDispenseSpeed. (Refer to Table 10-1 for a definition of this variable.)
- 10. In **then delay**, enter the variable or value to specify the length of the delay in milliseconds (ms) after the mix operation. The pod pauses for the specified duration before continuing with the next operation. The default value for this field is **0** ms.
- 11. Check **Set Dispense Height** to enter the variable or expression for the height of the tips during dispense.

**Note:** If Set Dispense Height is not selected, the mix operation occurs at the current height and the tips do not move vertically before dispensing.

- In Go to, enter the variable or value to specify the height of the tip for the dispense operations, in millimeters (mm). The default value for this field is the context variable C\_\_\_MixDispenseHeight. (Refer to Table 10-1 for a definition of this variable.)
- In from, enter the variable or value to specify from where the height in Go to is measured. The default value for this field is the context variable
   C\_\_MixDispenseFrom. (Refer to Table 10-1 for a definition of this variable.)

#### 10.5.10 Prewet

Prewet aspirates, then dispenses the liquid to prepare tips for the main aspiration. Prewet should only be used in the Aspirate operation of a pipetting template. The delay occurs between the aspirate and dispense of the Prewet.

**Note:** This template step performs operations only when **Prewet** is selected in the **Technique Editor** for the technique being used during method run.

To configure Prewet:





Figure 10-17. Prewet step configuration

- In volume, enter the variable or value to specify the volume of liquid used to prewet the tips. The default value for this field is the context variable
   C\_PrewetOverage. (Refer to Table 10-1 for a definition of this variable.)
- In speed, enter the variable or value to specify the speed of the pod in the D-axis in microliters per second (µl/sec) during the prewet operation. The default value for this field is the context variable C\_\_Speed. (Refer to Table 10-1 for a definition of this variable.)
- 4. In pause, enter the length of time to delay between aspirating and dispensing the prewet liquid. The pod pauses for the specified duration after aspirating the prewet liquid and before dispensing the prewet liquid. The default value for this field is the context variable C\_\_PrewetDelay. (Refer to Table 10-1 for a definition of this variable.)



### 10.5.11 Tip Touch

If Tip Touch is selected in the technique, the tip touches the side of the well and performs any specified delays. The tip then moves back to the center of the well.

**Note:** This template step performs operations only when Tip Touch is selected in the Technique Editor for the technique being used during method run.

To configure Tip Touch:



1. Insert **Tip Touch** into the pipetting method. The template step configuration appears (Figure 10-18).

Pipetting T	emplate Ed	itor	
New	X Remove	Copy Paste Rename Variables	Dial for adjusting the angle of movement
Q	Moving	Pipetting Template: New Template	
Comment	Dispense Mix Prewet	<ul> <li>Aspirate</li> <li>Aspirate 0 µL for Air Gap at =C_Speed</li> <li>Prewet</li> <li>Aspirate =C_Volume µL at =C_Speed</li> <li>Moving Aspirate</li> <li>End</li> <li>Dispense</li> <li>Dispense Air Gap at =C_Speed</li> <li>Moving Dispense</li> <li>Moving Dispense</li> <li>To Touch</li> <li>Ford</li> </ul>	from =CTipTouchHeightFrom then move to the side of the well at =CTipTouchTh degrees, then delay =CTipTouchDelay ms before returning to the center of the well. Specify tip touch speed =CTipTouchSpeed %
Aspirate Aspirate Moving Aspirate Dispense Air Gap	Axes Move Axes Move Pause Bulk Dispense	Mix Mix End TipTouch Performs a tip touch to aspirate, delay, then	■ Blowout during tip touch ■CSpeed ≈ Tip Touch Configuration Tip Touch settings are configured here.
Dispense		dispense inquids.	OK Cancel Apply

Figure 10-18. Tip Touch step configuration

- In move to height, enter the variable or value to specify the desired height, in millimeters (mm), at which to touch the well edge. The default value for this field is the context variable C\_\_\_TipTouchHeight. (Refer to Table 10-1 for a definition of this variable.)
- In from, enter the variable or value to specify from where the move to height is measured for the tip touch location. The default value for this field is the context variable C\_\_\_TipTouchHeightFrom. (Refer to Table 10-1 for a definition of this variable.)

4. In at, enter the desired variable or value to specify the angle along the well edge at which to touch. The default value for the angle is the context variable
 C\_\_\_\_\_\_TipTouchTheta. (Refer to Table 10-1 for a definition of this variable.)

OR

In the graphic representing the well, position the dial at the desired angle at which the Move to measurement is performed. The value in from the well center, at displays the current angle, in degrees.

- 5. In **then delay**, enter the variable or value to specify the length of the delay before returning to the center of the well. The default value for this field is **0** milliseconds (ms).
- 6. To control the tip touch speed, check Specify tip touch speed.

**Note:** If Specify tip touch speed is not selected, the tip touch operation occurs at the operation speed as defined in the technique.

- In the text box below Specify tip touch speed, enter the variable or value to specify the speed of the pod during the tip touch operation as a percentage of the maximum speed of the pod. The default value for this field is the context variable C\_\_\_TipTouchSpeed. (Refer to Table 10-1 for a definition of this variable.)
- 8. To add a blowout, check **Blowout during tip touch**.

**Note:** The blowout occurs during the vertical move configured at the beginning of the tip touch.

 In the text box below Blowout during tip touch, enter the variable or value to specify the speed of the pod during the blowout operation as a percentage of the maximum speed of the pod. The default value for this field is the context variable C\_\_\_Speed. (Refer to Table 10-1 for a definition of this variable.)

#### 10.5.12 Axes Move

Axes Move moves the tips to a defined place within a well. For example, the template steps may move the tips 10% from the well center and 90 degrees around the edge of the well.

To configure Axes Move:



1. Insert **Axes Move** into the pipetting method. The template step configuration appears (Figure 10-19).



Figure 10-19. Axes Move step configuration

- 2. In **Move to**, enter the variable or value to specify where within the well to move the tip, either as a percentage of the distance from the center of the well to the well edge or in centimeters (cm). The default for this field is **0**.
- 3. Select the units for the variable or value entered in **Move to**: % (percentage) or cm (centimeters). The default value for this field is %.
- 4. In **from the well center, at**, enter the variable or value to specify the angle at which the Move to measurement is performed. The default value for the angle is **0** degrees.

OR

In the graphic representing the well, position the dial at the desired angle at which the Move to measurement is performed. The value in from the well center, at displays the current angle, in degrees.

- 5. In **and**, enter the variable or value to specify the height of the tip, in millimeters (mm). The default value for this field is **0** mm.
- 6. In **from**, enter the variable or value to specify from where the height in and is measured. The default value for this field is **Bottom**.
- 7. In **at speed**, enter the variable or value to specify the speed of the pod during the axes move as a percentage of the maximum speed of the pod. The default value for this field is **0**%.

#### 10.5.13 Pause

Pause allows the template to stop operations temporarily. Pause is configured in milliseconds only.

To configure Pause:



1. Insert the **Pause** template steps into the pipetting method. The template steps configuration appears (Figure 10-20).

Pipetting T	emplate Ed	litor							
	$\mathbf{X}$		r	<b>N</b> -					
New	Remove	Copy Paste	Rename	Variables		-			
	⊌‡	Pipetting Template:				Duration: 0		ms	
Comment	Moving Dispense	Aspirate							
F	₩‡	Aspirate 0 µL f	or Air Gap at	=CSpeed					
If	Mix	Aspirate =C	Volume µL at	=CSpeed					
C	<b> </b>	Htt Moving Aspiral	e			Pa	use Cor	nfigurat	ion
	Prewet	Dispense	ap at =CS	peed		Enter	the millis	seconds t	.0
<b>∀</b> T Aspirate Air	Tip Touch	Dispense =C_	_Volume µL al se	t =CSpeed		pause	operation	ns.	
Gap		Tip Touch							
<b>VT</b> Aspirate	Axes Move	Mix Move to 0%, (	)°,0mm						
₩‡	0	End	Р	ause	]				
Moving Aspirate	Pause		Places	a pause in					
U+	<u>í</u>		an ope	ration.					
Dispense Air Gap	Bulk Dispense				-				
<b>⊳</b> +									
Dispense									
						ОК	Ca	ancel	Apply

Figure 10-20. Pause step configuration

2. In **Duration**, enter the variable or value for the length of the pause, in milliseconds (ms). The default value for this field is **0** ms.

## 10.5.14 Bulk Dispense (FX and NX-S8 only)

Bulk Dispense dispenses system fluid through the system tubing using the tips and syringes of the Span-8 Pod.

To configure Bulk Dispense:



1. Insert **Bulk Dispense** into the pipetting method. The template step configuration appears (Figure 10-21).

Pipetting T	emplate Ed	itor							
	×		C2	r an	- لا				
New	Remove	Сору	Paste	Rename	Variables				
Comment	Dispense	Pipetting New Te	) Template: mplate				•	Dis vol	spense system liquid with Iume =CVolume μL
F If	Moving Dispense	Na Aspira Unita Unita Unita Unita Unita Unita Na Aspira	ate spirate =C rewet spirate =C ause for 0 ms oving Aspirat	Volume µL at Volume µL at ; ;e	=CSpeed =CSpeed			ats the	speed =C_DispenseSpeed µL/s en delay =C_DispenseDelay ms speed pump
Loop Ut Aspirate Air Gap	Mix Hix Prewet	● Ei Dispe H, D H, D H, D H, M	nd nse ispense =C_ ispense =C_ oving Dispen p Touch	_Volume µL a _Volume µL a se	t =CSpeed t =CSpeed				
Aspirate	Tip Touch	Mix ₩ix ₩ix ₩im ₩im ₩im ₩im	ispense =C_ nd ove to 0%, ( ix nd	_Volume µL o 0° , 0 mm	f System Liquid .	at =CDisper	nseSpeed		Bulk Dispense Configuration Bulk Dispense settings are configured here.
Dispense Air Gap	Pause Pause Bulk Dispense	]←		Bul Places in an o	lk Dispe a bulk dis peration.	nse spense			
									OK Cancel Apply

Figure 10-21. Bulk Dispense step configuration

In volume, enter the variable or value to specify the amount of system fluid to dispense, in microliters (μL). The default value for this field is the context variable C\_\_\_Volume. (Refer to Table 10-1 for a definition of this variable.)

- In speed, enter the variable or value to specify the speed of the pod in microliters per second (μl/sec) during the bulk dispense operation. The default value for this field is the context variable C\_\_Speed. (Refer to Table 10-1 for a definition of this variable.)
- 4. In **delay**, enter the variable or value to specify the length of the delay in milliseconds (ms) after the bulk dispense operation. The pod pauses for the specified duration before continuing with the next operation. The default value for this field is the context variable C\_\_DispenseDelay. (Refer to Table 10-1 for a definition of this variable.)
- 5. Depending on instrument configuration, speed pump or purge pump appears.
  - **FX** Choose **speed pump** to use this optional device.
  - > *NX-S8* Choose **purge pump** to use it.



# 11.1 Overview

- **FX**, **NX-S8** Only the Span-8 Pod can be used to create well patterns.
- > **3000** Only a Single-Tip Pipette Tool can be used to create well patterns.

Creating certain methods requires the probes to access specific wells in labware. Well Patterns (Figure 11-1) is an editor which allows patterns for accessing specific wells to be created and stored. There are also well patterns already created in the editor which may be used, copied, or revised to create new patterns. Any well pattern stored in Well Patterns may be renamed or deleted.

The patterns created and stored in Well Patterns are accessible in the Transfer and Combine steps (refer to Section 15.3, *Configuring Transfer and Combine Steps*). Well patterns may also be created in the Transfer and Combine steps and then stored in Well Patterns.

Well patterns, along with information about tip and labware types; liquid types; and pipetting templates and techniques, are stored as part of a project file. Project files store a history of all changes, additions, and deletions of items from the project file. Refer to Chapter 6, <u>Understanding and Using Project Files</u>, for more information on project files.

**Note:** When Beckman Coulter Accounts & Permissions is enabled, only users with the **Develop Projects** permission assigned may open, edit, create, and delete well patterns (refer to Chapter 2, *Using Accounts & Permissions*).



To access the Well Pattern Editor, choose **Project>Well Pattern Editor**. Well Patterns appears (Figure 11-1).

Figure 11-1. Well Patterns

Use the Well Pattern Editor for:

- <u>Creating a New Well Pattern</u> (Section 11.2)
- <u>Copying a Well Pattern</u> (Section 11.3)
- <u>Deleting a Well Pattern</u> (Section 11.4)
- <u>Renaming a Well Pattern</u> (Section 11.5)

# **11.2 Creating a New Well Pattern**

New well patterns are created based on the type of labware accessed by the specific probes.

**Note:** To make sure the desired pattern is not already created, view the patterns in Well Patterns by highlighting the well pattern in the list displayed in Well Patterns (Figure 11-1).

To create a new well pattern:

From Well Patterns (Figure 11-1), choose New. New Pattern appears (Figure 11-2).



Figure 11-2. New Pattern

- 2. From the list of labware, choose a labware type on which to base the new pattern.
- 3. In Pick a name for your pattern:, rename the new pattern, if desired.

**Note:** By default, the first new pattern is named after the number of wells on the labware type. Subsequent patterns using the same labware type are named after the number of wells plus a number. For example, the first new pattern created for a 96 well would be Pattern96, the second new pattern would be Pattern96\_1, and the third new pattern would be Pattern96\_2.

4. Choose **OK**. The labware grid based on the labware type appears in Well Patterns.

5. Create the new pattern from the labware grid by dragging the mouse and using Ctrl and Shift on the keyboard.

**Note:** To create a new pattern, select wells while holding down the Shift key. To deselect wells, use the Ctrl key to toggle the wells from select to deselect. To invert the well selection, drag the mouse across the pattern while holding down the Ctrl key.

**Note:** If Shift is not held down when dragging and selecting, the previous selection will be deleted when subsequent selections are made.

6. Choose **Save** to save the changes. Well Patterns closes and all the changes made in the dialog are saved.

OR

Choose **Cancel** to cancel the changes. A Warning (Figure 11-3) appears. Choose **Yes**. Well Patterns closes and all the changes made in the dialog are cancelled.

₩arning	X
	You will lose all the changes that you made in this dialog. Are you sure you want to do this?
	Yes <u>N</u> o

Figure 11-3. Warning when Cancel is chosen

# 11.3 Copying a Well Pattern

Copying a well pattern creates an exact copy of it. The copied well pattern can be modified to create a new well pattern.

To copy a well pattern:

- 1. From Well Patterns (Figure 11-1), highlight the desired pattern from the Well Pattern List.
- 2. Choose **Copy**. Copy Pattern (Figure 11-4) appears.

Copy Pattern	×
Please enter a name:	
Pattern384	
OK Cancel	

Figure 11-4. Copy Pattern

3. Rename the pattern, if desired.

**Note:** By default, the first new pattern is named after the number of wells on the labware type. Subsequent patterns using the same labware type are named after the number of wells plus a number. For example, the first new pattern created for a 96 well would be Pattern96, the second new pattern would be Pattern96\_1, and the third new pattern would be Pattern96\_2.

4. Choose **OK**. The copied pattern appears in Well Patterns.

**Note:** If desired, modify the well pattern.

5. Choose **Save** to save the changes. Well Patterns closes and all the changes made in the dialog are saved.

OR

Choose **Cancel** to cancel the changes. A Warning (Figure 11-3) appears. Choose **Yes**. Well Patterns closes and all the changes made in the dialog are cancelled.

# 11.4 Deleting a Well Pattern

To delete a well pattern stored in Well Patterns:

- 1. From Well Patterns (Figure 11-1), highlight the desired pattern from the Well Pattern List.
- 2. Choose **Delete**. A Warning (Figure 11-5) appears, confirming the deletion.



Figure 11-5. Warning confirming deletion of a pattern

- 3. Choose **OK**. The pattern is deleted from the list of patterns in Well Patterns.
- 4. Choose **Save** to save the changes. Well Patterns closes and all the changes made in the dialog are saved.

OR

Choose **Cancel** to cancel the changes. A Warning (Figure 11-3) appears. Choose **Yes**. Well Patterns closes and all the changes made in the dialog are cancelled.
# 11.5 Renaming a Well Pattern

Renaming a labware type provides the opportunity to assign a descriptive name to a well pattern for easy identification.

To rename a well pattern in Well Patterns:

- 1. From Well Patterns (Figure 11-1), highlight the desired pattern from the Well Pattern List.
- 2. Choose **Rename**. Rename Pattern (Figure 11-6) appears.

Rename Pattern	×
Pattern384	
OK	Cancel

Figure 11-6. Rename Pattern

- 3. Rename the pattern.
- 4. Choose **OK**. The pattern is renamed in the list of patterns in Well Patterns.
- 5. Choose **Save** to save the changes. Well Patterns closes and all the changes made in the dialog are saved.

OR

Choose **Cancel** to cancel the changes. A Warning (Figure 11-3) appears. Choose **Yes**. Well Patterns closes and all the changes made in the dialog are cancelled.



### 12.1 Overview

A method is a series of steps controlling the operations of the Biomek instrument. When steps are inserted into a method, they represent operations performed during the method run. The steps available for insertion into a method are located on the left of the Biomek main editor in the Step Palettes (Figure 12-1).

**Note:** When Beckman Coulter Accounts & Permissions is enabled, only users with Develop Methods permission may develop, open, and edit methods (refer to Chapter 2, <u>Using Accounts & Permissions</u>).

Biomek Software does a substantial amount of the method-building work while at the same time allowing direct and precise control over the building of methods.

Methods are saved as part of a project file, and may also have revisions created by checking in the method. A revision of a method is connected to the current revision for all project items. When a specific revision of a method is opened, it uses the revision of all project items that was current at the time the method was checked in.

**Note:** The appearance of the main editor (Figure 12-1), such as the Current Deck Display and step palettes, varies based on the current instrument file and project file. The current instrument and project file names are displayed in the status bar of the main editor.



Figure 12-1. Biomek Software main editor for method creation

Using methods to control the operations of the Biomek instrument includes:

- <u>Creating a New Method</u> (Section 12.2).
- <u>Inserting Steps in a Method</u> (Section 12.3).
- <u>Using Additional Step Palettes</u> (Section 12.4).
- <u>Copying, Cutting, and Pasting Steps in a Method</u> (Section 12.5).
- <u>Deleting Steps in a Method</u> (Section 12.6).
- <u>Using Undo and Redo in Method Building</u> (Section 12.7).
- <u>Entering and Viewing Method Properties</u> (Section 12.8).
- <u>Saving a Method</u> (Section 12.9)
- <u>Checking In a Method</u> (Section 12.10).
- *Validating a Method* (Section 12.11)
- <u>Signing a Method</u> (Section 12.12).
- <u>Viewing Method History</u> (Section 12.13).
- **Opening a Saved Method** (Section 12.14).
- <u>Checking Out a Method</u> (Section 12.15).
- Importing and Exporting Methods (Section 12.16).
- <u>Running, Pausing, and Stopping a Method</u> (Section 12.18).
- <u>Printing a Method</u> (Section 12.19).
- <u>Improving a Method</u> (Section 12.20).

# 12.2 Creating a New Method

The main editor (Figure 12-1) is the starting point for creating a liquid-handling method using the Biomek instrument. Method steps from the step palette are inserted into the Method View in a linear fashion.

When steps are inserted into a method, the configuration associated with that step appears on the right side of the Biomek main editor (Figure 12-2).

**Note:** Steps not configured appropriately generate errors when the method is validated or run.

To create a new method:

- 1. Launch Biomek Software. The main editor appears (Figure 12-1).
- 2. From the toolbar, select the **New Method** icon.

OR

From the File menu, choose **New**.

OR

Drag and drop a step, such as Instrument Setup, into the Method View. A new method is created and assigned the default name Method# [New] (Figure 12-2).





Figure 12-2. Biomek Software main editor for dual-pod Biomek FX instrument

There are two steps automatically present in every method created and executed by Biomek Software:



- Start the first step in a method; allows creation of global variables that are applicable to the entire method, as opposed to a variable applicable only to one step within the method.
- Finish the final step in a method; provides options to clear the Biomek instrument deck, clear the devices on the deck, unload tips from the pod, clear all global variables, and generate a report on data sets.

#### 12.2.1 Configuring the Start Step

Start

Use the **Start** step to define variables used throughout the entire method, as opposed to variables defined within individual steps. A variable is a value that has been assigned a name (Figure 12-3).

**Note:** Variables defined in individual steps, such as the Let step, are applicable only to that step and any nested substeps. Refer to Section 17.6, *Let Step*, for more information about the Let step.

To use a variable created in the Start step, that variable name is entered into the configuration of the step using the variable. In the example, the variable AspValue, created in the Start configuration (Figure 12-4), is entered in the  $\mu$ L (volume) field of the Transfer step.

The Biomek Software internally substitutes the name with its value, but the name remains in the field. In the example (Figure 12-4), AspValue is substituted internally by the value '10'. The value of '10' is never displayed, only the variable name AspValue. The method can be run subsequently with a different value for AspValue, and the software automatically replaces AspValue with the new value.

Variables created in the **Start** step can also give a prompt at the start of a method run allowing new values to be entered for each variable at runtime.

**Note:** Refer to Chapter 13, <u>Using Variables and Expressions in a Method</u>, for more information on using variables in a method.

To configure a variable in the Start step:

1. Select **Start** in the Method View. The Start step configuration appears (Figure 12-3).



Figure 12-3. Start step and configuration



2. Click under **Variable Name** (Figure 12-4) to position the cursor in the Variable Name field.

Figure 12-4. Start step with variables configured

3. Enter a Variable Name.

**Note:** Variable names must begin with a letter, may only use alphanumeric characters (0-9, A-Z) and the underscore (\_), and may not exceed 255 characters. Variable names are not case sensitive.

- 4. Enter the desired default Value for the variable.
- 5. Check the box under Overridable (Figure 12-4) if the variable can be overridden by a previously defined value. For example, a variable Test is created in the Start step with a value of 10. A Let step later in the method assigns a value of 15 to the Test variable. If Overridable is selected in the Let step, the previously existing value from the Start step of 10 overrides the value in the Let step. If Overridable is not selected, however, the variable is assigned the value of 15 defined in the Let step.

6. Check the box under **Prompt** (Figure 12-4) to activate a prompt for the variable at run time. Each time the method is run, a prompt similar to the one shown in Figure 12-5 asks for verification of the variables and their values. Verification of the variables provides an opportunity to change the default values associated with the variables each time the method is run, if desired.

**Note:** A separate prompt appears for each variable that has **Prompt** selected.

Biomek® Software	_	
Enter a value to use for 'AspValue'		
25		
,	OK	
		9/24/2003 2:04:06 PM

Figure 12-5. Prompt to enter value for variable at runtime

7. Repeat steps 2-6 for each variable.

**Note:** To get a new line for another variable, press **Enter** with the cursor in the Value field of the last defined variable.

### 12.2.2 Configuring the Finish Step

🎽 Finish

The Finish step is the last step in every Biomek method. The Finish step indicates the end of a method and provides opportunities to control pod, deck, and device status upon completion of a method.

When highlighted in the Method View:

- The Finish step configuration is displayed on the right side of the Biomek main editor.
- The software runs an internal validation to check for errors in the method.
- An Estimated Time to Completion (ETC) of the method is displayed in the status bar at the bottom of the Biomek main editor.

**Note:** When the Finish step is highlighted in the Method View, the software estimates the real time required to complete the entire method. When any other step is highlighted in the Method View, the length of time displayed in the ETC field represents the time required to complete the method up to the selected step. If an error is found in validation, ETC is displayed as Failed.

Use the Finish step to re-initialize the software state of the deck and external hardware devices, and to physically remove tips from the pod when a method is completed.

The default configuration for the Finish step is all options selected (Figure 12-6). If the options are not selected, errors may occur due to conflicts between the labware in a new method and the labware from the previous method. The error occurs because software and hardware remain in their final state after a method is run, unless told otherwise.

**Note:** Each of the Finish step options is independent of the others, so any combination of the options may be selected.

**Note:** If the Finish step configuration options are not used, steps must be added to the new method to reset the software and hardware.

The options of the Finish step reset the hardware and software so a new method can populate the deck as required, without encountering labware errors. The Finish step also allows for reports to be generated on labware data from data sets (refer to Chapter 14, <u>Using Sample Tracking and Data Sets in a Method</u>).



Figure 12-6. The Finish step configuration for a Biomek FX instrument

#### 12.2.2.1 Clear Current Instrument Setup of All Labware After the Method Completes

The Current Deck Display represents the status of the labware on the deck upon completion of a method run (Figure 12-6). Typically, a clear deck is needed at the start of a method.

To clear the deck in the software after a method, select **Clear current instrument** setup of all labware after method completes in the Finish step configuration. This ensures a clear deck for the next method.

**Note:** To use the same deck setup for a subsequent method, do not select **Clear current instrument setup of all labware after method completes** in the Finish step configuration. The labware positions, volumes, and marks carry over to the next method. Use the **As Is** option in the **Instrument Setup** step in the subsequent method to leave those positions in their current state.

# 12.2.2.2 Clear Current Device Setup of All Labware After the Method Completes

The Device Setup remembers the contents of the devices on the deck at completion of a method run (Figure 12-6). Typically, devices are clear of labware at the start of a method.

To clear all devices of labware, select **Clear current device setup of all labware after method completes** in the Finish step configuration. This ensures clear devices for the next method.

**Note:** Insert an Instrument Setup step at the beginning of the next method to specify where labware is located on the deck (refer to Section 15.2, *Instrument Setup Step*). Use the Deck Editor to specify the location of devices on the deck (refer to Chapter 5, *Preparing and Managing the Deck*), and use the Device Setup step to configure the contents of the external devices. Refer to Section 22.4, *Using External Devices with Biomek Laboratory Automation Workstations*, and Section 22.5, *Device Setup Step*, for more information about configuring external devices.

#### 12.2.2.3 Unload Disposable Tips from All Pods and Wash Any Fixed Tips After the Method Completes

Tips remain on the pod(s) or tools unless they are removed during a step or in the Finish step (Figure 12-6). Typically, the pod or tool needs to be clear of tips at the start of a method.

FX, NX-MC, NX-S8 — Disposable tips can be removed from any Multichannel Pod and any probes on a Span-8 Pod. If any probes on a Span-8 Pod are equipped with fixed tips, they can be washed at the conclusion of the method.

**Note:** To wash any fixed tips on a Span-8 Pod, the deck must have a Span-8 Wash ALP accessible by the Span-8 Pod (refer to the *ALPs User's Manual*, Chapter 18, *Span-8 Tip Wash ALP*) and configured in the Deck Editor (refer to Chapter 5, *Preparing and Managing the Deck*).

> **3000** — Disposable tips can be removed from any pipetting tool.

To clear all pods of disposable tips and wash any fixed tips, select **Unload disposable tips from all pods and wash any fixed tips after method completes** in the Finish step configuration.

# 12.2.2.4 Clear All Global Variables After the Method Completes

Global variables are set in the Start or Set Global steps. A variable is a value that has been assigned a name and can be referenced repeatedly during a method. Global variables are carried over into subsequent methods unless they are cleared in the Finish step.

To clear all global variables, select **Clear all global variables after the method completes** in the Finish step configuration.

# 12.2.2.5 Unload the Current Tool After the Method Completes (3000 Only)

When using Biomek Software to create methods for the Biomek 3000 instrument, an additional option to unload whatever tool is installed on the Biomek 3000 head assembly at the end of the method is available (Figure 12-7). Selecting this option instructs the instrument to unload any tool installed on the Biomek 3000 head assembly upon completion of the method to ensure no tool is installed at the start of the next method.

If Unload the current tool after method completes is not selected, it may be necessary to add additional steps to the start of any subsequent methods to unload or change the tool on the head assembly.

To unload the tool at the end of the method, select **Unload the current tool after the method completes** in the Finish step configuration. Any tool loaded onto the Biomek 3000 head assembly is removed and placed back onto the tool rack at the conclusion of the method.

🌵 Biomek® Software - Method6* [New]		
File Edit Project Instrument Execution O	ptions Help	
Instrument	Clear current instrument setup of all labware after the method completes	
Setup	Clear current device setup of all labware after the method completes	
Transfer	Unload disposable tips from all pods and wash any fixed tips after the method completes	
Combine	Clear all global variables after the method completes Additional option unload a tool from	to
Move Labware	Unload the current tool after the method completes Biomek 3000 head assembly at the en	l d of
Pause	a method.	
Comment		
	P1 P2 P3 P4 P5 P6 P7	
	1 Y	
Method6* Biomek3000 Biomek3000 ETC: 0:0	00:03	

Figure 12-7. Finish step options for a Biomek 3000 instrument

#### 12.2.2.6 Configuring a Report in the Finish Step

The Finish step may be configured to create a report on data sets at the end of the method. Refer to Chapter 14, <u>Using Sample Tracking and Data Sets in a Method</u>, for more information on using data sets in a method.

**Note:** To create a report on data sets after any transfer or pipetting operation, a **Reporting** step may be inserted into a method at any time (refer to Section 24.6, *Configuring the Reporting Step*)

To choose the report style and location of a report at the end of a method:

1. From the Finish step, check **No Reporting** (Figure 12-7) to open the **Reporting** configuration (Figure 12-8).

🌵 Biomek® Software - Method2* [New]		_D×
File Edit Project Instrument Execution (	Options Help	
		_
Instrument	Clear current instrument setup of all labware after the method completes	
Setup	Clear current device setup of all labware after the method completes	
Transfer	A Reporting	◄
	Beport on all labware data after the method completes	
Combine	Report Style:	
	Report Location:	
Move Labware		
Pause		
Q		
	TL1         P4         P8         P12         P16           P1         P5         P9         P13         P17           P2         P6         P10         P14         P18           P3         P7         P11         P15         P19	
Method2* BiomekFX BiomekFX ETC: 0:00:1	19	

Figure 12-8. Reporting enabled on Finish step

2. Select **Report on all labware data after the method completes** to enable reporting.

- 3. From **Report Style**, choose one of the following:
  - Text File
  - Per-Plate HTML Files
  - Per-Plate Text Files
  - SQL Server (refer to Section 24.6.2, <u>Configuring a SQL Server Report</u> <u>Style</u>)

**Note:** Refer to Section 24.6.1, <u>*Report Styles*</u>, for examples and information on the types of reports that may be generated.

4. In **Report Location**, browse to find the location where the file is saved. Reporting on the Finish step is configured.

#### 12.2.2.6.1 Generating a Report When a Method Stops

When a method is stopped or aborted and reporting is configured in the Finish step, a prompt appears asking if a report should be generated as configured in the Finish step (Figure 12-9) (refer to Section 12.2.2.6, *Configuring a Report in the Finish Step*).

Biomek Software	
Method aborted. Generate report on all labware as configured in the Finish step?	
Yes <u>N</u> o	2/16/2004 6:19:07 PM
	2/16/2004 6:19:07 PM

Figure 12-9. Prompt when method is aborted

To generate a report when a method stops:

From Figure 12-9, choose **Yes**. The report is generated as configured in the Finish step.

#### 12.2.2.7 Completing Finish Step Configuration

Once all the desired options on the Finish step have been selected, the method can be run. When the method is completed, any actions selected in the Finish step are performed.

### 12.3 Inserting Steps in a Method

Steps can be added to a method in a sequential order or inserted at any location in the Method View during the method building process; however, steps must represent the desired order of execution when the method is run.

To insert a step in a method, choose one of the following procedures:

- 1. Select the step above the desired insertion point (Figure 12-10).
- 2. Click the desired step on the step palette. The step is inserted below the selected step.



Figure 12-10. Inserting a step into a method

OR

Drag and drop the desired step from the step palette to the desired location in the Method View. A black bar appears at the insertion point (Figure 12-11).



Figure 12-11. Inserting a step by drag and drop

### 12.3.1 Understanding Techniques

All pipetting operations in a method are controlled using techniques. A technique instructs the Biomek instrument in performing pipetting operations, such as aspirate, dispense, and mix. Technique selections are made anytime liquid needs to be aspirated or dispensed during a method.

A technique stores a set of values, such as aspirate and dispense height, pod speed, and tip touch, used in pipetting operation. Biomek Software also stores a set of properties, such as labware type and liquid type, related to each technique. Based upon these values and properties, the appropriate technique is selected automatically for the pipetting operation. This auto-selection can be overridden and a technique selected manually for each operation, if desired.

Pipetting options used in the selected technique are represented visually under the Auto-Select field in the step configuration. These pipetting options may include:





• Tip Touch — allows tips to touch a designated location briefly in a well after aspirating or dispensing.

**Note:** Movement of a tip touch first centers the tip in the well, performs the pipetting operation, moves to the side of the well, performs a specified delay, and returns to the center of the well for dispensing.

• Mix prior to aspirating liquid or Mix after dispensing liquid — mixes the liquid before aspirating or after dispensing.



• Follow liquid level when aspirating or dispensing — is accomplished with calculations that determine the difference in height of a liquid level after an aspirate or dispense operation is completed. The Z and D axis are coordinated according to that liquid level.



- Aspirate a trailing air gap after leaving the liquid draws air into tips after aspirating liquid, and blows out that air before dispensing.
- Blowout all leading air gaps draws air into tips before aspirating liquid, and blows out that air after dispensing to ensure all liquid has been dispensed.

**Note:** Refer to Chapter 9, <u>Understanding and Creating Techniques</u>, for more information on pipetting techniques.

# 12.4 Using Additional Step Palettes

Additional steps, other than those displayed on the Basic Step Palette, may be added to a method by displaying one of the existing step palettes or creating a new step palette using the Step Palette Builder (refer to Chapter 29.5, <u>Using the Step Palette</u> <u>Builder</u>).

To display an existing step palette:

• From the Options menu, select **Toolbars** and then select the desired step palette to display.

**Note:** The step palettes available depend on instrument type and configuration.

OR

Right-click any empty palette space, and select the desired step palette to display from the step palette menu (Figure 12-12).



Figure 12-12. Step palette menu

**Note:** To create a custom step palette or add steps to existing step palettes, select **Palette Builder** to open the Step Palette Builder.

### 12.5 Copying, Cutting, and Pasting Steps in a Method

One or more steps can be copied, cut, and pasted within a method or between methods in a sequential order. Steps can be inserted at any location in the Method View during the method building process; however, they must represent the desired order of execution when the method is run.

The originating method and the destination method must use the same deck and instrument setups when copying, cutting, and pasting steps between methods. Verifying that the deck and instrument setups are the same in both methods ensures that Biomek Software knows where labware is physically located on the deck. If the decks do not match, steps copied from another method may need to be modified to work in the new method.

To copy, cut, and paste one or more steps in a method:

- 1. Highlight a step in the Method View for copy (or cut).
- 2. To select multiple steps to copy (or cut):
  - While holding down **Shift**, select the final step in the series to copy (or cut). All the steps between the first and second steps selected are now highlighted.
  - While holding down **Ctrl**, select additional steps to copy (or cut). All the selected steps are highlighted.
- 3. From the Edit menu, select Copy (or Cut).

OR

Right-click on any of the selected steps and select **Copy** (or **Cut**) from the menu that appears.

**Note:** If Cut is selected, the highlighted step(s) are removed from the Method View. If any other step is **Copied** or Cut prior to **Pasting** the first step in the Method View, the first copied or cut step is lost or overwritten.

- 4. If the step is to be pasted in a different method, open the desired method (refer to Section 12.14, *Opening a Saved Method*).
- 5. Highlight the step below where the copied or cut step(s) are to be pasted.
- 6. From the Edit menu, select **Paste**.

OR

Right-click the step below where the copied or cut step(s) are to be pasted and select **Paste** from the menu that appears. The step(s) are inserted above the highlighted step.

# 12.6 Deleting Steps in a Method

Steps can be deleted from a method during the method-building process if they are no longer desired.

To delete a step from a method:

- 1. Highlight the step in the Method View.
- 2. To select multiple steps to delete:
  - While holding down **Shift**, select the final step in the series to delete. All the steps between the first and second steps selected are now highlighted.
  - While holding down **Ctrl**, select additional steps to delete. All the selected steps are highlighted.
- 3. Press **Delete** on the keyboard.

OR

From the Edit menu, select Clear. The highlighted step(s) are deleted.

OR

Right-click on any of the selected steps and select **Clear** from the menu that appears.

- 4. If Ask for confirmation before removing a step from a method is selected in Preferences, a prompt appears asking for confirmation each time a step is deleted from a method (Figure 12-13).
- 5. Choose **Yes** to delete the step from the method.

OR

Choose **No** to cancel the deletion.

Confirm	×
2	You are about to remove steps from your method. Do you wish to continue?
	<u>Y</u> es <u>N</u> o

Figure 12-13. Confirmation to delete steps in a method

### 12.7 Using Undo and Redo in Method Building

The Biomek Software maintains a history of actions performed during method building. By using the Undo and Redo features located on the toolbar, this history can be accessed and the actions undone or redone while building a method. Hovering the mouse over the Undo or Redo button on the toolbar displays a tool tip indicating what action will be undone or redone.

The following actions are stored in the history and may be undone and redone while building a method:

- Adding a step.
- Removing a step.
- Moving a step.
- Configuring a step.
- Changing active selection.

**Note:** The active selection is the currently selected (highlighted) step within the method. Selecting another step in the method changes the active selection.

### 12.7.1 Using Undo

Undo is used to undo the most recent method-building action. Multiple actions may be undone by performing consecutive Undo actions; there is no limit to how many actions can be undone.

To undo an action:

Select **Undo** from the Edit menu

OR

```
S)
```

C1

Click **Undo** on the toolbar

**Note:** When Undo is used while configuring a step, it undoes all actions made to the configuration. For example, while configuring Instrument Setup step and placing labware on deck positions, choosing Undo will undo the entire Instrument Setup step and not just the last deck position that was configured.

### 12.7.2 Using Redo

**Redo** is used to redo an action that has been undone using the **Undo** feature. By performing consecutive **Redo** actions, multiple actions can be redone if multiple actions were undone.

**Note:** If another action is performed after undoing an action, **Redo** is disabled and the previously undone action cannot be redone.

To redo an undone action:

• Select **Redo** from the Edit menu

OR

Click **Redo** on the toolbar

# 12.8 Entering and Viewing Method Properties

Method Properties allows for entry of a description of the method that can be viewed or modified.

**Note:** Method properties can be viewed, but not modified when running a method in validated mode (refer to Section 12.14, <u>Opening a Saved Method</u>). Validated mode is available only when Beckman Coulter Accounts & Permissions is enabled (refer to Chapter 2, <u>Using Accounts & Permissions</u>).

To enter or view Method Properties:

1. From the File menu, select **Properties**. Method Properties appears (Figure 12-14).

Method Properties		×
<u>A</u> uthor:		
BECKMANCOULTER		
Description:		
Enter a description of the method here.		
	ОК	Cancel

Figure 12-14. Method Properties

- 2. In Author, enter the name or user name of the method designer.
- 3. In Description, enter a description of the method.
- 4. Choose **OK** to save changes and close Method Properties.

OR

Choose Cancel to close Method Properties without saving changes.

### 12.9 Saving a Method

Methods may be saved at any time during their development. Saving a method only saves the current status of the method and does not create a revision or check in project items as revisions. If any of the project items, such as labware class definitions or techniques, changes after the method is saved, when the method is opened next, it uses the latest definitions.

Methods are saved as elements of the project file, not as separate files. To use a method in another project file or on another system, it must be exported to a Biomek method file (\*.bmf) and imported into the desired project (refer to Section 12.16, *Importing and Exporting Methods*).

To save a method:

1. Choose the **Save Method** icon on the toolbar.

OR

From the File menu, choose Save. Save Method appears (Figure 12-15).

**Note:** Save Method appears only the first time the method is saved; subsequent saves to the method do not prompt for a new file name. To save an existing method with a different file name, choose **Save As** from the File menu.

Save Method As		X
Enter method name:		
Method2		
ОК	Cancel	

Figure 12-15. Save method

- 2. In Enter method name, enter the name under which the method is saved.
- 3. Choose **OK**. The method is saved with the specified method name and **Save** Method closes. The title bar of the Biomek Software main editor displays the method name as MethodName [Development] (Figure 12-16).

OR

Choose Cancel. Save Method closes without saving the method.



Figure 12-16. Biomek Software main editor after saving method ExampleMethod

### 12.10 Checking In a Method

Checking in a method is similar to saving a method, but in addition creates a new revision of the method, checks in all project items of the project file, and creates new revisions for any items that have changed. A relational link is created between the method and all items in the project file.

**Note:** Methods must be saved once before being checked in. A prompt to save appears automatically if the method has not been saved before.

To check in a method:

 From the File menu, choose Check In. If any project items have changed since the last revision, Check In appears, listing all project items that have changed (Figure 12-17).

Check In			
Select the project items to	check in:		
Project Item	Change	Last Check In	Check In Time
🖃 🥙 Labware Classes			
– 🗹 Biomek2000HDRSt	🔊 Reverted	Berkeley Rattan	10/6/2003 2:44:22 PM
- 🗹 BiomekFXHDRCircu	🔊 Reverted	Berkeley Rattan	10/6/2003 2:44:22 PM
— 🗹 BiomekFXHDRStati	🔊 Reverted	Berkeley Rattan	10/6/2003 2:44:22 PM
🗏 🖳 FanShroud	🔊 Reverted	Berkeley Rattan	10/6/2003 2:44:22 PM
J			
	ОК	Cancel	

Figure 12-17. Check In

2. Select the project items to check in as new revisions. By default, all project items in Check In are selected.

**Note:** When a method is checked in, the title bar updates the method title to show the current revision of the method and that it is in **Development** mode. Methods can only be validated or opened in validated mode when Beckman Coulter Accounts & Permissions is enabled (refer to Section 12.11, <u>Validating a</u><u>Method</u>).

3. Choose **OK** to check in the new revisions of project items.

Note: If Accounts & Permissions is not enabled, check in is complete.

OR

Choose Cancel to close Check In without checking in the method.

4. If Accounts & Permissions is enabled, Check-In appears (Figure 12-18). In Reason, enter information or notes about the method or project revision being checked in. Text entered in Reason is date and time stamped, and stored in the audit log of all user activity.

5. If Accounts & Permissions is configured to require password checks for signing and check-in, enter the user **Password**.

Check-In		
User Name: BerkeleyR		
Checking In: Project (BiomekFX)		
Reason: New labware		
Password         Appears only when Accounts & Permissions is configured to         require password checks (refer to Section 2.2, <u>Installing and</u> Setting the Level of Support For Accounts & Permissions).		
Password:		
UK Cancel		

Figure 12-18. Check-In

6. Choose **OK** to check in the method or project. If project items were modified, only the project was checked in. Check-In appears a second time to check in the method.

OR

- 7. Choose **Cancel** to close Check-In without checking in the method or project.
- 8. Repeat steps 4 6 to check in the method, if necessary.

### 12.11 Validating a Method

When Beckman Coulter Accounts & Permissions is enabled, methods may be validated. A validated method is a revision of a method that is checked in, approved with an electronic signature, and protected from further modification. Revisions of project items required to run the validated method are also checked in and protected from further modification. This ensures that validated method runs are reproducible.

Validating a revision of a method is not the same as validating a method before a run to internally test for errors before the method is actually run. To validate a method to test for errors before it is run, check **Validate the current method before running it** in **Preferences** (refer to Section 29.2, <u>*Changing Display Preferences*</u>).

**Note:** Only users with Validate Methods permission can validate methods. Refer to Chapter 2, <u>Using Accounts & Permissions</u> for more information about Beckman Coulter Accounts & Permissions. Permissions can be configured so that a user may be permitted to run only validated methods (refer to Section 12.14, <u>Opening a Saved</u> <u>Method</u>).

To validate a method:

1. From the File menu, choose **Validate**. If any project items have changed since the last revision, **Check In** appears, listing all project items that have changed (Figure 12-17).

**Note:** Methods must be saved at least once before being checked in. A prompt to save appears automatically if the method has not yet been saved.

2. Select the project items to check in as new revisions. By default, all project items in Check In are selected.

**Note:** All project items used in the method must be checked in order for the correct revision of each item to be used in a validated method run.

 Choose OK to check in the new revisions of project items. Check-In appears (Figure 12-18).

OR

Choose **Cancel** to close Check In without checking in and validating the method.

- 4. In Reason, enter information or notes about the method or project revision being checked in. Text entered in Reason is date and time stamped, and stored in the audit log of all user activity.
- If Accounts & Permissions is configured to require password checks for signing and check-in, enter the user **Password** (refer to Section 2.2, <u>Installing and</u> <u>Setting the Level of Support For Accounts & Permissions</u>).
- 6. Choose **OK** to check in the method or project. If project items were modified, only the project was checked in. Check-In appears a second time to check in the method. Repeat steps 4 6.

OR

Choose Cancel to close Check-In without checking in or validating the method.

7. After the project and method are checked in, Signature appears (Figure 12-19). In Reason, enter information or notes about the method being validated. Text entered in Reason is date and time stamped, and stored in the audit log of all user activity.

Signat	ure	
User I	User Name: BerkeleyR	
Signin	ıg:	Method-1A
Reaso	on:	Method is approved for use.
Password Appears only when Accounts & Permissions is configured to require password checks (refer to Section 2.2, <u>Installing and</u> <u>Setting the Level of Support For Accounts &amp; Permissions</u> ).		
Passv	vord:	******
		OK Cancel

Figure 12-19. Signature

- 8. If Accounts & Permissions is configured to require password checks for signing and check-in, enter the user **Password** (refer to Section 2.2, *Installing and Setting the Level of Support For Accounts & Permissions*).
- 9. Choose **OK** to validate the method.

OR

10. Choose **Cancel** to close Signature without validating the method.

**Note:** After validating a method, it remains in development mode. To use the method in validated mode, it must be closed and then opened in validated mode. Refer to Section 12.14, *Opening a Saved Method* for more information about opening methods in validated mode.

# 12.12 Signing a Method

When Beckman Coulter Accounts & Permissions is enabled, methods that are checked in or validated may be signed. Signing allows users to add comments with an electronic signature to the audit trail for the method open in the editor. Signing is useful when laboratory policies require personnel to review and sign off on procedures; for example, a supervisor could approve a new validated method via the electronic signature generated by signing the method in Biomek Software.

**Note:** Refer to Chapter 2, <u>Using Accounts & Permissions</u> for more information about Beckman Coulter Accounts & Permissions.

To sign a method:

- Ensure the method has been checked in (refer to Section 12.10, <u>Checking In a</u> <u>Method</u>).
- 2. From the File menu, choose Sign. Signature appears (Figure 12-20).

Signat	ure		
User I	Name:	ame: BerkeleyR	
Signin	ig:	Method-1A	
Reas	on:	Method is approved for use.	
Password Appears only when Accounts & Permissions is configured to require password checks (refer to Section 2.2, <u>Installing and</u> <u>Setting the Level of Support For Accounts &amp; Permissions</u> ).			
Passv	vord:	OK Cancel	

Figure 12-20. Signature

- 3. In Reason, enter information or notes about the method being signed. Text entered in Reason is date and time stamped, and stored in the audit log of all user activity.
- If Accounts & Permissions is configured to require password checks for signing and check-in, enter the user **Password** (refer to Section 2.2, <u>Installing and</u> <u>Setting the Level of Support For Accounts & Permissions</u>).
- 5. Choose **OK** to sign the method.

OR

6. Choose **Cancel** to close Signature without signing the method.

## **12.13 Viewing Method History**

Revision and run history for the current method can be viewed at any time. Method history is viewed in two windows:

- History view all checked in and validated revisions of the current method (refer to Section 12.13.1, <u>Viewing Revision History</u>).
- Validated Run History view information about each validated run of the current method (refer to Section 12.13.2, <u>Viewing Validated Run History</u>).

**Note:** Validated methods and Validated Run History are available only when Beckman Coulter Accounts & Permissions is enabled (refer to Chapter 2, <u>Using</u><u>Accounts & Permissions</u>).

#### 12.13.1 Viewing Revision History

A revision history that tracks each time the current method has been checked in or validated can be viewed when the method is open.

**Note:** Methods can be validated only when Beckman Coulter Accounts & Permissions is enabled (refer to Chapter 2, <u>Using Accounts & Permissions</u>).

To view the history for a method:

1. From the File menu, choose **History**. History appears (Figure 12-21).

**Note:** Revision history details are listed in a tree structure that lists information by run and type. Branches may be expanded or collapsed using the + and - buttons to the left.

The Changed By and Comment columns appear only when Accounts & Permissions is enabled.

History						
History of Method-1A:						
		#	Change	Changed By	Time of Change	Comment
	•	2	🖌 Checked In	🔦 Berkeley Rattan	10/7/2003 1:46:39 PM	Added addtional labware.
			🔦 Validated	🔦 Marcia Q. Madsen	10/7/2003 1:49:01 PM	Passed testing.
		1	🖌 Checked In	🛐 Berkeley Rattan	10/6/2003 4:01:56 PM	New & Improved
			🔦 Validated	🛐 Berkeley Rattan	10/6/2003 4:02:35 PM	New & Improved
/						

Figure 12-21. Method history

2. When finished viewing the history, choose **OK** to close History.

### 12.13.2 Viewing Validated Run History

Validated methods run in validated mode generate a run history that can be viewed when the method is open. A run history includes access to all dialog alerts, log reports, worklists, and other files associated with or generated by the method during the run.

Many of the files listed in the run history can be viewed, saved, and printed (refer to Section 12.13.2.1, *Viewing, Saving, and Printing Run History Details*).

**Note:** Validated methods and their run history are available only when Beckman Coulter Accounts & Permissions is enabled. Refer to Section 2, <u>Using Accounts & Permissions</u> and Section 12.15, <u>Checking Out a Method</u>.

**Note:** Text files, such as \*.csv files used by the Worklist, Define Pattern, and Transfer From File steps, appear in the run history and may be opened directly from the window. However, when a project or validated method is exported, they are excluded from the Biomek import (\*.imp) or method (\*.bmf) file. To use a \*.csv file in a method exported to another system, it must be copied to the system along with the Biomek import or method file.

To view the validated run history for a method:

1. From the File menu, choose **Run History**. Validated Run History appears (Figure 12-22).

**Note:** Run history details are listed in a tree structure that lists information by run and type. Branches may be expanded or collapsed using the + and - buttons to the left.

Validated Run History					
Runs for Method-1A					
Run Information	User	Comment			
E- ▶ 🛃 2	🔦 Berkeley Rattan	Added addtional labware.			
È- ▶ 10/7/2003 1:59:36 PM	ឮ Marcia Q. Madsen				
🔁 📂 Dialog					
🗌 🗌 🕞 The left pod should hav	ឮ Marcia Q. Madsen	ок			
🗄 📂 Log					
Errors	🔦 Marcia Q. Madsen				
1	ឮ Berkeley Rattan	New & Improved			
Run history details	ОК				

Figure 12-22. Validated Run History displayed in validated mode

2. When finished viewing the history, choose **OK** to close History.

#### 12.13.2.1 Viewing, Saving, and Printing Run History Details

Many of the files listed in the run history details can be viewed, saved, and printed.

To view and print log and other text files:

Right click on the desired file and choose **Open** from the menu. The file opens in **Print Preview** with printing and viewing options (Figure 12-23).

To save a file listed in the run history:

<b>Note:</b> Dialog details cannot be viewed or saved.	
--	--

- 1. Right click on a file and choose **Save** from the menu that appears.
- 2. Save files in text format by appending **.txt** to the end of the filename for best file compatibility.

🏫 Print Preview	
Print Q Q T5% Change view magnification	⊆lose
Zoom in and out Errors	<u>•</u>
Method = Method-1A Logged in user = REGERBIG Started 10/07/2003 13:59:37 Unit serial number = Podl head serial number = <unknown> Wo validation date. 10/07/2003 13:59:42,Run ended.</unknown>	
	► //.

Figure 12-23. Print Preview

### 12.14 Opening a Saved Method

Methods saved in a project file can be opened only in that project file. Opening a method using the **Open** command opens it in development mode using the current revision for all project items. A revision of a method can be opened in read-only mode using the **Check Out** command (refer to Section 12.15, <u>Checking Out a</u> <u>Method</u>).

**Note:** Methods may also be opened from **Recent Methods** on the **File** menu. A deleted method may be opened from the **Recycled** folder. Opening a deleted method automatically restores it.

To open a saved method:

1. Choose **Open Method** on the toolbar

OR

From the **File** menu, choose **Open**. Open Method appears, listing all methods saved in the project file (Figure 12-24).

- 2. Right-click on a method and select **History** to view the history for the selected method.
- 3. Use Check Out (refer to Section 12.15, <u>*Checking Out a Method*</u>) to open a specific revision of a method in read-only mode.
- 4. Choose **View** and select the desired option, **Icon**, **List**, or **Details**, to change how methods are displayed in **Open Method**. **Icon** is the default view selection.

Open Method						
	Select a meth	nod to open:				View 🝷
My Methods						<b>_</b>
<b>S</b>	Advanced	Aspirate1	Method 1	Method-1A	Method1	
Recycled		C)	. CU	C)	1 CU	
	Method10	Method11	Method12	Method13	Method14	
	Method2	Method3	Validate Available only & Permissions	<b>d Methods</b> y when Accou s is enabled.	nts ethod5	
						T
	Method Name:	Method 1				OK
	Show Methods:	All Methods			•	Cancel
	🔽 Open metho	d in validated r	mode	J		

Figure 12-24. Open saved methods

5. Select the desired method to open.

**Note:** When Accounts & Permissions is enabled, validated methods are denoted by a circled check (Figure 12-24). Only validated methods appear in Open Method for users that only have Run Validated Methods permission.

Users with Develop Methods permission have the option to Open method in validated mode. Only previously validated methods can be opened in validated mode. Methods opened in validated mode are locked and cannot be modified.

6. Choose **OK**. The selected method is opened in the main editor.

OR

Choose Cancel to close Open Method without opening a method.

# 12.15 Checking Out a Method

Methods that have been checked in may be checked out. Checking out a method allows any revision of the method to be opened in read-only mode. The links created when the method was checked in allows Biomek Software to open the method with the same project items as when it was checked in. A method revision that is checked out may also be exported to a Biomek method file for use in another project or on a different system (refer to Section 12.16.1, *Exporting a Method*).

**Note:** Validated methods may be checked out in validated mode when Accounts & Permissions is enabled.

To check out a method:

1. From the File menu, choose **Check Out**. Check Out Method appears (Figure 12-25).

Check Out Method	d	
	Select a method to open:	View 🝷
My Methods		
	Method1 Method1b Method3 Transfer 4a	
Recycled	Validated method options and Changed by are visit only when Accounts & Permissions is enabled.	ole
		_
	Select a revision to check out:	
	# Changed By Time of Change Comment	
	2 2 Berkeley 10/22/2003 3:47:34 PM added more laby	vare
	1 🖸 Berkeley /10/21/2003 2:14:34 PM initial design	
	Method Name: Method1	ок
	Show Methods: All Methods Ca	ancel

Figure 12-25. Check Out Method (Accounts & Permissions enabled)

2. In Select a method to open, select the desired method. All revisions of the selected method appear in Select a revision to check out.

**Note:** When Accounts & Permissions is enabled, validated methods are denoted by a circled check (Figure 12-25). Only validated methods appear in Check Out Method for users that only have Run Validated Methods permission.

Users with Develop Methods permission also have the option to Open method in validated mode. Only previously validated methods can be checked out in validated mode. When methods are checked out in validated mode, Accounts & Permissions generates the audit log during method runs.

- 3. In Select a revision to check out, select the desired revision.
- 4. Choose **OK** to check out the selected revision of the method.

OR

Choose Cancel to close Check Out Method.

### **12.16 Importing and Exporting Methods**

Methods can be freely shared between projects or transferred between computers by exporting and importing Biomek method files. A method file is created by exporting a method from the active project file, and contains the method and all records required to execute the method. This allows the method to execute correctly when the method file is imported into a different project.

**Note:** When Beckman Coulter Accounts & Permissions is enabled, users with the Run Validated Methods or Develop Methods permission may export methods, but only those with the Develop Methods permission may import methods into a project (refer to Chapter 2, <u>Using Accounts & Permissions</u>).

**Note:** Projects may also be imported and exported. Refer to Section 6.10, *Importing and Exporting Project Files*, for more information.

This section covers:

- *Exporting a Method* (Section 12.16.1).
- <u>Exporting All Methods Associated With a Project File</u> (Section 12.16.2).
- <u>Importing Methods</u> (Section 12.16.3).
#### 12.16.1 Exporting a Method

The method currently open in Biomek Software can be exported to a Biomek method file (\*.bmf). The method file contains all records necessary to allow the method to be imported into another project file and executed.

**Note:** Method files exported from Biomek 3.0 or later are not compatible with earlier versions of the Biomek FX software.

**Note:** When Accounts & Permissions is enabled, users with Run Validated Methods or Develop Methods permission may export the currently open method. However, when validated methods are exported and later imported into different projects, they are no longer considered validated and open in development mode. After being imported, they can then be validated by users with Validate Methods permission.

**Note:** Text files, such as \*.csv files used by the Worklist, Define Pattern, and Transfer From File steps, appear in the run history and may be opened directly from the window. However, when a project or validated method is exported, they are excluded from the Biomek import (\*.imp) or method (\*.bmf) file. To use a \*.csv file in a method exported to another system, it must be copied to the system along with the Biomek import or method file.

To export the currently open method:

1. From the File menu, choose **Export**. Export Method appears (Figure 12-26).

Export Metho	d	<u>?</u> ×
Save in: 🔂	Biomek 💌 🗢 🗈 💣 🎫	
() Method 01	a.bmf	
, File name: Save as type:	Method 02a Save Biomek® Method Files (*.bmf)	2

Figure 12-26. Export Method

- 2. In Save in, browse to the desired location to save the exported Biomek method file.
- 3. In File name, enter a name for the method file.
- 4. Choose **Save** to export and save the method file.

OR

Choose **Cancel** to close Export Method without exporting the method.

#### 12.16.2 Exporting All Methods Associated With a Project File

All methods associated with the currently open project file may be exported at the same time. Method files are exported to a folder selected or created by the user.

**Note:** When Accounts & Permissions is enabled, users with Run Validated Methods or Develop Methods permission may export the currently open method. However, when validated methods are exported and later imported into different projects, they are no longer considered validated and open in development mode. After being imported, they can then be validated by users with Validate Methods permission.

To export all saved methods:

1. From the File menu, choose **Export All**. Browse For Folder appears (Figure 12-27).

Browse For Folder	? ×
Select a path for the exported methods:	
Desktop My Documents This Computer Coal Disk (C:) Compact Disk (D:) Compact Disk (F:) Compact Disk (F:)	•
Folder: Local Disk (C:)	
OK Cancel New Fo	lder //

Figure 12-27. Browse For Folder

2. Use the tree view to browse to the folder where the exported method files will be saved.

OR

Choose **New Folder** to create a new folder for the method files. The new folder appears in the tree view with the name highlighted.

**Note:** New Folder is available only when Biomek Software is run on a Windows 2000 or Windows XP system.

3. If a new folder has been created, enter a new name for it.

OR

To change the name of an existing folder, select the folder, click on the folder name a second time, and enter the new name.

4. Choose **OK** to export all methods to the selected folder.

OR

Choose **Cancel** to stop the operation without exporting any methods.

## 12.16.3 Importing Methods

Biomek method files previously exported from a project file can be imported into the project file currently active. Importing a method file also imports all the necessary records to execute the method.

Multiple methods can be selected and imported simultaneously.

To import methods into the project file:

1. From the File menu, choose Import. Import Method appears (Figure 12-28).

Import Meth	od	? ×
Look in: 🕞	Methods 💽 🗢 🗈 📸 🖽	•
🕼 NX Multich	annel 10.bmf	
📲 🖗 NX Transfe	er 01.bmf	_
📲 🖗 NX Transfe	er O1a.bmf	_
		_
1		
File name:	0r	en
Eller of hereit		
Files of type:		

Figure 12-28. Import Method

- 2. In Look in, browse to the location where the desired Biomek method file(s) is saved.
- 3. Select the desired method file(s) to import. To import several method files, hold the **CTRL** key while selecting each method desired to import.
- 4. Choose **Open** to import the selected method file(s). **Confirm** appears (Figure 12-29).

OR

Choose **Cancel** to close Import Method without importing the method file.

Confirm		×
2	Do you wish to import "NX Multichannel 10.bmf" into the current project?	
	<u>Y</u> es <u>N</u> o	

Figure 12-29. Confirming the method file(s) to import

5. Choose **Yes** to confirm the desired method file is selected. Confirm appears (Figure 12-30).

OR

Choose **No** to cancel the import operation without importing the method file.

Confirm	×
Q	Do you wish to import the corresponding project items for each method?
	<u>Y</u> es <u>N</u> o

Figure 12-30. Choosing to import project items corresponding to the method

6. Choose Yes to import the method and project items used by the method, but not currently stored in the active project file. Check In Project appears, listing only project items used by the method that do not currently exist in the current project. These project items will be imported into the current project and linked to the method.

**Note:** Check In Project may not appear if all project items in the current project file are compatible with the imported method.

Items in the current project that are not used by the imported method are not linked to the method.

OR

Choose **No** to import the method but not the project items. No project items are linked to the method. The import operation is complete.

**Note:** Importing project items used by the method is recommended. Choosing No may prevent the imported method from executing properly.

Check In Project				
Select the project items to check in:				
Project Item	Change	Last Check In	Check In Time	
🖃 🗹 🤌 Labware Classes				
AB384WellReactio	🛨 New			
- 🗹 AP384_30uL	🖉 Restored		2/3/2004 1:29:07 PM	
🖵 🔽 Greiner384Lid	Modified		1/21/2004 5:40:14 PM	
	ОК	Cancel		

Figure 12-31. Check In Project

7. Choose **OK** to check in the revisions of project items associated with the method. If Accounts & Permissions is enabled, Check-In appears (Figure 12-32).

**Note:** If Accounts & Permissions is not enabled, importing the method is complete.

OR

Choose **Cancel** to close Check In without checking in the project items.

- 8. In Reason, enter information or notes about the method or project revision being checked in. Text entered in Reason is date and time stamped, and stored in the audit log of all user activity.
- 9. If Accounts & Permissions is configured to require password checks for signing and check-in, enter the user **Password**.

Check-In					
User Name:	BerkeleyR				
Checking In:	Project (BiomekFX)				
Reason:	New labware				
Only requir <u>Settin</u> Password:	Password     appears when Accounts & Permissions is configured to     re password checks (refer to Section 2.2, Installing and     g the Level of Support For Accounts & Permissions).				

Figure 12-32. Check-In

Choose OK to check in the project items associated with the imported method(s).
OR

Choose **Cancel** to close Check-In without checking in project items associated with the imported method.

# 12.17 Deleting and Restoring Methods

Methods may be deleted from a project file, but are not removed from the project file and may be restored at any time.

## 12.17.1 Deleting Methods

To delete a saved method:

1. Choose **Open Method** on the toolbar.

OR

From the **File** menu, choose **Open**. Open Method appears, listing all methods saved in the project file (Figure 12-33).

- 2. Right-click on a method and select **History** to view the history for the selected method.
- 3. Choose **View** and select the desired option, **Icon**, **List**, or **Details**, to change how methods are displayed in Open Method. Icon is the default view selection.



Figure 12-33. Open saved methods

4. Right-click the desired method to delete and select **Delete**. A confirmation appears (Figure 12-34).



Figure 12-34. Confirm deletion of selected method

5. Choose **Yes** to confirm deletion of the selected method. The method is removed from My Methods and placed in Recycled.

OR

Choose **No** to keep the selected method.

## 12.17.2 Restoring Methods

To restore a deleted method:

1. Choose **Open Method** on the toolbar.

OR

From the **File** menu, choose **Open**. Open Method appears, listing all methods saved in the project file (Figure 12-33).

2. Right-click on a method and select **History** to view the history for the selected method.

Open Method						
	Select a meth	iod to open:				View 👻
My Methods		1			C)	<b>_</b>
Ĩ	Advanced	Aspirate1	Method 1	Method-1A	Method1	
Recycled		C)	1.C	100	C)	
	Method10	Method11	Method12	Method13	Method14	
		C)	1	1	C	
	Method2	Method3	Method4	Method4a	Method5	
		1		1	C.	-
	Method Name:	Method 1				ОК
	Show Methods:	All Methods			•	Cancel
	🔽 Open metho	d in validated m	iode			

3. Choose **View** and select the desired option, **Icon**, **List**, or **Details**, to change how methods are displayed in Open Method. **Icon** is the default view selection.

Figure 12-35. Open saved methods

4. Choose **Recycled** on the left side of **Open Method** to display all the methods that have previously been deleted (Figure 12-36).

5. Choose **View** and select the desired option, **Icon**, **List**, or **Detail**, to change how methods are displayed in Open Method. **Icon** is the default view selection.

Open Method			
Open Method My Methods	Select a meth Method4	od to open:	View •
	Method Name:	Method4	OK. Cancel

Figure 12-36. Recycled methods

6. Right-click the desired method to restore and choose **Restore**. The method is moved from **Recycled** to **My Methods**.

**Note:** A method may be opened directly from the **Recycled** folder by selecting the method and choosing **OK**. A confirmation appears stating that the method will be restored prior to opening the method. Choose **OK** to restore and open the method.

## 12.18 Running, Pausing, and Stopping a Method

Methods may be run on the Biomek instrument or in simulation mode immediately after they are built. If desired, methods may be configured to check for internal errors and confirm that labware on the instrument deck matches the layout specified in the method before every run.

This section covers:

- <u>Running a Method</u> (Section 12.18.1).
- <u>Pausing a Method in Progress</u> (Section 12.18.2).
- <u>Stopping a Method in Progress</u> (Section 12.18.3).
- <u>Snapping a Continuation (FX, NX only)</u> (Section 12.18.4).
- <u>Disabling Steps Within a Method</u> (Section 12.18.5).
- <u>Purging Air From Span-8 Pod Before a Method Run (FX and NX-S8 Only)</u> (Section 12.18.6).

#### 12.18.1 Running a Method

After a method has been built, it may be run immediately on the Biomek instrument or in Simulate mode. In simulate mode, the software runs the method and checks for configuration errors in each step. While the method is running, the Biomek Simulator shows an animated 3-D model of the instrument performing the method.

When simulate mode is chosen, validated methods run in validated mode are recorded in Run History. Validated methods run in development mode are not logged. Validated methods are only available when Accounts & Permissions is enabled.

**Note:** When Beckman Coulter Accounts & Permissions is enabled, only users with Develop Methods or Run Validated Methods permission may run methods. Users with Develop Methods permission may run any method saved on the system; users with Run Validated Methods permission may only run validated methods. Refer to Section 2.3.1, <u>Assigning Biomek Software Permissions</u> for more information about the permissions available to users of Biomek Software.

To run a Biomek method:

- 1. From Hardware Setup in Port, choose:
  - Simulate to run the method in simulated mode.

OR

- Com1 to run the method on the instrument.
- 2. If running the method on the instrument, from the Instrument menu, choose **Home All Axes** to home the pod(s), if necessary. The pod(s) must be homed before performing methods each time the instrument is powered on.

**Note:** Refer to the instrument hardware manual for more information about homing the pod(s).

3. To start the method run, on the toolbar, click the **green arrow** button.

OR

From the Execution menu, choose Run.

The method run begins. Depending on how Biomek Software is configured, two events may precede the execution of steps in the method:

 If Validate Before Run is selected in Options>Preferences, the method is validated internally to check for errors (refer to Section 29.2.1, <u>Configuring</u> <u>General Options</u>).

**Note:** Validating the method for errors is a different process from validating a revision of a method when Accounts & Permissions is enabled. Refer to Section 12.11, *Validating a Method* for more information about validating a revision of a method.

• If Pause to confirm setup? is selected in the Instrument Setup step, a deck confirmation prompt appears (Figure 12-37).



CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height and failure to choose each ALP correctly in the software may result in collisions between pod(s) and ALPs during operation.

To confirm that the deck setup is correct:

1. Visually confirm the deck setup on the instrument matches the display.

Biomek® Software
P1 P3
P6 P7
The left and should have up the landed
i ne iert pod snouio nave no tips loaded.
Does the Biomek® Software deck match the above layout, including the labware and their locations?
If yes, choose OK to continue the method. If no, choose Abort to stop the method.
OK     Abort       9/24/2003 6:16:47 PM

Figure 12-37. Deck layout confirmation prompt for a Biomek 3000 method

2. Choose **OK** if the setup is acceptable.

OR

Choose **Abort** to close the deck confirmation prompt and cancel the method run.

**Note:** To set up the deck correctly, either set up labware on the instrument deck to match the layout displayed in the deck confirmation prompt, or change the Deck Layout in the Instrument Setup step (refer to Section 15.2.4, *Populating the Deck with Labware*).

The method runs as soon as the deck setup is confirmed. The method run can be visually followed in the Method View; steps are highlighted in green as they are executed.

When a method runs in simulate mode, the Biomek Simulator appears and shows a 3-D animation of the method being performed (refer to Section 12.18.1.1, <u>Viewing the</u> <u>Method Run In the Biomek Simulator</u>).

**Note:** If Accounts & Permissions is configured to automatically log off users after a set period of operating system inactivity, the possibility exists that a user may be logged off automatically during a method run. When a user is logged off, the method run continues, but any prompts, such as the deck layout confirmation (Figure 12-37), are hidden from view until the same user, or another user with permission to run methods, logs on.

#### 12.18.1.1 Viewing the Method Run In the Biomek Simulator

When a method is run in simulate mode, the Biomek Simulator appears and shows an animated 3-D model of the instrument performing the method (Figure 12-38).



Figure 12-38. The Biomek Simulator showing a method run on an NX-MC

Five methods of controlling the simulator view are available:

	=			
1.5	_	9		
i.e	-	ø	ı	

From the toolbar, choose **Simulation Display** to toggle the Biomek Simulator on and off while the method is executing.

**Note:** The simulator automatically launches when the method run begins and closes when it is finished. If the animation does not run smoothly, try reducing the image quality setting in Hardware Setup (refer to Section 4.4.1, *Configuring Simulator Settings*).

- Change the view by placing the cursor in the simulator display, then clicking and dragging the mouse in the desired direction of rotation.
- Change the view by clicking the cursor in the simulator display and using the keyboard control keys defined in Camera (refer to Section 4.4.2, <u>Configuring Camera Controls</u>).
- Use the camera navigation controls to change, save, and reset the view (Figure 12-39). The navigation controls are displayed when enabled in Hardware Setup or when toggled on using the simulator controls (refer to Section 12.18.1.1.1, <u>Using the Simulator Controls</u>).

**Note:** Additional simulator parameters may also be configured in Hardware Setup (refer to Section 4.4, *<u>Configuring the Biomek Simulator</u>*).

Press the tilde (~) key to access the simulator controls (refer to Section 12.18.1.1.1, <u>Using the Simulator Controls</u>). The simulator controls provide the ability to change the simulation playback speed, toggle the camera navigation controls on and off, and modify the lighting that illuminates the simulated instrument.



Figure 12-39. Camera navigation controls

#### 12.18.1.1.1 Using the Simulator Controls

The simulator controls provide the ability to change the simulation playback speed, toggle the camera navigation controls on and off, and modify the lighting that illuminates the simulated instrument.

To access and use the simulator controls:

1. During a simulated method run, press the tilde (~) key to toggle the simulator controls on. The simulator controls appear (Figure 12-40).



Figure 12-40. Biomek Simulator with all simulator controls enabled

- 2. Enter a new **Warp Factor** and choose **Apply** to change the simulator playback speed, if desired. Warp Factor 1 runs the simulation in real time, Warp Factor 2 at twice normal speed, and so on.
- 3. Choose Lights to toggle the lighting controls on, if necessary.
- 4. Create a new light or choose an existing light from the pull-down menu, and use the lighting controls to modify the lighting properties as desired. Table 12-1 provides a description of each lighting control.

**Note:** Modifying the lighting parameters is not recommended because the default parameters cannot be restored if changed or deleted.

- Choose Navigation to toggle the camera navigation controls on, if necessary. Refer to Section 12.18.1.1, <u>Viewing the Method Run In the Biomek Simulator</u>, for detailed information about using the navigation controls.
- 6. Press the tilde ( $\sim$ ) key to toggle the simulator controls off, if desired.

Control	Description	Additional Information		
<b>Global Light Properties</b>				
Local View	Select to subtly change the intensity of light.			
Global Ambient	Choose the color of ambient light present in the simulator view.			
Lights	·			
(new light)	Create a new light.			
(delete light)	Delete the light currently selected in the pull-down menu.	A light that has been deleted cannot be restored.		
(save light)	Save the configuration of the light currently selected in the pull-down menu.			
Draw Lights	Select to display the light sources in the simulator view.			
Properties	•			
Name	Change or enter a new name for the selected light.			
Enabled	Select to enable (turn on) the selected light.			
Туре	Choose the type of light: <b>Positional</b> or <b>Directional</b> .	Positional light radiates from the source in all directions; directional light radiates from the source in a single direction.		
Direction	Enter X, Y, and Z coordinates for the direction from which the selected light is directed at the instrument.			
Light	Choose colors for the <b>Ambient</b> , <b>Diffuse</b> , and <b>Specular</b> properties of the light.	Ambient and Diffuse define the color of the 3-D model of the instrument when the light is enabled. Specular defines the color of specular highlights (glare).		
Attenuation	Enter values in each field to define the fall off from the center of the light	Available only when <b>Positional</b> is the type of light selected.		
	ingin.	Entering higher values causes the light to fall off quicker.		
Spotlight	Choose to cast a tightly focused beam of light at the 3-D model of the instrument.	Available only when <b>Positional</b> is the type of light selected.		
Spot Direction	Enter X, Y, and Z coordinates for the direction from which the spotlight is directed at the instrument.	Available only when Spotlight is the type of light selected.		

Table 12-1.	Simula	tor Ligl	hting	Control	ls
-------------	--------	----------	-------	---------	----

Control	Description	Additional Information
Spot Exponent	Enter a value to set the focus of the spotlight.	Available only when Spotlight is the type of light selected.
		A low value distributes the light evenly across the illuminated area; a high value concentrates more light in the center of the spotlight.
Spot Cutoff	Enter a value to set the size of the spotlight cone, which determines the total surface area covered by the spotlight.	Available only when Spotlight is the type of light selected.

## Table 12-1. Simulator Lighting Controls

## 12.18.2 Pausing a Method in Progress

Pause causes the method to halt after the Biomek instrument has completed the move in progress. For example, if the instrument is in the process of loading tips when Pause is selected, it finishes loading the tips before the system halts. When the method is resumed, it continues as though it were never halted.

**Note:** A method cannot be edited during a pause.

Use one of the following procedures to pause a method:



#### CAUTION: No changes to the Biomek state are permitted while a method is paused. Changes can be made to the labware contents, but not the deck or the devices.

From the Execution menu, choose Pause.

OR



Choose **Pause** on the toolbar. The instrument completes the move in progress and stops.

FX, NX-MC, NX-S8 — The instrument deactivates the light curtain when the method is paused.

To resume the method run, choose **Pause** again. The light curtain is reactivated, and the method resumes from the point where it was paused.

#### 12.18.3 Stopping a Method in Progress



# WARNING: The light curtain is a safety device. Use it to stop a method only in an emergency.

Use **Stop** to halt a method during its run when there is no intent to resume method execution. If the pod is in the process of a move, the operation is not completed. Since the method is halted, this option allows edits to the method or changes to the hardware and deck.

**Note:** Stop may not halt operations already in progress on external devices, such as Stacker Carousels.

Use one of the following procedures to stop a method during a run:

• From the Execution menu, choose **Stop**.

OR

• Choose **Stop** on the toolbar.

OR

- FX, NX-MC, NX-S8 Break (violate) the light curtain by no more than one inch, and choose Abort.
- > **3000** Press the STOP button on the front of the instrument

**Note:** Use standard procedures to rerun the method.

#### 12.18.4 Snapping a Continuation (FX, NX only)

Another way to halt a method in progress is by snapping a Continuation. Continuations stop the current method and allow corrections in the middle of a method run. The method can be restarted from the point where the method was halted. For more information about Continuations, refer to Section 25.11, <u>Using</u> <u>Continuations (not available on a 3000)</u>.

To snap a continuation, choose **Snap** to create a Continuation (refer to Section 25.11, *Using Continuations (not available on a 3000)*).

## 12.18.5 Disabling Steps Within a Method

Individual steps within a method can be disabled for execution. When the method is run, Biomek Software ignores any disabled steps.

Disabling steps is useful for experimenting with the configuration of certain steps during method development, or when only a subset of the steps in a method need to be executed in a method run for an assay.

**Note:** If a method is checked out in validated mode, steps cannot be disabled or enabled.

To disable a step in a method:

- 1. Select the desired step to disable.
- 2. From the Edit menu, choose **Disable step**.

OR

Right-click the step and choose **Disable step** from the menu. The step icon changes to grayscale with a red X to indicate the step is disabled.





To enable a previously disabled step:

- 1. Select the disabled step to enable.
- 2. From the Edit menu, choose Enable step.

OR

Right-click the step and choose **Enable step** from the menu. The red X disappears and the step icon changes to color to indicate the step is enabled.

## 12.18.6 Purging Air From Span-8 Pod Before a Method Run (FX and NX-S8 Only)

When a Span-8 Pod is idle, air bubbles form in the tubing and syringes and interfere with pipetting accuracy. These air bubbles may form prior to beginning a method run or while the pod is sitting idle during a method run. Therefore to accurately transfer liquid, air *must* be purged from the tubing and syringes of the Span-8 Pod. The purging process draws system fluid through the tubing and syringes until all air bubbles are removed.

Purging the tubing and syringes of air is accomplished through Manual Control (refer to Section 6.5.7, *Purging Air from the Syringes and Tubing*) or by inserting a Span-8 Wash Tips step (refer to Section 18.8.1, *Configuring the Span-8 Wash Tips Step for a Passive Wash*) in a method before the Span-8 Pod moves:

- Use Manual Control immediately before running a method or insert a Span-8 Wash Tips step at the beginning of the method to purge the tubing and syringes of air.
- A Span-8 Wash Tips step using approximately 10 to 15 mL of system fluid may be inserted after the Instrument Setup step to purge the tubing and syringes of air.
- When using a dual-pod system, it may be necessary to insert another Span-8 Wash Tips step into the method immediately before a Span-8 Pod begins to move if the pod has been idle.

# 12.19 Printing a Method

Printing a hard copy of a method is useful if a computer monitor is some distance from the instrument, it is necessary to show the method steps to someone away from the lab, or a hard-copy record of the method is needed to recreate the method. All of the step configurations are listed in the hard copy.

To print a Biomek method in sequential text form:

- 1. Select the **File** menu.
- 2. Choose **Print**. A print-out similar to Figure 12-42 is produced at the default printer.

**Note:** Use **Print Preview** from the File menu to open a dialog which displays how the method appears when printed.

	C:\Program Files\Biom	ek FX\Methods\Tutorial5.bmt
hod hor:		
ine the following ispense = 25 spirate = 10	values for this method	1:
cription:		
trument Setup		
k: Deckl		
se to confirm lay ify that the pod	out. is set up in its defaul	t configuration.
ms:		
Nothing : Nothing		
Nothing		
: Nothing : Nothing		
Nothing		
Nothing		
: Nothing		
: Nothing		
: Nothing		
Nothing		
Nothing		
Reservoir with 1	00000. uL of Water	
BCFlat95 with an	unknown amount of an u	inknown liquid.
Nothing		
Nothing		
Nothing		
Nothing · MDG6 200ml tim :	or Heathara time 0 ti	
. AFFD_200002 cip .		
nsfer		
ng pod "Podl", ex	ecute the following tra	insfer:
m: "Revr" using th	he following technique:	
se the following ;	pipetting template: Def	fault Template
lowout: frue	. 0	
alibration Slope:	1	
inimum Pipetting 1	feight: .2 mm	
rewet: False	-	
ollow Liquid: Fal	se	
eight: 1 mm from '	the bottom	
iz frue	100*	
in Appirate Speed	: 100%	
ix Dispense Speed	: 100*	
ix Dispense Heigh	t: 2 mm from the bottom	n
ix Count: 3		
ix Volume: 50 uL		
peration speed: 1	00%	
ip Touch: True		
railing Air Gap:	0 uL	

Figure 12-42. Printed method

## 12.20 Improving a Method

Biomek provides procedures that allow a method to be observed in order to make adjustments to improve the way it runs.

Here are three procedures to improve the way a method runs:

- <u>Running the Method at Slow Speed</u> (Section 12.20.1).
- <u>Snapping a Continuation (FX and NX Only)</u> (Section 12.20.2).
- <u>Using Logging</u> (Section 12.20.3).

#### 12.20.1 Running the Method at Slow Speed

To slow down the pod speed to provide more time to examine the method in progress:

- 1. From the Instrument menu, choose **Hardware Setup**. Hardware Setup appears (Figure 12-43).
- 2. On the left side of Hardware Setup, select the desired pod.
- 3. Expand the Additional Pod Settings.
- 4. Set the Speed Limit to **20%**. This is the recommended speed for observing a method as it runs (Figure 12-43).

Figure 12-43. Changing the method speed in Hardware Setup

- 5. Choose **Accept** to save the changes and close **Hardware Setup**.
- 6. From the Execution menu, choose **Run** to start the method.

**Note:** Another way to run the method slowly is to use Single Step on the Execution menu. Single Step allows the unit to move one operation at a time by clicking the launch button for each move. Refer to Section 25.12, *Performing Single Operations Within Steps*, for more information about using Single Step.

#### 12.20.2 Snapping a Continuation (FX and NX Only)

Possible changes in a method may be desired after observing a method. Snapping a Continuation stops the method in progress, allows settings to be changed, and then continues the run. Another option is to snap a Continuation after the run has been observed at slow speed to return the speed to 100%. This causes the first part of the method to run slowly to observe the execution, but the rest of the method runs at full speed.

**Note:** Refer to Chapter 25, <u>*Handling and Preventing Errors*</u>, for more information about continuations.



- 1. Once the method is running, choose **Snap Continuation** from the toolbar.
- 2. Modify the method as desired.
- 3. Start the continuation.

#### 12.20.3 Using Logging

Text format logs are used for close examination of every operation that occurs during method execution in order to modify the run to improve the method. These generated logs give detailed information, including the date and time.

There are several logs available in the Biomek Software:

- Details Provides comprehensive information on every operation the instrument performs.
- Errors Lists errors for a method, if any.
- Pipetting Lists aspirate and dispense operations.
- Span8Pipetting captures only Span-8 pipetting operations that occur during a method run, including location and labware name or type.
- Span8Transfer captures only Span-8 transfer operations that occur during a method run, including location and labware name or type.

**Note:** The addition of SampleID data sets also allows the Span8Pipetting and Span8Transfer logs to include well identification information.

- FX, NX-S8 Span8Pipetting and Span8Transfer logs are available only for these instruments.
- UnifiedPipetting Captures pipetting operations, along with Sample IDs for wells, performed by any pods, including information on where the aspirate or dispense operation occurs.
- UnifiedTransfer Captures transfer operations, along with Sample IDs for wells, performed by any pods, including information on the source and destination labware.

To turn logging on or off:

- 1. From the toolbar, choose **Options**.
- 2. Choose Log Configuration.
- 3. Check or uncheck the desired Log(s).

**Note:** Refer to Chapter 26, <u>*Generating and Using Log Data*</u>, for more information on generating and using log data.

# Using Variables and Expressions in a Method

## 13.1 Overview

A variable is a value that has been assigned a name. A value can contain text strings, arrays, expressions, objects, operators, or numerical constants. Any value assigned a name can be referenced repeatedly during a method by inserting the variable name in any applicable field in a step configuration.

An expression is a one-line combination of alphanumeric characters and/or variables combined using mathematical operations. An expression can be used anywhere a variable can be used, and it may use one or more variables.

**Note:** For example the variable Volume may be required in a field, but accuracy may require additional information. The variable Volume may be modified into an expression, such as Volume+ 10 or Volume/4.

.Advantages to using variables include:

- Several variables can be created in one step; for example, the Start step can be used to define multiple variables that are available to every step within a method, while the Let and Worklist steps can be used to create multiple variables that can be used by their substeps.
- Several variables can be used in one step.
- A name can be entered as a variable or expression in any step configuration field where the value associated with the variable or expression is required. Biomek Software then interprets the variable or expression and replaces it with the associated value.
- A variable can be used as many times as necessary; for example, multiple steps within Let and Loop steps can all use the same variable.

**Note:** To make variables available to all steps in a method, define the variables in the **Start** step.

- Variables used extensively in a method need only be changed in the step they are defined to change the value in every step in the method that uses the variable.
- When a method is run more than once, the value associated with that variable can be changed in the step, and the associated values within the method automatically change accordingly.

## 13.2 Using Variables

A variable consists of a name and a value. To use a variable, enter an equal sign (=) followed by the variable name into the step configuration requiring the variable. For example, in Figure 13-1, the variable =AspValue, created in the Start step configuration of the method (Figure 13-2), is entered in the Volume field.

The Biomek Software internally substitutes the name with its value, but the name remains in the field. In the example, the variable AspValue is substituted internally by the value 25. The value of 25 is never displayed, only the variable name.

Variable values are not assigned a unit of measure. When a variable is identified in a step, the unit of measure is defined by the requirements of the step. For example, if a variable named Aspirate is assigned a value of 5, and **=AspValue** is entered in a field that requires microliters, the software interprets the value of 5 as 5 microliters; likewise, if **=AspValue** is entered in a field that requires seconds, the software interprets the value as 5 seconds.

**Note:** If a variable is used in a pipetting step with an auto-selected technique, and the value of the variable changes such that a new technique would be selected, the technique is automatically updated when the step is executed.

**Note:** Enter the variable name preceded by the equal (=) sign in any field where the value it represents is appropriate.



Figure 13-1. Transfer step using the AspValue variable

## 13.2.1 Creating Variables

Variables may be created in several steps. Refer to the section for the step for more specific information on creating variables with that step:

- <u>Configuring the Start Step</u> (Section 12.2.1).
- <u>Loop Step</u> (Section 16.9).
- <u>Run Method Step</u> (Section 17.3).
- <u>Worklist Step</u> (Section 17.4).
- <u>Let Step</u> (Section 17.6).
- <u>Create Group Step</u> (Section 21.3).
- <u>Set Global Step</u> (Section 21.7).
- <u>Define Procedure Step</u> (Section 21.8.1).
- <u>Run Procedure Step</u> (Section 21.8.2).



Figure 13-2. Variables configured in the Start step

#### 13.2.2 Using Variables in a Method

To use a variable created previously in the method, the variable name is entered into the configuration of the step. In the example, the variable AspValue, created in the Start configuration (Figure 13-2), is entered in the  $\mu$ L (volume) field of the Transfer step (Figure 13-3).

**Note:** Enter the Variable Name in any field where the value it represents is appropriate.

The Biomek Software internally substitutes the name with its value, but the name remains in the field. In the example (Figure 13-2), AspValue is substituted internally by the value 25. The value of 25 is never displayed, only the variable name AspValue.

Variables previously configured in any step may be used in any field where the value of the variable is a valid value for the field. For example, a variable with a value of 50 would be appropriate in a Volume field, but not in a field asking for a deck position.

Any variables used in a method must be defined prior to the step in which they are used, or an error occurs. For example, if the variable AspValue is not defined when it is entered in the Transfer step, an error results.

**Note:** Keep this in mind when writing a method that will be run in another method using the Run Method step. If the Run Method step in the other method specifies any variable values, those variables must be created in the method for method validation. When the method is run through the Run Method step, it is run using the variable values specified in the Run Method step.

**Note:** Variables created with some steps are only applicable for that step and its substeps. For example, a variable created with a Let step may only be used for substeps of the Let step.

To use a previously created variable in a step configuration:

- 1. Select the appropriate field in the desired step configuration.
- 2. Enter the variable name, preceded by an equal sign. For example, **=AspValue** in the Volume field of the Transfer step (Figure 13-3) indicates that a volume equal to the value of the variable AspValue is dispensed in the selected destination labware.

**Note:** Variables may be used in any field which the value is a valid entry. For example, a variable that has a text value cannot be entered in a field that is expecting a numerical input.



Figure 13-3. Transfer step using the AspValue and AspValue2 variables defined in the Start step

## 13.3 Using Expressions

An expression is a one-line combination of variables, such as Aspirate; mathematical operators, such as addition (+); string constants, such as "P4"; and numeric constants. An expression can be used anywhere a variable can be used and may contain one or more variables; for example, the variable Volume may be required in a field, but accuracy may require additional information. The variable Volume may be modified into an expression, such as Volume + 10 or Volume/4.

Like variables, expressions start with an equal sign and can contain operators allowed in VBScript programming, such as addition (+), division (/), exponentiation (^), concatenation of strings (&), and functions like random (Rnd), rounding (Round), or absolute value (Abs).

An expression can be entered in any step configuration field where the value associated with the expression is required. Biomek Software interprets the expression and replaces it with the associated value.

Three examples of expressions are:

- If a variable called my\_volume is defined, transferring from a 96-well microplate to each quadrant of a 384-well microplate could be accomplished by placing =my\_volume / 4 in the volume field.
- If the three variables VolumeA, VolumeB, and VolumeC are defined to represent the volumes transferred to a destination microplate from three different sources, mixing the entire contents of the destination microplate could be accomplished by configuring a Mix step using the expression = VolumeA + VolumeB + VolumeC in the volume field.
- In a Loop step, a variable called count is created and appended to the letter "P" in a step configuration as the expression = "P" & count. If the Loop is configured to start at 1 and end at 10, the Loop results would be P1, P2, ..., P10.

# Using Sample Tracking and Data Sets in a Method

## 14.1 Overview

Sample tracking means the contents of the plate or tube are known at any time in the method and decisions may be made during transfer or pipetting operations based on the contents of the plate or tube.

Sample tracking is possible in Biomek Software using data sets, which store specific information about wells or tubes (refer to Section 14.1.2, *Data Set Definition*).

#### 14.1.1 Sample Tracking

Sample tracking allows information about a sample in an individual well to be moved along with the sample when it is moved to another well. Sample tracking is enabled through the following functions of Biomek Software:

- association of a sample ID with wells or tubes through data sets.
- automatic copying of the SampleID data set (and any other data sets) during transfer and pipetting operations.
- reporting of information.

#### 14.1.2 Data Set Definition

Data sets allow information about a sample in an individual well or tube to be tracked when the sample is moved to another well or tube. During any transfer or pipetting operation, all data sets associated with a plate are automatically copied from the source plate to the destination plate; however, only the wells in the data set affected by the transfer or pipetting operation are copied to the destination plate. The information in the data set can be used in decision-making; for example, during a **Transfer** or **Combine** step, a data set can be used to determine from which wells or tubes liquid should be aspirated or dispensed.

#### 14.1.3 Data Set Steps in the Biomek Software

Four steps in Biomek Software allow data sets to track samples. Knowledge of these steps is necessary to understand how data sets are used in Biomek Software to track samples:

- Create Data Set specifies data in a data set (refer to Section 24.3, <u>Configuring the Create Data Set Step</u>).
- Data Set Management renames, removes, copies, or modifies the properties of a data set (refer to Section 24.4, <u>Configuring the Data Set</u> <u>Management Step</u>)
- Data Set Processing applies a transformation expression to an existing data set to create a new data set (refer to Section 24.5, <u>Configuring the Data</u> <u>Set Processing Step</u>).
- Reporting generates a report on data sets at any point during a method (refer to Section 24.6, <u>Configuring the Reporting Step</u>).

**Note:** Refer to Chapter 24, <u>Using the Data Sets Step Palette</u>, for complete information on displaying the Data Sets Step Palette and configuring the data sets steps.

#### 14.1.4 Example of Sample Tracking Using Data Sets

While there are numerous ways in which sample tracking can be implemented using data sets, the following is an example of using data sets to track sample ID:

If a sample ID is associated with a tube and an amount is transferred from the tube to specified wells on a microplate, the **Sample ID** data set for the tube is copied to the specified wells on the microplate. The report will show the sample ID in the tube and the sample ID in the specified wells.

## 14.2 Understanding How Data Sets Are Created

The information contained in a data set is chosen in the **Create Data Set** step (refer to Section 24.3, *Configuring the Create Data Set Step*); however, some data sets may be generated in other ways through the software.

Biomek Software creates data sets in several ways:

• automatically (refer to Section 14.2.1, <u>Understanding the Volume Data Set</u>).

**Note:** The Volume data set is created automatically.

- using the Instrument Setup step (refer to Section 15.2.6, <u>Creating the</u> <u>SampleID Data Set Using the Instrument Setup Step</u>).
- using the Create Data Set step (refer to Section 24.3, <u>Configuring the</u> <u>Create Data Set Step</u>).
- using the SILAS step (refer to Section 22.8.3, <u>Creating Data Sets from</u> <u>SILAS Messages</u>).
- using VB scripting (refer to Section 28.3, <u>Scripting Data Sets</u>).

**Note:** Creating data sets using scripting requires prior knowledge of Visual Basic scripting.

**Note:** While Biomek Software retains information on all data sets and allows the information to be used anytime during the method, the specific information on a data set can only be viewed in a report configured in the **Reporting** (refer to Section 24.6, *Configuring the Reporting Step*) or Finish steps (refer to Section 12.2.2, *Configuring the Finish Step*).

#### 14.2.1 Understanding the Volume Data Set

The volume of each well or tube is automatically tracked and stored for each piece of labware in a data set called Volume. The Volume data set is read-only and cannot be changed or manipulated, but may be used in other steps in the method to specify well patterns or volumes.

The Volume data set is reported when reports on data sets are generated and configured in the Reporting step (refer to Section 24.6, <u>Configuring the Reporting</u>. <u>Step</u>). The Volume data set allows the volume of a well or tube to be known at any time during a method by inserting a Reporting step after any transfer or pipetting operation.

## 14.3 Using Data Sets in a Transfer or Combine Step

- FX, NX-S8 Only the Span-8 Pod can be used to define patterns with data sets in a Transfer or Combine step.
- 3000 Only a Single-Tip Pipette Tool can be used to define patterns with data sets in a Transfer or Combine step.

Not only are data sets automatically copied from the source plate to the destination plate during all transfer and pipetting operations, data sets may also be used in a **Transfer** or **Combine** step to define patterns for aspirating and dispensing liquid or to specify transfer volumes to specific wells. However, to use data sets to define well patterns for the **Transfer** and **Combine** steps, a data set must first be created for the source or destination labware in the **Create Data Set** step (refer to Section 24.3, *Configuring the Create Data Set Step*).

**Note:** The well patterns defined using data sets are for use during the transfer operation for the specific **Transfer** or **Combine** step; they cannot be saved in the Well Pattern Editor.

Data sets can be used in a Transfer or Combine step for:

- <u>Defining Well Patterns Using Data Sets</u> (Section 14.3.1).
- <u>Specifying Volumes Using Data Sets</u> (Section 14.3.2).

#### 14.3.1 Defining Well Patterns Using Data Sets

Data sets may be used in a method to define well patterns for aspirating and dispensing liquid using the **Transfer** and **Combine** steps.

To use data sets to configure source or destination labware in a **Transfer** or **Combine** step:

1. Insert a **Transfer** or **Combine** step into the Method View anywhere after the specific data set that is used has been created (Figure 14-1).



Figure 14-1. Transfer step in Method View after data set is created

2. In Use Pod, select the Span-8 Pod. Use Probes is displayed (Figure 14-1).

**Note:** If using a single-pod instrument, **Pod1** is the only available selection.

- 3. In Use Probes, select the probes of the Span-8 Pod to use for the transfer.
  - 3000 Use Probes is not displayed.
- 4. Configure the Tip Handling as desired.

**Note:** Refer to Section 15.3.1, <u>*Configuring Tip Handling*</u>, for information on configuring Tip Handling.

5. Select the Source or Destination labware.



6. Double-click on the graphic of the labware to zoom in (Figure 14-2).



7. Select **Use Data Set** to define a pattern using a data set.

**Note:** All data sets defined for the specified Source or Destination labware are available for selection.
- 8. In where its values, select the desired option:
  - are equal to wells that have a value for the selected data set that are equal to a specified value or string are included in the pattern.

Note: Strings used with are equal to must be entered as ="string".

- are greater than wells that have a value for the selected data set that are greater than a specified value are included in the pattern.
- are greater than or equal to wells that have a value for the selected data set that are greater than or equal to a specified value are included in the pattern.
- are less than wells that have a value for the selected data set that are less than a specified value are included in the pattern.
- are less than or equal to wells that have a value for the selected data set that are less than or equal to a specified value are included in the pattern.
- are not equal to wells that have a value for the selected data set that are not equal to a specified value or string are included in the pattern.

**Note:** Strings used with are not equal to must be entered as **="string"**.

• contains — wells that contain a specified numeric or text string within their value for the selected data set are included in the pattern.

Note: Strings used with contains must be entered as ="string".

- 9. Enter a value to complete the expression.
- 10. Double-click the graphic of the labware or choose **Zoom Out** to return to the **Transfer** step configuration.

# 14.3.2 Specifying Volumes Using Data Sets

Data sets may be used in a method to specify transfer volumes using the **Transfer** and **Combine** steps.

To use data sets to configure volumes in a Transfer or Combine step:

1. Insert a **Transfer** or **Combine** step into the Method View anywhere after the specific data set that is used has been created (Figure 14-3).

🌵 Biomek® Software - Method7* [New]	
File Edit Project Instrument Execution Options Help	
□☞☆◼◨◓▯◾▫▫〃	
💼 🧟 Start	Use god Pod2 v for transfer. Use probes 1 2 3 4 5 6 7 8
Create Instrument Setup	A Tip Handling
Data Set Setup	V Load AP96_200uL V tips and unload them V when the transfer is done.
	Wash tips in Water : 3 cycles of 110% %
Data Set Management Transfer	✓ Wash tips with 2 mL of system liquid after dispensing 1 mL to waste. □ Speed Pump
Finish	✓ Wash tips between transfers.
Data Set Processing Combine	Click here to add a source.
III L	
Reporting Move Step Labware	
18	
Pause	
~	
	Stop when finished with Destinations  Advanced
	Beplicate each well 1 time.
	© Dispense up to 1 → time per draw.
	C Aspirate at most 0
	♥ Transfer Details
	TL1 P4 💹 P16
	P1 Emple P9 P13 P17
	P2 5ample P10 P14 P18
	P3 P7 P11 P15 P19
Method7* BiomekFX BiomekFX ETC: 0:00:03	

Figure 14-3. Transfer step in Method View after data set is created

2. In Use Pod, select the Span-8 Pod.

**Note:** If using a single-pod instrument, **Pod1** is the only available selection.

- 3. In Use Probes, select the probes of the Span-8 Pod to use for the transfer.
  - > **3000** Use Probes is not displayed.
- 4. Configure the Tip Handling as desired.

**Note:** Refer to Section 15.3.1, <u>Configuring Tip Handling</u>, for information on configuring Tip Handling.

5. Select the Source or Destination labware.

**Note:** For a Transfer step, Volume is configured in the destination labware; for a Combine step, Volume is configured in the source labware.

- 6. In Volume, choose one of the following:
  - To retrieve all values of a data set, enter
     =positions.PositionName.labware.DataSets("name").GetAll, where PositionName is the name of the deck position assigned by the Deck Editor (for example, p5 or p8) and name is the name of the data set from which to retrieve values.
  - To get a single value from the data set, enter
     =positions.PositionName.labware.DataSets("name")("A1"), where "A1" is the position in the data set of the desired value.
  - To get a subset of values from the data set, enter
     =positions.PositionName.labware.DataSets("name")(array(" A1", "B1, "C1")), where "A1", "B1", and "C1" are the positions in the data set of the desired values.

**Note:** Data sets can also be used in an expression using the above and the ListMath functions (refer to Section 10.4.1, *About ListMath*). For example, to normalize the volume of all wells on the plate at P5 to 250 µL, enter:

= Listmath.Sub (250, positions.p5.labware.DataSets("Volume").GetAll).

# Using the Basic Step Palette

# 15.1 Overview

The Basic Step Palette incorporates the basic steps that provide essential functionality for the Biomek instruments, including general instrument setup, pipeting operations, and labware manipulation.



CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height and failure to choose each ALP correctly in the software may result in collisions between pod(s) and ALPs during operation.

The steps available in the Basic Step Palette are:



Instrument Setup — specifies the type of labware used for a liquid transfer, the position the labware occupies on the Biomek instrument deck, and some labware content configuration.

**Note:** The type and position of ALPs and other hardware on the deck is specified in the Deck Editor.



• Transfer — comprises tip load, aspirate, dispense, and unload tip functions in one step to transfer liquid from a single source to one or more destinations.



• Combine— similar to a Transfer step, except Combine transfers liquids from one or more sources to a single destination.



- Move Labware moves labware via automatic manipulation using the gripper on the Biomek instrument.
  - > **NX-S8** The optional gripper must be installed to move labware.



٠

- Pause halts instrument interaction with a lab position for a specified amount of time or the entire deck for an indefinite period of time.
- Comment
- Comment documents the method or adds instructions in the Method View.

# 15.2 Instrument Setup Step



# CAUTION: An inaccurate Instrument Setup may result in pod and labware collisions, or in inappropriate pipetting.

When preparing the Biomek instrument for a method, it is important to understand a

critical concept that applies every time the instrument is set up: obstacle avoidance. When building a method, always tell the software what labware and ALPs are on the deck, and the deck position each occupies. In the **Instrument Setup** step, labware is selected and positioned on the Biomek instrument deck.

As the bridge and pod move during method execution, the pod rises just enough to clear the objects sitting on the deck, and not much higher; therefore, if the software is not aware of what and where objects are on the deck, a hardware crash may occur.

Insert an Instrument Setup step to specify the labware, devices, and ALPs that are on the deck and the deck position each occupies. The Instrument Setup step can be inserted immediately after the Start step (Figure 15-1), or at any point in the method flow.

The Sample ID data set is also created in the Instrument Setup step by configuring the option Pause for bar code (refer to Section 15.2.6, <u>Creating the SampleID Data</u> <u>Set Using the Instrument Setup Step</u>).

**Note:** It is not necessary to add the **Instrument Setup** to a method if the current method is using the same deck setup as the previously run method, and the setup was not cleared in the **Finish** step of the previous method.



To insert an Instrument Setup step into a method, drag and drop Instrument Setup into the Method View (Figure 15-1).



Figure 15-1. Instrument Setup step and configuration



CAUTION: To make sure the Biomek pod avoids all obstacles during its travel, always specify the labware, ALPs, and devices that are on the deck, and the deck position each occupies.

Configuring the Instrument Setup step includes:

- <u>Selecting a Deck Layout</u> (Section 15.2.1).
- <u>Verifying Deck Layout</u> (Section 15.2.2).
- <u>Configuring Pod Setup Options</u> (Section 15.2.3).
- <u>Populating the Deck with Labware</u> (Section 15.2.4).
- <u>Configuring Tip and Labware Properties</u> (Section 15.2.5).
- <u>Creating the SampleID Data Set Using the Instrument Setup Step</u> (Section 15.2.6).
- <u>Creating Custom Labware</u> (Section 15.2.7).

## 15.2.1 Selecting a Deck Layout



CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height, and failure to choose each ALP correctly in the software may result in collisions between pod(s) and ALPs during operation.

The Instrument Setup step uses a Deck Layout as a map for the placement of labware, ALPs, and external hardware devices, such as stacker carousels.

A Deck Layout is a blueprint, or map, of the Biomek instrument deck. Biomek Software provides two predefined Deck Layout options: Deck1 and Standard. These two deck layouts represent the basic deck layout for the specific instrument with only standard deck positions.

Note: The Standard deck layout is read-only and may not be modified.

When an Instrument Setup step is inserted into a method, the default Deck Layout set in the Deck Editor is automatically selected. If another deck layout other than the default deck is desired, choose any existing Deck Layout in Deck from the drop-down list.

**Note:** Refer to Chapter 5, <u>*Preparing and Managing the Deck*</u>, for more information on creating and modifying a deck layout and setting the default deck layout.

# 15.2.2 Verifying Deck Layout

It is possible to verify the Deck Layout selected for a method before the method is run by selecting the Pause to Confirm Setup? option (Figure 15-2). The default selection for the Pause to Confirm Setup? option is On.

Verification of the Deck Layout is recommended when a method is run repeatedly, or when modifications have been made to a method. A prompt appears (similar to the one below) when Pause to Confirm Setup? is selected prior to running the method. The prompt displays the deck layout and the labware type and name (if assigned a name in Labware Properties) of the labware on each position.

Biomek® Software		_	-		
AP96_200uL_LI	Rsvr - 7777 BCFullReservoi	P8	P12		P16
AP96_200uL_LI	Deep1 BCDeep96Rour	Deep2 - BCDeep96Rour	P13	w	P17
P2	Dest1 - BCFlat96	Dest2 - BCFlat96	P14		P18
РЗ	Dest3 - BCFlat96	Dest4 - BCFlat96	P15		P19
The left and should have no tins	loaded				
The right pod should have no tip	is loaded.				
Does the Biomek® Software de	ck match the above layo	out, including the lat	oware and their loca	itions?	
If yes, choose OK to continue the If no, choose Abort to stop the n	e method. nethod.				
		K <u>A</u> borl	t		

Figure 15-2. Pause to Confirm Setup? Deck Layout confirmation prompt

## 15.2.3 Configuring Pod Setup Options

Verify Pod Setup? is used to configure the current state of the pod or tool on the Biomek instrument at the beginning of the Instrument Setup step. However, selecting Tips Loaded or Tool Loaded informs the Biomek Software that tips or a tool are loaded on the instrument; it does not instruct the instrument to load tips or a tool.

Verify Pod Setup? is recommended when multiple methods are run one following another, when a single method is run repeatedly, or when tips are not loaded during the method.

Options for Verify Pod Setup depend on the instrument and pod type.

- **FX**—With a Multichannel Pod, has two options to verify the pod setup:
  - Pod must have the gripper retracted instructs the instrument to retract the gripper on the pod. The default setting is **On**.
  - Tips Loaded informs the software that tips are loaded at the time of the Instrument Setup step. The default setting for the Tips Loaded option is **Off**.
- **FX**—With Span-8 Pod, has only one option to verify the pod setup:
  - Tips Loaded informs the software that tips are loaded at the time of the Instrument Setup step. The default setting is **Off**.
- > **3000** Has two options to verify the pod setup:
  - Tool Loaded informs software that a tool is loaded on the head assembly at the time of the Instrument Setup step. The default setting is Off.
  - Tips Loaded informs the software that tips are loaded onto the tool at the time of the Instrument Setup step, if Tool Loaded is **On** and a pipetting tool is selected. The default setting is **Off**.
- > **NX-MC** Has two options to verify the pod setup:
  - Pod must have the gripper retracted instructs the instrument to retract the gripper on the pod. The default setting is **On**.
  - Tips Loaded informs the software that tips are loaded at the time of the Instrument Setup step. The default setting is **Off**.
- > **NX-S8** Has only one option to verify pod setup:
  - Tips Loaded informs the software that tips are loaded at the time of the Instrument Setup step. The default setting is **Off**.

To verify the pod setup prior to a method run:

1. Select Verify Pod Setup?.

Configure...

2. Choose **Configure** to change the Pod Setup, if desired. Pod Setup appears (Figure 15-3).

Pod Setup Left Pod
Pod must have the gripper retracted
Tips Loaded AP96_200uL
Right Pod
🗖 Tips Loaded 📃 🔽
OK Cancel

Figure 15-3. Pod Setup prompt for dual-pod Biomek FX instrument

- FX If the instrument is a dual-pod system, Pod Setup provides options for the Left Pod and the Right Pod.
- 3. Uncheck **Pod must have the gripper retracted**, if the gripper position is not critical.
- 4. Select **Tips Loaded**, if tips must be loaded at the beginning of the method. The **Tips Loaded** field has three states:
  - Checked indicates tips are loaded
  - Unchecked indicates tips are not loaded
  - Gray-checked instructs the software to leave the tips in their current state, whether tips are loaded or not loaded
- 5. If **Tips Loaded** is selected, select the tip type loaded at the beginning of the method.
- 6. Choose **OK** to save the changes made to the **Pod Setup**.

# 15.2.4 Populating the Deck with Labware

The labware available for selection is displayed graphically below Labware Category (Figure 15-4). A specific type of labware can be viewed in the graphical display, or all types of labware available can be viewed simultaneously, using the Labware Category filter.



Figure 15-4. Populating the Deck Layout of a Biomek 3000 instrument

To populate the Deck Layout with labware, select from the graphical display below Labware Category and add them to the Deck Layout display.

**Note:** When populating the deck with labware, consider the tip to labware accessibility of the instrument pod or tool as described in the hardware manual for the specific instrument.

- 1. To display labware, select the type of labware desired in **Labware Category**. Labware Category choices include:
  - Any lists all types of labware available, including lids and deck positions reserved for swapping labware.
  - Custom lists any labware stored with defined properties (refer to Section 15.2.7, <u>Creating Custom Labware</u>).
  - Lid lists lids associated with labware available.
  - Reservation reserves deck positions for specific purposes; for example, to enable swapping of labware between positions or tip loading for the Span-8 Pod.
  - Reservoir lists reservoirs available.
  - TipBox lists types of tips available.
  - Titerplate lists types of microplates available.
  - Tuberack lists types of tube racks available.

**Note:** Labware types and their characteristics are defined in the Labware Type Editor. Refer to Chapter 7, <u>Creating and Modifying Tip and Labware Types</u>, for information using the Labware Type Editor. If the Hide Labware option was chosen while defining a labware type, it will not be displayed in the Instrument Setup step.



# CAUTION: Do not place labware other than a tip box on a tip loader position (TL#).

2. To place labware on the Deck Layout, drag and drop each desired labware graphic to the desired position on the Deck Layout display.

#### OR

Click on the labware graphic, and then on the desired position in Deck Layout display. The same type of labware can be added to as many deck positions as required by continuing to click on deck positions.

#### OR

Click on the labware graphic, and then click and drag the mouse over multiple deck positions in the Deck Layout display. This places the labware in all the highlighted positions.

**Note:** To move labware to a different position, drag the desired labware to the new position in the Deck Layout.

3. To remove unwanted labware from the Deck Display during setup, drag and drop the labware to the **Clear** (trash) icon (Figure 15-4).

OR

Right-click the unwanted labware and select **Delete** from the menu that appears.

### 15.2.4.1 Changing Instrument Setup During a Method

When changing the instrument setup during a method, a second Instrument Setup step must be added. The additional Instrument Setup step reflects modifications to the current state of the deck (Figure 15-5).



Figure 15-5. A second Instrument Setup step with deck positions specified As Is

Additional Instrument Setup steps can be inserted into a method as many times as necessary to complete a liquid-handling process, and to accommodate the addition of labware to the deck throughout a method.

The following options are available to help modify all instrument setups, either globally or by individual deck positions. These options are:

- As Is individual deck positions retain their current state, whether empty or occupied by labware or a device.
- Toggle toggles all empty deck positions to the As Is state and from the As Is state to their initial status, allowing those deck positions to retain their empty state.
- Clear clears individual deck positions, whether empty or occupied by labware.
- Clear All clears all deck positions, whether empty or occupied by labware or a device.

#### 15.2.4.1.1 As Is and Toggle

As Is indicates that a deck position retains its current state, whether empty or occupied by labware or a device. Previously populated deck positions are indicated with the As Is option in the subsequent Instrument Setup step configuration. When As Is is applied to a deck position, the position becomes grayed out in the Deck Layout display (Figure 15-5).

As Is may also be used at the start of a method to indicate that deck positions retain their state from a previously run method that did not clear the deck of all labware in the Finish step.

To indicate that a deck position retains its current state:



1. Click As Is.

- 2. Click any deck position on the Deck Layout, and it is displayed As Is.
- 3. Repeat step 2 for each desired deck position.

OR

Click on a deck position, and then click and drag the mouse over multiple deck positions in the Deck Layout display. This toggles all the highlighted positions to the **As Is** state.

Toggle toggles all empty deck positions in the Instrument Setup Display between As Is, preserving the deck position's current state (Figure 15-6), and Clear, which empties a deck position. When Toggle is selected, deck positions labeled As Is revert to Clear.

**Note:** Adding labware to a deck position that is set to As Is removes the As Is status and replaces that position with the added labware. Any labware at that position is assumed to be removed from the deck and replaced with the new labware.

To indicate that a deck position retains its current state:

Toggle	5
- i <u>O</u> ggie	·

Click **Toggle**, and any deck position not occupied by a piece of labware in the Instrument Setup display is toggled to the As Is state.



Figure 15-6. Toggle applied to the Deck Layout

#### 15.2.4.1.2 Clear and Clear All

**Clear** removes labware or the **As Is** designation from individual deck positions. There is no confirmation for **Clear**, so once **Clear** is chosen, and a deck position is selected, the position is cleared of any labware or the **As Is** option (Figure 15-7).



Figure 15-7. Clear Deck Layout designation applied to deck position TL1

The Clear All option removes all labware and As Is designations from the Instrument Setup. Clear All displays a prompt before the deck is cleared of all labware and As Is designations.

**Note:** If a deck position that contained labware from a previous **Instrument Setup** step is set to **Clear**, the labware is assumed to have been removed from the deck and that position no longer contains any labware.

To remove labware from the Instrument Setup configuration:



- 1. Choose **Clear** to clear a single deck position of labware or an As Is designation.
- 2. Click on the deck position in the Deck Layout display.
- 3. Continue to click on individual deck positions to empty them.

OR

Drag and drop labware from Deck Layout into Clear.

OR

Right-click on the labware in the Deck Layout display and select **Delete**.

Clear <u>A</u>ll

- 4. To clear the entire deck at one time, choose Clear All.
- 5. A confirmation prompt asking for verification appears. Choose **Yes** to remove all labware from the deck (Figure 15-8).

Confirm	X
?	Are you sure you want to clear the deck?
	<u>Y</u> es <u>N</u> o

Figure 15-8. Confirmation prompt to Clear All labware from the Deck Layout

# 15.2.5 Configuring Tip and Labware Properties

Labware Properties is used to configure properties for each piece of labware added to the Deck Layout. The information provided in Labware Properties is used when a pipetting technique is selected, and when tips are loaded and unloaded.

To access Labware Properties:

1. Double-click on a piece of labware in the Deck Layout.

OR

Right-click on a piece of labware in the Deck Layout and choose **Properties** to open Labware Properties.

**Note:** If there is a stack of labware at the selected position, a submenu appears from Properties listing labware from bottom to top.

The configuration options in Labware Properties provide the following information, if it applies to the type of labware selected:

- Name assigned to a piece of labware, if desired
- Labware type
- Maximum volume per well
- Bar code
- Sense liquid level options
- Volume for each well, if known The default volume for a labware type may be set in the Labware Type Editor (refer to Section 7.3.6, <u>Editing Labware</u> <u>Type Properties</u>).
- Liquid type
- Where to dispose of tips
- Where to dispose of tip boxes
- Maximum number of times tips may be loaded
- Available tips

**Note:** Variables and expressions may be entered in any field that can be configured. Refer to Section 13.2, *Using Variables*, and Section 13.3, *Using Expressions*, for more information on using variables and expressions.

## 15.2.5.1 Configuring Labware Properties for Tips

When the labware selected for configuration is a tip box, Labware Properties provides the ability to configure the following information:

1. Double-click the tip box to configure in the Deck Layout.

#### OR

Right-click the tip box to configure in the Deck Layout and select **Properties** from the menu. Labware Properties for tips appears (Figure 15-9).

Labware Properties	
Name:	Labware Type: AP96_200uL
Bar Code:	
When empty, send to: Home>	Unload Tips Into: <a>Tipbox&gt;</a>
Load no more than 1 time	
♡ Show Available Tips	
	OK Cancel

Figure 15-9. Labware Properties for tips

2. In **Name**, enter the name assigned to the tip box, if applicable. Naming a tip box forces the instrument to look for the specific box of tips, rather than any tip box on the deck that contains the specified tip type.

**Note:** Multiple tip boxes can be given the same name. This is useful for creating a pool of tip boxes for a specific use.

#### OR

Leave Name blank. When Name is blank, tips are accessed by the information displayed in the Labware Type field.

- 3. In Labware Type, verify the tip type.
- 4. In **Bar Code**, enter the bar code. Use the bar code field to identify a specific tip box in certain methods, such as plate replication. This field may be left blank.
- 5. In When empty, send to, select a final destination for the tip box at the end of the method.
- 6. In Unload Tips Into, select the location reserved for tip disposal.
- 7. In Load no more than, enter the number of times tips can be loaded onto the pod during the method.

8. Choose **Show Available Tips** to display a graphic of the tip box (Figure 15-10).

abware Properties
Name: Labware Type: AP96_200uL
Bar Code:
When empty, send to: <home> 🗾 Unload Tips Into: <tipbox></tipbox></home>
Load no more than
A Hide Available Tips
1 2 3 4 5 6 7 8 9 10 11 12
96 usable tips.
OK Cancel

Figure 15-10. Labware Properties for tips with Available Tips shown

- 9. On the graphic of the tip box, select the usable tips. By default, all tips are selected. This can be used to indicate missing tips or tips that are otherwise not used in the method.
- 10. Choose **Hide Available Tips** to collapse the graphic of the tip box.
- 11. Choose **OK** to save Labware Properties and return to the Instrument Setup step configuration.

#### 15.2.5.2 Configuring Labware Properties of Microplates, Reservoirs, and Tube Racks

When the labware selected for configuration is a microplate, reservoir, or tube rack, Labware Properties provides the opportunity for configuration (Figure 15-11).

1. Double-click the piece of labware to configure in Deck Layout.

OR

Right-click the piece of labware to configure in Deck Layout and select **Properties** from the menu. Labware Properties for microplates, reservoirs, and tube racks appears (Figure 15-11).

Labware Properties		
Name:	Labware Type: BCFlat96	Maximum Volume: 362.76 µL
Bar Code:		
Labware contains a Known 💌 volume: 0	⊥µL of liquid type:	•
• Sense the liquid level the first time a well with	Unknown or Nominal volume is accessed "from	the Liquid".
🔿 Sense the liquid level every time a well is acce	ssed "from the Liquid".	
▽ Show Labware Volumes		
		OK Cancel



- 2. Verify the Labware Type.
- 3. In Name, enter a name for the labware.

**Note:** When a deck is populated by numerous pieces of labware, naming labware is recommended. Names should be descriptive of the contents of the labware or the work being accomplished during the method. Naming labware in a meaningful fashion may reduce confusion.

- 4. In **Bar Code**, enter the bar code to identify a specific plate in certain methods, such as plate replication. This field may be left blank.
- 5. Make a selection in **Labware contains**. This information is used by many of the techniques supplied with Biomek Software that aspirate or dispense at a certain offset from the liquid level.

**Note:** Some pods or tools can detect the liquid level using liquid-level sensing technology.

- If Known is selected, the liquid level is not detected during method run and the entered value is used during validation and method run. Known Volume should be supplied whenever possible.
- If Unknown is selected, the liquid level is detected during method run if required by the technique, and the wells are assumed to be full when validating the method.
- If Nominal is selected, the liquid level is detected during method run, but the volume in the wells is assumed to be the entered value when validating the method.
- 6. Enter the **Volume**, if **Nominal** or **Known** volume is selected. A value entered in Volume is assigned to each well on the selected labware.

- 7. Select the **Liquid Type** contained in the labware. Specifying liquid type is useful when Biomek Software auto-selects a pipetting technique for any aspirate and dispense operations on this piece of labware. This technique is auto-selected based on the physical factors of the liquid, as well as the physical attributes of the labware, and the pod or tool performing the operation. For more information on liquid types, refer to Chapter 8, <u>Understanding and Creating Liquid Types</u>.
- 8. Select **Sense the liquid level the first time a well with Unknown or Nominal volume is accessed "from the Liquid"** to use liquid level sensing to determine the liquid level only the first time it accesses a well with an Unknown or Nominal volume from the liquid. When the same well on the piece of labware is accessed, the liquid level is calculated internally based on the result of the earlier liquid level sense and the amount aspirated or dispensed to the well in previous steps.

OR

Select **Sense the liquid level every time a well is accessed "from the Liquid"** to use liquid level sensing to determine the liquid level every time it accesses a well with an Unknown or Nominal volume from the liquid.

**Note:** Liquid level sensing options apply only if the pod or tool accessing the labware is liquid level sensing capable. If not, the selected option is ignored.

 If the labware does not contain uniform volumes, choose Show Labware Volumes to display a graphic of the labware (Figure 15-12) and assign two or more volumes to wells on the same piece of labware.

Labware Properties Labware Type: BCFlat96 Name: Maximum Volume: 362.76 µL Bar Code: Labware contains a Known ▼ volume: 0 μL of liquid type: • Sense the liquid level the first time a well with Unknown or Nominal volume is accessed "from the Liquid". C Sense the liquid level every time a well is accessed "from the Liquid" Hide Labware Volumes Volume 2 3 4 5 6 7 8 9 10 11 12 1 0 А В С D Е F G Amount (µL): Н Set ⊆lear 96 selected wells. ОK Cancel

**Note:** If a volume for all wells is specified in step 6, proceed to step 15.

Figure 15-12. Labware Properties for a 96-well microplate with Labware Volumes shown

**Note:** Show Labware Volumes may appear different depending on the number of wells on the selected labware.

10. Select the desired wells on the graphic.

**Note:** Hold down **Ctrl** to toggle wells between selected and deselected without affecting the selection status of other wells. Click a column or row heading to select all wells of a column or row.

- 11. Enter the volume to assign to the currently selected wells in **Amount**.
- 12. Choose **Set** to assign the current Amount to the currently selected wells. The wells are given a color code that corresponds to the list of set volumes under Volume.
- 13. Repeat steps 10-12 until all desired wells have been assigned the desired volume. Clear sets the volume for all wells back to zero.
- 14. Choose Hide Labware Volumes to collapse the graphic of the labware.
- 15. Choose **OK** to save Labware Properties and return to the Instrument Setup step configuration.

# 15.2.5.3 Configuring Labware Properties for a Biomek 3000 Tool Rack

The Tool Rack on the Biomek 3000 instrument stores various interchangeable tools for head assembly on the instrument deck during a method. The Biomek 3000 can load and unload tools from the Tool Rack to give the instrument different capabilities. The tools available to the head assembly are configured from Labware Properties in an Instrument Setup step.

Since a gripper tool may be placed physically on the Gripper Tool Rack only (refer to the *Biomek*® 3000 Laboratory Automation Workstation User's Manual, Chapter 3.2, <u>Installing the Gripper Tool in the Tool Rack</u>), it must be configured in Instrument Setup using a Gripper Tool Rack.

Since the HDR tool occupies all five positions of a standard tool rack and cannot be placed on the gripper tool rack (refer to the *Biomek*® *3000 Laboratory Automation Workstation User's Manual*, Chapter 5.2.2, *Installing the HDR Tool on a Tool Rack*), it must be configured in Instrument Setup occupying all five positions of a Tool Rack. However, if the tool rack is installed on the left end of the Biomek 3000 deck, and there is no disposal unit installed on the left side, the HDR tool body may be installed in slot 1 or 2 of the tool rack. Additional tools may then be installed in slots 4 and 5 of the tool rack.

To configure the tools available at a Tool Rack in Instrument Setup:

1. Double-click the piece of labware to configure in Instrument Setup

OR

Right-click the piece of labware to configure in Instrument Setup and select **Properties** from the menu. Labware Properties for the Biomek 3000 Tool Rack appears (Figure 15-13).



Figure 15-13. Labware Properties for a Biomek 3000 Tool Rack

2. Locate the desired tool on the right side of Labware Properties and drag it to one of the unoccupied (black) positions on the Tool Rack. The selected tool is placed in the position (Figure 15-14).



Figure 15-14. Labware Properties for a Biomek 3000 Tool Rack with tools loaded

- 3. Repeat step 2 until the Tool Rack is full or all desired tools have been placed on the Tool Rack.
- 4. Choose **OK** to save the Labware Properties for the Tool Rack.

# 15.2.6 Creating the SampleID Data Set Using the Instrument Setup Step

The Instrument Setup step contains an option to **Pause for bar code input?** during a method run (Figure 3-11). When this option is selected, the method pauses at the Instrument Setup step and allows for per-well or per-tube bar codes to be entered for all labware on the deck. Using the specified bar codes, a data set called SampleID is created for each piece of labware.



Figure 15-15. Enabling bar code input in the Instrument Setup step

When Pause for bar code input? is selected, Bar code input (Figure 15-16) appears during the method run at the Instrument Setup step.

If Pause to confirm setup? is also selected, Bar code input appears after the Deck Confirmation prompt (refer to Section 15.2.2, *Verifying Deck Layout*).

Bar code input	×
To input bar code ID information: Select a position from the deck display below. In the Labware ID field enter the bar code for the labware. To select a well other than A1, click on the well you want displayed. In the Wel/Tube ID field, enter the bar code and press Enter. After pressing Enter, the next well will be automatically selected and its bar code displayed.	
Position	
Labware ID	
Well/Tube ID	
Increment by Row	
C Increment by Column Set	
P4 P6 Samples2	
Close	

Figure 15-16. Bar code input

If desired, only labware IDs or only sample IDs may be entered in **Bar code input**. It is not necessary to enter both labware and sample IDs for any piece of labware.

**Note:** Labware must be unstacked to assign Labware ID and/or Well/Tube ID in Bar code input.

To enter labware and sample IDs:

5. In **Position**, select the deck position of the labware to specify labware or sample IDs.

OR

Click on the desired piece of labware in the Deck Display. A graphic of the selected labware type appears (Figure 15-17).



Figure 15-17. Entering sample IDs for labware in Bar code input

- 6. In Labware ID, enter the plate bar code or name.
- 7. Select **Increment by Row** to enter sample IDs by row.

OR

Select **Increment by Column** to enter sample IDs by column.

**Note:** The Increment selection determines which well is automatically selected when a sample ID is set by choosing **Set** or pressing **Enter**.

- 8. Select the well or tube in the labware graphic to assign a sample ID.
- 9. In Well/Tube ID, enter the bar code or sample ID to assign to the selected well or tube.

- 10. Press **Enter** or choose **Set** to assign the Labware ID and Well/Tube ID to the selected labware and well or tube. The well or tube is assigned the sample ID and turns green to indicate a sample ID has been assigned to that well or tube. The next well or tube on the labware is automatically selected.
- 11. After each Well/Tube ID assignment and/or Labware ID assignment, press **Enter** or choose **Set**.

**Note:** If using a handheld bar code reader, ensure the bar code on the plate or tube ends with a carriage return and line feed.

- 12. Repeat steps 4-6 to assign a sample ID to all desired wells or tubes of the labware.
- 13. Repeat steps 1-7 to assign sample IDs to each desired piece of labware.
- 14. After sample IDs have been entered to all desired labware, choose **Close**. The SampleID data set is created for every piece of labware on the deck.

Each Instrument Setup step with Pause for bar code input? selected allows wells or tubes to be assigned IDs. If a new ID is assigned to a well or tube that previously had a SampleID created, the new value overwrites the old value. To maintain a record of all SampleIDs for a well or tube, use the Data Set Management step to rename or create a copy of the SampleID data set before the additional Instrument Setup step assigns the new sample ID (refer to Section 24.4, <u>Configuring the Data</u> <u>Set Management Step</u>).

## 15.2.7 Creating Custom Labware

The **Custom** labware category offers the option to create and reuse labware with defined and stored Labware Properties. New labware stored in the **Custom** labware category maintains the Labware Properties defined when the labware was created or added to the deck, including the name, bar code, liquid type, and volume information.

Custom labware creates an instance of the labware type containing the attributes of the working revision of the labware type it is based on at the time the custom labware is created and the configured Labware Properties. Custom labware is saved to the instrument file.

**Note:** The default volume for a labware type can be specified in the labware definition stored in the Labware Type Editor (refer to Section 7.3.6, *Editing Labware Type Properties*).

To create Custom labware:

- 1. Highlight the **Instrument Setup** step.
- 2. Add the desired labware to create the custom labware from to the deck; for example, if creating a custom reservoir, add a reservoir to the deck. Refer to Section 15.2.4, *Populating the Deck with Labware*.
- Configure the Labware Properties for that piece of labware. (Refer to Section 15.2.5.1, <u>Configuring Labware Properties for Tips</u> and Section 15.2.5.2, <u>Configuring Labware Properties of Microplates, Reservoirs, and Tube Racks</u> to configure Labware Properties.)



4. In Labware Category, display the **<Custom>** labware category. If no custom labware has been created, no labware is displayed.

Figure 15-18. Custom labware category displayed

 Drag and drop the labware graphic from the Deck Layout to the <Custom> Labware Category display. If the <Custom> labware category is not visible, dragging and dropping the labware has no effect.

**Note:** Multiple pieces of labware, or stacks of labware, cannot be added to the <**Custom>** labware category simultaneously; each piece of labware must be added individually.



Figure 15-19. Custom labware created

6. To delete a piece of labware from the <**Custom>** labware category, drag the labware from the <**Custom>** labware category display to Clear.

**Note:** Custom labware added to the <**Custom**> labware category in the Instrument Setup step is visible in the Device Setup (refer to Section 22.5, *Device Setup Step*) and Stacker Carousel Setup (refer to Section 23.3, *Using the Stacker Setup Step*).

# **15.3 Configuring Transfer and Combine Steps**

The **Transfer** and **Combine** steps configure multiple aspirate and dispense operations using the selected pod or tool to transfer liquid from all specified source wells to all specified destination wells. The **Transfer** step transfers liquid from a single source to one or more destinations, while the **Combine** step transfers liquid from one or more sources to a single destination.

The **Transfer** and **Combine** steps may be used with any Biomek instrument; however, the capabilities of the steps vary depending on the pod or tool configuration of the instrument:

- **FX, NX-MC** Using the Multichannel Pod, the same volume is transferred from each of the 96 or 384 channels in each aspirate and dispense cycle.
- FX, NX-S8 Using the Span-8 Pod, a different volume may be aspirated from each well of a source labware and a different volume may be dispensed to each destination. The pod may also use only selected probes to aspirate from specified wells on the source labware and dispense to specified wells on the destination labware.
- 3000 The pipetting tool aspirates from all channels at once (one for a single-channel tool or eight for an eight-channel tool). The tool transfers the same volume from each channel in each aspirate and dispense cycle.

**Note:** If there are multiple sources or destinations, each source or destination may have a different volume transferred, but all wells on any single source or destination must have the same volume transferred. To transfer different volumes to or from wells on the same piece of labware, configure multiple sources or destinations using the same labware but specify different wells and volumes.



The **Transfer** and **Combine** steps perform four liquid-handling functions through the step configuration, eliminating the need to insert multiple steps to complete the liquid transfer:

- Load tips onto pod or tool.
- Aspirate liquid from source(s).
- Dispense liquid to destination(s).
- Unload tips from pod or tool.

The Transfer and Combine step configuration includes:

- <u>Configuring Tip Handling</u> (Section 15.3.1).
- <u>Configuring Source Labware</u> (Section 15.3.2).
- <u>Configuring Destination Labware</u> (Section 15.3.3).
- Configuring Transfer Details (Section 15.3.4).

Insert a Transfer or Combine step into the Method View (Figure 15-20).

3000 — If a specific pipetting tool is desired, it must first be loaded using either a Change Tool (refer to Section 20.3, <u>Change Tool Step</u>) or a Load Tool step (refer to Section 20.4, <u>Load Tool Step</u>).

**Note:** The appearance of the **Transfer** or **Combine** step configuration varies depending on the instrument and pod selection.



Figure 15-20. Transfer step configuration for a Multichannel Pod

## 15.3.1 Configuring Tip Handling

Configuring Tip Handling in the Transfer and Combine steps is not required, because the default Tip Handling configuration locates and loads clean tips automatically every time a Transfer or Combine steps is initiated and unloads them when the step is completed. It is recommended, however, that Tip Handling configuration be supplied if cross contamination or conserving consumables is a concern, or a specific tip type should be used.

Tip Handling is also used to configure washing or changing tips between transfer operations. Tips may be washed or changed between sources or between destinations. Options for washing or changing tips vary according to the instrument and configuration.

> **3000** — Tip wash options are not available for this instrument.

To configure Tip Handling (Figure 15-20) for the Transfer or Combine steps:

- 1. If using a dual-pod *FX* instrument, specify the **Pod** to perform the transfer operation. The pod configured as the default pod is displayed in **Pod**. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.
- If using a Span-8 Pod on *FX* or *NX-S8*, select the probes used in the transfer by clicking the probe numbers in Use Probes. Any combination of probes may be used as long as all selected probes are using the same type of tips and syringes.

OR

Right-click any of the probe numbers in **Use Probes** and make a selection from the menu. Options include:

- Use Disposable Tips selects all probes with disposable tips
- Use Fixed Tips selects all probes with fixed tips
- Use Selection allows for custom selection

**Note:** If the probes selected are equipped with fixed tips, the Load and unload them/leave them on options (steps 4 to 6) are unavailable and the fixed tips are used for the transfer. Proceed to step 7.

**Note:** Disposable tips are selected by default. If the pod has only fixed tips, all fixed tips are selected by default. For example, if the pod is equipped with fixed tips on probes 1-4 and disposable tips on probes 5-8, probes 5-8 are selected by default. However, if probes 5-8 are also equipped with fixed tips, all eight probes are selected by default.

- 3. If Tip Handling is not already displayed, click the arrow or sentence summary to expand the Tip Handling configuration (Figure 15-20).
- 4. Select **Load** to load new tips at the start of the **Transfer** or **Combine** step.

**Note:** If Load is not selected, the Transfer or Combine step uses previously loaded tips, rather than loading new tips at the start of the step.

If Load is selected and the tips loaded to the pod have been used, an error may occur. To increase the number of uses allowed for tips, increase the Usages count in the Labware Properties configuration in the Instrument Setup step (refer to Section 15.2.5, *Configuring Tip and Labware Properties*).

5. If Load is selected, select the type of tips to load from the drop-down list, or from the Current Deck Display.

**Note:** If tips are selected from the Current Deck Display, and the tips are named, Load displays the name, not the tip type. If a name was not assigned to the tips, the tip type is displayed in Load.

It is recommended that tips not be named in the **Instrument Setup** step, because this could restrict the instrument from locating available tips.

6. To return tips to the tip box after the **Transfer** or **Combine** step is completed, select **unload them**.

OR

To leave tips on after the Transfer or Combine steps is completed, select **leave** them on.

**Note:** To conserve tips when using the **Transfer** or **Combine** steps within a Loop step, uncheck Load and select leave them on (steps 4 and 6 above). Add a New Tips step to the method prior to the Loop step, and an Unload Tips step after the Loop. This loads new tips before the Loop step, uses the same set of tips for all Transfer or Combine steps completed as part of the Loop, and unloads the tips when the Loop step is completed.

- 7. Choose **Wash tips in** to use a Multichannel Channel Tip Wash ALP to wash disposable or fixed tips.
  - > **3000** Options to wash tips are not available. Proceed to step 15.

**Note:** If Wash tips in is not selected, proceed to step 11.

**Note:** A Multichannel Tip Wash ALP must be added to the deck and configured in the Deck Editor before washing can occur. Clicking on the ALP while the Transfer and Combine steps is being configured places a check in Wash tips in, and populates Wash tips in with the solution configured for the Multichannel Tip Wash ALP in Labware Properties on the Instrument Setup step (refer to Section 15.2.5, <u>Configuring Tip and Labware Properties</u>).

8. In **Wash tips in**, select a solution for the wash cycle if the field does not already contain a solution.

**Note:** The wash solution specified in the **Transfer** and **Combine** steps must match the solution selected in the Multichannel Tip Wash ALP configuration, or an error occurs.

- 9. Choose the number of cycles to aspirate and dispense during wash.
- 10. In %, provide the maximum volume of fluid contained in the tips in previous steps; for example, if the maximum volume of fluid transferred is 50  $\mu$ L, and the % is set for 110%, the Transfer and Combine steps washes the tips with 55  $\mu$ L of solution.

**Note:** If the % sign is deleted in the % field, the field label changes to µL. Any value entered is now interpreted as a specific volume, not a percentage.

11. If using a Span-8 Pod on *FX* or *NX-S8*, choose Wash tips with to use the passive Span-8 Tip Wash ALP to wash fixed tips.

**Note:** Only fixed tips can be washed with the Span-8 Tip Wash ALP. If using disposable tips or Wash tips with is not selected, proceed to step 15.

**Note:** A Span-8 Tip Wash ALP must be added to the deck and configured in the Deck Editor before washing can occur. Clicking on a Span-8 Tip Wash ALP while the Transfer step is being configured places a check in Wash tips with, and uses default values of 2 mL of system liquid after dispensing 1 mL to waste.
- 12. In **Wash tips with**, enter the volume of system fluid to use when washing the outside of the tips.
- 13. In **after dispensing**, enter the volume of system fluid to use when washing the inside of the tips.
- 14. If using a Span-8 Pod on *FX*, select **Speed Pump** to use the speed pump when flushing the tips with system fluid, if desired.

OR

If using a Span-8 Pod on *NX-S8*, select **purge pump** to use the purge pump when flushing tips with system fluid, if desired.

**Note:** The speed pump or purge pump is an optional device that can be used to shorten the time it takes to wash tips. A speed pump or purge pump must be added as a device and configured with the Span-8 Pod in Hardware Setup to use it when washing tips.

- 15. If using a Multichannel Pod on *FX* or *NX-MC*, or a *3000*, select Change/Wash tips between sources, to change or wash tips each time a new source well is accessed.
  - > **3000** Tips can only be changed.

**Note:** Change/Wash tips between sources changes or washes tips any time it is required to aspirate liquid from a different well or wells than the previous aspirate operation. For example, if aspirating from quadrants of a 384-well microplate with a 96-channel head, Biomek Software instructs the instrument to use new tips when aspirating from each quadrant.

- If using a Multichannel Pod on *FX* or *NX-MC*, or a *3000*, select Change/Wash tips between destinations, to change or wash tips each time a new destination well is accessed.
  - ➤ 3000 Tips can only be changed.

**Note:** Change/Wash tips between destinations changes or washes tips after all dispense operations, except the last. Tips are removed after the last dispense operation only if unload them is selected in Tip Handling. If there is only one Destination Labware configured, Change tips between destinations does not change tips.

17. If using a Span-8 Pod on *FX* or *NX-S8*, select Change/Wash tips between transfers, to change or wash tips between each liquid transfer.

The Tip Handling configuration can be collapsed to allow more room during Source and Destination Labware configuration. When collapsed, the tip handling configuration is displayed as a sentence summary (Figure 15-21). Use the sentence summary to verify that tip handling has been configured appropriately.

To collapse or expand the Tip Handling configuration, click the arrow to the left of Tip Handling or the textual display (Figure 15-21).



Figure 15-21. Combine step — Adding Source Labware

## 15.3.2 Configuring Source Labware

A 'source' is a group of wells accessed at one time by the Biomek instrument to aspirate liquid; for example:

FX, NX-MC — A Multichannel Pod aspirates from either 96 or 384 wells at a time, depending on the head installed.

**Note:** If configuring a 384-well microplate using a 96-channel pod or a 1536-well microplate using a 384-channel pod as source labware, any combination of quadrants may be used. For example, if it is desired to transfer liquid from the first, second, and third quadrants but not the fourth, the source labware may be configured to transfer only from those quadrants.

- FX, NX-S8— A Span-8 Pod aspirates from one to eight wells at a time, depending on the selected probes.
- 3000 Depending on the tool loaded onto the head assembly, aspirates from either 1 or 8 wells at a time.

**Note:** If there are multiple sources in a **Combine** step, each source may have a different volume transferred, but all wells on each source must have the same volume transferred. To transfer different volumes from wells or quadrants on the same microplate, configure multiple sources using the same source plate but specifying different well or quadrants and volumes.

The Source Labware configuration for the Transfer and Combine steps includes specifying:

- Labware type
- Labware location
- Liquid amount to transfer (for Combine step only)
- Liquid type
- Well pattern or quadrants to access

**Note:** Quadrants are applicable only when accessing a 384-well microplate using a 96-channel pod or a 1536-well microplate using a 384-channel pod.

- Technique selection
- Pipetting height

**Note:** A single source is configured for the **Transfer** step, because the **Transfer** step moves liquid from a single source to one or more destinations. Multiple sources can be configured for the **Combine** step, because the **Combine** step moves liquid from multiple sources to a single destination.

To configure Source Labware for the Transfer and Combine steps:

- 1. Select **Click here to add a source** (Figure 15-21). The Source Labware configuration appears.
- 2. Click on the desired piece of labware in the Current Deck Display. The labware type and position for that piece of labware is entered automatically into the Source Labware configuration.

**Note:** If the selected labware is named, the name of the labware appears instead of the deck position.

OR

Select a **Source Labware** (Figure 15-22) type and specify the deck position.

**Note:** To configure additional Source Labware for a Combine step, select **Click here to add a source**, and then select another piece of labware from the Current Deck Display. A Destination labware must be configured before configuring additional source labware. If an incorrect piece of labware is selected, right-click on the labware title and select **Delete** on the menu that appears, or change the Source Labware selection by clicking anywhere in the source labware configuration area, and selecting another piece of labware.





3. Verify the deck position of the labware.

**Note:** A bright blue outline appears in the Current Deck Display around the labware designated as the **Source Labware**.

4. To specify the wells or quadrants in a microplate to access, double-click the source labware in the step configuration to zoom in on the labware (Figure 15-23).

**Note:** To call up a menu for Zoom and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows selection of wells or quadrants as targets for aspirate and dispense operations. Specify Selection as Text may also be used to enter variables or expressions.

**Note:** Specify Selection as Text is not applicable for 96-well microplates, reservoirs, or 384-well microplates accessed by a 384-channel pod.





- > **FX, NX-MC** Using a Multichannel Pod, some options are not available
- > **3000** Using a Multi-Tip Pipette Tool, some options are not available.

- 5. Select the wells from which to transfer using one of the following techniques:
  - FX, NX-MC A Multichannel Pod can only select quadrants of a microplate.
  - **FX, NX-S8** A Span-8 Pod can select individual wells.
  - 3000 A multi-tip pipette tool always selects groups of wells in columns of 8 while a single-tip pipette tool can select individual wells.
    - Choose Copy Pattern and select the desired pattern to use a previously defined well pattern created using the Well Patterns Editor (refer to Chapter 11, <u>Creating Well Patterns</u>).

**Note:** The pattern must be compatible with the head or tool installed and the labware type.

 Select Use pattern and choose the previously defined pattern from the list to use a pattern created in a Well Patterns Editor or Define Pattern step.

**Note:** Refer to Section 18.10, <u>*Define Pattern Step (including 3000)</u></u>, for more information about creating patterns using the Define Pattern step.</u>* 

- Define a well pattern using data sets (refer to Section 14.3.1, <u>Defining</u> <u>Well Patterns Using Data Sets</u>).
- Create a custom well pattern from the labware grid by dragging the mouse and using Ctrl and Shift on the keyboard.

**Note:** Hold **Shift** and click a well to select additional wells without deselecting any wells. Hold **Ctrl** and click a well to toggle the selection status of selected wells without affecting the status of other wells. If Shift or Ctrl is not held down when selecting wells, any previous selection is deleted.

**Note:** To save the new pattern, right click the labware graphic and select **Save Pattern** from the menu. Enter a name for the pattern in New Pattern and choose **OK**.

- Select the **Direction** in which source wells are mapped to destinations. Down first, then left to right is the default selection.
  - **FX, NX-S8** Direction options only apply when using a Span-8 Pod.

**Note:** Direction controls how source wells are mapped to the destination, not the physical order in which wells are transferred. The **Transfer** step automatically determines the most efficient way to complete the transfer.

• Down first, then left to right — goes down each column from top to bottom, then goes right to the next column (Figure 15-24)



Figure 15-24. Wells accessed down first, then left to right

#### OR

• To the right first, then top to bottom — goes across each row from left to right, then goes down to the next row (Figure 15-25)



Figure 15-25. Wells accessed to the right first, then top to bottom

**Note:** The direction can also be selected when the labware is zoomed out by using the Down, then right or Right, then down buttons beneath the labware.



7. Select the first well accessed in **Start**. At first selected well is the default selection.

FX, NX-S8 — Start options only apply when using a Span-8 Pod.

**Note:** Start controls how source wells are mapped to the destination, not the physical order in which wells are transferred. The **Transfer** step automatically determines the most efficient way to complete the transfer.

• At first selected well — moves in the selected Direction starting from the first well selected on the labware

OR

• After last marked well — moves in the selected Direction starting from the first selected well on the labware after the marked well

**Note:** The selected labware must have marks from a previous step with Mark last well that is used selected.



**Note:** The start location can also be selected when the labware is zoomed out by using the Start at selection or Start at last mark buttons beneath the labware.

**Note:** When transferring liquid, the first accessed source well is transferred to the first accessed destination well, the second accessed source well is transferred to the second accessed destination well, and so forth.

8. De-select Mark last well that is used, if it is not desired to mark wells. This option must be selected to use any options later in the method that make use of marks. For example, if Mark last well that is used is selected and the Transfer step runs out of sources before using all destinations, it marks the last well that had liquid transferred to it. Another Transfer step is transferring liquid to the same plate. By selecting After last marked well as the Start condition, the first source well will be transferred to the first unused destination well and continue according to the specified Direction. Mark last well that is used is selected by default.

> **3000**— Marks only apply when using a single-tip pipette tool.



**Note:** Mark last well that is used can also be selected when the labware is zoomed out by using the Set mark button beneath the labware.

- 9. Choose **Zoom Out** to return to the step configuration screen.
- For a Combine step, enter the amount of liquid (μL) to aspirate from the Source Labware. This will also be the amount of liquid dispensed into the destination labware. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue (Figure 15-26).

**Note:** Refer to Section 14.3.2, <u>Specifying Volumes Using Data Sets</u>, for information on specifying volumes using a data set.

11. In **Using liquid type**, select the type of liquid in the Source Labware (Figure 15-26).



Figure 15-26. Combine step — Source Labware configuration

12. Keep the auto-selected **Technique** (refer to Section 15.3.2.1, <u>Auto-Selection of</u> <u>a Technique</u>).

**Note:** The software automatically selects a **Technique** based on the liquid type, labware type, tip type, and volume being aspirated.

OR

Manually select the desired **Technique**, or customize the selected **Technique** (refer to Section 15.3.2.2, *Customizing and Saving a Technique*).

13. Leave the height as is to use the settings specified in the pipetting technique.

OR

Set the aspirate height manually using one of the following techniques:

• Set the aspirate height by positioning the cursor over the graphic of a tip inside a well (Figure 15-26). Click and drag the graphic up or down. The tip moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.

 $\bigcirc$ 

**Note:** The cursor changes to a hand when positioned over the graphic.

- Position the cursor in the graphic of a tip inside a well, and adjust the height using the ↑ and ↓ keys.
- Right-click on the graphic of a tip inside a well, and a menu appears. Select **Custom Height**, and **Custom Height** appears (Figure 15-27). Insert the **Height** in millimeters (mm) and, in **from**, select a reference point from the drop-down list.

**Note:** The phrase (Overrides Technique) appears below the graphic when the tip height is changed by manipulating the graphic (Figure 15-26), or when the technique is changed with the Customize option.

Custom Height	
Height:	mm
from Bottom	•
ОК	Cancel

Figure 15-27. Custom Height prompt

14. For a Combine step, repeat steps 1 to 13 for each Source Labware desired.

**Note:** A Destination Labware must first be selected before additional Source Labware can be configured.

#### 15.3.2.1 Auto-Selection of a Technique

Auto-Select option is the default setting in the Transfer and Combine steps configuration. Disabling Auto-Select provides access to a list of predefined techniques, as well as allowing access to the Technique Editor so changes can be made to existing techniques.

**Note:** For more information on creating, editing, and saving techniques, refer to Section 15.3.2.2, *Customizing and Saving a Technique*, and Chapter 9, *Understanding and Creating Techniques*.

#### 15.3.2.2 Customizing and Saving a Technique

Disabling Auto-Select provides access to a list of predefined techniques, as well as allowing access to the Technique Editor so changes can be made to existing techniques.

**Note:** For more information on creating, editing, and saving techniques, refer to Chapter 9, <u>Understanding and Creating Techniques</u>.

To select a **Technique** other than the technique selected automatically by Biomek Software:

- 1. Uncheck Auto-Select. The Technique: field is activated.
- 2. Select a **Technique:** from the list of predefined techniques.
- 3. Click **Customize** to modify the technique, if desired. The Technique Editor appears (Figure 15-28).

Technique Editor - [Custom]		
Pipetting Template: Span-8		
General Dispense Calibration Liquid Level Sensing Liquid Type		
Move within the well at 5 % speed.		
Dispense at 1.5 mm from the Bottom		
✓ Follow liquid level when aspirating or dispensing liquid		
☐ <u>I</u> ouch tips on the sides of the wells		
☑ Blowout all leading air gaps		
🥅 Mix after dispensing liquid		
Mi <u>x</u> 10 μL 1 time,		
Aspirate at 0 mm from the Liquid $\checkmark$ at 100 $\mu$ L/s.		
Dispense at 0 mm from the Liquid $\checkmark$ at 100 $\mu$ L/s.		
OK Cancel		

Figure 15-28. Technique Editor — Dispense options

**Note:** The Aspirate tab is virtually the same as the Dispense tab, with the exception of the Aspirate a trailing air gap after leaving the liquid and Prewet the tips pipetting options which appear on the Aspirate tab. When dispense operations are being configured, only the Dispense tab is displayed, while the Aspirate tab is displayed only when configuring aspirate operations. The Liquid Level Sensing tab only appears if the pod or tool supports liquid level detection.

4. Make desired changes in the applicable tabs of the Technique Editor.

- 5. Click **OK** to save the edited technique and return to the **Transfer** and **Combine** steps.
- 6. Click **Save As** to save an edited technique for use in this method and others.

**Note:** If Save As is not selected, the technique is saved with the current method, but it is not accessible to other methods or other steps within the current method.

### 15.3.3 Configuring Destination Labware

A 'destination' is a group of wells accessed at one time by the Biomek instrument to dispense liquid; for example:

FX, NX-MC — A Multichannel Pod aspirates from either 96 or 384 wells at a time, depending on the head installed.

**Note:** If configuring a 384-well microplate using a 96-channel pod or a 1536-well microplate using a 384-channel pod as source labware, any combination of quadrants may be used. For example, if it is desired to transfer liquid from the first, second, and third quadrants but not the fourth, the source labware may be configured to transfer only from those quadrants.

- FX, NX-S8 A Span-8 Pod dispenses to one to eight wells at a time, depending on the selected probes.
- 3000 Depending on the tool loaded onto the head assembly, aspirates from either 1 or 8 wells at a time.

**Note:** If there are multiple destinations, each destination may have a different volume transferred, but all wells on each destination must have the same volume transferred. For example, to transfer different volumes to quadrants on the same 384-well plate, configure multiple destinations using the same destination plate but specifying different quadrants and volumes.

The Destination Labware configuration for the Transfer and Combine steps includes specifying:

- Labware type
- Labware location
- Liquid amount to transfer (for Transfer step only)
- Liquid type
- Well pattern or quadrants to access

**Note:** Quadrants are applicable when accessing a 384-well microplate using a 96-channel pod or a 1536-well microplate using a 384-channel pod.

- Technique selection
- Pipetting height

**Note:** Multiple destinations can be configured for the **Transfer** step, because the **Transfer** step moves liquid from a single source to one or more destinations. A single destination is configured for the **Combine** step, because the **Combine** step moves liquid from one or more sources to a single destination.

To configure Destination Labware for the Transfer and Combine steps:

1. Select **Click here to add a destination** (Figure 15-29). The Destination Labware configuration appears.



Figure 15-29. Transfer step — adding destination labware

2. Click on the desired piece of labware in the Current Deck Display. The labware type and position for that piece of labware is entered automatically into the **Destination Labware** configuration.

**Note:** If the selected labware is named, the name of the labware appears instead of the deck position.

OR

Select a **Destination Labware** (Figure 15-30) type and deck position.

**Note:** Once the first **Destination Labware** has been selected, the **Source** Labware parameters are collapsed to display a sentence summary. To reopen the Source labware parameters, click anywhere in the **Source Labware** configuration area.

**Note:** Subsequent labware selections modify the active Source or Destination Labware configuration. To configure additional Destination Labware, select **Click here to add a destination**, or select outside a labware configuration, and then select another piece of labware from the Current Deck Display. If an incorrect piece of labware is selected, right-click on the labware title and select **Delete** on the menu that appears.





3. Verify the deck position of the labware.

**Note:** A bright yellow outline appears in the Current Deck Display around the labware designated as the **Destination Labware**.

4. To specify the wells or quadrants of a microplate to access, double-click the source labware in the step configuration to zoom in on the labware.

**Note:** To call up a menu for Zoom and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows selection of wells or quadrants as targets for aspirate and dispense operations. Specify Selection as Text may also be used to enter variables or expressions.

**Note:** Specify Selection as Text is not applicable for 96-well microplates, reservoirs, or 384-well microplates accessed by a 384-channel pod.

- 5. Select the wells from which to transfer using one of the following techniques:
  - FX, NX-MC A Multichannel Pod can only select quadrants of a microplate.
  - > **FX, NX-S8** Can select individual wells.
  - 3000 A multi-tip pipette tool always selects groups of wells in columns of 8 while a single-tip pipette tool can select individual wells.
    - Choose Copy Pattern and select the desired pattern to use a previously defined well pattern created using the Well Patterns Editor (refer to Chapter 11, <u>Creating Well Patterns</u>).

**Note:** The pattern must be compatible with the head or tool installed.

 Select Use pattern and choose the previously defined pattern from the list to use a pattern created in a Well Patterns Editor or Define Pattern step.

**Note:** Refer to Section 18.10, *Define Pattern Step (including 3000)*, for more information about creating patterns using the **Define Pattern** step.

- Define a well pattern using data sets (refer to Section 14.3.1, <u>Defining</u> <u>Well Patterns Using Data Sets</u>).
- Create a custom well pattern from the labware grid by dragging the mouse and using Ctrl and Shift on the keyboard.

**Note:** Hold **Shift** and click a well to select additional wells without deselecting any wells. Hold **Ctrl** and click a well to toggle the selection status of selected wells without affecting the status of other wells. If Shift or Ctrl is not held down when selecting wells, any previous selection is deleted.

**Note:** To save the new pattern, right click the labware graphic and select **Save Pattern** from the menu. Enter a name for the pattern in New Pattern and choose **OK**. Select the **Direction** in which source wells are mapped to destinations. Down first, then left to right is the default selection.

**FX, NX-S8** — Direction options only apply when using a Span-8 Pod.

**Note:** Direction controls how source wells are mapped to the destination, not the physical order in which wells are transferred. The **Transfer** step automatically determines the most efficient way to complete the transfer.

• Down first, then left to right — goes down each column from top to bottom, then goes right to the next column (Figure 15-31)



Figure 15-31. Wells accessed down first, then left to right

#### OR

• To the right first, then top to bottom — goes across each row from left to right, then goes down to the next row (Figure 15-32)



Figure 15-32. Wells accessed to the right first, then top to bottom



**Note:** The direction can also be selected when the labware is zoomed out by using the Down, then right or Right, then down buttons beneath the labware.

7. Select the first well accessed in **Start**. At first selected well is the default selection.

FX, NX-S8 — Start options only apply when using a Span-8 Pod.

**Note:** Start controls how source wells are mapped to the destination, not the physical order in which wells are transferred. The **Transfer** step automatically determines the most efficient way to complete the transfer.

• At first selected well — moves in the selected Direction starting from the first well selected on the labware

OR

 After last marked well — moves in the selected Direction starting from the first selected well on the labware after the marked well

**Note:** The selected labware must have marks from a previous step with Mark last well that is used selected.



**Note:** The start location can also be selected when the labware is zoomed out by using the Start at selection or Start at last mark buttons beneath the labware.

**Note:** When transferring liquid, the first accessed source well is transferred to the first accessed destination well, the second accessed source well is transferred to the second accessed destination well, and so forth.

8. De-select Mark last well that is used, if it is not desired to mark wells. This option must be selected to use any options later in the method that make use of marks. For example, if Mark last well that is used is selected and the Transfer step runs out of sources before using all destinations, it marks the last well that had liquid transferred to it. Another Transfer step is transferring liquid to the same plate. By selecting After last marked well as the Start condition, the first source well will be transferred to the first unused destination well and continue according to the specified Direction. Mark last well that is used is selected by default.

> FX, NX-S8 — Marks only apply when using a Span-8 Pod.



**Note:** Mark last well that is used can also be selected when the labware is zoomed out by using the **Set mark** button beneath the labware.

9. Choose **Zoom Out** to return to the step configuration screen.

 For a Transfer step, enter the amount of liquid (μL) to dispense into the Destination Labware. This will also be the amount of liquid aspirated from the source labware. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue (Figure 15-33).

**Note:** Refer to Section 14.3.2, <u>Specifying Volumes Using Data Sets</u>, for information on specifying volumes using a data set.

🌵 Biomek	(B Software - Transfer* [Revision]	1: Development]		
File Edit	Project Instrument Execution Op	otions Help		
່ 🗅 🖻 (	- <b>6</b> () () () () () () () () () () () () ()			
æ	Start	Use god Pod2 💌 for transfer.	Use probes 1 2 3	4 5 6 7 8
Instrument	🐔 Instrument Setup	▽ Load Span_8_200uL tips, change betwe	een transfers, and unload	them when finished.
Setup	Transfer 50 µL from			Amount to dispense
Transfer	Finish	Ω 	-2.00 mm from liquid	Liquid type selection
	-	Destination: Dest1		
Combine	Adjust dis	spense height	╶п┢╵п╽	BCFlat96 at Dest1
				50 µL of Water ▼
Move Labware				Iechnique: Span-8 Low 80
8		7441600	1.50 mm from bottom	Ser
Conti	nue adding		here to add a des	stination
Desti	ination Labware by		nere to add a des	Technique selection
clicki	ng this button, or by	Stop when finished with destinations. Dis	spense up to 1 time per d	raw. <u>R</u> eplicate each well 1 time.
- select	ing more labware in			9898989
the C	urrent Deck Display.			
		Pierce Pesti Pi	13 <u>P17</u>	
			15 P19	
Transfer*	BIOMEKHX BIOMEKHX ETC: 0:00:03	J		

Figure 15-33. Transfer step — destination labware technique configuration

- 11. Verify that the liquid type indicated is correct. The liquid type dispensed to the Destination Labware should match the liquid type in the Source Labware configuration.
- Keep the auto-selected **Technique** (refer to Section 15.3.2.1, <u>Auto-Selection of</u> <u>a Technique</u>).

**Note:** The software automatically selects a **Technique** based on the liquid type, labware type, tip type, and volume being aspirated.

OR

Manually select the desired **Technique**, or customize the selected **Technique** (refer to Section 15.3.2.2, *Customizing and Saving a Technique*).

13. Leave the height as is to use the settings specified in the pipetting technique.

#### OR

Set the dispense height manually using one of the following techniques:

• Set the dispense height by positioning the cursor over the graphic of a tip inside a well (Figure 15-33). Click and drag the graphic up or down. The tip moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.

**Note:** The cursor changes to a hand when positioned over the graphic.

- Position the cursor in the graphic of a tip inside a well, and adjust the height using the ↑ and ↓ keys.
- Right-click on the graphic of a tip inside a well, and a menu appears. Select **Custom Height**, and **Custom Height** appears (Figure 15-27). Insert the **Height** in millimeters (mm) and, in **from**, select a reference point from the drop-down list.

**Note:** The phrase (Overrides Technique) appears below the graphic when the tip height is changed by manipulating the graphic, or when the technique is changed with the Customize option.

14. For a Transfer step, repeat steps 1 to 13 for each additional piece of Destination Labware.

#### 15.3.3.1 Auto-Selection of a Technique

Auto-Select option is the default setting in the Transfer and Combine steps configuration. Disabling Auto-Select provides access to a list of predefined techniques, as well as allowing access to the Technique Editor so changes can be made to existing techniques.

**Note:** For more information on creating, editing, and saving techniques, refer to Section 15.3.2.2, *Customizing and Saving a Technique*, and Chapter 9, *Understanding and Creating Techniques*.

#### 15.3.3.2 Customizing and Saving a Technique

Disabling Auto-Select provides access to a list of predefined techniques, as well as allowing access to the Technique Editor so changes can be made to existing techniques.

**Note:** For more information on creating, editing, and saving techniques, refer to Chapter 9, <u>Understanding and Creating Techniques</u>.

To select a **Technique** other than the technique selected automatically by Biomek Software:

- 1. Uncheck Auto-Select. The Technique: field is activated.
- 2. Select a **Technique:** from the list of predefined techniques.



 Click Customize to modify the technique, if desired. The Technique Editor appears (Figure 15-34).

Technique Editor - [Custom]		
Pipetting Template: Span-8		
General Dispense Calibration Liquid Level Sensing Liquid Type		
Move within the well at $5$ % speed.		
Dispense at 1.5 mm from the Bottom 💌		
Eollow liquid level when aspirating or dispensing liquid		
☐ <u>I</u> ouch tips on the sides of the wells		
☑ Blowout all leading air gaps		
Mix after dispensing liquid		
Mi <u>x</u> 10 μL 1 time.		
Aspirate at 0 mm from the Liquid $\checkmark$ at 100 $\mu$ L/s.		
Dispense at 0 mm from the Liquid 💌 at 100 µL/s.		
OK Cancel		

**Note:** refer to Chapter 9, <u>Understanding and Creating Techniques</u>, for more information about the Technique Editor.

Figure 15-34. Technique Editor — Dispense options

**Note:** The Aspirate tab is virtually the same as the Dispense tab, with the exception of the Aspirate a trailing air gap after leaving the liquid and Prewet the tips pipetting options which appear on the Aspirate tab. When dispense operations are being configured, only the Dispense tab is displayed, while the Aspirate tab is displayed only when configuring aspirate operations.

**Note:** If the pod performing the pipetting operation is a Span-8 Pod, there is an additional Liquid Level Sensing tab displayed as part of the Technique Editor.

- 4. Make desired changes in the applicable tabs of the Technique Editor.
- 5. Click **OK** to save the edited technique and return to the **Transfer** and **Combine** steps.
- 6. Click **Save As** to save an edited technique for use in this method and others.

**Note:** If Save As is not selected, the technique is saved with the current method, but it is not accessible to other methods or other steps within the current method.

#### 15.3.4 Configuring Transfer Details

Transfer Details (Figure 15-35) includes specifying:

- a stop condition for transfer operations.
- the number of replicates
  - **FX, NX-S8** Replicates are available only with a Span-8 Pod.
- a repeat pipetting configuration.
- a maximum tolerance in timing between transfer operations (Figure 15-36).



Figure 15-35. Transfer step — additional Transfer Details

To configure Transfer Details for the Transfer and Combine steps:

- 1. Click the arrow or sentence summary below the labware configuration section to expand Transfer Details.
- 2. In **Stop when finished with**, choose to determine when the liquid transfer operation stops: after Sources, Destinations, or Either.
  - Sources the Transfer or Combine is complete when it runs out of sources
  - **Destinations** the Transfer or Combine is complete when it runs out of destinations
  - Either the Transfer or Combine is complete when it runs out of either sources or destinations

**Note:** The default setting for Stop when finished with is Destinations to avoid cross contamination.

**Note:** If there are more source wells than destination wells and **Stop when** finished with is set to **Sources**, after dispensing to the final destination it will go to the first selected destination and continue until it runs out of sources. If **Destinations** is selected, it will stop after dispensing to the last destination and not use the remaining sources. If **Either** is selected, it will stop after dispensing to the last destination and not use the remaining sources.

If marks are being used, the last source and/or destination wells used will be marked. Future operations may continue from the marked wells.

- 3. In **Replicate each well**, enter the number of destination wells to transfer each source well to. Each source well is transferred to the specified number of consecutive destination wells.
  - FX, NX-S8 Replicates are available only with a Span-8 Pod.
- Select one of the repeat pipetting configuration options: Dispense up to or Aspirate at most.

**Note:** If Tip Handling is configured to Change tips between destinations, the repeat pipetting options are unavailable for selection.

**Note:** Only one repeat pipetting option may be selected. Either **Dispense** up to or **Aspirate at most** may be selected, but not both.

5. In **Dispense up to**, select the number of times the instrument is allowed to dispense per draw (aspirate). For example, if dispensing 75 μL to each of six microplates, three aspirate operations of 150 μL can each dispense into two destinations. In this case, **Dispense up to** is set to no fewer than two.

**Note:** The default setting in **Dispense** up to is one, which does not allow a repeat pipetting operation. Settings of two or more allow repeat pipetting.

OR

In **Aspirate at most**, select the maximum volume to aspirate from the source for repeat pipetting. For example, if dispensing 75  $\mu$ L to each of six microplates, three aspirate operations of 150  $\mu$ L can each dispense into two destinations. In this case, **Aspirate at most** is set to no less than 150  $\mu$ L.

- 6. In **Replicates**, enter the number of times each source well should be transferred to the destination labware. For example, if **Replicates** is set at **3**, the first source well is transferred to the first three destination wells; the second source well is transferred to the next three destination wells; and so forth.
- Select Advanced to change the default setting of applying a Just-In-Time (JIT) block to aspirate and dispense operations. Just In Time synchronizes the execution of the substeps within the Transfer or Combine step. The Advanced Options prompt appears (Figure 15-36).

✓ Tie Aspirates to Dispenses with a Just-In-Time step.			
OK Cancel			

Figure 15-36. Advanced button — JIT Block

- 8. Uncheck **Tie Aspirate to Dispense with a JIT Block** to deactivate the JIT block.
- 9. Click **OK** to save any changes.

The Transfer Details configuration can be collapsed to allow more room during Source and Destination Labware configuration. When collapsed, the Transfer Details configuration is displayed as a sentence summary. Use the sentence summary to verify that Transfer Details has been configured appropriately.

To collapse or expand the **Transfer Details** configuration, click the arrow to the left of **Transfer Details** or the textual display.

# 15.4 Move Labware Step

- > **3000** The gripper tool is required to move labware.
- > **NX-S8** The optional gripper must be installed to move labware.

The Move Labware step moves labware from one position on the Biomek instrument deck to another position. Move Labware can also remove labware from the deck by placing it in an appropriate disposal position, or to an external hardware device without halting the Biomek instrument or the method.

## 15.4.1 Inserting a Move Labware Step



Insert a Move Labware step into the Method View (Figure 15-37).

File Edit Project Instrument Execution Options Help Start Instrument Setup Wove Laboware Finish Wove the entire stack of laboware from the stack. Move the topmost Move	🌵 Biomek	® Software - Method3* [New]		L X
Start         Instrument Setup         Move Labware         Finish         Move the entire stack of labware at the source position.         Move the topmost         piece of labware from the stack.         Move the topmost         Pause         Comment         Instrument Setup         Move Labware         Move the topmost         Pause         Comment         Instrument         Instrument Setup         Move tabware         Move Labware         Move tabware         Instrument         Instrument         Instrument         Instrument         Instrument         Move Labware         Move the topmost         Pause         Instrument         Instrument <tr< th=""><th>File Edit</th><th>Project Instrument Execution</th><th>Options Help</th><th></th></tr<>	File Edit	Project Instrument Execution	Options Help	
Start   Instrument Setup   Move Labware   Finish   Wove the entire stack of labware at the source position. Move the tentire stack of labware at the source position. Move the tentire stack of labware from the stack. Move Labware Move Labware from the stack. Move the tentire stack of labware at the source position. Move the tentire stack of labware at the source position. Move the tentire stack of labware at the source position. Move the tentire stack of labware is moved from stacks on the deck. <b>Move Labware step Configuration</b> The stacking options configure how labware is moved from stacks on the deck. <b>Connent Current deck display</b> The origination position is outlined in blue. The destination location is outlined in yellow.	0 🖻 (	1 8 8 4 X B 8		
Instrument Setup	æ	🚦 Start	Using pod Pod1	
Wove Labware   Finish     Move the online     Move the topmost     Move the topmost     Discourse     Move the topmost        Discourse                 Parent <th>TL Instrument</th> <th>🐔 Instrument Setup</th> <th>Move labware from P6 💌 to 🖭</th> <th></th>	TL Instrument	🐔 Instrument Setup	Move labware from P6 💌 to 🖭	
Finish       Move stack, leaving the bottom piece of labware at the source position.         Move from the topmost       piece of labware from the stack.         Move Labware Step Configuration       The stacking options configure how labware is moved from stacks on the deck.         Move from stacks on the deck.       Image: Start and the origination position is outlined in blue. The destination location is outlined in yellow.	Berup	📐 Move Labware	Move the entire stack of labware.	
Transfer       Move the topmost       piece of labware from the stack.         Move       Move Labware step configuration         Move       The stacking options configure how labware is moved from stacks on the deck.         Pass       Image: Comment         Comment       Image: Comment         Image: Comment       Image: Comment	10 M	Einish	O Move stack, leaving the bottom piece of labware at the source position.	
Work       Move Labware step configuration         The stacking options configure how labware is moved from stacks on the deck.         Pause         Comment         Image: Comment	Transfer		C Move the topmost 1 piece of labware from the stack.	
Comment       Move Labware step configuration         Move Labware step configuration       The stacking options configure how labware is moved from stacks on the deck.         Pause       Image: Comment         Comment       Image: Comment         Image: Comment       Image: Comment </td <td></td> <td></td> <td><u> </u></td> <td></td>			<u> </u>	
Move Labovare       The stacking options configure how labovare is moved from stacks on the deck.         Pause       Image: Comment         Comment       Image: Comment         Image: Comment       Image: Comment	Combine		Move Labware step configuration	
Move Tabware Torment The origination position The origination position is outlined in blue. The destination location is outlined in yellow.			The stacking options configure how labware is	
Pause Pause Comment Comment Comment Current deck display The origination position is outlined in blue. The destination location is outlined in yellow. Mathed 21 BiomedEV Elici 0.00.02	Move		moved from stacks on the deck.	
Pause Comment Pause Comment Pause Comment Pause Comment Pause Comment Pause	18			
Comment	<b>L</b> Pause			
Comment	E			
Comment	4			
P1       P13       P17         P2       P14       P18         P15       P19       United in blue. The destination location is outlined in yellow.	Comment		Current deck dis	play 🗐
P1       P13       P17       is outlined in blue. The destination location is outlined in yellow.         Mathed21       BinnelEY       ETC: 0.00002       ETC: 0.00002			The origination positive of the prigration posit	ition
P2       P14       P18       destination location is outlined in yellow.         Mathed2t BiomelEY ETC: 0.000/02       EtC: 0.000/02			<b>P1 [</b> ] <b>[</b> ] <b>P13 P17 [</b> ] is outlined in blue <b>[</b> ]	The
Method 21 Biometery ETC: 0.00002			P2 P14 P18 destination location	is
Mathed <sup>34</sup> Biomeley ETC: 0:00:02			P3 P15 P19 outlined in vellow	
Mathad3* BiomelEY ETC: 0:00:02				
TECHOUS DIVINEN A DIVINEN A LTC, U(UU(U)	Method3*	BiomekFX BiomekFX ETC: 0:00:	3	

Figure 15-37. Move Labware step and configuration

## **15.4.2** Configuring the Move Labware Step



CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height, and failure to choose each ALP correctly in the software may result in collisions between pod(s) and ALPs during operation.

To configure a Move Labware step, complete the following steps:

- 3000 To use the Move Labware step with the Biomek 3000, a gripper tool must be added and configured in Hardware Setup (refer to the Biomek® 3000 Laboratory Automation Workstation User's Manual, Section 12.3, <u>Adding and Removing Tools and Devices</u>) and placed on a tool rack in an Instrument Setup step (refer to Section 15.2, <u>Instrument Setup Step</u>).
- If using a dual-pod *FX* instrument, specify the Pod to move the labware in Using Pod. The pod configured as the default pod is automatically selected. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.
- 2. In Move labware from, select the original deck position for the labware from the Current Deck display. This instructs the Biomek instrument to Move labware from a specific deck position in preparation for leaving it at a final destination.
- 3. In to, select the final deck position for the labware from the Current Deck display. This instructs the Biomek instrument to move the labware to a final deck position.
- 4. Select the desired option for moving stacked labware:
  - FX, NX-MC Gripper on the Multichannel Pod can move up to four plates at a time.
  - NX-S8 The optional gripper, if installed, can move up to four plates at a time.
  - > **3000** Gripper tool can move up to three microplates at a time.
    - Move the entire stack of labware moves all labware in the stack; this option should be selected when moving a single unstacked piece of labware.
    - Move stack, leaving the bottom piece of labware at the source position moves all labware in the stack except for the bottom piece.

**Note:** Selecting Move stack, leaving the bottom piece of labware at the source position when the source deck position contains only one piece of labware results in an error.

 Move the topmost . . .piece(s) of labware from the stack moves only the specified number of labware from the top of the stack.

**Note:** Refer to Section 7.3.6.4.1, *<u>Biomek Stacking Rules</u>*, for complete information on using stacks on Biomek instruments.

## 15.5 Pause Step

The **Pause** step is used to either halt a deck position or device during a method for a specified amount of time, or pause the instrument for an indefinite period of time. During the pause, the instrument may still move and act on other deck positions. Depending on the purpose of the pause, the step can be configured in one of two ways:

- To incubate a piece of labware in a specific position for a specific amount of time, configure the desired time in seconds to hold that position idle and unavailable for interaction with the instrument (Figure 15-38). Refer to Section 15.5.1, *Configuring a Pause for a Specific Deck Position or Pod*.
- To manipulate the deck by hand during a method, configure the step to pause the instrument for an indefinite period of time (Figure 15-40). When the **Pause** step halts the instrument, the amber indicator lights on the front of the unit are activated, and a prompt appears. The prompt displays the text inserted in the **Pause** step configuration and requires acknowledgement before the method continues. Refer to Section 15.5.2, <u>Pausing the Whole</u> <u>System and Displaying a Message</u>.

**Note:** When the **Pause** step occurs during a method run, the Biomek pod remains in the position of the last action completed before the pause. If the purpose of the **Pause** step is to move labware on or off the deck, a **Move Pod** step must be inserted in the method prior to the **Pause** step so that the pod is out of the way (refer to Section 16.11, *Move Pod Step*).

Insert a Pause step into the Method View (Figure 15-38).

0

Pause



Figure 15-38. Inserting a Pause step in a method

## 15.5.1 Configuring a Pause for a Specific Deck Position or Pod

To configure a Pause for a specific deck position:

- 1. Choose **Pause** in the **Pause** step configuration (Figure 15-38).
- 2. In Pause, select the deck position or pod in the Current Deck Display or from the drop-down list (Figure 15-39).

**Note:** To pause the instrument for a specified length of time, select **the whole system** in Pause.



Figure 15-39. Pausing a deck position for a specified length of time

3. Choose the length of the pause, in seconds.

**Note:** The method continues with operations not affected by the **Pause**, when possible. If the next step is dependent upon the deck position or pod paused, the method waits until the **Pause** has been completed.

## 15.5.2 Pausing the Whole System and Displaying a Message

To configure a Pause for the entire Biomek instrument and display a message:

1. Select **Pause the whole system and display this message:** in the Pause step configuration (Figure 15-40).



Figure 15-40. Pause step that provides instruction

2. Enter the message to display when the Pause is initiated.

**Note:** When Pause the whole system and display this message: is selected, a prompt (Figure 15-41) appears in the Biomek Software with the text entered in the Pause step configuration. When the action requested is completed, click **OK** to continue the method, or select Abort to halt the method without completing it.

Biomek® Software	
Add BCFlat96 microplates to P17 and P18. Remove reservoir from P4.	
<u> </u>	
	9/30/2003 2:23:37 PM

Figure 15-41. Pause message prompt

# 15.6 Comment Step

The **Comment** step documents a method and/or adds instructions in the Method View. Text inserted into the **Comment** step configuration may indicate the purpose of a segment of a method, prompt for an action, or convey any other desired information in the Method View. The Comment step does not initiate any actions on the Biomek instrument and is used only to provide descriptive information and notes to a method.

**Note:** Comments are printed when a method is printed. Choose **Print** from the File menu to print a method (refer to Section 12.19, *Printing a Method*).

Insert a Comment step into the Method View (Figure 15-42).



To configure Comment:

- 1. In **Description**, enter short summary of the comment. The text entered in **Description** is used as the step caption in the Method View.
- 2. In **Comment**, enter the full text for the comment.
- 3. Click the next step, and the Comment step in the Method View displays the text entered in Description (Figure 15-42).



Figure 15-42. Comment step after text is inserted

# Using the Intermediate Step Palette

# 16.1 Overview

Steps on the Intermediate Step Palette provide individual control over liquid-handling functions outside the **Transfer** and **Combine** steps, including tip handling operations, flow of control, tip and tip box relocation options, and pod movement.



CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height and failure to choose the correct ALP in the Deck Editor may result in collisions between pod(s) and ALPs during operation.

The steps available in the Intermediate Step Palette are:

preparation for the Dispense step.

FX, NX-S8 — The pipetting and tip handling steps on the Intermediate step palette cannot be used with the Span-8 Pod; refer to Chapter 18, <u>Using</u> <u>the Span-8 Step Palette</u>, to perform similar functions.

Aspirate — aspirates a specified amount of liquid from a single source in

- Aspirate
- Dispense dispenses a specified amount of liquid into a single destination, following the Aspirate step.





Mix — mixes the contents of a piece of labware using repeated aspirate and dispense.



- Wash washes tips by aspirating and dispensing repeatedly at a Tip Wash ALP.
  - 3000 The Wash step cannot be used with the Biomek 3000 and does not appear on the Intermediate Step Palette.

•



New Tips— loads new tips to the head.

- Unload Tips
- Unload Tips unloads tips from the head.
- Loop
- Loop executes one or more steps a configured number of times.



• Cleanup — places tips and tip boxes at specified locations.



• Move Pod — moves the pod to a specified deck location.



• Group Step — groups a series of steps in a nested fashion under a logical name that appears in the Method View.

When one of the above steps is added to a method, the configuration associated with that step appears on the right side of the Biomek Software main editor, in step configuration (Figure 16-1).

**Note:** Steps not configured appropriately generate errors when the method is validated or run.

# 16.2 Displaying the Intermediate Step Palette

In order to add intermediate steps to a method, display the Intermediate Step Palette (Figure 16-1).

- FX, NX-S8 Only the Loop, Cleanup, Move Pod, and Group steps appear on the Intermediate step palette for Span-8 only instruments.
- > **3000** The Wash step does not appear on the Intermediate step palette.





To display the Intermediate Step Palette, complete the following:

• Right-click any empty palette space, and the Step Palette menu appears. Select **Intermediate**.

OR

• From the toolbar, select **Options>Toolbars>Intermediate**.

# 16.3 Aspirate Step

NX-S8 — To aspirate using a Span-8 Pod, refer to Section 18.4, <u>Span-8</u> <u>Aspirate Step</u>.

The Aspirate step removes liquid from source labware in preparation for dispensing to destination labware. Each tip of the head or tool descends into the wells of the source labware and aspirates the desired volume. The Aspirate step is often used in conjunction with the Dispense, New Tips, and Unload Tips steps (Figure 16-2), but it can be used with any combination of steps. Aspirate provides more direct control over pipetting operations than the Transfer or Combine steps because any step or combination of steps can complete the liquid transfer operation started with the Aspirate step.

3000 — A Load Tool or Change Tool step must be inserted and configured to load an appropriate pipetting tool prior to an Aspirate step in the method (refer to Section 20.4, <u>Load Tool Step</u>). Depending on the tool loaded, either a single well or eight wells in a column are selected when aspirating with the Biomek 3000 instrument.

**Note:** A Loop step and variable can be used to aspirate to more wells on a microplate (refer to Section 16.9, *Loop Step*).

The Aspirate step configuration includes specifying:

- Labware type accessed during aspirate
- Position of the labware on the deck
- Type of liquid to aspirate
- Volume to aspirate
- Pod that performs aspirate
- Tip refresh, if desired, and the type of tips used
- Technique selection
- Quadrant or well(s) accessed, if applicable
  - FX, NX-MC Quadrants are applicable only when accessing a 384well microplate using a 96-channel pod or a 1536-well microplate using a 384-channel pod.
  - 3000 Either a single well or eight wells in a column are selected, depending on the tool loaded.

Insert an **Aspirate** step into the method (Figure 16-2).



👫 Biomek® Software - Method5\* [New] <u>\_ | × | </u> Aspirate step configuration File Edit Project Instrument Execution Options Help Start Ð 5 Instrument Setup Aspirat 🚺 New Tips Ô 🕦 Aspirate from H20 Dispen Dispense to Dest1 æ Unload Tips Mi 7777 3.20 mm from bottom [Overrides Technique] 伵 Finish Pod: Pod1 • Labware Type: Reservoir • • Refresh Tips AP96\_200uL -Position: H20 Liquid Type: Water • Method View <u>V</u>olume: 50 μL Ne Note that tips are loaded Customize... Save As. Auto-Select prior to the Aspirate step Iechnique: Reservoir  $\mathbf{T}$ Unle and unloaded after the Dispense step. **Refresh Tips** Refresh Tips is unchecked when loading tips prior to > the Aspirate step. Refresh Tips loads a new box of tips Cleanup prior to the Aspirate operation. **Current Deck Display** The Current Deck Display is \_ ⊠∭∭∭ used for labware selection, 120 Eth P12 P16 and it displays the status of Dest1 Dest2 P13 the deck upon completion of P10 P14 P18 the previous step. P11 P15 Method5\* BiomekFX BiomekFX ETC: 0:00:16

Figure 16-2. Aspirate step and configuration

## **16.3.1** Configuring the Aspirate Step

When an Aspirate step is added to a method, the Aspirate step configuration appears (Figure 16-2).

To configure the Aspirate step:

1. Click on the desired piece of labware in the Current Deck Display from which to aspirate. Information for that piece of labware is entered automatically into the Aspirate step configuration.

OR

Select a Labware Type (Figure 16-2).

2. Verify the **Position** of the labware.

**Note:** A bright blue outline appears in the Current Deck Display around the selected labware.

3. Indicate the Liquid Type contained in the selected labware.

**Note:** The liquid type defaults to the liquid specified for the labware in Instrument Setup.

- Enter the Volume aspirated from the selected labware. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue (Figure 16-2).
- 5. Select the well(s) or quadrant to aspirate from by highlighting the desired quadrant on the labware graphic.
  - FX, NX-MC Quadrants may be selected when aspirating from a 384well plate using a 96-channel pod or a 1536-well plate using a 384channel pod.
  - 3000 Either a single well or eight wells in a column are selected, depending on the tool loaded.

**Note:** To zoom in on the labware, double-click on the labware. To call up a menu for Zoom In and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows the selection of a quadrant as a target for the aspirate operation. Specify Selection as Text is not applicable for 96-well microplates with a Multichannel Pod on *FX* or *NX-MC*, but is applicable with pipetting tools on the **3000**. It is never applicable with reservoirs.

- 6. If using a dual-pod FX instrument, specify the Multichannel Pod performing the aspirate operation from the drop-down list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.
- 7. Check **Refresh Tips**, if desired, and specify the **tip type**. Refresh Tips loads new tips to the pod whether tips are currently loaded or not.

**Note:** If Biomek Software determines that tips are already loaded, the default setting for **Refresh Tips** is Off. If Biomek Software does not determine tips are loaded, the default setting for **Refresh Tips** is On.
8. The software automatically selects a Pipetting Technique based on the liquid type, labware type, tip type, and volume being transferred. To override this selection, deselect **Auto-Select** and choose the desired **Technique** 

OR

Choose **Customize** and configure the technique as desired in the **Technique** Editor.

**Note:** For more information on customizing, configuring, and saving a technique, refer to Section 12.3.1, *<u>Understanding Techniques</u>*.

9. In some cases, it may be necessary to override the aspiration height. Leave the height as is to use the settings specified in the pipetting technique.

OR

Set the aspirate height manually using one of the following techniques:

 Set the aspirate height by positioning the cursor over the graphic of a tip inside a well (Figure 16-3). Click and drag the graphic up or down. The tip moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.

**Note:** The cursor changes to a hand when positioned over the graphic. The phrase (Overrides Technique) appears below the graphic when the tip height is changed by manipulating the graphic (Figure 16-3) or when the technique is changed with the Customize option. For more information on selecting and editing a technique for this method, refer to Chapter 9, <u>Understanding and Creating Techniques</u>.

- Adjust the aspirate height by selecting the graphic of a tip inside a well and using the ↑ and ↓ keys. The textual representation of the height, which is displayed below the graphic, adjusts in 0.1 millimeters (mm) increments.
- Right-click on the graphic of a tip inside a well, and a menu appears. Select **Custom Height**, and the Custom Height prompt appears (Figure 16-4). Insert the height in mm and, in from, select a reference point from the drop-down list. Click **OK** to save the edits and return to the Aspirate step configuration.

**Note:** For more information on creating, editing, and saving techniques for use with other methods, refer to Chapter 9, <u>Understanding and Creating Techniques</u>.



🌵 Biomek® Software - Method5	* [New]	×
File Edit Project Instrument Ex	ecution Options Help	
Start		
Aspirate 😤 Instrument :	Setup	
New Tips	Adjust the tip height for	
Dispense Aspirate fro		
👍 🝈 Dispense to	Desti	
Mix Unload Tips	3.20 mm from bottom	
Finish	[Overrides Technique]	
Wash	Labware type: Heservoir Pod Pod Pod Pod	
atta	Liquid Type: Water	
New Tins	<u>V</u> olume: 50 μL	
	R Auto-Select Outromize   Height display	
VVVV		-
Childad hips		
	Notifies that the technique is not	
Loop	using the default settings	
Cleanup		
Move Pod		
Group	pest1 pest2 p13 p17	
	P2 P6 P10 P14 P18	
	P3 P7 P11 P15 P19	
Method5* BiomekFX BiomekFX E	TC: 0:00:16	

Figure 16-3. Aspirate step — Adjusting the aspirate height

Custom Height					
Height:	mm				
from Bottom	•				
ОК	Cancel				

Figure 16-4. Custom Height prompt

10. Select a step that occurs after the Aspirate step or the **Finish** step to validate the step configuration.

## 16.4 Dispense Step

FX, NX-S8 — To dispense using a Span-8 Pod, refer to Section 18.5, <u>Span-8 Dispense Step</u>.

The Dispense step dispenses liquid into a single destination labware following aspirate. Each tip of the head or tool descends into the wells of the destination labware and dispenses the desired volume. The Dispense step is often used in conjunction with the Aspirate, New Tips, and Unload tips steps, but it can be used with any combination of steps. Dispense provides more direct control over pipetting operations than the Transfer or Combine steps because any step or combination of steps can initiate the liquid transfer operation.

**Note:** Tips must be loaded and have liquid in them from a previous step, such as an Aspirate step, prior to using a Dispense step.

**Note:** A Loop step and variable can be used to dispense from more wells on a microplate (refer to Section 16.9, *Loop Step*).

The Dispense step configuration includes specifying:

- Labware type accessed during dispense
- Position of the labware on the deck
- Type of liquid to dispense
- Volume to dispense
- Pod that performs dispense
- Empty tips, if desired
- Technique selection
- Quadrant or well(s) accessed, if applicable
  - FX, NX-MC Quadrants are applicable only when accessing a 384well microplate using a 96-channel pod or a 1536-well microplate using a 384-channel pod.
  - 3000 Either a single well or eight wells in a column are selected, depending on the tool loaded.



Insert a **Dispense** step into the method (Figure 16-5).



Figure 16-5. Dispense step and configuration

## 16.4.1 Configuring the Dispense Step

When a Dispense step is added to a method, the Dispense step configuration appears (Figure 16-5).

To configure the Dispense step:

1. Click on the desired piece of labware in the Current Deck Display. Information for that piece of labware is entered automatically into the **Dispense** step configuration.

OR

Select a Labware Type (Figure 16-5).

2. Verify the **Position** of the labware.

**Note:** A bright yellow outline appears in the Current Deck Display around the selected labware.

3. Indicate the Liquid Type contained in the selected labware.

**Note:** The liquid type defaults to the liquid specified for the labware in Instrument Setup.

- Enter the Volume dispensed to the selected labware. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue (Figure 16-5).
- 5. Select the well(s) or quadrant to dispense to by highlighting the desired well(s) or quadrant on the labware graphic.
  - FX, NX-MC Quadrants may be selected when dispensing to a 384-well plate using a 96-channel pod or a 1536-well plate using a 384-channel pod.
  - 3000 Either a single well or eight wells in a column are selected when dispensing with the Biomek 3000 instrument, depending on the tool loaded.

**Note:** To zoom in on the labware, double-click on the labware. To call up a menu for Zoom In and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows the selection of a quadrant as a target for the aspirate operation. Specify Selection as Text is not applicable for 96-well microplates with a Multichannel Pod on *FX* or *NX-MC*, but is applicable with pipetting tools on the **3000**. It is never applicable with reservoirs.

- 6. If using a dual-pod *FX* instrument, specify the Multichannel **Pod** performing the dispense operation from the drop-down list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or from the Current Deck Display.
- 7. Check **Empty Tips**, if desired. Empty Tips expels the Trailing Air Gap, the fluid contained in the tip, and the Blow Out all at one time, rather than as separate operations.

Note: The default setting for Empty Tips is Off.

8. The software automatically selects a Pipetting Technique based on the liquid type, labware type, tip type, and volume being transferred. To override this selection, deselect **Auto-Select** and choose the desired **Technique** 

OR

Choose **Customize** and configure the technique as desired in the **Technique** Editor.

**Note:** For more information on customizing, configuring, and saving a technique, refer to Section 12.3.1, <u>Understanding Techniques</u>.

9. In some cases, it may be necessary to override the aspiration height. Leave the height as is to use the settings specified in the pipetting technique.

#### OR

Set the aspirate height manually using one of the following techniques:

• Set the aspirate height by positioning the cursor over the graphic of a tip inside a well (Figure 16-6). Click and drag the graphic up or down. The tip moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.

**Note:** The cursor changes to a hand when positioned over the graphic. The phrase (Overrides Technique) appears below the graphic when the tip height is changed by manipulating the graphic (Figure 16-6) or when the technique is changed with the Customize option. For more information on selecting and editing a technique for this method, refer to Chapter 9, <u>Understanding and Creating Techniques</u>.

- Adjust the aspirate height by selecting the graphic of a tip inside a well and using the ↑ and ↓ keys. The textual representation of the height, which is displayed below the graphic, adjusts in 0.1 millimeters (mm) increments.
- Right-click on the graphic of a tip inside a well, and a menu appears. Select **Custom Height**, and the Custom Height prompt appears (Figure 16-7). Insert the height in mm and, in from, select a reference point from the drop-down list. Click **OK** to save the edits and return to the **Aspirate** step configuration.

**Note:** For more information on creating, editing, and saving techniques for use with other methods, refer to Chapter 9, <u>Understanding and Creating Techniques</u>.





Figure 16-6. Dispense step — Adjusting the dispense height

Custom Height	
Height:	mm
from Bottom	•
OK	Cancel

Figure 16-7. Custom Height prompt

10. Select a step that occurs after the **Dispense** step or the **Finish** step to validate the step configuration.

## 16.5 Mix Step

**FX, NX-S8** — The Mix step cannot be used with a Span-8 Pod.

The Mix step provides a mixing motion for the contents of a piece of labware by performing one or more aspirate and dispense operations within a single microplate. The Mix step can be used to mix the contents of a piece of labware that has been inactive for an extended length of time, prior to aspirating.

3000 — A Load Tool or Change Tool step must be configured with an appropriate pipetting tool prior to a Mix step in the method (refer to Section 20.4, *Load Tool Step*). Depending on the tool loaded, either a single well or eight wells in a column are selected when mixing.

**Note:** A Loop step and variable can be used to mix more wells on a microplate (refer to Section 16.9, *Loop Step*).

The Mix step configuration includes specifying:

- Labware type accessed during mix
- Position of the labware on the deck
- Type of liquid mixed
- Volume mixed
- Pod performing the mix
- Tip refresh, if desired, and the type of tips to load
- Number of aspirate and dispense cycles to perform during the mix process
- Technique selection
- Quadrant or wells accessed, if applicable
  - FX, NX-MC Quadrants are applicable only when accessing a 384well microplate using a 96-channel pod or a 1536-well microplate using a 384-channel pod.
  - 3000 Either a single well or eight wells in a column are selected, depending on the tool loaded.



Insert a **Mix** step into the method (Figure 16-8).



Figure 16-8. Mix step and configuration

## 16.5.1 Configuring the Mix Step

When a Mix step is added to a method, the Mix step configuration appears (Figure 16-8).

To configure the Mix step:

1. Click on the desired piece of labware in the Current Deck Display (Figure 16-8). Information for that piece of labware is entered automatically into the Mix step configuration.

OR

Select a Labware Type from the drop-down list.

2. Verify the **Position** of the labware.

**Note:** A bright blue outline appears in the Current Deck Display around the selected labware.

- 3. Select the well(s) or quadrant to mix by highlighting the desired well(s) or quadrant on the labware graphic.
  - FX, NX-MC Quadrants may be selected when mixing wells in a 384-well plate using a 96-channel pod or a 1536-well plate using a 384-channel pod.
  - 3000 Either a single well or eight wells in a column are selected when mixing wells, depending on the tool loaded.

**Note:** To zoom in on the labware, double-click on the labware. To call up a menu for Zoom In and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows the selection of a quadrant as a target for the aspirate operation. Specify Selection as Text is not applicable for 96-well microplates with a Multichannel Pod on *FX* or *NX-MC*, but is applicable with pipetting tools on the **3000**. It is never applicable with reservoirs.

- 4. Indicate the Liquid Type contained in the selected labware.
- Enter the Volume mixed in the selected labware. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue.
- 6. If using a dual-pod **FX** instrument, specify the Multichannel **Pod** performing the mix operation from the drop-down list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list.
- 7. Check **Refresh Tips**, if desired, and specify the **tip type**. Refresh Tips loads new tips to the pod whether tips are currently loaded or not.

**Note:** If Biomek Software detects that tips are already loaded, the default setting for **Refresh** Tips is Off. If Biomek Software does not detect tips, the Default setting **Refresh** Tips is On.

- 8. In **Mix**, enter the number of times the mix operation is performed.
- 9. The software automatically selects a Pipetting Technique based on the liquid type, labware type, tip type, and volume being transferred. To override this selection, deselect **Auto-Select** and choose the desired **Technique**

OR

Choose **Customize** and configure the technique as desired in the **Technique** Editor.

**Note:** For more information on customizing, configuring, and saving a technique, refer to Section 12.3.1, *<u>Understanding Techniques</u>*.

10. Select a step that occurs after the Mix step or the **Finish** step to validate the step configuration.

## 16.6 Wash Step

- FX, NX-S8 To wash tips on a Span-8 Pod, refer to Section 18.8, <u>Span-8</u> <u>Wash Tips Step</u>.
- 3000 The Wash step is not for use with the Biomek 3000 and does not appear on the Intermediate Step Palette.

The Wash step combines aspirate and dispense operations into one step that cleans tips using the Multichannel Tip Wash ALP. Use the Deck Editor to add and configure the Wash Station ALP.

The Wash step configuration includes specifying:

- Pod performing the wash operation.
- Wash station accessed.
- Type of liquid contained in the wash station.
- Volume used in the wash process.
- Number of wash cycles completed.

Insert a **Wash** step into the method (Figure 16-9).





Figure 16-9. Wash step and configuration

### 16.6.1 Configuring the Wash Step

When a Wash step is added to a method, the Wash step configuration appears (Figure 16-9).

To configure the Wash step:

- 1. If using a dual-pod *FX* instrument, specify the Multichannel **Pod** performing the wash operation from the drop-down list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or from the Current Deck Display.
- 2. Click on the Wash Station device in the Current Deck Display. Information for that wash station is entered automatically into the Wash step configuration.

OR

Select a Wash Station device from the drop-down list.

**Note:** The Wash Station ALP can be added to the deck and configured for use with the Deck Editor.

**Note:** A Wash Station does not have to be specified as long as the Liquid Type is selected. If no Wash Station is specified, the software determines the wash station with the specified Liquid Type that can be accessed by the pod.

3. Verify the Liquid Type contained in the selected Wash Station.

**Note:** The Liquid Type contained in the Wash Station ALP is configured in the Instrument Setup step. If the Wash Station is selected from the Current Deck Display, the information is automatically inserted.

4. In Volume, provide the volume of fluid to use when washing the tips.

**Note:** Enter the volume as a percent (including the percent sign, %) to specify the volume as a percent of the maximum volume of fluid contained in the tips in previous steps. For example, if the maximum volume of fluid transferred is 50  $\mu$ L, and the % is set for 110%, the **Wash** step washes the tips with 55  $\mu$ L of solution. However, if no fluid has been transferred and the volume is entered as a percent, no action is taken and tips are not washed.

- 5. In Wash Cycles, select the number of aspirate and dispense operations completed during the wash operation.
- 6. Select a step that occurs after the Wash step or the **Finish** step to validate the step configuration.

## 16.7 New Tips Step

FX, NX-S8 — To load new tips onto a Span-8 Pod, refer to Section 18.6, <u>Span-8 New Tips Step</u>.

The New Tips step instructs the instrument to load new tips at a specific point in a method. New Tips is used when precise tip control is required. New Tips is used in conjunction with the Unload Tips steps. Also, since tips must be loaded prior to aspirate, the New Tips step should be positioned prior to any Aspirate step if tips have not been previously loaded.

3000 — A Load Tool step must be configured with an appropriate pipetting tool prior to a New Tips step in the method (refer to Section 20.4, <u>Load Tool</u> <u>Step</u>).

The New Tips step is used in place of the Refresh Tips option in the Mix and Aspirate steps or the Load tips option in the Transfer and Combine steps, when more tip control is desired. New Tips is particularly useful with the Loop and Worklist steps; for example, when a Worklist step is performing a transfer that accesses the same tip box repeatedly at different points throughout the worklist, the tips can be loaded and unloaded as many times as their tip configuration allows.

Loading new tips is recommended when:

- The current tips have manipulated a caustic fluid.
- The current tips have been loaded the maximum number of times allowed in their Labware Properties.

**Note:** To increase the number of uses allowed for each set of tips, increase the Load no more than count in the Labware Properties configuration in the Instrument Setup step.

The integrity of the current tips is in question.



Insert a **New Tips** into the method (Figure 16-10).

🕼 Biomek® Software - Aspirate1* [Development		J×
File Edit Project Instrument Execution Options I		
Apprate Depense	Pod Pod V Ip: AP96_2004	
Dispense to Dest1	New Tips Step Configuration	
Wash New Tips		
TTTT Luload Tips		
Loop Cleanup		
Move Pod		
	TL1 Ref Eth P12 P16 Dest1 Dest2 P17 P2 P6 P10 P14 P18 P3 P7 P11 P15 P19	
Aspirate1* BiomekFX BiomekFX ETC: 0:00:03		

Figure 16-10. New Tips step and configuration

To configure New Tips:

- 1. If using a dual-pod *FX* instrument, specify the Multichannel **Pod** requiring new tips from the drop-down list. The pod configured as the default pod is displayed in **Pod**. If the other pod is desired, select the pod from the drop-down list or from the Current Deck Display.
- 2. Select the **Tips** loaded in the New Tips step from the drop-down list.

OR

Select the tip box from the Current Deck Display. Selecting tips from the Current Deck Display automatically updates Tips in the Step Configuration.

**Note:** The name of a specific tip box can be entered in Tips if tip boxes were assigned names in the Instrument Setup step. If a name is specified, the New Tips step only uses tip boxes with that name. However, if the tip box type is used, New Tips uses any available tip box of that type.

3. Select a step that occurs after the New Tips step or the **Finish** step to validate the step configuration.

## 16.8 Unload Tips Step

FX, NX-S8 — To unload tips from a Span-8 Pod, refer to Section 18.7, <u>Span-8 Tip Discard Step</u>.

The Unload Tips step instructs the instrument to unload tips at a specific point in a method. Unload Tips is used when precise tip control is required. Unload Tips is often used in conjunction with the New Tips steps, but it can be used with any combination of steps. It is acceptable to use the Unload Tips step multiple times throughout a method.

Unload Tips is used when tip integrity is in question or when cross-contamination between labware is a concern. Unload Tips can be used in place of the Unload (Tips) option in the Transfer and Combine steps, and is particularly useful with Loop and Worklist steps; for example, when a Worklist step performs a transfer accessing the same tip box repeatedly, the tips can be loaded and unloaded as many times as the tip configuration allows with the New Tips and Unload Tips steps.

Unloading tips is recommended when:

- The current tips have manipulated a caustic fluid.
- The current tips have been loaded the maximum number of times allowed in their Labware Properties.

**Note:** To increase the number of uses allowed for each set of tips, increase the Load no more than count in the Labware Properties configuration in the Instrument Setup step.

- The integrity of the current tips is in question.
- To avoid cross-contamination of samples.



Insert an **Unload Tips** step into the Method View (Figure 16-11).

Biomek® Software - Aspirate1 [Development]  File Edit Project Instrument Execution Options Hel	þ.	. <u> </u>
Start	Eod Pod1	
Aspirate 🐔 Instrument Setup	<u>↑</u>	
🚓 🕕 New Tips		
Dispense Aspirate from Resr	Unload Tips Configuration	
Dispense to Dest1		
Mix Unload Tips		
Wash Finish		
New Tips		
9999		
VVV Unload Tips		
0		
Loop		
Cleanup		
Move Pod		
Group		
	P12      P16        Seen      Dest1	
	P2 P6 P10 P14 P18	
	P3 P7 P11 P15 P19	
		=
Aspirate1 BiomekFX BiomekFX ETC: 0:00:31		

Figure 16-11. Unload Tips step and configuration

To configure Unload Tips:

- 1. If using a dual-pod *FX* instrument, specify the Multichannel **Pod** requiring tip removal from the drop-down list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or Current Deck Display.
- 2. Select a step that occurs after the Unload Tips step or the **Finish** step to validate the step configuration.

## 16.9 Loop Step

The Loop step executes any steps located between the Loop and End Loop icons as many times as specified in the Loop Step Configuration; for example, if a method requires multiple repetitions of the same aspirate and dispense operations, the Loop step allows the repetition of those steps as many times as required. The steps requiring repetition, in this case the Aspirate and Dispense steps, are 'substeps' of the Loop step.



When the Loop step is added to a method, the Loop icon appears in the Method View, as well as the End Loop icon. All steps placed between the Loop and End Loop icons are contained within the Loop and are repeated as many times as indicated in the Loop Step Configuration (Figure 16-12).

**Note:** While the Loop step is executing, a number appears in parentheses behind each substep within the Loop step, indicating the current iteration of the loop.

**Note:** Variables defined in a Loop step apply only to the Loop and its substeps.



Insert a **Loop** step into the Method View (Figure 16-12).



Figure 16-12. Loop step and configuration

# 16.9.1 Configuring a Loop Step and Creating a New Variable

To configure a Loop step and create a new variable:

1. In Variable, insert a variable for the Loop step and any substeps (Figure 16-13). Variable creates a variable that can be used in the Step Configurations for any substep of the Loop (Figure 16-14). A variable name cannot contain a space or a punctuation mark, such as a period (.), a comma (,), or an ampersand (&).

**Note:** An entry in Variable is optional. The Loop step can function without a Variable.



Figure 16-13. Loop step — variable created in Variable

2. Insert the **Start** value for the Loop count (Figure 16-13).

**Note:** A Loop usually begins counting at '1'; for example, if the Increment is '25', Start is '1', and End is '100', four passes of the loop are completed for the Increment values of 1, 26, 51, and 76. It is occasionally desirable, however, to count the passes of a loop by the Increment value; for example, if the Start value is changed from '1' to '25', four passes of the loop are completed for the Increment values of 25, 50, 75, and 100.

3. Insert the **End** value for the Loop count.

**Note:** Each time the method completes a loop, its count is advanced. The method drops out of the Loop when the next pass would exceed the End number; for example, if the Increment is '25', Start is '1', and End is '100', four passes of the loop are completed for the Increment values of 1, 26, 51, and 76. The Loop stops at 76 because the value of the next pass would be 101, which exceeds the End value of 100.

4. Insert an **Increment** value for the Loop count. The Increment advances the iterations of the Loop from the Start value to the End value. Variables can be used in Increment; however, the variable must be preceded by an equal (=) sign (Figure 16-14).

**Note:** When the variable is accessed in the substeps, the value of the variable is incremented until the value reaches the End value, without going over. For example, if the Increment for Count is 25, and if Start is '1', End is '100', four iterations of the loop are completed for the values 1, 26, 51, and 76. The variable 'Count' is used in the volume fields of the Aspirate and Dispense substeps of the Loop step (Figure 13-22).

5. Select **End Loop** in the Method View to add steps to the Loop.

- 6. Insert the desired substeps between Loop and End Loop.
- 7. Configure the substeps, using the variable created in Variable where applicable .

🌵 Biomek	® Software - Aspirate1* [Development]		_ 🗆 🗙
File Edit	Project Instrument Execution Options Hel		
Aspirate		Labware Type: Reservoir Pod1	
New Tips New Tips Linload Tips Cleanup Cleanup Move Pod	Finish	Liquid Type: Vater The variable 'Count' used in Volume: -Count used in the Aspirate step within a Loop step. Iechnique: Convol Reservoir	
	<u>× &gt;</u>	Ess      Eth      P12      P16        Vvic      Dest1      Dest2      P17        P2      P6      P10      P14      P18        P3      P7      P11      P15      P19	11
Aspirate1*	BiomekFX BiomekFX ETC: 0:00:22		



**Note:** In the example, the variable 'Count' is used in the volume fields of the Aspirate and Dispense substeps of the Loop step (Figure 16-14).

# 16.9.2 Configuring a Loop Step Using an Existing Variable

To configure a Loop step without creating a new variable, but using an existing variable:

1. Leave Variable blank (Figure 16-15).

**Note:** An entry in Variable is optional. The Loop step can function without an entry in Variable.

M Biomek® Software - Aspirate1* [Development] File Edit Project Instrument Execution Options Help	
Start  Set in	Variable Variable is left blank.
Aspirate	
Dispense Dispense Dispense	'AspValue 'AspValue 'AspValue 'AspValue 'AspValue' is being
Aspirate from Resr	used as an Increment
Mix Dispense to Dest1	for the Loop step.
Finish	
New Tips	
4444	
Loop	
Cleanup	
Move Pod	
Group	
	E** P12 P16
	Swap Dest1 Dest2 P17
	P3 P7 P11 P15 P19
Achievent Biomekey Biomekey ETC: 0:00:22	14 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1

Figure 16-15. Loop step — using a previously created variable as the Increment

2. Insert the **Start** value for the Loop count.

**Note:** A Loop usually begins counting at '1'; for example, if **Increment** is '=AspValue2' and AspValue2 has an assigned value of 25, and if Start is '1' and End is '100', four iterations of the loop are completed for the values 1, 26, 51, and 76. It is occasionally desirable, however, to count the passes of a loop by the **Increment** value; for example, if the Start value is changed from '1' to '25', four passes of the loop are completed for the **Increment** values of 25, 50, 75, and 100. 3. Insert the **End** value for the Loop count.

**Note:** Each time the method completes a loop, its count is advanced. The method drops out of the Loop when the next pass would exceed the End number; for example, if Increment is '=AspValue2' and AspValue2 has an assigned value of 25, and if Start is '1' and End is '100', four iterations of the loop are completed for the values 1, 26, 51, and 76. The Loop stops at 76 because the value of the next pass would be 101, which exceeds the End value of 100.

4. Insert a previously created variable in **Increment** (Figure 16-15). The value of the variable is used as the **Increment** to advance the iterations of the Loop from the Start value to the End value. Variables must be preceded by an equal (=) sign in the Loop step configuration, as well as in the substep configurations.

**Note:** When the variable is accessed in the substeps, the value of the variable is incremented until the value reaches the End value, without going over. For example, if Increment is '=AspValue2' and AspValue2 has an assigned value of 25, and if Start is '1' and End is '100', four iterations of the loop are completed for the values 1, 26, 51, and 76.

- 5. Select **End Loop** in the Method View to add steps to the Loop.
- 6. Insert the desired substeps between Loop and End Loop.

7. Configure the substeps using the previously created variable where applicable.

iomek® Software - Aspirate1\* [Development] Edit Project Instrument Execution Options Help \_ 🗆 🗙 🏟 Bio Start Þ 🐔 Instrument Setup Aspirat 🕕 New Tips Ô 🕗 Loop from 25 to 100 step =AspValue Dispense n Aspirate from Resr Þ Dispense to Dest1 Mix 3.70 mm from bottom [Overrides Technique] End Loop Pod: Pod1 Labware Type: Reservoir • Unload Tips Wash Resr • 🔲 Refresh Tips Position: The variable 'AspValue' Water • Finish Liquid Type: is used in the volume =AspValue ⊻olume μL 🖌 New Tips field of the substeps VVVV Auto-Select Cu Iechnique: Low-vol Reservoir Unload Tip: 0 Cleanup <u>m</u> Nove Pod ۲ Group ×111 Resr Eth P12 P16 P17 P6 P14 P19 P11 P15 Aspirate1\* BiomekFX BiomekFX ETC: 0:00:22

**Note:** The variable 'AspValue' is used in the volume fields of the Aspirate and Dispense substeps of the Loop step (Figure 16-16).

Figure 16-16. Loop step — the 'AspValue' variable is used as the Volume for the substeps

## 16.10 Cleanup Step

The Cleanup step is used to direct the instrument to dispose of tips and tip boxes. Normally, when a method is completed, tips and tip boxes are left in their final positions, regardless of their originating positions. The Cleanup step can return tips and tip boxes to their original deck position or hardware device; place them in a trash receptacle; or leave them wherever they are when the method is completed.

The Cleanup Step Configuration includes:

- With no configuration options selected, all tips are sent to their configured location at the end of the method.
- When Send tip boxes to their configured final destinations is checked, all the tip boxes that came from off deck and all those that were configured to go to trash are sent to those final destinations.
- When Send boxes that started on deck back to their original locations is checked, in addition to the operations resulting from the box checked above, any tip boxes that started on the deck are returned to their starting position. (This option is activated only when Send tip boxes to their configured final destinations is selected.)

Cleanup resets the hardware by physically returning tips and tip boxes to their original positions, while the Finish step only unloads tips and resets the software to its original state at the completion of the method. Use the Cleanup step with the Finish step to reset both the hardware and the software to the state they were in at the onset of the method, or to clear the deck of used tips and tip boxes. The Cleanup step is useful when the same method is run repeatedly, one time after another, and when trying to track which tips and tip boxes were used for which fluids and labware.

Cleanup

Insert a **Cleanup** step into the Method View (Figure 16-17).



Figure 16-17. Cleanup step and configuration

To configure the Cleanup step:

1. Select **Send tip boxes to their configured final destination**, if desired. This sends the tip boxes to the final destination configured in the Labware Properties for that tip box.

**Note:** The default setting for the Cleanup step is both options Unchecked. Inserting the Cleanup step without selecting either option instructs the Biomek instrument to unload any tips currently on the pod to the final destination configured in Labware Properties for those tips. 2. Select **Send tip boxes that started on deck back to their original locations**, if desired. This returns the tip boxes to their starting location, regardless of whether they originated on the deck or in an external hardware device, such as a stacker carousel.

**Note:** Send tip boxes that started on deck back to their original locations is only available when Send tip boxes to their configured final destination is selected.

3. Select a step that occurs after the Cleanup step or the **Finish** step to validate the step configuration.

## 16.11 Move Pod Step

The Move Pod step moves the pod to a deck position that does not hamper access to the labware, ALPs, and devices on the deck.

Use Move Pod to ensure that the pod does not interfere with activities taking place on the deck, such as the addition, removal, or relocation of labware or devices.



Insert a **Move Pod** step into the Method View (Figure 16-18).



Figure 16-18. Move Pod step and configuration

To configure the Move Pod step:

1. Specify the **Pod** being moved from the drop-down list.

OR

Select the **Pod** from the Current Deck Display. The tip diagram on the left of the Current Deck Display represents **Pod1**, while the tip diagram on the right represents **Pod2**. Selecting a pod from the Current Deck Display automatically updates **Pod** in the Step Configuration.

**Note:** In Pod, Pod1 is the default for a one-pod Biomek FX or Biomek 3000 system. In a two-pod Biomek FX system, the pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.

2. In Location, select the deck position from the drop-down list.

OR

Select the **Location** from the Current Deck Display. Selecting a location from the Current Deck Display automatically updates **Location** in the Step Configuration.

**Note:** A bright blue outline appears in the Current Deck Display around the selected location.

- 3. Insert the X Offset value in centimeters (cm).
- 4. Insert the **Y** Offset value in centimeters (cm).

**Note:** Selecting the deck position without offset values is usually sufficient to relocate the pod; however, X and Y Offset values can be used to provide a more specific position.

The final position of the pod when using X and Y Offsets is the deck position specified in Location *PLUS* the values inserted into the X and Y Offsets. For example, when framing an ALP, the labware positioned on the ALP is 'offset' from the zero (0) point, or back-left corner, of the ALP to the actual edge of the labware itself.

5. Select a step that occurs after the Move Pod step or the **Finish** step to validate the step configuration.

## 16.12 Group Step

The Group step gathers steps performing similar functions or steps achieving a specific result together. Group reduces the confusion that can arise when viewing large methods by consolidating and applying a single label to a number of steps. Multiple Group steps can be used within a method, and they can be placed in a Group step together to further abbreviate the steps displayed in the Method View.

Advantages to grouping steps are:

- The steps contained in the Group step are hidden from the Method View by collapsing the Group step.
- The Group step is collapsed by default when a method is opened. This means long methods appear shorter and more concise.
- Multiple steps can be identified by a single label.
- All steps related to a specific operation can be grouped together under a single label to identify the operation they perform.

Insert a **Group step** into the Method View (Figure 16-19).



Figure 16-19. Group step and configuration

When a Group step is added to a method, the Group Step Configuration appears (Figure 16-19).

To configure the Group step:



- 1. Double-click on the Group step to expand it so that the End icon is visible.
- 2. In Group Label, enter text to identify the group.

**Note:** The text inserted in Group Label appears in the Group steps caption in the Method View.

- 3. Insert steps between the Group and End icons.
- 4. Double-click on the Group step to collapse it so that the steps contained within the Group step are hidden from view.

Note: The Group step is collapsed by default when the method opens.

# Using the Advanced Step Palette

#### 17.1 **Overview**

The steps presented in the Advanced Step Palette (Figure 17-1) provide tools for more complex management of methods. These steps are used to perform the following:

- Repeat a series of steps using a list of specified values without entering and ٠ configuring the steps repeatedly
- Access and run a method within a method
- Conditionally execute steps based on an aspect of the environment, such as • the volume in a microplate

**Note:** The steps available in the Advanced Step Palette do not use the pod directly and may be used with any single- or dual-pod instrument.

The steps available in the Advanced Step Palette are:

• Run Method — accesses and runs a method within a method.



Worklist — fills in variables based on a worklist file. •





Just in Time — synchronizes the execution of steps.



ΣÈ Let.

•

Let — defines variables for its substeps.

•



If — evaluates a condition within a method and runs either the "then" substeps or the "else" substeps according to the condition.



• Script — runs a list of commands providing customized control over the instrument.



• Scripted Let —similar to Script step, with the exception that it allows variables to be extended outside the script and used in the method.

When one of the above steps is inserted into a method, the configuration options associated with that step appear on the right side of the Biomek Software main editor, in Step Configuration.

**Note:** Steps not configured appropriately may generate errors when the method is validated or run.

# **17.2 Displaying the Advanced Step Palette**

In order to add advanced steps to a method, display the Advanced Step Palette (Figure 17-1).

🌵 Biomek	® Software	- U ×
File Edit	Project Instrument Execution Options Help	
🗋 🗅 💕 🖞		
Run Method Worklist Just In Time Let If	Method View      The method is built      step by step in this      area.      When a step is highlighted in the Method View,      set the corresponding configuration in this area.	
	TL1 P4 P8 P12 P16	
Scripted Let	Advanced Step Palette P13 P17	
	Displays the steps available for insertion into a method.	
	Additional step palettes include Basic, Intermediate,	
	Span-8, Stacker Carousel, and Devices.	
1110	BiomekFX BiomekFX	

Figure 17-1. Biomek Software main editor with Advanced Step Palette displayed

To display the Advanced Step Palette:

- Right-click any empty palette space, and select Advanced from the menu.
  OR
- From the toolbar, select **Options>Toolbars>Advanced**.

## 17.3 Run Method Step

The Run Method step executes a previously created method within the method in progress. Run Method saves time by assigning variables accessible by the previously created method and by preventing unnecessary duplication of steps. If the same series of steps is used in a number of methods, a method can be created containing only those steps and accessed using a Run Method step. Multiple Run Method steps can be used within a method. Methods referenced by a Run Method step must be stored in the active project file.

Although it is not visible in the Method View, the Run Method step 'inserts' all of the steps from the referenced method into the current method at the point the Run Method step appears. All of the steps contained in the submethod are completed as a part of the current method; they are not completed as a separate method.

**Note:** A 'submethod' is a method run within another method. It is recommended that the submethod use the same **Instrument Setup** step configuration as the method in progress. If the submethod and the method in progress do not use the same **Instrument Setup**, an **Instrument Setup** step must be added to the method in progress to ensure that all labware for both methods is visible to the instrument.

The advantages of using Run Method are:

- The ability to run a method within a method
- Allows repetition of steps configured in a previously created method without recreating those steps
- Variables assigned in the Run Method step are accessible by the submethod
- Controls the size of the current method and reduces confusion by listing only the Run Method step in the Method View, not all of the steps accessed by the Run Method step

Insert Run Method into the Method View (Figure 17-2).



👫 Biomek® Software - Method2 [Development]				
File Edit Project Instrument Execution Options Help				
] 🗅 🚅 🕯	1 8 8 6 X 9 8 4 ×			
	Start	File Name: Advanced	•	
🔊 Run Method	🚮 Run Advanced	Overridable Prompt Variable Name Value		
	Finish			
63		Run Method Step Configuration		
Just In Time	Method View	previously created method and then runs it as a submethod within	L	
ΣC	The method displayed	the current method. Variables configured in the Run Method		
Let	sten Note that the Run	Step Configuration are accessible by the submethod.		
If	Method step shows the name of the submethod.			
Script				
		TL1 P4 P8 P12 P16		
Scripted Let		P1      P5      P9      P13      P17        P2      P6      P10      P14      P18        P3      P7      P11      P15      P19		
Method2 Bio	omekFX BiomekFX ETC: 0:00:02			

Figure 17-2. Run Method step and configuration

## 17.3.1 Configuring the Run Method Step

To configure the Run Method step:

1. In File Name, select a submethod to run within the method in progress. The Run Method step shows the name of the submethod.

**Note:** Only submethods stored in the same project file as the method in progress are available. To access a submethod stored in another project file, import the desired submethod into the project (refer to Section 6.10, *Importing and Exporting Project Files*).

2. Configure any variables as needed (refer to Section 13.1, *Overview*).

## 17.4 Worklist Step

The Worklist step uses a text file to supply multiple values for one or more variables (Figure 17-3). Worklist is useful when repetition of the same step(s) is required, but a variable, or variables, needs to change each time the step(s) is executed. When a group of steps using the variable(s) defined in a text file are placed inside a Worklist step, Worklist automatically performs each step once for each line in the text file. The steps using the variable(s) defined in the text file are substeps within the Worklist step in the Method View.

		Values Used During the Worklist Step		
<b>Loop Variable</b> The variable assigned in the Worklist configuration that holds	Value assigned during each pass of the Worklist for each of the steps within the Worklist step.		pass of e steps	
the current iteration of the Worklist 'loop'.		index	Location	Amount
Rows of Data		1	P5	25
Each row of data represents a pass		2	P6	50
for the step(s) nested within the Worklist step.		3	P7	100

Figure 17-3. Worklist example

**Note:** Values read from the text file must be formatted correctly in order for Biomek Software to create the worklist with the desired row and column formatting. Each row in the text file represents a row in the worklist. In each row, columns are designated by separating individual values with commas.

Worklist text files may be created in a database, spreadsheet, or text editor. When creating a worklist in a database or spreadsheet, save the file in comma-separated values (\*.csv) format. When using a text editor or word processor, manually separate values on a single line with commas and save the file in text (\*.txt) format.

**Note:** Quotation marks may be used in the text file, but they must come in pairs. The Worklist step looks at the first set of quotation marks and interprets that everything after them is included as a row of data up to the second set of quotation marks. If quotation marks are desired in a row of data, use another set of quotation marks around the data within the quotations marks. For example, type ""this is a comment" in the text file and "this is a comment" will appear in the row of data in the Worklist step.

The Worklist step operates like a combined Let/Loop step because it 'lets' a variable in the Worklist text file use different values for each pass of the Worklist 'loop'. Worklist 'loops' through the substeps until each step has used each line of the Worklist text file specified.



When Worklist is added to a method, the Worklist icon and an End Worklist icon appear in the Method View. Any step(s) inserted between the Worklist and End Worklist icons use the variables contained in the Worklist text file configured in the Worklist step.
Advantages of using a Worklist include:

- Data in a text file is accessible by any Biomek Software method.
- Selecting the text file in the Worklist Step Configuration copies all of the variable data from the text file into the Worklist Step Configuration without individually entering all of the variables and their values in each of the steps performed inside the Worklist step (Figure 17-4).
- A dynamic link exists between the method and the text file. The Worklist step accesses the text file each time the method is run; therefore, any updates made to the text file are included in the next method run.

**Note:** The text file configured in the Worklist step must maintain its original file path, or the method will not be able to find the text file. If the text file path changes, update the Worklist Step Configuration to reflect the new file path.



Figure 17-4. Worklist step and configuration



Insert **Worklist** into the Method View (Figure 17-4).

#### 17.4.1 Configuring the Worklist Step

To configure Worklist, complete the following:

1. Click the Browse... button following File Name. Open appears (Figure 17-5).

Open				?	×
Look jn:	🔄 Biomek Software 💽		Ť		
worklist.cs	v)				
File <u>n</u> ame:	worklist.csv			<u>O</u> pen	
Files of type:	Comma Delimmited Text File (*.txt; *.csv)	<b>•</b>		Cancel	

Figure 17-5. Worklist — text file selection

2. Browse to and select the text file accessed during the method run.

**Note:** The file path and the file name of the text file are included in the Worklist step in the Method View.

- 3. Click **Open** to add the file to the Worklist configuration and return to the main Biomek Software window.
- 4. Enter a name for the **Loop Variable**. The loop variable holds the current iteration number of the worklist.
- 5. Select **Loop entire worklist** to use all rows of the worklist.

OR

Select **Loop from line** and enter the start and end row numbers to use a specific range of rows from the worklist.

## 17.5 Just in Time Step

Just In Time synchronizes the execution of the substeps within the Just In Time step (Figure 17-6).

The steps within the Just In Time step are queued in the order in which they appear in the Method View, but it is possible to execute two or more steps simultaneously. In Figure 17-6, for example, the first two substeps are completed, then deck position P8 is paused. While the pause on P8 is in process, the next four steps are completed because they do not require access to deck position P8. Finally, the last two steps are completed once the pause is finished because they require access to deck position P8.



**Note:** An incubation period is accomplished using the Pause step (refer to Section 15.5, *Pause Step*).



Figure 17-6. Just In Time step

Since Just In Time is a part of the Transfer, Combine, and Move Labware steps, it is not necessary to use these steps in conjunction with the Just In Time step.

Use Just In Time when an Aspirate operation and its subsequent Dispense operation must occur immediately following an incubation period (Figure 17-6) or when a method requires that a preparatory operation should not occur too early with respect to another related operation.

**Note:** Any steps placed between the Just In Time and End Just In Time icons are configured in their individual Step Configurations, but they are completed when appropriate within the Just In Time step.



Insert **Just In Time** into the Method View. The **Just In Time** icon appears in the method, as well as an End Just In Time icon (Figure 17-6).

To configure Just In Time, enter the steps requiring synchronized timing between the Just In Time and End Just In Time icons.

**Note:** The Just In Time step does not affect the Estimated Time to Completion for the method when its substeps are already synchronized steps, such as the Transfer, Combine, or Move Labware steps.

## 17.6 Let Step

The Let step is used to create variables accessible only to the sub-steps of the Let step (Figure 17-7). The variables can be used by multiple steps within the Let step, and they allow configuration of the steps by specifying explicit values for things like volumes and heights. For example, instead of typing the number of microliters into the volume field of an Aspirate step, the name of a variable is inserted. Every time the Aspirate step is executed, the current value of that variable is used as the volume.



Figure 17-7. Let step and configuration



When the Let step is added to a method, the Let and End Let icons appear (Figure 17-7). Any steps placed between the Let and End Let icons are substeps controlled by the Let step, but they are configured through their individual Step Configurations. Substeps are enqueued for execution in the order of appearance in the Method View; however, it is possible to execute two or more steps simultaneously.

Any of the substeps within the Let step can use variables created in the Let Step Configuration. Only the Name is displayed in the step configuration, not the Value associated with that Name. The Biomek Software automatically uses the associated Value when the step is executed.

## 17.6.1 Configuring the Let Step

Configuring the Let step involves creating the variables used by the substeps within the Let step (Figure 17-7).



Insert Let into the Method View (Figure 17-7).

To configure the Let step:

- 1. Insert the variables desired (refer to Section 13.1, *Overview*).
- 2. Enter desired steps between Let and End Let icons.
- 3. Configure the substeps between Let and End Let.

# 17.7 If Step

The lf step controls which steps in a method are executed based on a condition, such as the volume of liquid in a piece of labware. When lf is run, Biomek Software tests the lf condition, then processes the appropriate block of substeps based on the results of the test.

The substeps of If are:

- Then substeps following Then are processed if the condition is true.
- Else substeps following Else are processed if the condition is false.
- End the End substep terminates each If, Then, and Else block of steps (Figure 17-8).



To insert an lf step into a method, insert **lf** into the Method View. The lf Step Configuration appears (Figure 17-8).



Figure 17-8. If step and configuration

#### 17.7.1 Configuring the If Step

To configure the lf step:

1. Enter the **Condition** in the configuration.

**Note:** A Condition must be a valid VBScript expression and can refer to variables defined in other steps.

- 2. Highlight **End** under the Then substep.
- 3. Add and configure the steps performed if the condition is true.
- 4. Highlight **End** under the Else substep.
- 5. Add and configure the steps to perform if the condition is false.

**Note:** Any steps placed between the Then and End or Else and End icons are configured in their individual Step Configurations using the same procedures as though they were a main step in the method.

## 17.8 Script Step

**Note:** The information in this section is intended only to assist in gaining an understanding of what the scripting function does. It is not intended to be a tutorial on scripting. The **Script** step is provided only for the use of methods developed within Beckman Coulter at this time. This function is not currently supported for end-user applications.

More information is available at *Masching* and in *VBScript in a Nutshell*, 2nd Edition, Matt *ds*, Paul Lomax, and Ron Petrusha 2003, O'Reilly & Associates, Inc.

Note: Refer to Chapter 28, *Scripting*, for other information on scripting.

While the editors and templates in Biomek Software provide the means to perform most operations, it is possible that some highly specialized liquid-handling applications are beyond this scope. Although not an activity typically required by users, Biomek Software uses the Script step for writing script to accommodate these advanced needs.

A Biomek Software script consists of lines of code that provide instructions for instrument operations. The lines of code, or commands, may be typed into the Script Step Configuration. Running the method invokes a Visual Basic® script interpreter that evaluates each line of code and performs the requested operation.

Dragging a Script step from the Method View to the Step Palette makes it available as a step to any method (refer to Chapter 29, <u>Changing Window Appearance</u>). To make the values defined in a Let step available to the Script step, place the Script step within the Let step.

The Script step provides a configuration area in which a description of the script step may be entered and lines of code may be written (Figure 17-9).



Insert Script into the Method View (Figure 17-9).

Note: Place the Script step within a Let block of steps (refer to Section 17.6, Let <u>Step</u>) to use the variables defined in the Let step in the Script.



Figure 17-9. Script step and configuration

#### 17.8.1 Configuring the Script Step

To configure the Script step:

1. In **Description**, enter a description for the operation or purpose of the Script Step.

**Note:** The Description replaces the word "Script" in the Script step caption in the Method View.

2. Position the cursor in the Script Step Configuration.

**Note:** The lines of code can be entered in a word processor and then copied and pasted into the Script Step Configuration using standard Windows<sup>™</sup> procedures. Use standard Windows<sup>™</sup> techniques for editing scripts.

3. Write and edit the script.

Note: The Biomek Software scripting feature is based on Visual Basic Scripting, a subset of the Microsoft® Visual Basic programming language. Any code interpreted by Visual Basic Scripting can be interpreted by the instrument and may be included in a method. More Visual Basic Scripting information is available at //msdn.microsoft.com/scripting and in the book, *VBScript in a Nutshell*, 2nd ion, Matt Childs, Paul Lomax, and Ron Petrusha 2003, O'Reilly & Associates, Inc. For more specific Biomek Software scripting help, contact Beckman Coulter Technical Support.

## 17.9 Scripted Let Step

**Note:** The information in this section is intended only to assist in gaining an understanding of what the scripting function does. It is not intended to be a tutorial on scripting. The **Scripted Let** step is provided only for the use of methods developed within Beckman Coulter at this time. This function is not currently supported for end-user applications.

More information is available at *Markovication and the state of the s* 

Note: Refer to Chapter 28, Scripting, for other information on scripting.

The Scripted Let step is similar to the Script step (refer to Section 17.8, *Script Step*), except it allows for variables created within a Script to be extended outside the Script and used in the method like variables created in a Let step (refer to Section 17.6, *Let Step*). Two additional commands are available to set variables — Extend and WeakExtend.

- Extend sets the variable name to the specified value for all steps contained within the Scripted Let. The proper syntax for Extend is Extend "Variable Name", Value.
- WeakExtend sets the variable name to the specified value for all steps contained within the Scripted Let step if the specified variable does not already have a value. This is similar to a Let variable with Overridable selected. The proper syntax for WeakExtend is WeakExtend "Variable Name", Value.

The Scripted Let step provides a configuration area in which a description of the script step may be entered and lines of code may be written.



Insert Scripted Let into the Method View (Figure 17-10).

**Note:** Place the Scripted Let step within a Let block of steps (refer to Section 17.6, *Let Step*) to use the variables defined in the Let step in the Script.



Figure 17-10. Scripted Let step and configuration

#### **17.9.1** Configuring the Scripted Let Step

To configure the Scripted Let step:

1. In **Description**, enter a description for the operation or purpose of the Scripted Let Step.

**Note:** The Description replaces the words "Scripted Let" in the Scripted Let step caption in the Method View.

2. Position the cursor in the Scripted Let Step Configuration.

**Note:** The lines of code can be entered in a word processor and then copied and pasted into the **Scripted Let** Step Configuration using standard Windows® procedures. Use standard Windows techniques for editing scripts.

3. Write and edit the script.

**Note:** The Biomek Software scripting feature is based on Visual Basic Scripting, a subset of the Microsoft® Visual Basic programming language. Any code interpreted by Visual Basic Scripting can be interpreted by the instrument and may be included in a method. More Visual Basic Scripting information is available at //msdn.microsoft.com/scripting and in the book, *VBScript in a Nutshell*, 2n dition, Matt Childs, Paul Lomax, and Ron Petrusha 2003, O'Reilly & Associates, Inc. For more specific Biomek Software scripting help, contact Beckman Coulter Technical Support.

 Insert any substeps between Scripted Let and End Scripted Let that utilize any variables defined in the Scripted Let.

**Note:** Variables extended outside of a script in the Scripted Let step are only available to the substeps within the Scripted Let.

# Using the Span-8 Step Palette

## 18.1 Overview

- FX, NX-S8 Only the Span-8 Pod may be used with steps on the Span-8 step palette.
- ➤ 3000 The Serial Dilution, Transfer From File, and Define Pattern steps can all be used with pipetting tools on the Biomek 3000.

Steps on the Span-8 Step Palette provide individual control over liquid-handling functions, including tip handling and pipetting operations, using the Span-8 Pod outside the Transfer and Combine steps. Additional steps, such as Serial Dilution and Transfer From File, configure larger operations involving multiple liquid-handling functions.

The **Transfer** and **Combine** steps on the Basic Step Palette are used with the Span-8 Pod to configure larger operations involving multiple liquid-handling functions when strict control of every operation is not necessary.

**Note:** For information on using and configuring the **Transfer** and **Combine** steps using a Span-8 Pod, refer to Section 15.3, *<u>Configuring Transfer and Combine Steps</u>*.



CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height and failure to choose the correct ALP in the Deck Editor may result in collisions between pod(s) and ALPs during operation.

The steps available in the Span-8 Step Palette are:



- Serial Dilution performs a series of dilutions on a single microplate.
  - > **3000** Serial Dilution can be used in methods on the Biomek 3000.



• Span-8 Aspirate — aspirates a specified amount of liquid from a single source in preparation for the Span-8 Dispense step.



•

•

Span-8 Dispense — dispenses a specified amount of liquid into destination labware, following the Span-8 Aspirate step.

Span-8 New Tips — loads new tips to the Span-8 probes.

Span-8 Tip Discard — unloads tips from the Span-8 probes.

- Span-8 New Tips
- Span-8 Tip Discard
- Span-8 Wash Tips



Define Pattem

- Span-8 Wash Tips washes tips by flushing tips with system fluid at a Span-8 Tip Wash ALP or by aspirating and dispensing at a Wash Station 96 ALP. The Span-8 Wash Tips step is also used to purge air from system tubing and syringes during a method.
- Transfer From File performs well-to-well transfers using a commaseparated data file.
  - 3000 Transfer From File can be used in methods on the Biomek 3000.
- Define Pattern create a method-specific well pattern manually or by reading well information from a file.
  - > **3000** Define Pattern can be used in methods on the Biomek 3000.

When one of the above steps is added to a method, the configuration associated with that step appears on the right side of the Biomek Software main editor, in Step Configuration (Figure 18-1).

**Note:** Steps not configured appropriately in Step Configuration generate errors when the method is validated or run.

# 18.2 Displaying the Span-8 Step Palette

In order to add Span-8 steps to a method, display the Span-8 Step Palette (Figure 18-1).

3000 — Only the Serial Dilution, Transfer From File, and Define Pattern steps are displayed for a Biomek 3000 instrument.



Figure 18-1. Biomek main editor with Span-8 Step Palette displayed

To display the Span-8 Step Palette, complete the following:

• Right-click any empty palette space, and the Step Palette menu appears. Select **Span-8**.

OR

• From the menu bar, select **Options>Toolbars>Span-8**.

## 18.3 Serial Dilution Step (including 3000)

The Serial Dilution step is used to perform multiple dilutions of a sample on a single microplate. The Serial Dilution step transfers liquid from wells on a microplate to other wells on the same microplate and may also add diluent to those wells. The microplate may be prepared by the Serial Dilution step or in the same method prior to the Serial Dilution step.

Serial dilution completes multiple dilutions by performing the following actions in sequence:

- Transfer a volume of diluent from the diluent labware to all selected wells except the first row or column (unless specified otherwise).
- Transfer a volume of sample from the first selected well to the next selected well.
- Mix the solution.

**Note:** The Serial Dilution step does not automatically mix the solution. Select **Customize** to add mixing to the technique or create a new technique in the Technique Editor. Refer to Section 9.3, <u>*Creating New Techniques*</u>, for information on creating and modifying techniques.

- Transfer a volume of solution to the next selected well.
- Mix the solution.
- Repeat transfer and mix operations until all selected wells are used.

**Note:** The labware may be previously prepared, either by hand or by using the Biomek instrument, with a specific volume of diluent in the selected wells. If the labware is previously prepared, skip the step that adds diluent to the wells.

The Serial Dilution step specifies:

- Pod that performs the dilution.
- Probes that are used during dilution.
- Tips used during dilution.
- Labware type accessed during serial dilution.
- Position of the labware on the deck.
- Liquid type of the labware.
- Wells in the selected labware to use for dilution.
- Volume being diluted.
- Direction in which wells are accessed during dilution.
- Technique.
- Diluent labware type accessed during serial dilution.
- Position of the diluent labware on the deck.
- Liquid type of the diluent labware.
- Wells in the selected diluent labware to use for dilution.
- Dilution ratio.

Insert a **Serial Dilution** step into the method (Figure 18-2).



🌔 Biomek	® Software - Method8* [New]		
File Edit	Project Instrument Execution Op	ptions Help	
∣ດອ≀	1 🖬 🖪 🖨 🕺 🌆 🛍 🖌		
<b>Seriel</b>	Start	Pod: Pod2 Use Tips : AP96_200uL_LLS V Probes: 1 2 3 4 5 6 7 8	<u>^</u>
Span-8 Aspirate	Serial Dilution	Labware Type:	
Span-8 Dispense	Method View	Direction : Left to Right Discard extra volume from last wells	]
Span-8 New Tips	The method displayed contains a Serial Dilution step.	Wash tips in Water : 3 cycles of 110% μL	
Span-8 Tip Discard		Image of the second	-
Wash Tips		TL1 P4 Divert P12 P16 Sorial Dilution S	ton
From File		P1 P5 Source P17 P17 Configuration P2 P6 P10 P14 P18 P3 P7 P11 P15 P19	ieb I
Method8*	BiomekFX BiomekFX ETC: 0:00:03	3 J	

Figure 18-2. Serial Dilution step and configuration

#### **18.3.1** Configuring the Serial Dilution Step

When a Serial Dilution step is added to a method, the Serial Dilution Step Configuration appears (Figure 18-2).

**Note:** If it is desired to have diluent in any unused wells as a standard, tips can be conserved by using a **Transfer** step to transfer diluent to the unused wells and selecting **Leave them on** in the **Tip Handling** configuration.

To configure Serial Dilution:

1. If using a dual-pod *FX* instrument, specify the **Pod** performing the dilution operation by selecting a Span-8 Pod from the list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.

OR

If using a **3000**, insert and configure a Load Tool or Change Tool step to load an MP20 or MP200 tool in the method before the Serial Dilution.

**Note:** Refer to Section 20.3, <u>*Change Tool Step*</u> or Section 20.4, <u>*Load Tool Step*</u> for information about configuring the desired step.

2. If using a Span-8 pod on the **FX** or **NX-S8**, select the probes used in the dilution by highlighting the desired numbers next to **Probes**. Any combination of probes may be used as long as all selected probes are using the same type of tips and syringes.

**Note:** All disposable tips are selected by default. If there are only fixed tips, all tips are selected by default. For example, if the Span-8 Pod is equipped with fixed tips on probes 1-4 and disposable tips on probes 5-8, probes 5-8 will be selected by default. However, if probes 5-8 are also equipped with fixed tips, all eight probes are selected by default.

3. In **Use Tips**, select the tip type loaded from the drop-down list.

OR

Select the tip box from the Current Deck Display with the desired tip type. Selecting tips from the Current Deck Display automatically updates Tips in the Step Configuration.

FX, NX-S8 — If the selected Probes are equipped with fixed tips, Use Tips is set to Fixed and is not configurable.

**Note:** If tip boxes were assigned names in the **Instrument Setup** step, enter that name in **Use Tips**. If tips were not assigned names, select the tip type. It is recommended that tips not be named in the **Instrument Setup** step because this could restrict the step from locating available tips.

4. Click on the desired piece of labware in the Current Deck Display in which the serial dilution is performed. Information for that piece of labware is entered automatically into the Serial Dilution Step Configuration.

OR

Select a **Labware Type** from the list (Figure 18-3).

**Note:** The labware contains the sample that is diluted in the first row or column of selected wells, depending on the direction of dilution selected.

🌵 Biomek® Software - Method8* [New]		
File Edit Project Instrument Execution O	ptions Help	
Start	Pod: Pod2 Use Tips	: AP96_200uL_LLS 💌
serial 😤 Instrument Setup	Probes: 1 2 3 4 5 6 7 8	
Pilwinn		Labware Tupe: BCElat96
	<b></b>	
span-8 Aspirate		
¥		Liquid Type: Water
Span-8		Volume: 100 µL
Dispense T T	<u> </u>	Direction : Left to Right
99	<u> </u>	I Discard extra volume from last wells
Span-8 New Tips	0000000000000	
R	🔽 Wash tips in Water 💽 : 3 👘 cycle	s of 110% µL
Span-8 Tip	🗌 Change tips between transfers	
Discard 🕅 🕅	Auto-Select	Customize Save As
പ്പ്	Technique: Span.8	
Span-8 Wash Tips		Ľ
		DIE
Transfer		
From File	P1 P5 P13	
	P2 P6 P10 P14	P18
Define Pattern	P3 P7 P11 P15	P19
Method8* BiomekFX BiomekFX ETC: 0:00:0:	3	

Figure 18-3. Serial Dilution step configuration with labware configured

5. Verify the **Position** of the labware.

**Note:** A green outline appears in the Current Deck Display around the selected labware.

- 6. Select the Liquid Type being diluted in the selected labware.
- 7. Select the wells on the labware to use in the serial dilution. Select the wells according to the procedures described in Chapter 11, <u>Creating Well Patterns</u>.

**Note:** To zoom in on the labware, double-click on the labware. To call up a menu for Zoom In and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows the selection of the first well in a microplate

as a target for the dispense operation. Specify Selection as Text may also be used to enter variables or expressions. Wells are numbered left to right and down (for example, the first row is wells 1-12, the second row wells 13-24, etc., for a 96-well microplate).

 Enter the Volume diluted in each well of the selected labware. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue. Otherwise, just enter the desired value.

**Note:** This is the volume of each diluted solution that is transferred to the next well for the next dilution.

- Select the **Direction** to dilute. The specified Volume is transferred from each well to the next selected well in the chosen **Direction** for the next dilution. Available options are:
  - Left to right dilutes across rows of microplate (Figure 18-4).
  - Top to bottom dilutes down columns of microplate (Figure 18-5).
  - > **3000** The direction must be Left to right.

🏥 Biomek	® Software - Method8* [New]							_ 🗆 🗙
File Edit	Project Instrument Execution Op	tions Help						
0000	1 🖬 🖪 🥔 🕹 🖪 🖻 🖷 🖌	n ca 🖉		× EI				
$\overline{\cap}$	Start	<u>P</u> od:	Pod2	-	Use Ti	ps : AP96_200uL_L	LS 🔽	1
Serial Dilution	🐔 Instrument Setup	Probes:	1234	5678				
1	Serial dilution of 10					Labware Type:	BCFlat96	•
∎ Span-8	💈 Finish	66			00	Position:	P8	•
Aspirate	-	00		000	00	Liquid Type:	Water	_
_ĭ.		<b>ÓÓ</b>	<b>DOOO</b>	000	00	<u>V</u> olume:	100	μL
Span-8 Dispense		66		000	00	Direction :	Left to Right	•
J.L		$\bigcap$	$\sim$	$\sim \sim \sim$	$\mathbf{\mathbf{v}}$	🦕 🔽 Discard extr	a vol <b>a</b> ne from last w	ells
Span-8					$\dot{\mathbf{OO}}$	)		
		🗖 Wash t	ips in Water		3 <del>- </del> T		le is transfer	red in
Spap-8 Tip	Leftmost selected	well <sup>pe</sup>	e tips between tran	sfers		lirection of a	rrows as ind	icated in
Discard	Discard Contains sample to dilute. Select Direction. Diluent is added to well						d to well	
44	Contains sample to al	т				and mixed be	efore transfer	ring to
Span-8 Wash Tips		_lechnique:	Span-8		r	next selected	well.	
R		×		D4 Diluer		DIG	666661	
Transfer								
From File			P1	P5	P13			
$\sim$			P2	P6 P10	P14	P18		
Define Pattern	4		P3	P7 P11	P15	P19		
Method8*	BiomekFX BiomekFX ETC: 0:00:03							

Figure 18-4. Serial Dilution from left to right



Figure 18-5. Serial Dilution from top to bottom

- 10. Select **Discard extra volume from last wells** to aspirate the extra Volume from the last wells and discard it at the Span-8 Tip Wash ALP so that each well contains the same volume of solution.
- 11. If using an *FX* or *NX-S8*, choose **Wash tips in** to use the active 96-Channel Tip Wash ALP to wash the tips.
  - > **3000** Tip wash options are not available. Proceed to step 19.

**Note:** A WashStation96 ALP must be added to the deck and configured in the Deck Editor before washing can occur. Clicking on a WashStation96 ALP while the Serial Dilution step is being configured places a check in Wash tips in, and populates Wash tips in with the solution configured for the WashStation96 ALP in Labware Properties on the Instrument Setup step.

**Note:** If Wash tips in is not selected, proceed to step 15.

12. In **Wash tips in**, select a solution for the wash cycle if the field does not already contain a solution.

**Note:** The wash solution specified in the Serial Dilution step must match the solution selected in Labware Properties for the WashStation96 ALP on the Instrument Setup step configuration, or an error occurs.

13. Choose the number of cycles to aspirate and dispense during wash.

14. In %, provide the volume of wash fluid to use when washing tips as a percentage of the maximum volume of fluid contained in the tips in previous steps; for example, if the maximum volume of fluid transferred is 50  $\mu$ L, and the % is set for 110%, the **Transfer** step washes the tips with 55  $\mu$ L of solution.

**Note:** If the % sign is deleted in the % field, the field label changes to µL. Any value entered is now interpreted as a specific volume, not a percentage.

15. If using an *FX* or *NX-S8* with fixed tips installed, choose **Wash tips with** to use the passive Span-8 Tip Wash ALP to wash fixed tips.

**Note:** A Span-8 Tip Wash ALP must be added to the deck and configured in the Deck Editor before washing can occur. Only fixed tips can be washed with the Span-8 Tip Wash ALP. If using disposable tips and washing is desired, go back to 11. If Wash tips with is not selected, proceed to step 19.

- 16. In **Wash tips with**, enter the volume of system fluid to use when washing the outside of the tips.
- 17. In **after dispensing**, enter the volume of system fluid to use when washing the inside of the tips.
- 18. If using a Span-8 Pod on *FX*, select **Speed Pump** to use the speed pump when flushing the tips with system fluid, if desired.

OR

If using a Span-8 Pod on *NX-S8*, select **purge pump** to use the purge pump when flushing tips with system fluid, if desired.

**Note:** The speed pump or purge pump is an optional device that can be used to shorten the time it takes to wash tips. A speed pump or purge pump must be added as a device and configured with the Span-8 Pod in Hardware Setup to use it when washing tips.

19. Choose **Change tips between transfers** to change tips between transferring diluent and sample.

**Note:** Change tips between transfers is available only when disposable tips are installed on all probes.

20. The software automatically selects a **Technique** based on the liquid type, labware type, tip type, and volume being transferred. To override this selection, deselect **Auto-Select** and choose the desired **Technique**.

OR

Choose **Customize** and configure the technique as desired in the **Technique** Editor.

**Note:** For more information on customizing, configuring, and saving a technique, refer to Section 12.3.1, *<u>Understanding Techniques</u>*.

**Note:** It is recommended a technique include mixing for best results. Select **Customize** to add mixing to the technique or create a new technique in the **Technique Editor**. Refer to Section 9.3, <u>*Creating New Techniques*</u>, for information on creating and modifying techniques.

21. If the labware is previously prepared with diluent, select a step that occurs after the Serial Dilution step or the **Finish** step to validate the step configuration.

OR

If the labware is not previously prepared with diluent, configure **Diluent Properties**. Proceed to Section 18.3.2, *Configuring Diluent Properties*.

## 18.3.2 Configuring Diluent Properties

Diluent Properties are configured if the sample microplate is not previously prepared with diluent. The Diluent Properties configuration allows for diluent to be transferred from a piece of labware to the wells used for the dilution.

The Diluent Properties configuration can be collapsed to allow more room during configuration. When collapsed, the diluent properties configuration is displayed as the text Diluent Properties.

To collapse or expand the Diluent Properties configuration, click the arrow to the left of Diluent Properties or the textual display.

To configure Diluent Properties for the Serial Dilution step:

1. Click the arrow to the left of Diluent Properties or the textual display to expand the Diluent Properties configuration (Figure 18-6).



Figure 18-6. Serial Dilution — Diluent Properties configuration

2. Select **Add diluent before transfer** to activate the Diluent Properties configuration.

3. Click on the desired piece of labware in the Current Deck Display to select the labware with the diluent. Information for that piece of labware is entered automatically into the Diluent Properties configuration.

OR

Select a **Diluent Labware Type** from the list (Figure 18-7).

**Note:** Reservoirs are the only labware type valid for diluent retrieval.

👫 Biomek® Software - Method8* [New]		- D X
File Edit Project Instrument Execution C	Options Help	
Start	Add diluent before transfer	<b>_</b>
serial 😤 Instrument Setup	Diluent Labware Tune: Beservo	ir
Pilution Serial dilution of 10	Diluent Position: P8	
	Diluent Fosition. 10	
Span-8 Aspirate	Diluteri Liquia ype. Water	
8	Dilution Hatto : I: 2.0	wells
Span-8		
Dispense		
99	North Sec in Materia	
Span-8 New Tips		
	Change tips between diluent transfers	
Span-8 Tip	Customize	
Discard T T	Iechnique: Span-8	
	♥ Diluent Properties	_
Span-8 Wash Tips		
Transfer		
From File		
Pattern 4	P3 P7 P11 P15 P19	
Method8* BiomekFX BiomekFX ETC: 0:00:0	3	



- 4. Verify the **Diluent Position** of the diluent labware.
- 5. Select the **Diluent Liquid Type** of the diluent in the diluent labware.
- 6. Enter the desired **Dilution Ratio**. If using a variable, enter an equal sign (=) followed by the variable name; for example, =AspValue. Otherwise, just enter the desired value.

**Note:** The Dilution Ratio is set up as a ratio of volume of sample to total solution (sample plus diluent) volume. For example, a Dilution Ratio of 1:2 indicates that for every microliter of sample there are two microliters of solution, or a 50 percent dilution (one microliter of diluent added).

7. Select the wells on the labware from which to aspirate diluent. Select the wells according to the procedures described in Chapter 11, *Creating Well Patterns*.

**Note:** To zoom in on the labware, double-click on the labware. To call up a menu for Zoom In and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows the selection of the first well in a microplate as a target for the dispense operation. Specify Selection as Text may also be used to enter variables or expressions. Wells are numbered left to right and top to bottom (for example, the first row is wells 1-12, the second row wells 13-24, etc., for a 96-well microplate).

8. Select **Add Diluent to leftmost/topmost selected wells** if it is desired to use the first selected well for the first dilution. Enough diluent to achieve the entered Dilution Ratio is added to the first selected well.

**Note:** If the dilution Direction is configured as **Left to Right**, Add diluent to leftmost selected wells is available (Figure 18-4). If the dilution Direction is configured as **Top to Bottom**, Add diluent to topmost selected wells is available (Figure 18-5).

- If using an *FX* or *NX-S8*, choose Wash tips in to use the active 96-Channel Tip Wash ALP to wash the tips.
  - > **3000** Tip wash options are not available. Proceed to step 17.

**Note:** A WashStation96 ALP must be added to the deck and configured in the Deck Editor before washing can occur. Clicking on a WashStation96 ALP while the Serial Dilution step is being configured places a check in Wash tips in, and populates Wash tips in with the solution configured for the WashStation96 ALP in Labware Properties on the Instrument Setup step.

**Note:** If Wash tips in is not selected, proceed to step 13.

10. In **Wash tips in**, select a solution for the wash cycle if the field does not already contain a solution.

**Note:** The wash solution specified in the Serial Dilution step must match the solution selected in Labware Properties for the WashStation96 ALP on the Instrument Setup step configuration, or an error occurs.

- 11. Choose the number of cycles to aspirate and dispense during wash.
- 12. In %, provide the volume of wash fluid to use when washing tips as a percentage of the maximum volume of fluid contained in the tips in previous steps; for example, if the maximum volume of fluid transferred is 50  $\mu$ L, and the % is set for 110%, the Transfer step washes the tips with 55  $\mu$ L of solution.

**Note:** If the % sign is deleted in the % field, the field label changes to µL. Any value entered is now interpreted as a specific volume, not a percentage.

 If using an *FX* or *NX-S8* with fixed tips installed, choose Wash tips with to use the passive Span-8 Tip Wash ALP to wash fixed tips.

**Note:** A Span-8 Tip Wash ALP must be added to the deck and configured in the Deck Editor before washing can occur. Only fixed tips can be washed with the Span-8 Tip Wash ALP. If using disposable tips and washing is desired, go back to 9. If Wash tips with is not selected, proceed to step 17.

- 14. In **Wash tips with**, enter the volume of system fluid to use when washing the outside of the tips.
- 15. In **after dispensing**, enter the volume of system fluid to use when washing the inside of the tips.
- 16. If using a Span-8 Pod on *FX*, select **Speed Pump** to use the speed pump when flushing the tips with system fluid, if desired.

OR

If using a Span-8 Pod on *NX-S8*, select **purge pump** to use the purge pump when flushing tips with system fluid, if desired.

**Note:** The speed pump or purge pump is an optional device that can be used to shorten the time it takes to wash tips. A speed pump or purge pump must be added as a device and configured with the Span-8 Pod in Hardware Setup to use it when washing tips.

17. Choose **Change tips between diluent transfers** to change tips between each diluent transfer.

**Note:** Change tips between diluent transfers is available only when disposable tips are installed on all probes.

 The software automatically selects a Technique based on the liquid type, labware type, tip type, and volume being transferred. To override this selection, deselect Auto-Select and choose the desired Technique.

OR

Choose **Customize** and configure the technique as desired in the **Technique** Editor.

**Note:** For more information on customizing, configuring, and saving a technique, refer to Section 12.3.1, <u>Understanding Techniques</u>.

**Note:** It is recommended to use a technique that includes mixing for best results. Select **Customize** to add mixing to the technique or create a new technique in the Technique Editor.

19. Select a step that occurs after the Serial Dilution step or the **Finish** step to validate the step configuration.

## 18.4 Span-8 Aspirate Step

The Span-8 Aspirate step (Figure 18-8) removes liquid from source labware in preparation for dispensing to destination labware. The Span-8 Aspirate step is often used in conjunction with the Span-8 Dispense, Span-8 New Tips, and Span-8 Discard Tips steps, but it can be used with any combination of steps. Span-8 Aspirate provides more direct control over pipetting operations than the Transfer or Combine steps because it offers more explicit control over individual probe aspiration volumes and well access.

**Note:** If the Span-8 Pod is equipped with disposable tips, it is necessary to load tips to the probes using a Span-8 New Tips step prior to aspirating.

The Span-8 Aspirate Step Configuration includes specifying:

- Labware type accessed during aspirate.
- Position of the labware on the deck.
- Type of liquid to aspirate.
- Volume to aspirate.
- Pod that performs the aspirate.
- Probes that are used during aspirate.
- Spacing between probes during aspirate.
- Tip refresh, if desired, and the type of tips used.
- Wells in the selected labware from which to aspirate.
- Technique.
- Pipetting height.

Insert a **Span-8 Aspirate** step into the method (Figure 18-8).

8

Span-8 Aspirate



Figure 18-8. Span-8 Aspirate step and configuration

**Biomek Software User's Manual** 

#### 18.4.1 Configuring the Span-8 Aspirate Step

When a Span-8 Aspirate step is added to a method, the Span-8 Aspirate Step Configuration appears (Figure 18-8).

To configure the Span-8 Aspirate step:

1. Click on the desired piece of labware in the Current Deck Display. Information for that piece of labware is entered automatically into the Span-8 Aspirate Step Configuration.

OR

🌵 Biomek® Software - Method2\* [New] \_ 🗆 🗙 File Edit Project Instrument Execution Options Help ▶ □□ × 🖃 Start 1 ₹<sub>n</sub> The second secon Instrument Setup Serial Dilutior Instrument Setup 2 Aspirate 50 µL at P7 R Finish Span-8 Dispense Span-8 New Tip: 3 Transfer -5 4 Combine 2.00 mm from bottom [Overrides Technique] Labware Type: BCFlat96 Pod1 • Pod: • Move Labwan P7 Position: • 1 2 3 4 5 6 7 8 Probes: 8 Spacing: 2 Water Liquid Type: • 5pan-8 Tip Discard 50 Individual Volumes <u>V</u>olume: μL Pause ▼ Refresh Tips AP96\_200uL\_LLS ▼ Span-8 Wash Tips Auto-Select Customize... Commen Iechnique: Span-8 Low 80 R Ŧ Transfer From File \*\*\*\*\*\*  $\diamond$ Define Patterr • Method2\* BiomekNX Span BiomekNX-Span ETC: 0:00:03

Select a **Labware Type** from the drop-down list (Figure 18-9).

Figure 18-9. Span-8 Aspirate step configuration with labware configured

2. Verify the **Position** of the labware.

**Note:** A bright blue outline appears in the Current Deck Display around the selected labware.

- 3. Select the Liquid Type contained in the selected labware.
- 4. If using a dual-pod *FX* instrument, specify the **Pod** performing the aspirate operation by selecting a Span-8 Pod from the list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.

5. Select the **Probes** used in the aspirate step by highlighting the desired probe numbers. Any combination of probes may be used as long as all selected probes are using the same type of tips and syringes.

**Note:** All disposable tips are selected by default. If there are only fixed tips, all tips are selected by default. For example, if the Span-8 Pod is equipped with fixed tips on probes 1-4 and disposable tips on probes 5-8, probes 5-8 are selected by default. However, if probes 5-8 are also equipped with fixed tips, all eight probes are selected by default.

6. Select the **Spacing**. Spacing is the number of wells between probes. For example, if a spacing of **3** is selected, the probes hit every third well.

**Note:** When aspirating from a 384-well microplate, it is not possible to aspirate from adjacent wells. A minimum spacing of 2 wells is required. When aspirating from some tube racks, it must aspirate from adjacent wells and probe spacing is not possible.

7. Select the wells on the labware to aspirate from by hovering the mouse over the graphic representing the labware. The wells highlighted in yellow indicate the wells accessed based on the selected probes and spacing. Click to select the wells. The wells appear blue after they are selected.

For example, if aspirating from a 384-well plate, probes 1-4 are selected, and Spacing is set at 4, wells are selected as shown in Figure 18-10.



Figure 18-10. Example of how probe and spacing selection affects well selection

**Note:** To zoom in on the selected labware, double-click on the labware. To call up a menu for Zoom In and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows the selection of the first well in a microplate as a target for the aspirate operation. The first selected probe (lowest probe number) aspirates from the specified well. Remaining probes aspirate from wells based on the specified well and Spacing. Specify Selection as Text may also be used to enter variables or expressions. Wells are numbered left to right and down (for example, the first row is wells 1-12, the second row wells 13-24, etc., for a 96-well microplate).

 Enter the Volume aspirated from the selected labware. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue. The Volume is aspirated by all selected Probes.

Probe 3:    0    µL      Probe 4:    0    µL      Probe 5:    0    µL      Probe 6:    0    µL      Probe 7:    0    µL      Probe 8:    0    µL	Probe <u>1</u> : 0 Probe <u>2</u> : 0	μL μL	
Probe 5:  0  µL  the step configuration appears grayed out in    Probe 7:  0  µL  Individual Volumes.	Probe <u>3</u> : 0 Probe <u>4</u> : 0	μL μL	A probe that is not selected on
Probe <u>7</u> :  0  μL  out in    Probe <u>8</u> :  0  μL  Individual    Volumes.	Probe <u>5</u> : 0 Probe <u>6</u> : 0		the step configuration appears grayed
	Probe <u>7</u> : 0 Probe <u>8</u> : 0	μL μL	out in Individual Volumes.

9. To aspirate a different volume in each probe, select **Individual Volumes** and choose **Edit**. Individual Volumes appears (Figure 18-11).

Figure 18-11. Individual Volumes

- 10. Enter the volume each probe is to aspirate in Individual Volumes. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue. A volume may only be entered for the probes that are selected on the Span-8 Aspirate step configuration. Deselected probes appear grayed out in Individual Volumes.
- 11. Choose OK. Individual Volumes closes.
- 12. Check **Refresh Tips**, if desired, and specify the **tip type** to load new tips to the pod whether tips are currently loaded or not.

 The software automatically selects a Technique based on the liquid type, labware type, tip type, and volume being aspirated. To override this selection, deselect Auto-Select and select the desired Technique (Figure 18-12).

OR

Choose **Customize** and configure the technique as desired in the **Technique** Editor.

**Note:** For more information on customizing, configuring, and saving a technique, refer to Section 12.3.1, <u>Understanding Techniques</u>.

**Note:** If the technique is auto-selected and the volumes specified differ enough to require different techniques, an aspiration is performed for each unique technique.

🌵 Biomek@	® Software	- Method2* [New]		
File Edit F	Project Inst	rument Execution	Options Help	
		Start		
Serial Dilution	Instrument Setup	🐔 Instrumer	nt Setup	for Span-8 Aspirate
Ŷ	R	Aspirate 5	i0 µL at P7	
Span-8 Aspirate	Transfer	🎽 Finish		30000000000
Î	-			
Span-8 Dispense	Combine			Height display
66				Labware Type: BCFlat96 Pod1
Span-8 New Tips	Move Labware			Position: P7 Probes: 1 2 3 4 5 6 7 8
	0			Liquid Type: Water Spacing: 2 👗
Span-8 Tip Discard	Pause			Volume: 50 µL In Notifies that the technique is
$\mathbf{T}\mathbf{T}$				
Span-8 Wash Tips	Comment			✓ Auto-Select
				Iechnique: Span-8 Low 80
Transfer From File				
Define Pattern				selection p2 p11
		•	•	
Method2*	BiomekNX Sc	an BiomekNX-Spen	ETC: 0:00:03	
p iccitouz	Promoran of	an pioneway span	L.C. 0.00.05	

Figure 18-12. Span-8 Aspirate step — Adjusting the aspirate height

14. Leave the tip height as is to use the settings specified in the pipetting technique.

OR

Set the aspirate height manually using one of the following techniques:

• Set the aspirate height by positioning the cursor over the graphic of a tip inside a well (Figure 18-12). Click and drag the graphic up or down. The tip moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.

Note: graphic

**Note:** The cursor changes to a hand when positioned over the graphic.

- Position the cursor in the graphic of a tip inside a well, and adjust the height using the ↑ and ↓ keys.
- Right-click on the graphic of a tip inside a well, and a menu appears. Select **Custom Height**, and **Custom Height** appears (Figure 18-13). Insert the **Height** in millimeters (mm) and, in **from**, select a reference point from the drop-down list.

Custom Height	
Height:	mm
from Bottom	<b>~</b>
OK	Cancel

Figure 18-13. Custom Height prompt

**Note:** The phrase (Overrides Technique) appears below the graphic when the tip height is changed by manipulating the graphic (Figure 18-12), or when the technique is changed with the Customize option.

- 15. If desired, right-click on the graphic of a tip inside a well and choose how the tip height is measured:
  - **Measure from Liquid** The tip height is measured from the surface of the liquid in the well.

**Note:** With standard tips, Measure from Liquid requires that a known volume of liquid be entered in Labware Properties for the labware to aspirate from. If the volume of liquid is not known, liquid level sensing (LLS) tips must be used to measure the tip height from the surface of the liquid in the well (refer to Section 9.4.5, <u>Setting Liquid</u> Level Sensing Values (FX, 3000, and NX-S8 only)).

- **Measure from Bottom** The tip height is measured from the bottom of the well.
- **Measure from Top** The tip height is measured from the top of the well.
- 16. Select a step that occurs after the Span-8 Aspirate step or the **Finish** step to validate the step configuration.
## 18.5 Span-8 Dispense Step

The Span-8 Dispense step dispenses liquid into a single destination labware following aspirate. The Span-8 Dispense step is often used in conjunction with the Span-8 Aspirate, Span-8 New Tips, and Span-8 Discard Tips steps, but it can be used with any combination of steps. Span-8 Dispense provides more direct control over pipetting operations than the Transfer or Combine steps because it offers more explicit control over individual probe dispense volumes and well access.

Tips must be loaded and have liquid in them from a previous step, such as a Span-8 Aspirate step, prior to using a Span-8 Dispense step.

The Span-8 Dispense Step Configuration includes specifying:

- Labware type accessed during dispense.
- Position of the labware on the deck.
- Type of liquid to dispense.
- Volume to dispense.
- Pod that performs the dispense.
- Probes that are used during dispense.
- Spacing of the probes during dispense.
- Empty tips, if desired.
- Wells in the selected labware to dispense.
- Technique.
- Pipetting height.
- Disposal of excess liquid into the reservoir section of the nearest Span-8 Wash ALP.



Insert a **Span-8 Dispense** step into the method (Figure 18-14).

( <sup>1</sup> ) Biomek® Software - Span-8 Steps* [Development]	IX
File Edit Project Instrument Execution Options Help	
Start Start	
set 🙀 Instrument Setup	
Aspirate 50 µL at Source	
Labware Type: Pod2	
Protest         1         2         3         4         5         5         2         0	
Wietnoa View	
Span® Tip Discard Excess Individual Volumes Edt	
displayed contains a	
Span-8 Dispense Zutorite. Stare As.	
step. The location of	
Transfer the dispense	
Operation is specified	
Define Uperation is specified Span-8 Dispense Step Configuration	
Method View.	
Do Desti Dest	
P3 P7 P11 P15 TR1	
Span-8 Steps* BiomekFX BiomekFX ETC: 0:00:24	

Figure 18-14. Span-8 Dispense step and configuration

#### 18.5.1 Configuring the Span-8 Dispense Step

When a Span-8 Dispense step is added to a method, the Span-8 Dispense Step Configuration appears (Figure 18-14).

To configure the Span-8 Dispense step:

1. Click on the desired piece of labware in the Current Deck Display. Information for that piece of labware is entered automatically into the Span-8 Dispense Step Configuration.

OR

Select a Labware Type from the list (Figure 18-15).

🌵 Biomek® Software - Span-8 Steps [Development]		_0×
File Edit Project Instrument Execution Options Help		
Start		
Serial 😪 Instrument Setup		
Aspirate 50 ult at Source		
Aspirate Dispense 50 µL at Dest1		
Finish		
Span-8 Dispense	2.70 mm from bottom	
X X	[Dverrides Technique]	
Span-8	Labware Type: BCFlat96  Pod: Pod2	
New Tips	Position: Dest1 Probes: 1 2 3 4 5 6 7 8	
MU .	Liquid Type: Water Spacing I T Empty Tips	
Span-8 Tip Discard	Volume: 50 μL Discard Excess Individual Volumes Edit	
Y.V.		
Span-8	V Auto-Select Customize Save As	
(Vash lips	Technique: Span-8 Low 80	
Turnels		
From File		
Define Pattern		
	······,	
	P2 P6 Pest1 Pest2	
	P3 P7 P11 P15 TR1	
Span-8 Steps BiomekFX BiomekFX ETC: 0:00:24		

Figure 18-15. Span-8 Dispense step configuration with labware configured

2. Verify the **Position** of the labware.

**Note:** A bright yellow outline appears in the Current Deck Display around the selected labware.

- 3. Select the Liquid Type contained in the tips.
- 4. If using a dual-pod *FX* instrument, specify the **Pod** performing the dispense operation by selecting a Span-8 Pod from the list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.

5. Select the **Probes** that will be used in the dispense step by highlighting the desired probe numbers. Any combination of probes may be used as long as all selected probes are using the same type of tips and syringes.

**Note:** All disposable tips are selected by default. If there are only fixed tips, all tips are selected by default. For example, if the Span-8 Pod is equipped with fixed tips on probes 1-4 and disposable tips on probes 5-8, probes 5-8 are selected by default. However, if probes 5-8 are also equipped with fixed tips, all eight probes are selected by default.

6. Select the **Spacing**. Spacing is the number of wells between probes. For example, if a spacing of **3** is selected, the probes will hit every third well.

**Note:** When dispensing to a 384-well microplate, it is not possible to dispense to adjacent wells. A minimum spacing of 2 wells is required. When dispensing to tube racks, it must dispense to adjacent tubes and probe spacing is not possible.

7. Check **Empty Tips**, if desired, to expel the **Trailing Air Gap**, the fluid contained in the tip, and the Blow Out all at one time, rather than as separate operations.

#### Note: The default setting for Empty Tips is Off.

8. Select the wells on the labware to dispense to by hovering the mouse over the graphic representing the labware. The wells highlighted in yellow indicate the wells accessed based on the selected probes and spacing. Click to select the wells. The wells appear blue after they are selected.

For example, if dispensing to a 384-well plate, probes 1-4 are selected, and Spacing is set at 4, wells are selected as shown in Figure 18-16.

2	Spacing of four leaves three empty wells between selected wells.

Figure 18-16. Example of how probe and spacing selection affects well selection

**Note:** To zoom in on the selected labware, double-click on the labware. To call up a menu for Zoom In and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows the selection of the first well in a microplate as a target for the dispense operation. The first selected probe (lowest probe number) dispenses from the specified well. Remaining probes dispense from wells based on the specified well and Spacing. Specify Selection as Text may also be used to enter variables or expressions. Wells are numbered left to right and down (for example, the first row is wells 1-12, the second row wells 13-24, etc., for a 96-well microplate).

- Enter the Volume dispensed to the selected labware. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue. The Volume is dispensed from all selected Probes.
- 10. To dispense a different volume in each probe, select **Individual Volumes** and choose **Edit**. Individual Volumes appears (Figure 18-17).

Probe <u>1</u> : 0		μL		
Probe <u>2</u> : 0		μL		
Probe <u>3</u> : 0		μL		
Probe <u>4</u> : 0		μL		A probe that is not selected on
Probe <u>5</u> : 0		μL		the step
Probe <u>6</u> : 0		μL		configuration
Probe <u>7</u> : 0		μL		out in
Probe <u>8</u> : 0		μL		Individual Volumes.
ОК	Can	cel	-	

Figure 18-17. Individual Volumes

- 11. Enter the volume each probe is to dispense in Individual Volumes. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue. A volume may only be entered for the probes that are selected on the Span-8 Dispense step configuration. De-selected probes appear grayed-out in Individual Volumes.
- 12. Choose OK. Individual Volumes closes.

 The software automatically selects a Technique based on the liquid type, labware type, tip type, and volume being dispensed. To override this selection, deselect Auto-Select and select the desired Technique (Figure 18-18).

OR

Choose **Customize** and configure the technique as desired in the **Technique** Editor.

**Note:** For more information on customizing, configuring, and saving a technique, refer to Section 12.3.1, *<u>Understanding Techniques</u>*.

**Note:** If the technique is auto-selected and the volumes specified differ enough to require different techniques, a dispense is performed for each unique technique.



Figure 18-18. Span-8 Dispense step — Adjusting the dispense height

14. Leave the tip height as is to use the settings specified in the pipetting technique.

OR

Set the aspirate height manually using one of the following techniques:

Set the aspirate height by positioning the cursor over the graphic of a tip inside a well (Figure 18-18). Click and drag the graphic up or down. The tip moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.

**Note:** The cursor changes to a hand when positioned over the graphic.

- Position the cursor in the graphic of a tip inside a well, and adjust the height using the ↑ and ↓ keys.
- Right-click on the graphic of a tip inside a well, and a menu appears. Select **Custom Height**, and **Custom Height** appears (Figure 18-19). Insert the **Height** in millimeters (mm) and, in **from**, select a reference point from the drop-down list.



Figure 18-19. Custom Height prompt

**Note:** The phrase (Overrides Technique) appears below the graphic when the tip height is changed by manipulating the graphic (Figure 18-18), or when the technique is changed with the Customize option.



- 15. If desired, right-click on the graphic of a tip inside a well and choose how the tip height is measured:
  - **Measure from Liquid** The tip height is measured from the surface of the liquid in the well.

**Note:** With standard tips, **Measure from Liquid** requires that a known volume of liquid be entered in **Labware Properties** for the labware to dispense to. If the volume of liquid is not known, liquid level sensing (LLS) tips must be used to measure the tip height from the surface of the liquid in the well.

- **Measure from Bottom** The tip height is measured from the bottom of the well.
- **Measure from Top** The tip height is measured from the top of the well.
- 16. Check **Discard Excess** to dispense any excess liquid into the reservoir section of the nearest Span-8 Wash ALP.

Note: The default setting for Discard Excess is Off.

17. Select a step that occurs after the Span-8 Dispense step or the **Finish** step to validate the step configuration.

# 18.6 Span-8 New Tips Step

The Span-8 New Tips step instructs the Biomek instrument to load new tips to the Span-8 Pod at a specific point in a method. Tips may also be loaded onto the probes of the Span-8 Pod automatically as part of a Transfer, Combine, or Span-8 Aspirate step. Span-8 New Tips is used when precise tip control is required. Span-8 New Tips is used in conjunction with the Span-8 Discard Tips step (refer to Section 18.7, *Span-8 Tip Discard Step*).

Loading new tips is recommended when:

- Current tips have been used and carryover must be avoided.
- Current tips have manipulated a caustic fluid.
- Current tips are not seated properly.
- Integrity of the current tips is in question.

Insert a Span-8 New Tips into the method (Figure 18-20).



Biomek® Software - Span-8 Steps* [Development]     File Edit Project Instrument Execution Ontions Help		- D ×
Seine Seine Spans Aspire Seine Diperse Diperse Diperse Diperse Seine Seine Diperse Seine	Pote Pod2 Ipx: AP96.2004 Prote: E E E E E E E Span-8 New Tips Step Configuration	
New Tips		
Span-8 Tip Discard The method		
LL displayed contains a		
Wash Tips Span-8 New Tips		
step prior to the		
From File Span-8 Aspirate		
Define Pattern		
	TL1 P4 P8 P12 P16	
	P1 P5	
	P3 P7 P11 P15 TR1	
Span-8 Steps* BiomekFX BiomekFX ETC: 0:00:03		

Figure 18-20. Span-8 New Tips step and configuration

### 18.6.1 Configuring the Span-8 New Tips Step

When a Span-8 New Tips step is added to a method, the Span-8 New Tips Step Configuration appears (Figure 18-20).

To configure Span-8 New Tips:

- 1. If using a dual-pod *FX* instrument, specify the **Pod** requiring new tips by selecting a Span-8 Pod from the list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.
- 2. Select the **Tips** to load in the **Span-8** New Tips step.

OR

Select the tip box from the Current Deck Display with the desired tip type. Selecting tips from the Current Deck Display automatically updates Tips in the Step Configuration.

**Note:** If tip boxes were assigned names in the Instrument Setup step, enter that name in Tips. If tips were not assigned names, select the tip type. It is recommended that tips not be named in the Instrument Setup step because this could restrict the Span-8 New Tips step from locating available tips.

3. Select the **Probes** that need new tips by highlighting the probe numbers.

**Note:** All disposable tips are selected by default. If there are only fixed tips, all tips are selected by default. For example, if the Span-8 Pod is equipped with fixed tips on probes 1-4 and disposable tips on probes 5-8, probes 5-8 will be selected by default. However, if probes 5-8 are also equipped with fixed tips, all eight probes are selected by default.

Note: Probes equipped with fixed tips cannot have tips loaded.

4. Select a step that occurs after the Span-8 New Tips step or the **Finish** step to validate the step configuration.

## 18.7 Span-8 Tip Discard Step

The Span-8 Tip Discard step instructs the Biomek instrument to unload tips from a Span-8 Pod at a specific point in a method. Tips may also be discarded from the probes of the Span-8 Pod automatically as part of a Transfer, Combine, or Span-8 Dispense step. Span-8 Tip Discard is used when precise tip control is required. Span-8 Tip Discard is often used in conjunction with the Span-8 New Tips step, but it can be used with any combination of steps.

Span-8 Tip Discard is used when tip integrity is in question or when crosscontamination between labware is a concern. Span-8 Tip Discard can be used in place of the Unload (Tips) option in the Transfer and Combine steps, and is particularly useful with Loop and Worklist steps.

Unloading tips is recommended when:

- Current tips have been used and carryover must be avoided.
- Current tips have manipulated a caustic fluid.
- Current tips are not seated properly.
- Integrity of the current tips is in question.
- To avoid cross-contamination of samples.

Insert a **Span-8 Tip Discard** step into the method (Figure 18-21).





Figure 18-21. Span-8 Tip Discard step and configuration

#### 18.7.1 Configuring the Span-8 Tip Discard Step

When a Span-8 Tip Discard step is added to a method, the Span-8 Tip Discard Step Configuration appears (Figure 18-21).

To configure Span-8 Tip Discard:

- 1. If using a dual-pod *FX* instrument, specify the **Pod** requiring tip removal by selecting a Span-8 Pod from the list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.
- 2. Select the **Probes** that will discard tips by highlighting the probe numbers.

**Note:** All disposable tips are selected by default. If there are only fixed tips, all tips are selected by default. For example, if the Span-8 Pod is equipped with fixed tips on probes 1-4 and disposable tips on probes 5-8, probes 5-8 will be selected by default. However, if probes 5-8 are also equipped with fixed tips, all eight probes are selected by default.

Note: Probes equipped with fixed tips cannot discard tips.

3. Select a step that occurs after the Span-8 Tip Discard step or the **Finish** step to validate the step configuration.

## 18.8 Span-8 Wash Tips Step

The Span-8 Wash Tips step flushes fixed tips with system fluid and cleans tips using the Span-8 Tip Wash ALP (refer to the *ALPs User's Manual*, Chapter 18, <u>Span-8 Tip</u> <u>Wash ALP</u>), or washes disposable tips by performing multiple aspirate and dispense cycles at a 96-Channel Tip Wash ALP (refer to the *ALPs User's Manual*, Chapter 13, <u>Multichannel Tip Wash ALPs</u>). Use the Deck Editor to add and configure the appropriate wash ALP to the deck (refer to Chapter 5, <u>Preparing and Managing the</u> <u>Deck</u>).

The Span-8 Wash Tips step can be configured to perform either a passive wash or an active wash. A Passive Wash uses the Span-8 Tip Wash ALP, a Passive ALP, to wash fixed tips. An Active Wash uses a 96-Channel Tip Wash ALP, an Active ALP, to wash disposable or fixed tips.

**Note:** It is recommended that a **Span-8** Wash Tips step be inserted at the start of every method to purge the tubing and syringes of air. If the Span-8 Pod is idle during the method, it may be necessary to insert another **Span-8** Wash Tips step into the method immediately before a Span-8 Pod begins to move.

**Note:** If using a system with mixed fixed and disposable tips, it is necessary to insert separate Span-8 Wash Tips steps to purge each set of four probes used in the method.

The Span-8 Wash Tips Step Configuration includes specifying:

- Pod performing the wash operation.
- Wash station position on the deck.
- Type of wash to perform: Active Wash or Passive Wash.
- Volume of system fluid used in the wash process.
- Technique, liquid type, and number of wash cycles if an Active Wash is chosen.
- Delay before proceeding with method.
- Which probes to wash.



Insert a **Span-8 Wash Tips** step into the method (Figure 18-22).



Figure 18-22. Span-8 Wash Tips step — Passive Wash configuration

#### 18.8.1 Configuring the Span-8 Wash Tips Step for a Passive Wash

When a Span-8 Wash Tips step is added to a method, the Span-8 Wash Tips Step Configuration appears (Figure 18-22).

A Passive Wash uses the Span-8 Tip Wash ALP to cleanse fixed tips. To perform a Passive Wash, a Span-8 Tip Wash ALP must be physically on the deck and configured in Instrument Setup (refer to Chapter 4, <u>Configuring Hardware Setup</u>) and the Deck Editor (refer to Chapter 5, <u>Preparing and Managing the Deck</u>).

The Span-8 Tip Wash ALP has two parts: an open drain for dispensing waste, and eight wells used to clean the tips with system fluid. System fluid washes the inside of the tips and is dispensed through the open drain. Then the tips fit into the wells and the outside of the tips are washed with system fluid.



#### CAUTION: Do not purge the system without mandrels installed and tubing attached to disposable or fixed tips. Purging the system without the mandrels installed and the tubing attached to tips may cause corrosion in the tip interface.

**Note:** The Span-8 Wash Tips step Passive Wash can be used at the beginning of a method or during a method in which the Span-8 Pod is idle to purge the tubing and syringes of air rather than using Manual Control.

**Note:** All disposable tips must be discarded prior to initiating a Span-8 Tip Wash Passive Wash to clean the probes or purge the system.

To configure the Span-8 Wash Tips step for a Passive Wash:

- 1. If using a dual-pod *FX* instrument, specify the **Pod** performing the wash operation by selecting a Span-8 Pod from the list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.
- 2. Select a **Position** from the drop-down list.

**Note:** The Span-8 Wash ALP is added to the deck and configured for use with the Deck Editor (refer to Chapter 5, *<u>Preparing and Managing the Deck</u>*).

Note: If a wash station is not selected, the closest available wash station is used.

3. Select the **Probes** to wash.

**Note:** All disposable tips are selected by default. If there are only fixed tips, all tips are selected by default. For example, if the Span-8 Pod is equipped with fixed tips on probes 1-4 and disposable tips on probes 5-8, probes 5-8 will be selected by default. However, if probes 5-8 are also equipped with fixed tips, all eight probes are selected by default.

**Note:** All disposable tips must be discarded prior to initiating a Span-8 Tip Wash Passive Wash to clean the probes or purge the system.

4. Select **Only wash tips that have been used** to automatically wash only the probes have been used. The software determines from the method which probes require washing and washes only those tips.

**Note:** When using the Passive Wash to purge the tubing and syringes of air, make sure that Only wash tips that have been used is not selected to flush all probes with system fluid.

5. Select **Passive Wash** to wash the fixed tips using a Span-8 Wash ALP. The **Passive Wash** configuration appears (Figure 18-22).

**Note:** When using the **Passive Wash** to purge system tubing of air bubbles for probes equipped with disposable tips, make sure that no tips are loaded to the probes.

6. In **Dispense**, enter the volume of system fluid, in milliliters (mL), to use to wash the inside of the tips. The inside of the tips are flushed with the specified volume of system fluid and dispensed into the open drain on the Span-8 Tip Wash ALP.

**Note:** When using the **Passive Wash** to purge the tubing and syringes of air, use approximately 10 to 15 mL of system fluid.

7. In **Wash**, enter the volume of system fluid, in milliliters (mL), to use to wash the outside of the tips. The outside of the tips are washed with the specified volume of system fluid while the tips are positioned in the wells of the Span-8 Tip Wash ALP.

**Note:** If Dispense tip contents only is selected, any liquid in the tips is dispensed to the open drain, but the tips are not flushed with system fluid. The Dispense to waste and Wash while in the wells options are unavailable when Dispense tip contents only is selected.

8. In **Delay**, enter the time, in milliseconds (ms), to hold the pod over the Span-8 Wash ALP after each dispense.

Note: The default Delay is 300 milliseconds (ms).

9. If using a Span-8 Pod on *FX*, select **Speed Pump** to use the speed pump when flushing the tips with system fluid, if desired.

OR

If using a Span-8 Pod on *NX-S8*, select **purge pump** to use the purge pump when flushing tips with system fluid, if desired.

**Note:** The speed pump or purge pump is an optional device that can be used to shorten the time it takes to wash tips. A speed pump or purge pump must be added as a device and configured with the Span-8 Pod in Hardware Setup to use it when washing tips.

10. Select a step that occurs after the Span-8 Wash Tips step or the **Finish** step to validate the step configuration.

#### 18.8.2 Configuring the Span-8 Wash Tips Step for an Active Wash

When a Span-8 Wash Tips step is added to a method, the Span-8 Wash Tips Step Configuration appears (Figure 18-22).

An Active Wash uses the 96-Channel Tip Wash ALP to cleanse disposable or fixed tips. To perform an Active Wash, a 96-Channel Tip Wash ALP must be physically on the deck and configured in Instrument Setup (refer to Chapter 4, <u>Configuring</u> <u>Hardware Setup</u>) and the Deck Editor (refer to Chapter 5, <u>Preparing and Managing</u> <u>the Deck</u>).

FX — In a dual-pod system, the 96-Channel Tip Wash ALP must be located in a deck position accessible by the Span-8 Pod to use Active Wash options.

To configure the Span-8 Wash Tips step for an Active wash:

- 1. If using a dual-pod *FX* instrument, specify the **Pod** performing the wash operation by selecting a Span-8 Pod from the list. The pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.
- 2. Click on the **Wash Station 96 ALP** in the Current Deck Display. Information for that wash station is entered automatically into the **Wash** Step Configuration.

OR

Select a Position.

**Note:** The Wash Station 96 ALP is added to the deck and configured for use with the Deck Editor (refer to Chapter 5, <u>*Preparing and Managing the Deck*</u>).

**Note:** If a wash station is not selected, the closest available wash station is used.

3. Select the **Probes** to wash.

**Note:** All disposable tips are selected by default. If there are only fixed tips, all tips are selected by default. For example, if the Span-8 Pod is equipped with fixed tips on probes 1-4 and disposable tips on probes 5-8, probes 5-8 will be selected by default. However, if probes 5-8 are also equipped with fixed tips, all eight probes are selected by default.

4. Select **Only wash tips that have been used** to automatically wash only the tips that have been used. The software determines from the method which tips require washing and washes only those tips.

5. Select **Active wash** to wash the disposable or fixed tips using a 96-Channel Tip Wash ALP. The **Active wash** configuration appears (Figure 18-23).

👘 Biomek® Software - Span-8 Steps* [Development]		
File Edit Project Instrument Execution Options Help		
Start	Pod: Pod2 Probes: 1 2 3 4 5 6 7 8	Week Ontione
serial 🥰 Instrument Setup	Position: 🔽 🔽 Only wash tips that have been used	wash Options
	C Passive Wash	Options to wash tips
	Active Wash	using a Passive wash
Aspirate Span-8 Wash Lips	⊻olume: 110% %	or an Active wash
Aspirate 50 µL at Source	Wash Cycles: 4 times	
span-8 🕴 Dispense 50 µL at Dest1	Ljquid Type: Water	
Span-8 Tip Discard	Tel Auto Calcat	
	Tashainan Du/ashina	
₩ <b>&gt;</b>	T T	
Span-8 Tip Discard		
	Ener 9 Week Tine	
Span-8	Span-8 wash rips	
Wash Tips	Step Active Wash	
	Configuration	
From File		
Define		
Paven		
	TL1 P4 P8 P12 P16	
		10000000
	P1 P5	
	Do Do Do Dosti	
	P3 P7 P11 P15 TR1	
Span-8 Steps* BiomekFX BiomekFX ETC: 0:00:06		

Figure 18-23. Span-8 Wash Tips step — Active Wash configuration

6. Enter the **Volume** of fluid to use when washing the tips.

**Note:** Enter the volume as a percent (including the percent sign, %) to specify the volume of wash fluid as a percent of the maximum volume of fluid contained in the tips in previous steps. For example, if the maximum volume of fluid transferred is 50  $\mu$ L, and the % is set for 110%, the Wash step washes the tips with 55  $\mu$ L of solution. However, if no fluid has been transferred and the volume is entered as a percent, no action is taken and tips are not washed.

- 7. Specify in Wash Cycles the number of aspirate and dispense operations completed during the wash operation.
- 8. Verify the Liquid Type contained in the selected 96-Channel Tip Wash ALP.

**Note:** The Liquid Type contained in the 96-Channel Tip Wash ALP is configured in the Instrument Setup step. If the Wash Station is selected from the Current Deck Display, the information is automatically inserted.

9. Select a step that occurs after the Span-8 Wash Tips step or the **Finish** step to validate the step configuration.

# 18.9 Transfer From File Step (including 3000)

**Transfer From File** allows a specified volume to be transferred from a specified source well to a specified destination well by reading data from a comma-delimited text file (.txt or .csv). A comma-delimited text file is a text file that specifies the values of a table by separating each column with a comma and each row with a return.

3000 — Transfer From File may only be used with a single-tip pipette tool.

The Transfer From File configuration specifies:

- Pod that performs the transfer.
- Probes that are used during the transfer.
- Tip handling.
- File properties.
- Source labware.
- Destination labware.
- Transfer details.



Insert a **Transfer From File** step into the Method View. The **Transfer From File** Step Configuration appears (Figure 18-27).

Biomek® Software - Method18* [New]     Ela Edit Drviert Instrument Evention Onlines Help	
Start Use god Pod2 v for transfer. Use probes 1 2 3 4 5 6 7 8	
serial 😪 Instrument Setup	
Dilution Transfer From File	
Span-8 Finish Water 3 gruppes of 110% %	
Change tips between transfers.	
y Span-8 Disense A File Properties	
Pod and Probe Configuration	
Span-8 New Tips The pod and probes performing the will be read again when the method is run): File has a header row	
Iquid transfer must be specified. A         Tip Handling Configuration	7
Span-8 Pod on an <b>FX</b> or <b>NX-S8</b> or Tip Handling parameters specify tip	
pipetting tool on a <b>3000</b> must be type, whether tips remain on the pod a	t
wash Tips used for the transfer operation.	
File specifies destination position in column change tips, and tip wash options.	
From File	
Dispense up to 1 interper draw.	Advanced
Pattern	
P1 P4 P8 P12 W P16	
P2 P5 P9 P13 P17	
Mathol18* BiomakEY BiomakEY FTC: 0:00:03	=

Figure 18-27. Transfer From File step and configuration

#### 18.9.1 Configuring Tip Handling

Configuring Tip Handling in the Transfer From File step is optional, because the Biomek instrument locates and loads clean tips automatically every time a Transfer From File step is initiated. It is recommended, however, that Tip Handling configuration be supplied if cross contamination or conserving consumables is a concern.

Tip Handling is also used to configure washing fixed tips using the Span-8 Wash ALP or disposable tips using the 96-Channel Tip Wash ALP. Refer to Section 18.8, *Span-8 Wash Tips Step*, for more information on washing tips on the Span-8 Pod.

To configure Tip Handling (Figure 18-27) for the Transfer From File step:

 If using a dual-pod *FX* instrument, specify the **Pod** performing the transfer operation. The pod configured as the default pod is displayed in **Pod**. If the other pod is desired, select the pod from the drop-down list or the Current Deck Display.

OR

If using a **3000**, configure a Load Tool or Change Tool step with a P20 or P200L tool in the method before the Transfer From File.

**Note:** Refer to Section 20.3, <u>*Change Tool Step*</u> or Section 20.4, <u>*Load Tool Step*</u> for information about configuring the desired step.

 If using a Span-8 pod on the *FX* or *NX-S8*, select the probes used in the transfer by clicking the probe numbers in Use Probes. Any combination of probes may be used as long as all selected probes are using the same type of tips and syringes.

OR

Right-click any of the probe numbers in **Use Probes** and make a selection from the menu. Options include:

- Use Disposable Tips selects all probes with disposable tips.
- Use Fixed Tips selects all probes with fixed tips.
- Use Selection allows for custom selection.

**Note:** Disposable tips are selected by default. If the Span-8 Pod has only fixed tips, all tips are selected by default. For example, if the Span-8 Pod is equipped with fixed tips on probes 1-4 and disposable tips on probes 5-8, probes 5-8 are selected by default. However, if probes 5-8 are also equipped with fixed tips, all eight probes are selected by default.

**Note:** If the probes selected are equipped with fixed tips, the Load and unload them/leave them on options (steps 4 and 5) are unavailable and the fixed tips are used for the transfer. Proceed to step 6.

3. If Tip Handling is not already displayed, click the arrow or sentence summary to expand the Tip Handling configuration.

4. Check Load and select the type of tips used in the transfer to load tips at the start of the Transfer From File step. If Load is unchecked, the Transfer From File step uses previously loaded tips, rather than loading new tips at the start of the Transfer From File step. Unchecking Load deactivates initial tip loading for the Transfer From File step.

**Note:** Tips are selected from the drop-down list, or from the Current Deck Display. If tips are selected via the Current Deck Display, and the tips are named, Load displays the name, not the tip type. If a name was not assigned to the tips, the tip type is displayed in Load.

It is recommended that tips not be named in the **Instrument Setup** step, because this could restrict the instrument from locating available tips.

**Note:** If Load is checked and the tips loaded on the pod have been used, an error may occur. To increase the number of uses allowed for tips, increase the Load no more than count in the Labware Properties configuration in the Instrument Setup step.

5. To discard tips after the Transfer From File step is completed, select **unload them**.

OR

To leave tips on after the Transfer From File step is completed, select **leave** them on.

**Note:** The Load and unload them/leave them on fields (steps 4 and 5 above) are useful when multiple Transfer From File steps are used within a method; for example, in the first Transfer From File step, **check Load** and select **leave them on**. In the second Transfer From File step, **uncheck Load** and select **leave them on**. In the third Transfer From File step, **uncheck Load** and select **leave them on**. In the third Transfer From File step, **uncheck Load** and select **leave them on**. In the third Transfer From File step, **uncheck Load** and select **unload them**. This loads new tips for the first Transfer From File step, uses the same set of tips for all three Transfer From File steps, and unloads the tips when the third Transfer From File step is completed. This is useful when trying to conserve consumables.

- 6. If using an *FX* or *NX-S8*, choose **Wash tips in** to use the active 96-Channel Tip Wash ALP to wash the tips.
  - 3000 Tip wash options are not available for the Biomek 3000. Proceed to step 14.

**Note:** If Wash tips in is not selected, proceed to step 10.

**Note:** A WashStation96 ALP must be added to the deck and configured in the Deck Editor before washing can occur. Clicking on a WashStation96 ALP while the Transfer From File step is being configured places a check in Wash tips in, and populates Wash tips in with the solution configured for the WashStation96 ALP in Labware Properties on the Instrument Setup step.

7. In **Wash tips in**, select a solution for the wash cycle if the field does not already contain a solution.

**Note:** The wash solution specified in the **Transfer From File** step must match the solution selected in Labware Properties for the Wash Station ALP on the Instrument Setup step configuration, or an error occurs.

- 8. Choose the number of cycles to aspirate and dispense during wash.
- 9. In %, provide the volume of wash fluid to use when washing tips as a percentage of the maximum volume of fluid contained in the tips in previous steps; for example, if the maximum volume of fluid transferred is 50  $\mu$ L, and the % is set for 110%, the **Transfer** step washes the tips with 55  $\mu$ L of solution.

**Note:** If the % sign is deleted in the % field, the field label changes to  $\mu$ L. Any value entered is now interpreted as a specific volume, not a percentage.

10. If using an *FX* or *NX-S8* with fixed tips installed, choose **Wash tips with** to use the passive Span-8 Wash ALP to wash fixed tips.

**Note:** A Span-8 Wash ALP must be added to the deck and configured in the Deck Editor before washing can occur. Clicking on a Span-8 Wash ALP while the Transfer From File step is being configured places a check in Wash tips with, and uses default values of 2 mL of system liquid after dispensing 1 mL to waste.

**Note:** Only fixed tips can be washed with the Span-8 Wash ALP. If using disposable tips or Wash tips with is not selected, proceed to step 14.

- 11. In **Wash tips with**, enter the volume of system fluid to use when washing the outside of the tips.
- 12. In **after dispensing**, enter the volume of system fluid to use when washing the inside of the tips.
- 13. If using a Span-8 Pod on *FX*, select **Speed Pump** to use the speed pump when flushing the tips with system fluid, if desired.

OR

If using a Span-8 Pod on *NX-S8*, select **purge pump** to use the purge pump when flushing tips with system fluid, if desired.

**Note:** The speed pump or purge pump is an optional device that can be used to shorten the time it takes to wash tips. A speed pump or purge pump must be added as a device and configured with the Span-8 Pod in Hardware Setup to use it when washing tips.

14. Select **Change/Wash tips between transfers** to change or wash tips between liquid transfers, if desired.

**Note:** If the probes used for transfer are equipped with disposable tips and Wash tips in is not selected, the option to Change tips between transfers is available. If the probes used for transfer are equipped with fixed tips or disposable tips with Wash tips in selected, the option to Wash tips between transfers is available.

The Tip Handling configuration can be collapsed to allow more room during File Properties, Source Labware, and Destination Labware configuration. When collapsed, the Tip Handling configuration is displayed as a sentence summary (Figure 18-28). Use the sentence summary to verify that Tip Handling has been configured appropriately.

To collapse or expand the Tip Handling configuration, click the arrow to the left of Tip Handling or the textual display (Figure 18-28).

🌵 Biomek	® Software •	- Method18* [New]				
File Edit	Project Instr	ument Execution Option	Help			
🗅 📽 (		🕘 🎖 🖻 🛍 🗠 🖉				Pod and probe
	Stari	t	Use pod Pod2 💌 for tran	fer. Use probes 1 2 3 4 5	6 7 8	selection
Serial	🧠 Instr	rument Setup	▼ Load AP96_200uL tips, change bet	een transfers, and unload them when	finished.	
Dilution	Tran	sfer From File	A File Properties		<u>▲</u>	
Υ.		L	File Name			7
Span-8 Aspirate		n	Example data (file will be read again wh	en the me Collapsed	Tip Handling	
Ĭ			Please enter a filename.	config	uration.	
Span-8 Dispense						
ΪÏ						
Span-8					-	
New Tips		Arrow to c	Dllapse/	nn <u> </u> 0	<u>리</u> ㅋ	
IMU Sava & Tin		expand Tip	Handling	in column U	-1	
Discard		• •	Elle contains destination well inform	tion in column	고 키	
പ്പ			File contains volume information in	olumn	7	
Span-8 Wash Tips			Skip zero volume transfers		-	
Transfer Erom File			Source: <file specif<="" td=""><td>ied position&gt;</td><td>Watan fram - file</td><td></td></file>	ied position>	Watan fram - file	
			Dispense up to 1 time per up	aw Draw	water from sine	Advanced
Define			C Aspirate at most 0 µL c	er transfer for repeated dispensing.		
Pattern						
			Transfer Details			
			TL1	P7 P11	P15	
			P1	P4 P8 P12	w P16	
				P5 P9 P13	P17	
			P3	P14	TR1	
Mathadta	Binnel/EV	Pierreliev ETC: 0:00:00				
Pietnoa18	BIOMEKPX	BIOTHER AT JETC: 0:00:03				

Figure 18-28. Transfer From File step — Tip Handling collapsed

### 18.9.2 Configuring File Properties

Transfer From File uses a comma-separated value file to configure parts of the transfer operation. The File Properties configuration is needed to specify the file to use and which columns indicate what information.

The file must contain at least information on source wells accessed, but may also specify the position on deck or name of source labware, position on deck or name of destination labware, destination wells accessed, and volume transferred. If all information is specified by the data file, each row is a single well-to-well transfer.

To configure File Properties for the Transfer From File step:

1. If File Properties is not already displayed, click the arrow or textual display to expand the File Properties configuration (Figure 18-29).

🌵 Biomek® Software - Method18* [New]		_ 🗆 🗙
File Edit Project Instrument Execution Options Help		
▋▋▆▆▋▙▟▕▓▝▆▝▋▝▖᠅⊘▕▌		
Start	Use pod Pod2 for transfer. Use probes 1 2 3 4 5 6 7 8	
Serial 🐔 Instrument Setup	$\overline{\mathrm{v}}$ Load AP96_200uL tips, change between transfers, and unload them when finished.	
😚 🎑 Transfer From File	A File Properties	
span-8 Finish	File Name	
Aspirate	Example data (file will be read again when the method is run): 🛛 🧮 File has a header row	
File Properties	Please enter a filename.	
<sup>c</sup> File Properties specifies		
<sup>sp</sup> the file to use for the	File specifies source position in column	
transfer information and	File contains source well information in column	
spar Dis the type of information	✓     File specifies destination position in column	
ine type of information	File contains destination well information in column	
stored in each column.	File contains volume information in column	
Wash	Skip zero volume transfers	
Transfer	Source: <file position="" specified=""></file>	-
From File	Draw Water from < file	<u>•</u>
	C Dispense up to 1 📑 time per draw.	Advanced
Pattern	C Aspirate at most U µL per transfer for repeated dispensing.	
	♥ Transfer Details	
	TL1 P7 P11 P15	
	P1 P4 P8 P12 W P16	
	P2 P5 P9 P13 P17	
	P3 P14 TR1	
Method18* BiomekFX BiomekFX ETC: 0:00:03		

Figure 18-29. Transfer From File — File Properties

2. In **File Name**, enter the full path and file name of the data file from which to read the transfer information.

OR

Choose the browse button next to File Name and select the directory and file from which to read the transfer information in Open (Figure 18-30). The selected file is read and Example data is displayed (Figure 18-31).

**Note:** The data file must be a comma-delimited text file (.txt or .csv).

)pen			? ×
Look jn:	Methods	• Ē	₫ 🖩 🖻
📲 transferfi	le.csv		
	[		Open
File name:			
File <u>n</u> ame:	 		

Figure 18-30. Open dialog



Figure 18-31. Transfer From File — Example data displayed

3. If the file read has a header row identifying the columns, select **File has a header row**. The first row of the data file is used to identify the columns (Figure 18-32).



Figure 18-32. Transfer From File — Example data with header row

4. Select **File specifies source position in column** if a column of the data file stores information about the position of source labware and select the desired column from the list.

**Note:** The source position column may contain either deck positions (P4, P7, etc.) or labware names (Source1, Dest1, etc.). If information is stored as labware names, there must be labware on the deck with the specified labware name (Figure 18-32).

5. In **File contains source well information in column**, select the column of the data file that stores information about source wells from which to transfer from the list.

**Note:** The source well column may specify wells with either alphanumeric well addresses (A1-H12 for a 96-well plate) or numeric well numbers (1-96 on a 96-well plate). The same data file may contain both alphanumeric and numeric well information (Figure 18-32).

6. Select **File specifies destination position in column** if a column of the data file stores information about the position of destination labware and select the desired column from the list.

**Note:** The destination position column may contain either deck positions (P4, P7, etc.) or labware names (Source1, Dest1, etc.). If information is stored as labware names, there must be labware on the deck with the specified labware name (Figure 18-32).

7. Select **File contains destination well information in column** if a column of the data file stores information about destination wells to which to transfer and select the desired column from the list.

**Note:** The destination well column may specify wells with either alphanumeric well addresses (A1-H12 for a 96-well plate) or numeric well numbers (1-96 on a 96-well plate). The same data file may contain both alphanumeric and numeric well information.

8. Select **File contains volume information in column** if a column of the data file stores information about the volume to transfer and select the desired column from the list.

**Note:** All volumes are specified in microliters ( $\mu$ L).

9. Select **Skip zero volume transfers** to ignore any rows that specify a volume of **0** when performing the **Transfer From File** step.

The File Properties configuration can be collapsed to allow more room during Source Labware and Destination Labware configuration. When collapsed, the File Properties configuration is displayed as text.

To collapse or expand the File Properties configuration, click the arrow to the left of File Properties or the textual display.

#### 18.9.3 Configuring Source Labware

A 'source' is a group of wells accessed at one time by the Biomek instrument to aspirate liquid; for example:

- FX, NX-S8— A Span-8 Pod aspirates from one to eight wells at a time, depending on the selected probes.
- 3000 Using a single-tip pipette tool, aspirates from a single well at a time.

The **Transfer From File** step configures multiple aspirate and dispense operations using the selected probes to transfer all source wells to the destination labware.

The Source Labware configuration for the Transfer From File step includes choosing:

- the labware type
- the labware location, if not specified by the file
- the liquid type
- the technique
- the pipetting height

**Note:** Only one piece of **Source Labware** is configured, because the **Transfer** From File step moves liquid from a single source to one or more destinations. However, if the source position is specified by the file, multiple sources are accessed for the transfer, but all sources are configured by a single source configuration. To configure Source Labware for the Transfer From File step:

1. Click on the **Source** configuration area to activate it (Figure 18-33).



Figure 18-33. Transfer From File step — Source Labware selection

2. If the source position is not specified by the file, click on the desired piece of labware in the Current Deck Display (Figure 18-33). Information for that piece of labware is entered automatically into the Source Labware configuration.

OR

If the source position is specified by the data file, select the labware type of the source labware. All sources must use the same labware type.

**Note:** Reservoirs may not be used as **Source Labware**.

**Note:** Once Source Labware has been selected, any subsequent selections become Destination Labware. If an incorrect piece of labware is selected, **right-click on the labware title** and select **Delete** on the menu that appears. If the Source Labware is incorrect, activate the Source Labware configuration by clicking anywhere in the source labware configuration area, and select another piece of labware.

3. Verify the deck position of the labware, if not specified by the data file.

**Note:** A bright blue outline appears in the Current Deck Display around the labware designated as the **Source Labware**.

4. In **Using liquid type**, select the type of liquid in the Source Labware (Figure 18-34).

**Note:** The default liquid type is the liquid type specified as the default in the Liquid Type Editor.



Figure 18-34. Transfer From File step — Source Labware configuration

5. De-select Mark last well that is used, if it is not desired to mark wells. This option must be selected to use any options later in the method that make use of marks. For example, if Mark last well that is used is selected and the Transfer From File step runs out of sources before using all destinations, it will mark the last well that had liquid transferred to it. Another Transfer From File step is transferring liquid to the same plate. By selecting After last marked well as the Start condition, the first source well will be transferred to the first unused destination well.

**Note:** Mark last well that is used can also be selected when the labware is zoomed out by using the **Set mark** button beneath the labware.

Note: Mark last well that is used is selected by default.

 Keep the auto-selected **Technique** (refer to Section 15.3.2.1, <u>Auto-Selection of</u> <u>a Technique</u>).

**Note:** The software automatically selects a **Technique** based on the liquid type, labware type, tip type, and volume being aspirated.

OR

Manually select the desired **Technique**, or customize the selected **Technique** (refer to Section 15.3.2.2, *Customizing and Saving a Technique*).

7. Leave the height as is to use the settings specified in the pipetting technique.

OR

Set the aspirate height manually using one of the following techniques:

Set the aspirate height by positioning the cursor over the graphic of a tip inside a well. Click and drag the graphic up or down. The tip moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.



**Note:** The cursor changes to a hand when positioned over the graphic.

- Position the cursor in the graphic of a tip inside a well, and adjust the height using the ↑ and ↓ keys.
- Right-click on the graphic of a tip inside a well, and a menu appears. Select **Custom Height**, and **Custom Height** appears (Figure 18-35). Insert the **Height** in millimeters (mm) and, in **from**, select a reference point from the drop-down list.

**Note:** The phrase (Overrides Technique) appears below the graphic when the tip height is changed by manipulating the graphic, or when the technique is changed with the Customize option.

Custom Height	
Height:	mm
from Bottom	•
ОК	Cancel

Figure 18-35. Custom Height prompt

### 18.9.4 Configuring Destination Labware

A 'destination' is a group of wells accessed at one time by the Biomek instrument to dispense liquid; for example:

- FX, NX-S8 A Span-8 Pod dispenses to one to eight wells at a time, depending on the selected probes.
- > **3000** Using a single-tip pipette tool, dispenses to a single well at a time.

The Transfer From File step configures multiple aspirate and dispense operations using the selected probes to transfer all source wells to the destination labware when using the Span-8 Pod.

The **Destination Labware** configuration includes choosing:

- the labware type
- the labware location, if not specified in the data file
- the wells on labware used, if not specified in the data file
- the liquid amount to transfer, if not specified in the data file
- the liquid type
- the technique
- the pipetting height

**Note:** Multiple pieces of Destination Labware can be configured for the Transfer From File step, because the Transfer step can move liquid from a single source to one or more destinations. However, if the destination position is specified in the data file, all destinations are configured by a single destination configuration.

To configure Destination Labware for the Transfer From File step:

1. If the destination position is not specified by the data file, click on the desired piece of labware in the Current Deck Display. Information for that piece of labware is entered automatically into the **Destination Labware** configuration.

OR

If the destination position is specified by the data file, select the **Destination Labware** type of the destination labware. All destinations must use the same labware type.

**Note:** Once the first **Destination Labware** has been selected, the **Source** Labware parameters are collapsed to display a sentence summary. To reopen the Source labware parameters, click anywhere in the Source Labware configuration area.

**Note:** Subsequent labware selections modify the active Source or Destination Labware configuration. To configure additional Destination Labware, select **Click here to add a destination**, or select outside a labware configuration, and then select another piece of labware from the Current Deck Display. If an incorrect piece of labware is selected, right-click on the labware title and select **Delete** on the menu that appears.

If either the destination position or destination wells are specified by the data file, additional destinations cannot be configured.



Figure 18-36. Transfer From File step — Destination Labware selection

2. Verify the deck position of the labware, if not specified by the data file.

**Note:** A bright yellow outline appears in the Current Deck Display around the labware designated as the **Destination Labware**.



Figure 18-37. Transfer From File step — Destination Labware configuration

- Enter the amount of liquid (μL) dispensed into the Destination Labware, if not specified by the data file. If using a variable or expression, enter an equal sign (=) followed by the variable name or expression; for example, =AspValue (Figure 18-37).
- Verify that the liquid type indicated is correct. The liquid type dispensed to the Destination Labware should match the liquid type in the Source Labware configuration.
- 5. Deselect Mark last well that is used, if it is not desired to mark wells. This option must be selected to use any options later in the method that make use of marks. For example, if Mark last well that is used is selected and the Transfer From File step runs out of sources before using all destinations, it will mark the last well that had liquid transferred to it. Another Transfer From File step is transferring liquid to the same plate. By selecting After last marked well as the Start condition, the first source well will be transferred to the first unused destination well. Mark last well that is used is selected by default.



**Note:** Mark last well that is used can also be selected when the labware is zoomed out by using the Set mark button beneath the labware.

 Keep the auto-selected **Technique** (refer to Section 15.3.2.1, <u>Auto-Selection of</u> <u>a Technique</u>).

**Note:** The software automatically selects a **Technique** based on the liquid type, labware type, tip type, and volume being aspirated.

OR

Manually select the desired **Technique**, or customize the selected **Technique** (refer to Section 15.3.2.2, *Customizing and Saving a Technique*).

7. Leave the height as is to use the settings specified in the pipetting technique.

OR

Set the dispense height manually using one of the following techniques:

Set the dispense height by positioning the cursor over the graphic of a tip inside a well. Click and drag the graphic up or down. The tip moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.

Ş

**Note:** The cursor changes to a hand when positioned over the graphic.

- Position the cursor in the graphic of a tip inside a well, and adjust the height using the ↑ and ↓ keys.
- Right-click on the graphic of a tip inside a well, and a menu appears. Select **Custom Height**, and **Custom Height** appears (Figure 18-35). Insert the **Height** in millimeters (mm) and, in **from**, select a reference point from the drop-down list.

**Note:** The phrase (Overrides Technique) appears below the graphic when the tip height is changed by manipulating the graphic, or when the technique is changed with the Customize option.

Custom Height	
Height:	mm
from Bottom	•
ОК	Cancel

Figure 18-38. Custom Height prompt

- 8. If destination wells are not specified by the data file, go to Section 18.9.4.1, *Configuring Destination Wells*.
- 9. Repeat steps 1-8 for each additional piece of **Destination Labware** if neither destination positions nor destination wells are specified by the data file.
## 18.9.4.1 Configuring Destination Wells

If destination wells were not specified by the data file, further configuration of the destination labware is required.

To complete the Destination Labware configuration:

1. Double-click the **Destination Labware** graphic in the step configuration to zoom in on the labware.

**Note:** To call up a menu for Zoom In and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows the selection of wells in a microplate as targets for the transfer operation as comma-delimited text. Wells are numbered left to right and down (for example, the first row is wells 1-12, the second row wells 13-24, etc., for a 96-well plate). Specify Selection as Text may also be used to enter variables or expressions.

- 2. Select the wells that are destinations for the transfer using one of the following techniques:
  - ► FX, NX-S8 Can select individual wells.
  - > **3000** A single-tip pipette tool can select individual wells.
    - Choose Copy Pattern and select the desired pattern to use a previously defined well pattern created using the Well Patterns Editor (refer to Chapter 11, <u>Creating Well Patterns</u>).

**Note:** The pattern must be compatible with the head or tool installed.

• Select **Use pattern** and choose a previously defined pattern from the list to use a pattern created in the Well Patterns Editor or a Define Pattern step.

**Note:** Refer to Section 18.10, *Define Pattern Step (including 3000)*, for more information about creating patterns using the **Define Pattern** step.

- Define a well pattern using data sets (refer to Section 14.3.1, <u>Defining</u> <u>Well Patterns Using Data Sets</u>).
- Create a custom well pattern from the labware grid by dragging the mouse and using Ctrl and Shift on the keyboard.

**Note:** Hold **Shift** and click a well to select additional wells without deselecting any wells. Hold **Ctrl** and click a well to toggle the selection status of selected wells without affecting the status of other wells. If Shift or Ctrl is not held down when selecting wells, any previous selection is deleted.

**Note:** To save the pattern, right click the labware graphic and select **Save Pattern** from the menu. Enter a name for the pattern in New Pattern and choose **OK**. Select the **Direction** in which destination wells are mapped to sources. Down first, then left to right is the default selection.

**Note:** Direction controls how source wells are mapped to the destination, not the physical order in which wells are transferred. The **Transfer From File** step automatically determines the most efficient way to complete the transfer.



 Down first, then left to right — goes down each column from top to bottom, then goes right to the next column (Figure 18-39)



Figure 18-39. Wells accessed down first, then left to right

#### OR

To the right first, then top to bottom — goes across each row from left to right, then goes down to the next row (Figure 18-40)



Figure 18-40. Wells accessed to the right first, then top to bottom

**Note:** The direction can also be selected when the labware is zoomed out by using the Down, then right or Right, then down buttons beneath the labware.

≯⁄↓

4. Select the first well accessed in **Start**. At first selected well is the default selection.

**Note:** Start controls how source wells are mapped to the destination, not the physical order in which wells are transferred. The **Transfer From File** step automatically determines the most efficient way to complete the transfer.



• At first selected well — moves in the selected Direction starting from the first well selected on the labware

OR



• After last marked well — moves in the selected Direction starting from the first selected well on the labware after the marked well

**Note:** The selected labware must have marks from a previous step with Mark last well that is used selected.

**Note:** The start location can also be selected when the labware is zoomed out by using the Start at selection or Start at last mark buttons beneath the labware.

**Note:** When transferring liquid, the source well in the first row of the data file is transferred to the first accessed destination well, the second source well is transferred to the second accessed destination well, and so forth.

5. Choose **Zoom Out** to return to the step configuration screen.

## 18.9.5 Configuring Transfer Details

Transfer Details includes specifying:

- a repeat pipetting configuration (Figure 18-41).
- the maximum tolerance in timing between transfer operations (Figure 18-42).



Figure 18-41. Transfer From File step - configuring additional Transfer Details

To configure Transfer Details for the Transfer From File step:

- 1. Click the arrow or sentence summary below the labware configuration section to expand Transfer Details.
- 2. Select one of the repeat pipetting configuration options: Dispense up to or Aspirate at most.

**Note:** Only one repeat pipetting option may be selected. Either Dispense up to or Aspirate at most may be selected, but not both.

3. In **Dispense up to**, select the number of times the Biomek instrument is allowed to dispense per aspirate. For example, if dispensing 75  $\mu$ L to each of six microplates, three aspirate operations of 150  $\mu$ L can each dispense into two destinations. In this case, **Dispense up to** is set to no fewer than two.

OR

In **Aspirate at most**, select the maximum volume to aspirate from the source for repeat pipetting. For example, if dispensing 75  $\mu$ L to each of six microplates, three aspirate operations of 150  $\mu$ L can each dispense into two destinations. In this case, Aspirate at most is set to no less than 150  $\mu$ L.

**Note:** The default repeat pipetting setting is **Dispense** up to one time per draw, which does not allow a repeat pipetting operation. Settings of two or more allow repeat pipetting.

4. Click **Advanced** to change the default setting of applying a Just-In-Time (JIT) block to aspirate and dispense operations. The Advanced Options prompt appears (Figure 18-42).

Advanced Options				
🔽 🛛 ie Aspirates to Dispe	enses with a Just-In-Time step.			
ОК	Cancel			

Figure 18-42. Advanced button — JIT Block

- 5. Uncheck **Tie Aspirate to Dispense with a JIT Block** to deactivate the JIT block.
- 6. Click **OK** to save any changes.

The Transfer Details configuration can be collapsed to allow more room during Source and Destination Labware configuration. When collapsed, the Transfer Details configuration is displayed as a sentence summary. Use the sentence summary to verify that Transfer Details has been configured appropriately.

To collapse or expand the **Transfer Details** configuration, click the arrow to the left of **Transfer Details** or the textual display (Figure 18-41).

# 18.10 Define Pattern Step (including 3000)

Define Pattern is used to create patterns and assign names to them for use in other steps in the method, such as a Transfer or Combine step. Well patterns defined using the Define Pattern step are method specific and embedded as part of the method.

Embedding the well patterns as part of the method has several advantages:

- Sharing methods among other users is simpler because the pattern information is stored in the method, not in the project.
- Common well patterns can be modified for individual methods without changing the well pattern in the project.
- The pattern definition can be changed while keeping the same pattern name, making it simple to modify methods by changing the well pattern definition. Since the **Transfer** and **Combine** steps use the pattern name, changes to the pattern are automatically updated in the **Transfer** and **Combine** steps.

**Note:** This is particularly useful when the pattern is defined by a file using rows that match a bar code or other value. In a Loop step, the pattern could change based on the bar code of the current labware in a group created with a **Create Group** step. Each iteration of the Loop would transfer using a different pattern as defined by the file in the **Define Pattern** step.

Insert a **Define Pattern** step into the method. The **Define Pattern** Step Configuration appears (Figure 18-43).



Figure 18-43. Define Pattern step and configuration

Define Pattern has two different configuration options. It can define a pattern by manually specifying wells or reading from a file.

## 18.10.1 Configuring the Define Pattern Step by Specifying the Wells Manually

To configure the Define Pattern step by manually specifying the wells:

- 1. In Create a pattern named, enter a name for the pattern.
- 2. In Model it after the labware type, select the labware type for the pattern.
- 3. Select Specify the wells manually.
- 4. On the graphic representing the labware, select the wells to be included in the pattern. All wells are selected by default (Figure 18-43).

**Note:** Click on a well to select it and deselect the current selection. Hold down **Shift** and click on additional wells to select wells without affecting the current selection. Hold down **Ctrl** and click on wells to toggle wells between selected and deselected without affecting the current selection.

5. To use or modify a well pattern defined in the Well Patterns Editor, select **Copy Pattern** and choose the pattern from the list.

## 18.10.2 Configuring the Define Pattern Step Reading from a File

To configure the Define Pattern step by reading well information from a file:

- 1. In Create a pattern named, enter a name for the pattern.
- 2. In Model it after the labware type, select the labware type for the pattern.
- 3. Select **Read from a file**. The Read from a file configuration options appear (Figure 18-44).

Biomek® Software - define pattern* [Development] File Edit Project Instrument Execution Ontions Help		_ 🗆 ×
		File selection
Start	Create a pattern named	Used to select the file to
serial 🦓 Instrument Setup	Model it after the labware type BCFlat96	read to create the pattern.
😚 🔊 Define Pattern	C Specify the wells manually	
span-8	Read from a file	<b>n</b> .
Aspirate		
Span-8	Example data (ne will be read again when the method is fun).	
Dispense		
Span-8		
New Tips		
Span-8 Tip	C Use all rows	
Discard T T		
LL Span-8	Wells are specified by Alphanumerics (e.g. A1-H12) in column #2	
Wash Tips		
Transfer		
From File		
Define		
Pattern		
	TL1 P4 P12 P16	
	P2 P6 P10 P14 P18	
	P3 P7 P11 P15 TR1	
(I) <b>F</b>		
define pattern* BiomekFX BiomekFX ETC: 0:00:03		

Figure 18-44. Define Pattern step — Read from a file configuration

4. Enter the full path and file name of the data file from which to read the pattern information.

OR

...

Choose the browse button and select the directory and file from which to read the pattern information in Open (Figure 18-45). The selected file is read and Example data is displayed (Figure 18-46).

**Note:** The data file must be a comma-delimited text file (.txt or .csv).

Open						?	x
Look jn:	🔄 Methods		-	£	Ċ		
📲 definepatt	ern.csv						٦
🛐 example.c	sv						
🗒 My Workli	st.txt						
📲 transferfile	.CSV						
, Els	1				_		1
File <u>n</u> ame:	ļ					<u>U</u> pen	L
Files of type:	Comma Delimited T	ext File (*.txt; *.cs	sv)	-		Cancel	1

Figure 18-45. Open dialog

Biomek® Software - define pattern* [Development] File Edit Project Instrument Execution Options Help		
Start	Create a pattern named	
serial 😴 Instrument Setup	Model it after the labware type BCFlat96	
Pilution S Define Pattern	C Specify the wells manually	
	Read from a file	
Aspirate Example Data	C:\Program Files\Biomek Software\definepattern.csv	
The selected file	Example data (hie will be read again when the method is run):	
Dispense is read and data		
displayed in the	BC02 74 BC03 81	
Example data	RM4 6	
Fell section.	C Use all rows	
Discard	Use rows where column Barcode matches • the Barcode matches from matches	
<u>unu</u>		
Span-8 Wash Tips	Weils are specified by [Alphanumerics (e.g., Al+H12) 🔽 in column   Weil	
Transfer From File		
Define Pattern		
	P12 P16	
	P2 P6 P14 P18	
		2000
		=
define pattern*  BiomekFX  BiomekFX  ETC: 0:00:03		

Figure 18-46. Define Pattern step - Example data displayed

5. If the file read has a header row identifying the columns, select **File has a header row**. The first row of the data file is used to identify the columns (Figure 18-47).

🌵 Biomek® Software - define pattern* [Development		_ 🗆 ×
File Edit Project Instrument Execution Options Help		
	File has a header row	
Start	Create a pattern named Uses the first row of the data	
Instrument Setur	Model it after the labware type BCFlat96	
Dilution	C Specify the wells manually	
Define Pattern	Geod from a file	
Span-8 Finish	P\Program Files\Binmek Software\definenation csy	
Aspirate -		
Span-8 Dispense		
44	BC02 74	
Span-8 New Tins	8C03 81	
R.		
Span-8 Tip		
Discard D D		
ப்ப	C the value	
Span-8 Wash Tips	Wells are specified by Alphanumerics (e.g., A1·H12) 💌 in column   Well	
	<u> </u>	
Transfer		
From File	Specify the lines of	
	the date Classes	
Define Pattern	the data file to use.	
		TTTTTNNN)
	Dest1 Dest1	
	P2 P6 P14 P18	
	P7 P11 P15 TR1	
		<u> </u>
define pattern*  BiomekFX  BiomekFX  ETC: 0:00:03		

Figure 18-47. Define Pattern step — Example data with header row

- 6. Beneath the Example data, select which lines from the data file to use when defining the pattern.
  - To use all rows of the data file, select Use all lines.
  - To use specific wells where a certain column matches a specified bar code or value, refer to Section 18.10.2.1, <u>Specifying Lines to Use</u>.
- 7. In Wells are specified by, select whether wells are identified using alphanumeric well addresses (for example, A1-H12 for a 96-well plate) or numeric well numbers (for example, 1-96 for a 96-well plate), and specify the column that contains the well information.
- 8. Choose a step that occurs after the Define Pattern step or the Finish step to validate the step configuration.

#### 18.10.2.1 Specifying Lines to Use

Sometimes it may be desired to only use specific rows from the data file. For example, if a data file contains information on several plates, the file can be configured to use only the rows pertaining to the current plate using the bar code, deck position, or other factor.

To use selected lines from the data file:

- 1. Select **Use lines where column** to activate the configuration.
- 2. In **column matches**, select the column from the data file that must match the specified value to use the row.
- 3. Select whether the selected column must match the **Bar code** or **Name** of the labware selected in **from** and specify a deck position or labware name to match against in **from**.

OR

Select **the value** and enter a variable or expression that must match the column selected.

The rows that match the specified condition are used to define the well pattern using the column that specifies the well information.

## 18.10.3 Using a Defined Pattern in a Transfer or Combine Step

Once a pattern has been created using a Define Pattern step, it may be used in any Transfer or Combine step within the method that uses a Span-8 Pod on the *FX* or *NX-S8* or a pipetting tool on the *3000*.

To use a defined pattern in a Transfer or Combine step:

- 1. Select the desired **Transfer** or **Combine** step in the Method View. The step configuration appears.
- 2. Select the desired **Source** or **Destination** configuration to display the configuration for the selected labware.
- 3. Double-click the graphic of the labware to zoom in on it.
- 4. Select **Use pattern** to use a pattern created in a **Define Pattern** step.

5. Choose the desired pattern created in a Define Pattern step from the list (Figure 18-48). During method run, the configured step uses the specified pattern.



Figure 18-48. Transfer step using Sample pattern

**Note:** Only patterns that are defined in the current method or from a previously run method that did not clear global variables appear in the list.

6. Choose a step that occurs after the configured **Transfer** or **Combine** step or the Finish step to validate the step configuration.

# Using the HDR Step Palette (FX, 3000 only)

# 19.1 Overview



CAUTION: Do not attempt to access a 96-Channel or 384-Channel Tip Wash ALP with a Multichannel Pod equipped with an HDR Tool Body. The gripper may crash and damage the pod, HDR Tool Body, or Tip Wash ALP.

The HDR Step Palette provides access to steps used to control the HDR Tool Body on the Biomek FX instrument or the HDR tool on a Biomek 3000 instrument. The HDR steps include the means to transfer liquid, wash pins, and move labware using the HDR Tool Body.

The steps available in HDR Step Palette include:



• HDR Transfer — transfers liquid from a single source to one or more destinations using the HDR Tool Body



• HDR Combine — transfers liquid from a one or more sources to a single destination using the HDR Tool Body



 HDR Tool Cleaning — cleans pins on the HDR Tool Body by dipping the pins in one or more reservoirs and drying them at the HDR Pin Drying ALP



• HDR Move Labware — moves labware using the gripper on the Multichannel Pod with the HDR Tool Body installed.

> **3000** — HDR Move Labware is not available.

**Note:** In addition to the HDR steps, the **Device Action** step may be used to control the wash pumps for the circulating reservoirs or the fan for the HDR Pin Drying ALP independently during a method (refer to Section 22.6, *Device Action Step*).

# **19.2 Displaying the HDR Step Palette**

In order to add HDR steps to a method, display the HDR Step Palette (Figure 19-1).

📫 Biomek® Software	
File Edit Project Instrument Execution Options Help	
HDR Tool HDR	
TL1     P4     P8     P12     P16       P1     P5     P9     P13     P17       P2     P6     P10     P14     P18       P3     P7     P11     P15     P19	
BiomekFX  BiomekFX	

Figure 19-1. Biomek main editor with HDR Step Palette displayed

To display the HDR Step Palette, complete the following:

• Right-click any empty palette space, and the Step Palette menu appears. Select **HDR**.

OR

• From the menu bar, select **Options>Toolbars>HDR**.

## 19.3 Using the HDR Transfer and Combine Steps

The HDR Transfer and HDR Combine steps transfer a volume of sample from each well of a source labware to each well of the destination labware. Each destination well may contain up to 36 samples per well in a 6 x 6 array for a 96-well microplate or up to 16 samples per well in a 4 x 4 array for a 384-well microplate.

**Note:** Only a 384-pin plate with 0.015" post pins can access quadrants of a 1536-well microplate.

The HDR Transfer step transfers liquid from a single source to one or more destinations, while the HDR Combine step transfers liquid from a one or more sources to a single destination.

A liquid transfer operation deposits samples in the same array position of each destination well, starting in the upper left corner position. Each subsequent liquid transfer operation deposits samples in the next available array position, going across each row before moving down to the next row of the array (Figure 19-2).



Figure 19-2. Order of transfer to array positions in destination wells with a 3x3 array

**Note:** Destination labware should be placed on a Positive Position ALP during gridding applications. Refer to the *ALPs User's Manual*, Chapter 15, *Positive Position ALP*, for more information on the Positive Position ALP.

The HDR Transfer and HDR Combine steps perform liquid-handling functions through the step configuration (Figure 19-3). An HDR Transfer step transfers sources such that each destination receives exactly one source; an HDR Combine step transfers each source to exactly one destination.

- If there are the same number of sources and destinations (wells x array positions), all sources are transferred and each destination receives exactly one source.
- If there are more sources than destinations (wells x array positions), an HDR Transfer step transfers only enough sources so that each destination receives exactly one source (some sources are unused); an HDR Combine step transfers all sources to exactly one destination (some destinations receive more than one source).
- If there are fewer sources than destinations (wells x array positions), an HDR Transfer step transfers all sources until each destinations receives exactly one source (some sources are used multiple times); an HDR Combine step transfers all sources to exactly one destination (not all destinations receive a source).

**Note:** The HDR Transfer and HDR Combine steps must be used with either a Multichannel Pod equipped with the HDR Tool Body on the Biomek FX instrument, or an HDR tool installed on the head assembly of a Biomek 3000 instrument.

The HDR Transfer or HDR Combine step configuration includes specifying:

- Pod that performs the transfer.
  - **FX only** The pod that performs the transfer is specified.
- Pin washing procedure between transfers, if desired.

**Note:** Pin washing between transfers requires that a procedure is defined using the Define Procedure step (refer to Section 19.3.1, <u>Washing Pins</u> <u>Between Transfers</u>).

- Source labware configuration(s).
- Destination labware configuration(s).

To configure an HDR Transfer step:



1. Insert an HDR Transfer or HDR Combine step into the Method View (Figure 19-3).

約 Biomek® Software - Method2* [New]	
File Edit Project Instrument Execution Options He	p
Start	Use pod Pod1 ror transfer. Pod selection
HDR Instrument Setup	
HDR Transfer	Wash between transfers with protocol
HDR Move Labware Transfer	Click here to add a source.
HDR Tool Cleaning Combine	
HDR Move Transfer Labware	Click this button to add Source
	Labware configuration.
Pause	
Comment	
	P2 P2 P14 P18 P1 P18 P1 P1 P15 P19
Method2* BiomekFX BiomekFX ETC: 0:00:10	

Figure 19-3. HDR Transfer step configuration

- 2. In **Use pod**, specify the pod with the HDR Tool Body used to perform the transfer operations.
  - > **3000** Use pod does not appear in the step configuration.
- 3. Select **Wash between transfers with protocol** to wash pins between transfers using a specified procedure.

**Note:** Pin washing between transfers requires that a procedure is defined using the Define Procedure step (refer to Section 19.3.1, <u>Washing Pins Between</u> <u>Transfers</u>).

- 4. In Wash between transfers with protocol, select a previously defined procedure to use to wash pins between transfers.
- 5. Configure Source and Destination labware as described in Sections 19.3.2, <u>Configuring Source Labware</u>, and 19.3.3, <u>Configuring Destination Labware</u>.

## 19.3.1 Washing Pins Between Transfers

Pins are washed between transfers in an HDR Transfer or HDR Combine step using a wash procedure defined in a Define Procedure step (refer to Section 21.8.1, <u>Define</u> <u>Procedure Step</u>). After defining the procedure, it is selected in the HDR Transfer or HDR Combine step configuration.

To wash pins between transfers in an HDR Transfer or HDR Combine step:

1. Insert a **Define Procedure** step in the Method View. The **Define Procedure** Step Configuration appears (Figure 19-4).

**Note:** Define Procedure is not on any of the default step palettes and must be added to a step palette from the Specialty step category in the Palette Builder. Refer to Section 29.5, *Using the Step Palette Builder*, for more information on adding Define Procedure to a step palette.



Figure 19-4. Define Procedure step and configuration

2. In **Procedure**, enter a name for the procedure.

**Note:** The procedure name may contain only alphanumeric characters and the underscore (\_), and is not case-sensitive.



- 3. Double-click the **Define Procedure** icon in the Method View to open the procedure. The End Procedure icon appears below the **Define Procedure** icon.
- 4. Insert the **HDR Tool Cleaning** step between the Define Procedure and End Procedure icons.

- 5. Configure the **HDR Tool Cleaning** step as desired to perform all pin washing between transfers (refer to Section 19.4, *Using the HDR Tool Cleaning Step*).
- 6. If desired, insert and configure additional steps between the Define Procedure and End Procedure icons.

**Note:** All steps included in this procedure are performed between every transfer of an HDR Transfer or HDR Combine step.

 Double-click the **Define Procedure** icon in the Method View to collapse the procedure and hide all steps between the Define Procedure and End Procedure icons.

**Note:** After defining the procedure and configuring all steps within the procedure as desired, drag it onto one of the step palettes to create a custom step. When writing other methods, insert the custom **Define Procedure** step into the Method View to use the same pin washing procedure and ensure consistent pin washing across methods.

8. When configuring an HDR Transfer or HDR Combine step, select the procedure created above in Wash between transfers with protocol.

## 19.3.2 Configuring Source Labware

A 'source' is a group of wells accessed at one time by the HDR Tool Body; for example, a 96-pin plate accesses 96 wells at a time while a 384-pin plate accesses 384 wells at a time.

**Note:** If configuring a 384-well microplate for access by a 96-pin plate or a 1536well microplate for access by a 384-pin plate as source labware, any combination of quadrants may be used. For example, if it is desired to transfer liquid from the first, second, and third quadrants but not the fourth, the source labware may be configured to transfer only from those quadrants.

**Note:** Only a 384-pin plate with 0.015" post pins can access quadrants of a 1536-well microplate.

In the HDR Transfer and HDR Combine steps, the HDR Tool Body dips the pins in the source labware at a specified height for a specified length of time a specified number of times to pick up samples by liquid adhesion.

**Note:** The volume of sample that adheres to the pins is dependent on many factors, including environmental factors, the liquid type, the surface area of the pins immersed in the sample, the length of time the pins are in contact with the sample, pin washing technique, and dryness of the pins. Conduct tests to determine the optimal settings to transfer the desired volume of a specific liquid type with a specific pin size.

The Source Labware configuration for the Transfer step includes choosing:

- Labware type
- Labware location
- Dipping configuration
- Quadrants used, if applicable

**Note:** Quadrants are applicable when accessing a 384-well microplate using a 96-pin plate or a 1536-well microplate using a 384-pin plate. Only a 384-pin plate with 0.015" post pins can access quadrants of a 1536-well microplate.

**Note:** A single source is configured for the HDR Transfer step, because the HDR Transfer step moves liquid from a single source to one or more destinations. Multiple sources can be configured for the HDR Combine step, because the HDR Combine step moves liquid from multiple sources to a single destination.

To configure Source Labware for the HDR Transfer and HDR Combine steps:



CAUTION: Do not access labware positioned on a 1 x 5 Passive ALP with the HDR Tool Body. The gripper may crash into the ALP.



CAUTION: Do not access labware on a Stirring ALP with the HDR Tool Body. The magnetic stirrer may bend the pins or interfere with the liquid transfer performance of the pins.

1. Select **Click here to add a source** (Figure 19-3). The Source Labware configuration appears (Figure 19-5).

🌵 Biomek	® Software	- HDR T	[ransfer* [[	Developme	ent]		
File Edit	Project Inst	rument	Execution	Options H	lelp		
] 🗅 📽 🕯	i B 🖪		K 🖻 💼	K) CH (	0		
45		8	Start			Use pod Pod1 ror transfer.	
HDR Combine	Instrument Setup		Instrumer	nt Setup			
	<u>a</u>	<b>S</b>	HDR Tran	nsfer		Wash between transfers with protocol	
HDR Move Labware	Transfer	000	-inish			Source:	-
	4						
HDR Tool Cleaning	Combine						
						• •	
HDR Transfer	Move Labware					Dipping Configuration	
	Pause					Click here to add a destination.	
	Q						
	Comment						
m						P1     m     pource     P13     P17       P2     PMSO     Dest     P14     P18       P3     Eth     P11     P15     P19       P4     Bleach     P12     P16     P20	
	<b>C</b>		Levi leve a				
HDR Trans	rer* Biomek	FX Biom	NEKFX JETC: I	0:00:10			

Figure 19-5. HDR Transfer step — Source Labware configuration

2. Click on the desired piece of labware in the Current Deck Display. The labware type and position for that piece of labware is entered automatically into the Source Labware configuration (Figure 19-6).

OR

Select a **Source Labware** type and specify the deck position.

**Note:** To configure additional Source Labware for an HDR Combine step, select **Click here to add a source**, and then select another piece of labware from the Current Deck Display. If an incorrect piece of labware is selected, right-click on the labware title and select **Delete** on the menu that appears, or change the Source Labware selection by clicking anywhere in the source labware configuration area, and selecting another piece of labware.





3. Verify the deck position of the labware.

**Note:** A bright blue outline appears in the Current Deck Display around the labware designated as the **Source Labware**.

4. To specify quadrants in a microplate to access, double-click the source labware in the step configuration to zoom in on the labware.

**Note:** To call up a menu for Zoom In and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows selection of quadrants as targets for aspirate and dispense operations. Specify Selection as Text may also be used to enter variables or expressions (refer to Section 13.2, *Using Variables* and Section 13.3, *Using Expressions*).

**Note:** Specify Selection as Text is not applicable for 96-well microplates, reservoirs, or 384-well microplates accessed by a 384-channel pod.

5. Select the desired quadrant(s) on the zoomed-in graphic of the labware.

Note: Selecting any well automatically selects all wells in that quadrant.

**Note:** To select multiple quadrants, click and drag over the desired quadrants or hold down **Ctrl** or **Shift** and select the desired quadrants.

Zoom Out

- 6. Choose **Zoom Out** to return to the step configuration screen.
- 7. Choose **Dipping Configuration** to configure the dipping operation. Dipping Configuration appears (Figure 19-7).

**Note:** The Dipping Configuration is similar to a pipetting technique. It specifies the method in which liquid is picked up by the pins.

Dipping Configuration	
Dip Cycles : 🚺	
Move down to	Move up to
0.00 mm from bottom	-1.00 mm from top
Move down at	Move up at
50 % speed	50 % speed
Pause for 0 seco	nds after dipping
ОК	Cancel

Figure 19-7. Dipping Configuration

8. In Dip Cycles, enter the number of times the pins should descend into the wells to pick up liquid.

**Note:** For Dip Cycles greater than **1**, Move up to must be configured.

9. In Move down to, specify the height the pins descend to for each dipping cycle.

**Note:** By default, Move down to is measured from bottom of the well. To measure from top of the well or from liquid, right click the graphic of the pin in the well and make the desired selection from the menu.

**Note:** The Move down to height can go as much as 2 mm below the bottom of the well as long as the pod can descend that far. Descending below the bottom of the well potentially corrects for any slight imperfections in the labware or ALP heights and can be done safely because the pin push plate is in a floating position, allowing the pins to move up and down freely without resistance. To specify a height below the bottom of the well, right-click the graphic of the pin in the well and select Custom Height. Enter a negative value in Custom Height to move the pins below the bottom of the well.

- 10. In Move down at, enter the speed of the descent of the HDR Tool Body as a percentage of the maximum speed of the pod.
- 11. In Move up to, specify the height the pins ascend to between each dipping cycle.

**Note:** If Dip Cycles is set to **1**, Move up to is disabled.

**Note:** By default, Move up to is measured from bottom of the well. To measure from top of the well or from liquid, right click the graphic of the pin in the well and make the desired selection from the menu.

- 12. In Move up at, enter the speed of the ascent of the HDR Tool Body as a percentage of the maximum speed of the pod.
- 13. In Pause for, enter the length of time the pod pauses at the down position for each Dip Cycle to pick up liquid.
- 14. Choose **OK** to save the Dipping Configuration and return to the source labware configuration.

15. Leave the pin height as is to use the settings for the Move down to height specified in the Dipping Configuration.

#### OR

Set the pin height manually using one of the following methods:

- Position the cursor over the graphic of a pin inside a well. Click and drag the graphic up or down. The pin moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.
- Select the graphic of a pin inside a well, and use the ↑ and ↓ keys. The textual representation of the height, which is displayed below the graphic, adjusts in 0.1 millimeter (mm) increments.
- Right-click on the graphic of a pin inside a well, and a menu appears. Select Custom Height, and Custom Height appears (Figure 19-8). Insert the Height in millimeters (mm) and, in from, select a reference point from the drop-down list. Choose OK.



**Note:** The cursor changes to a hand when positioned over the graphic.

Custom Height	
Height:	mm
from Bottom	•
OK	Cancel

Figure 19-8. Custom Height prompt

16. Repeat steps 1 to 15 for each additional piece of Source Labware.

**Note:** A Destination Labware must be selected before additional Source Labware can be configured.

## **19.3.3 Configuring Destination Labware**

A 'destination' is a group of wells accessed at one time by the HDR Tool Body; for example, a 96-pin plate accesses 96 wells at a time while a 384-pin plate accesses 384 wells at a time.

Destination wells may be configured to hold an array of destination sites, allowing for a greater number of samples on a single microplate. All destination wells must have the same array configuration.

**Note:** If configuring a 384-well microplate using a 96-pin plate or a 1536-well microplate using a 384-pin plate as source labware, any combination of quadrants may be used. For example, if it is desired to transfer liquid to the first, second, and third quadrants but not the fourth, the destination labware may be configured to transfer only to those quadrants.

➤ 3000 — 1536-well plates are not used.

**Note:** Only a 384-pin plate with 0.015" post pins can access quadrants of a 1536-well microplate.

In the HDR Transfer and HDR Combine steps, the HDR Tool Body dips the pins in the destination labware at a specified height for a specified length of time a specified number of times to deposit samples by liquid adhesion. Multiple samples can be deposited in a single destination well in an array format.

**Note:** The volume of sample that deposits from the pins to the destination labware is dependent on many factors, including environmental factors, the liquid type, the surface area of the pins in contact with the labware, and the length of time the pins are in contact with the labware. Conduct tests to determine the optimal settings to transfer the desired volume of a specific liquid type with a specific pin size.

The Destination Labware configuration includes choosing:

- Labware type
- Labware location
- Dipping configuration
- Array configuration for gridding

**Note:** Destination labware should be placed on a Positive Position ALP during gridding applications. Refer to the *ALPs User's Manual*, Chapter 15, *Positive Position ALP*, for more information on the Positive Position ALP.

Quadrants used, if applicable

**Note:** Quadrants are applicable when accessing a 384-well microplate using a 96-pin plate or a 1536-well microplate using a 384-pin plate. Only a 384-pin plate with 0.015" post pins can access quadrants of a 1536-well microplate.

▶ **3000** — 1536-well plates are not used.

**Note:** Multiple destinations can be configured for the HDR Transfer step, because the HDR Transfer step moves liquid from a single source to one or more destinations. A single destination is configured for the HDR Combine step, because the HDR Combine step moves liquid from one or more sources to a single destination.

To configure Destination Labware for the HDR Transfer step:



CAUTION: Do not access labware positioned on a 1 x 5 Passive ALP with the HDR Tool Body. The gripper may crash with the ALP.

CAUTION: Do not access labware on a Stirring ALP with the HDR Tool Body. The magnetic stirrer may bend the pins or interfere with the liquid transfer performance of the pins.

1. Select **Click here to add a destination** (Figure 19-9). The Destination Labware configuration appears.



Figure 19-9. HDR Transfer step — Adding Destination Labware

2. Click on the desired piece of labware in the Current Deck Display. The labware type and position for that piece of labware is entered automatically into the **Destination Labware** configuration (Figure 19-6).

OR

Select a **Destination Labware** type and specify the deck position.

**Note:** Once the **Destination Labware** has been selected, the **Source Labware** parameters are collapsed to display a sentence summary. To reopen the **Source** labware parameters, click anywhere in the **Source Labware** configuration area.

**Note:** To configure additional Destination Labware for an HDR Transfer step, select **Click here to add a destination**, and then select another piece of labware from the Current Deck Display. If an incorrect piece of labware is selected, right-click on the labware title and select **Delete** on the menu that appears, or change the **Destination Labware** selection by clicking anywhere in the destination labware configuration area, and selecting another piece of labware.





3. Verify the deck position of the labware.

**Note:** A bright yellow outline appears in the Current Deck Display around labware designated as **Destination Labware**.

4. To specify the quadrants of a microplate to access, double-click the destination labware in the step configuration to zoom in on the labware.

**Note:** To call up a menu for Zoom and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows selection of quadrants as targets for aspirate and dispense operations. Specify Selection as Text may also be used to enter variables or expressions (refer to Section 13.2, *Using Variables* and Section 13.3, *Using Expressions*).

**Note:** Specify Selection as Text is not applicable for 96-well microplates, reservoirs, or 384-well microplates accessed by a 384-pin plate.

5. Select the desired quadrant(s) on the zoomed-in graphic of the labware.

Note: Selecting any well automatically selects all wells in that quadrant.

**Note:** To select multiple quadrants, click and drag over the desired quadrants or hold down **Ctrl** or **Shift** and select the desired quadrants.

- 6. Choose **Zoom Out** to return to the step configuration screen.
- 7. Choose **Dipping Configuration** to configure the operation. Dipping Configuration appears (Figure 19-11).

**Note:** The Dipping Configuration is similar to a pipetting technique. It specifies the method in which liquid is deposited by the pins.

Dipping Configuration	
Dip Cycles : 🚺	
Move down to	Move up to
0.00 mm from bottom	-1.00 mm from top
Move down at	Move up at
50 % speed	50 % speed
Pause for 0 s	econds after dipping
OK	Cancel

Figure 19-11. Dipping Configuration

8. In Dip Cycles, enter the number of times the pins should descend into the wells to deposit liquid.

**Note:** For Dip Cycles greater than **1**, Move up to must be configured.

9. In Move down to, specify the height the pins descend to for each dipping cycle.

**Note:** By default, Move down to is measured from bottom of the well. To measure from top of the well or from liquid, right click the graphic of the pin in the well and make the desired selection from the menu.

**Note:** The Move down to height can go as much as 2 mm below the bottom of the well as long as the pod can descend that far. Descending below the bottom of the well potentially corrects for any slight imperfections in the labware or deck position heights and can be done safely because the pin push plate is in a floating position, allowing the pins to move up and down freely without resistance. To specify a height below the bottom of the well, right-click the graphic of the pin in the well and select Custom Height. Enter a negative value in Custom Height to move the pins below the bottom of the well.

- 10. In Move down at, enter the speed of the descent of the HDR Tool Body as a percentage of the maximum speed of the pod.
- 11. In Move up to, specify the height the pins ascend to between each dipping cycle.

**Note:** If Dip Cycles is set to **1**, Move up to is disabled.

**Note:** By default, Move up to is measured from bottom of the well. To measure from top of the well or from liquid, right click the graphic of the pin in the well and make the desired selection from the menu.

- 12. In Move up at, enter the speed at which the HDR Tool Body moves up as a percentage of the maximum speed of the pod.
- 13. In Pause for, enter the length of time the pod pauses at the down position for each Dip Cycle to deposit liquid.
- 14. Choose **OK** to save the Dipping Configuration and return to the source labware configuration.

15. Choose **Array Configuration** to configure arrays for the destination wells. Array Configuration appears (Figure 19-12).

Array Configuration	
Make an array with th	e following characteristics:
	Array dimensions:
$\left( \cdot \right)$	X distance between spot centers: 0.0 mm
	Y distance between spot centers: 0.0 mm
	OK Cancel

Figure 19-12. Array Configuration

**Note:** It is recommended that destination labware is placed on a Positive Position ALP during gridding applications for best results. Refer to the *ALPs User's Manual*, Chapter 15, *Positive Position ALP*, for more information on the Positive Position ALP.

16. In Array dimensions, select the desired size of the array. The first value changes the number of rows in the array and the second value changes the number of columns in the array.

**Note:** The maximum Array dimension is **6x6** when transferring to a 96-well microplate or **4x4** when transferring to a 384-well microplate.

17. In X distance between spot centers, enter the distance in millimeters (mm) between the centers of adjacent spots in each row.

**Note:** The X distance between spot centers updates automatically as the second value in Array dimensions changes.

18. In Y distance between spot centers, enter the distance in millimeters (mm) between the centers of adjacent spots in each column.

**Note:** The Y distance between spot centers updates automatically as the first value in Array dimensions changes.

19. Choose **OK** to save the Array Configuration and return to the Destination Labware configuration.

20. Leave the pin height as is to use the settings for the Move down to height specified in the Dipping Configuration.

#### OR

Set the pin height manually using one of the following methods:

• Position the cursor over the graphic of a pin inside a well. Click and drag the graphic up or down. The pin moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.

**Note:** The cursor changes to a hand when positioned over the graphic.

- Select the graphic of a pin inside a well, and use the ↑ and ↓ keys. The textual representation of the height, which is displayed below the graphic, adjusts in 0.1 millimeter (mm) increments.
- Right-click on the graphic of a pin inside a well, and a menu appears. Select Custom Height, and Custom Height appears (Figure 19-8). Insert the Height in millimeters (mm) and, in from, select a reference point from the drop-down list. Choose OK.

Custom Height	
Height:	mm
from Bottom	•
ОК	Cancel

Figure 19-13. Custom Height prompt

21. For an HDR Transfer step, repeat steps 1 to 20 for each additional piece of Destination Labware.



# **19.4 Using the HDR Tool Cleaning Step**

The HDR Tool Cleaning step washes pins on the HDR Tool Body. The HDR Tool Cleaning step uses a "dunk and dry" methodology to wash pins by dipping the pins in one or more reservoirs with wash solutions and drying the pins over a fan.

**Note:** To clean pins between transfers of an HDR Transfer or HDR Combine step, the HDR Tool Cleaning step must be inserted in a Define Procedure step.

The HDR Tool Cleaning step configuration is similar to that of the HDR Transfer or HDR Combine step. Instead of Source and Destination configurations, however, a series of wash operations — either dipping the pins into a wash reservoir or drying the pins at the HDR Pin Drying ALP — are configured. Cleaning operations are performed in the order they are configured.

The HDR Tool Cleaning step configuration includes specifying:

- Pod that performs the cleaning operations.
- Cleaning operation configuration(s).

To configure an HDR Tool Cleaning step:



1. Insert an HDR Tool Cleaning step into the Method View (Figure 19-14).



Figure 19-14. HDR Tool Cleaning step configuration

- 2. In Use pod, specify the pod with the HDR Tool Body used to perform the transfer operations.
  - > **3000** Use pod does not appear in the step configuration.
- 3. Configure Cleaning Operations as described in Section 19.4.1, *Configuring* <u>Cleaning Operations</u>.

## **19.4.1 Configuring Cleaning Operations**

The HDR Tool Cleaning step is configured through a series of cleaning operations. There are two types of Cleaning Operations:

- Wash Operations pins are dipped in wash solution in a reservoir, deepwell microplate, or other labware type (refer to Section 19.4.1.1, <u>Configuring a Wash Operation</u>).
- Drying Operations pins are air-dried by a fan at the HDR Pin Drying ALP on a Biomek FX instrument or the HDR Fan on the tool rack on a Biomek 3000 instrument (refer to Section 19.4.1.2, <u>Configuring a Drying</u> <u>Operation</u>).

As many Cleaning Operations as desired can be configured in the HDR Tool Cleaning step.

**Note:** Cleaning operations are performed in the order they are listed in the step configuration.

## 19.4.1.1 Configuring a Wash Operation

A Wash operation dips pins in a piece of labware a specified number of times for a specified length of time. Usually a reservoir is used as Wash labware, but pins can be dipped in any labware type.

To configure a wash operation for the HDR Tool Cleaning step:



CAUTION: Do not attempt to access a 96-Channel or 384-Channel Tip Wash ALP with a Multichannel Pod equipped with an HDR Tool Body. The gripper may crash and damage the pod, HDR Tool Body, or Tip Wash ALP.



CAUTION: Do not access labware positioned on a 1 x 5 Passive ALP with the HDR Tool Body. The gripper may crash with the ALP.



CAUTION: Do not access labware on a Stirring ALP with the HDR Tool Body. The magnetic stirrer may bend the pins or interfere with the liquid transfer performance of the pins.

1. Select **Click here to add a cleaning operation** (Figure 19-14). The **Cleaning Operation** configuration appears.



Figure 19-15. HDR Tool Cleaning step - new cleaning operation configuration

2. Click on the piece of labware with the desired wash solution in the Current Deck Display. The labware type and position for that piece of labware is entered automatically into the Cleaning Operation configuration (Figure 19-14).

**Note:** If the selected labware is named, the name of the labware appears instead of the deck position.

OR

Select a **Cleaning** Labware type (Figure 19-14) and deck position.

**Note:** Subsequent labware selections modify the active Cleaning Operation configuration. To configure additional Cleaning Operations, select **Click here to add a cleaning operation**, or select outside a cleaning operation configuration, and then select another piece of labware from the Current Deck Display. If an incorrect piece of labware is selected, right-click on the labware title and select **Delete** on the menu that appears.



Figure 19-16. HDR Tool Cleaning step - configuration for a wash operation

3. If the **Cleaning** labware selected is a static reservoir or microplate, verify the labware type and deck position of the labware.

**Note:** A bright blue outline appears in the Current Deck Display around the labware designated for the first **Cleaning** operation; a yellow outline appears in the Current Deck Display around labware designated for any subsequent **Cleaning** operations.

#### OR

On an **FX** instrument, if the Cleaning labware selected is a circulating reservoir (Figure 19-14):

• In Turn pump on for, enter the length of time in seconds to turn the wash pump on before the HDR Tool Body moves to the circulating reservoir.

Note: The default value for Turn pump on for is 2 seconds.

• In Leave pump on for, enter the length of time in seconds to leave the wash pump on after the HDR Tool Body moves away from the circulating reservoir.

**Note:** The default value for Leave pump on for is **2** seconds.

4. To specify the quadrants of a microplate to access, double-click the source labware in the step configuration to zoom in on the labware.

**Note:** To call up a menu for Zoom and a Specify Selection as Text option, right-click on the labware.

Specify Selection as Text allows selection of quadrants as targets for aspirate and dispense operations. Specify Selection as Text may also be used to enter variables or expressions (refer to Section 13.2, *Using Variables* and Section 13.3, *Using Expressions*).

**Note:** Specify Selection as Text is not applicable for 96-well microplates, reservoirs, or 384-well microplates accessed by a 384-pin plate.

5. Select the desired quadrant(s) on the zoomed-in graphic of the labware.

Note: Selecting any well automatically selects all wells in that quadrant.

**Note:** To select multiple quadrants, click and drag over the desired quadrants or hold down **Ctrl** or **Shift** and select the desired quadrants.

6. Choose **Zoom Out** to return to the step configuration screen.
7. Choose **Dipping Configuration** to configure the wash operation. Dipping Configuration appears (Figure 19-17).

Dipping Configuration	
Dip Cycles : 🚺	
Move down to	Move up to
0.00 mm from bottom	-1.00 mm from top
Move down at	Move up at
50 % speed	50 % speed
Pause for 0	seconds after dipping
ОК	Cancel

**Note:** The Dipping Configuration is similar to a pipetting technique. It specifies the method in which pins are washed.

Figure 19-17. Dipping Configuration

8. In Dip Cycles, enter the number of times the pins should descend into the wells.

Note: For Dip Cycles greater than 1, Move up to must be configured.

9. In Move down to, specify the height the pins descend to for each dipping cycle.

**Note:** By default, Move down to is measured from bottom of the well. To measure from top of the well or from liquid, right click the graphic of the pin in the well and make the desired selection from the menu.

**Note:** The Move down to height can go as much as 2 mm below the bottom of the well as long as the pod can descend that far. Descending below the bottom of the well potentially corrects for any slight imperfections in the labware or deck position heights and can be done safely because the pin push plate is in a floating position, allowing the pins to move up and down freely without resistance. To specify a height below the bottom of the well, right-click the graphic of the pin in the well and select Custom Height. Enter a negative value in Custom Height to move the pins below the bottom of the well.

10. In Move down at, enter the speed of the descent of the HDR Tool Body as a percentage of the maximum speed of the pod.

11. In Move up to, specify the height the pins ascend to between each dipping cycle.

Note: If Dip Cycles is set to 1, Move up to is disabled.

**Note:** By default, Move up to is measured from bottom of the well. To measure from top of the well or from liquid, right click the graphic of the pin in the well and make the desired selection from the menu.

- 12. In Move up at, enter the speed at which the HDR Tool Body moves up as a percentage of the maximum speed of the pod.
- 13. In Pause for, enter the length of time the pod pauses at the down position for each Dip Cycle to wash pins.
- 14. Choose **OK** to save the Dipping Configuration and return to the source labware configuration.
- 15. Leave the pin height as is to use the settings for the Move down to height specified in the Dipping Configuration.

OR

Set the pin height manually using one of the following methods:

• Position the cursor over the graphic of a pin inside a well. Click and drag the graphic up or down. The pin moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.



**Note:** The cursor changes to a hand when positioned over the graphic.

- Selecting the graphic of a pin inside a well, and use the ↑ and ↓ keys. The textual representation of the height, which is displayed below the graphic, adjusts in 0.1 millimeter (mm) increments.
- Right-click on the graphic of a pin inside a well, and a menu appears. Select Custom Height, and Custom Height appears (Figure 19-8). Insert the Height in millimeters (mm) and, in from, select a reference point from the drop-down list. Choose OK.

Custom Height	
Height:	mm
from Bottom	•
ОК	Cancel

Figure 19-18. Custom Height prompt

 Configure additional Cleaning Operations following the procedures in Sections 19.4.1.1, <u>Configuring a Wash Operation</u> and 19.4.1.2, <u>Configuring a Drying</u> <u>Operation</u>.

#### 19.4.1.2 Configuring a Drying Operation

To configure a drying operation for the HDR Tool Cleaning step:

1. Select **Click here to add a cleaning/drying operation**. The Cleaning Operation configuration appears (Figure 19-19).



Figure 19-19. HDR Tool Cleaning step - new cleaning operation configuration

2. Click on the position with the fan (either the HDR Pin Drying ALP on the Biomek FX instrument or the tool rack with the HDRFan tool installed on the Biomek 3000 instrument) in the Current Deck Display. Information for the fan is entered automatically into the Cleaning Operation configuration.

**Note:** Subsequent labware selections modify the active Cleaning Operation configuration. To configure additional Cleaning Operations, select **Click here to add a cleaning operation**, or select outside a cleaning operation configuration, and then select another piece of labware from the Current Deck Display. If an incorrect piece of labware is selected, right-click on the labware title and select **Delete** on the menu that appears.



Figure 19-20. HDR Tool Cleaning step — configuration for a drying operation

3. In Turn fan on for, enter the length of time in seconds to turn the fan on before the HDR Tool Body moves to the HDR Pin Drying ALP.

**Note:** The default value for Turn fan on for is **2** seconds.

- 3000 Turn on for is not available because the fan unit is not controlled by Biomek Software.
- 4. In Dry for, enter the length of time in seconds to keep the HDR Tool Body at the HDR Pin Drying ALP to dry.

Note: The default value for Dry for is 5 seconds.

5. In Leave fan on for, enter the length of time in seconds to leave the fan on after the HDR Tool Body moves away from the HDR Pin Drying ALP.

Note: The default value for Leave fan on for is 2 second	s.
--	----

- 3000 Leave on for is not available because the fan unit is not controlled by Biomek Software.
- 6. Leave the pin height as is to use the default setting.

OR

Set the pin height manually using one of the following methods:

- Position the cursor over the graphic of a pin inside a well. Click and drag the graphic up or down. The pin moves with the cursor, and the height displayed below the graphic is adjusted as the graphic is manipulated.
- Selecting the graphic of a pin inside a well, and use the ↑ and ↓ keys. The textual representation of the height, which is displayed below the graphic, adjusts in 0.1 millimeter (mm) increments.
- Right-click on the graphic of a pin inside a well, and a menu appears. Select Custom Height, and Custom Height appears (Figure 19-8). Insert the Height in millimeters (mm) and, in from, select a reference point from the drop-down list. Choose OK.

 $\bigcirc$ 

**Note:** The cursor changes to a hand when positioned over the graphic.

Custom Height				
Height:	mm			
from Bottom	•			
OK	Cancel			

Figure 19-21. Custom Height prompt

 Configure additional Cleaning Operations following the procedures in Sections 19.4.1.1, <u>Configuring a Wash Operation</u> and 19.4.1.2, <u>Configuring a Drying</u> <u>Operation</u>.

## 19.5 Using the HDR Move Labware Step (FX only)

The HDR Move Labware step moves labware from one position on the Biomek FX deck to another position. HDR Move Labware can also remove labware from the deck by placing it in a Disposal ALP or moving it to an external hardware device without halting the Biomek FX or the method, and without causing a light curtain violation.

**Note:** The HDR Move Labware step should be used only when moving labware with a Multichannel Pod that has the HDR Tool Body installed. If the pod is equipped with any other multichannel head, use the Move Labware step found on the Basic Step Palette (refer to Section 15.4, *Move Labware Step*). A validation error results if the inappropriate Move Labware step is used.



Insert an HDR Move Labware step into the Method View (Figure 19-22).



Figure 19-22. HDR Move Labware step and configuration

#### 19.5.1 Configuring the HDR Move Labware Step

To configure an HDR Move Labware step, complete the following steps:



CAUTION: Make sure the correct ALP is chosen when configuring the deck setup in the Deck Editor. ALPs vary in height, and failure to choose each ALP correctly in the software may result in collisions between pod(s) and ALPs during operation.



CAUTION: Do not access labware positioned on a 1 x 5 Passive ALP with the HDR Tool Body. The gripper may crash with the ALP.

1. In Using Pod, select the pod moving the labware.

**Note:** In Using Pod, Pod1 is the default for a one-pod Biomek FX system. In a two-pod Biomek FX system, the pod configured as the default pod is displayed in Using Pod. If the other pod is desired, select the pod from the drop-down list. Only a Multichannel Pod with the HDR Tool Body installed may use the HDR Move Labware step.

- 2. In Move labware from, select the original deck position for the labware from the Current Deck Display. This instructs the Biomek FX to Move labware from a specific deck position in preparation for leaving it at a final destination.
- 3. In to, select the final deck position for the labware from the Current Deck Display. This instructs the Biomek FX to move the labware to a final deck position.
- 4. Select the desired option for moving stacked labware:
  - Move the entire stack of labware moves all labware in the stack; this option should be selected when moving a single unstacked piece of labware.
  - Move stack, leaving the bottom piece of labware at the source position moves all labware in the stack except for the bottom piece.

**Note:** Selecting Move stack, leaving the bottom piece of labware at the source position when the source deck position contains only one piece of labware results in an error.

 Move the topmost . . .piece(s) of labware from the stack moves only the specified number of labware from the top of the stack.

**Note:** Refer to Section 7.3.6.4.1, *<u>Biomek Stacking Rules</u>*, for complete information on using stacks with the Biomek FX instrument.

# Using the Biomek 3000 Step Palette

## 20.1 Overview

The Biomek 3000 Step Palette provides steps for use specifically with a Biomek 3000 workstation, including tool loading and unloading and aspirate and dispense operations with the Wash System.

The steps available in the Biomek 3000 Step Palette are:



- Change Tool unloads the tool currently loaded on the Biomek 3000 head assembly to an empty tool rack position and loads the specified tool.
- Load Tool locates the specified tool on the deck and loads it to the Biomek 3000 head assembly.



Load Tool Step

• Unload Tool — unloads the tool currently loaded on the Biomek 3000 head assembly to an empty tool rack position.



Bulk Dispense

- Remove to Waste aspirates liquid to waste using a wash tool.
- Bulk Dispense performs a bulk dispense operation from a wash liquid container using a wash tool.

•



Purge Wash Tool — purges the tubing of the wash system and wash tool.



• Wash Plate — perform a plate washing operation using a wash tool.

When one of the above steps is added to a method, the configuration for the step appears on the right side of the Biomek main editor, in step configuration (Figure 20-1).

**Note:** Steps not configured appropriately in Step Configuration generate errors when the method is validated or run.

## 20.2 Displaying the Biomek 3000 Step Palette

In order to add Biomek 3000 steps to a method, display the Biomek 3000 Step Palette (Figure 20-1).



Figure 20-1. Biomek Software main editor with Biomek 3000 Step Palette displayed

To display the Biomek 3000 Step Palette, complete the following:

• Right-click any empty palette space, and the Step Palette menu appears. Select **Biomek3000**.

OR

• From the toolbar, select **Options>Toolbars>Biomek3000**.

## 20.3 Change Tool Step

The Change Tool step is used to instruct the Biomek 3000 workstation to unload the tool installed on the head assembly and load a different tool. The Biomek 3000 workstation automatically locates an empty tool rack position, unloads the tool at the empty position, locates the desired tool, and loads the tool to the head assembly.

**Note:** The Transfer, Combine, and Move Labware steps automatically change to an appropriate tool to perform the operation. However, if a specific tool is desired for an operation, a Change Tool or Load Tool step should be inserted prior to the operation to guarantee the desired tool is used.



Insert a Change Tool step into the method. The Change Tool step configuration appears (Figure 20-2).



Figure 20-2. Changing tools during a method

To configure the Change Tool step configuration:

- 1. In Change to Tool, specify the tool to load after unloading the current tool. All tools located on the deck in a tool rack (as configured in Labware Properties for the tool rack in an Instrument Setup step) are available for selection.
- 2. Select a step that occurs after the Change Tool step or the **Finish** step to validate the step configuration.

## 20.4 Load Tool Step

The Load Tool step is used to instruct the Biomek 3000 workstation to load a tool to the Biomek 3000 head assembly when no tool is currently installed. The Biomek 3000 workstation automatically locates the desired tool in a tool rack and loads the tool to the head assembly.

**Note:** The Transfer, Combine, and Move Labware steps will automatically change to an appropriate tool to perform the operation. However, if a specific tool is desired for an operation, a Change Tool or Load Tool step should be inserted prior to the operation to guarantee the desired tool is used.



Insert a Load Tool step into the method. The Load Tool step configuration appears (Figure 20-3).



Figure 20-3. Loading tool during a method

To configure the Load Tool step configuration:

- 1. In Load Tool, specify the tool to load to the head assembly. All tools that are located on the deck in a tool rack (as configured in the Labware Properties for the tool rack in an Instrument Setup step) are available for selection.
- If the tool selected in Load Tool is a wash tool, in Liquid, select the liquid type to purge the tubing and wash tool with after loading the tool. Biomek Software automatically performs a purge operation similar to a <u>Purge Wash Tool Step</u> (Section 20.8) with the selected Liquid when the tool is loaded.



Figure 20-4. Load Tool step configuration with wash tool selected

3. Select a step that occurs after the Load Tool step or the **Finish** step to validate the step configuration.

## 20.5 Unload Tool Step

The Unload Tool step is used to instruct the Biomek 3000 workstation to unload the tool installed on the head assembly. The Biomek 3000 workstation automatically locates an empty tool rack position and unloads the tool at the empty position.

A tool must be unloaded before loading a new tool. Many steps, such as Transfer, Combine, and Move Labware automatically unload the current tool and load a more appropriate tool for the operation It is only necessary to use an Unload Tool step when it is desired to unload the current tool prior to loading a specific tool to the head assembly using a Load Tool step.

**Note:** The Biomek 3000 head assembly can be instructed to automatically unload any tools installed at the end of a method in the Finish step.



Insert an Unload Tool step into the method. The Unload Tool step configuration appears (Figure 20-5).



Figure 20-5. Unloading a tool during a method

The Unload Tool step requires no additional configuration. Select a step that occurs after the Unload Tool step or the **Finish** step to validate the step configuration.

#### 20.6 Remove to Waste Step

The **Remove to Waste** step is used to aspirate liquid using a single-channel or eightchannel wash tool to an off-deck waste container. The wash tools has two needles for each channel: a long aspiration needle and a shorter dispense needle. Liquid is aspirated from the wells using the aspiration needles by applying a vacuum for a specified amount of time and removed to an off-deck waste container.

**Note:** A Wash Unit must be installed on the Biomek 3000 workstation and connected to an external vacuum pump or vacuum source to aspirate liquid using the Remove to Waste step.



Insert a Remove to Waste step into the method. The Remove to Waste step configuration appears (Figure 20-6).

🌵 Biomek	® Software	e - Method1* [New]	ND.
File Edit	Project Ins	Istrument Execution Options Help	
]∟≌≀	n B  B.		
Change Tool Step Load Tool Step Unload Tool Step Unload Tool Step Waste	Instrument Setup Transfer Combine Move Labware Dause	Start Instrument Setup Load Tool Step Remove to Waste Finish Finish Labware Type: Pod: Pod1 Labware Type: Pod1	
Purge Wash Tool	Comment	Aspirate for 1 s	
Method1*	Biomek3000	0 Biomek3000 ETC: 0:00:28	

Figure 20-6. Remove to Waste step configuration

**Note:** A wash tool must be loaded onto the Biomek 3000 head assembly using either a Load Tool or Change Tool step prior to the Remove to Waste step.

**Note:** If a 6-port value is installed with the Wash Unit, a Device Action step must be inserted to configure the wash liquids available from each port.

To configure the Remove to Waste step:

 Click on the desired piece of labware in the Current Deck Display. The Labware Type and Position for that piece of labware is entered automatically into the Remove to Waste step configuration.

OR

Select a **Labware Type** and **Position** from the drop-down list.

2. On the plate map, select the wells from which to aspirate liquid to waste.

**Note:** Hold **Shift** and click a well to select additional wells without deselecting any wells. Hold **Ctrl** and click a well to toggle the selection status of selected wells without affecting the status of other wells. If Shift or Ctrl is not held down when selecting wells, any previous selection is deleted.

- 3. In Liquid Type, select the liquid type to aspirate.
- 4. In Aspirate for, enter the length of time in seconds to apply vacuum and aspirate liquid to waste for each well.
- 5. Select a step that occurs after the **Remove to Waste** step or the **Finish** step to validate the step configuration.

## 20.7 Bulk Dispense Step

The Bulk Dispense step is used to dispense liquid using a single-channel or eightchannel wash tool. The wash tools have two needles for each channel: a long aspiration needle and a shorter dispense needle. Liquid is dispensed from the source wash liquid container through the dispense needles. The volume dispensed is limited only by the amount of liquid in the source container.

After dispensing, a Backup Volume is aspirated by the dispense needles to prevent droplets from forming on the dispense needles. The Backup Volume is set in Hardware Setup (refer to the *Biomek*® 3000 Laboratory Automation Workstation User's Manual, Section 12.4.3, <u>Configuring Wash Tools</u>).

**Note:** A Wash Unit must be installed on the Biomek 3000 workstation to dispense liquid using the Bulk Dispense step.



Insert a Bulk Dispense step into the method. The Bulk Dispense step configuration appears (Figure 20-7).

the Biomek® Software - Method1* [New]		
File Edit Project Instrument Execution Options Help		
Ett 😤 📓 Start		
Change Instrument Setup		
Load Tool Step	2 2	
Load Tool Step Bulk Dispense		
Finish		
Unload Tool Combine		
	Labware Type: Pod: Pod1	
Remove to Move Waste Labware	Position:	
	Liquid Type: Water	
Buik Pause Dispense	Zoinme: In the shear lon x	
Tool Comment		
Wash Plate		
	P1 P3 P5 P7	
Method1* Biomek3000 Biomek3000 ETC: 0:00:28		

Figure 20-7. Bulk Dispense step configuration

**Note:** A wash tool must be loaded onto the Biomek 3000 head assembly using either a Load Tool or Change Tool step prior to the Bulk Dispense step.

**Note:** If a 6-port value is installed with the Wash Unit, a **Device Action** step must be inserted to configure the wash liquids available from each port.

To configure the Bulk Dispense step:

1. Click on the desired piece of labware in the Current Deck Display. The Labware Type and Position for that piece of labware is entered automatically into the Bulk Dispense step configuration.

OR

Select a Labware Type and Position from the drop-down list.

2. On the plate map, select the wells from which to aspirate liquid to waste.

**Note:** Hold **Shift** and click a well to select additional wells without deselecting any wells. Hold **Ctrl** and click a well to toggle the selection status of selected wells without affecting the status of other wells. If **Shift** or **Ctrl** is not held down when selecting wells, any previous selection is deleted.

3. In Liquid Type, select the liquid type to dispense.

**Note:** If using the Wash Tool Dispense step with a six-port valve, the liquid type must be one of the liquid types configured for the six-port valve in the Device Action step.

- 4. In Volume, enter the volume to dispense in microliters ( $\mu$ L).
- 5. In Speed, enter the speed at which to dispense.
- 6. Select a step that occurs after the Bulk Dispense step or the **Finish** step to validate the step configuration.

#### 20.8 Purge Wash Tool Step

The **Purge Wash Tool** step is used to rinse the tubing and needles of the tools before use. A purge is automatically performed when a wash tool is loaded or when liquids are switched using the 6-port valve. The **Purge Wash Tool** step enables another purge to occur at any time during a method. The **Purge Wash Tool** step is useful for rinsing the tubing and wash tool when changing wash liquids using a 6-port valve.

During a purge, a wash liquid is dispensed through the dispense tubing and dispense needles on the wash tool. At the same time, wash liquid is aspirated using the aspirate needles on the wash tool and removed to waste through the aspirate tubing. The purge process effectively rinses the wash system tubing and wash tool, preparing it for use with the specified wash liquid.

After a purge operation, the dispense needles aspirate a **Return Volume** to prevent droplets from forming on the dispense needles. The **Return Volume** is set in Hardware Setup (refer to the *Biomek*® 3000 Laboratory Automation Workstation User's Manual, Section 12.4.3, <u>Configuring Wash Tools</u>).

**Note:** It is recommended to insert a **Purge Wash Tool** step using distilled or deionized water at the end of each method that uses the wash tool to maintain the wash system.

**Note:** A Wash Unit must be installed on the Biomek 3000 workstation to purge using the **Purge Wash Tool** step.



Insert a Purge Wash Tool step into the method. The Purge Wash Tool step configuration appears (Figure 20-8).

🌵 Biomek File Edit	(*) Biomek® Software - Method1* [New]				
] 🗅 📽 (	≝́∎∣&	ခံ 🌡 🖻 🛍 က က 🖉			
<b>₽</b> ₽Ŷ	48	Start	Liquid: Default		
Change Tool Step	ער Instrument Setup	Instrument Setup	Purge 1000 uL per channel		
Ьţ	R	Load Tool Step	at 100 % speed		
Load Tool Step	Transfer	Purge 1000 uL at 100%			
ti)	45	Finish			
Unload Tool Step	Combine				
Waste	Labware				
Bulk	<b>D</b> ana				
Dispense	Fause				
Purge Wash Tool	Comment				
Wash Plate					
			uuuuuu,		
				1	
Method1*	Biomek3000	Biomek3000 ETC: 0:00:28			

Figure 20-8. Wash Tool Purge step configuration

**Note:** A wash tool must be loaded onto the Biomek 3000 head assembly using either a Load Tool or Change Tool step prior to the Wash Tool Purge step.

**Note:** If a 6-port value is installed with the Wash Unit, a **Device Action** step must be inserted to configure the wash liquids available from each port.

To configure the Purge Wash Tool step:

1. In Liquid, select the liquid type to use for the purge.

**Note:** If using the Wash Tool Dispense step with a six-port valve, the liquid type must be one of the liquid types configured for the six-port valve in the Device Action step.

- 2. In Purge, enter the volume to dispense from each channel in microliters ( $\mu$ L).
- 3. In Speed, enter the speed at which to purge.
- 4. Select a step that occurs after the Purge Wash Tool step or the **Finish** step to validate the step configuration.

#### 20.9 Wash Plate Step

The Wash Plate step is used to perform plate washing operations using a wash tool on the Biomek 3000 head assembly. Plate washing is performed as part of many assays to remove any unbound substrates before further processing.

A typical plate washing procedure performs the following actions:

- dispense liquid into wells from dispense needles on the wash tool.
- after dispensing, aspirate a Backup Volume with dispense needles to prevent droplets from forming on the dispense needles. The Backup Volume is set in Hardware Setup (refer to the *Biomek*® 3000 Laboratory Automation Workstation User's Manual, Section 12.4.3, <u>Configuring Wash</u> <u>Tools</u>).
- pause for a specified length of time.
- aspirate liquid from the wells using the aspirate needles on the wash tool and remove it to waste.
- repeat a specified number of times.

**Note:** A Wash Unit must be installed on the Biomek 3000 workstation to wash plates using the Wash Plate step.



Insert a Wash Plate step into the method. The Wash Plate step configuration appears (Figure 20-9).

👫 Biomek® Software - Method1* [New]				
File Edit		rrument Execution Options Heip		
	d Hila			
H\$		Start		
Change Tool Step	Instrument Setup	Instrument Setup		
Ы		Load Tool Step	っっ	
Load Tool Step	Transfer	Wash Plate		
Ħ		Finish		
Unload Tool Step	Combine			
- CP			Labware Type:	
Remove to Waste	Move Labware		Position:	
Ö.	8		Liquid Type: Water Wash 1 times	
Bulk	Pause		Volume: 0 µL Speed: 50 %	
	A		Aspirate for 1 s Pause after dispense 0 s	
Purge Wash	Comment			
1001				
West Flor				
Wash Plate	]			
			[mmm],	
			P7	
			· · ·	
Method1*	Biomek3000	Biomek3000 ETC: 0:00:28		

Figure 20-9. Wash Plate step configuration

To configure the Wash Plate step:

**Note:** A wash tool must be loaded onto the Biomek 3000 head assembly using either a Load Tool or Change Tool step prior to the Wash Plate step.

**Note:** If a 6-port valve is installed with the Wash Unit, a **Device Action** step must be inserted to configure the wash liquids available from each port (refer to Section 22.6.1.1, *Configuring the Device Action Step for 6-Port Valve (3000 only)*).

1. Click on the desired piece of labware in the Current Deck Display. The Labware Type and Position for that piece of labware is entered automatically into the Wash Plate step configuration.

OR

Select a **Labware Type** and **Position** from the drop-down list.

2. On the plate map, select the wells of the microplate to wash.

**Note:** Hold **Shift** and click a well to select additional wells without deselecting any wells. Hold **Ctrl** and click a well to toggle the selection status of selected wells without affecting the status of other wells. If **Shift** or **Ctrl** is not held down when selecting wells, any previous selection is deleted. Wells are selected individually or in rows, depending on the wash tool used.

3. In Liquid Type, select the liquid type to use for plate washing.

**Note:** If using the Wash Plate step with a 6-port valve, the liquid type must be one of the liquid types configured for the 6-port valve in the Device Action step.

- 4. In Wash ... times, enter the number of cycles to perform to wash the plate.
- 5. In Volume, enter the volume to dispense to each well from the dispense needles during each cycle of the plate wash in microliters (μL).
- 6. In Speed, enter the speed at which to dispense.
- 7. In Aspirate for, enter the length of time in seconds to apply vacuum and aspirate liquid to waste for each well using the aspirate needles.
- 8. In Pause after dispense, enter the length of time after the dispense to pause before beginning the aspirate operation.
- 9. Select a step that occurs after the Wash Plate step or the **Finish** step to validate the step configuration.



## 21.1 Overview

The steps presented in the Specialty Step Palette allow more control over methods. These steps are used to:

- Set variables and make non-incremental changes to variables.
- Break out of loops.
- Create and use groups of labware.
- Run executables.
- Define and run procedures.

**Note:** The steps available in the Specialty Step Palette do not use the pod directly and may be used with any single or dual-pod configuration.

Steps available in Specialty include:

• Create Group — creates a group of labware.



• Next Labware — accesses the next piece of labware in a group of labware created using Create Group.



Next abware

Run Program — runs any executable file during a method.



Next Item — binds a series of values to a global variable.



٠

٠

Set Global — defines a global variable that can be used in subsequent steps of a method.

Define Procedure — creates a series of steps that can be run at any



- Run Procedure runs a series of steps previously created in a Define Procedure step.
- Define Procedure



• Break — breaks out of one or more loops.

point during a method using a Run Procedure step.

# 21.2 Displaying the Speciality Step Palette

To add specialty steps to a method, display the Specialty Step Palette (Figure 21-1).



Figure 21-1. Biomek Software main editor with Specialty Step Palette displayed

To display the Specialty Step Palette:

- Right-click any empty palette space, and select **Specialty** from the menu.
   OR
- From the toolbar, select **Options>Toolbars>Specialty**.

# 21.3 Create Group Step

The **Create Group** step creates and names a group of labware that may be accessed during a method run using a **Next Labware** step.

#### 21.3.1 Configuring the Create Group Step



To insert a **Create Group** step into a method, insert the **Create Group** in the Method View. The **Create Group** Step Configuration appears (Figure 21-2).

MBiomek® Software - Method3* [New] File Edit Project Instrument Execution Options Help		- O ×
	JX           Group:	
Greate Group Create Group		
Next Labware	Create Group configuration	
Run T	Create Group is used to create a	
Nethod View	to that group.	
The method displayed contains a <b>Create</b>		
Group step. The group	Move the first labware to ving Pod1	
caption.		
Procedure		
Break		
	P7 P11 P15 P19	
Method3* BiomekFX BiomekFX ETC: 0:00:03		

Figure 21-2. Create Group step and configuration

To configure the Create Group step:

- 1. In **Group**, enter a name for the group of labware (Figure 21-3).
- 2. Click on labware in the Current Deck Display to add them to the group.

🌵 Biomek® Software - Method3* [New]		- <b>-</b> ×
File Edit Project Instrument Execution Options Help	*	Enter the name of the group here.
Start	Group: Sources	k display to add them to the group
🐨 🧟 Instrument Setup	Position: P5	Delete
Group Create Group "Sources"	Position: P9	Delete
	Position: P6	Delete
Labware	Position: P10	Delete
Gal		<b>_</b>
Provides an optio	n to allow a Multichannel	
Pod or Gripper to	ol to move labware to the	Delete labware
Next Item appropriate deck	position	from the group
		here
Set Global		
	Move the first labware to	Pod1
Run Procedure		
Define Procedure		
Break		
Current Deck		DIO DIC
Display		P12 P16
Clicking on labware		P13 P17
here adds them to the		
group.		P14 P18
	P7 P11	P15 P19
	<u> </u>	
Method3* BiomekFX BiomekFX ETC: 0:00:03		

Note: Choose Delete to remove the labware from the group.

Figure 21-3. Create Group step configuration with labware added

3. Choose **Move the first labware to** to allow the pod to move labware to a specific position, if desired.

**Note:** Selecting Move the first labware to is equivalent to inserting a Next Labware step right after the Create Group step. The group variable is assigned the value of the first labware in the group and is moved to the specified position.

- FX When selecting Move the first labware to, the chosen pod must be a Multichannel Pod; these pods have a gripper that can grasp and move labware.
- 3000, NX-S8 To use a gripper tool, it must be installed in Hardware Setup, configured in Instrument Setup, and available on the deck.

4. If Move the first labware to is selected, select the deck position to which to move the first labware.

**Note:** Click on any empty deck position in Current Deck Display. Move the first labware to is selected, and the deck position chosen in the Current Deck Display is selected.

If a deck position is not selected, it leaves the labware at its current location.

5. If **Move the first labware to** is selected, select the pod used to move the first labware to the selected position.

**Note:** Choosing the position and pod is a necessary option if a Stacker Carousel presents a piece of labware that must be moved to a deck position.

- FX The chosen pod must be a Multichannel Pod; these pods have a gripper that can grasp and move labware.
- 3000, NX-S8 To use a gripper tool, it must be installed in Hardware Setup, configured in Instrument Setup, and available on the deck.

## 21.4 Next Labware Step

Next Labware assigns the next labware in the group as the value of the global variable created in the Create Group step. The global variable created in the Create Group step is not assigned a value until a Next Labware step has assigned a value to the global variable. The Next Labware step also has options to get the next labware only when a pattern is completed or to break out of the current loop when all labware is used.

#### 21.4.1 Configuring the Next Labware Step



Insert the **Next Labware** step in the Method View. The **Next Labware** Step Configuration appears (Figure 21-4)

the Biomek® Software - Specialty Method FX* [Development]		_ 🗆 🗙
File Edit Project Instrument Execution Options Help		
JD ☞ ╈ █ Q & # N N N N N N N N N N N N N N N N N N		
Start	Group: Sources	
Create 🐔 Instrument Setup	To Location:	
💑 🌑 Create Group "Sources"	Pod: Pod1	
Next O Loop from 1 to 100 step 1	Get the next labware only if the last pattern used on the labware was completed	
Col Transfer 20 µL from P4 to P5	Break out of the current loop if there are no more unused labware in the given group	
Run 🕐 Next Labware		
End Loop		
Next Item Finish		
Set Global		
Run Procedure		
Define Procedure		
Break		
	P13 P17	
	P14 P18	
Specialty Method FX* BiomekFX BiomekFX ETC: 0:00:29		

Figure 21-4. Next Labware step and configuration

To configure the Next Labware step:

1. Choose the **Group** from the pull-down menu.

**Note:** All the groups created in the current method are listed.

2. In **To Location**, choose the deck position to which the next listed labware in the selected **Group** is moved by selecting from the list

OR

Click on the empty deck position in the Current Deck Display.

- 3. Check **If the labware is on deck leave it where it is** to leave the labware in its current location instead of moving it to the **To Location**, if desired.
- 4. Choose the **Pod** moving the labware to the specified **To Location**.

**Note:** Choosing the position and pod is necessary when a Stacker Carousel presents a piece of labware that must be moved to a deck position.

- FX On the Biomek FX, the chosen pod must be a Multichannel Pod; these pods have a gripper that can grasp and move labware.
- 3000, NX-S8 To use a gripper tool, it must be installed in Hardware Setup, configured in Instrument Setup, and available on the deck.
- 5. Choose **Get the next labware only if the last pattern used on the labware was completed** to allow the next piece of labware in the group to be accessed only if all the selected wells on the current piece of labware have been accessed in a **Transfer** or **Combine** step.
- 6. Choose **Break out of the current loop if there are no more unused labware in the given group** to break out of the current loop and allow the method to continue when all the labware in the group has been accessed.

#### 21.4.2 Using Group Variables in a Method

Group names configured in a Create Group step and assigned a value with a Next Labware step may be used in any deck position field in any step within the method.

To use a group variable in another step:

- 1. Select a field that requires a deck position in the desired step configuration. For example, the group **Sources** is used to specify the deck location of the destination for the **Transfer** step (Figure 21-5).
- 2. Enter the group name, preceded by an equal sign. For example, **=Sources** in the at field of the destination in the **Transfer** step indicates a volume is transferred to the labware at the deck position indicated by the group variable **Sources**.

**Note:** Group names may be used in any field where a deck position is a valid value for that field.

Stat	() Biomek	® Software · Project Insh	- Specialty ument Exe	Metho	d FX* [New] Ontions Help	_□×
Stat   Arren   Arren <th>000</th> <th></th> <th><b>€</b> } % ⊑</th> <th>b 🖻</th> <th></th> <th></th>	000		<b>€</b> } % ⊑	b 🖻		
Correction Average   Average Average   Average Average <th></th> <th></th> <th>æ</th> <th>00</th> <th>Start</th> <th>Use god Pod1 ransfer.</th>			æ	00	Start	Use god Pod1 ransfer.
With With With With With With With With	Create	Aspirate	Instrument	<b>E</b>	Instrument Setup	A Tip Handling
New Wash hops in Wash   New Wash   Non	Group		Setup	۲	Create Group "Sources"	✓ Load AP96_200uL ▼ tips and unload them Using a group defined in
Universe   With   With </th <th>Next</th> <th>Dispense</th> <th>Transfer</th> <th>Õ</th> <th>Loop from 1 to 100 step 1</th> <th>Wash tips in Water I 3 Character Group step</th>	Next	Dispense	Transfer	Õ	Loop from 1 to 100 step 1	Wash tips in Water I 3 Character Group step
Run Mix Condense   Run Mix Condense   With Mix Condense   Mix Mix   Mix	Labware	do			Transfer 20 µL from P4 to	Change ups between sources.
With With   With With   With With   Set Global With   New Tits Parse   With With   With With   Definish With   With With   With With   With With   Definish With   With With   With With   With With   Definish With   With With   With With   Definish With   With With   With With   With With   Definish With   With W	Run	Mix	Combine		Next "Sources"	Destination: =Sources
Next Item Wish   Wish Wish   Set Clobal   New Tips   Passe   Wish Wish   Wish Wish   New Tips   Passe   Wish   New Tips   Passe   Wish   Wish   New Tips   Passe   Wish   Wish   New Tips   Passe   Wish   Wish   Wish   New Tips   Passe   Wish   Wish   Wish   Wish   New Tips   Passe   Wish   Wish   Wish   Wish   Wish   Wish   New Tips   Passe   Wish   Wis		87 B			End Loop	
Set Global   New Tps   Puss   Procedure   Unload Tps   Comment     Define   Define <th>Next Item</th> <th>Wash</th> <th>Move Labware</th> <th>8</th> <th>Finish</th> <th></th>	Next Item	Wash	Move Labware	8	Finish	
Set Global New Tips   Procedure   Window Tips   Define   Define<	(MA)		8			Technique: Low-Volume
Note   Procedure   Outpoint   Define   Define<	Set Global	New Tips	Pause			2.40 mm from bottom [Overrides Technique]
Procedure       Unload Type       Comment         Define Procedure       Image: Stop when finished wijh Destinations Image: Stop when finished wijh Destinating Image: Stop when finished wijh Destinatio	5	0000				Click here to add a destination
Stop wen insured win [Jestinations]       Advanced         Define Procedure       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]         Break       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]         Break       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]         Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]         Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]         Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]       Image: Stop wen insured win [Jestinations]         Image: Stop went insured win [Jestinations]       Image: Stop went insured win [Jestinations]       Image: Stop went insured win [Jestinations]       Image: Stop went insured win [Jestinations]         Image: Stop went insured win [Jestinations]       Image: Stop went insured win [Jestinations]       Image: Stop went insured win [Jestinations]       Image: Stop went insured win [Jestinations]         Image: Stop went insured win [Jestination]       Im	Run Procedure	Unload Tips	Comment			
Define       Log         Procedure       Log         With Procedure       Aspirate at most         Cleanup       Image: Cleanup         Move Pod       P8         Group       P12         P14       P18         P21       P15         P19	Ø	0				G Disnerse up to 1 time per draw.
Presk       Cleanup         Move Pod       P8       P12       P16         Group       P13       P17       P14       P18         P21       P15       P19       P19	Define Procedure	Loop				C Aspirate at most 0
Break         Cleanap           Move Bod         P8           Group         P13           P14         P18           P21         P15           P19						♥ Transfer Details
Image: Second	Break	Cleanup				
Move Pod           Group           TR1           P13           P14           P15           P19						P8 P12 P16
Image: Constraint of the second sec		Move Pod				
Group Group P14 P18 P21 P15 P19						TR1 P13 P17
P21 P15 P19		Group				P14 P18
				•	<b>I</b>	
		e al denak	er teu e			=

Figure 21-5. Transfer step using the Sources group defined in a Create Group step

# 21.5 Run Program Step

Run Program executes any application within a method. It can be used, for example, to run a custom data logger, display a read-me file or special instructions at specified times during a method, or access a database.

#### 21.5.1 Configuring the Run Program Step



Insert the **Run Program** step in the Method View. The **Run Program** Step Configuration appears (Figure 21-6).

th Biomek® Software - Method4* [New]		×
File Edit Project Instrument Execution Options Help		
Start	Open so that it is Invisible 👻	
Rup Program	to the user, starting in the directory	
Group	with as parameters	
🕐 📓 Finish	after the resource <everything> vis available.</everything>	
Next		
Cabinate	When the program is started:	
GI	C Allow the method to continue independently	
Run	Prote the resource until the program completes     C. Plack all mathed activity including lack shead until the program completes	
	Didux an method address including index anead unkin the program completes	
Next Item	named	
Set Global	This program will take approximately 1 seconds	
	All and a field and the back and the field and the second in an inter-	
	Allow the light cuttain to be broken while the program is funning	
Procedure		
Define Procedure		
Break		
	TL1 P4 P8 P12 P16	
	TR1 P5 P9 P13 P17	
	P20 P6 P10 P14 P18	
	P21 P7 P11 P15 P19	
Method4* BiomekFX BiomekFX ETC: 0:00:02		

Figure 21-6. Run Program step and configuration
To configure the Run Program step:

1. In Open, enter the full path and file name for the program to execute during method run

OR

Choose Browse and select the desired program from Open.

**Note:** Any executable file (.exe, .bat, etc.) in any accessible directory (local or network) may be selected in the Run Program step. Web sites can be accessed (http://www.beckmancoulter.com) or email (mailto:jdoe@example.com) sent.

2. Select how the program should be run in so that it is. Options are Visible or Invisible.

**Note:** Any application that is Visible is run and displayed in a visible window. Any application that is Invisible is run in the background with no visual display. Certain executables, such as Notepad or Calculator, cannot be executed as Invisible.

3. In starting in the directory, specify the directory to search for the parameter entered in with.

**Note:** When the Run Program step is executed, it searches the indicated directory for the specified parameter. Results depend on the application run and whether or not the specified parameter is found.

- 4. In with, enter any parameters used when running the program. For example, if the selected program is Microsoft® Word, the full path and file name of a Word document can be entered as a parameter. The program runs and opens the Word file entered as a parameter in with.
- 5. In after the resource, select a pod, deck position, or ALP. The program executes when the selected resource is available.

**Note:** Select **<Everything>** to run the program when everything on the deck is available.

- 6. Make a selection in When a program is started:
  - Allow the method to continue independently after starting the program, the method continues independent from the program; that is, it does not wait for the program to finish before continuing with the method. This is useful for displaying nonfatal information.
  - Hold the resource until the program completes the selected resource is paused while the program is running and may not be accessed until the program is finished and closed.
  - Block all method activity including look-ahead until the program completes the method pauses and waits for the program to finish before enqueueing the next step and continuing with the method. This is useful when data from the executable is used to make choices in a Biomek method; for example, a LIMS system driving a hit picking method.

7. In This program will take approximately, enter the estimated length of time to run the program in seconds.

**Note:** The time entered in This program will take approximately is only used to calculate the ETC and does not affect the method run or actual run time for the method.

8. If desired, select **Allow the light curtain to be broken while the program is running**. While the program is running, the Biomek instrument is paused and the light curtain may be broken without causing an error. This is useful for integrating third-party devices onto a Biomek system.

**Note:** Allow the light curtain to be broken while the program is running is available when the resource <Everything> is selected and Allow the method to continue independently is deselected.

- > 3000 Does not have light curtain.
- 9. Select a step that occurs after the Run Program step or the Finish step to validate the step configuration.

## 21.6 Next Item Step

The Next Item step is used to iterate over a sequence of values. When configured, the Next Item step names a global variable, provides a list of VB script expressions, and specifies behavior when the list is exhausted.

When the Next Item step is used in a method, the named global variable may be assigned a new value. Each expression is evaluated to form a list of values.

If the named global variable is not assigned a value, it is assigned the first value of the list. If the global variable is assigned a value, the step searches for a matching value in the list and performs one of the following actions based on the results of the search:

- If a match is found, the global variable is assigned the next value in the list.
- If the matching value is the last value in the list, the action to take when the list is exhausted is applied.
- If no match is found, the global variable is assigned the first value in the list.

**Note:** Use a Set Global step prior to the first appearance of a Next Item step to create the variable used in the Next Item step and assign it no value (leave the Value field empty). This guarantees that the first Next Item step assigns the first value in the list to the variable.

The Next Item step is often used with a Loop step, with the value of the variable changing with each iteration of the loop.

**Note:** Refer to Section 16.9, *Loop Step*, for more information on using the Loop step.

### 21.6.1 Configuring the Next Item Step



Insert **Next Item** into the Method View. The **Next Item** Step Configuration appears (Figure 21-7).



Figure 21-7. Next Item step and configuration

To configure the Next Item step:

1. Select a previously defined Variable

OR

Enter a new Variable name.

2. Enter the **Values** of the Variable, separating each with a comma.

**Note:** Deck positions must be entered in double quotes. For example, deck position P4 is entered as **"P4"** in Values. Deck positions may also be entered by clicking on each position in the Current Deck Display.

**Note:** All Values must be unique. If a value is repeated in the list, an error occurs during validation.

- 3. Indicate the action to take when all values have been used:
  - Break out of current loop ends the current loop and continues with the method; Next Item must be placed inside a Loop step or a procedure that is run from inside a Loop step to use this option
  - Set variable to sets the variable to a specified value and uses that value for the variable
  - Start over with the first value sets the variable to the first value and continues; loops the Next Item values

# 21.7 Set Global Step

The Set Global step is used to create a global variable accessible to all steps following the Set Global step in a method. It allows steps to be parameterized by specifying explicit values for fields like volumes and heights. For example, instead of typing the number of microliters into the volume field of an Aspirate step, the name of a variable is inserted. Every time the Aspirate step is executed, the current value of that variable is used as the volume.

Any of the steps following the Set Global step can use variables created in the Set Global Step Configuration. A step uses a global variable by referencing it in its step configuration. The Biomek Software automatically uses the associated value when the step is executed.

**Note:** For information on variables and expressions, refer to Chapter 13, <u>Using</u> <u>Variables and Expressions in a Method</u>.

### 21.7.1 Configuring the Set Global Step

Configuring the Set Global step involves naming the variable and setting its initial value.



Insert **Set Global** into the Method View. The **Set Global** Step Configuration appears (Figure 21-8).



Figure 21-8. Set Global step and configuration

To configure the Set Global step:

- 1. Select **Set Global** in the Method View. The **Set Global** Step Configuration appears.
- 2. Enter the **Variable** name (refer to Chapter 13, <u>Using Variables and Expressions</u> <u>in a Method</u>).
- 3. Enter the desired **Value** for the Variable.

**Note:** The value of the variable is assigned the appropriate units when used in a step configuration. For example, if a variable AspValue with a value of 10 is used in the Volume field of a Transfer step, the software interprets this as 10 microliters ( $\mu$ L).

**Note:** Deck positions must be entered in double quotes. For example, deck position P4 is entered as **"P4"** in Values.

# 21.8 Defining and Running Procedures

The Define Procedure and Run Procedure steps are used to create a series of steps that may be used multiple times in a method. A procedure is created by adding and configuring steps within a Define Procedure step.

Variables may be created in the **Define Procedure** step and used to configure the steps in the procedure. Variables created in the **Define Procedure** step configuration can be overridden in the **Run Procedure** step configuration.

Multiple procedures can be defined in a method, and a procedure can be run multiple times within a method.

The advantages of using procedures are:

- The ability to run the same steps multiple times within a method but configuring them only once
- The size of the current method is controlled which reduces confusion by listing only the Run Procedure step in the Method View, not all of the steps accessed by the Run Procedure step

### 21.8.1 Define Procedure Step

The Define Procedure step is used to create procedures that may be called any time during a method with a Run Procedure step.

Define Procedure

End Procedure

When the Define Procedure step is added to a method, the Define Procedure icon appears. Double-click the icon to expand the procedure and display the End Procedure icon. Any steps placed between the Define Procedure and End Procedure icons are run when the procedure is called with a Run Procedure step. These steps are configured through their individual step configurations.

The procedure may be collapsed to conserve space in the Method View by doubleclicking the Define Procedure icon.

### 21.8.1.1 Configuring the Define Procedure Step



Insert **Define Procedure** in the Method View. The **Define Procedure** Step Configuration appears (Figure 21-9). **Define Procedure** is added to a method collapsed.





To configure the Define Procedure step:

1. In **Procedure**, enter a name for the procedure.

**Note:** The procedure name may contain only alphanumeric characters and the underscore ( ), and are not case-sensitive.

 Insert the variables desired (refer to Chapter 13, <u>Using Variables and Expressions</u> <u>in a Method</u>).

**Note:** Variables created in a **Define Procedure** step are accessible only to the substeps within that procedure.

- 3. Double-click the **Define Procedure** icon in the Method View to open the procedure. The End Procedure icon appears below the **Define Procedure** icon.
- 4. Insert the desired steps between Define Procedure and End Procedure.
- 5. Configure the steps between Define Procedure and End Procedure as desired.
- 6. Double-click the **Define Procedure** icon to collapse the procedure and hide all steps between the **Define Procedure** and **End Procedure** icons.

### 21.8.2 Run Procedure Step

The Run Procedure step executes a procedure previously created with the Define Procedure step. The step inserts all of the steps from the referenced procedure into the current method at the point the Run Procedure step appears. The called procedure then behaves exactly as if the steps were configured at the point where the procedure is called.

The Run Procedure step may also assign new values to variables created in the called procedure. These values override any value configured in the Define Procedure step. This allows a procedure to be used as a template that can be customized each time it is performed by a Run Procedure step by changing the values of the variables.

### 21.8.2.1 Configuring the Run Procedure Step



Insert **Run Procedure** in the Method View. The **Run Procedure** Step Configuration appears (Figure 21-10).

**Note:** The procedure to run in a Run Procedure step must be defined in a Define Procedure step earlier in the method. or in a previously run method that did not clear all global variables in the Finish step



Figure 21-10. Run Procedure step and configuration

- 1. Select the desired procedure from the list of defined procedures. The variables used in that procedure are displayed.
- 2. Enter the desired values for the variables (refer to Chapter 13, *Using Variables and Expressions in a Method*).

**Note:** Variables in a Run Procedure step are accessible only to the called procedure.

### 21.9 Break Step

The Break step is used to break out of one or more loops during a method. Break is usually used as part of an If step to break out of a Loop under certain conditions. When Break is placed under the Then substep and the If statement is true, the Break step ends the Loop.

**Note:** The Break step must be placed inside of a LOOP step or a procedure run from inside a LOOP step or errors are generated during validation or method run.

**Note:** Refer to Section 16.9, *Loop Step*, for more information on using a Loop step. Refer to Section 17.7, *If Step*, for more information on using an If Step.

### 21.9.1 Configuring the Break Step



To insert a Break step into a method, insert the **Break** step inside a Loop step in the Method View. The Break Step Configuration appears (Figure 21-11).



Figure 21-11. Break step and configuration

To configure the Break step:

Select the number of Levels to break out of.

**Note:** Levels refers to the number of nested Loop steps that are discontinued. Break terminates the number of Loop steps specified in Levels and continues with the method.

For example, a method contains a Loop step that contains a second Loop step nested. Within the second Loop step is an If step with a Break step with Levels set at 2 under the Then branch. When the If statement in the second Loop step is true, the Break step terminates both Loop steps and continues with the method. If Levels was set to 1, the Break step terminates the second Loop step and continues with the first Loop step.

**Note:** There must be a number of nested Loop steps greater than or equal to the number of Levels or an error occurs during method validation and run.

# Using the Devices Step Palette

### 22.1 Overview

The steps presented in the Devices Step Palette (Figure 22-1) integrate active ALPs and other devices with the Biomek instrument. The integration allows Biomek Software to control the device.

After an active ALP or device is integrated, Biomek Software understands the attributes of the device, displays the configuration for the device, and receives information for operating the device.

**Note:** Devices must be installed and configured in Hardware Setup. Refer to the instrument hardware manual for more information.

**Note:** Devices must also be configured on the deck in the Deck Editor (refer to Chapter 5, <u>*Preparing and Managing the Deck*</u>).

The steps displayed on the Devices Step Palette include:

of labware required for an SPE operation.

- SPE
- Device Setup configures labware contained in external devices that hold multiple pieces of labware.

SPE — allows a method to easily assemble and disassemble the stack





- Device Action controls devices during a method run.
  - FX controls devices and active ALPs such as the Tip Wash ALP or Stirring ALP.
  - 3000 controls devices such as the autolatch, 6-port valve, vacuum, Wash System, and MicroMix.
  - NX controls devices and active ALPs such as the Source/Waste Sensor and Drainable/Refillable Reservoir ALP.



•

•

- Tip Loader configures a Tip Loader ALP location and status.
  - FX, NX-MC The Tip Loader ALP is used only with a Multichannel Pod.

The following step is displayed when a SILAS device and the software associated with it are installed on the system:



SILAS<sup>™</sup> Module step — controls operation of the associated SILAS Module, such as a barcode reader or a SILAS consumer that has been added to a step palette.

**Note:** The icon for a SILAS Module step varies according to the device the step represents; for example, a Barcode Reader SILAS Module step displays an icon of a barcode reader.

When a device integration step is added to a method, the configuration associated with that step appears on the right side of the Biomek main editor in Step Configuration (Figure 22-1). Steps not configured appropriately can generate errors when the method is validated or run.

# 22.2 Displaying the Devices Step Palette

In order to add device steps to a method, the Devices Step Palette must be displayed (Figure 22-1).



Figure 22-1. Main editor for a Biomek FX with Devices Step Palette displayed

To display the Devices Step Palette, complete the following:

• Right-click any empty palette space, and the Step Palette menu appears. Select **Devices** from the menu.

OR

From the toolbar, select Options>Toolbars>Devices

## 22.3 SPE Step

The Solid Phase Extraction (SPE) ALP applies a vacuum to remove liquid from a filter microplate. The SPE step is a convenient way of executing the assembly and disassembly of the SPE ALP system. The configuration for the SPE step informs Biomek Software of the:

- Operation required of the SPE step (Figure 22-2)
- Location of the filtered microplate used by the SPE ALP
- Locations of the SPE vacuum manifold and receiver
- Location of the SPE ALP on the deck
- Pod constructing/destroying the SPE stack
  - **FX**, **NX-MC** The SPE step is for use with a Multichannel Pod; the gripper must be used to assemble and disassemble the SPE ALP.
  - > **3000** A gripper tool must be installed and configured on the deck.
  - NX-S8 The SPE step may only be used on the NX Span-8 with gripper; the gripper must be used to assemble and disassemble the SPE ALP.

**Note:** The SPE ALP can only be used on a DNA Preparation Deck Layout and occupies two deck positions.

Insert a SPE step into the Method View (Figure 22-2).





Figure 22-2. SPE step and configuration

### 22.3.1 Configuring the SPE Step

When an SPE step is added to a method, the SPE Step Configuration appears (Figure 22-2).

To configure the SPE step:

- 1. Select the **Operation** performed by the SPE ALP from the options listed below:
  - Create assembles the various pieces of an SPE stack
  - **Destroy** disassembles an SPE stack
- 2. Select the location of the **Filter Source** used by the SPE stack. The **Filter Source** is the location on the deck of the filtered microplate that is placed on top of the SPE ALP stack.
- 3. Select the location of the **Manifold Source** for the SPE stack. The **Manifold Source** is the location of the Manifold, which is referred to as the Collar in the *ALPs User's Manual* (refer to the *ALPs User's Manual*, Chapter 16, <u>Solid Phase</u> <u>Extraction (SPE) Vacuum Manifold ALP</u>).

4. Select the location of the **Receiver Source** for the SPE stack. The **Receiver** Source is any filtered microplate that is positioned inside the SPE Collar and the SPE ALP. If no receiver microplate is defined, the fluid flows through the source filtered microplate into the SPE ALP Base. The fluid is removed from the base via the vacuum hose (refer to the *ALPs User's Manual*, Chapter 16, <u>Solid Phase</u> <u>Extraction (SPE) Vacuum Manifold ALP</u>).

**Note:** Filter Source, Manifold Source, and Receiver Source all change to Destination when Operation is set to Destroy.

5. Select **Receiver plate has a holder** if the receiving microplate is positioned in a Filter Holder. The Filter Holder is used to keep the tips of a filtered microplate from touching the base of an ALP stand. A filter microplate without a Filter Holder can be positioned inside a 1x1 ALP without damaging the microplate.

**Note:** Labware, including the Filter Holder, is positioned on the Deck Layout in the Instrument Setup step.

- 6. Select the **Location of SPE ALP** on the Biomek deck.
- 7. Specify the **Pod** accessing the SPE stack.

**Note:** In Pod, Pod1 is the default for a one-pod Biomek system. In a two-pod system, the pod configured as the default pod is displayed in Pod. If the other pod is desired, select the pod from the drop-down list.

8. Select a step that occurs after the SPE step or the **Finish** step to validate the step configuration.

# 22.4 Using External Devices with Biomek Laboratory Automation Workstations

External devices may be used during a method run. In order to use these devices in a method, there are three components that must be configured for proper operation of the device. These components are:

- Device Editor sets the attributes of the device within the Biomek Software (refer to Section 22.4.1, <u>Configuring Devices Using the</u> <u>Device Editor</u>).
- Device Setup Step configures the labware within a device for use in a method (refer to Section 22.5, <u>Device Setup Step</u>).
- SILAS Module configures the operation of the device for the method (refer to Section 22.8.1, <u>Configuring the SILAS Module</u>).

# 22.4.1 Configuring Devices Using the Device Editor

Use the **Device Editor** to configure external devices for use with the Biomek instrument. Each external device may contain different setup options, such as the number and type of labware the device is capable of holding. In order to use the device, this information must be provided through the **Device Editor**.

**Note:** For specific operating instructions for external devices, refer to the user's manual for each specific device.

To open the Device Editor:

Choose Instrument>Device Editor. Device Editor appears (Figure 22-3).

Device Editor									
Select a device to configure or control from the list of devices. Then click the button for what you want to do.									
Device: Barco	Device: BarcodeReader								
Configuration Options	Action Commands Light Curtain Access								
	Close								

Figure 22-3. Device Editor

To configure a device for use with the Biomek instrument:

- 1. In Device, select an installed device.
- 2. Choose the desired option:
  - Configuration Options configures the device to be used with the Biomek instrument. Refer to the device user's manual for more information about available options.
  - Action Commands select and manually operate individual device operations. Refer to the device user's manual for more information about available operations.
  - **Light Curtain Access** (Figure 22-4) allows light curtain violations caused by a device action to be ignored.

BarcodeReader L	ight Curtain Access	
✓ Device Crosses	s Light Curtain	
Access <u>S</u> ide:	Left	•
Access <u>E</u> xtent:	0.0	cm
OK	Cancel	

Figure 22-4. Light Curtain Access configuration

To configure Light Curtain Access:

- FX, NX Device Crosses Light Curtain is deselected by default. When selected, light curtain violation parameters must be specified. Light curtain violations occurring within the specified parameters are ignored; violations occuring outside the parameters stop the method run or action in progress.
- > **3000** does not have a light curtain.
- 1. Select Device Crosses Light Curtain to enable Light Curtain Access.
- 2. In Access Side, choose the side of the deck where the device action passes through the light curtain: Left, Center, Right.
- 3. In Access Extent, enter the extent in centimeters that the device passes through the light curtain.
- 4. Choose **OK** to close Light Curtain Access and save changes.

OR

Choose **Cancel** to close Light Curtain Access without saving changes. Device Editor appears.

# 22.5 Device Setup Step

The Device Setup step is used to configure the labware contained in multiple slot external devices, such as a carousel (Figure 22-5). A Device Setup step only configures one device, so multiple Device Setup steps must be used when multiple external devices are used.

Device Setup is used to select the type of labware contained in each slot in the device, and it allows the Labware Properties to be set for each piece of labware in the device.

**Note:** The software included with an external device must be installed before the device can be configured in the **Device Setup** step and accessed by the Biomek instrument.



Insert a Device Setup step into the Method View (Figure 22-5).



Figure 22-5. Device Setup step and configuration

### 22.5.1 Configuring the Device Setup Step

When a Device Setup step is inserted into a method, the Device Setup Step Configuration appears (Figure 22-5). The Device Setup Step Configuration includes:

- Configure Selection displays the Labware Properties for the selected piece of labware.
- Delete removes labware from the selected slot.
- Add Lid communicates to the software that a lid is included with a piece of labware, such as a tip box or microplate with a lid. When Add Lid is applied to any piece of labware, an asterisk appears next to the labware indicating that a lid is present.
- Remove Lid communicates to the software that a lid has been removed from the labware. When Remove Lid is applied to any piece of labware with an asterisk next to its name, the asterisk disappears indicating that a lid is no longer present.

To configure the Device Setup step:

- Select the **Device** from the options listed. The configuration for the device is displayed in Step Configuration.
- 2. Select a Labware Category.
- 3. Select a position, or multiple positions, within the configuration of the device to populate with labware.
- 4. Select a piece of labware from the graphical labware display.

OR

Drag and drop a piece of labware from the graphical labware display to the desired position in the device.

- Right-click a piece of labware in the Step Configuration and select Configure Selection. Labware Properties appears for the type of labware being configured (Figure 22-6 and Figure 22-7). Use Labware Properties to configure each piece of labware within the device (refer to Section 22.5.1.1, <u>Configuring Labware Properties for Tips and Labware</u>).
- 6. Choose **OK** to save Labware Properties and return to the Device Setup Step Configuration.
- 7. Use **Add Lid** and **Remove Lid** as necessary to configure the lid status for the labware in the device.

**Note:** When Add Lid is selected, it indicates that a lid is included with a piece of labware. An asterisk appears next to the labware in the graphical representation of the stack (Figure 22-5).

When **Remove Lids** is used on a piece of labware preceded by an asterisk, the asterisk disappears indicating that the lids are not present.

# 22.5.1.1 Configuring Labware Properties for Tips and Labware

Specific tips and labware used in a method can be configured using Labware Properties. The information provided in Labware Properties is used when a pipetting technique is selected, and when tips are loaded and unloaded.

To access Labware Properties:

Double-click on a piece of labware in the Device Setup Step Configuration.

OR

Choose Configure Selection.

OR

Right-click on a piece of labware in the Device Setup Step Configuration, and choose **Configure Selection**. Labware Properties appears (Figure 22-6).

Labware Properties	
Name:	Labware Type: AP96_200uL
Bar Code:	
When empty, send to: Home>	<tipbox></tipbox>
Load no more than 1 time	
$\nabla$ Show Available Tips	
	OK Cancel

Figure 22-6. Labware Properties for tips

#### 22.5.1.1.1 Configuring Labware Properties for Tips

When the labware selected for configuration is tip box, Labware Properties provides the ability to configure (Figure 22-6):

- Name assigned to the tip box
- Bar code
- Where to dispose of tips
- Where to dispose of tip boxes
- Maximum number of times tips can be loaded
- Available tips

**Note:** Variables and expressions may entered in any field that can be configured. Refer to Chapter 13, <u>Using Variables and Expressions in a Method</u>, for more information on using variables and expressions. To configure Labware Properties:

- 1. In Labware Type, verify the tip type.
- 2. In Name, enter name for the tip box, if applicable.

OR

Leave Name blank. When Name is blank, tips are accessed by the information displayed in the Labware Type field.

**Note:** Naming tip boxes forces the instrument to look for a specific box of tips, rather than any tip box containing the specified tip type, located anywhere on the deck. Multiple tip boxes can be given the same name. This is used to create a pool of tip boxes for a specific use.

3. In Bar Code, enter the bar code.

**Note:** Use the bar code field to identify a specific tip box in certain methods, such as plate replication. This field may be left blank.

- 4. In When empty, send to, select a final destination for the tip box at the end of the method.
- 5. In Unload Tips Into, select the location reserved for tip disposal.
- 6. In Load no more than, enter the number of times tips can be loaded onto the pod during the method.
- 7. Choose Show Available Tips to display a graphic of the tip box (Figure 22-7).

Labw	are Pro	opertie	25									
<u>N</u> ame	:											Labware Type: AP96_200uL
<u>B</u> ar C	Iode:											
<u>₩</u> her	n empty	, send	to:	ome>				-	Unloa	d Tips	Into:	<tipbox></tipbox>
Load	n <u>o</u> mor	e than	1			- -	time					
≜ Hi	de Avail	lable Tip	s									
	1	2	3	4	5	6	7	8	9	10	11	12
А		0		0	0	0			0			
В	Õ	Õ	Õ	ŏ	Õ	Õ	ŏ	Õ	Õ	ŏ	Õ	Ŏ
С	õ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ē	Ö
D	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ē	Ö
E	õ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ē	Ö
F	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ě	Ö
G	õ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ē	Ŏ
н	Ó	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ŏ	ē	Ŏ
96 usa	able tips											
												OK Cancel

Figure 22-7. Labware Properties for tips with available tips shown

8. On the graphic of the tip box, select the usable tips. This can be used to indicate missing tips or tips that are otherwise not used in the method. By default, all tips are selected.

- 9. Choose Hide Available Tips to collapse the graphic of the tipbox.
- 10. Choose **OK** to save Labware Properties and return to the Device Setup step configuration.

#### 22.5.1.1.2 Configuring Labware Properties for Labware

For labware other than tip boxes, Labware Properties provides the ability to configure (Figure 22-8):

- Name assigned to the labware
- Bar code
- Maximum volume per well
- Volume for each well, if known
- Liquid type

**Note:** Variables and expressions may entered in any field that can be configured. Refer to Chapter 13, <u>Using Variables and Expressions in a Method</u>, for more information on using variables and expressions.

Labware Properties			
Name:	Labware Type: BCFlat96	Maximum Volum	e: 362.76 µL
Bar Code:			
Labware contains a Known 💌 volume: 0	μL of liquid type:		•
• Sense the liquid level the first time a well with	Unknown or Nominal volume is accessed "from	n the Liquid".	
O Sense the liquid level every time a well is acce	ssed "from the Liquid".		
▼ Show Labware Volumes			
		ок	Cancel



To configure Labware Properties:

- 1. Verify the Labware Type.
- 2. In Name, enter a name for the labware.

**Note:** When a deck is populated by numerous pieces of labware, naming labware is recommended. Names should be descriptive of the contents of the labware or the work being accomplished during the method. Naming labware in a meaningful fashion may reduce confusion.

3. In Bar Code, enter the bar code.

**Note:** Use the bar code field to identify a specific plate in certain methods, such as plate replication. This field may be left blank.

- 4. Make a selection in Labware contains. Options are Unknown, Nominal, and Known. This information is used by many of the techniques supplied with Biomek Software which aspirate or dispense liquid a certain offset from the liquid level. Some pods or tools can detect the liquid level using liquid-level sensing technology.
  - If Known is selected, the liquid level is not detected during method run and the entered value is used during validation and method run. Known Volume should be supplied whenever possible.
  - If Unknown is selected, the liquid level is detected during method run if required by the technique, and the wells are assumed to be full when validating the method.
  - If Nominal is selected, the liquid level is detected during method run, but the volume in the wells is assumed to be the entered value when validating the method.
- 5. Enter the **Volume**, if **Nominal** or **Known Volume** is selected. A value entered in Volume is assigned to all wells.
- 6. Select the Liquid Type contained in the labware. The liquid type is useful information when Biomek Software auto-selects a technique for any aspirate and dispense operations acted upon this piece of labware. The technique auto-selected to aspirate and dispense the liquid is selected based on the physical factors of the liquid, as well as the physical attributes of the labware. For more information on liquid types, refer to Chapter 8, <u>Understanding and Creating Liquid Types</u>.
- 7. Select Sense the liquid level the first time a well with Unknown or Nominal volume is accessed "from the Liquid" to use liquid level sensing to determine the liquid level only the first time it accesses a well with an Unknown or Nominal volume from the liquid. Any instances the same well on the piece of labware is accessed, the liquid level is calculated internally based on the result of the earlier liquid level sense and the amount aspirated or dispensed to the well in previous steps.

OR

Select **Sense the liquid level every time a well is accessed "from the Liquid"** to use liquid level sensing to determine the liquid level every time it accesses a well with an Unknown or Nominal volume from the liquid.

**Note:** Liquid level sensing options only apply if the pod or tool accessing the labware is capable of sensing the liquid level. If not, the option selected is ignored.

8. If the labware does not contain uniform volumes, choose **Show Labware Volumes** to display a graphic of the labware (Figure 22-9).

**Note:** If Known Volume or Nominal Volume was selected in **Labware contains a**, Show Labware Volumes activates.

Labware Properties									
Name: BCFlat96 Maximum Volume: 362.76 µL									
Bar Code:									
Labware contains a Known 💌 volume: 0 👘 ul of liquid type:									
€ Sense the liquid level the first time a well with Unknown or Nominal volume is accessed "from the Liquid".									
C Sense the liquid level every time a well is accessed "from the Liquid".									
A Hide Labware Volumes									
1 2 3 4 5 6 7 8 9 10 11 12 Volume									
A 000000000000									
96 selected wells.									
OK Cancel									

Figure 22-9. Labware Properties for 96-well microplate Labware Volumes shown

9. Select the desired wells on the graphic.

**Note:** Hold down **Ctrl** to toggle wells between selected and deselected without affecting the selection status of other wells.

- 10. Enter the volume to assign to the currently selected wells in **Amount**.
- 11. Choose **Set** to assign the current Amount to the currently selected wells. The wells are given a color code that corresponds to the list of set volumes under Volume.

Note: Clear sets the volume for all wells back to zero.

- 12. Repeat steps 10-12 until all desired wells have been assigned the desired volume.
- 13. Choose Hide Labware Volumes to collapse the graphic of the labware.
- 14. Choose **OK** to save Labware Properties and return to the Device Setup step configuration.

# 22.6 Device Action Step

The Device Action step is used to configure the actions of active ALPs and devices.

- FX controls actions, such as operating the wash pump for the Wash ALP, stirring actions for the Stirring ALP, shaking actions for the Microplate Shaking ALP, and positioning actions for the Positive Position ALP.
- 3000 controls devices, such as the autolatch, 6-port valve, vacuum, Wash System, and MicroMix.
- NX controls devices and active ALPs, such as the Source/Waste Sensor and Drainable/Refillable Reservoir ALP.

**Note:** Any devices installed in Hardware Setup, except Tip Loaders and Device Controllers, are listed under Device in the Device Action Step Configuration.



Insert a **Device Action** step into the Method View (Figure 22-10).



Figure 22-10. Device Action step and configuration

### 22.6.1 Configuring the Device Action Step

When Device Action is added to a method, the associated configuration appears.

The configuration options available for the **Device Action** step depend upon the device selected for configuration.

The Device Action step is used to control the following ALPs and devices:

- <u>Configuring the Device Action Step for 6-Port Valve (3000 only)</u> (Section 22.6.1.1).
- <u>Configuring the Device Action Step for an Autolatch (3000 only)</u> (Section 22.6.1.2).
- <u>Configuring the Device Action Step for a Drainable/Refillable Reservoir</u> <u>ALP (FX, NX only)</u> (Section 22.6.1.3).
- <u>Configuring the Device Action Step for the MicroMix (3000 only)</u> (Section 22.6.1.4).
- <u>Configuring the Device Action Step for the Microplate Shaking ALP (FX,</u> <u>NX only)</u> (Section 22.6.1.5).
- <u>Configuring the Device Action Step for a Multichannel Tip Wash ALP</u> (FX, NX-MC only) (Section 22.6.1.6).
- <u>Configuring the Device Action Step for a Positive Position ALP (FX, NX only)</u> (Section 22.6.1.7).
- <u>Configuring the Device Action Step for a Source/Waste Sensor (FX and NX only)</u> (Section 22.6.1.8)
- <u>Configuring the Device Action Step for a Speed Pump (FX only)</u> (Section 22.6.1.9).
- <u>Configuring the Device Action Step for the Stirring ALP (FX. NX only)</u> (Section 22.6.1.10).
- <u>Configuring the Device Action Step for the Vacuum (3000 only)</u> (Section 22.6.1.11).
- <u>Configuring the Device Action Step for the Wash System (3000 only)</u> (Section 22.6.1.12).

# 22.6.1.1 Configuring the Device Action Step for 6-Port Valve (3000 only)

To configure the Device Action step for the 6-port valve:

1. From **Device**, select the 6-port valve performing the required operation.

Note: Biomek Software refers to the 6-port valve as SixPortValve.

OR

2. Click on the 6-port valve in the Current Deck Display to select it. The configuration for the 6-port valve appears (Figure 22-11).



Figure 22-11. Device Action step for the 6-port valve

- 3. In **Command**, choose:
  - Select Wash Liquid to enter the liquid number in Liquid Number.
  - Device Setup to select the liquid type for the six ports from the dropdown menus. Figure 22-12 appears.

(h Biomek) File Edit	® Software Project Inst	- Method1* [New] rument Execution Ontions Held						_ <b>_</b> _×
	n 🖬 🖪							
aze.	E	Start	De	evice:	SixPortValve 💌			
Instrument	Device	🥷 Instrument Setup	Cor	mmand:	Device Setup			
Setup	Setup	SixPortValve Device	Setup Lig	quid 1	Water 💌			
Transfer	Device	🖁 Finish	Liq	quid 2	Water			
-	Action	_	Liqu	quid 3	Water			
Combine	SPE		Liq	quid 4	Water 💌			
			Liq	quid 5	Water 💌			
Move Labware			Liq	quid 6	Water 💌			
Pause								
Q								
Comment								
						œ		
					P8	P1 P2	P3	MM1
							D7	NID IO
						P6		
			3	8 I I				
Method1*	Biomek3000	Biomek3000 ETC: 0:00:03						

Figure 22-12. Device Action step for the 6-port valve with Device Setup chosen

# 22.6.1.2 Configuring the Device Action Step for an Autolatch (3000 only)

To configure the Device Action step for the Auto-Latching Tip Rack Holder or autolatch:

1. From **Device**, select the autolatch performing the required operation.

OR

Click on the autolatch in the Current Deck Display to select it. The configuration for the autolatch appears (Figure 22-13).

**Note:** The deck position has a blue border in the Current Deck Display and is displayed to the right of **Device**.



Figure 22-13. Device Action step for the autolatch

2. In **Command**, indicate the desired state for the autolatch: Open or Close.

### 22.6.1.3 Configuring the Device Action Step for a Drainable/Refillable Reservoir ALP (FX, NX only)

To configure the Device Action step for the Drainable/Refillable Reservoir ALP:

1. From **Device**, select the Drainable/Refillable Reservoir ALP performing the operation.

**Note:** Biomek Software refers to the ALP as DrainableRefillableReservoir.

OR

Click on the desired ALP in the Current Deck Display. The deck position is highlighted by a blue border and named to the right of **Device**. The configuration for the ALP appears (Figure 22-14).

🗄 Biomek	® Software	- DrainRes	01* [D	evelopment]						
<u>File E</u> dit	Project Ins	trument E <u>x</u> e	ecution	Options Help		. 0.				
	i R   B.	. <b>€</b>   % [	b 🖻	∽ ~ Ø  ► II	日米	Berkele	ey Rattan 🕵			
*			0	Start		Device:	DrainableRefillableReservoir0	▼ P1		
SPE	Aspirate	M Run Method	The second secon	Instrument Setup		Command:	Fill	•		
E	*		1	Fill Reservoir						
[15] Device			T	Aspirate from P1						
Setup	Dispense	Workiist	X	Dispense to P7						
Devrice	æ	Lust In	12	Einich						
Action	Mix	Time								
1111	43	Σ¢Ι								
Tip Loader	Wash	Let								
all a		F								
Instrument Setup	New Tips	If								
(CL	9999	ł								
Transfer	Unload Tips	Script								
-	0	<u>S</u>								
Combine	Loop	Scripted Let								
						()i				 
Move	Cleanup	Define								
Labware	Ĩ	Pattern								
8	Move Pod							P4	P10	
Pause								P5	P11	
Q	Group					N.		P3 P6	P9 P12	
Comment		1	•		Þ	N.				
DrainRes01	I* BiomekN	K MC Biome	kNX-MC	ETC: 0:00:03						

Figure 22-14. Device Action step for the Drainable/Refillable Reservoir ALP

2. In Command, choose the operation performed by the ALP:

- **Fill** fills the reservoir to the maximum level.
  - **Drain** drains the reservoir.
  - **AssertSensor** has the sensor check for the presence of liquid in the reservoir. Sensor State appears in the configuration.
  - **AutoFill** keeps the reservoir supplied with liquid during the method run. AutoFill should be appears in the configuration.

- 3. If AssertSensor is selected, in Sensor State, choose
  - **On** allows the method run to proceed when sufficient liquid is detected in the reservoir.
  - Off allows the method run to proceed when liquid is not detected in the reservoir.
- 4. If AutoFill is the selected Command, in AutoFill should be, choose:
  - **On** turns AutoFill on for the duration of the method run.
  - Off turns AutoFill off. Liquid remains in the reservoir.

**Note:** To turn off AutoFill and drain the reservoir, insert a Device Action step configured to Drain the reservoir. Draining the reservoir is useful for reducing evaporation during a long pause in a method run. For example, insert a Device Action step configured to Drain the reservoir right before the pause. When the pause ends, insert a Device Action step configured to turn AutoFill On. The reservoir will be supplied with liquid during subsequent pipetting operations in the method run.

# 22.6.1.4 Configuring the Device Action Step for the MicroMix (3000 only)

To configure the Device Action step for the MicroMix:

1. From **Device**, select the MicroMix performing the required operation.

OR

Click on the MicroMix in the Current Deck Display to select it. The configuration for the MicroMix appears.

**Note:** The deck position has a blue border in the Current Deck Display and is displayed to the right of **Device**.

- 2. In **Command**, choose:
  - Shake to configure a shake operation (refer to Section 22.6.1.4.1, <u>Configuring Shake</u>).
  - Custom Shake to configure a custom shake operation (refer to Section 22.6.1.4.2, <u>Configuring Custom Shake</u>).
  - On to turn on the MicroMix
  - Off to turn off the MicroMix.

### 22.6.1.4.1 Configuring Shake

To configure a shake operation for the MicroMix:

1. In **Command**, choose Shake.

Biomek® Software - Method1* [New]		
File Edit Project Instrument Execution Options Help		
	ſ	
Start 👔	Device:	MicroMix1 MM2, MM1
Instrument Setup	Command:	Shake
Setup Setup T MicroMix1 Shake	Amplitude	5
	Amplitude	
Transfer Device Action	Duration	5 minutes
all a	Form	1
Combine SPE		
Move		
Lauware 10		
Pause		
Comment		
	ETT A	
		P6 P7 MM2
	l â l	
Method1* Biomek3000 Biomek3000 ETC: 0:00:03		

Figure 22-15. Device Action step for the MicroMix with Shake chosen

2. In **Amplitude**, enter the magnitude of power.

**Note:** The selection for Amplitude ranges from 1 to 9; the minimum amplitude is 1 and the maximum is 9.

- 3. In **Duration**, enter the time in minutes.
- 4. In **Form**, enter the form number.

**Note:** Sixty Forms (preset shaking cycle patterns) are available. See the *Appendix* in the *MicroMix 5 Shaker User's Manual* for a list of the available forms.

#### 22.6.1.4.2 Configuring Custom Shake

To configure a custom shake operation:

1. In **Command**, choose Custom Shake (Figure 22-16).

Biomek® Software - Method1* [New]     Ela Edit Project Instrument Evention Onlines Halp		_ 🗆 🗙
Start	Device: MicroMix1 MM2 MM1	
The Instrument Set in		
Setup Setup		
	Amplitude 5	
Transfer Device Action	Duration 5 minutes	
9 x	Mode Clockwise, consta 💌	
Combine SPE	Start Frequency 10 Hz	
	End Frequency 10 Hz	
Move Labware	Ramp Rate 1 Hz/sec	
	Gap Time 1 seconds	
Pause		
Comment		
	P8 P1 P2 P3 MM1	
Method1* Biomek3000 Biomek3000 ETC: 0:00:03		

Figure 22-16. Device Action step for the MicroMix with Custom Shake chosen

2. In **Amplitude**, enter the magnitude of power.

**Note:** The selection for Amplitude ranges from 1 to 9; the minimum amplitude is 1 and the maximum is 9.

- 3. In **Duration**, enter the time in minutes.
- 4. In **Mode**, choose one of the following:
  - Clockwise, constant direction, constant speed.
  - Counter-clockwise, constant direction, constant speed.
  - Ramp frequency up, ramp frequency down, gap, change direction.
  - Ramp frequency up, gap, change direction, ramp frequency down, gap, change direction.
5. In **Start Frequency**, enter the starting orbital speed.

Note: The allowable range of Start Frequency is 10 to 30.

6. In End Frequency, enter the ending orbital speed.

Note: The allowable range of End Frequency is 10 to 30.

7. In **Ramp rate**, enter the speed of change from the **Start Frequency** to the End Frequency.

**Note:** The allowable range of **Ramp rate** is 1 to 250.

8. In **Gap time**, enter the non-shaking period prior to change of direction in seconds.

**Note:** The allowable range of **Gap time** is 0 to 2.55.

#### 22.6.1.5 Configuring the Device Action Step for the Microplate Shaking ALP (FX, NX only)

To configure the Device Action step for the Microplate Shaking ALP (Figure 22-17):

1. From **Device**, select the Microplate Shaking ALP performing the required operation.

**Note:** The Biomek Software refers to the Microplate Shaking ALP as ShakerALP.

OR

Click on the Microplate Shaking ALP on the Current Deck Display to select it. The configuration for the Microplate Shaking ALP appears (Figure 22-17).

**Note:** The deck position has a blue border in the Current Deck Display and is displayed to the right of **Device**.

#### 22-26 Using the Devices Step Palette



Figure 22-17. Device Action step for the Microplate Shaking ALP

2. In **Command**, indicate the operation performed by the device: Shake, Timed Shake, Off, Clamp, or Unclamp.

**Note:** Shake and Timed Shake activate the Microplate Shaker ALP and require further configuration, as described below. Off deactivates the Microplate Shaker ALP and does not require further configuration. Clamp and Unclamp tighten and loosen the clamp on the Microplate Shaker ALP and do not require further configuration.

- 3. Select the **Shaking speed**, which is in the range of 1-100 indicating a percentage of the Microplate Shaking ALP's maximum shaking speed.
- 4. In **Time to reach full speed**, select how long (in seconds) it should take to reach the full shaking speed.

**Note:** The Microplate Shaking ALP stops shaking automatically when pipetting to or from the ALP and restarts when the pipetting operation is finished. The Microplate Shaking ALP stops shaking when a gripper operation is performed, but does not restart when the operation is completed. The Microplate Shaking ALP stops shaking when the Finish step is executed.

5. For a Timed Shake action only, select how long (in seconds) to shake the microplate in **Time to shake**.

#### 22.6.1.6 Configuring the Device Action Step for a Multichannel Tip Wash ALP (FX, NX-MC only)

To configure the **Device Action** step for the wash pump associated with a Multichannel Tip Wash ALP (Figure 22-18):

**Note:** A Multichannel Tip Wash ALP may be a 96-Channel or 384-Channel Tip Wash ALP.

1. From **Device**, select the wash pump associated with the Multichannel Tip Wash ALP performing the required operation.

OR

Click on the Multichannel Tip Wash ALP in the Current Deck Display to select it. The configuration for the Multichannel Tip Wash ALP appears (Figure 22-18)

**Note:** The deck position has a blue border in the Current Deck Display and is displayed to the right of **Device**.





2. In **Command**, indicate the desired state for the Multichannel Tip Wash ALP: On or Off.

**Note:** The Multichannel Tip Wash ALP stops washing when the Finish step is executed.

## 22.6.1.7 Configuring the Device Action Step for a Positive Position ALP (FX, NX only)

To configure the Device Action step for the Positive Position ALP:

1. From **Device**, select the Positive Position ALP performing the required operation.

OR

Click on the Positive Position ALP in the Current Deck Display to select it. The configuration for the Positive Position ALP appears (Figure 22-19).

**Note:** The deck position has a blue border in the Current Deck Display and is displayed to the right of **Device**.

🏥 Biomek	® Software	- Method7* [New	1										- 🗆 ×
File Edit	Project Inst	trument Execution	Options Help	-									
	s Hig	X≞∎	<b>v</b> ~ <b>0</b>   <b>v</b> = 🗉 [						-				
R	all	Start			Device:	PositivePos	itionALP0	•					
SPE	ער" Instrument	🖏 Instrum	ent Setup		Command:	Clamp		•					
E.	G	Positive	PositionALP0 Clamp			-			-				
Device	Transfer	👸 Finish											
1				De	vice A	ction	Step C	onfig	uratio	on			
Device	Combine			The	Devic	e Actio	n Step	Confi	guratio	on			
				spe	cifies a	n opera	tion ne	rforme	ed by fl	he			
Tip Loader	Move			Pos	itive Po	sition	ALP						1.00
	Labware			100		, one of the second sec	. 121 .						
	Pause												
	Comment												
		1				TL1			P8				7
						Asp1			P9	P13	w	P17	
						Disp1		P6	P10	P14		P18	
						P3		P7	P11	P15		P19	
Method7*	RiomekEV	BiomekEY ETC: 0.0	0.04										



- 2. In **Command**, indicate the desired action for the Positive Position ALP:
  - Clamp tightens the clamp to hold the labware in place
  - Unclamp releases the clamp to allow the labware to be removed from the ALP
  - VerifyLabware verifies that labware is currently on the Positive Position ALP; an error displays and the method stops if no labware is found
  - VerifyNoLabware verifies that there is currently no labware on the Positive Position ALP; an error displays and the method stops if labware is found

#### 22.6.1.8 Configuring the Device Action Step for a Source/ Waste Sensor (FX and NX only)

To configure the Device Action step for a Source/Waste Sensor:

1. From Device, select the **SourceWasteSensor** performing the operation. Configuration options for the sensor appear (Figure 22-20).

Image: Second Project   Image: Second Proje	🌵 Biomek(	🖲 Software	- DrainRest	01* [New]				_ 🗆 🗙
Start   Start   Log   Normand   Start   Claure   Normand   Aspirate from P1   Dispense to P7   Dispense to P7 <t< th=""><th>File Edit I</th><th>Project Insl</th><th>rument Exe</th><th>ecution Options Help</th><th></th><th>u Dathar 🗛</th><th></th><th></th></t<>	File Edit I	Project Insl	rument Exe	ecution Options Help		u Dathar 🗛		
Start Second Mathematical Second Association of the second Associatio			 					
SEE Log Run Muhor SEE Log Run Muhor Series State Command Series State Command Series State Com Series State Com Mive Ded Jut, In Source WasteSensorO Ass Series State Com Mive Ded Jut, In Source WasteSensorO Ass Series State Com Transfor Source WasteSensorO Ass Series State Com Transfor Source WasteSensorO Ass Series State Com Transfor Source WasteSensorO Ass Source WasteSensorO Ass Series State Com Transfor Source WasteSensorO Ass Source WasteSensorO Ass Source WasteSensorO Ass Series State Com Transfor Source WasteSensorO Ass Source Wa	~	5	d	Start	Device:	SourceWasteSensor0		
Survey Survey   Survey <t< th=""><th>SPE</th><th>Loop</th><th>Run Method</th><th>🐔 Instrument Setup</th><th>Command:</th><th>AssertSource</th><th></th><th></th></t<>	SPE	Loop	Run Method	🐔 Instrument Setup	Command:	AssertSource		
Divide Dispense to P7	E			SourceWasteSensor0 Ass	Sensor Stat	e On 💌		
Sarajo     Cleange     Worklet     Dispense to P7       Dispense to P7     Dispense to P7       Dispense     Dispense to P7       Sarajo     Dispense to P7       Dispense to P7     Dispense to P7       Sarajo     Dispense to P7       Dispense to P7 </th <th>Devrice</th> <th></th> <th></th> <th>Aspirate from P1</th> <th></th> <th></th> <th></th> <th></th>	Devrice			Aspirate from P1				
Dispense to P/   Diverbed   Dispense to P/	Setup	Cleanup	Worklist					
Device Move Pod   Actin Image: Finish	C	m	63	Dispense to P7				
Image: Spanse Spans	Device Action	Move Pod	Just In Time	📓 Finish				
Group Let   Serial Jintrument   Serial Jintrument   Serial Soipit   Soipit Soipit   Soipit Soipit   Soipit Soipit   Soipit Soipit   Soipit Soipit   Jintrument Soipit   Soipit Soipit   Soipit Soipit   Jintrument Soipit   Soipit Soipit   Jintrument Soipit   Jintrument Soipit   Soipit Soipit   Jintrument Soipit   Jintrument Soipit   Jintrument Soipit   Jintrument Soipit   Jintrument Jintrument			Σē					
Seide   Seide </th <th></th> <th>Group</th> <th>Let</th> <th></th> <th></th> <th></th> <th></th> <th></th>		Group	Let					
Instrument Series       Srid Dilution         Single Transfer       Span-8 Aspirate         Span-8 Aspirate       Scriptel Let         Sorrige       Span-8 Digense         More Labouare       Span-8 Digense         Transfer       Transfer         Pause       Transfer         Transfer       Transfer         Tra		~						
Instrument Setup       Serial Dilution       III         Image: Setup Dilution       Image: Setup Dilution       Image: Setup Dilution         Image: Setup Dilution       Image: Setup Dilution       Image: Setup Dilution         Image: Setup Dilution       Image: Setup Dilution       Image: Setup Dilution         Image: Setup Dilution       Image: Setup Dilution       Image: Setup Dilution         Image: Setup Dilution       Image: Setup Dilution       Image: Setup Dilution         Image: Setup Dilution       Image: Setup Dilution       Image: Setup Dilution         Image: Setup Dilution       Image: Setup Dilution       Image: Setup Dilution         Image: Define Dilution       Image: Setup Dilution       Image: Setup Dilution         Image	The second							
Image: Stand Approx       Sop         Sop       Sop         Combine       Spand         Dispense       Sope         Move       Spand         Sope       Sope         Dispense       Image: Sope         Pause       From File         Pattern       Image: Sope         Define       Image: Sope         Define       Image: Sope         Pattern       Image: Sope	Instrument Setup	Serial Dilution						
Transfer Span-8 Aspirate Somble Combine Disperse Nove Labware Wash Tips Pause Pause Pause From File Define Pause Comment Define Pause From File Pause Transfer Pause Pause Transfer	ß	Ŷ						
Image: Spanse Sorped Lat         Some Spanse Sorped Lat         Image: Spanse Labware Wash Tips         Image: Pause From File         Image: Pa	Transfer	Span-8	Script					
Combine Span 8 Sciped Let Move Labware Wash Tips Pause Pause Pause Pause Comment Define Pattern Nove Labware Paus		Aspirate	(Sh					
Combine     Disperse       Move     Span-8       Labware     Wash Tips       Pause     From File       Pause     From File       Ocriment     Define       Define     Define		∬ Snan-8	Scripted Let					
Image: Span P       Move: Span P       Span P       Pause       Frankier       Frankier       Pause	Combine	Dispense						
Move Labware Wash Tips     Span-8 Wash Tips       Image: Span-8 Wash Tips     Image: Span-8 Wash Tips <tr< th=""><th><math>\rightarrow</math></th><th>പ്പ്</th><th></th><th></th><th></th><th></th><th>TR1</th><th></th></tr<>	$\rightarrow$	പ്പ്					TR1	
Pause     Transfer       From File     P10       Omment     P5       P11     P3       P3 P6     P9	Move Labware	Span-8 Wash Tips						
Pause Transfer From File Comment Define Pattern I	8						P4 P10	
Comment Pattern Comment	Pause	Transfer Erom Eile						
Comment Dafine P3 P6 P9 P12								
Volument     Pattern     Image: Construction of the		Define					P3 P6 P9 P12	
	Comment	Pattern						
DrainRes01* BiomekNX MC BiomekNX-Span ETC: 0:00:03	DrainRes01	* BiomekN	K MC Biomek	NX-Span ETC: 0:00:03				

Figure 22-20. Device Action for a Source/Waste Sensor

- 2. In Command, select the operation to perform:
  - **AssertSource** has the source container sensor check for the presence of liquid.
  - **AssertWaste** has the waste container sensor check for the presence of liquid.
- 3. In Sensor State, set the sensor action:
  - **On** allows the method run to continue if sufficient liquid is detected in the source container, or if the liquid level in the waste container is sufficiently low.
  - Off allows the method run to continue when insufficient liquid is detected in the source container, or if the waste container is filled to the maximum level allowed.

## 22.6.1.9 Configuring the Device Action Step for a Speed Pump (FX only)

The Speed Pump is used in conjunction with the Span-8 Wash ALP. To configure the **Device Action** step for the Speed Pump:

1. From **Device**, select the Speed Pump performing the required operation.

OR

Click on the Span-8 Wash ALP in the Current Deck Display to select it. The configuration for the Speed Pump appears (Figure 22-21).

**Note:** The deck position has a blue border in the Current Deck Display and is displayed to the right of **Device**.

Biomek® Software - Method7* [New]     File Edit Project Instrument Execution Options Help		
Start	Device: SpeedPump0	
SPE Instrument Setup	Command: On	
SpeedPump0 On		
Device Setup Transfer		
	Device Action Step Configuration	
Action Combine	The Device Action Step Configuration	
Tin Loader Move	specifies an operation performed by a	
Labware	Speed I ump.	
Pause		
Comment		
	Asp1 P9 P13 W P17	
	P6 P10 P14 P18	
	P3 P7 P11 P15 P19	
Method7* BiomekFX BiomekFX ETC: 0:00:04		

Figure 22-21. Device Action step for the Speed Pump

2. In **Command**, indicate the desired state for the Wash Station ALP: On or Off.

Note: The Speed Pump stops when the Finish step is executed.

## 22.6.1.10 Configuring the Device Action Step for the Stirring ALP (FX, NX only)



CAUTION: When setting the height for pipetting operations on the Stirring ALP, the presence of the stir bar must be taken into consideration. Pipetting operations that do not account for the height of the stir bar could damage the tips.

To configure the Device Action step for the Stirring ALP (Figure 22-22):

1. From **Device**, select the Stirring ALP performing the required operation.

Note: Biomek Software refers to the Stirring ALP as StirrerALP.

OR

Click on the Stirring ALP on the Current Deck Display to select it. The configuration for the Stirring ALP appears (Figure 22-22).

**Note:** The deck position has a blue border in the Current Deck Display and is displayed to the right of **Device**.



Figure 22-22. Device Action step for the Stirring ALP

2. In **Command**, indicate the operation performed by the device: Stir or Off.

**Note:** The Stir command requires further configuration, as described below. Off deactivates the Stirring ALP and does not require further configuration.

3. Select the **Stirring speed**, which is in the range of 1-100 indicating a percentage of the Stirring ALP's maximum stirring speed.

- 4. In **Time to reach full speed**, select how long (in seconds) it should take to reach the full stirring speed.
- 5. In **Stop for Pipetting?**, select whether or not to stop the stirring operation for pipetting. If **Yes** is selected, when performing pipetting operations to labware on the Stirring ALP it stops stirring while the tips access the labware and resumes stirring when finished.

**Note:** The Stirring ALP stops stirring when the Finish step is executed.

## 22.6.1.11 Configuring the Device Action Step for the Vacuum (3000 only)

To configure the Device Action step for the vacuum:

1. From **Device**, select the vacuum performing the required operation.

OR

Click on the vacuum in the Current Deck Display to select it. The configuration for the vacuum appears (Figure 22-23).

**Note:** The deck position has a blue border in the Current Deck Display and is displayed to the right of **Device**.

Biomek® Software - Method1* [New]         File Edit Project Instrument Execution Options Help         Instrument         Device         Setup         Instrument Setup         Instrument Setup         Instrument Setup         Endition         Device         Action         Setup         Endition         Setup         Instrument Setup         Endition         Device         Action         Setup         Combine         Spe	Device: Vacuum P Command: Close
Move Labware Pause	Device Action Step Configuration The Device Action Step Configuration specifies an operation performed by the vacuum.
	P8 P1 P2 P3 MM1 P6 P7 MM2
Method1* Biomek3000 Biomek3000 ETC: 0:00:03	

Figure 22-23. Device Action step for the vacuum

- 2. In **Command**, choose:
  - Close to close the vacuum.
  - Open to open the vacuum.
  - Timed Vacuum to select the vacuum time in seconds. Figure 22-24 appears.

👫 Biomek® Software - Method1* [New]			٦×
File Edit Project Instrument Execution Options Help			
	ſ		
Start	Device:	Vacuum 🔽 P4	
Instrument Setup	Command:	Timed Vacuum	
Vacuum Close	for	1 8	
Transfer Device Artino			
<b>3</b>			
Combine SPE			
Move Labware			
12			
Pause			
Comment			
		P8 P1 P2 P3 MM1	
	<u>}</u>		
Method1* Biomek3000 Biomek3000 ETC: 0:00:03			

Figure 22-24. Device Action for the vacuum with Timed Vacuum chosen

## 22.6.1.12 Configuring the Device Action Step for the Wash System (3000 only)

To configure the Device Action step for the Wash System:

1. From **Device**, select the Wash System performing the required operation.

OR

Click on the Wash System in the Current Deck Display to select it. The configuration for the Wash System appears (Figure 22-25).

**Note:** The deck position has a blue border in the Current Deck Display and is displayed to the right of **Device**.



Figure 22-25. Device Action for the Wash System

- 2. In **Command**, choose:
  - InitializeWash to initialize the wash.
  - Latch to open or close the latch on the wash by choosing **Open** or **Close** from **Direction** (Figure 22-26).
  - Aspirate to aspirate the wash liquid by entering **Speed** and **Volume** (Figure 22-27).
  - Dispense to dispense the wash liquid by entering **Speed** and **Volume** (Figure 22-27).

🔅 Biomek	® Software	e - Met	hod1* [New]	]									_ 🗆 🗙
File Edit	Project In:	strument	Execution	Options Help									
1000	r H I		8 B B	≌ ¤ Ø   ▶ [									
æ	E.	8	Start		Device:	Wash		•					
42	A SI	æ	Instrume	nt Setun	Command	Latch		<b>_</b>					
Setup	Device Setup	143	iniod dime		command.								
R	1	0	Wash Lat	tch	Direction	Open	-						
Transfer	Device	000	Finish										
-	Action												
Combine													
	DPE												
Move													
Labware													
Pause													
Comment													
	]												
						_			asha				
						_	P8		3 P1	P2	P3	MM1	
								******					
									010100100100	P6	P7	MM2	
					No.								
		-			 r								=
Method1*	Biomek300	Biom	ek3000 ETC:	0:00:03									

Figure 22-26. Device Action for the Wash System with Latch chosen

### 22-36 Using the Devices Step Palette

Biomek® Software - Method1* [New] File Edit Project Instrument Execution Options Help		
Instrument       Start         Instrument       Setup         Instrument       Wash Aspirate         Image: Spe       Spe         Move       Spe         Labware       Spe         Image: Spe       Spe         Combine       Spe         Pause       Spe         Comment       Spe	Device: Vash	
	P8 P1 P2 P3 MM1	

Figure 22-27. Device Action for the Wash System with Aspirate chosen

## 22.7 Tip Loader Step (FX, NX-MC only)

The Tip Loader step provides more control over a Tip Loader ALP than other steps using the Tip Loader ALP, such as New Tips and Unload Tips (Figure 22-28).

FX — the Tip Loader is used only for loading and unloading tips to a Multichannel Pod.

The Tip Loader step is used when precise control over tip loading and unloading is required, such as when performing multiple aspirate and dispense operations with each operation requiring a specific set of tips.

**Note:** A usable tip box must be present on the Tip Loader ALP before using the Tip Loader step. The **Tip Loader** step does not automatically locate a tip box prior to performing an operation.



👘 Biomek® Software - Method7\* [New] <u>\_ | | ×</u> Project Instrument Execution Options He 8 Start Load • and the second s <u>T</u>ips Z. TL1 • 🐔 Instrument Setup Location Instrume Setup SPE Pod1 • <u>P</u>od Load tips from TL1 E R Device Setup Device Action Finish Transfe -**Tip Loader Step Configuration** Combine The Tip Loader Step Configuration Tip Loade indicates the operation performed by Move Labwan the Tip Loader, the deck location of the Tip Loader used, and the pod used with Pause the Tip Loader. Q Commen Method7\* BiomekFX BiomekFX ETC: 0:00:04

Insert a Tip Loader step into the Method View (Figure 22-28).

Figure 22-28. Tip Loader step and configuration

### 22.7.1 Configuring the Tip Loader Step

When the **Tip Loader** step is added to a method, the associated configuration appears (Figure 22-28).

To configure Tip Loader:

1. In Tips, specify whether to Load or Unload tips.

**Note:** If the Tips field indicates that a load is performed, the Tip Loader caption in Method View indicates the operation with "Load 96 Tips from …"; however, if Unload is requested, the caption reads, "Unload 96 Tips from …".

- 2. Specify the **Location** of the Tip Loader on the deck.
- 3. Specify the **Pod** performing the tip load operation from the drop-down list.

OR

Select the **Pod** from the Current Deck Display. The tip diagram on the left of the Current Deck Display represents **Pod1**, while the tip on the right represents **Pod2**. Selecting a pod from the Current Deck Display automatically updates **Pod** in the Step Configuration.

FX — The pod can be selected from the Current Deck Display only if the instrument has two Multichannel Pods; Span-8 Pod cannot be used with this step.

**Note:** In Pod, Pod1 is the default for a single-pod system. In a dual-pod system, the pod configured as the default pod is initially displayed in Pod.

## 22.8 SILAS Module Step

Any external device requiring a SILAS module for operation during a method is configured using a SILAS step. SILAS steps operate devices on a Biomek deck during a method run by communicating between Biomek Software and the SILAS modules.

### 22.8.1 Configuring the SILAS Module

**Note:** Before inserting a SILAS step into a method, it must be placed on a step palette. Refer to Chapter 29, <u>Changing Window Appearance</u>, for more information.

When configuring a SILAS Module, any required labware movement must be known and the appropriate selection must be made. All SILAS modules for external devices are configured using the same procedure, but some devices may have additional fields to configure.

To configure a SILAS Module Step:

1. Insert a SILAS Module Step into the method in the Method View (Figure 22-29).

**Note:** The SILAS Module Step is labeled according to the device it represents; for example, the SILAS Module Step for the barcode reader is labeled as a Barcode Reader step.





2. Select **Configure Action**. Configure the operation of the device as desired.

- 3. Select **Update** after completing the action configuration.
- 4. Select the appropriate Labware movement.

Labware options include:

- Retrieving From Module retrieves labware from the device
- Sending To Module sends labware from the deck to the device
- No Change labware remains in the current position after operations are completed
- 5. Select **Crosses Light Curtain** if the action taken by the device causes something to cross the light curtain. If selected, the instrument disables the light curtain while this command is executing.
  - > **3000** This instrument does not have a light curtain.

**Note:** Refer to Section 22.4.1, <u>*Configuring Devices Using the Device Editor*</u>, to configure the Crosses Light Curtain option.

 Select Wait for Runtime Data, if required. Wait for Runtime Data allows Biomek Software to use data resulting from an action to make decisions on operations. Refer to Section 22.8.2, *Configuring Runtime Data*, for more information.

**Note:** When Wait for Runtime Data is checked, the Biomek instrument will not validate beyond this step until it is completed.

### 22.8.2 Configuring Runtime Data

Runtime Data allows Biomek Software to read specific data about labware and to perform operations that meet specific criteria during a method. For example, the results of a bar code read can determine the specific operations performed on the labware using an lf step.

**Note:** Information entered in Runtime Data Defaults during method validation is used because real data has not yet been obtained by Biomek Software.

To configure and use Runtime Data:

1. Select Simulated Data. Runtime Data Defaults appears (Figure 22-30).

Runtime Data Defaults								
Set Labware Barcode								
Random Number								
C Specific String: B	arcode							
Other Properties:								
Property	Default Value							
ОК	Cancel							

Figure 22-30. Runtime Data Defaults

- 2. Choose Set Labware Barcode to provide data for future decisions.
- 3. Choose one of the following:
  - Random Number sets the Barcode property to a random number
  - **Specific String** sets the Barcode property to the string indicated
- 4. In the **Property** field, enter the name of the property to set.
- 5. Enter the **Default Value** for the property when the property is not present for the labware.
- 6. Choose **OK** when configuration is complete.

### 22.8.3 Creating Data Sets from SILAS Messages

Data sets can be created from messages sent by a SILAS step for a device. For example, the results of a measurement from a microplate reader can be stored in a data set created from the SILAS step.

To create a data set from a SILAS message using a SILAS step:

1. Insert the desired SILAS step into the Method View.



Figure 22-31. SILAS step with data sets chosen

- 2. Select **Configure Action** and configure the action of the device as desired.
- Select Wait for Runtime Data, if required. Wait for Runtime Data allows Biomek Software to use data resulting from an action to make decisions on operations (refer to Section 22.8.2, *Configuring Runtime Data*).

**Note:** When Wait for Runtime Data is checked, the Biomek instrument will not validate beyond this step until the step is completed.

- 4. Select **Generate Data Set From Runtime Data** to create a data set from the results of the configured action.
- 5. In Data Set Name, enter a name for the data set.

**Note:** If the configured action produces more than one result per well, such as a multiple wavelength or kinetic read, a data set is created for each result and the Data Set Name is appended with an incremental number; for example, DataSet1, DataSet2, DataSet3, and so forth.

- 6. Select **The data in the new Data Set should be tracked during pipetting** to allow the system to track the data during pipetting operations.
- 7. Choose **Simulated Data**. Runtime Data Defaults appears (Figure 22-32).

**Note:** Information entered in Runtime Data Defaults during method validation is used because real data has not yet been obtained by Biomek Software (refer to Section 22.8.2, *Configuring Runtime Data*).

Runtime Data Defaults								
🔽 Set Labware Bar Code	☑ Set Labware Bar Code							
Random Number								
C Specific String: B	arcode							
Other Properties:								
Property	Default Value							
1								
OK	Cancel							

Figure 22-32. Runtime Data Defaults

- 8. In Property, enter the Data Set Name.
- 9. In Default Value, enter a default value to use for the data set during method validation. This value is used for all wells on the plate.

**Note:** At run time, the actual values of the data set will be used. The Default Value is only used for method validation.

# Using the Stacker Carousel Step Palette

## 23.1 Overview

Steps on the Stacker Carousel Step Palette allow labware to be moved to and from the Stacker Carousel and the Biomek deck. The Stacker Carousel and steps expand labware capacity and increase walk-away automation during a method.

**Note:** Refer to the hardware manual for the specific instrument for detailed information on integrating and framing a Stacker Carousel on a Biomek Laboratory Automation Workstation.

The steps available in the Stacker Carousel Step Palette are:



• Present — configures from which hotel labware is presented to the deck and bar code reading options.



• Stacker Setup — configures the placement and properties of labware in the hotels on the Stacker Carousel.



• Store — configures to which hotel labware is stored from the deck.

## 23.2 Displaying the Stacker Carousel Step Palette

In order to use the Stacker Carousel steps in a method, display the Stacker Carousel Step Palette (Figure 23-1).

🏥 Biomek® Software - Method1 [New]		
File Edit Project Instrument Execution Options Help		
D ☞ ☆ Ħ B. ❹ % ħ B. ♡ ∾ Ø ► I	□□ ×	
Start Overridab	able Prompt Variable Name V	alue
Present E Finish		
IN THE REPORT OF		
Stacker		
10		
Store		
×		
		P3
	P4 P5 P6	5 P7
Method1 Biomek3000 Biomek3000 ETC: 0:00:00		

Figure 23-1. Biomek main editor with Stacker Carousel Step Palette displayed

To display the Stacker Carousel Step Palette, complete the following:

• Right-click any empty palette space and the step palette menu appears. Select **StackerCarousel**.

OR

• From the toolbar, select **Options>Toolbars>StackerCarousel**.

## 23.3 Using the Stacker Setup Step

Use the Stacker Setup step to:

- configure the placement of labware in the hotels on the Stacker Carousel.
- configure the properties of the labware in the hotels.

## 23.3.1 Configuring the Placement of Labware in the Hotels

To configure the placement of labware in the hotels, labware is dragged from the Labware Category and dropped onto the slots of the hotels. As labware is dragged to the slot, the options Drop One, Fill Slot, and Fill Stack appear to allow the hotel to be quickly configured with labware.

The slots in hotels visually indicate the pieces of labware contained in each slot. A piece of labware that cannot be placed in a particular slot due to the configuration rules cannot be dropped into the slot (refer to Section 23.3.1.1, *Following the Rules to Configure the Placement of Labware in Hotels*).

## 23.3.1.1 Following the Rules to Configure the Placement of Labware in Hotels

These rules must be followed to configure the placement of labware in the hotels:

- Labware cannot be placed in a None hotel.
- Labware that cannot be stacked cannot be substacked. For example, in normal operation a tipbox cannot be stacked on a BCFlat96 plate, so in the Stacker Setup step, a tipbox cannot be substacked on a BCFlat96 plate in a hotel.

Note: Substacking refers to stacking labware in the same slot of the hotel.

- Labware that is taller than the hotel's slot height cannot be placed in the hotel.
- A substack of labware that is taller than the hotel's slot height cannot be placed in hotel.
- A substack of labware cannot be placed in an HD (high-density) hotel.

To configure the placement of labware in the hotels on the Stacker Carousel:

- \_ 8 × Executio Options Device Stacker1 Configuration From File Start Ш Stacker 10 Stacker 10 Stacker 20 Stacker HD 🐔 Instrument Setup Labware Category • Stacker Carousel Setup <Any> Finish Stacker Setup T MANN CTuberack 10mm BCTuberack 12mm BCTuberack 13mm Clipside D **Stacker Carousel** llar\_36mm **Step Palette** Tuberack B3K BCTuberack B3K thod1\* Biomek3000 Biomek3000 ETC: 0:00:04
- 1. Drag and drop a **Stacker Setup** step into the Method View (Figure 23-2).

Figure 23-2. Stacker Setup step configuration

2. Choose the appropriate Stacker Carousel in **Device**.

Note: Multiple Stacker Carousels may be installed on a Biomek instrument.

**Note:** The Stacker Carousel must be installed and configured in Hardware Setup to appear in Device. Refer to the hardware manual for the specific instrument for detailed information on integrating and framing a Stacker Carousel on a Biomek instrument.

3. Drag and drop labware into the slots of the desired hotels.

**Note:** As labware is dropped into a slot, a dialog with options Drop One, Fill Slot, and Fill Stack appears (Figure 23-3). Choose one.

Drop	One
Fill Slo	)t
Fill Sta	ack

Figure 23-3. Dialog to drop labware into hotels

OR

Check **Configuration from File** to use a stacker configuration from a .csv file. **File** and **Browse** appear to allow the desired file to be chosen.

**Note:** Comma-Separated Value Files (.csv) (Table 23-1) are files that include data from a table that can be imported into another application. In this case, the .csv file used with the **Stacker Setup** step must contain Hotel, Slot, and LabwareType, and it may contain Name, LiquidType, UnloadTipsInto, BoxFinalLocation, Volume, BarCode, and SlotOrder.

 Table 23-1.
 Example of a Comma-Separated Value File

Hotel	Slot	Slot Order	Labware Type	Name	Volume	Liquid Type	Bar Code	Unload TipsInto	BoxFinal Location
А	3	1	AP96200µL			na		<tipbox></tipbox>	<home></home>
А	1	1	BCFlat96	Dest1		Water			
А	2	1	BCFlat96	Dest2	120	Water			
А	4	1	BCFlat96	Dest3	130	Water			
В	1	1	BCFlat96		110	Water			
В	2	1	BCFlat96		120	Water			
В	3	1	BCFlat96		130	Water			
В	4	1	BCFlat96		140	Water			
В	5	1	BCFlat96		150	Water			
В	6	1	BCFlat96		160	Water			
С	1	1	BCFlat96			Water			
D	1	1	BCFlat96			Water			

## 23.3.2 Configuring the Properties of the Labware in the Hotels

Each piece of labware in a hotel that will eventually be brought onto the instrument deck is configured using Labware Properties. The information provided in Labware Properties is used when a technique is selected, and when tips are loaded and unloaded.

To access Labware Properties:

1. Double-click on a piece of labware in the Stacker Setup.

OR

Right-click on a piece of labware in the Stacker Setup to open the menu to Labware Properties (Figure 23-4).

Properties					
Сору	Ctrl+C				
Paste	⊂trl+∀				
Delete	Del				

Figure 23-4. Menu accessing Labware Properties

**Note:** Right-clicking a piece of labware also provides the options of **Copy**, **Paste**, and **Delete** to populate the hotels with labware. Choosing **Copy** after labware has been configured in Labware Properties allows configured labware to be copied. Then choosing **Paste** allows a copied version of the configured labware to be pasted into another space on a hotel. Choosing **Delete** will delete the labware from the hotel.

**Note:** Right-clicking a tip box also provides the option of **Remove Tips** to configure an empty tip box.

The configuration options in Labware Properties provide the following information, if it applies to the type of labware selected:

- Name assigned to a piece of labware, if desired
- Labware type
- Maximum volume per well
- Bar code
- Volume for each well, if known
- Liquid type
- Where to dispose of tips
- Where to dispose of tip boxes
- Maximum number of times tips can be loaded
- Available tips

**Note:** Variables and expressions may be entered in any field that can be configured. Refer to Chapter 13, *Using Variables and Expressions in a Method*, for more information on using variables and expressions.

#### 23.3.2.1 Configuring Tip Properties

When the labware selected for configuration is a tip box, Labware Properties provides the ability to configure the following information

1. Double-click the tip box to configure in the Stacker Setup Display

#### OR

Right-click the tip box to configure in the Stacker Setup Display and select **Properties** from the menu. Labware Properties for tips appears (Figure 23-5):

Labware Properties	
Name:	Labware Type: AP96_200uL
Bar Code:	
When empty, send to: Home> Inload Tips Into:	<tipbox></tipbox>
Load no more than 1	
∇ Show Available Tips	
	OK Cancel

Figure 23-5. Labware Properties for a tip box

2. In **Name**, enter the name that is assigned or will be assigned to the tip box in the Stacker Setup step, if applicable.

OR

Leave Name blank. When Name is blank, tips are accessed by the information displayed in the Labware Type field.

**Note:** Naming tip boxes forces the instrument to look for a specific box of tips, rather than any tip box containing the specified tip type located anywhere on the deck.

**Note:** Multiple tip boxes can be given the same name. This is used to create a pool of tip boxes for a specific use.

- 3. In Labware Type, verify the tip type.
- 4. In Bar Code, enter the bar code.

**Note:** Use the bar code field to identify a specific tip box in certain methods, such as plate replication. This field may be left blank.

- 5. In When empty, send to, select a final destination for the tip box at the end of the method.
- 6. In Unload Tips Into, select the location reserved for tip disposal.
- 7. In Load no more than, enter the number of times tips can be loaded onto the pod during the method.

8. Choose **Show Available Tips** to display a graphic of the tip box (Figure 23-6).

Labw	are Pr	opertie	es									
<u>N</u> ame	e: 🗌											Labware Type: AP96_200uL
<u>B</u> ar C	Code:											
<u>W</u> hei	n empty	y, send t	to: <ho< td=""><td>ome&gt;</td><td></td><td></td><td></td><td>-</td><td><u>U</u>nlo</td><td>ad Tips</td><td>Into:</td><td><tipbox></tipbox></td></ho<>	ome>				-	<u>U</u> nlo	ad Tips	Into:	<tipbox></tipbox>
Load	n <u>o</u> moi	re than	1			÷	ime					
⊿ Hi	de Ava	ilable Ti	ps									
	1	2	3	4	5	6	7	8	9	10	11	12
A		•	•	•	•	•		0	0	•	•	
В		•	•	•	•	•	•	•	0	•	•	
С		•	•	•	•	•	•	•	0	•	•	
D		•	•	•	•	•	•	•	0	•	•	
E	•	•	•	•	•	•	•	•	•	•	•	
F	•	•	•	•	•	•	0	0	0	•	0	
G	•	0	•	•	•	•	0	0	0	0	0	•
Н		•	•	•	•	•	0	0	0	0	0	
96 usable tips.												
												OK Cancel

Figure 23-6. Labware Properties for tips with Show Available Tips activated

9. On the graphic of the tip box, select the usable tips. By default, all tips are selected.

**Note:** This can be used to indicate missing tips or tips that are otherwise not to be used in the method.

- 10. Choose **Hide Available Tips** to collapse the graphic of the tip box.
- 11. Choose **OK** to save Labware Properties.

#### 23.3.2.2 Configuring Labware Properties

When the labware selected for configuration is a microplate or small tube rack, Labware Properties provides the ability to configure the following information:

1. Double-click the piece of labware to configure in the Stacker Setup Display

OR

Right-click the piece of labware to configure in the Stacker Setup Display and select **Properties** from the menu. Labware Properties for microplates, reservoirs, and tube racks appears (Figure 23-7).

Labware Properties							
Name:	Labware Type: BCFlat96	Maximum Volume: 362.76 µL					
Bar Code:							
Labware contains a Unknown 💌 volume: 0	μL of Jiquid type:	•					
• Sense the liquid level the first time a well with Unknown or Nominal volume is accessed "from the Liquid".							
Sense the liquid level every time a well is accessed "from the Liquid".							
V Show Labware Volumes							
		OK Cancel					

Figure 23-7. Labware Properties for microplates, reservoirs, and tube racks

- 2. Verify the Labware Type.
- 3. In **Name**, enter a name for the labware.

**Note:** When a deck is populated by numerous pieces of labware, naming labware is recommended. Names should be descriptive of the contents of the labware or the work being accomplished during the method. Naming labware in a meaningful fashion may reduce confusion.

4. In **Bar Code**, enter the bar code.

**Note:** Use the bar code field to identify a specific plate in certain methods, such as plate replication. This field may be left blank.

- 5. Make a selection in Labware contains. Options are Unknown, Nominal, and Known. This information is used by many of the techniques supplied with Biomek Software which aspirate or dispense liquid a certain offset from the liquid level. Some pods or tools can detect the liquid level using liquid-level sensing technology.
  - If Known is selected, the liquid level is not detected during method run and the entered value is used during validation and method run. Known Volume should be supplied whenever possible.
  - If Unknown is selected, the liquid level is detected during method run if required by the technique, and the wells are assumed to be full when validating the method.
  - If Nominal is selected, the liquid level is detected during method run, but the volume in the wells is assumed to be the entered value when validating the method.
- 6. Enter the Volume, if Nominal or Known volume is selected.

#### **Note:** A value entered in Volume is assigned to all wells.

- 7. Select the Liquid Type contained in the labware. The liquid type is useful information when Biomek Software auto-selects a pipetting technique for any aspirate and dispense operations acted upon this piece of labware. The pipetting technique auto-selected to aspirate and dispense the liquid is selected based on the physical factors of the liquid, as well as the physical attributes of the labware. For more information on liquid types, refer to Chapter 8, <u>Understanding and Creating Liquid Types</u>.
- 8. Select **Sense the liquid level the first time a well with Unknown or Nominal volume is accessed "from the Liquid"** to use liquid level sensing to determine the liquid level only the first time it accesses a well with an Unknown or Nominal volume from the liquid. Any instances the same well on the piece of labware is accessed, the liquid level is calculated internally based on the result of the earlier liquid level sense and the amount aspirated or dispensed to the well in previous steps.

#### OR

Select **Sense the liquid level every time a well is accessed "from the Liquid"** to use liquid level sensing to determine the liquid level every time it accesses a well with an Unknown or Nominal volume from the liquid.

**Note:** Liquid level sensing options only apply if the pod or tool accessing the labware is capable of sensing the liquid level. If not, the option selected is ignored.

9. If the labware does not contain uniform volumes, choose **Show Labware Volumes** to display a graphic of the labware (Figure 23-8).

Labware Properties							
Name: Labware Type: BCFlat96 Maxim	ium Volume: 362.76 µL						
Bar Code:							
Labware contains a Known 💌 volume: 150 👘 µL of liquid type:	•						
📀 Sense the liquid level the first time a well with Unknown or Nominal volume is accessed "from the Liq	uid".						
O Sense the liquid level every time a well is accessed "from the Liquid".							
A Hide Labware Volumes							
1 2 3 4 5 6 7 8 9 10 11 12	Volume						
A 000000000000	100						
	150						
	Amount (µL):						
No selected wells.	<u> </u>						
	OK Cancel						

Figure 23-8. Labware Properties for microplates, reservoirs, or tube racks with Show Labware Volumes activated

**Note:** Show Labware Volumes may appear different depending on the number of wells on the selected labware. The graphic shown is for a 96-well microplate.

10. Select the desired wells on the graphic.

**Note:** Hold down **Ctrl** to toggle wells between selected and deselected without affecting the selection status of other wells.

**Note:** Clicking on a column or row heading selects all wells in that column or row.

- 11. Enter the volume to assign to the currently selected wells in **Amount**.
- 12. Choose **Set** to assign the current Amount to the currently selected wells. The wells are given a color code that corresponds to the list of set volumes under Volume.
- 13. Repeat steps 10-12 until all desired wells have been assigned the desired volume.

Note: Clear sets the volume for all wells back to zero.

- 14. Choose Hide Labware Volumes to collapse the graphic of the labware.
- 15. Choose **OK** to save Labware Properties and return to the Stacker Setup step configuration.

## 23.4 Using the Present Step

Use the Present step for:

- <u>Configuring from which Hotel Labware is Presented</u> (Section 23.4.1).
- <u>Configuring Bar Code Reading Options</u> (Section 23.4.2).

### 23.4.1 Configuring from which Hotel Labware is Presented

To configure a **Present** step in a method:

- 1. Drag and drop a **Present** step into the Method View.
- 2. Choose a hotel and the desired stacker carousel from **Present labware from** hotel...of.... The chosen hotel is highlighted (Figure 23-9).

🕼 Biomek® Software - Method1* [New]	_ 8 ×
File Edit Project Instrument Execution Options Help	
Start Present labware from hotel	
Present Instrument Setup Do not read bar code	
Stacker Carousel Setup Stacker 10 Stacker 20 Stacker HD	
Statler Setup Present	
Store	
A B C D	
p6 p7	
Method1* [Biomek3000 Biomek3000 ETC; 0:00:04	

Figure 23-9. Present Step configuration

Note: Multiple Stacker Carousels may be installed on a Biomek instrument.

 Choose Do not read bar code, Read bar code for logging, or Read bar code for decision making (refer to Section 23.4.2, <u>Configuring Bar Code</u> <u>Reading Options</u>).

**Note:** The step in the Method View is more descriptive once the step is configured and Finish selected; for example "Present from hotel B of Stacker 1."

### 23.4.2 Configuring Bar Code Reading Options

The bar code reading options available in the Present step are defined as follows:

- Do not read bar code the bar code is not read when the Stacker Carousel presents the plate.
- Read bar code for logging the bar code is captured in the pipetting logs (refer to Sections 26.2.3, *Pipetting Log*, 26.2.4, *Span8Pipetting Log (FX.* <u>NX-S8 only)</u>, and 26.3.6, *UnifiedPipetting Log Contents*).
- Read bar code for decision making the results of the bar code read determine the next operation of the method. The software waits for the results of the bar code before continuing method execution.

**Note:** Because the Read bar code for decision making option creates a dependent operation (similar to the If Step and Then and Else), it impacts the validation process during method building. The method validation stops at this point as it waits for the results of the bar code read before it continues; however, since there is no actual bar code read during validation the software does not know what operation to validate next. To make sure the validation process finishes satisfactorily, in Read bar code for decision making using. . . for method validation, type in a value to use as the bar code during method validation.

For example, to check for an exact bar code, use the lf step with the condition

= (Labware("SC1").Properties.Barcode="SRC") or to check for plates whose bar code begin with SRC, use the If Step with the condition

= (Left(Labware("SC1").Properties.Barcode,3) =
"SRC").

## 23.5 Using the Store Step

Use the **Store** step to configure to which hotel on the Stacker Carousel labware is stored from the deck.

To configure a Store step in a method:

- 1. Drag and drop a **Store** step into the Method View.
- 2. Choose a hotel and the desired stacker carousel from **Store labware in hotel..** .of .... The chosen hotel is highlighted (Figure 23-10).

Biomek® Software - Method1* [New]	
File Edit Project Instrument Execution Options Help	
Start Store labovare in hotel of Stacker1	
🕎 🧠 Instrument Setup	
10 10 Stacker Carousel Setup	
Stacker to Stacker to Stacker to Stacker to	
Setup in the research in the ratio of a setup in the r	
store Finish	
DUCE PE P7	
U Method1* Biomek3000 Biomek3000 ETC: 0:00:15	

Figure 23-10. Store Step configuration

**Note:** The step in the Method View is more descriptive once the step is configured and Finish selected; for example "Store to hotel B of Stacker 1."

# Using the Data Sets Step Palette

## 24.1 Overview

Steps on the Data Sets Step Palette allow data sets to be created, managed, used, and reported in Biomek methods. Data sets allow sample tracking which means the contents of the plate or tube are known at any time in the method and decisions may be made during transfer or pipetting operations based on the contents of the plate or tube.

**Note:** Refer to Chapter 14, <u>Using Sample Tracking and Data Sets in a Method</u>, for more information on using sample tracking and data sets in a method.

The steps available on the Data Sets are:

• Create Data Set — specifies data in a data set.



Data Set Management

190

 Data Set Management — renames, removes, copies, or modifies the properties of a data set.



• Data Set Processing — applies a transformation expression to an existing data set to create a new data set.



• Reporting — generates a report on data sets at any point during a method.

## 24.2 Displaying the Data Sets Step Palette

In order to add data set steps to a method, display the Data Sets Step Palette (Figure 24-1).



Figure 24-1. Biomek Software main editor with Advanced Step Palette displayed

To display the Data Sets Step Palette:

- Right-click any empty palette space, and select **DataSets** from the menu. OR
- From the toolbar, select **Options>Toolbars>DataSets**.
# 24.3 Configuring the Create Data Set Step

The **Create Data Set** step is used to create data sets from a text file or a database table. The specified file or database table is read and compared to either the plate bar code and well number of the plate or a data set specifying the sample ID. Data sets are created from the specified columns of the file or database table using the rows that matched the plate bar code and well number or data set specifying the sample ID.

Note: The text file must reside on the local computer.

Note: SQL Server 2000 SP3 is the version requirement to import the datatbase table.

File Edit Project Instrument Execution Options Help	
▐▋▆▖▋▋▙▟▕▓▆▆▏▖▖▖┛▏▶▝▌▌▕▓	
Contraction Contr	File Options     Read from a Database.      Read from a file.     Read from a Database.      Read from a Database.      Read from a file.     Read from a Database.      Read from
Output Options	TL1         P4         XXX         P16           P1         Samuel P9         P13         P17           P2         Samuel P10         P14         P18           P3         P7         P11         P15         P19

Insert a Create Data Set step into the Method View (Figure 24-2).

Figure 24-2. Create Data Set step configuration

The **Create Data Set** step consists of three collapsible configuration areas. Each area can be collapsed or expanded independently from each other to display the configuration options for that section (Figure 24-2). To collapse or expand any of the configuration areas, click on the sentence summary.

Configuring the Create Data Set step includes configuring:

- the file or database to read (refer to Section 24.3.1, <u>Configuring File</u> <u>Options</u>).
- the columns of the file or database table to compare (refer to Section 24.3.2, *Configuring Input Options*).
- the columns of the file or database table used to create data sets (refer to Section 24.3.3, <u>Configuring Output Options</u>).

## 24.3.1 Configuring File Options

File Options is used to select a comma- or tab-delimited text file (refer to Section 24.3.1.1, <u>Reading from a Text File</u>) or a database table (refer to Section 24.3.1.2, <u>Reading From a Database Table</u>) to use to create data sets for a piece of labware.

#### 24.3.1.1 Reading from a Text File

To configure File Options to read from a file:

 If not expanded, select the sentence summary to display File Options (Figure 24-3).

**Note:** By default, File Options is expanded when a Create Data Set step is inserted.



Figure 24-3. Create Data Set step — File Options displayed

- 2. Under File Options, select Read from a file.
- 3. In Read the file, enter the full directory path and file name of the text file to read and press **Enter**. The first four rows of the selected data file are displayed in File Preview.

Note: Browse to select the file using standard Windows techniques.

**Note:** The text file must reside on the local computer.

4. In Begin reading at row, select the first row of the data file that should be read. File Preview updates. 5. Select **The file has a header row** if a header row is included in the data file. The first row that is read is used for column headings in File Preview (Figure 24-3). For example, if Begin reading at row is set to 5 and The file has a header row is selected, the contents of row 5 are used as the column headings.

**Note:** If The file has a header row is not selected, column headings are given a generic label Column#.

6. Select the appropriate file type option: Comma-Delimited or Tab-Delimited. File Preview updates accordingly.

#### 24.3.1.2 Reading From a Database Table

File Options can be configured to read data from a database table stored in the SQL Server database.

**Note:** SQL Server is a Microsoft product that allows queries and provides analysis, and is required to run Biomek Software. The installation of this product is the responsibility of the user and is not supported by Beckman Coulter Service Engineers. SQL Server 2000 SP3 is the version requirement.

To configure File Options to read from a database:

1. If necessary, select the sentence summary to display File Options (Figure 24-4).

(† Biomek)	® Software	- Method10* [New]		_ # ×
File Edit I	Project Ins ⊨î⊒⊒l िå	trument Execution Options Help ▲ LA Ban (R) い ⊂ 20 ► □□ □ 次	IE 31	
		Start	/ File Options	
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	The second	Contract Set in	Read from a file.     C Read from a Database.	
Create Data Set	Instrument Setup	The Croate Data Set	Bead the file	
Ħ	A		Begin reading at row 1 Comma-Delimited	
Data Set Management	Transfer		The file has a header row. C Tab-Delimited	
關	all's		File Preview	
Data Set Processing	Combine			
锢	$\Rightarrow$			
Reporting Step	Move Labware			
	Pause		∇ Source position not selected.	
	Q		∇ Destination position not selected.	
	Comment			
			Т.1 Р4 💹 Р16	
			P1 Sample P9 P13 P17	
			P2 Eample P10 P14 P18	
			P3 P7 P11 P15 P19	
]]				=
Method10*	BiomekFX	BIOMERTX [EIC: 0:00:03]		

**Note:** File Options is expanded when a Create Data Set step is inserted.

Figure 24-4. Create Data Set step — File Options displayed

2. Under File Options, select **Read from a Database**. File Options changes to display the database options (Figure 24-5).



Figure 24-5. Reading from a database

3. Browse to find the location of the desired database on that server. SQL Server Connection (Figure 24-6) appears on top of the Create Data Set step configuration.

**Note:** SQL Server is a Microsoft product that allows queries and provides analysis. The installation of this product is the responsibility of the user and is not supported by Beckman Coulter Service Engineers. SQL Server 2000 SP3 is the version requirement.

SQL Server Connection	
[local] DepartmentServer	User:
	Password:
	Use NT Authentication
	Connect
Databases:	
OK.	Cancel

Figure 24-6. SQL Server Connection

- 4. Select the desired server from the list of servers on the left.
- 5. Enter the **User** name and **Password** to log in to the selected server.

OR

Select Use NT Authentication.

**Note:** The administrator of the SQL Server Connection sets the logging protocol for user and password or NT authentication.

6. Choose **Connect**. A list of the available databases appears in **Databases** (Figure 24-7).

SQL Server Connection	
(local) DepartmentServer	User: sa
	Password: *********
	Use NT Authentication
	Connect
Databases:	
DataSets	
OK.	Cancel

Figure 24-7. Connected to SQL Server

7. In **Databases**, select the desired database found on the server to use to create data sets.

OR

Repeat steps 3-6 to connect to a different server, if desired.

- 8. Choose **OK** to accept the server and close **SQL** Server Connection.
- 9. In Read from the table (Figure 24-5), select the database table to read. The first four rows of the selected database table are displayed in Table Preview (Figure 24-5).

## 24.3.2 Configuring Input Options

Input Options (Figure 24-9) is used to configure the columns of the file or database table to compare to the plate bar code and well number or data set specifying the sample ID of the specified plate. Rows that match the plate bar code and well number or data set specifying the sample ID are used to create the data set(s).

By default, Input Options is collapsed when a Create Data Set is inserted into a method (Figure 24-8).

Image: Biomek® Software - Method10* [New]	
File Edit Project Instrument Execution Options Help	
▯▯ਫ਼ਜ਼ੑੑੑੑੑੑੑੑੑੑੑੑੑੑੑੑੑੑੑਫ਼ੑੑੑੑੑੑੑੑੑੑੑੑੑੑੑ	
Start	A File Options
Instrument Set in	Read from a file.     C Read from a Database.
Data Set Setup	
Create Data Set	Read the tie
Data Set Transfer Finish	Begin reading at row 1 호 📀 Comma-Delimited
	The file has a header row.
Stars was	File Preview.
Processing Combine	
Reporting Move Collapsed Input	
Options	
Click on the sentence	
	∇ Source position not selected.
summary to expand	∇ Destination position not selected.
Comment Input Options.	
	P1 P1 P1 P13 P17
	P3 P7 P11 P15 P19
Method10* BiomekFX BiomekFX ETC: 0:00:03	

Figure 24-8. Create Data Set step — expanding Input Options

(h) Biomeki	® Software	- Method2* [New	ป									
File Edit	Project Insl	trument Execution	Options Help		_	_						
	i B B	a % b 🖻	n α Ø   ► II									
	æ	Start 🔋		Begin r	eading at row	1 🌻	•	C <u>o</u> mma-De	limited			1
Create Data Sat	ිරු Instrument	🐔 Instrum	ent Setup	🗖 The	e file has a <u>h</u> eac	der row.	C	<u>I</u> ab-Delimi	ed			
Ħ	<u>a</u>	Create D	)ata Set	File Preview:								
Data Set Management	Transfer	🔓 Finish		L1	L2	L1	L3	L1	Plate4	F12	_	
Π				L1	L1	L2	L2	L1	Plate4	J17		
Data Set	Combine			L2	L1	L1	L3	L1	Plate1	N21		
E												
Reporting	Move			A Input Opti	ions							
Step	Labware			For the Jabw	are at	💌 , stad	k <u>d</u> epth 0					
				Match th	ne <u>P</u> late Barcod	e and Well N	imberj					
	Pause			For the la	abware's <u>b</u> arcoo	de, use the fie			-			
				and for t	he well <u>n</u> umber,	use the field			•			
	Comment			C Match th	ne <u>S</u> ample ID							
		_		For the S	Sample [D, use	the field			-			
				to <u>m</u> atch	n with the Diata S	Set 🗌		~	[			
				⊽ Destinatio	on position not s	elected.						_
					TL1				I F	15		
					P1		-   P8	P12	1			
					P2		P9	P13	í			
				l V I r	P3	P6	P10	P14	i í			
												=
Method2*	Biomek FX	Study 14 BiomekF	X ETC: 0:00:03									

Click on the sentence summary to expand Input Options (Figure 24-9).

Figure 24-9. Create Data Set step — Input Options displayed

#### 24.3.2.1 Matching Plate Bar Code and Well Number

The specified file or database table is compared to the plate bar code and well number to determine which rows of the file or database table to use to create the data set. Only the rows of the file or database table that match the plate bar code are used. The column for well number is used to assign the correct value of the data set to each well.

To configure Input Options to match the plate bar code and well number:

1. In For the labware at, select the deck position of the labware to compare the plate bar code and well number or data set specifying the sample ID to the file or database table (Figure 24-9).

OR

Click on the desired labware in the Current Deck Display.

2. If the labware is in a stack, enter the **stack depth**.

**Note:** Depth refers to a specific piece of labware in a stack. To use the top piece of labware, enter **0**; to use the second from the top piece of labware, enter **1**; to use the third from the top piece of labware, enter **2**; and so on. Refer to Section 7.3.6.4.1, *Biomek Stacking Rules*, for information on using stacks.

- 3. Select **Match the Plate Barcode and Well Number** to compare the plate bar code and well number of the selected plate to the file or database table (Figure 24-9).
- 4. In For the labware's barcode, use the field, select the column of the file or database table to compare to the plate bar code.

**Note:** For the labware's barcode, use the field is optional. If no field is specified, all rows of the file or database table are assumed to apply to that plate. If a well number is repeated, the last row of the file or database table to use that well number is used.

5. In and for the well number, use the field, select the column of the file or database table to compare to the well number.

#### 24.3.2.2 Matching Sample IDs

The specified file or database table is compared to a data set that specified the sample IDs for the selected plate to determine which rows of the file or database table to use to create the data set. Only the rows of the file or database table that match one of the sample IDs on the plate are used to create data sets.

To configure Input Options to match the plate bar code and well number:

1. In For the labware at, select the deck position of the labware to compare the plate bar code and well number or data set specifying the sample ID to the file or database table (Figure 24-10).

OR

Click on the desired labware in the Current Deck Display.

🌵 Biomek(	® Software	- Method2* [New	1								
File Edit I	Project Ins	trument Execution	Options Help		- 1						
jD ⊯≞ ∎			∽ ~ Ø   ►	※  厘目	<u>u.</u>						_
<b>E</b>	æ	Start		Begir	reading at row	1 👤	(	Comma-D	elimited		<b>_</b>
Create	Instrument	🕷 Instrum	ent Setup	Π Τ	he file has a <u>h</u> e	ader row.	(	〕 <u>I</u> ab-Delim	ited		
Data Set	Setup	Create D	ata Set	File Previe	W:						
Data Set		Finish			1.0		1.0			510	
Management	l ransher			LI L1	L2 L1	L1 L2	L3 L2	L1	Plate4 Plate4	J17	-
57(AB)				L2	L1	L1	L3	L1	Plate1	N21	
Processing	Combine										
锢	$\rightarrow$			A Innut 0	otions						
Reporting Step	Move Labware			For the lat	ware at	<b>T</b> st	ck depth 🛙				
	0			O Match	the Plate Barco	ode and Well I	lumber				
	Pause			For the	labware's <u>b</u> arc	ode, use the f	eld		•		
				and fo	r the well <u>n</u> umbe	er, use the fiel			-		
	Comment			<ul> <li>Match</li> </ul>	the <u>S</u> ample ID	_			_		
		í		For the	Sample [D, us	e the field			-		
				to <u>m</u> at	sh with the Dati	a Set		*	1		
				⊽ Destina	tion position no	t selected.					•
					TL1			M IW	3. F	15	
					P1						
					P2		P9	P13			
					P3	P6	P10	) P14			
 Method2*	Piomek FY	- Study 14 Rioselu	X ETC: 0:00:02	 							
rieulouz.	Diomek FA -	- Juday 14 DIUMERF	v jere, 0,00:03 j								

Figure 24-10. Create Data Set step — Input Options displayed

2. If the labware is in a stack, enter the **stack depth**.

**Note:** Depth refers to a specific piece of labware in a stack. To use the top piece of labware, enter **0**; to use the second from the top piece of labware, enter **1**; to use the third from the top piece of labware, enter **2**; and so on. Refer to Section 7.3.6.4.1, *Biomek Stacking Rules*, for information on using stacks.

- 3. Select **Match the Sample ID** to compare a data set specifying the sample IDs for the selected plate to the file or database table (Figure 24-10).
- 4. In For the Sample ID, use the field, select the column of the file or database table to compare to the data set specifying sample IDs for the plate.
- 5. In to match with the Data Set, select the data set that specifies the sample IDs for the plate.

## 24.3.3 Configuring Output Options

Output Options is used to configure the plate for which data sets are created and to specify the columns of the file or database table to use to create data sets.

By default, Output Options is collapsed when a Create Data Set is inserted into the Method View.

🌵 Biomek(	® Softwar	e - Method2* [New]										
File Edit I	Project In La 🔲 🗖	strument Execution Options Help		a								
豳		Start	Begin reading at row 1 🚖 📀 C <u>o</u> mma-Delimited									
Create Data Set	Instrument Setup	🐔 Instrument Setup		he file has a <u>h</u> ea	ader row.	0	<u>T</u> ab-Delimi	ted				
		Treate Data Set	File Previe	w:								
Data Set	Turnefu	Finish	1.1	1.2	11	10	11	Distad				
Management	i ransrer	-	L1	L2 L1	L1 L2	L3 L2	L1	Plate4	J17			
5(ab)			L2	L1	L1	L3	L1	Plate1	N21			
Processing	Combine											
睴	$\rightarrow$		A Input D	ntiana								
Reporting Step	Move Labware		Carlles lab			-1. d1. 0	_					
	8		C Match	the Plate Barco	de and Well N	ck geptn ju						
	Pause		For the	labware's barci	ode, use the fi	eld		-				
	A		and fo	the well <u>n</u> umbe	r, use the field			-				
	Comment		<ul> <li>Match</li> </ul>	the <u>S</u> ample ID								
		Collapsed Output	For the	Sample [D, use	the field		1	2				
		Options	to <u>m</u> ati	sh with the Data	Set		7					
		Click on the sentence	▼ Destina	tion position not	selected.						Ţ	
		summary to expand								1-1-		
		Output Options.		ТІ 1		h Marine	1 837	1 P	15			
			147	P1		P8	P12					
				P2	ſ	 Р9	P13	) (				
				P3	PD	P10	P14					
Method2*	Biomek FX	Study 14 BiomekFX ETC: 0:00:03										

Figure 24-11. Create Data Set step — expanding Output Options

To configure Output Options:

1. Click on the sentence summary to expand the Output Options configuration (Figure 24-12).

🌵 Biomek	B Software	- Method2* [New	ı]		
File Edit	Project Inst	trument Execution	Options Help		
່ 🗅 📽 (	n 🖬 🖻	🧉 🐰 🖻 🖻	∽ ∩ Ø   ► [		
₩.	æ	Start		♥ Filename has not been set.	
Create	Instrument	🐔 Instrume	ent Setup	$\nabla$ Match the Plate Barcode using the field and the Well Number using the field for the labware at P4, stack depth 0.	
Data Set	Setup	Create D	)ata Set	Create 0 Data Sets on the labware at P4, stack depth 0.	
Data Set	Transfor	Finish		For the labware at, stack depth 0	
Management	manaren			Available Fields Data Sets To Be Created	
Data Set	Combine			Plate	
Processing					
Reporting	Move			X4 X5	
Step	Labware			Add All Remove All	
	Pause				
	Comment				
				P1 Sample P8 P12	
				P2 Sample P9 P13	
Method2*	Biomek FX	Study 14 BiomekF	X ETC: 0:00:03		

Figure 24-12. Create Data Set step — Output Options

2. In For the labware at, select the deck position of the labware for which data sets are created.

OR

Click on the desired labware in the Current Deck Display.

3. If the labware is in a stack, enter the **stack depth**.

**Note:** Depth refers to a specific piece of labware in a stack. To use the top piece of labware, enter **0**; to use the second from the top piece of labware, enter **1**; to use the third from the top piece of labware, enter **2**; and so on. Refer to Section 7.3.6.4.1, *Biomek Stacking Rules*, for information on using stacks.

4. Under Available Fields, select the desired columns from the file or database table from which to create data sets. The selected fields appear highlighted.

5. Select ---> to move the selected fields to Data Sets to be Created.

**Note:** To add all columns from Available Fields to Data Sets to be Created, select **Add All**.

 To remove an option from Data Sets to be Created, select the desired data set to remove and select <---. The selected field(s) are moved back to Available Fields.

**Note:** To remove all columns from Data Sets to be Created, select **Remove** All.

#### 24.3.3.1 Setting Data Set Properties

Properties for each data set can be modified in Output Options by right-clicking on the data set name under Data Sets To Be Created. Properties that can be modified are:

- data set name
- including the data set in reports
- tracking the data set during pipetting

To modify properties of a data set:

1. In Data Sets To Be Created, right-click the desired data set. The data set properties menu appears.

<u>S</u>et Destination Name ✓ Include in Reports ✓ Irack Data While Pipetting

Figure 24-13. Changing data set properties

- 2. Select the desired option from the menu:
  - Select **Set Destination Name** and enter a new name for the data set to rename the data set.
  - Select Include in Reports to allow the data set to be included in the reports chosen in the Reporting (refer to Section 24.6, <u>Configuring the Reporting Step</u>) or Finish steps (refer to Section 12.2.2, <u>Configuring the Finish Step</u>).
  - Select Track Data While Pipetting to allow the system to track the data during pipetting operations.

**Note:** When Track Data While Pipetting is selected, the data set is copied to the destination labware during pipetting.

3. Repeat steps 1 and 2 to select additional options, if desired.

# 24.4 Configuring the Data Set Management Step

The Data Set Management step is used to manage data sets that have been created automatically or created using the Instrument Setup step, the Create Data Set step, the SILAS step, or Visual Basic scripting.

**Note:** Data sets are not created or processed in the Data Set Management step.

Managing data sets includes using the Data Set Management step for:

- <u>Copying a Data Set</u> (Section 24.4.1).
- <u>*Renaming a Data Set*</u> (Section 24.4.2).
- <u>*Removing a Data Set*</u> (Section 24.4.3).
- <u>Changing the Properties of a Data Set</u> (Section 24.4.4).

#### 24.4.1 Copying a Data Set

To copy a data set:

1. Insert a **Data Set Management** step into the Method View. The **Data Set** Management step configuration appears (Figure 24-14).

🌔 Biomek	® Software	- Meth	nod7* [New]									
File Edit	Project Inst	rument	Execution O	ptions Help								
] D ⊯ (		<b>a</b>	1. B 🖻 🛛	ດ ຕ 🖉	▶ □□ 二 ×							
田田	48	8	Start			For the	e Jabware at	•	, depth	:		
Create Data Set	بن Instrument Setup		Instrument	t Setup		Сору	▼ the	Data Set <u>n</u> am	ed 📃	•	to 🚺	
田	R	田	Data Set M	lanagemen	t							
Data Set Management	Transfer	000	Finish									
镾	ag											
Data Set Processing	Combine											
睴												
Reporting Step	Move Labware											
	Pause											
	Q											
	Comment											
							1					
							TL1		P8 F	P12	P16	
						$\left  \sum_{i=1}^{n} \right $		P5	P9 F	P13	P17	
								P7	P11 C	P14	P18	
						N.				1.5	1.15	

Figure 24-14. Choosing Copy in Data Set Management step

- 2. From **For the labware at**, choose the labware where the data set is copied.
- 3. If in a stack, enter the **depth** of the labware.

**Note:** Depth refers to a specific piece of labware in a stack. To use the top piece of labware, enter **0**; to use the second from the top piece of labware, enter **1**; to use the third from the top piece of labware, enter **2**; and so on. Refer to Section 7.3.6.4.1, *Biomek Stacking Rules*, for information on using stacks.

- 4. From the drop-down menu, choose **Copy**.
- 5. From **the Data Set named**, choose the data set to copy.

**Note:** Each data set must have a unique name. The method displays an error during validation if two data sets have the same name.

6. At **to**, enter the name of the copied data set. A copy of the data set is created with the entered name.

## 24.4.2 Renaming a Data Set

Rename renames a data set. Renaming data sets is an important function for sample tracking because only one data set named Sample ID can exist on a single piece of labware. When another data set is created from the Instrument Setup step using sample IDs, it overwrites the previous data set.

Rename a data set when sample IDs from the Instrument Setup are necessary for a transfer operation and then another set of sample IDs is necessary for a second transfer operation. For example, the name of the first Sample ID data set can be used to maintain a record of the first transfer operation and then the second Sample ID data set may be renamed before the second transfer operation.

**Note:** Each data set must have a unique name. The method displays an error during validation if two data sets have the same name.

**Note:** Refer to Section 15.2.6, <u>Creating the SampleID Data Set Using the Instrument</u> <u>Setup Step</u>, for more information on creating the Sample ID data set using the Instrument Setup step.

To rename a data set:

- 1. Insert a **Data Set Management** step into the Method View. The **Data Set** Management step configuration appears (Figure 24-14).
- 2. From For the labware at, choose the labware where the data set is renamed.
- 3. If in a stack, enter the **depth** of the labware.

**Note:** Depth refers to a specific piece of labware in a stack. To use the top piece of labware, enter **0**; to use the second from the top piece of labware, enter **1**; to use the third from the top piece of labware, enter **2**; and so on. Refer to Section 7.3.6.4.1, *Biomek Stacking Rules*, for information on using stacks.

N Biomek® Software - Method7* [New]	×
Image: Set Processing   Data Set Processing <t< th=""><th>ware at P1 , gepth 0 : The Data Set named I I is .</th></t<>	ware at P1 , gepth 0 : The Data Set named I I is .
	TL1 P8 P12 P16 P5 P9 P13 P17 P14 P18 P7 P11 P15 P19

4. From the drop-down menu, choose **Rename**. Figure 24-15 appears.

Figure 24-15. Choosing Rename in Data Set Management step

5. From the Data Set named, choose the data set to rename.

**Note:** Each data set must have a unique name. The method displays an error during validation if two data sets have the same name.

6. At **to**, enter the new name of the data set. The data set is renamed.

## 24.4.3 Removing a Data Set

When a data set is no longer needed, it may be removed.

To remove a data set:

- 1. Insert a **Data Set Management** step into the Method View. The **Data Set** Management step configuration appears (Figure 24-14).
- 2. From **For the labware at**, choose the labware where the data set is removed.
- 3. If in a stack, enter the **depth** of the labware.

**Note:** Depth refers to a specific piece of labware in a stack. To use the top piece of labware, enter **0**; to use the second from the top piece of labware, enter **1**; to use the third from the top piece of labware, enter **2**; and so on. Refer to Section 7.3.6.4.1, *Biomek Stacking Rules*, for information on using stacks.

4. From the drop-down menu, choose **Remove**. Figure 24-16 appears.

Image: Biomek® Software - Method7* [New]		- 🗆 🗡
File Edit Project Instrument Execution Options Help		
TIME Contraction Start	For the Jabware at P1 , depth 0 :	
Create Data Set Setup	Remove The Data Set named	
Data Set Management Transfer		
Data Set Combine		
Processing		
Reporting Move Step Labware		
Pause		
Comment		
	TL1         P8         P12         P16           P5         P9         P13         P17           P14         P18	
	P7 P11 P15 P19	
		=
Method7* BiomekFX BiomekFX ETC: 0:00:03		

Figure 24-16. Choosing Remove in Data Set Management step

5. From **the Data Set named**, choose the data set to remove. The data set is removed.

### 24.4.4 Changing the Properties of a Data Set

When Change is chosen in the Data Set Management step, two options are available:

- Including or not including the data set in reports.
- Tracking or not tracking the data in the data sets during pipetting.

To change the reporting properties of a data set:

- 1. Insert a **Data Set Management** step into the Method View. The **Data Set** Management step configuration appears (Figure 24-14).
- 2. From **For the labware at**, choose the labware where the data set properties are changed.
- 3. If in a stack, enter the **depth** of the labware.

**Note:** Depth refers to a specific piece of labware in a stack. To use the top piece of labware, enter **0**; to use the second from the top piece of labware, enter **1**; to use the third from the top piece of labware, enter **2**; and so on. Refer to Section 7.3.6.4.1, *Biomek Stacking Rules*, for information on using stacks.

4. From the drop-down menu, choose Change. Figure 24-17 appears.

🕅 Biomek	® Software	- Method7* [New]		- 🗆 🗵
D ⊯ (	n 🖪 🖪		※ [ 正 ]	
<b>E</b>	and the second s	Start	For the Jabware at P10 , depth 0 :	
Create Data Set	ער Instrument Setup	🐔 Instrument Setup	Change T the Data Set named as follows:	
田	A	Data Set Management	✓ The Data Set should be included in reports.	
Data Set Management	Transfer	ğ Finish	✓ The data in the Data Set should be tracked during pipetting.	
镾				
Data Set Processing	Combine			
睴				
Reporting Step	Move Labware			
	Pause			
	Q			
	Comment			
			TL1 P8 P12 P16	
			P5 P9 P13 P17	
Mathad7*	PierrekEY			
prietnoa (*	рютекн-х	DOMERTA JETC: 0:00:03		

Figure 24-17. Change chosen on Data Set Management step configuration

5. From the Data Set named, choose the data set to be changed.

- 6. Ensure **The Data Set should be included in reports** is checked to allow the data set to be included in the reports chosen in the **Reporting** or **Finish** steps.
- 7. Ensure **The data in the Data Set should be tracked during pipetting** is checked to allow the system to track the data during pipetting operations.

**Note:** If the data set applies only to the source labware and should not be tracked, do not check The data in the new Data Set should be tracked during pipetting. Not checking this option may be useful for complex methods that use data sets, but may not require the data set to be transferred to the destination labware.

## 24.5 Configuring the Data Set Processing Step

The Data Set Processing step allows a transformation expression to be applied to an existing data set to create a new data set. The expression is applied to each well or tube in the destination data set and the result is stored in the corresponding well or tube of the newly created data set.

**Note:** A transformation expression is an operation stated in symbolic form (refer to Section 24.5.1, *Understanding Expressions*).

**Note:** The destination data set is the result of the transformation expression applied to the existing data set.

The applied expression may result in numeric or Boolean values for the new data set (Table 24-2). These values are presented in a report on data sets that is configured in the **Reporting** (refer to Section 24.6, *Configuring the Reporting Step*) or Finish steps (refer to Section 12.2.2, *Configuring the Finish Step*).

**Note:** Boolean values are expressed as True or False in the reports configured in the Reporting or Finish steps.

Insert a **Data Set Processing** step into the Method View. The **Data Set Processing** step configuration appears (Figure 24-18).



Figure 24-18. Data Set Processing step configuration

To configure the Data Set Processing step:

- 1. At For the labware at, choose the labware where the new data set is located.
- 2. If in a stack, enter the **depth** of the labware.

**Note:** Depth refers to a specific piece of labware in a stack. To use the top piece of labware, enter **0**; to use the second from the top piece of labware, enter **1**; to use the third from the top piece of labware, enter **2**; and so on. Refer to Section 7.3.6.4.1, *Biomek Stacking Rules*, for information on using stacks.

3. In Create a new Data Set named, enter the name of the new data set.

**Note:** Each data set must have a unique name. The method displays an error during validation if two data sets have the same name.

4. In with each element defined by the expression:, enter the expression.

**Note:** Refer to Section 24.5.1, *<u>Understanding Expressions</u>*, for information on common symbols used in expressions.

**Note:** An equal (=) sign must precede the expression entered in the Data Processing step.

5. At Allow the Data Set named. . . to be used in the expression, choose the previously created data set to be used in the expression.

Note: The previously created data sets are available for selection.

**Note:** Leave Allow the Data Set named. . .to be used in the expression blank to not use an existing data set as the basis for the transformation.

- Ensure The new Data Set should be included in reports is checked to allow the new data set to be included in reports chosen in the Reporting or Finish steps.
- 7. Ensure **The data in the new Data Set should be tracked during pipetting** is checked to allow the data from the new data set to be tracked during pipetting operations.

**Note:** If the expression is applied only to the source labware and should not be tracked, do not check The data in the new Data Set should be tracked during pipetting. Not checking this option may be useful for complex methods that use data sets, but may not require the data set to be tracked.

## 24.5.1 Understanding Expressions

Expressions, using symbols, allow a customized operation to be performed in the Data Set Processing step.

Using expressions in the Data Set Processing step includes understanding:

- some common symbols necessary to create an expression (Table 24-1).
- results of expressions on data sets (Table 24-2).

**Note:** An equal (=) sign must precede an expression entered in the Data Set Processing step.

**Note:** Refer to Section 13.3, <u>Using Expressions</u>, for more information on using expressions.

Table 24-1	Some common	symbols used	l in ex	nressions
1 a 0 10 2 - 1.		symbols used	а пп сл	10105510115

Symbol	Definition
+	addition
-	subtraction
*	multiplication
/	division; returns a rational number
\	integer division; returns an integer, no remainder; for example, $13 \ 5$ returns a value of 2
a MOD b	returns the remainder after a is divided by b; for example, 13 MOD 5 returns a value of 3
>	greater than
<	less than
=	equal to
<>	not equal to
^	exponentiation; for example, $a \wedge b = a^b$

Table 24-2. Some expressions and the results

Expression	Result
= DataSetName > 5	creates a new data set where each element is True if the value in the existing data set is greater than 5 and False if the value is less than or equal to 5
= DataSetName * 1.05	creates a new data set where each value is the product of the existing data set multiplied by 1.05

Note: Element refers to the value associated with a well or tube.

# 24.6 Configuring the Reporting Step

The Reporting step (refer to Section 24.6, <u>Configuring the Reporting Step</u>) allows a report to be generated on data sets at any point during a method. While specifying the data desired from the data sets may be accomplished in the Create Data Set, Data Set Management, Data Set Processing, and Script steps, the report style and saved location of the report are chosen in the Reporting step.

The report contains data from all data sets that are configured to be included in reports and the type may be selected from four styles (refer to Section 24.6.1, *Report Styles*).

Since the Volume data set (refer to Section 14.2.1, <u>Understanding the Volume Data</u> <u>Set</u>) is reported in every report, the volume in each well in a microplate or tube in a rack can be tracked throughout a method if a **Reporting** step is inserted after every transfer or pipetting operation.

**Note:** The report style and saved location of the report on data sets generated at the end of a method may also be chosen in the Finish step (refer to Section 12.2.2, *Configuring the Finish Step*).

Insert a **Reporting** step into the Method View. The **Reporting** step configuration appears (Figure 24-19).



Figure 24-19. Reporting step configuration

To choose the report style and location of the report:

- 1. From **Report Style**, choose one of the following:
  - Text File
  - Per-Plate HTML Files
  - Per-Plate Text Files
  - SQL Server (refer to Section 24.6.2, <u>Configuring a SQL Server Report</u> <u>Style</u>)

**Note:** Refer to Section 24.6.1, <u>*Report Styles*</u>, for examples and information on the types of reports that may be generated.

2. In **Report Location**, browse to find the location where the file is to be saved. The **Reporting** step is configured.

## 24.6.1 Report Styles

•

The report styles on data sets that may be generated are:

Text File — a single text file listing all plates in the system. The columns in this file include the SequentialID, plate name, plate bar code, well number, sample ID (if the SampleID data set is created), and volume. A column for each data set that has reporting enabled is also included (Figure 24-20).

**Note:** The SequentialID is a number assigned to each plate. This number is the only way to identify a plate in a report if the plate name or bar code were not specified.

🖺 Report.txt - Notepad	- 🗆 ×
<u>File Edit Search Help</u>	
<pre>\$equentialID,Name,Barcode,Well,Volume,My Data</pre>	
1,MyLabware,ABCDEF,A1,35,198	
1,MyLabware,ABCDEF,A2,35,196	
1,MyLabware,ABCDEF,A3,35,194	
1,MyLabware,ABCDEF,A4,35,192	
1, MyLabware, ABCDEF, A5, 35, 190	
1, MyLabware, ABCDEF, A6, 35, 188	
1, MyLabware, ABCDEF, A7, 35, 186	
1, MyLabware, ABCDEF, A8, 35, 184	
11, MyLadware, HBCDEF, H9, 35, 182	
1, MyLadware, HBGDEF, HT0, 35, 180	
1,MULADWARE,HBGDEF,HII,35,178	
1,MULADWARE,HBGDEF,H12,35,170	
1 Mul shuseo APCDEE D2 25 172	
1 Mul ahware ABCDEF B3 35 170	
1 Mulahware ABCDEF B& 35 168	
1. Mul abware - ABCDEF - 85 - 35 - 166	
1. Mul ahware. ABCDEF. B6.35.164	
1.MuLabware.ABCDEF.B7.35.162	
1.MuLabware.ABCDEF.B8.35.160	
1.MuLabware.ABCDEF.B9.35.158	
1,MyLabware,ABCDEF,B10,35,156	
1,MyLabware,ABCDEF,B11,35,154	
1,MyLabware,ABCDEF,B12,35,152	
1,MyLabware,ABCDEF,C1,35,150	
1,MyLabware,ABCDEF,C2,35,148	
1,MyLabware,ABCDEF,C3,35,146	
1,MyLabware,ABCDEF,C4,35,144	
1,MyLabware,ABCDEF,C5,35,142	
1,MyLabware,ABCDEF,C6,35,140	
1,MyLabware,ABCDEF,C7,35,138	
1,MyLabware,ABCDEF,C8,35,136	
1,MyLabware,ABCDEF,C9,35,134	
1,MyLabware,ABCDEF,C10,35,132	
1,MyLabware,ABCDEF,C11,35,130	<u> </u>

Figure 24-20. Report saved as Text File

• Per-Plate HTML Files — a folder containing an HTML file for each plate in the system. Each file contains the plate name, plate bar code, and an HTML table indexed by well number that includes the sample ID (if the SampleID data set is created) and volume. An entry for each data set that has reporting enabled is also included. The SequentialID is listed at the top of the file (Figure 24-21).

**Note:** The **SequentialID** is a number assigned to each plate. This number is the only way to identify a plate in a report if the plate name or bar code were not specified.

								Con	tents	of₿	arco	ie=A
Vo M	lume	_										
1413	' Dati	a 									_	
	1	2	3	4	5	6	7	8	9	10	11	12
A	35	35	35	35	35	35	35	35	35	35	35	35
	198	196	194	192	190	188	186	184	182	180	178	176
В	35	35	35	35	35	35	35	35	35	35	35	35
	174	172	170	168	166	164	162	160	158	156	154	152
C	35	35	35	35	35	35	35	35	35	35	35	35
	150	148	146	144	142	140	138	136	134	132	130	128
D	35	35	35	35	35	35	35	35	35	35	35	35
	126	124	122	120	118	116	114	112	110	108	106	104
Е	35	35	35	35	35	35	35	35	35	35	35	35
	102	100	98	96	94	92	90	88	86	84	82	80
F	35	35	35	35	35	35	35	35	35	35	35	35
	78	76	74	72	70	68	66	64	62	60	58	56
G	35	35	35	35	35	35	35	35	35	35	35	35
	54	52	50	48	46	44	42	40	38	36	34	32
H	35	35	35	35	35	35	35	35	35	35	35	35
	30	28	26	24	22	20	18	16	14	12	10	8

Figure 24-21. Report saved as Per-Plate HTML Files

•

Per-Plate Text Files — a folder containing a text file for each plate in the system. Each file contains the SequentialID, plate name, plate bar code, and columns that include the well number, and sample ID (if the SampleID data set is created). A column for each data set that has reporting enabled is also included.

**Note:** The SequentialID is a number assigned to each plate. This number is the only way to identify a plate in a report if the plate name or bar code were not specified.

**Note:** The text files in the folder for each plate in the system look similar to Figure 24-20.

SQL Server — a single database table listing all plates in the system. The columns of this table include the SequentialID, plate name, plate bar code, well number, and sample ID (if the SampleID data set is created). A column for each data set that has reporting enabled is also included. The table is created with a time-date stamp appended to the name (Figure 24-22).

**Note:** The **SequentialID** is a number assigned to each plate. This number is the only way to identify a plate in a report if the plate name or bar code were not specified.

**Note:** SQL Server is a Microsoft product that allows queries and provides analysis. The installation of this product is the responsibility of the user and is not supported by Beckman Coulter Service Engineers. SQL Server 2000 SP3 is the version requirement.

**Note:** Refer to Section 24.6.2, <u>*Configuring a SQL Server Report Style,*</u> for specific information on configuring a SQL Server report.

SequentialID	Name	Barcode	Well	Volume	
1		BARCODE11	A1	0	
1		BARCODE11	A2	0	
1		BARCODE11	A3	0	
1		BARCODE11	A4	0	
1		BARCODE11	A5	0	
1		BARCODE11	A6	0	
1		BARCODE11	A7	0	
1		BARCODE11	A8	0	
1		BARCODE11	A9	0	
1		BARCODE11	A10	0	
1		BARCODE11	A11	0	
1		BARCODE11	A12	0	
1		BARCODE11	B1	0	
1		BARCODE11	B2	0	
1		BARCODE11	B3	0	
1		BARCODE11	B4	0	
1		BARCODE11	B5	0	
1		BARCODE11	B6	0	

Figure 24-22. Report saved as SQL Server

## 24.6.2 Configuring a SQL Server Report Style

When SQL Server is chosen as the report style, the server, database, and table name are configured.

**Note:** SQL Server is a Microsoft product that allows queries and provides analysis. The installation of this product is the responsibility of the user and is not supported by Beckman Coulter Service Engineers. SQL Server 2000 SP3 is the version requirement.

Note: Refer to Figure 24-22 for an example of a SQL Server report style.

To configure a SQL server report style:

1. In Report Style, choose SQL Server. Figure 24-23 appears.

Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Consider Project Instrument       Image: Consider Project Instrument       Image: Consider Project Instrument         Image: Cons	In Biomek® Software - Method?* [New] File Edit Droject Instrument Execution Onlines Help		_ 🗆 🗵
Oreate Data Set     Start     Generate report on all labware in the system.       Oreate Data Set     Instrument Setup     Report Style:       Image: Set of the system     Image: Set of the system       Image: Set of the system     Image: Set of the system       Image: Set of the system     Image: Set of the system       Image: Set of the system     Image: Set of the system       Image: Set of the system     Image: Set of the system       Image: Set of the system     Image: Set of the system			
Management     Tander     Totol 1       Data Set Processing     Combine     Database:       Nove Edbware     Nove Labware     Note: This table will be created with a time-date stamp appended to its name.       Note: This table will be created with a time-date stamp appended to its name.     Note: This table will be created with a time-date stamp appended to its name.	Create Data Set Processing Reporting Step Pause Comment	Generate report on all laborate in the system. Report Style: SILSave: Serve: Database: Table Name: Note: This table will be created with a time-date stamp appended to its name.	
TL1 P8 P12 P16 P5 P9 P13 P17 P14 P18 P7 P11 P15 P19		TL1         P8         P12         P16           P5         P9         P13         P17           P14         P18         P19	

Figure 24-23. SQL Server report style chosen

2. Browse to find the location of the desired database on that server. SQL Server Connection (Figure 24-24) appears on top of the Reporting step configuration.

SQL Server Connection	
(local) DepartmentServer	User:
	Password:
	Use NT Authentication
	Connect
Databases:	
OK.	Cancel

Figure 24-24. SQL Server Connection

- 3. Highlight the desired server.
- 4. Enter the **User** and **Password**.

OR

Check Use NT Authentication.

**Note:** The administrator of the SQL Server Connection sets the logging protocol for user and password or NT authentication.

5. Choose **Connect**. A list of the available databases appears in Databases (Figure 24-25).

SQL Server Connection		- U ×
(local) DepartmentServer	User: sa	
	Password:	
	Use NT Authentication	
	Connect	
, Databases:		
DataSets		
OK	Cancel	

Figure 24-25. List of databases in SQL Server Connection

6. From **Databases**, highlight the desired database.

7. Choose **OK**. SQL Server Connection closes and the chosen server and database are entered in the Reporting step configuration.

Figure 24-26. Server and database chosen

8. From the **Reporting** step configuration (Figure 24-19), enter the name of the **Table Name**.

**Note:** Table names must begin with a letter, and only letters, numbers and underscores (\_) can be used.

**Note:** The table is created with a time-date stamp appended to the name.

# Handling and Preventing Errors

## 25.1 Overview

Biomek Software provides extensive error handling and error prevention capabilities to help make sure the method being built, edited, set up, and run executes appropriately. Included in the Biomek Software error handling and prevention capabilities are:

- Error notification and cause messages.
- Error recovery options.
- Constant internal validation during method building and editing.
- Internal validation before running a method.

## 25.1.1 Error Message Overview

There are several types of error messages that may display while building and running methods using Biomek Software. The types of error messages generated depend on the type of error that has occurred. While errors may originate from more than one source, most errors fall into the following general categories:

- <u>Configuration Errors</u> (Section 25.2) occur when a device or step is not configured correctly during method development or instrument configuration.
- <u>Validation Errors</u> (Section 25.3) occur when a method or step is validating.
- <u>Manual Control Errors</u> (Section 25.4) occur when a manual control movement requested of the ALPs, bridge, pod, head, or gripper is not compatible with Biomek instrument capabilities.
- <u>Run-Time Errors</u> (Section 25.5) occur while a method is running.
- <u>Light Curtain Errors</u> (Section 25.6) occur when there is a light curtain violation.
- <u>Liquid Level Sensing Errors</u> (Section 25.7) occur while attempting to determine the liquid level in a piece of labware during a method run.

- <u>Clot Detection Error Message (NX-S8 only</u>) (Section 25.8) occurs when a clot has been detected
- <u>Multiple Error Messages</u> (Section 25.9) numerous errors may occur during validation or method run.
- <u>Observed Errors</u> (Section 25.10) an error observed by the operator as the method runs.

## 25.1.2 Error Recovery Overview

Error recovery is the process of making changes to ensure a method can be completed. When changes are made and no subsequent errors are detected, the method completes. Depending on the conditions causing the error, the error message displays the appropriate error recovery options available.

Error recovery options may include:

- Fixing the error and continuing the method from the point of interruption.
- Stopping the current method, making modifications, and running the entire method from the beginning.
- Ignoring the error and continuing with the method.
- If a method using a dual-pod **FX** instrument is in progress when an error occurs, one pod can continue its tasks as long as its operations are not dependent upon the pod with the error.

### 25.1.3 Error Prevention Overview

Errors may be prevented by validating a method during the method-building process. Validation detects a large class of common errors before they occur by internally simulating the method prior to execution. This internal simulation provides an opportunity to change the method before it is run.

Operations for validating a method include:

- Validating the method immediately before a run starts by selecting Validate the current method before running it in Preferences.
- Individually validating steps by highlighting subsequent steps or the Finish step in the Method View.
- Individually validating operations by using the Single Step tool, available from the Execution menu.

# 25.2 Configuration Errors

Configuration errors occur when steps, labware, or devices are not configured correctly during method development or system setup.

## 25.2.1 Configuration Error Messages

When a configuration error is encountered during system setup, an error message may be displayed immediately (Figure 25-1).



Figure 25-1. Example of a configuration error generated due to a Liquid Type Editor configuration problem

A configuration error notification can also be generated as soon as a device configuration entry is made; for example, an incorrect entry in the Deck Editor appears red rather than black (Figure 25-2).



Figure 25-2. Example of a configuration error generated from the Deck Editor

When a step is not configured correctly, the step causing the error is highlighted in red in the Method View (Figure 25-3), and a message describing the error in detail is displayed at the bottom of the Biomek method editor and in a tool tip when the cursor is hovered over the step.





**Note:** The Error Message status bar may be hidden by choosing **Options>Toolbars** and deselecting **Method Error**.

To view multiple lines for a long error message, click the **More** button (Figure 25-3) to expand the error display to show the entire error message (Figure 25-4).

Click the **Less** button (Figure 25-4) to collapse the error display back to a single line (Figure 25-3).



Figure 25-4. Example of an expanded error message

## 25.2.2 Recovering from Configuration Errors

To recover from configuration errors, ensure that all of the facts pertaining to the step, labware, or device configuration have been supplied and are acceptable values. For example, make sure that volumes do not exceed the maximum capacity of the specified tips or labware.

## 25.3 Validation Errors

Validating a method prior to execution internally simulates the method to test for errors. If an error is detected during validation, an error message displays information about the error.

**Note:** In this section, validation refers only to running methods in simulation mode to test for step or instrument configuration errors. It does not refer to validating a revision of a method using Beckman Coulter Accounts & Permissions. Refer to Section 12.15, <u>Checking Out a Method</u> for more information about validating checked in revisions of methods.

**Note:** Validation errors can occur due to step configuration or instrument configuration errors, as noted in Section 25.2, *Configuration Errors*.

## 25.3.1 Validation Error Messages

Errors encountered during validation may display the following information in the error messages:

- Source
- Type
- Cause
- Location
- Error Code
- Recovery Option(s)
- Date and Time

## 25.3.2 Validating a Method Before Run

When Validate the current method before running it is checked in Preferences (Figure 25-5), a method is internally simulated to test for errors when the method is run. If no errors are detected during this validation, the method is executed. If an error is encountered during this validation, the process stops and an error message displays information about the nature of the error.

When Validate the current method before running it is selected, the first step causing an error is highlighted in the Method View (Figure 25-6). Validate the current method before running it is enabled by default.
To prevent the system from slowing down due to unnecessary memory consumption, use Look ahead up to [1800] seconds in the method while it is running (refer to Section 25.3.2.1, <u>Using Look ahead up to [1800] seconds in the method while it is running</u>).

Preferences	General
General	✓ Validate the current method before running it.
Errors	Ask for confirmation before removing a step from a method.
	Automatically save all Editor settings listed below. Save changes to the screen position of the Editor. Save changes to the toolbar and step palette layout of the Editor. Save changes to the steps configured within step palettes of the Editor. Look ahead up to 1800 seconds in the method while it is running. Method Validation Simulates methods internally to test for errors before the method is run.
	OK Cancel Reset

Figure 25-5. Validate the current method before running it option in Preferences



Figure 25-6. Example of a validation error message generated during method development as a result of incomplete step configuration

To use Validate the current method before running it:

- 1. From the toolbar, choose **Options>Preferences**. Preferences appears (Figure 25-5).
- 2. Make sure Validate the current method before running it is checked.
- 3. Choose **OK**.
- 4. Run the method.

**Note:** Validating a complex or lengthy method could result in occupying system resources for a long period of time. Validate the current method before running it can be disabled prior to running a previously validated method, or a method not requiring validation, by using Options>Preferences and deselecting Validate the current method before running it.

## 25.3.2.1 Using Look ahead up to [1800] seconds in the method while it is running

When a method is run, the software translates the steps of the method into "to do" lists of actions to be performed on each device. This translation process occurs while the method is running. Since methods can be arbitrarily long, the lists of actions can consume large amounts of computer memory, possibly slowing the system.

To prevent unnecessary slowing of the system due to memory consumption, the Look ahead up to [1800] seconds in the method while it is running option in Preferences (Figure 25-5) designates a time-based governor for suspending the translation process. The time in seconds entered into Look ahead up to [1800] seconds in the method while it is running is a threshold at which the translation process may be suspended so that the "to do" lists do not become too large. If each currently active device requires more than the time-designated threshold to complete its "to do" list, the suspension may occur. (The translation process is never suspended during a Just In Time step.) Once the system determines that some device may finish its list earlier than this threshold, it automatically resumes the translation process.

**Note:** To prevent problems with executing actions that should occur simultaneously, such as in a dual-pod system, it is important to enter a high enough Look ahead up to [1800] seconds in the method while it is running time.

#### 25.3.2.2 Validation Error Messages Detected During Method Development

During method building, Biomek Software validates constantly. Errors that are detected during this process are displayed when a subsequent step in the method is highlighted. When the Finish step is highlighted, the entire method is validated. The step causing the error is highlighted in red in the Method View, and a message describing the error in detail is displayed in the status bar at the bottom of the method editor as well as a tool tip when the mouse pointer is hovered over the step.

### 25.3.2.3 Validation Error Messages Detected During Method Run

Validation errors may also be detected during method execution. The display of the error message depends on whether Validate the current method before running it is selected or not (refer to Section 25.3.2, *Validating a Method Before Run*).

If Validate the current method before running it is selected, the method is validated before it starts. If an error is found, it displays an error message similar to Figure 25-7.



Figure 25-7. Example of a validation error message detected before method run

If a method has not been previously validated and the Validate the current method before running it option is not selected, it displays an error message similar to Figure 25-8.

Biomek® Software	
Dialogs The left pod should ha Validation failure in Ne	Validation failure in New Tips: No usable AP96_200uL tip boxes found which Pod1 can reach.
	Abort Snap

Figure 25-8. Example of a validation error message detected during method run

## 25.3.3 Recovering from Validation Errors

To recover from Validation errors encountered during method execution, choose one of the options provided on the error message. Validation error messages have either the OK recovery option or both Abort and Snap recovery options.

Validation error message recovery options include:

- Abort stops the method after Biomek instrument completes the move in progress.
- Snap stops the method after Biomek instrument completes the move in progress and creates a Continuation method (refer to Section 25.11.2, <u>Snapping a Continuation</u>).
- If the method is in the process of performing a move, such as aspirating or dispensing, the Biomek instrument completes the move; however, the instrument may not complete the entire step. For example, if the Biomek instrument is in the process of performing a **Transfer** step and an error occurs while it is aspirating, selecting **Abort** or **Snap** results in completing the aspiration but not the **Transfer** step.
  - > **3000** Snapping a Continuation is not available.
- OK stops the method, allowing changes.

**Note:** OK is presented only as an option for a validation error detected before the run has started if Validate the current method before running it is turned on (refer to Section 25.3.2, *Validating a Method Before Run*).

### 25.3.3.1 Choosing Abort

Choose Abort to stop a method after completing the move in progress and leave the method open for editing.

To use Abort:

- 1. Read the error message to determine the cause of the problem.
- 2. Choose **Abort** to make the necessary changes in the original method file.

### 25.3.3.2 Choosing Snap (not available on a 3000)

Snap allows the method to be modified without changing the original method. When Snap is chosen, a Continuation method is created. A Continuation method is a new method consisting of the step causing the error, any steps not yet completed, and, if applicable, any incomplete substeps of the step which generated the error (refer to Section 25.11.2, *Snapping a Continuation*).

To use Snap:

- 1. Read the error message to determine the cause of the problem.
- 2. Choose **Snap** to create a Continuation method (refer to Section 25.11.2, *Snapping a Continuation*).

## 25.4 Manual Control Errors

Manual Control errors occur when a manual control movement requested of the bridge, pod, head, or gripper is not compatible with Biomek instrument capabilities.

# 25.4.1 Manual Control Error Messages and Recovery

Manual Control errors have only one recovery option, and that is to choose OK on the error message (Figure 25-9). Manual Control error messages should be carefully read to determine whether the problem lies in the hardware or software. Once that determination has been made, appropriate changes can be made to recover from the error.

To recover from Manual Control errors:

- 1. Read the error message to determine the problem.
- 2. Choose **OK**, then make the necessary changes.

Error	×
$\otimes$	Move aborted. It violates the following constraint(s): Z max=-2.200
	<u> </u>

Figure 25-9. Example of a Manual Control error because the Biomek FX tried to move the pod too far on the Z axis

## 25.5 Run-Time Errors

Run-Time errors result from problems that occur while a method is executing; for example, tips cannot load because the tiploader rods are not operating correctly. Run-Time error recovery options are presented on each error message.

## 25.5.1 Run-Time Error Messages

Run-Time error messages (Figure 25-10 and Figure 25-14) display the:

- Source
- Device
- Error Message
- Error Number
- Recovery Option(s)
- Date and Time



Figure 25-10. An example of a Run-Time error due to undetermined gripper state

## 25.5.2 Recovering from Run-Time Errors

To recover from errors, choose one of the Recovery Options and correct the problem that caused the error.

The options available for recovering are:

- Abort stops the method after Biomek instrument completes the move in progress.
- Retry attempts to complete the move in progress when the error was encountered and complete the method (refer to Section 25.5.2.2, <u>*Choosing*</u><u>*Retry*</u>).



# WARNING: The Ignore error recovery option is potentially dangerous since almost every action depends upon the successful completion of previous actions. Choose Ignore at your own risk.

- Ignore skips the move in progress and proceeds with the rest of the method, if possible (refer to Section 25.5.2.3, <u>Choosing Ignore</u>).
- Snap stops the method after the Biomek instrument completes the move in progress and creates a Continuation method (refer to Section 25.10.1.3, <u>Snapping a Continuation (not available on a 3000)</u>).

**Note:** If the method is in the process of performing a move, such as aspirating or dispensing, the Biomek instrument completes the move; however, the instrument may not complete the entire step. For example, if the Biomek instrument is in the process of performing a Transfer step and an error occurs while it is aspirating, selecting Abort or Snap results in completing the aspiration but not the remainder of the Transfer step.

> **3000** — Snapping a Continuation is not available.

After the errors are corrected, the method can be rerun or continued from where the error occurred.

#### 25.5.2.1 Choosing Abort

Choose Abort to stop a method after completing the move in progress and leave the method open for editing.

To recover from errors using Abort:

- 1. Read the error message to determine the cause of the problem.
- 2. Choose Abort to make the necessary changes in the original method file.

#### 25.5.2.2 Choosing Retry

Choose **Retry** to attempt to complete the step that is in progress when the error was encountered. If the step is completed successfully, the method continues.



CAUTION: Resuming a method assumes that the Biomek instrument is in the same state as when the error occurred. The pod may be moved to deal with a problem, but no changes can be made to the Biomek instrument deck.

To recover from errors using **Retry**:

- 1. Read the error message to determine the cause of the problem.
- 2. Examine the instrument to verify it is safe to restart the method.
- 3. Choose **Retry**. The instrument attempts to perform the move in progress when the error occurred, and, if successful, the method continues. If the method cannot continue after choosing **Retry**, another error message appears.

#### 25.5.2.3 Choosing Ignore



WARNING: The Ignore error recovery option is potentially dangerous since almost every action depends upon the successful completion of previous actions. Choose Ignore at your own risk.

Choose **Ignore** to bypass the error and proceed with the method. The move in progress when the hardware error is detected does not complete, but the method attempts to continue the run.

To recover from errors using Ignore:

- 1. Read the error message to determine the cause of the problem.
- 2. Choose **Ignore**. The error is ignored, and the method continues to run, if possible. If the method cannot continue after choosing **Ignore**, another error message appears.

**Note:** Ignore may be useful for failed SILAS steps such as failed bar code reads or device time-out errors that do actually complete.

#### 25.5.2.4 Choosing Snap (not available on a 3000)

Snap allows the method to be modified without changing the original method. When Snap is chosen, a Continuation method is created (refer to Section 25.11, <u>Using</u> <u>Continuations (not available on a 3000)</u>).

To recover from errors using Snap:

- 1. Read the error message to determine the cause of the problem.
- 2. Choose **Snap** to create a Continuation method (refer to Section 25.11.2, *Snapping a Continuation*).

## 25.6 Light Curtain Errors

The light curtain is a perimeter diffuse-reflective safety component that stops the Biomek FX or Biomek NX instrument when objects more than one inch in diameter penetrate the light curtains protective zone. Light curtain errors occur when an object penetrates (or violates) the light curtain protective zone and a light curtain error message is displayed.

3000 — Does not have a light curtain, although pressing the STOP button on the front rail of the instrument allows recovery options similar to those for a light curtain violation (refer to Section 25.6.3, <u>Using the Stop Button on</u> <u>the Biomek 3000</u>).

A light curtain violation does not stop the entire instrument. Certain on-deck devices and all devices controlled by SILAS may continue to operate despite a light curtain violation. However, new commands will not be issued once a light curtain violation has occurred and the system will come to a stop once all currently issued commands are completed.

## 25.6.1 Light Curtain Error Messages

Light curtain error messages appear when there is a light curtain violation. The light curtain error message displays:

- a graphic.
- an OK button.
- error message text.

## 25.6.2 Recovery Options for Light Curtain Errors

Light curtain errors can be recovered from by:

- Removing the obstruction and responding to the light curtain error message.
- Aborting a method during a light curtain violation using the buttons on the toolbar.
- Snapping a continuation during a light curtain violation using the buttons on the toolbar.

**Note:** If a light curtain violation occurs during a pause, the light curtain error must be resolved before the method can be resumed.

## 25.6.2.1 Removing the Obstruction and Responding to the Light Curtain Error Message

When a light curtain error message appears, the following text is displayed in the error:

"A light curtain violation has occurred. When you are clear of the light curtain, press OK. The pods will move back to where they were before the light curtain violation and the method will resume. If you wish to abort the method, you may press the stop button on the toolbar."

To recover from a light curtain error message:

- 1. Remove the violating object from the light curtain.
- 2. Click **OK**. The pod position is restored to its position before the light curtain violation and the method continues. If the light curtain violation remains, another error message is displayed.

**Note:** The system will not check or restore the position of the D axis. More specifically, the system will not resume a pipetting action from the point the light curtain error occurred. The entire pipetting operation must be restarted; it cannot be completed from where it left off.

FX — If using a dual-pod system, each pod will be moved back to position one at a time to avoid collisions.

#### 25.6.2.2 Aborting a Method During a Light Curtain Violation

Abort stops the method and leaves it open for editing. The method may not be continued from the point the error occurred.

To abort during a light curtain violation, click **Abort** on the toolbar.

## 25.6.2.3 Snapping a Continuation During a Light Curtain Violation

Snap stops the method and creates a Continuation method. A Continuation method allows the method to be modified without changing the original method.

To snap a continuation during a light curtain violation, click **Snap Continuation** on the toolbar. A Recovery step is inserted at the start of the continuation method that will restore pod positions.



CAUTION: The partially completed steps assume the pods are in the same position they were in when the method was halted. Removing the Recovery step without also removing these partially completed steps could result in a crash.

If it is not desired for the pod(s) to restore position, delete the Recovery step and any partially complete steps that come after it in the continuation.

**Note:** For more information about continuations, refer to Section 25.11, <u>Using</u> <u>Continuations (not available on a 3000)</u>.

## 25.6.3 Using the Stop Button on the Biomek 3000

Pressing the emergency STOP button on the front rail of the Biomek 3000 stops movement of the bridge and head assembly.

When the STOP button is pressed, Figure 25-11 appears.



Figure 25-11. Choosing STOP on the front rail of the Biomek 3000

Two options are possible after pressing the emergency STOP:

• Choose **OK** to resume the method.

**Note:** If the STOP button is pressed while a tool is being loaded, the instrument must be homed before the method is resumed.

• From the toolbar, choose **Execution>Stop** to abort the method.

## 25.7 Liquid Level Sensing Errors

The technologies for Liquid Level Sensing are different on the Biomek FX and NX instruments and the Biomek 3000 which accounts for the difference in error messages and recovery options for the instruments.

## 25.7.1 Liquid Level Sensing on the FX and NX-S8

If the liquid level in a piece of labware is not known, the Span-8 Pod can determine the liquid level when Liquid Level Sensing (LLS) capable tips are used.

LLS tips are used to determine the liquid level in a piece of labware by detecting a shift in the capacitance of a probe. The LLS tip moves to a specified height within a well and then slowly moves down into the well. When the tip contacts liquid, there is a large change in the capacitance detected. The liquid level is sensed by determining the height at which this change in capacitance occurs.

## 25.7.2 Liquid Level Sensing on the Biomek 3000

The P200L Single-Tip Pipette Tool includes patented technology that sonically detects the liquid level. Liquid level sensing is performed using an acoustic process involving a transmitter and receiver within the single-channel tools. The transmitter emits a sound wave through the tip that bounces back when it contacts liquid. The receiver detects the wave as it bounces back past the end of the tip.

## 25.7.3 Recovering from Liquid Level Sensing Errors

Liquid level sensing error messages appear when no liquid is sensed or when the amount of liquid sensed is incompatible with the operation performed. Liquid level sensing error messages display one of the following error messages:

"There is insufficient volume in well ## for this operation. Check mandrel # before continuing."

"There is too much volume in well ## for this operation. Check mandrel # before continuing."

"Validation failure in Move to #. Consecutive senses did not provide a stable value. This can be caused by bubbles in the well or by the meniscus of the liquid."

Liquid level sensing errors display a variety of recovery options in the error message, based on the instrument and type of liquid level sensing error encountered.

#### 25.7.3.1 No Liquid is Detected on a FX or NX-S8

When no liquid is detected on a FX or NX-S8, the error message provides information on the probes (mandrel) generating the error and the following recovery options:

- Seek Again Repeats liquid level sensing operation.
- Seek Slower— Repeats liquid level sensing with tip descending into the labware at a slower rate of speed.
- Move Up Positions tip higher in well and repeats liquid level sensing.
- Snap [a Continuation] Stops the method after the instrument completes the move in progress and creates a Continuation method (refer to Section 25.11, <u>Using Continuations (not available on a 3000)</u>).
- Abort Stops the method after the instrument completes the move in progress.

## 25.7.3.2 Liquid Level Sensing Returning an Unstable Value on a FX or NX-S8

When a consistent value is not detected after numerous attempts to verify the liquid level on a FX or NX-S8, the error message (Figure 25-12) provides information on the pipetting operation generating the error, specific information on the error received, possible causes for the error, and the following recovery options:

- Abort Stops the method after the instrument completes the move in progress
- Snap [a Continuation] Stops the method after the instrument completes the move in progress and creates a Continuation method (refer to Section 25.11, <u>Using Continuations (not available on a 3000)</u>)

Biomek® Software	
Validation failure in Move to 0 cm, 0°, =C_Height mm: Consecutive senses di	d not provide a stable
This can be caused by bubbles in the well or by the meniscus of the liquid.	
<u>Abort</u>	
	3/9/2001 10:20:44 AM

Figure 25-12. Inconsistent liquid level value detected error message

#### 25.7.3.3 Liquid Level is Incompatible with the Operation

When the liquid level detected is incompatible with the pipetting operation, the error message is dependent on the instrument type (refer to Section 25.7.3.3.1, *Incompatible Pipetting Operation on a FX or NX-S8* and 25.7.3.3.2, *Incompatible Pipetting Operation on a 3000*) and provides information on the specific probe (mandrel) generating the error and recovery options.

## 25.7.3.3.1 Incompatible Pipetting Operation on a FX or NX-S8

The following recovery options are available:

- Seek Again Repeats liquid level sensing operation.
- Pipette Air Pipettes air when liquid is not available.
- Pipette From Bottom Moves tip to bottom of well and continues pipetting operation.
- Snap [a Continuation] Stops the method after the instrument completes the move in progress and creates a Continuation method (refer to Section 25.11, <u>Using Continuations (not available on a 3000)</u>).
- Abort Stops the method after the instrument completes the move in progress.

#### 25.7.3.3.2 Incompatible Pipetting Operation on a 3000

The following recovery options are available:

- Pipette Air Pipettes air when liquid is not available.
- Pipette From Bottom Moves tip to bottom of well and continue pipetting operation.
- Abort Stops the method after the instrument completes the move in progress.

## 25.8 Clot Detection Error Message (NX-S8 only)

In order for clots to be detected, conductive disposable tips (called LLS in the Tip Type Editor) or fixed tips must be used and the technique must include clot detection. Biomek Software views the detection of a clot as an error and provides recovery options when a clot is detected.

## 25.8.1 Retrying Has Not Resolved the Clot

If the clot has not been resolved by retrying and Prompt user for further input is selected in the technique, an error message appears providing information on the probes (mandrel) generating the error and the following recovery options:

- Dispense, Aspirate Again, then Retry Detection Dispenses the liquid back into the well, aspirates again, and retries clot detection.
- Ignore and Continue Ignores the error and continues the method.

**Note:** Ignoring the error and continuing the method when a clot exists may contaminate the deck.

- Snap [a Continuation}— Stops the method after the instrument completes the move in progress and creates a Continuation method (refer to Section 25.11, <u>Using Continuations (not available on a 3000)</u>).
- Abort Stops the method after the instrument completes the move in progress.

Biomek® Software			
A clot was detected on probe #4.			1
Dispense, Aspirate Again, then Retry Detection	Ignore and Continue	Snap 2	Abort

Figure 25-13. Clot detection error

## 25.9 Multiple Error Messages

When multiple error messages are generated during method validation, a dialog appears in the **Instrument Setup** prompt that contains a list of the errors (Figure 25-14). Specific information about each error in the list is displayed in the message area by highlighting the message in the error list. Recovery options for each message are also displayed.

Biomek® Software	
Dialogs The left pod should ha Validation failure in Ne	Validation failure in New Tips: No usable AP96_200uL tip boxes found which Pod1 can reach.
	Abort Snap
Error list	Message area
	Recovery options

Figure 25-14. Error message dialog with multiple errors

When multiple errors are encountered, the error messages appear in prompts similar to the examples displayed in Figure 25-15 and Figure 25-16.



Figure 25-15. An example of multiple error messages



Figure 25-16. An example of multiple error message generated during Run-Time

## 25.9.1 Recovering from Multiple Errors

Multiple error messages should be read carefully to determine where the problem lies: in the hardware, software, or in the method. Once that determination has been made, appropriate changes can be made to recover from the errors. Errors can be resolved in any order. Refer to Section 25.5.2, *Recovering from Run-Time Errors*, for more information on error recovery options.

## 25.10 Observed Errors

Occasionally, errors are observed while watching the Biomek instrument execute a run. For example, the instrument may attempt to aspirate from an empty or missing microplate.

Observed errors may result from setup, hardware, or software issues.

## 25.10.1 Recovering from Observed Errors

The Biomek Software provides three options for recovering from observed errors while the method is running:

- Pause halts a method without causing a light curtain violation.
- Stop halts a method when there is no intent to resume the method from the point where it was stopped.
- Snap stops the method after Biomek instrument completes the move in progress and creates a Continuation method (refer to Section 25.11, <u>Using</u> <u>Continuations (not available on a 3000)</u>).
  - > **3000** Snapping a Continuation is not available.

#### 25.10.1.1 Pausing a Method

Use Pause to halt a method without causing a light curtain violation. Pause causes the method to halt after the Biomek instrument has completed the move in progress. For example, if the Biomek instrument is in the process of loading tips when Pause is selected, it finishes loading the tips before the system halts. When the method is resumed, it continues as though it were never halted.

**Note:** A method cannot be edited during a pause.

Use one of the following procedures to pause a method:



CAUTION: No changes to the Biomek instrument state are permitted while a method is paused. Changes can be made to the labware contents, but not the deck or the Biomek devices.

1. Choose Execution>Pause.

OR



Choose **Pause** on the toolbar. The instrument completes the move in progress, stops, and deactivates the light curtain.

- 2. Make the necessary changes to correct the observed error.
- 3. To resume the method run, choose **Pause** again. The light curtain is reactivated, and the method resumes from the point where it was paused.

#### 25.10.1.2 Stopping a Method



## WARNING: The light curtain is a safety device. Use it to stop a method only in an emergency.

Use **Stop** to halt a method during its run when there is no intent to resume method execution. If the pod is in the process of a move, the operation is not completed. Since the method is halted, this option allows edits to the method or changes to the hardware and deck.

**Note:** Stop may not halt operations already in progress on external devices, such as Stacker Carousels.

Use one of the following procedures to stop a method during a run:

Choose Execution>Stop.

OR

• Choose **Stop** on the toolbar.

OR

FX, NX-S8 — Break (violate) the light curtain by no more than one inch, and Abort (refer to Section 25.6, <u>Light Curtain Errors</u>).

OR

3000 — Press the STOP button on the front of the instrument (refer to Section 25.6.3, <u>Using the Stop Button on the Biomek 3000</u>) and choose Stop from the toolbar.

Note: Use standard procedures to rerun the method.

## 25.10.1.3 Snapping a Continuation (not available on a 3000)

Another way to recover from observed errors is by snapping a Continuation. Snapping a Continuation stops the current method and allows corrections in the middle of a method run. The method can be restarted from the point where the method was halted.

To snap a continuation, choose **Snap** to create a Continuation (refer to Section 25.11, *Using Continuations (not available on a 3000)*).

## 25.11 Using Continuations (not available on a 3000)

Continuation methods allow error recovery or method modification after a method has begun execution. A Continuation is a temporary method with steps generated automatically that complete the original method from the point at which the Continuation method was snapped. A Continuation method is displayed in the Method View and can be edited; however, it cannot be saved and used as a regular method.

FX — If two pods are in use, one pod can continue to perform operations after an error occurs on the other pod. In this case, it is acceptable to allow the pod that is still performing operations to complete what it can before snapping a Continuation.

**Note:** When Beckman Coulter Accounts & Permissions is enabled, only users with permission to run the method may snap, edit, and run continuations (refer to Chapter 2, *Using Accounts & Permissions*).

## **25.11.1 Characteristics of Continuations**

Continuation methods have several distinct characteristics such as:

- A Continuation method may not have a Start step.
- Icons for steps unavailable for editing appear gray, not colored. If a step containing multiple operations, such as **Transfer** or **Combine**, is in progress when a Continuation method is created, the incomplete operations are displayed as substeps under the main step at the start of the Continuation method in a Recovery Step (Figure 25-17). For example, if the **Transfer** step is in progress when a Continuation is snapped, the first step in the Continuation method is **Transfer**. Double-click on the **Transfer** step to view the substeps.

**Note:** For light curtain violations, the Recovery step also includes steps to restore the pod to its position when the Light Curtain violation occurred.

- When a Continuation method is run successfully, the Continuation automatically closes.
- Only the last Continuation method generated is saved. Each subsequent Continuation method overwrites the previous Continuation method.



#### WARNING: To eliminate the possibility of replacing a Continuation method before corrections are made, it is recommended that a Continuation method be edited and run as soon as it is created.

• A Continuation method that contains a partially completed loop appears as one or more LOOP steps. The first LOOP is the partially completed loop iterations, while the last LOOP step represents the remaining iterations of the loop. The number appearing with the first LOOP indicates the iteration number the loop was on when the Continuation method was created (Figure 25-17).

**Note:** There are many dependencies among actions taken by different steps; therefore, take great care in modifying steps or substeps in a Continuation method.



Figure 25-17. A Continuation method snapped during a Loop step

## 25.11.2 Snapping a Continuation



CAUTION: A Continuation method assumes that the instrument is in the same state as when the Continuation was snapped. If this is not true, the Continuation method must be edited to account for the changes.

A Continuation method is a temporary method created when Snap is chosen on an error message or when Snap Continuation is chosen from the toolbar during a method run.

To abort the current method and create a Continuation method:

1. Choose **Execution>Snap Continuation**. Snap Continuation Information appears (Figure 25-18).

OR

Choose **Snap** from an error dialog.

OR



Choose **Snap Continuation** on the toolbar. Snap Continuation Information appears (Figure 25-18).

Inform	ation	×
į)	You have chosen to snap a continuation. This will close the current method and replace it with a new method that will complete th remaining work. If you have made unsaved changes to the current method, you will have a chance to save them before it closes	ne ;.

Figure 25-18. Snap Continuation Information

2. Choose **OK**. The **Confirm** prompt (Figure 25-19) appears if there are unsaved changes in the original method.

Close Method Confirmation				
The Method "Sample Method" has been modified. You may save the changes to the working revision or check in the changes.				
Save	⊆heck In	Discard Changes	Cancel Method Close	

Figure 25-19. Confirm saving changes to method

- 3. If prompted to save, choose from among the following options:
  - Choose **Save** to save and close the current method and to open the Continuation method.

**Note:** Refer to Section 12.9, *Saving a Method* for more information. After the method is saved, the Continuation method opens.

• Choose **Check In** to save and check in the method and open the Continuation method.

**Note:** Refer to Section 12.10, <u>*Checking In a Method*</u>, for more information. After the method is saved and checked in, the Continuation method opens.

- Choose **Discard Changes** to close the original method without saving the changes. Any unsaved changes in the method are lost and the Continuation opens and is active for editing.
- Choose Cancel Method Close to keep the original method active without saving and create a Continuation method for later use (refer to Section 25.11.3, *Loading and Running a Continuation*).

**Note:** Once the Continuation method is opened, the title changes in the titlebar indicating that the Continuation method is now active; for example, <Method Name> Cont'd.

**Note:** To access a Continuation method that is saved but not executed, follow the procedures outlined in Section 25.11.3, *Loading and Running a Continuation*.

- 4. Edit the **Continuation** method as needed (perhaps to correct an error).
- 5. Run the **Continuation** method.
- 6. Repeat steps 1 through 5 as many times as necessary to correct all errors.

When the Continuation method is run successfully, the Continuation method closes and cannot be loaded and run again.

## 25.11.3 Loading and Running a Continuation

If the Cancel Method Close option is chosen when snapping a Continuation (refer to Section 25.11.2, *Snapping a Continuation*), a Continuation method is created and saved for later use. To use the Continuation method later, it must be loaded.

**Note:** Continuation methods performed successfully are automatically deleted. Load Continuation is available only if a Continuation method exists. If a second Continuation is snapped before the first Continuation method is run, the first Continuation method is lost.

To load and run a Continuation method:

- 1. Choose **File>Load Continuation**. The Continuation method loads and appears in Method View.
- 2. Edit the Continuation method by adding, deleting, and editing steps.



3. Run the **Continuation**.

**Note:** If the Continuation method runs successfully, the Continuation method closes, and the Continuation is deleted. If further errors occur while running the Continuation method, handle them as outlined in this chapter.

## 25.12 Performing Single Operations Within Steps

The Single Step option performs single operations within Biomek steps, such as Aspirate or New Tips. Single Step pauses the Biomek instrument between each operation in a step, allowing visual verification that the operation is correct. Performing single operations can help when fine tuning a method.

To open Single Step:

Choose **Execution>Single Step**. Single Step appears (Figure 25-20):

Single Step			
Start the method from the execution menu; pending actions will appear below.			
	I Single Step		
	Launch All		
	E <u>s</u> it		

Figure 25-20. Single Step prompt

Single Step allows a method to be executed, one operation at a time. For example, to visually verify that tips are loading properly, use Single Step to walk through the New Tips step operation by operation. It is important to move slowly when using Single Step. Moving too quickly through the method may by-pass steps that need verification.

When visually verifying steps using Single Step, there are multiple operations that occur. For example, the New Tips step may walk through operations that include:

- Moving to a tip box.
- Extending the gripper.
- Gripping the tip box.
- Lifting the tip box.
- Moving to the tip loader.
- Placing the tip box on the tip loader.
- Retracting the gripper.
- Preparing to load tips.
- Loading Tips.

To run a method using Single Step:

- 1. Open the Single Step prompt. This can be done during a method run.
- 2. Start the method using the **Start** button on the toolbar. The method will stop for the first movement of a pod or bridge and the prompt will display the operation to perform (Figure 25-21).

Single Step		
Pod1 Absolute Move X : -1.431 Y : -4.986		
	Launch	
		 ☑ Single Step
		Launch All
		(E <u>x</u> it



- 3. To perform the next move, choose Launch.
- 4. To perform all steps in the method after verifying the desired steps, uncheck **Single Step**.

**Note:** Choose **Launch All** to launch all operations listed in the Single Step prompt. When there is more than one Launch button available, Launch All simultaneously presses all the Launch buttons.

5. Choose Exit to leave Single Step at any time.

# Generating and Using Log Data

## 26.1 Overview

Logs provide text records of a method run. The contents of the text record, or log file, is based upon the type of log requested. Logs include information on liquid transfer operations, errors, and gripper movements, as well as information contained in data sets.

Log types include:

- Details captures every operation that occurs during a method run, including Instrument Setup, absolute moves, pod movements, and gripper movements.
- Errors captures any errors that occur during a method run.
- Pipetting captures pipetting operations completed by a pod during a method run, deck location, and labware name or type.
  - **FX** available only for instruments with a Multichannel pod.
  - > **3000** available with all pipetting tools.
  - > **NX-MC** available for this instrument.
  - > **NX-S8** not available for this instrument.
- Span8Pipetting captures only Span-8 pipetting operations that occur during a method run, including location and labware name or type.
- Span8Transfer captures only Span-8 transfer operations that occur during a method run, including location and labware name or type.
  - FX Span8Pipetting and Span8Transfer are available only for instruments with a Span-8 pod.
  - 3000, NX-MC Span8Pipetting and Span8Transfer are not available for these instruments.
  - > **NX-S8** available for this instrument.

**Note:** The addition of SampleID data sets also allows the Span8Pipetting and Span8Transfer logs to include well identification information.

- UnifiedPipetting captures pipetting operations, along with Sample IDs for wells, performed by any pod, including information on the where the aspirate or dispense operation occurs.
- UnifiedTransfer captures transfer operations, along with Sample IDs for wells, performed by any pod, including information on the source and destination labware.

Log files are used to generate reports and capture the history of a method run. Log files can be printed using a standard text editor or by inserting them into a spreadsheet program, such as Microsoft® Excel.

## 26.2 Types of Logs

The seven types of logs available capture either a complete account of a method or specific operations that occur during a method run. The type of log captured is indicated by the file name in the Logs subdirectory. Logs is a subdirectory of the Biomek directory.

To access Logs:

Browse to C:\Documents and Settings\All Users\Documents\Biomek\Logs.

OR

Choose **Start>Beckman Coulter>Tools>Biomek Files** to open the Biomek directory and access Logs.

Examples of the naming procedures for log files are:

- Details Details03-09-2000 17.11.20.log
- Errors Errors03-27-2000 10.27.41.log
- Pipetting Pipetting03-09-200 17.11.20.log
- Span8Pipetting Span8Pipetting03-05-2001 16.51.18.log
- Span8Transfer Span8Transfer03-05-2001 16.51.18.log
- UnifiedPipetting UnifiedPipetting09-27-2002 12.11.41.log
- UnifiedTransfer UnifiedTransfer09-27-2002 12.11.41.log

### 26.2.1 Details Log

The **Details** log provides comprehensive information about each move made by the Biomek instrument. It is used when trying to identify and correct movement problems for any pod.

Refer to Section 26.3.1, *Details Log Contents*, for specific information about the Details Log contents.

**Note:** The default setting for the **Details** log is **Off**.

**Note:** The **Details** log is most useful to Beckman Coulter Service Engineers when troubleshooting an instrument.

## 26.2.2 Errors Log

The Errors log lists any errors or interruptions that occur during a method run. It is used to get the exact error message to include in a report or a discussion with Technical Support.

Refer to Section 26.3.2, *Errors Log Contents*, for specific information about the Errors log contents.

**Note:** The default setting for the Errors log is On.

## 26.2.3 Pipetting Log

The **Pipetting** log lists aspirate and dispense operations that are executed during a method run. It is used when validating a method or generating a comprehensive report about the pipetting operations used during a method.

- **FX** available only for instruments with a Multichannel pod.
- > **3000** available with all pipetting tools.
- > **NX-MC** available for this instrument.
- > **NX-S8** not available with this instrument.

Refer to Section 26.3.3, *<u>Pipetting Log Contents</u>*, for specific information about the Pipetting log contents.

Note: The default setting for the Pipetting log is Off.

### 26.2.4 Span8Pipetting Log (FX, NX-S8 only)

The Span8Pipetting log lists aspirate and dispense operations performed by the Span-8 Pod during a method run. It is used when validating a method or generating a comprehensive report about Span-8 pipetting operations used during a method. The SampleID data sets allows the Span-8Pipetting log to include the sample ID assigned in the Instrument Setup step for wells on labware. The sample ID is captured in the Well Identification field.

Refer to Section 26.3.4, <u>Span-8Pipetting Log Contents (FX, NX-S8 only)</u>, for specific information about the Span8Pipetting Log contents.

- FX default setting is On when a Span-8 Pod is installed and configured in Instrument Setup. However, a Span8Pipetting log is available only after a method has been run with a Span-8 pod.
- NX-S8 default setting is On. However, a Span8Pipetting log is available only after a method has been run with a Span-8 pod.

## 26.2.5 Span8Transfer Log (FX, NX-S8 only)

The Span8Transfer log lists liquid transfer operations performed by the Span-8 Pod during a method run. It is used when validating a method or generating a comprehensive report about Span-8 liquid transfer operations used during a method. The SampleID data sets allows the Span-8Transfer log to include the sample ID assigned in the Instrument Setup step for wells on labware. The sample ID is captured in the Well Identification field.

Refer to Section 26.3.5, <u>Span-8Transfer Log Contents (FX, NX-S8 only)</u>, for specific information about the Span8Transfer Log contents.

- FX default setting is On when a Span-8 Pod is installed and configured in Instrument Setup. However, a Span8Pipetting log is available only after a method has been run with a Span-8 pod.
- NX-S8 default setting is On. However, a Span8Pipetting log is available only after a method has been run with a Span-8 pod.

## 26.2.6 UnifiedPipetting Log

The UnifiedPipetting log lists aspirate and dispense operations, along with Sample IDs for wells, performed by any pod. It includes information on the labware where the aspirate or dispense operation occurs and is used when validating a method or generating a comprehensive report about pipetting operations used during a method.

Refer to Section 26.3.6, *UnifiedPipetting Log Contents*, for specific information about the UnifiedPipetting log contents.

**Note:** The default setting for the UnifiedPipetting log is Off.

## 26.2.7 UnifiedTransfer Log

The UnifiedTransfer log lists transfer operations, along with Sample IDs for wells, performed by any pod. It includes information on the source and destination labware and is used when validating a method or generating a comprehensive report about liquid transfer operations used during a method.

Refer to Section 26.3.7, <u>UnifiedTransfer Log Contents</u>, for specific information about the UnifiedTransfer log contents.

**Note:** The default setting for the UnifiedTransfer log is Off.

## 26.3 Log Contents

Each of the logs provide specific types of information; however, all log files provide the following information in their header:

- Method name and file path.
- User.
- Method start date and time.
- Unit serial number.
- Pod serial number.
- Last date the pod was validated (by a Beckman Coulter Service Engineer).
- A blank line (used as a separation between the header and the information captured during the method run).

Each line below the header can contain the following information:

- Date the information was captured.
- Time the information was captured.
- Information captured during the method run that is specific to the type of log requested.

**Note:** The information captured during the method run is based on the type of log generated. More specifically, when a **Pipetting Log** is generated, pipetting information such as the pod performing the operation, the type of pipetting operation performed, and the labware position and name are captured, along with other pipetting specific information.

The last line of a log file contains the date and time the method run ended.

## 26.3.1 Details Log Contents

Each line below the header of a Details log contains the following information:

- Date/Time when the error occurred.
- Action Completed action performed by the Biomek instrument.
- Device Completing the Action the device performing the action (if applicable).

## 26.3.2 Errors Log Contents

Each line below the header of an Errors log contains the following information:

- Date/Time when the error occurred.
- Error Description provides the exact error message.

### 26.3.3 Pipetting Log Contents

Each line below the header of a Pipetting log contains the following information:

- Date/Time when the error occurred.
- Pod Name which pod performed the operation (Pod1 or Pod2).
- Position the location on the deck (P1, P2...).
- Labware Name name assigned to the labware in Labware Properties.
- Labware Barcode barcode assigned to the labware in Labware Properties.
- Labware Section the quadrant accessed.

**Note:** Quadrants are numbered 1 through 4. A 96 well microplate is always Quadrant 1.

Amount — the amount of fluid pipetted.

**Note:** If no information was supplied for a particular field, the field is blank in the Pipetting log.

# 26.3.4 Span-8Pipetting Log Contents (FX, NX-S8 only)

Each line below the header of a Span-8Pipetting log contains the following information:

- Date/Time when the error occurred.
- Pod Name which pod performed the operation (Pod1 or Pod2).
- Operation the pipetting operation performed (aspirate or dispense).
- Position the location on the deck (P1, P2...).
- Labware Name name assigned to the labware in Labware Properties.
- Labware Barcode barcode assigned to the labware in Labware Properties.
- Well Number the well number pipetted to or from.

**Note:** Wells are numbered left to right, top to bottom for all types of labware (Figure 26-1).

- Well Identification sample ID assigned in the Instrument Setup step for wells on labware.
- Probe Number the probe used for the pipetting operation.

**Note:** Probes are numbered 1 through 8, from the back of the Biomek instrument to the front.

- Amount the amount of fluid pipetted.
- Technique the name of the technique used to govern the pipetting operation, or Custom if a customized technique was used.

**Note:** If no information was supplied for a particular field, the field is blank in the Span-8Pipetting log.



Figure 26-1. Well numbering in log files

# 26.3.5 Span-8Transfer Log Contents (FX, NX-S8 only)

Each line below the header of a Span-8Transfer log contains the following information:

- Date/Time when the error occurred.
- Pod Name which pod performed the operation (Pod1 or Pod2).
- Source Position location of the source labware for the pipetting operation.
- Source Labware Name name assigned to the source labware in Labware Properties.
- Source Well Number the number of the well pipetted from in the source labware.

**Note:** Wells are numbered left to right, top to bottom for all types of labware (Figure 26-1).

- Source Well Identification sample ID assigned in the Instrument Setup step for wells on labware.
- Destination Position location of the destination labware for the pipetting operation.
- Destination Labware Name name assigned to the destination labware in Labware Properties.

Destination Well Number — the number of the well pipetted to in the destination labware.

**Note:** Wells are numbered left to right, top to bottom for all types of labware (Figure 26-1).

- Destination Well Identification —sample ID assigned in the Instrument Setup step for wells on labware.
- Probe Number the probe used for the pipetting operation.

**Note:** Probes are numbered 1 through 8, from the back of the Biomek instrument to the front.

• Amount — the amount of fluid pipetted.

•

• Technique — the name of the technique used to govern the pipetting operation, or **Custom** if a customized technique was used.

**Note:** If no information was supplied for a particular field, the field is blank in the Span-8Transfer log.

## 26.3.6 UnifiedPipetting Log Contents

Each line below the header of a UnifiedPipetting log (Figure 26-2) contains the following information:

- Date/Time when the pipetting operation occurred.
- Pod Name which pod performed the pipetting operation (Pod1 or Pod2).
- Type of Pipetting Operation aspirate or dispense.
- Source Position location of the source labware for the pipetting operation.
- Labware Name name assigned to the labware in Labware Properties.
- Labware Bar Code bar code assigned to the labware in Labware Properties.
- Labware Well Number— number of the well pipetted from in the labware.

**Note:** Wells are numbered left to right, top to bottom for all types of labware (Figure 26-1).

- Labware Sample ID for Well sample ID assigned in Instrument Setup step for well for labware.
- Probe or Tip Number probe or tip used for the pipetting operation.
  - FX probes for the Span-8 Pod are numbered 1 through 8, from the back the instrument to the front. Tips for the Multichannel Pod are numbered left to right, top to bottom (like well numbering) for all types of labware.
  - 3000 Tips for the Multi-Tip Pipette Tools are numbered 1 through 8, from the back of the Biomek instrument to the front.
  - NX-MC tips for the Multichannel Pod are numbered left to right, top to bottom (like well numbering) for all types of labware.
  - NX-S8 probes for the Span-8 Pod are numbered 1 through 8, from the back the instrument to the front.

- Volume amount of fluid pipetted.
- Technique the name of the technique used to govern the pipetting operation, or **Custom** if a customized technique was used.

**Note:** If no information was supplied for a particular field, the field is blank in the UnifiedPipetting log.

🗐 UnifiedPipetting09-11-2002 15.24.11.log - Notepad	
<u>File E</u> dit <u>S</u> earch <u>H</u> elp	
Method = C:\Program Files\Biomek FX\Methods\Method48.bmt	
Logged in user = BAROGERS	
Started 09/11/2002 15:24:11	
Unit serial number =	
Pod1 head serial number = None	
No validation date.	
Pod2 head serial number = None	
No validation date.	
80/11/2882 15-25-48 Pod1 Achiesto PE Platon 18181 1 c1 1 28 Low-Holumo	
80/11/2002 13.23.40,7001,85µ1400,73,71400,131,51,150,1920,100-001000 80/11/2002 15.25.40,8041 (crivate DE Plated 10101,151,152,20,20,100-001000	
07/17/2002 15.25.40,1001,n5p110(2,15,110(0)),2,52,25,25,20,100 V01000 100/11/2002 15.25.40 Pod1 0cnivate D5 Plated 10101,2,52,12,20,100 V01000	
07/17/2002 15.25.40,1001,152,114(5,15,114(5,15,14)),3,3,20,20,20,20,20,20,20,20,20,20,20,20,20,	
60/11/2002 15:25:40 Pod1 denirate PS Platea 10101,5 5 20 Low-Unline	
19/11/2002 15:25:40, Pod1. Aspirate. P5. PlateA. 10101.6.,6.20.1 ov-Unlume	
09/11/2002 15:25:40, Pod1.Aspirate.P5.PlateA.10101.77.20.Low-Volume	
09/11/2002 15:25:40.Pod1.Aspirate.P5.PlateA.10101.8.8.20.Low-Volume	
09/11/2002 15:25:40,Pod1.Aspirate.P5,PlateA.10101.99.20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,10,,10,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,11,,11,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,12,,12,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,13,,13,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,14,s3,14,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,15,s4,15,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,16,,16,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,17,,17,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,18,,18,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,19,,19,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,20,,20,Low-Volume	
09/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,21,,21,20,Low-Volume	
109/11/2002 15:25:40,Pod1,Aspirate,P5,FlateA,10101,22,,22,20,Low-Volume	
109/11/2002 15:25:40,Pod1,Aspirate,P5,PlateA,10101,23,,23,20,Low-Volume	
109/11/2002 15:25:40,7001,H5p1rate,F5,F1ateH,10101,24,,24,20,L0W-U01ume	
109/11/2002 15:25:40,7001,HSp1/ate,F5,F1ateH,10101,25,,25,20,L00-V01UMe	
107/11/2002 15:25:40,7001,45p1/ate,75,71ate4,10701,20,20,20,100-001000	
107/11/2002 10.20.40,F001,H5µ1FdLE,F5,F1dLEH,10101,27,50,27,20,L0W-V010ME	الكر

Figure 26-2. UnifiedPipetting log
### 26.3.7 UnifiedTransfer Log Contents

Each line below the head of a UnifiedTransfer log (Figure 26-3) contains the following information:

- Date/Time when the transfer occurred.
- Pod Name which pod performed the transfer (Pod1 or Pod2).
- Source Position location of the source labware for the transfer operation.
- Source Labware Name name assigned to the source labware in Labware Properties.
- Source Labware Bar Code bar code assigned to the source labware in Labware Properties.
- Source Labware Well number of the well pipetted from in the source labware.

**Note:** Wells are numbered left to right, top to bottom for all types of labware (Figure 26-1).

- Source Labware Sample ID for Well sample ID assigned in Instrument Setup step for well for source labware.
- Destination Position location of the destination labware for the transfer operation.
- Destination Labware Name name assigned to the destination labware in Labware Properties.
- Destination Labware Bar Code bar code assigned to the destination labware in Labware Properties.
- Destination Labware Well the number of the well transferred to in the destination labware.
- Destination Labware Sample ID for Well sample ID assigned in Instrument Setup step for well for destination labware.

•

Probe or tip number — probe or tip used for the transfer operation.

- FX probes for the Span-8 Pod are numbered 1 through 8, from the back the instrument to the front. Tips for the Multichannel Pod are numbered left to right, top to bottom (like well numbering) for all types of labware.
- 3000 Tips for the Multi-Tip Pipette Tools are numbered 1 through 8, from the back of the Biomek instrument to the front.
- NX-MC tips for the Multichannel Pod are numbered left to right, top to bottom (like well numbering) for all types of labware.
- NX-S8 probes for the Span-8 Pod are numbered 1 through 8, from the back the instrument to the front.
- Volume amount of fluid transferred.

**Note:** If no information was supplied for a particular field, the field is blank in the UnifiedTransfer log.

🖺 UnifiedTransfer09-11-2002 15.24.11.log - Notepad	_ 🗆 ×
<u>File Edit S</u> earch <u>H</u> elp	
Method = C:\Program Files\Biomek FX\Methods\Method48.bmt	<b></b>
Logged in user = BAROGERS	
Started 09/11/2002 15:24:11	
Unit serial number =	
Podl head serial number = None	
NO VAILOALION GALE.	
ruuz neau seriai number - Nune No usidation dato	
No variation date.	
09/11/2002 15:25:41.Pod1.P5.PlateA.10101.1.s1.P9.PlateB.12344.1.s1.1.20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,2,s2,P9,PlateB,12344,2,s2,2,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,3,,P9,PlateB,12344,3,,3,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,4,,P9,PlateB,12344,4,,4,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,5,,P9,PlateB,12344,5,,5,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,6,,P9,PlateB,12344,6,,6,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,7,,P9,PlateB,12344,7,,7,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,8,P9,PlateB,12344,8,,8,20	
09/11/2002 15:25:41,7001,75,7LaTEH,10101,9,,79,7LaTEB,12344,9,,9,20	
07/11/2002 15:25:41;FUU1;F5;F1dLEH;10101;10;;F3;F1dLEB;12344;10;;10;20 60/11/2002 15:25:41;FUU1;F5;F1dLEH;10101;10;F3;F1dLEB;12344;10;;10;20	
07/11/2002 15.25.41,7001,75,714001,11,77,71400,77,71400,12344,11,,11,20	
69/11/2002 15:25:41, Pod1, P5, Plated, 10101, 13, P9, PlateB, 12344, 13, 13, 20	
09/11/2002 15:25:41.Pod1.P5.PlateA.10101.14.53.P9.PlateB.12344.14.53.14.20	
09/11/2002 15:25:41, Pod1, P5, PlateA, 10101, 15, 54, P9, PlateB, 12344, 15, 54, 15, 20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,16,P9,PlateB,12344,16,,16,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,17,,P9,PlateB,12344,17,,17,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,18,,P9,PlateB,12344,18,,18,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,19,,P9,PlateB,12344,19,,19,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,20,,P9,PlateB,12344,20,,20,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,21,,P9,PlateB,12344,21,,21,20	
09/11/2002 15:25:41,Pod1,P5,PlateA,10101,22,P9,PlateB,12344,22,,22,20	
09/11/2002 15:25:41,7001,75,7LaTEH,10101,23,79,7LaTEB,12344,23,,23,20	
00/11/2002 15:25:41,7001,75,71dt0H,10101,24,77,71dt0B,12344,24,,24,20 00/14/2009 15:25:41,7001,75,71dt0H,10101,24,77,71dt0B,12344,24,,24,20	
07/11/2002 12.22.41,7001,72,7141241,10101,22,773,714126,12344,22,,22,20 00/11/2002 12.25.41 Pod1 PC Plated 10101 26 PQ PlateR 1924/ 26 26 20	
00/11/2002 12-25-31,001,12,114ten,10101,20,117,114ten,12044,20,2044,20,20	-

Figure 26-3. UnifiedTransfer log

# 26.4 Selecting Log Types

To generate log(s) during a method run, the only information required is a selection of the type of log(s) desired.

To select log(s):

1. Choose **Options>Log Configuration**. Log Configuration appears (Figure 26-4).

Log Configuration	
Logs:	
Details  Frrors Pipetting Span8Pipetting Span8Transfer UnifiedPipetting UnifiedTransfer	
ОК	Cancel

Figure 26-4. Select the log type(s) generated here

**Note:** The default setting for Errors is On. The default settings for Details, Pipetting, UnifiedPipetting, and UnifiedTransfer logs are Off.

- FX Span8Pipetting and Span8Transfer logs only appear when a Span-8 Pod is installed and configured for use and after a method has been run with a Span-8 pod. When these conditions are met, the default settings for the Span8Pipetting and Span8Transfer logs are On.
- 3000, NX-MC Span8Pipetting and Span8Transfer logs are not available for these instruments.
- NX-S8 Span8Pipetting and Span8Transfer logs only appear after a method has been run. The default settings for the Span8Pipetting and Span8Transfer logs are On.
- 2. Check the logs desired.

# 26.5 Viewing Log Files

View log files when creating reports or obtaining detailed information about a method. Log files are viewed in a standard text processor, such as Notepad (Figure 26-5), or in a spreadsheet, such as Microsoft Excel (Figure 26-6).

📱 Span8Pipetting03-05-2001 16.51.18.log - Notepad 📃 🗖 🗖
<u>File Edit S</u> earch <u>H</u> elp
Method = C:\Program Files\Biomek FX\Methods\Span-8 Tutorial 3.bmt 🔺
Logged in user = Lawrenme
Started 03/05/2001 16:51:18
Unit serial number =
Pod1 head serial number = None
No validation date.
03/05/2001 10:51:24,P001,HSp1rate,P12,S0Urce,,50,,1,200,Gustom
03/05/2001 10:51:24,P001,HSp1rate,P12,S0Urce,,02,,2,200,Gustom
03/05/2001 10:51:24,000,HSp1rate,P12,S00rce,,/4,,3,200,USC0M
03/05/2001 10:51:24,7001,HSp1rdL0,712,300r00,80,,4,200,60500M
03/03/2001 10.51.25,F001,DISPENSE,F13,DESCI,,10,,1,200,Stalluaru
02/02/2001 10.21.22,F001,D15pense,F13,D05(1,,40,,2,200,Standard 02/02/2001 14.551.25 Dodt Dispanse D19 Doct1 (A) 9 200 Standard
03/05/2001 10.51.25,r001,Dispense,r13,Dest1,04,,3,200,Standard
63/65/2001 16-51-26 Pod1 Acritate P12 Source 2 1 100 Recerunir
13/05/2001 16-51-26 Pod1 Assirate P12 Source 26 2 150 Reservoir
13/15/2001 16:51:26 Pod1 Asirate P12 Source 50 3.200 Reservoir
03/05/2001 16:51:26.Pod1.Aspirate.P12.Source.74.4.250.Reservoir
03/05/2001 16:51:26.Pod1.Dispense.P13.Dest1311.100.Standard
03/05/2001 16:51:26,Pod1,Dispense,P13,Dest1,,43,,2,150.Standard
03/05/2001 16:51:26,Pod1,Dispense,P13,Dest1,,55,,3,200,Standard
03/05/2001 16:51:26,Pod1,Dispense,P13,Dest1,,67,,4,250,Standard
03/05/2001 16:51:44,Run ended.

Figure 26-5. Span8Pipetting Log viewed in Notepad

🗿 S	pan8Transfer(	)3-05-2	001 1	6.51.18.log	]									_ 🗆 X
	A	В	С	D	Е	F	G	Н	I	J	K	L	M	N
1	1 Method = C:\Program Files\Biomek FX\Methods\Span-8 Tutorial 3.bmt													
2	Logged in use	er = Lav	wrenm	ne										
3	Started 03/05	/2001 1	16:51:	18										
4	Unit serial nu	mber =												
5	Pod1 head se	erial nu	mber :	= None										
6	No validation	date.												
7														
31	3/5/01 16:51	Pod1	P12	Source		1		P13	Dest1		1		1	200
9	3/5/01 16:51	Pod1	P12	Source		13		P13	Dest1		25		2	200
10	3/5/01 16:51	Pod1	P12	Source		25		P13	Dest1		49		3	200
11	3/5/01 16:51	Pod1	P12	Source		37		P13	Dest1		73		4	200
12	3/5/01 16:51	Pod1	P12	Source		49		P13	Dest1		14		1	200
13	3/5/01 16:51	Pod1	P12	Source		61		P13	Dest1		38		2	200
14	3/5/01 16:51	Pod1	P12	Source		73		P13	Dest1		62		3	200
15	3/5/01 16:51	Pod1	P12	Source		85		P13	Dest1		86		4	200
16	3/5/01 16:51	Run ei	nded.											
	► ► Span8	Transfe	r03-0	5-2001 16.	.51.1 /				•					

Figure 26-6. Span8Transfer Log viewed in Microsoft Excel



## 27.1 Overview

Manual Control and Advanced Manual Control are used to control:

- Movement of the bridge, head, and gripper independently of a method.
- Pod when framing the deck or gripper and recovering from errors.
- Device Controllers and ALPs on a Biomek FX or Biomek NX manually.
- Devices on a Biomek 3000 manually.

For information on manually controlling Biomek instruments, ALPs, or other devices through Manual Control, refer to the appropriate hardware user's manual:

- FX <u>Biomek® FX Laboratory Automation Workstation User's Manual</u>
- Solution User's Manual <u>Biomek® 3000 Laboratory Automation Workstation User's Manual</u>
- NX-MC <u>Biomek® NX Multichannel Laboratory Automation Workstation</u> <u>User's Manual</u>
- NX-S8 <u>Biomek® NX Span-8 Laboratory Automation Workstation User's</u> <u>Manual</u>

Refer to the <u>ALPs User's Manual</u> or the appropriate integration manual for information on using Manual Control with specific ALPs or devices.

# 27.2 Accessing Manual Control

To open Manual Control, choose **Instrument>Manual Control**. An Information dialog (Figure 27-1) appears briefly as the connection is made with the Biomek instrument, immediately followed by Manual Control (Figure 27-2).

**Note:** Manual Control is available only when a method is not being executed. If a need for manual control is realized during a method run, stop the method using the Stop button or Snap Continuation button (refer to Chapter 25.11.2, <u>Snapping a</u> <u>Continuation</u>) on the toolbar before accessing Manual Control.

> **3000** — Snapping a Continuation is not available.

Informatio	on 🗵
٩	Connecting
	Cancel

Figure 27-1. Confirms Manual Control is connecting

Manual Control				
≶ <b>-</b> <u>A</u> dvanced Controls	<b>ff</b> Home All Axes	## Get <u>V</u> ersion	Stop	<b>E</b> žit
Click on a <u>p</u> o	sition to move Po	od1 💌 to it.		
TL1	P4 P	8 P12	)	216
P1	P5 P	9 P13		17
P2	P6 P	10 P14		218
P3	P7 P	11 P15	J F	219

Figure 27-2. Manual Control

# 27.3 Understanding the Options in Manual Control

Table 27-1 lists and describes the options in Manual Control:

Table 27-1. Manual Control Options

Option	Description
Advanced Controls	Accesses Advanced Manual Control which allows a pod, ALP, or device to be selected. Refer to the specific hardware manual for details on using Advanced Manual Control for a device.
Home All Axes	Gives the instrument a point of reference from which to make subsequent moves. For a single- pod system, home position is left, back. For a dual- pod system, home position for the first (left) pod is left, back and for the second (right) pod is right, back.
	<b>Note:</b> Pods should be homed each time the instrument is powered on. Depending on the type of pods on the system, a Warning appears. After confirming that the actions have been addressed properly, choose <b>OK</b> .
Get Version	Shows the current firmware version for installed devices, pods, and devices (Figure 27-3).
Stop	Stops a pod once movement has started after selecting the desired pod from Click on a position to move.
Click on a position to move	Moves the pod to a specific deck position after selecting the desired pod. Manual Control moves the pod to the top of the Z axis, then centers it over the selected position.
Exit	Closes Manual Control.



Figure 27-3. Information displaying firmware version



## 28.1 Overview

The Biomek scripting feature is based on Visual Basic Scripting, a subset of the Microsoft® Visual Basic programming language. Any Visual Basic Scripting code can be interpreted by Biomek Software and may be included in a method.

**Note:** For more Visual Basic Scripting information, go to http://msdn.microsoft.com/scripting.

The Advanced Step Palette provides a Script step and a Scripted Let step, which provide a configuration area in which code can be entered. These steps invoke a Visual Basic Scripting Engine that evaluates the code.

The Script step and Scripted Let step expose objects that are used internally by Biomek Software. This is accomplished through the use of Microsoft's Component Object Model (COM).

Data from a reader or LIMS file in a .csv file format may be translated into a data set using scripting. The data sets created from scripting behave like any other data set created in the **Create Data Set** step, from sample IDs in the **Instrument Setup** step, or in the **SILAS** step.

The sections in this chapter include:

•

- refer to Section 28.2, Using the Script Steps
- refer to Section 28.3, *Scripting Data Sets*

## 28.2 Using the Script Steps

A Script step, available from the Advanced Step Palette, provides a configuration area in which a description of the script step may be entered, and lines of code may be written.

**Note:** Refer to Section 17.8, <u>Script Step</u>, for information on configuring a Script step.

A Scripted Let step is also available from the Advanced Step Palette and provides a configuration area in which a description of the script step may be entered, and lines of code may be written. However, two additional commands may be entered in the Scripted Let step that allow for variables created in the script to be extended outside of the script and used in configuring other steps in the method. These two new commands are:

- Extend sets the variable name to the specified value for all steps contained within the Scripted Let. The proper syntax for Extend is Extend "Variable Name", Value.
- WeakExtend sets the variable name to the specified value for all steps contained within the Scripted Let step if the specified variable does not already have a value. This is similar to a Let variable with Overridable selected. The proper syntax for WeakExtend is WeakExtend "Variable Name", Value.

**Note:** Refer to Section 17.9, <u>Scripted Let Step</u>, for information on configuring a Scripted Let step.

## 28.3 Scripting Data Sets

**Note:** The information in this section requires an advanced skill level; it is not intended to offer instructions on learning VBScript. Scripting data sets is not supported by Beckman Coulter Service Engineers; however, information is available at http://msdn.microsoft.com/library/default.asp and in <u>VBScript in a Nutshell: A</u> <u>Desktop Quick Reference</u>, 2nd Edition, Matt Childs, Paul Lomax, and Ron Petrusha 2003, O'Reilly & Associates, Inc.

Data from a reader or LIMS file in a .csv file format may be translated into a data set using scripting. The data sets created from scripting behave like any other data set created in the Create Data Set step, from sample IDs in the Instrument Setup step, or in the SILAS step.

The procedure (refer to Section 28.3.1, *Procedure to Script Data Sets*) to script data sets includes:

- Creating a new data set to hold imported data.
- Opening a .csv file.
- Translating data from a .csv file to a new data set called Sample ID.
- Placing the translated data (the new data set called Sample ID) into the new data set to hold imported data.

### 28.3.1 Procedure to Script Data Sets

Execute the following script code:

```
' Create a new Data Set to hold our imported data.
Set DestinationDataSet = Labware("P5").DataSets.Add("SampleID")
' Create the Windows file system object. This object will allow us
' to find our data file on the disk.
Set FileSystemObject = CreateObject("Scripting.FileSystemObject")
' Check to see if the file is on the disk.
If Not FileSystemObject.FileExists("c:\demodata.csv") Then
  ' Call the system's error creator. This will cause a sensible
  ' error message to be shown to the person running the method.
  ' The first parameter is an error code number. 500 was chosen
  ' because it is not already in use by Windows.
  ' The second parameter is the source of the error.
  ' The third parameter is the description of the error.
 Err.Raise 500, "My Import Script", "File does not exist."
End If
' Open the data file.
Set FileStream = FileSystemObject.OpenTextFile("c:\demodata.csv")
' Loop over the rows in the file.
For i = 0 To 7
  ' Read a line out of the file.
 RowLine = File.Stream.ReadLine
  ' Split the values in the line into an array of values, using
  ' the comma as a delimiter.
  ' Note that the array contains values 0, 1, ..., 11.
 RowDataArray = Split(Rowline, ",")
  ' Loop over the columns in the row.
 For j = 0 To 11
    ' Put the data from the array into our new Data Set.
    ' The expression "(i*12) + (j+1)" converts from the 0-based
    ' loop variables to well identifiers.
   DestinationDataSet((i*12) + (j+1)) = RowDataArray(j)
 Next ' end the column loop
Next ' end the row loop
' Close the file when we're done.
FileStream.Close
```



## 29.1 Overview

The Biomek main editor (Figure 29-1) contains the basic tools necessary to create and run methods; however, there may be times when additional tools are needed. To address this situation, steps and step palettes may be added to the main editor as necessary. These additional steps and step palettes provide more functionality when creating and running methods. By using **Preferences** (Figure 29-12) and **Step Palette Builder** (Figure 29-2), the look of the main editor can be changed to fit the needs of the laboratory.

**Preferences** is used to alter the Method View and change the size of the steps (or step buttons) in a step palette. For example, graph lines and + and - buttons may be added to the Method View (Figure 29-1), and toolbar and step palette positions may be saved automatically when the appropriate settings are checked in **Preferences**.

The Step Palette Builder allows the addition, removal, and manipulation of steps on the step palettes. Steps may be added to any step palette or combined into a new step palette. Individual devices, such as an incubator, operating with a SILAS module must be added to a step palette after the SILAS module has been installed so the device can be used in a method and controlled by Biomek Software.

**Note:** The Current Deck Display can be manipulated like a toolbar, except it can be docked only below the user interface area.

Biomek® Software - Metho File Edit Project Instrument	od7* [New] Execution Options Help		_ 🗆 🗙
	6 <b>12   い ○ 0</b>   ▶ □ □   火		
Aspirate	Start Start Loop from 1 to 1 step 1	Yariable Start 1 End 1	
Dispense Transfer	Finish	Increment ]1	
Mix Combine Pre- Mix Combine Pre- Met Met Met Met Met Met Met Met Met Met	Method View eferences allows the ethod View to be altered. To options selected in eferences are displayed this Method View: splay graph lines tween steps of a ethod in the Method ew d splay + and - buttons in e Method View when panded or collapsed	X TL1 P4 P8 P12 P16	
Move Pod Group	bsteps are present in a ethod.	P1         P5         P9         P13         P17           P2         P6         P10         P14         P18           P3         P7         P11         P15         P19	
Method7* BiomekFX (4) Biomek	kFX ETC: 0:00:10		

Figure 29-1. Main editor for a Biomek FX

Step Palette Builder Palettes Steps	×
Advanced Advanced Basic Devices J Devices J Intermediate J Span-8 J StackerCarousel	<u>N</u> ew <u>P</u> roperties <u>D</u> elete
<b>Step Palette Builder</b> Allows the addition, removal, an manipulation of steps on the step palettes. A custom step palette m created to display frequently used on a single palette.	nd p nay be d steps
OK	Cancel

Figure 29-2. Step Palette Builder

**Note:** The available step palettes in the **Step Palette Builder** are based on the instrument.

3000 — the Span-8 Step Palette is available because the Serial Dilution, Transfer from File, and Define Pattern steps may be used with the instrument.

# 29.2 Changing Display Preferences

Preferences allows the main editor appearance to be customized. Preferences allows the customization of the main editor using options organized under General, View, and Errors.

To customize the appearance of the main editor:

1. Choose **Options>Preferences...** Preferences appears (Figure 29-3).

Preferences		4
Preferences	General	
General View	☑ Validate the current method before running it.	
Errors	🔲 Ask for confirmation before removing a step from a method.	
	<ul> <li>Automatically save all Editor settings listed below.</li> <li>Save changes to the screen position of the Editor.</li> <li>Save changes to the toolbar and step palette layout of the Editor.</li> <li>Save changes to the steps configured within step palettes of the Editor.</li> <li>Look ahead up to 1800 seconds in the method while it is running.</li> </ul>	
	OK Cancel <u>R</u> eset	

Figure 29-3. Preferences

2. Choose **General** to configure options concerning validating and saving methods (refer to Section 29.2.1, *Configuring General Options*).

OR

Choose **View** to configure options concerning the appearance of the Method View (refer to Section 29.2.2, *<u>Configuring View Options</u>*).

OR

Choose **Errors** to configure options concerning the notifications of errors (refer to Section 29.2.3, *Configuring Errors Options*).

## 29.2.1 Configuring General Options

General options concern validating and saving methods.

To configure General options:

- 1. From Preferences, highlight General (Figure 29-3).
- 2. Check the desired options using Table 29-1.
- 3. Choose **OK** to save the checked options.

OR

Choose **Cancel** to cancel the checked options.

OR

Choose **Reset** to reset all customizations, including custom labware, toolbar/palette organization, options chosen in **Preferences**, and position of the main editor.

Option	Description
Validate the current method before running it.	Simulates methods internally to test for errors before the method is run. If no errors are detected, the method is executed. If an error is detected, the process stops and an error message displays information about the error.
Ask for confirmation before removing a step from a method.	Displays a confirmation prompt when deleting a step from a method (Figure 29- 4).
Automatically save all Editor settings listed below.	Automatically saves the following three settings. (Enabled by default.)
Save changes to the screen position of the Editor.	When checked, automatically saves the size and position of the editor on the screen. (Available when Automatically save all Editor settings listed below is not selected.)
Save changes to the toolbar and step palette layout of the Editor.	When checked, automatically saves toolbar and step palette size, arrangement, and location. (Available when Automatically save all Editor settings listed below is not selected.)

### Table 29-1. General options

Option	Description
Save changes to the steps configured within step palettes of the Editor.	When checked, automatically saves steps on step palette (modified by Palette Builder).(Available when Automatically save all Editor settings listed below is not selected.)
Look ahead up to [1800] seconds in the method while it is running.	To prevent the system from bogging down due to unnecessary memory consumption, the software translate the steps of the method into "to do" lists of actions. This option prevents the software from bogging down by designating a length of time for suspending the translation process (refer to Section 25.3.2.1, <u>Using Look ahead up to [1800]</u> <u>seconds in the method while it is running</u> ).

Table 29-1. General options (Continued)

**Note:** When Automatically save all Editor settings listed below is not selected, and when Save changes to the screen position of the Editor, Save changes to the toolbar and step palette layout of the Editor, and Save changes to the steps configured within step palettes of the Editor are also not selected, the main editor opens at default size with the last saved configuration for the editor, toolbars, and palettes.

Confirm	×
?	You are about to remove steps from your method. Do you wish to continue?
	<u>Yes</u> <u>N</u> o

Figure 29-4. Confirm to delete steps from a method

## 29.2.2 Configuring View Options

View options concern the appearance of the Method View.

To configure View options:

1. From Preferences, highlight View (Figure 29-5).

Preferences		×
Preferences	View	
<ul> <li>General</li> <li>View</li> <li>Errors</li> </ul>	<ul> <li>Use large icons in the Method View.</li> <li>Use large icons and a caption for steps within a step palette.</li> <li>Use large icons and a caption for toolbar buttons.</li> </ul>	
	<ul> <li>Display graph lines between steps of a method in the Method View.</li> <li>Display + and - buttons in the Method View when expanded or collapsed substeps are present in a method</li> </ul>	
	OK Cancel <u>R</u> eset	

Figure 29-5. Preferences - View

- 2. Check the desired options using Table 29-2.
- 3. Choose **OK** to save the checked options.

OR

Choose **Cancel** to cancel the checked options.

OR

Choose **Reset** to reset all customizations, including custom labware, toolbar/palette organization, options chosen in **Preferences**, and position of the main editor.

Table 29-2. View Options

Option	Description
Use large icons in the Method View.	Displays text and icons in the Method View in a larger size. (Enabled by default.)
Use large icons and a caption for steps within a step palette.	Displays large icons with labels (descriptive text) under the steps on the step palettes. (Enabled by default.)
Use large icons and a caption for toolbar buttons.	Displays large toolbars with labels (descriptive text) when checked.
Display graph lines between steps of a method in the Method View.	Displays lines which connect steps in the Method View when checked.
Display + and - buttons in the Method View when expanded or collapsed substeps are present in a method.	Displays + and - in front of steps such as Loop which contain nested steps. Click the + or - to expand or collapse the main step.

## 29.2.3 Configuring Errors Options

**Error** options concern the notifications of errors. An option to play a .wav file when an error occurs is available while another option allows a program, such as an .exe file, to be run when an error occurs.

To configure Errors options:

1. From Preferences, highlight Errors (Figure 29-6).

Preferences	× ×
Preferences	Errors
General View Frrors	<ul> <li>Play a sound on errors during runs.</li> <li>Play this sound: <ul> <li>Play this sound:</li> <li>Play the sound three times</li> </ul> </li> <li>Launch a program on errors during runs. <ul> <li>Launch this program:</li> <li>Browse</li> </ul> </li> <li>Browse</li> <li>Send these parameters: <ul> <li>Click here for more information on parameters.</li> <li>Start in this directory:</li> <li>Browse</li> </ul> </li> <li>If a window appears, start in this state: Hide</li> </ul>
	OK Cancel <u>R</u> eset

Figure 29-6. Preferences - Errors

- 2. Choose **Play a sound on errors during runs** to run a .wav file when an error message appears.
- 3. In Play this sound, use Browse to find the desired .wav file.
- 4. Choose the desired .wav file. The desired file appears in Play this sound.
- 5. From Play the sound, choose one of the following options from the drop-down menu to play the sound the desired number of times when an error message appears:
  - once
  - twice
  - three times
  - repeatedly until dismissed

- 6. From intervals, choose one of the following options from the drop-down menu to play the sound at the desired intervals when an error message appears:
  - 10 second
  - 1
  - 5
  - 30
  - 1 minute
  - 5 minute
- 7. Choose **Launch a program on errors during runs** to run an .exe file when an error message appears.
- 8. In Launch this program, use Browse to find the desired .exe file.
- 9. Choose the desired file. The desired file appears in Launch this program.
- 10. In Send these parameters, enter the desired parameters using the information displayed in Figure 29-7.

**Note:** Choose **click here** to display Parameter Information (Figure 29-7). Choose **OK** to close Parameter Information. Highlighting the Parameter and Value and choosing OK does not enter the desired parameter; the desired parameters must be manually entered in Send these parameters.

Parameter Information			
The following parameters ara available for use:			
Parameter	Value		
%Error% %Method% %Project% %Instrument%	The error message displayed in the error dialog. The name of the current method. The name of the current project. The full path of the current instrument file.		
1 Parise and te 70			

Figure 29-7. Parameter Information

- 11. In Start in this directory, use Browse to start browsing in the desired directory.
- 12. From If a window appears, start in this state, choose one of the following options from the drop-down menu to display the style of the program:
  - Don't care the message appears in the default style of the program.
  - Maximize the message appears in the maximized state of the program.
  - Minimize the message appears in the minimized state of the program.
- 13. Choose **OK** to save the checked options.

OR

Choose **Cancel** to cancel the checked options.

OR

Choose **Reset** to reset all customizations, including custom labware, toolbar organization, options chosen in **Preferences**, and position of the main editor.

## 29.3 Toggling Toolbars and Step Palettes

The Biomek main editor provides toolbars and step palettes to allow quick access to method-building tools. Descriptions of the options in **Toolbars** are shown in Table 29-3.

To toggle toolbars and step palettes on and off:

1. Choose **Options>Toolbars**. A submenu appears listing the toolbars and step palettes (Figure 29-8).



Figure 29-8. Options for step palettes and toolbars

**Note:** The available step palettes in the **Step Palette Builder** are based on the instrument.

3000 — the Span-8 Step Palette is available because the Serial Dilution, Transfer from File, and Define Pattern steps may be used with the instrument.

**Note:** Any custom step palettes created with the Step Palette Builder also appear in the submenu (refer to Section 29.5.4, <u>*Creating a New Step Palette*</u>).

#### OR

Right-click in a gray area around the toolbars or step palettes. A menu listing only the step palettes appears (Figure 29-9).

	Advanced
~	Basic
	Devices
	Intermediate
	Span-8
	StackerCarousel
	Palette Builder

Figure 29-9. Step palette menu for a Biomek FX

Note: The step palette menu is based on the instrument; for example.

**Note:** Any custom step palettes created with the Step Palette Builder also appear in the submenu (refer to Section 29.5.4, *Creating a New Step Palette*).

2. Choose the desired toolbar or step palette to toggle it on or off. The corresponding toolbar or step palette appears or is hidden from view.

**Note:** A check to the left of a toolbar or step palette in **Options>Toolbars** indicates that it is currently displayed.

**Note:** Use the **Step Palette Builder** as an alternative method for displaying and hiding step palettes (refer to Section 29.5.2, *Displaying and Hiding Step Palettes*).

Option	Description	
Advanced	Toggles on or off the Advanced Step Palette.	
Basic	Toggles on or off the Basic Step Palette (Figure 29-10).	
Biomek3000	Toggles on or off the Biomek 3000 Step Palette.	
Deck Display	Toggles on or off the Deck Display (Figure 29-10).	
Devices	Toggles on or off the Devices Step Palette.	
Execution	Toggles on or off the part of the toolbar that is associated with the running of methods (Figure 29-10).	
Files	Toggles on or off the part of the toolbar that is associated with the building of methods (Figure 29-10).	
Intermediate	Toggles on or off the Intermediate Step Palette.	
Method Error	Toggles on or off the description of errors (Figure 29-10).	
Span-8	Toggles on or off the Span-8 Step Palette.	
Chasielty	3000 — the Span-8 Step Palette is available because the Serial Dilution, Transfer from File, and Define Pattern steps may be used with the instrument.	
Specialty	Toggles on or off the Specialty Step Palette.	
StackerCarousel	Toggles on or off the Stacker Carousel Step Palette.	





# 29.4 Customizing Step Palettes

Customize a step palette to add custom steps, change the view of the step palettes, or modify tool tips and labels of any step.

### 29.4.1 Displaying Large Step Palette and Toolbar Views

Labels for toolbars and step palettes can be displayed using **Preferences**. The Large Step Palette View and the Large Toolbar View display labels for the steps and tools (Figure 29-11).



Figure 29-11. Step palette views with and without labels

To toggle the step palette view:

1. Choose **Options>Preferences>View**. Figure 29-12 appears.

Preferences		×
Preferences Preferences General View Errors	View         ✓ Use large icons in the Method View.         ✓ Use large icons and a caption for steps within a step palette.         Use large icons and a caption for toolbar buttons.         □ Display graph lines between steps of a method in the Method View.         □ Display + and - buttons in the Method View when expanded or collapsed substeps are present in a method	×
1	OK Cancel <u>R</u> eset	

Figure 29-12. Preferences — main editor setting options

2. Choose **Use large icons and a caption for steps within a step palette** to toggle the step labels (Figure 29-12).

**Note:** The default for Use large icons and a caption for steps within a step palette is On.

3. Choose **Use large icons and a caption for toolbar buttons** to toggle the toolbar labels.

**Note:** The default for Use large icons and a caption for toolbar buttons is Off.

4. Choose OK.

### 29.4.2 Changing Step Properties

Change the properties associated with each step to make the step purpose easier to identify, or to distinguish custom, configured steps from standard steps. Each step has three properties:

- Tool Tip brief tip on the steps usage that displays when the mouse pointer hovers over the step.
- Label text describing operations of a step.
- Bitmap graphic displayed on the step button.

To change step properties:

- 1. On the step palette, right-click on the desired step.
- 2. Choose **Properties**. Step Properties appears (Figure 29-13).

Step Prope	erties		×
<u>T</u> ool Tip	Aspirates and Dispenses	Browse	<u>0</u> K
Label	Transfer	L	<u>C</u> ancel
<u>B</u> itmap	OStepUI.ocx,TRANSFER		

Figure 29-13. Step Properties

- 3. In Tool Tip, enter a new **Tool Tip**.
- 4. In Label, enter a new Label.
- 5. In Bitmap, choose a different graphic for the step button by clicking on **Browse**, and selecting the desired bitmap.
- 6. Choose **Open** to accept the new bitmap and return to **Step Properties**.
- 7. Choose **OK** to save changes to the **Step Properties** and return to the main editor.

### 29.4.3 Modifying a Step Caption in Method View

A step caption may be modified in the Method View. This allows differentiation of the steps by their captions.

To change a step caption in the Method View:

- 1. Highlight the desired step in the Method View.
- 2. Right-click on the step to open a menu.
- 3. Select Change Caption. Edit Step Caption appears (Figure 29-14).

Edit Step Caption	×
Change Instrument Setup to:	
Instrument Setup	
OK Cancel	

Figure 29-14. Change the step caption in Edit Step Caption

- 4. Enter the new step caption for the step.
- 5. Choose **OK**. The new caption appears at the appropriate spot in the Method View.



Figure 29-15. Modified step caption in Method View

# 29.5 Using the Step Palette Builder

The Step Palette Builder is used to create, delete, and modify step palettes. If a step is no longer needed on a step palette, it can be removed easily with the Step Palette Builder.

Use the Step Palette Builder to:

- Display a step palette.
- Hide a step palette.
- Add steps.
- Remove steps.
- Rearrange steps.
- Create a step palette.
- Modify step palette properties.
- Delete a step palette.

### 29.5.1 Accessing the Step Palette Builder

To access the Step Palette Builder:

Choose **Options>Palette Builder...** Step Palette Builder appears (Figure 29-16).

OR

- 1. Right-click in a gray area around the toolbars or step palettes. A menu listing available step palettes appears (Figure 29-9).
- 2. Choose Palette Builder....Step Palette Builder appears (Figure 29-16).

Step Palette Builder Palettes Steps	×
Image: Advanced         Image: Basic         Image: Devices         Image:	New Properties Delete
 	Cancel

Figure 29-16. Step Palette Builder

Note: The Step Palette Builder is based on the instrument.

3000 — the Span-8 Step Palette is available because the Serial Dilution, Transfer from File, and Define Pattern steps may be used with the instrument.

### 29.5.2 Displaying and Hiding Step Palettes

Display and hide step palettes using the Step Palette Builder.

To display or hide a step palette:

 Choose Options>Palette Builder....Step Palette Builder appears (Figure 29-16).

OR

Right-click in a gray area around the toolbars or step palettes. A menu listing available step palettes appears (Figure 29-9). Choose **Palette Builder...**.Step Palette Builder appears (Figure 29-16).

- 2. In the Step Palette Builder, click on the check box to the left of the desired palette to toggle the display on and off.
- 3. Choose **OK**. The selected step palettes are displayed or hidden from the Biomek main editor.

**Note:** As an alternative method for toggling a step palette on or off, use the Options>Toolbars menu (refer to Section 29.3, *Toggling Toolbars and Step Palettes*). This alternative method is available only when Palette Builder is not open.

### 29.5.3 Customizing Step Palettes

There are two main sources for steps added to step palettes:

- With Step Palette Builder not open, steps can be moved from the Method View onto a step palette.
- With Step Palette Builder open, steps can be moved from the Step Palette Builder to a step palette or moved between palettes.

**Note:** In order to use a SILAS module, such as the bar code reader, it must be placed on a step palette.

#### 29.5.3.1 Adding Non-Configured Steps to a Step Palette

To add steps to a step palette:

1. Choose **Options>Palette Builder...**.Step Palette Builder appears (Figure 29-16).

OR

Right-click in a gray area around the toolbars or step palettes. A menu listing available step palettes appears (Figure 29-9). Choose **Palette Builder...**.Step **Palette Builder** appears (Figure 29-16).

2. Choose the Steps tab (Figure 29-17).



Figure 29-17. Step Palette — Steps tab

3. In **Categories**, choose the desired category of steps. Steps from the selected category appear on the right side of the **Step Palette Builder**.

**Note:** Multiple selections can be highlighted at once in the Steps tab by holding down Ctrl or Shift.

4. Drag the desired step from **Steps** to a step palette on the main editor. The step is added to the step palette.

OR

Drag the desired step from one palette to another on the main editor.

5. From the Step Palette Builder, choose OK.

#### 29.5.3.2 Adding Configured Steps to a Step Palette

If a step configuration used in a method is to be used several times in various methods, add it to a step palette to save time during the method building process. By adding a configured step, any subsequent uses of the same configuration are added to a method easily.

To add configured steps to a step palette:

- 1. Configure the step in the Method View.
- 2. Highlight the step to add to the step palette.
- 3. Drag the step to the desired step palette. The addition of the step must be confirmed before it is added to the palette (Figure 29-18).



Figure 29-18. Confirmation for adding a step to a step palette

### 29.5.3.3 Rearranging Steps on a Step Palette

To rearrange steps on a step palette:

1. Choose **Options>Palette Builder...**Step Palette Builder appears (Figure 30-12).

OR

Right-click in a gray area around the toolbars or step palettes. A menu listing available step palettes appears (Figure 29-9). Choose **Palette Builder...**.Step Palette Builder appears (Figure 29-16).

- 2. Drag and drop the desired step on the step below the desired location on the palette. The step inserts itself above the step on which it was dropped.
- 3. From the Step Palette Builder, choose OK.

#### 29.5.3.4 Removing Steps From a Step Palette

Remove steps no longer needed to keep step palettes manageable.

To remove steps from a step palette:

1. Choose **Options>Palette Builder...**.Step Palette Builder appears (Figure 29-16).

OR

Right-click in a gray area around the toolbars or step palettes. A menu listing available step palettes appears (Figure 29-9). Choose **Palette Builder...** Step **Palette Builder** appears (Figure 29-16).

- 2. Drag and drop the desired step to a location where a step palette is not present.
- 3. From the Step Palette Builder, choose OK.

### 29.5.4 Creating a New Step Palette

Create custom step palettes to display frequently used or commonly used configured steps on a single palette.

To create a step palette:

1. Choose **Options>Palette Builder...**.Step Palette Builder appears (Figure 29-16).

OR

Right-click in a gray area around the toolbars or step palettes. A menu listing available step palettes appears (Figure 29-9). Choose **Palette Builder...**.Step **Palette Builder** appears (Figure 29-16).

2. Choose **New**. A new step palette appears in the upper left corner of the main editor (Figure 29-19).



Figure 29-19. New Step Palette

- 3. Change the new palettes properties (refer to Section 29.5.5, *Modifying Step Palette Properties*).
- 4. Add steps to the step palette (refer to Section 29.5.3.1, <u>Adding Non-Configured</u> <u>Steps to a Step Palette</u>).
- 5. From the Step Palette Builder, choose OK.

### 29.5.5 Modifying Step Palette Properties

The Step Palette Properties is the Name of the palette that appears in the Step Palette Builder and in Options>Toolbars.

To change a step palettes properties:

 Choose Options>Palette Builder....Step Palette Builder appears (Figure 29-16).

OR

Right-click in a gray area around the toolbars or step palettes. A menu listing available step palettes appears (Figure 29-9). Choose **Palette Builder...**.Step Palette Builder appears (Figure 29-16).

- 2. Select the desired palette.
- 3. Choose Properties.... Palette Properties appears (Figure 29-20).

Palette Properties		
<u>N</u> ame	Configured Steps	<u>0</u> K
		<u>C</u> ancel

Figure 29-20. Palette Properties

4. Type a descriptive name for the step palette.

**Note:** All alphanumeric and non-alphanumeric characters are allowed in palette names.

- 5. From Palette Properties, choose **OK** to save the name.
- 6. From Step Palette Builder, choose OK.

### 29.5.6 Deleting a Step Palette

Delete any step palettes that are no longer needed.

To delete a step palette:

1. Choose **Options>Palette Builder...**.Step Palette Builder appears (Figure 29-16).

OR

Right-click in a gray area around the toolbars or step palettes. A menu listing available step palettes appears (Figure 29-9). Choose **Palette Builder...**.Step Palette Builder appears (Figure 29-16).

- 2. Select the desired palette.
- 3. Choose **Delete**. The step palette is removed from the main editor.

**Note:** Deleting a step palette cannot be undone. There is no confirmation for deleting a step palette. If a step palette is accidentally deleted, choose **Cancel** in **Step Palette Builder**.

4. Choose OK.
# Index

#### **Numerics**

21 CFR Part 11 compliance. *See* Accounts & Permissions.

### Α

abort choosing to 25-10, 25-13 how to 25-13 Accept using in Hardware Setup 4-3 Account Management. See Accounts & Permissions Accounts & Permissions checking in methods 12-22 installing 2-2 logging onto 2-9 overview 2-1 permissions 2-6 setting the level of support 2-3 signing methods 12-26 system administration adding permissions 2-6 handling disabled accounts 2-10 overview restoring the administrator password 2-7 validating methods 12-24 viewing the audit log 2-8 viewing the logged on user's permissions 2-11 Active Wash defined 18-35

Add Device using in Hardware Setup 4-3 Advanced Manual Control defined 27-1 Advanced Step Palette displaying 17-3, 24-2 overview 17-1 steps displayed on 17-1 All defined 15-14 ALPs adding to deck 5-15 deleting from deck 5-14 properties defined in Deck Editor 5-16 setting properties in Deck Editor 5-16, 5-18 As Is defined 15-11 using 15-12 Aspirate Step configuration includes 16-4 configuring 16-6 defined 16-1 inserting into method 16-4, 16-5 inserting into pipetting template 10-21 pipetting template configuring 10-21 defined 10-21 inserting 10-21 using in 10-21 Auto-Select unchecking 9-45

### В

Bar Code reading options configuring in Present Step 23-13 Do not read bar code 23-13 Read bar code for decision making 23-13 Read bar code for logging 23-13 bar codes using data sets with 24-11 **Basic Step Palette** steps displayed on 15-1 Beckman Coulter Accounts & Permissions. See Accounts & Permissions Biomek 3000 Step Palette displaying 20-3 **Biomek Simulator** camera navigation buttons 4-7 configuring camera controls 4-9 configuring materials 4-11 configuring settings 4-4 previewing configuration changes 4-7 using camera navigation buttons 4-8 using lighting controls 12-49 using simulator controls 12-48 viewing a method run 12-46 Biomek software introduction 1-2 methods defined 1-2 related user's manuals 1-3 Blowout defined 9-11 defined in Transfer Step 12-14 Break Step configuring 21-21 defined 21-2 inserting into method 21-21 Levels defined 21-22 using 21-21 Bulk Dispense Step configuring 20-10 defined 20-1

### С

calibration of techniques 9-34 offset 9-35, 9-38 slope 9-35, 9-38 Cancel using in Hardware Setup 4-3 Cautions defined v listed ix Change Tool Step configuring 20-4 defined 20-1 check in discarding all project file changes since last check in 6-17 methods 12-22 project files 6-11 viewing revision history 12-27 check out methods 12-32 Cleanup Step defined 16-2, 16-30 inserting into method 16-31 Clear defined 15-11, 15-14 using 15-14 Clear All defined 15-14 in Instrument Setup defined 15-11 using 15-15 Combine Step configuring destination labware 15-44 configuring source labware 15-35 configuring Tip Handling 15-30 configuring Transfer Details 15-53 defined 15-1 four functions in 15-29 inserting into method 15-30 overview 15-29 Span-8 Pod using defined patterns 18-69 Comma-Separated Value Files (.csv) 23-5

Comment Step configuring 15-62 defined 15-2, 15-62 inserting into method 15-62 using in pipetting template 10-17 context variables defined 10-1 defining 10-10 table of 10-12 understanding 10-8 Continuation 25-25 characteristics of 25-25 defined 25-25 loading a 25-29 running a 25-29 snapping a 12-52, 12-57, 25-24, 25-25, 25-26 Create Data Set Step configuring 24-3 configuring File Options 24-4 configuring Input Options 24-9 configuring Output Options 24-13 Create Group Step configuring 21-4 defined 21-1 inserting into method 21-4 using 21-4 custom labware creating 15-26 cutoff velocity changing in Liquid Type Editor 8-8 changing in Technique Editor 9-33 defined 8-5, 9-32 equation to determine 8-8, 9-33 maximum 8-8, 9-32 minimum 8-8, 9-32 of syringes 9-32

### D

Data Set Management Step configuring 24-16 overview 24-16 Data Set Processing Step configuring 24-22 overview 24-22 using expressions in 24-24

data sets changing the properties of 24-20 copying 24-16 creating 14-3 creating from SILAS messages 22-42 definition 14-1 managing 24-16 overview 14-1, 24-1 performing sample tracking 14-2 processing 24-22 removing 24-19 renaming 24-17 scripting 28-1, 28-2 using in a method 14-4 using to define well patterns 14-5 using to specify volumes 14-8 Volume understanding 14-3 database table reading from to create data sets 24-5 deck adding ALPs to 5-15 adding positions to 5-15 creating a new 5-11 defined 5-1 deleting a 5-12 deleting ALPs from 5-14 deleting deck positions from 5-14 list in Deck Editor 5-10 modifying a 5-14 opening for viewing or modifying 5-10 populating with labware 15-9 renaming 5-13 restoring deck display 29-12 selection 5-2 setting ALP properties for 5-16 setting positions 5-16 setup confirmation 12-45 teaching (framing) defined 5-3 deck display toggling 29-12 Deck Editor accessing 5-4 defined 5-1 displayed 5-2 options defined 5-5

deck layout confirmation 15-6 selecting a 15-5 deck position defined in Deck Editor 5-16 deleting from deck 5-14 pause a 15-60 pausing a specific 15-60 setting properties in Deck Editor 5-19 default liquid type setting 8-11 **Define Pattern Step** advantages 18-64 configuring reading from a file 18-66 specify wells manually 18-65 defined 18-2 inserting into method 18-65 read from file specifying lines to use 18-69 using patterns in Transfer or Combine Steps 18-69 **Define Procedure Step** configuring 21-19 defined 21-2 inserting into method 19-6, 21-19 using 21-18 Define Procedure step creating HDR pin washing procedure 19-6 destination labware configuring in Combine Step 15-44 configuring in HDR Combine Step 19-13 configuring in HDR Transfer Step 19-13 configuring in Transfer From File Step 18-55 configuring in Transfer Step 15-44 defined 15-44, 18-55, 19-13 Details log contents 26-6 defined 26-1 overview 26-3 Detection 9-25 **Device** Action Step configuring 22-17 configuring for Drainable/Refillable Reservoir ALP 22-21

configuring for Microplate Shaking ALP 22-25 configuring for Positive Position ALP 22-28 configuring for Source/Waste Sensor 22-29 configuring for Speed Pump 22-30 configuring for Stirring ALP 22-31 configuring for Wash Station 384 ALP 22-27 configuring for Wash Station 96 ALP 22-27 defined 22-1, 22-16 inserting into method 22-16 Device Editor configuring devices with 22-7 **Device Setup Step** configuring 22-10 defined 22-1, 22-9 included in configuration 22-10 inserting into method 22-9 devices configuring using Device Editor 22-7 external using with Biomek FX 22-7 integration 22-2 **Devices Step Palette** displaying 22-3 integrating devices 22-1 overview 22-1 steps displayed on 22-1 **Dispense Step** configuring 16-10 defined 16-1, 16-9 inserting into method 16-10 inserting into pipetting template 10-25 pipetting templates configuring 10-25 defined 10-25 inserting 10-25 Do not Auto Select checking 9-8 Drainable/Refillable Reservoir ALP configuring Device Action step 22-21

# Ε

Else If Step substep 17-13 End If Step substep 17-13 error category configuration 25-3 defined 25-3 examples 25-3 recovery 25-5 light curtain 25-15 defined 25-15 examples 25-15 recovery 25-15 aborting the method 25-16 snapping a continuation 25-16 liquid level sensing 25-18 defined 25-18 recovery 25-18 manual control 25-11 defined 25-11 examples 25-11 recovery 25-11 multiple error messages 25-22 defined 25-22 examples 25-22 recovery 25-22 observed 25-23 defined 25-23 examples 25-23 recovery 25-23 pausing the method 12-51, 25-23 snapping a continuation 12-52, 25-24 stopping the method 12-52, 25 - 24run-time 25-12 defined 25-12 examples 25-12 recovery 25-13

validation 25-6 defined 25-6 examples 25-8 recovery 25-10 prevention Continuations 12-52, 25-24 Look ahead up to 1800 seconds in the method while it is running 25-8 overview 25-2 Single Step operations 25-30 validating a method before run 25-6 recovery options abort choosing to 25-13 defined 25-13 ignore choosing to 25-14 defined 25-13 overview 25-2 retry choosing to 25-14 defined 25-13 snap choosing to 25-14 defined 25-13 error handling overview 25-1 error message displayed in Method View 25-4, 25-5, 25-7 displayed in status bar 25-4, 25-5, 25-7 Errors log contents 26-6 defined 26-1 overview 26-4 export all methods associated with a project file 12-36 the currently open method 12-35 exporting project files 6-20

expressions defined 13-1 examples 13-6 rules for using 13-6 using 13-1, 13-6 using in the Data Set Processing Step 24-24 external devices configuring labware for 22-7 setting attributes 22-7 using SILAS Module 22-7 using SILAS Module Step to configure 22-39 using with Biomek FX 22-7

# F

**File Options** configuring in Create Data Set Step 24-4 **File Properties** configuring in Transfer From File Step 18-47 Finish Step 12-7 clearing devices of labware 12-9 clearing global variables 12-10 clearing instrument of labware 12-9 configuring 12-7 configuring a report in 12-11 defined 12-4 unloading and washing tips 12-9 Follow Liquid defined 9-11 defined in Transfer Step 12-14 framing (teaching) reasons for 5-3

### G

Group Step advantages to using 16-35 configuring 16-36 defined 16-2, 16-35 inserting into method 16-35 groups creating technique 9-39 using in a method 21-9

### Н

Hardware Setup accessing 4-2 overview 4-1 understanding the options in 4-3 using Accept in 4-3 using Add Device in 4-3 using Cancel in 4-3 using Home All Axes in 4-3 using Reconnect in 4-3 using Remove Device in 4-3 HDR Combine Step configuring destination labware 19-13 configuring source labware 19-7 defined 19-3 inserting into method 19-4 using 19-3 washing pins in 19-6 HDR Move Labware Step configuring 19-30, 19-31 defined 19-30 inserting into method 19-30 using 19-30 HDR pins defining wash procedure 19-6 washing between transfers 19-6 HDR Step Palette displaying 19-2 HDR steps defined 19-1 HDR System operating in software 19-1 using in a method 19-1 HDR Tool Cleaning Step cleaning operations 19-21 configuring drying operations 19-27 configuring wash operations 19-22 inserting into method 19-20 using 19-20 HDR Transfer Step configuring destination labware 19-13 configuring source labware 19-7 defined 19-3 inserting into method 19-4 using 19-3 washing pins in 19-6

history printing validated method run history 12-28, 12-29 viewing method revision 12-27 Home All Axes 4-3

## 

If Step configuring 17-14 defined 17-2, 17-13 inserting into method 17-13 substeps of 17-13 using in pipetting template 10-17 ignore choosing to 25-14 how to 25-14 import methods into projects 12-37 Import File exporting setting to 3-7 importing settings from 3-9 Import/Export Utility sharing instrument settings 3-6 importing project files 6-22 **Input Options** configuring in Create Data Set Step 24-9 matching plate bar code and well number 24-11 matching sample IDs 24-12 instrument file definition 1-1 restoring settings 3-5 settings 3-1 Instrument Setup Step changing during a method 15-11 defined 15-1 using additional 15-11 Intermediate Step Palette displaying 16-3 overview 16-1 steps displayed on 16-1

# J

Just In Time (JIT) Step configuring 17-10 defined 17-1, 17-9 inserting into method 17-10

# L

labware category defined 15-10 changing colors in Labware Type Editor 7-16 changing type properties in Labware Type Editor 7-14 copying a type in Labware Type Editor 7-11 creating custom 15-26 defining in Labware Type Editor 7-9 defining new types in Labware Type Editor 7-10 deleting a type in Labware Type Editor 7-12 populating deck with 15-9 properties 15-15 renaming a type in Labware Type Editor 7-13 specifying stackable items in Labware Type Editor 7-21 Labware Category Any defined 15-10 Custom defined 15-10 Lid defined 15-10 Reservation defined 15-10 Reservoir defined 15-10 TipBox defined 15-10 Titerplate defined 15-10 Tuberack defined 15-10 Labware Properties accessing 15-15, 23-6 accessing in Device Setup step 22-11 configuring for microplates 15-18 configuring for reservoirs 15-18 configuring for tube racks 15-18 configuring in Device Setup step 22-13

Known Volume defined 15-18, 22-14, 23 - 10Nominal Volume defined 15-18, 22-14, 23 - 10Unknown Volume defined 15-18, 22-14, 23-10 Labware Type Editor accessing 7-9 adding a new labware type in 7-10 categories in 7-14 changing colors of labware in 7-16 configuring a reservoir with added sections 7-18 configuring well properties in 7-26 copying a labware type in 7-11 deleting a labware type in 7-12 editing labware type properties in 7-14 overview 7-1 renaming a labware type in 7-13 specifying stackable items in 7-21 using Secure Stacking in 7-24 Let Step configuring 17-12 defined 17-1, 17-11 inserting into method 17-12 Liquid Level Sensing (LLS) defined 9-22, 9-27 errors Liquid Level is Incompatible with the Operation 25-20 Liquid Level Sensing Returning an Unstable Value 25-19 No Liquid is Detected 25-19 overview 25-18 recovering from 25-18 sensitivity defined 8-6, 9-23, 9-27 Liquid Type Editor accessing 8-4 configuration view 8-2 cutoff velocity defined 8-5 defined 8-1 liquid type view 8-2 overage defined 8-5

sensitivity defined 8-6 using the 8-3 liquid types copying/pasting in Liquid Type Editor 8-10 creating new 8-1 creating new in Liquid Type Editor 8-4 default 8-1 defined 8-3 overriding values using Technique Editor 9-30 removing/deleting from Liquid Type Editor 8-10 setting default type in Liquid Type Editor 8-11 understanding 8-1 ListMath defined 10-15 using in pipetting template steps 10-16 Load Tool Step configuring 20-5 defined 20-1 log types 26-1 Details log contents 26-6 defined 26-1 overview 26-3 Errors log contents 26-6 defined 26-1 overview 26-4 Pipetting log contents 26-7 defined 26-1 overview 26-4 Span-8 Pipetting log contents 26-7 defined 12-58, 26-1 overview 26-4 Span-8 Transfer log contents 26-8 defined 12-58, 26-1 overview 26-5 Unified Pipetting log contents 26-9 defined 26-2 overview 26-5

Unified Transfer log contents 26-11 defined 26-2 overview 26-5 logging configuring 12-58 defined 12-58 using 12-58 logging off system 2-11 logging onto system 2-9 logs content 26-6 defined 26-1 generating 26-13 generating data 26-1 selecting to generate 26-13 viewing 26-14 Look ahead up to 1800 seconds in the method while it is running defined 25-8 enabling 25-8 Loop Step configuring 16-24 creating a new variable in 16-24 defined 16-2, 16-23 End 16-23 incrementing 16-24 inserting into method 16-23 using existing variable in 16-27 using in pipetting template 10-17 using variables in 16-25 variables defined in 16-23

### Μ

main editor changing appearance of 29-1 displaying Step Palettes 29-11 toggling toolbars 29-11 Manual Control accessing 27-2 defined 27-1 dialog display 27-1, 27-2 method 12-1 checking in a 12-22 checking out a 12-32 copying steps in a 12-16 creating a new 12-3

creating techniques within 9-45 cutting steps in a 12-16 defined 1-2 deleting steps in a 12-17 exporting all associated with a project file 12-36 exporting the currently open 12-35 halting without light curtain violation 12-51, 25-23 importing into projects 12-37 improving way method is run 12-56 inserting steps in a 12-13 modifying techniques through steps 9-45 opening a saved 12-30, 12-40, 12-41 opening a validated 12-31, 12-33 pasting steps in a 12-16 printing 12-28, 12-29 printing a hardcopy of 12-55 purging air before a run with a Span-8 Pod 12-54 running at slow speed 12-56 saving a 12-20 scripting in 28-1 signing 12-26 stopping a 12-52, 25-24 using groups in 21-9 using Redo when building a 12-18 using Undo when building a 12-18 using variables in 13-4 validating 12-24 validating before a run 25-6 viewing revision history 12-27 Microplate Shaking ALP configuring in Device Action Step 22-25 Mix defined 9-11 defined in Transfer Step 12-14 Mix Step configuring 16-15 defined 16-1, 16-14 inserting into method 16-15 inserting into pipetting templates 10-28 pipetting templates configuring 10-28 defined 10-28 inserting 10-28

Move Labware Step configuring 15-56, 15-57 defined 15-1, 15-56 inserting into method 15-56 Move Pod Step configuring 16-34 defined 16-2 inserting into method 16-33 Multiple Dispense conditioning techniques for 9-36 conditioning volumes defined 9-37

### Ν

New Tips Step defined 16-2, 16-19 inserting into method 16-20 when to use 16-19 Next Item Step breaking out of current loop 21-15 configuring 21-14 defined 21-1 inserting into method 21-14 setting variable to value 21-15 starting over with first value 21-15 using 21-13 Next Labware Step configuring 21-7 defined 21-1 inserting into method 21-7 using 21-7

### 0

Obstacle Avoidance 15-3 offsets changing per-labware 5-21 Output Options configuring in Create Data Set Step 24-13 setting data set properties 24-15 overage defined 8-5

### Ρ

Passive Wash defined 18-35 patterns using in Transfer or Combine Steps 18-69 Pause Step configuring 15-61 configuring for a pod 15-60 configuring for specific deck position 15-60 defined 15-2, 15-58 inserting into method 15-59 inserting into pipetting templates 10-34 pausing whole system 15-61 pipetting templates configuring 10-34 defined 10-34 inserting 10-34 pausing a specific deck position 15-60 pod 15-60 the whole system 15-61 without light curtain violation 12-51, 25 - 23per-labware offset changing 5-21 Per-Plate HTML Files report style choosing 24-26 Per-Plate Text Files report style choosing 24-26 Piercing 9-29 Pipetting log contents 26-7 defined 26-1 overview 26-4 **Pipetting Technique** selecting a 12-14 Pipetting Template Editor accessing 10-2 Aspirate Air Gap Step configuring 10-18, 10-20 defined 10-20 inserting into pipetting template 10-20 using in pipetting template 10-20 Axes Move Step configuring 10-33 defined 10-33 inserting into pipetting template 10-33 **Bulk Dispense Step** configuring 10-35 defined 10-35 inserting into pipetting template 10-35

**Clot Detection Step** configuring 10-18 Dispense Air Gap Step configuring 10-24 defined 10-24 inserting into pipetting template 10-24 Moving Aspirate Step configuring 10-22 configuring final position 10-23 configuring initial position 10-23 defined 10-22 inserting into pipetting template 10-22 Moving Dispense Step configuring 10-26 configuring final position 10-27 configuring initial position 10-27 defined 10-26 inserting into pipetting template 10-26 overview 10-1 Prewet Step configuring 10-30 defined 10-30 inserting into pipetting template 10-30 Tip Touch Step configuring 10-31 defined 10-31 inserting into pipetting template 10-31 pipetting templates associating with technique 10-5 configuring steps in 10-2, 10-17 context variable reference 10-10 context variables in 10-8 copying and pasting 10-7 creating 10-2 creating new 10-4 default 10-1 defined 10-1 defining context variables 10-10 inserting variables 10-8 liquid type context information 10-11 modifying existing 10-7 renaming 10-7 understanding context variables 10-8 using Aspirate Air Gap Step in 10-20 using Aspirate Step in 10-21 using Axes Move Step in 10-33

using Bulk Dispense Step in 10-35 using Comment Step in 10-17 using Dispense Air Gap Step in 10-24 using Dispense Step in 10-25 using If Step in 10-17 using ListMath in 10-16 using Loop Step in 10-17 using Mix Step in 10-28 using Moving Aspirate Step in 10-22 using Moving Dispense Step in 10-26 using Pause Step in 10-34 using Prewet Step in 10-30 using Tip Touch Step in 10-31 viewing existing 10-3 pod Home All Axes 4-3 setup options 15-7 Pod Setup Options 15-7 Positive Position ALP configuring in Device Action Step 22-28 Preferences accessing 29-4 using to change main editor appearance 29-1 Present Step configuring bar code reading instructions 23-12 configuring from which hotel 23-12 Present step defined 23-1 Prewet defined 9-11 defined in Transfer Step 12-14 printing hardcopy of method 12-55 Procedures advantages of using 21-18 collapsing 21-18 defining 21-18 expanding 21-18 running 21-18 project file importing 6-22 renaming the active file 6-8

project files checking in 6-11 contents of 6-3 creating copies of 6-10 creating new 6-4 defined 1-2 deleting 6-6 discarding all changes since last check in 6-17 exporting 6-20 exporting all methods associated with 12-36 opening 6-5 overview 6-1 renaming saved files 6-9 restoring deleted 6-7 restoring deleted project items 6-14 reverting individual project items to a specific revision 6-19 reverting project contents to a previous state 6-15 viewing contents 6-13 Purge Wash Tool Step configuring 20-13 defined 20-2

## R

Reconnect using in Hardware Setup 4-3 Redo using in method building 12-18 **Remove Device** using in Hardware Setup 4-3 Remove to Waste Step defined 20-1 Remove to Waste step configuring 20-8 report styles Per-Plate HTML Files choosing 24-26 description 24-27 Per-Plate Text Files choosing 24-26 description 24-28

configuring 24-26 description 24-28 Text File choosing 24-26 description 24-26 types 24-26 Reporting step configuring 24-25 overview 24-25 reservoirs configuring with sections in Labware Type Editor 7-18 restoring deck display 29-12 instrument settings 3-5 instrument settings from backup files 3-10 project items 6-14 retry choosing to 25-14 how to 25-14 revisions. See check in Run Method Step advantages to using 17-4 configuring 17-5 defined 17-1, 17-4 inserting into method 17-5 Run Procedure Step configuring 21-20 defined 21-2 inserting into method 21-20 using 21-20 Run Program Step configuring 21-11 defined 21-1 inserting into method 21-10 using 21-10 Runtime Data configuring 22-41

SQL Server

choosing 24-26

li

### S

sample IDs matching in data sets 24-12 sample tracking definition 14-1 example using data sets 14-2 overview 14-1 Script Step configuring 17-17 defined 17-2, 17-15 inserting into method 17-16 Scripted Let Step configuring 17-20 defined 17-18 Extend defined 17-18, 28-2 extending variables 17-18 inserting into method 17-19 WeakExtend defined 17-18, 28-2 scripting data sets 28-1, 28-2 in methods 28-1 Secure Stacking configuring 7-25 stacking rules in 7-24 using 7-24 sensitivity defined 8-6 Serial Dilution Step collapsing diluent properties 18-12 configuring 18-6 defined 18-1 dilution ratio defined 18-13 expanding diluent properties 18-12 inserting into method 18-5 Set Global Step configuring 21-16 defined 21-2 inserting into method 21-16 using 21-16 signing methods 12-26 SILAS messages creating data sets from 22-42 SILAS Module Step configuring 22-39 configuring runtime data 22-41 defined 22-2 inserting into method 22-39

Single Step defined 25-30 example 25-30 opening 25-30 running 25-31 slope calibration 9-35, 9-38 snap choosing to 25-10, 25-14 how to 25-14 **Snap Continuation** using 12-57 snapping a Continuation 12-57 source labware configuring in Combine Step 15-35 configuring in HDR Combine Step 19-7, 19-14 configuring in HDR Transfer Step 19-7, 19-14 configuring in Transfer From File Step 18-51 configuring in Transfer Step 15-35, 15-45 Source/Waste Sensor configuring Device Action step 22-29 Span-8 Aspirate Step aspirating individual volumes 18-20 configuration 18-16 configuring 18-18 defined 18-1 inserting into method 18-17 Span-8 Dispense Step configuring 18-25 defined 18-2, 18-23 dispensing individual volumes 18-27 included in configuration 18-23 inserting into method 18-24 Span-8 New Tips Step configuring 18-32 defined 18-2, 18-31 inserting into method 18-31 Span-8 Pipetting log contents 26-7 defined 12-58, 26-1 overview 26-4 sample IDs added to 26-4, 26-5

Span-8 Pod purging air from the syringes and tubing 12-54 Span-8 Step Palette defined 18-1 displaying 18-3 steps displayed on 18-1 Span-8 Tip Discard Step configuring 18-34 defined 18-2 inserting into method 18-33 when to use 18-33 Span-8 Transfer log defined 12-58, 26-1 overview 26-5 Span-8 Wash Tips Step Active Wash configuring 18-39 defined 18-39 configuring a passive wash 18-36 configuring an active wash 18-39 defined 18-2 included in configuration 18-35 inserting into method 18-36 Passive Wash configuring 18-37 defined 18-36 using to purge system 18-37 SPE (Solid Phase Extraction) ALP defined 22-4 SPE Step configuring 22-5 defined 22-1, 22-4 inserting into method 22-5 Specialty Step Palette displaying 21-3 overview 21-1 **Specialty Steps** defined 21-1 Speed Pump configuring in Device Action Step 22-30 SOL Server report style configuring 24-29 Stacker Carousel Step Palette displaying 23-2 overview 23-1

Stacker Carousel, Biomek instrument using the Present Step with 23-12 using the Stacker Setup Step with 23-3 using the Store Step with 23-14 Stacker Setup Step configuring 23-3 defined 23-1 Start Step configuring 12-4 defined 12-4 using 12-4 Step Palette Builder accessing 29-19 using 29-18 **Step Palettes** adding configured steps 29-22 changing properties of 29-23 creating 29-20 creating new 29-23 customizing 29-14, 29-20 deleting 29-24 modifying 29-18 rearranging steps on 29-22 removing steps from 29-22 toggling 29-11 steps adding to Step Palettes 29-22 changing properties for identification on Step Palettes 29-16 copying 12-16 cutting 12-16 deleting 12-17 disabling 12-53 enabling 12-53 modifying caption in Method View 29-17 modifying techniques through 9-45 pasting 12-16 rearranging on Step Palettes 29-22 removing from Step Palettes 29-22 Stirring ALP configuring in Device Action Step 22-31 Store Step configuring 23-14 defined 23-1

submethod defined 17-4 syringes cutoff velocity 9-32

### Т

teaching (framing) reasons for 5-3 **Technique Browser** accessing 9-3 **Technique** Editor Aspirate tab 9-12 Calibration tab 9-34, 9-35, 9-37, 9-38 Dispense tab 9-14, 9-15 General tab 9-11 Liquid Level Sensing tab 9-19 Liquid Type tab 9-30, 9-31 Mix tab 9-17 setting values 9-9 **Technique Properties** setting 9-7 Techniques associating pipetting template with 10-5 auto-selection of 15-42, 15-51 calibrating 9-34 conditioning 9-36 context information defined 9-4 creating groups 9-39 creating new 9-5 creating within a method 9-46 customizing 15-43, 15-51 defined 9-1, 12-14 displaying available 9-4 how they work 9-1 icons defined 9-11 identifying 9-4 liquid level sensing error handling settings 9-23 general settings 9-20, 9-26 repetition settings 9-21 sensitivity settings 9-22, 9-27 modification restrictions 9-45 modifying through a method step 9-46 operation icons 9-11 overriding liquid type values 9-30 properties 9-5

removing/deleting 9-43 saving 15-43, 15-51 saving custom 9-48 selecting a pipetting 12-14 selecting manually 9-45 selection in Combine Step 15-41 selection in Transfer From File Step 18-53 selection in Transfer Step 15-41 selection within a method 9-46 setting aspirate values 9-12 setting clot detection values 9-25 setting dispense values 9-14 setting liquid level sensing values 9-19, 9-25, 9-29 setting mix after dispense values 9-16 setting mix before aspirate values 9-14 setting mix values 9-17 setting piercing values 9-29 setting properties 9-7 setting values for 9-9 setting values in Technique Editor 9-9 understanding 9-1 viewing description 9-10 viewing properties of 9-4 why they are valuable 9-2 Techniques, Modifying Saved 9-44 text file reading from to create data sets 24-4 Text File report style choosing 24-26 Then If Step substep 17-13 Tip Discard Step (Span-8 Pod) configuring 18-34 **Tip Handling** configuring in Combine Step 15-30 configuring in Transfer From File Step 18-43 configuring in Transfer Step 15-30 Tip Loader ALP step defined 22-2 Tip Loader Step configuring 22-38 defined 22-37 inserting into method 22-37

tip properties configuring in Device Setup step 22-11 **Tip Touch** defined 9-11 defined in Transfer Step 12-14 Tip Type Editor accessing 7-4 adding new tips in 7-5 changing tip properties in 7-6 overview 7-1 removing tips in 7-8 tips accessing properties 15-15 adding new in Tip Type Editor 7-5 capacity diagram 7-7 changing properties in Tip Type Editor 7-6 defining in Tip Type Editor 7-3 properties defined 7-6 properties diagram 7-7 properties displayed in Tip Type Editor 7-4 removing from type list in Tip Type Editor 7-8 type list displayed 7-4 Toggle defined 15-11 using 15-12 toolbars 29-11 options 29-12 toggling 29-11 **Trailing Air Gap** defined 9-11 defined in Transfer Step 12-14 **Transfer Details** configuring in Combine Step 15-53 configuring in Transfer From File Step 18-62 configuring in Transfer Step 15-53 Transfer From File Step configuring destination labware 18-55 configuring file properties 18-47 configuring source labware 18-51 configuring tip handling in 18-43 configuring transfer details 18-62 defined 18-2 inserting into method 18-42

Transfer Step configuring destination labware 15-44 configuring source labware 15-35 configuring Tip Handling 15-30 configuring Transfer Details 15-53 defined 15-1 four functions in 15-29 inserting into method 15-30 overview 15-29 selecting a pipetting technique 12-14 Span-8 Pod using defined patterns 18-69

# U

Undo using in method building 12-18 Unified Pipetting log contents 26-9 defined 26-2 overview 26-5 Unified Transfer log contents 26-11 defined 26-2 overview 26-5 Unload Tips Step defined 16-21 inserting into method 16-22 when to use 16-21 Unload Tips Step (Multichannel Pod) defined 16-2 Unload Tool Step configuring 20-7 defined 20-1

# V

Validate Before Run defined 25-6 enabling 25-8 validated methods opening 12-31, 12-33 printing run history 12-28, 12-29 validating methods 12-24 variables advantages to using 13-1 configuring 12-4, 13-3 context defining 10-10 table of 10-12 understanding 10-8 creating in Loop step 16-24 creating in the Start Step 12-4 defined 13-1 defined in Loop Step 16-23 defined in Run Method 13-3 extended from Scripted Let step 17-20 inserting into pipetting templates 10-8 naming restrictions 12-6 using 13-1, 13-2 using an existing variable in Loop Step 16-27 using for aspirate 13-2 using in a method 13-4 using in Loop Step 16-25 using in Transfer From File Step 18-57 using in Transfer Step 15-50 Volume data set understanding 14-3 volumes specifying with data sets 14-8

### W

Warnings defined v listed ix Wash Plate Step configuring 20-15 defined 20-2 Wash Step configuring 16-18 defined 16-1, 16-17 inserting into method 16-17 Well Pattern Editor accessing 11-2 copying a well pattern in 11-5 creating a new well pattern in 11-3 deleting a well pattern in 11-6 overview 11-1 renaming a well pattern in 11-7 well patterns defining with data sets 14-5 well properties configuring in Labware Type Editor 7-26 Worklist Step advantages to using 17-7 configuring 17-7, 17-8 defined 17-1, 17-6 inserting into method 17-7 using text file in 17-6 using values 17-6